



RULEMAKING ISSUE

November 26, 1990

(Affirmation)

SECY-90-387

For: The Commissioners

From: James M. Taylor, Executive Director for Operations

Subject: TRANSMITTAL OF REVISED PART 20

To transmit to the Commission the corrected version of the Purpose: revised Part 20 rule for final approval.

Category: Major action fo. affirmation.

Discussion:

On July 30, 1990, the Secretary issued a Staff Requirements Memorandum (SRM) containing the Commission's approval for issuing the revised 10 CFR Part 20 as a final rule. The SRM contained eight areas where additions, corrections, or other changes were requested. The modified Federal Register Notice is Enclosure A to this paper. Enclosure B contains copies of the SRM and the Federal Register Notice marked to indicate where all changes have been made to Enclosure 3 (Statement), Enclosure 4 (Rule), and Enclosure 5 (Appendices) of SECY-88-315. Enclosure B also contains a marked-up version of Enclosure 6 to SECY-88-315 (listing of changes from the proposed rule) as the SRM requested that a check be performed on whether all of these changes were addressed in the statement of considerations. Changes resulting from the annotated copy of SECY-88-315 that accompanied the SRM were also made and are identified in the markup. The sections relating to the implementation date have been revised to conform with the SRM of November 20, 1990 on the Part 20 implementation date.

Enclosure C is the final Regulatory Analysis (including the flexibility analysis). Enclosure D is the final Environmental Assessment, Enclosure E is the final Backfit Analysis, Enclosure F contains the letters to congressional committees informing them of the issuance of this rule and providing copies to them, and Enclosure G is the Press Release for the press briefing announcing the issuance of the revised Part 20.

NOTE: TO BE MADE PUBLICLY AVAILABLE WHEN THE FINAL SRM IS MADE AVAILABLE

DFOZ

Contact: H. T. Peterson, Jr., RES 49-23640

> 9012130388 901126 90-387 SEC1 PDC

The revised Part 20 contains the changes to SECY-88-315 that have been identified in SECY-89-267 and SECY-90-237 in addition to those changes requested in the SRM of July 30, 1990. We have also modified the effective date to be thirty days after publication in the Federal Register, except that the information collection requirements (recordkeeping and reporting) will not be effective until publication of the OMB approval.

The modifications have been reviewed by the Part 20 Working Group and Steering Committee and the entire document has been reviewed by a technical editor. An earlier draft incorporating the SECY-89-267 changes was reviewed by staff of the Regulatory Publications Branch, DFIPS, ADM and the Information & Records Management Branch, IRMB, IRM and their suggested changes have been incorporated into the enclosed Federal Register Notice.

The extensive table of radionuclide intake and concentration limits in Appendix B was compared with a pre-publication draft of Federal Guidance Report #11 (FRP #11) which also contains derived air concentrations and annual limits for occupational exposure. Adjustments were made by NRC and EPA to both sets of tables so that they are mutually consistent. Because of this, the Statement (Preamble) endorses the use of FRP #11 for calculating doses and determining compliance with 10 CFR Part 20 occupational limits (FRP #11 does not address exposure of members of the general population).

The changes made subsequent to the July 30, 1990 Staff Requirements Memorandum are as follows (item numbers correspond to items in the SRM):

- Replacement of the discussion of the Backfit Analysis in the statement with the revised text of the Commission's determination from staff's final backfit analysis and adding the paragraph from item #1 of the Staff Requirements Memorandum to both the FRN discussion and the final Backfit Analysis (Enclosure F).
- 2. (a)Modification of the effective date of the rule for NRC licensees from January 1, 1992 (SECY-88-315) to 30 days after publication, so that the January 1, 1991 codification of the existing Part 20 can be used by licensees who do not choose early implementation. The final date of implementation for NRC licensees is January 1, 1993.

- 2.(b)Addition of text to pages 40 and 54 to indicate that flexibility is provided for more precise dose evaluations, but this provides the same degree of health protection.
- 3. Modification of Section V of the Statement of Considerations (Preamble) and addition of a revised § 20.8 and a new § 20.9 (old § 20.8) to provide for early implementation of the revised Part 20 and guidance on relationship to license conditions and technical specifications based on the existing Part 20.
- 4. The basis of the dose design criterion for generallylicensed radioactive devices was not changed by the conforming amendments to Part 32. The design criterion remains at 10 percent of the occupational dose limit or 500 millirem per year. This design criterion is no longer equal to the dose limit for members of the general public (now 100 millirem per year). Design criteria for generally-licensed radioactive devices are being considered in connection with the reexamination of exemptions and general license conditions (See SECY-90-175 of May 14, 1990 and the Staff Requirements Memorandum of August 13, 1990).
- 5. The Federal Register Notice has been updated to reflect issuance of the Commission's Policy Statement on Below Regulatory Concern. These updates appear in the discussion of ongoing related activities or page 15 and in the discussion of public comments on BRC levels on page 25 of the Preamble (Statement of Considerations).
- 6. Staff has replaced the definition of 'natural background" with "background radiation" and included residual global fallout and radon in ambient concentrations within this definition. "Global fallout" could not be encompassed within the scope of "natural background" as it is manmade. Sources of radiation considered to be "background radiation" are excluded from coverage under Part 20 (See §§ 20.1 and 20.2).
- 7. As noted above, quality control checks have been carried out by comparison with the July 30, 1990 SRM; with the Enclosures in SECY-88-315, especially with Enclosure 6; with Enclosure 3 of SECY-89-267; and with SECY-90-237. A marked copy of Enclosure 6 (changes from the proposed rule) is in Enclosure B.

ENCLOSURE A FEDERAL REGISTER NOTICE

Enclosure A

à

[7590-01]

#### NUCLEAR REGULATORY COMMISSION

10 CFR PART 20

and 10 CFR Parts 2, 19, 31, 32, 34, 35, 39, 40, 50, and 61

RIN 3150 - AA38

Standards for Protection Against Radiation

AGENCY: Nuclear Regulatory Commission.

ACTION: Final rule.

SUMMARY: The Nuclear Regulatory Commission (NRC) is revising its standards for protection against ionizing radiation. This action is necessary to incorporate updated scientific information and to reflect changes in the basic philosophy of radiation protection. The revision conforms the Commission's regulations to the Presidential Radiation Protection Guidance to Federal Agencies for Occupational Exposure and to recommendations of national and international radiation protection.

EFFECTIVE DATE: This regulation becomes effective on (30 days after publication in the Federal Register). However, licensees may defer implementation of this rule until January 1, 1993. The information collection requirements are not effective until NRC publishes the OMB Clearance in the Federal Register.

ADDRESS: Copies of documents relating to the January 9, 1986 proposed rule (51 FR 1092) or this document may be examined and copied for a fee in the Commission's Public Document Room at 2120 L Street NW (Lower-Level), Washington, DC 20555.

FOR FURTHER INFORMATION CONTACT: Harold T. Peterson, Jr., Division of Regulatory Applications, Office of Nuclear Regulatory Research, U.S. Nuclear Regulatory Commission, Washington, DC 20555, Telephone (301)492-3640.

#### SUPPLEMENTARY INFORMATION:

#### I. Introduction

A. Purpose of the Revision

The purpose of this revision of 10 CFR Part 20 is to modify the NRC's radiation protection standards to reflect developments in the principles and scientific knowledge underlying radiation protection that have occurred since Part 20 was originally issued more than 30 years ago. These developments not only include updated scientific information on radionuclide uptake and metabolism, but also reflect changes in the basic philosophy of radiation protection. Incorporation of these changes will ensure that Part 20 continues to provide adequate protection of public healtn and safety.

It is also the purpose of this revision to implement the 1987 Presidential guidance on occupational radiation exposure (see Section II.D). The Atomic Energy Commission (AEC) and the NRC have followed past Federal radiation protection guidance, and conformance with the guidance is viewed by the Commission as being necessary to ensure that NRC licensees are using levels of protection comparable to those used by Federal agencies.

The AEC and the NRC have generally followed the basic radiation protection recommendations of the International Commission on Radiological Protection (ICRP) and its U.S. counterpart, the National Council on Radiation Protection and Measurements (NCRP), in formulating basic radiation protection standards. In 1977, ICRP issued revised recommendations for a system of radiation dose limitation. This system, which was described in ICRP Publication 26,<sup>1</sup> introduced a number of significant modifications to existing concepts and recommendations of the ICRP and the NCRP that are now being incorporated in the NRC regulations. In particular, this revision of Part 20 puts into practice

Recommendations of the International Commission on Radiological Protection, January 13, 1977, ICRP Publication No. 26 (1977). (Available for sale from Pergamon Press, Inc., Elmsford, NY 10523.)

recommendations from ICRP Publication 26 and subsequent ICRP publications. The Federal radiation protection guidance signed by the President on January 20, 1987, is also based upon the ICRP 1977 recommendations in ICRP Publication 26.

In adopting the basic tenets of the ICRP system of dose limitation, the Nuclear Regulatory Commission recognizes that, when application of the dose limits is combined with the principle of keeping all radiation exposures "as low as is reasonably achievable," the degree of protection could be significantly greater than from relying upon the dose limits alone.

B. Fundamental Radiation Protection Principles

The radiation protection standards in this part are based upon the assumptions that--

(1) Within the range of exposure conditions usually encountered in radiation work, there is a linear relationship, without threshold, between dose and probability of stochastic health effects (such as latent cancer and genetic effects) occurring;

(2) The severity of each type of stochastic health effect is independent of dose; and

(3) Nonstochastic (nonrandom) radiation-induced health effects can be prevented by limiting exposures so that doses are below the thresholds for their induction.

The first assumption, the linear nonthreshold dose-effect relationship, implies that the potential health risk is proportional to the dose received and that there is an incremental health risk associated with even very small doses, even radiation doses much smaller than doses received from naturally occurring radiation sources. These health risks, such as cancer, are termed stochastic because they are statistical in nature; i.e., for a given level of dose, not every person exposed would exhibit the effect. The second assumption means that when a stochastic effect is induced, the severity of the effect is not related to the radiation dose received. The third assumption implies that there are effects, termed nonstochastic effects, for which there is an apparent threshold; i.e., a dose level below which the effect is unlikely to occur. An

example of a nonstochastic effect is the formation of radiation-induced cataracts of the eyes.

The above assumptions are necessary because it is generally impossible to determine whether or not there are any increases in the incidence of disease at very low doses and low dose rates, particularly in the range of doses to members of the general public resulting from NRC-licensed activities. It is firmly established, both from animal studies and human epidemiological studies (such as those of the radium dial painters, radiologists, and the atomic bomb survivors) that there is an increased incidence of certain cancers associated with radiation exposure at high doses and high dose rates. However, whether these effects occur at very low doses and, if they occur, whether their occurrence is linearly proportional to dose are not firmly established. This creates considerable uncertainty in the magnitude of the risk at low doses and low dose rates. There is no clear human evidence of radiation-induced genetic damage to the children of irradiated parents. Such effects are inferred from studies of mice and nonmammalian species (e.g., fruit flies).

In the absence of convincing evidence that there is a dose threshold or that low levels of radiation are beneficial, the Commission believes that the assumptions regarding a linear nonthreshold dose-effect model for cancers and genetic effects and the existence of thresholds only for certain nonstochastic effects remain appropriate for formulating radiation protection standards and planning radiation protection programs.

## C. Background

Standards for radiation protection were originally issued by the former AEC in the late 1950s (22 FR 548, January 29, 1957) and republished in 1960. These standards have been modified since that time by a series of amendments relating to specific issues; however, no complete revision of Part 20 has been made since the original standards were issued.

The MRC issued an Advance Notice of Proposed Rulemaking (ANPRM) in the Federal Register of March 20, 1980 (45 FR 18023). This ANPRM requested comments

on possible topics that should be revised in a proposed revision of Part 20. The responses received to this announcement were considered in the formulation of the proposed revision.

During the development of this rule, early comments from licensees, labor unions, public interest groups, other Federal agencies, and scientific organizations were solicited, discussed, and considered in formulating the proposed rule. In addition, the NRC staff has benefited from its participation in several public meetings held by the Environmental Protection Agency (EPA) in connection with the guidance for occupational radiation exposure. The revised Part 20 and the Federal guidance on occupational exposure were developed in parallel and are both based primarily on the ICRP recommendations. The comments made in these EPA-sponsored meetings and those received by EPA on the draft guidance published by EPA in the January 23, 1981 Federal Register (46 FR 7836) were reviewed by the NRC staff and considered in preparing the proposed Part 20.

The NRC published the proposed revision of the 10 CFR Part 20 rule in the January 9, 1986 Federal Register (51 FR 1092). More than 800 sets of public comments were received on the proposed revision. The public comments on the proposed revision were categorized, analyzed, and taken into account in developing the final rule. The principal public comments and the NRC staff responses to them are discussed in Section VI.

II. Developments Since the Proposed Revision Was Issued

### A. ICRP 1985 Paris Meeting

In March 1985, the International Commission on Radiological Protection (ICRP) held a meeting in Paris, France, to review the work of the various ICRP task groups and committees. One of the outcomes of this meeting was an ICRP statement<sup>2</sup> that the ICRP intended the principal dose limit for members of

International Commission on Radiological Protection, "Statement from the 1985 Paris Meeting of the [ICRP]," British Journal of Radiology, Vol. 58, page 910: 1985; also Health Physics, 48(6): 828-829 (June 1985)

the general public to be 1 millisievert (100 millirems) in a year, rather than 5 millisieverts (500 millirems). This clarification has been taken into account for the limits adopted for members of the public in the final rule and is discussed more fully in the discussion on § 20.301.

A second recommendation of the ICRP made at that time concerned the appropriate quality factor for converting the absorbed dose from neutrons (in rads or grays) to a dose equivalent (in rems or sieverts). The ICRP statement recommended increasing the quality factor for high-energy neutrons by a factor of 2. The quality factor for fast neutrons, for example, would be increased from 10 to 20. This change has the effect of doubling the apparent biological effectiveness of high-energy neutrons. For reasons explained in the discussion of quality factors (see the discussion of § 20.4), the NRC has not adopted this recommendation in this revision of Part 20.

#### B. ICRP 1987 Washington Meeting

The primary focus of the statement issued by the ICRP following the 1987 meeting in Washington<sup>3</sup> was ICRP Publication No. 48.<sup>4</sup> That publication discussed higher transfer factors for transport of certain transuranic elements across the intestinal walls. These higher fractional absorption factors have been incorporated in revisions to the annual limits on intake (ALIs) and derived air concentrations (DACs) in Appendix B of the final rule. The changes resulting from the use of these revised factors would not change either the ingestion or inhalation ALIs for plutonium in the oxide or nitrate forms, but would lower the ALIs for other compounds or mixtures by a factor of 10. The transfer factor for the gut transfer of neptunium was found to be an order of magnitude lower than the value used in ICRP-30 and, consequently, the ingestion ALI can be increased by almost an order of magnitude. The transfer factors for

<sup>3</sup> International Commission on Radiological Protection, "ICRP Statement from 1987 Washington Meeting," Health Physics 53(3): 335-342 (1987).

<sup>4</sup> International Commission on Radiological Protection, "The Metabolism of Plutonium and Related Elements," ICRP Publication No. 48 (Available for sale from Pergamon Press, Inc., Elmsford, NY 10523.) (1986).

americium, curium, and californium were found to be a factor of 2 higher than the ICRP-30 value so the ingestion ALIs are reduced by a factor of 2. Parameters applicable to inhalation ALIs and DACs are less affected than the ingestion ALIs as the transfer from the gastrointestinal (GI) tract to the blood for these radionuclides generally is less significant than transfer from the lung to the blood.

## C. ICRP 1987 Como Meeting

Following its 1987 meeting in Como, Italy, the ICRP issued a statement<sup>5</sup> that reviewed the existing estimates of the biological risks of ionizing radiation and, in particular, the preliminary data from the reanalysis of the Hiroshima-Nagasaki atomic bomb followup studies. Reanalysis of these data indicated that the risks from gamma radiation are approximately a factor of 2 higher than previous estimates for the general population and are also higher, but by a smaller factor, for workers. The ICRP concluded in 1987 that this information alone was "not considered sufficient at that time to warrant a change in the dose limits for occupational exposure and, for the general population, the increase in risk indicated by the new data is not considered to require an immediate change in the recommended dose limits, following the reduction by the ICRP (in 1985) in the principal limit from 5 to 1 mSv in a year (from sources other than medical and natural background radiation)." The ICRP also noted that the potential higher risks indicated by the reanalysis of the atomic bomb data should not be a major consideration as the dose limits should not be of primary importance in controlling doses if the principle of keeping radiation exposures "as low as is reasonably achievable" is being practiced. This position has since been modified by the ICRP 1990 Statement (see Section II.I below).

D. Federal Radiation Protection Guidance on Occupational Exposure

On January 20, 1987, President Reagan approved revised guidance to Federal agencies for occupational radiation protection. This guidance, which was

<sup>5</sup> International Commission on Radiological Protection, "Statement from the 1987 Como Meeting of the [ICRP]," Health Physics, 54(1): 125-132 (1988).

published in the Federal Register (52 FR 2822; January 27, 1987), generally adopts the philosophy and methodology of ICRP Publications 26 and 30. The Part 20 revision was developed in parallel with the development of the guidance. Because of this parallel development, the proposed Part 20 rule conformed with the draft Federal guidance available at the time the proposed Part 20 rule was written. However, because of changes made to both the draft guidance and the draft Part 20 revision, there were a few differences between the guidance in its final published form and the proposed Part 20 revision. As discussed in the respective sections below, changes to the proposed rule have been made in order to implement the final version of the Federal guidance.

E. NCRP Report No. 91

On June 1, 1987, the National Council on Radiation Protection and Measurements (NCRP) issued a report<sup>6</sup> containing updated NCRP recommendations for radiation protection limits. These recommendations replace recommendations published in 1971. The majority of these recommendations are in accord with the 1977 recommendations of the ICRP and, consequently, were already reflected in the proposed Part 20 rule. There are, however, several NCRP recommendations that were not in the ICRP-26 recommendations. These NCRP recommendations are:

> (1) A general "guideline" that the cumulative effective dose equivalent to a worker should not exceed 1 times the worker's age in years; i.e.,  $1 \times N$  instead of the former 5(N - 18) formula;

(2) Use of committed effective dose equivalent for planning purposes and the use of annual (rather than committed) doses for post-(internal) exposure control;

National Council on Radiation Protection and Measurements (NCRP), "Recommendations on Limits for Exposure to Ionizing Radiation," NCRP Report No. 91 (June 1, 1987). (Available for sale from the NCRP, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)

(3) A monthly dose limit as well as a limit on total gestation dose to the embryo/fetus;

(4) Adoption of a 0.1-rem (1 mSv) effective dose equivalent limit for exposure of the general public with the condition that the "site operator" assess the total exposure to the most exposed individual if estimated or measured exposures exceed 25 percent of this limit (25 millirems or 0.25 mSv per year);

(5) The use of "reference levels" set up by the radiation user below the regulatory limits;

(6) A Negligible Individual Risk Level of 1 millirem (0.01 mSv) per year. This level is the "...average annual excess risk of fatal health effects attributable to irradiation, below which further effort to reduce radiation exposure to the individual is unwarranted" (NCRP No. 91, p. 43).

These NCRP recommendations were issued after publication of the proposed Part 20 rule and, consequently, there has not been an opportunity for public comment on them. For this reason, these NCRP recommendations are not being adopted in the revised Part 20 rule at this time.

F. The 1988 Report of the United Nations Scientific Committee on the Effects of Atomic Radiation (UNSCEAR-88)

The United Nations Scientific Committee on the Effects of Atomic Radiation has analyzed data on the sources and effects of atomic radiation and published a series of reports containing summaries of the sources of radiation, the doses received by workers and members of the general public from these sources, and an analysis of the potential health risks from exposure to ionizing radiation.

The latest report in this series is the 1988 report. The 1988 report<sup>7</sup> contains more recent information on the health risks of ionizing radiation determined from a reevaluation of the data on the survivors of the Hiroshima-Nagasaki atomic bombings. Based upon these data, the radiation risk at high doses and high dose rates is estimated to be 7.1 x 10"4 fatal health effects per rad (0.071 effects per gray). For estimating the risk from radiation doses below 100 rads, the UNSCEAR report recommended that a dose rate reduction factor be applied to account for the reduced effectiveness of lower doses and lower lose rates. This would lead to an estimated risk of fatality of between (0.7 to 3.5) x  $10^{-4}$  health effects per rad for low doses such as those encountered in routine occupational exposure and the even lower doses that might be received by members of the general public from NRC- (or Agreement State) licensed activities. The fatal cancer risk value associated with the 1977 ICRP recommendations, 1 is 1.25 x 10<sup>-4</sup> (the proposed Part 20 rule, 51 FR 1102, January 9, 1986) so that the risks as estimated by the 1988 UNSCEAR report for low doses are between 0.6 to 2.8 times higher than the earlier ICRP estimate. The implications of the increased risk are discussed in Section II.I.

G. The 1988 Report of the National Academy of Sciences' Committee on the Biological Effects of Ionizing Radiation (BEIR-IV)<sup>8</sup>

The 1988 BEIR-IV report supplements the 1980 BEIR-III report by providing a more detailed analysis of the risks from internal alpha-emitting radionuclides to complement the emphasis of the BEIR-III report on gamma and beta radiation. Revised risk estimates are given for intakes of radon, radium, polonium, thorium, uranium, and higher transuranic elements (e.g., plutonium).

 7 United Nations Scientific Committee on the Effects of Ionizing Radiation (UNSCEAR), "Sources, Effects and Risks of Ionizing Radiation, 1988 Report to the General Assembly, Sales Section, United Nations, NY 16017 (1988)
8 National Academy of Sciences-National Research Council, Committee on the Biological Effects of Ionizing Radiation "Health Risks of Radon and Other Internally Deposited Alpha-Emitters, (BEIR-IV)," National Research Council, National Academy Press, Washington, DC 20418 (1988). The radionuclide given the greatest emphasis in the BEIR-IV report is radon (radon-222), the gaseous decay product of radium-226. The radon dose conversion factor in the BEIR-IV report for exposure conditions representative of those of the general public is consistent with the value used to derive the airborne effluent concentration limit for radon-222 in Appendix B, Table 2 of the revised 10 CFR Part 20.

H. The 1990 Report of the National Academy of Sciences' Committee on the Biological Effects of Ionizing Radiation (BEIR-V)<sup>9</sup>

The BEIR-V report is another comprehensive reevaluation of the health risks of radiation exposure based upon the revised dose estimates for the survivors of the atomic bombings of Hiroshima and "agasaki. The BEIR-V report gives risk estimates for leukemia and non-leukemia (solid cancers) that are about two to five times higher than the estimates in the 1980 BEIR-III report. The BEIR-V report gives the following factors as the principal reasons for this increase: (1) use of different dose-response and risk projection models, (2) revised estimates of the doses to the individual survivors of the atomic bombings in Japan, and (3) improved epidemiological data from additional years of followup studies since the BEIR-III was completed in 1980.

The BEIR-V Committee uses the linear dose response model and the relative risk projection model to extrapolate the fatal tumor risk to future periods. The relative risk projection model assumes the risk to be proportional to the natural cancer incidence, which generally increases with age. Because of this dependence on age, the relative risk model generally predicts higher future (lifetime) risks than the absolute risk model which employs a constant added risk per year with increasing age. Estimates are given of the risk as a function of the time since the exposure occurred and the age and sex of the exposed person. The BEIR-V report, like the UNSCEAR-88 report, indicates that

<sup>9</sup> National Academy of Sciences-National Research Council, Committee on the Biological Effects of Ionizing Radiation, "Health Effects of Exposure to Low Levels of Ionizing Radiation, (DEIR-V)," National Research Council, National Academy Press, Washington, DC 20418 (1990).

a reduction factor should be applied to the risk estimates derived from high doses and dose rates in order to apply them to low dose and low dose-rate situations. Although neither the BEIR-V report nor the UNSCEAR-88 report recommends a specific value for this factor, both reports indicate that this factor should be greater than 2 (larger reduction factors would give a lower risk per unit dose). Assuming a factor of 2 reduction in the risk estimates derived from high doses and high dose rates, BEIR-V would give a lifetime risk of a radiation-induced cancer fatality of about 4 x 10<sup>-4</sup> fatal cancers/rem (0.04 per sievert) for workers and 5 x 10<sup>-4</sup> per rem (0.05 per sievert) for the general population, the higher value for the public being associated with the higher sensitivity and the longer period of elevated risk associated with the younger ages present in the general population. The value of 5 x 10<sup>-4</sup> is three times as large as the recommended value in the 1980 BEIR-III report and four times as large as the estimate in the 1977 ICRP-26<sup>1</sup> report (see Section II.F).

The BEIR-V report also summarized the data on the frequency of severe mental retardation found in the children of Hiroshima and Nagasaki atomic bomb survivors. These children were exposed <u>in utero</u> at gestational ages of 8-15 weeks and the tisk of severe mental retardation during this period is about  $4 \times 10^{-3}$  per rem with less of a risk at other gestational ages.

The estimates of genetic effects to the offspring of irradiated individuals remained similar to those in the 1972 BEIR-I and 1980 BEIR-III reports. As radiation-induced inherited abnormalities have not been observed directly in humans, estimates of genetic effects have been based primarily upon experimental studies with mice. These studies suggest that it would take a dose of about 100 rads to double the natural frequency of genetically transmitted diseases.

# I. ICRP 1990 Recommendations

On June 22, 1990, the International Commission on Radiological Protection issued a press release indicating that it would issue revised recommendations for radiation protection based upon the newer studies of radiation risks (such as those described in Sections F, G, and H above). The press release indicated

that the ICRP would recommend a reduction in the occupational dose limit from an equivalent of 5 rems per year to an average of 2 rems per year with some allowance for year-to-year flexibility. The ICRP dose limit for long-term exposure of members of the general public would remain equivalent to the level adopted in this revision of Part 20, 0.1 rem per year.

The Nuclear Regulatory Commission does not believe that additional reductions in the dose limits are urgently required by the latest radiation risk estimates. Few individuals in either the work force or in the general public are exposed at or near the limits, and most of these will not be exposed at such levels over long periods of time. Due to the practice of ALARA ("as low as is reasonably achievable"), the average radiation dose to occupationally exposed individuals is well below the limits in either the existing or revised Part 20 and also below the changes being considered by the ICRP. For example, in 1987 about 97 percent of the workers in nuclear power plants, industrial radiography, reactor fuel fabrication, and radioisotope manufacturing, four of the industries having the highest potential for occupational radiation exposures, were below an annual dose of 2 rems so that an immediate reduction in the occupational dose limits would result in only a small reduction in the population dose and in the potential health impact. Although the risk per unit dose is higher than previously thought, individual annual exposures averaged over a lifetime in the highest exposed groups in the working population appear to be about 2-3 rems per year (50-60% of the 5-rem annual limit). Therefore, a factor of 2 increase in the risk per unit dose would result in estimated potential risks associated with actual lifetime exposures that are comparable to the previous risk estimate applied to an assumed lifetime exposure of 5 rems per year.

As a result of the application of the ALARA philosophy to effluent release standards in Appendix I to 10 CFR Part 50 for nuclear power reactors and EPA's 40 CFR Part 190 for the uranium fuel cycle, doses from radioactive effluents from fuel cycle facilities are already much less than the 0.1 rem per year standard in the revised Part 20. The 0.1 rem per year remains as the level recommended by the ICRP for protection of the general public.

Until the final ICRP recommendations are published, and the need for further revisions in NRC standards established, the Commission believes it would be advisable to proceed with the promulgation of the proposed dose limits, rather than deferring the dose reductions that are already associated with the revised Part 20 rule. The Commission will carefully review the final recommendations of the International Commission on Radiological Protection, the comments of the scientific community and others on these recommendations, and the ICRP response to these comments. In addition, the Commission staff will review the recommendations of other expert bodies, such as the National Council on Radiation Protection and Measurements, and participate in the deliberations of the U.S. Committee on Radiation Research and Policy Coordination and any interagency task force convened by the Environmental Protection Agency to consider revised Federal radiation guidance. Any future reductions in the dose limits by the Commission would be the subject of a future rulemaking proceeding.

#### III. Issues Being Resolved Separately

As noted in the above discussion. there are several areas where the Commission believes a better scientific consensus is needed before adopting values different from those in the present Part 20. There are also several areas where issues raised in the public comments (see Section V) are being resolved in other NRC rulemaking proceedings because of either their scope, complexity, or timing. The following issues are being or will be resolved in other NRC rulemaking proceedings:

(1) Establishment of "Below Regulatory Concern (BRC)" levels (related to <u>de minimis</u> levels and a negligible level of risk). On June 27, 1990, the Commission announced the issuance of a policy statement on Below Regulatory Concern, which was subsequently published in the Federal Register on July 3, 1990 (55 FR 27522). This policy statement establishes the framework for the Commission to formulate rules and licensing decisions to exempt certain practices involving small quantities of radioactive materials from some or all regulatory controls. The BRC policy statement sets forth criteria for protection of both individuals (individual dose criteria) and population groups (a collective dose criterion).

(2) Limits for decommissioning of nuclear facilities and for residual radioactive contamination. This is being actively pursued by the NRC staff by developing criteria for residual contamination of soils and structures, which is one aspect of the implementation of the Below Regulatory Concern policy, and by NRC staff participation on an EPA Interagency Task Force on Residual Radioactivity.

(3) Limits and calculational procedures for dealing with the "hot particle" issue (small particles found in nuclear reactors that, because of their high activity and small size, produce high localized doses to skin). The NRC notes that the National Council on Radiation Protection and Measurements (NCRP) has recently issued new recommendations regarding "hot particles" in NCRP Report No. 106, "Limit for Exposure to 'Hot Particles' On the Skin," December 31, 1989. A modified NRC enforcement policy statement with regard to the "hot particle issue" was published in the July 31, 1990 Federal Register (55 FR 31113). The NCRP report, together with a forthcoming ICRP report on the biological effects of skin irradiation and other technical analyses, will be considered in a future rulemaking to set limits for skin irradiation.

(4) Modification of NRC incident notification requirements. A modification of the incident notification requirements was issued for public comment on May 14, 1990 (55 FR 19890). If this proposal is adopted as a final rule, it would modify both the existing Part 20 and this revision.

(5) Publication of a separate rule for large irradiators. A new Part 36 is being proposed for public comment. The detailed requirements for irradiators presently in the revised Part 20 (§ 20.603) will eventually be deleted and replaced by the provisions incorporated in the new Part 36.

There are also additional areas where the scientific basis is not yet resolved sufficiently to justify a change from current practice. These two areas require better scientific consensus on the appropriate position: (1) The need for and impact of a lifetime cumulative dose limit of 1 rem per year of age and (2) quality

factors, especially for neutrons, low-energy beta-emitters, and high-energy gamma photons. These issues will be reconsidered as consensus positions are reached by the scientific community.

IV. Need for Additional Regulatory Guidance

The Commission recognizes that the incorporation of many new concepts into Part 20 will require additional guidance and explanation on their application to practical problems in radiation protection. The Commission also notes the desirability of having such additional guidance available at the same time that the final rule is issued in effective form. However, it was impractical, both for reasons of scheduling and availability of resources, for these guides to be developed concurrently with Part 20. Some of the regulatory guides being developed or revised to assist in the implementation of the revised Part 20 are:

- (1) Content of Radiation Protection Programs at Nuclear Power Plants;
- (2) Interpretation of Bioassay Measurements (Draft Regulatory Guide 8.9, Revision 1),
- Criteria and Procedures for Summation of Internal and External Occupational Doses,
- (4) Acceptable Criteria for Planned Special Exposures and for Satisfying Documentation Requirements;
- (5) Methods and Parameters for Calculating the Dose to the Embryo/Fetus;
- (6) Instructions for Recording and Reporting Occupational Radiation Exposures (includes NRC Forms 4 and 5).

The Commission has instructed the staff to have these and other draft guides published for public comment early in 1991.

#### V. Implementation and Existing License Conditions

Section 20.8 of the rule provides that NRC licensees must implement the Part 20 rule on or before January 1, 1993. Licensees that adopt the provisions of this rule prior to the required implementation date are required to notify the NRC. Early implementation may benefit applicants for new licenses or license renewals as they could avoid having to adopt and implement one version of Part 20 for only a short period of time prior to the required implementation date of this revision. Licensees choosing early implementation must adopt the entire revised Part 20. Compliance will be required with the version of 10 CFR Part 20 codified in the Code of Federal Regulations on January 1, 1991 until January 1, 1993, or until the licensee notifies the Commission of early implementation of the revised Part 20.

License conditions and reactor technical specifications may contain citations to portions of the existing 10 CFR Part 20. After adoption of the revised Part 20 by the licensee or after January 1, 1993, the applicable section of the revised Part 20 that corresponds to the same topic should be used in place of any section of the Part 20 in effect on or before January 1, 1991 that is cited in the technical specifications or license conditions. When there is no corresponding section in the revised Part 20 to these cited provisions, the current license condition based on the Part 20 in effect on or before January 1, 1991 shall remain in force until there is a technical specification change, or license amendment or renewal. If a license condition or technical specification exempted a licensee from a provision of Part 20, it will be assumed to also exempt the licensee from the applicable provision of the revised Part 20. If the license condition or technical specification is more restrictive than the revised Part 20, it shall remain in force until it is modified by a technical specification change or license amendment or renewal.

The NRC will issue a regulatory guide that provides the section and paragraph identifiers in the revised Part 20 and the corresponding sections or paragraphs in the earlier Part 20. This document will issued shortly after the publication of this rule and will enable licensees to locate sections of the revised Part 20 that correspond to sections of the earlier Part 20 cited in license conditions and technical specifications.

# VI. Summary of Public Comments and Changes from Proposed Rule

The purpose of this section is to respond to comments raised on the proposed rule and to explain and highlight the changes made to the proposed rule. This section presents, for each paragraph or section of the rule, the principal public comments on the proposed rule, an NRC staff response to the comments (where appropriate), and a summary of the principal changes that were made to the proposed rule. This section has been arranged so that it corresponds to the structure of the rule. Although it follows the format of the final rule, the following text is not intended to create any additional requirement not already in the regulatory text.

## Subpart A -- General Provisions

# Section 20.1 Purpose.

Final Rule: A new sentence was added to convey the intent of the former § 20.9 in the proposed rule (which has been removed) that the regulations in Part 20 should not hinder a licensee's actions to protect health and safety in the event of an emergency. It is the Commission's intent that the regulations be observed to the extent practicable during emergencies, but that conformance with the regulations should not hinder any actions that are necessary to protect public health and safety such as lifesaving or maintaining confinement of radioactive materials.

In this regard, the Commission notes that the Federal guidance on occupational radiation protection states that those dose standards only apply to normal operating conditions. The Commission believes that the dose limits for normal operation should remain the primary guidelines in emergencies. However, the Commission also recognizes that, in an emergency, operations that do not conform to the regulations may have to be carried out to achieve the high-priority tasks of worker, public, and facility protection. The purpose of the addition to this section is to assure licensees that their first priority should be to carry out those actions that are necessary to protect workers and the public from radiation exposure, to perform lifesaving activities, to prevent or limit the spread of radioactive contamination or the release of radioactive materials to the environment, and to preserve an adequate margin of safety. In evaluating any ensuing violations and their severity, the Commission will consider on a case-by-case basis any extenuating circumstances.

#### Section 20.2 Scope.

Final Rule: The statement of scope remains essentially the same as in the proposed rule except that "background radiation" has replaced "natural background." This change was made to include residual global fallout and ambient radon levels within the definition of "background."

## Section 20.3 Definitions.

General: Because of the large number of comments that dealt primarily with wording changes or that question if the need for or the use of a particular definition, the individual comments will not be discussed separately. However, these comments did result in substantial revisions to many of the definitions that appeared in the proposed rule. Those definitions that were added, modified, or deleted as a result of the public comments are listed below.

Comment: Differentiation among different kinds of dose equivalents. The potential for confusion among different dose equivalents was noted. Commenters noted that effective dose equivalents, committed effective dose equivalents, and doses to the lens of eye, skin, or extramities were all expressed in units of rems or sieverts and may be difficult to distinguish from one another.

Response: In the final rule the NRC staff has applied unique names for these previously undesignated quantities including: eye dose equivalent, shallow-dose equivalent (skin), shallow-dose equivalent (extremities), and total effective dose equivalent. The ICRP did not give these quantities specific names. The use of characteristic names is intended to reduce confusion in using

these units. In this regard, it should be noted that the licensee is required to designate, in a clear and unambiguous manner, the quantities that are being recorded (see paragraph 20.1101(b)).

Final Rule: All the important definitions in the revised rule have been collected into one section, § 20.3 Definitions. Unlike the proposed rule, which employed groups of related terms ("Area Terms," "Dose Terms," "Monitoring Terms," etc.), all the definitions in the final rule are listed in strict alphabetical order. This organization also avoids the presence of "local definitions" that appear only in a specific section of the regulation.

1. New Terms. The following definitions have been added to the final rule. These definitions have been added to clarify the meaning of the terms:

- a. "Activity"
- b. "Background radiation"
- c. "Derived air concentration-hours" ("DAC-hours")
- d. "Dosimetry processor"
- e. "Entrance or access point"
- f. "Generally applicable environmental standard"
- g. "Individual monitoring device"
- h. "Quality factor"
- i. "Sanitary sewerage"
- j. "Total effective dose equivalent (TEDE)."

2. Revised Definitions. The following definitions have been revised or modified from the definition used in the proposed rule:

- a. "Absorbed dose"
- b. "Annual limit on intake"
- c. "Class"
- d. "Committed dose equivalent"
- e. "Committed effective dose equivalent"
- f. "Derived air concentration"
- g. "Dose equivalent"

- h. "Effective dose equivalent"
- i. "Embryo/fetus"
- j. "Eye dose equivalent"
- k. "Member of the public"
- 1. "Nonstochastic"
- m. "Person"
- n. "Planned special exposure"
- o. "Quarter"
- p. "Survey"
- q. "Weighting factor"
- r. "Working level"
- s. "Year"

3. Definitions and terms deleted. Two definitions were deleted because the terms no longer appear in the rule: "Collective effective dose equivalent" and "Roentgen." "Natural background" has been replaced by "Background radiation."

# Section 20.4 Units of Radiation Dose.

Comment: Choice of the system of units. Several commenters expressed a preference for retaining the older "special" units (the curie, rad, and rem) rather than allowing the use of the newer SI units. Reasons cited for retaining the older system included: present widespread use and licensee familiarity, potential for misunderstandings with the newer units, the need for worker retraining (particularly while learning the new ICRP system of dose limitation), and the costs assocized with changing recordkeeping systems. A smaller number of commenters favored changing over to the SI units: becquerels, grays, and sieverts.

Response: Although both the "special units" and the SI units appear in the text of Part 20 (to increase the familiarity of licensees with the SI units), the Commission has decided that adoption of the SI units at this time is not necessary. The Commission recognizes that the new terms and methodological approaches in the revised Part 20 are complex and that imposition of the SI

system of units on top of this complexity would further increase the potential for confusion. Consequently, at the present time, the recordkeeping, reporting, and notification requirements require the use of the "special units," the rad, the rem, and the curie. However, as the national move to metrication continues, as anticipated in Section 5164 of the Omnibus Trade and Competitiveness Act of 1988 (P.L. 100-418), at some later time there may be amendments to Part 20 that would require the use of SI units only (becquerels, grays, and sieverts).

Final Rule: The final Part 20 rule includes the International System of Units (SI units) for distance, area, and volume. The older "special units" are retained for activity (curie), absorbed dose (rad), and dose equivalent (rem).

Comment: Quality factors for neutrons. The quality factor is the conversion factor between the absorbed dose (rads) and the dose equivalent (rems). Several publications<sup>2,3,4,10,11</sup> have recommended changes in neutron quality factors that are a factor of 2 higher than those in proposed Part 20. These changes would raise the quality factor for fast neutrons from 10 to 20.

Response: Increases in the quality factor for neutrons are suggested by some animal experimental data on the relative biological effectiveness (RBE) of neutrons. However, there appears to be considerable uncertainty as to whether the data actually demonstrate an increase in the hazard of neutrons. Because the RBE is defined as a ratio of coses to produce equivalent biological effects, it is not clear whether the apparent increase in the neutron RBE is due to the increased effectiveness of neutrons or whether it actually results from the decreased effectiveness of the reference gamma radiation at low doses.

<sup>10</sup> International Commission on Radiological Units and Measurements, "The Quality Factor in Radiation Protection," ICRU Report No. 40 (1986). (Available for sale from ICRU Publications, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)

<sup>11</sup> International Commission on Radiological Protection, "Data for Use in Protection Against External Radiation," ICRP Publication No. 51 (January 1988). (Available for sale from Pergamon Press, Inc., Elmsford, NY 10523.)

Final Rule: The NRC has decided not to revise the neutron quality factor at this time but to defer any change until there is greater scientific consensus on the most appropriate value. A major consideration underlying this decision is that neutron exposures at most NRC-licensed facilities are currently small and the potential increase of a factor of 2 would not have a major health or regulatory impact.

The decision to defer any change is consistent with recommendations of the Committee on Interagency Radiation Research and Policy Coordination (CIRRPC) of the Office of Science and Technology Policy that there should not be a revision of the value of the neutron quality factor at this time without more study. This position is also reflected in papers from the United Kingdom National Radiological Protection Board (UK, XPB)<sup>12</sup> and a statement on the neutron quality factor from the British Committee on Radiation Units and Measurements.<sup>13</sup>

Comment: Table of neutron quality factors. Several commenters questioned the accuracy and timeliness of the table of neutron quality factors and fluence rates (to give dose equivalents of 1 rem) that appeared in the proposed rule. Some commenters suggested that there were more appropriate tables published by the NCRP or ICRP.

Response: The tables in the proposed and revised rules were taken from NCRP Report No  $38^{14}$  and are appropriate for the neutron dose equivalent at a soft tissue depth of 1 centimeter (which is the depth specified for the determination of the deep dose equivalent). There are newer tables from the ICR<sup>2</sup>,

<sup>12</sup> J.A. Dennis, "The Relative Biological Effectiveness of Neutron Radiation and Its Implications for Quality Factor and Dose Limitation," Nuclear Energy 20(2). 133-149 (1987).

<sup>13</sup> British Committee on Radiation Units and Measurements (BCRU), "Memorandum from the BCRU: Effective Quality Factor for Neutrons," Physics in Medicine and Biology 31 (7):797-799 (1986).

<sup>14</sup> National Council on Radiation Protection and Measurements, "Protection Against Neutron Radiation," NCRP Report No. 38 (January 1971). (Available for sale from the NCRP, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)

but these tables incorporate the factor of 2 increase in the neutron quality factor. (See the preceding discussion of the neutron quality factor.)

Subpart B -- Radiation Protection Programs

# Section 20.101 Radiation Protection Programs ["As Low As Is Reasonably Achievable" (ALARA)] (§ 20.102 of the Proposed Rule).

Comment: The concept of ALARA is a philosophical principle of radiation protection and, as such, it should not be made into a regulatory requirement. A primary objection to changing the status of ALARA from the hortatory suggestion in the current Part 20 ("licensees should") to a mandatory requirement ("licensees shall") is that there are no guidelines (except for light-waterreactor (LWR) effluents) as to what constitutes ALARA. Because of the subjective nature of an "ALARA level," there are problems in the retrospective evaluation of licensee performance by NRC inspectors and, at least in one case, interpretations by the courts concerning whether the levels arhieved were truly "as low as is reasonably achievable."

Response: There were a number of comments that expressed similar concerns regarding the proposed implementation of "ALARA." The emphasis on ALARA actions has been revised from detailed requirements to document all ALARA actions to a requirement to have a radiation protection program that includes measures to keep doses and intakes "as low as is reasonably achievable." This shift is to emphasize that the ALARA concept is intended to be an operating principle rather than an absolute minimization of exposures.

Comment: Any requirement for ALARA should include a lower bound. Many commenters felt that there should be a "floor" for ALARA necessary.

Response: The Commission agrees that there would be advantages to establishing such a "floor," below which efforts to further reduce doses would not

be necessary. An NRC policy statement on "Below Regulatory Concern" was annemced on June 27, 1990, and published in the Federal Register on July 3, 1990 27522). The Below Regulatory Concern levels in that policy statement delinance criteria below which additional licensee actions to further reduce doses would not be required. Specific rulemaking actions will be carried out to define operational thresholds for particular classes of activities such as disposal of very low-level contaminated materials. The BRC policy statement provides a framework for evaluating these case-specific actions. (See also discussion on § 20.304.)

Comment: Compliance with "ALARA-based" standards should constitute being ALARA. Several commenters supported the statement in the proposed Part 20 (§ 20.102(b)) that compliance with EPA's 40 CFR Part 190 and with Appendix I to 10 CFR Part 50 should constitute de facto compliance with the requirement to keep LWR effluents ALARA. Environmental Protection Agency (EPA) comments did not support this view.

Response: Appendix I to 10 CFR Part 50 defines ALARA levels of radioactive materials in LWR effluents. If the design objectives of Appendix I are met, it constitutes a demonstration that the effluents are ALARA and no additional effort is required to reduce the effluent levels. Although the EPA interprets 40 CFR Part 190 as an "ALARA-based" standard, it also believes that 40 CFR Part 190 constitutes an upper bound, not a lower bound, to ALARA efforts.<sup>15</sup> Consequently, compliance with 40 CFR Part 190 is not in itself sufficient to demonstrate that releases and doses are ALARA.

As Appendix I to Part 50 defines ALARA design objectives that constitute ALARA effluent levels, meeting these levels is sufficient to demonstrate ALARA

<sup>15</sup> Letter of January 7, 1986, from Sheldon Meyers, Director, Office of Radiation Frograms, Office of Air and Radiation, U.S. Environmental Protection Agency, to Sobert B. Minogue, Director, Office of Nuclear Regulatory Research, U.S. Nuclear Regulatory Commission. (This letter is reproduced as Enclosure B to the comments of the Environmental Protection Agency on the 10 CFR Part 20 revision (Docket PR-19, 20, 30 et al., 50 FR 51992, Comment # 769).)

that direct radiation from on sources (gamma radiation from external radwaste tanks and turbine gener rs ("turbine shine") is also ALARA, and that

effluent releases. In order for light-water reactors to demonstrate that doses from both effluents and direct radiation are ALARA, it is decessary to demonstrate that effluents meet the design objectives of Appendix I to 10 CFR Part 50, the total dose to any member of the public is within the numerical standards in 40 CFR Part 190. Meeting these conditions will constitute sufficient evidence that offsite doses from LWRs are ALARA and in conformance with both Appendix I and 40 CFR Part 190.

Comment: The NRC should establish "reference levels" in its rules. One commenter thought that the NRC should have "reference levels" for licensee action in Part 20.

Response: The Commission recognizes that licensees generally establish their own lower "reference levels" in order to keep from reaching and exceeding the Commission's formal dose limits. Based upon the public comments on the reference level for exposure of members of the public, which was in the proposed § 20.303, this ap, bach would not be favored by a majority of licensees. Several commenters viewed the reference level for the dose to members of the public as being applied exactly as if it were a limit. Consequently, if the NRC were to specify generic reference levels for licensee action, the impact might be similar to lowering the magnitude of the dose limits. The Commission believes that the use of the ALARA philosophy is a preferable means to keep exposures well below the limits established by the Commission.

Final Rule: The final rule establishes a requirement for all licensees to have a radiation protection program that includes provisions for keeping radiation doses ALARA. It is expressly intended that the level of this program and efforts to document it are commensurate with the size of the licensed facility and the potential hazards from radiation exposure and the intake of radioactive materials.

The requirement for a radiation protection program is not new; it was discussed in the proposed rule (under ALARA) and is consistent with requirements

in Part 33 (§§ 33.13, 33.14, and 33.15), Part 34 (§ 34.11), Part 35 (§§ 35.20-35.31), and Part 40 (§ 40.32) of the NRC regulations, with the information requested in Chapter 12 of Regulatory Guide 1.70, "Standard Format and Content of Safety Analysis Reports for Nuclear Power Plants," and with the conditions in most licenses issued by the Commission. The extent of this program and requirements for written records and procedures for operating the program are intended to be commensurate with the scope and potential hazards associated with the licensee's activities. The Commission recognizes the need to provide guidance on the scopes of radiation protection programs and such guidance will be prepared in the form of regulatory guides.

The Commission continues to emphasize the importance of the ALARA concept to an adequate radiation protection program. In order to strengthen this concept, the Commission has adopted a requirement that all licensees include provisions for maintaining radiation doses and intakes of radioactive materials as low as is reasonably achievable as part of their radiation protection programs. Compliance with this requirement will be judged on whether the licensee has incorporated measures to track and, if necessary, to reduce exposures and not whether exposures and doses represent an absolute minimum or whether the licensee has used all possible methods to reduce exposures. This shift in emphasis should reduce potential problems of retrospective evaluation of licensee performance under admittedly subjective criteria. However, the licensee should be able to demonstrate that periodic reviews of performance have been made and that efforts have been made to achieve ALARA. As noted above, the level of effort expended on the radiation protection programs should reflect the magnitude of the potential exposures, both the magnitude of average and maximum individual doses and, in facilities with large numbers of employees, collective (population) doses. A nuclear power reactor licensee would be expected to have a considerably larger program than a licensee with only small sealed sources.

The Commission has not adopted a requirement that a numerical cost-benefit analysis (optimization analysis) be used to demonstrate ALARA. The quantitative approach is useful for those situations where both costs and benefits (dose reduction) can be quantitated, such as in shielding design or analysis of

decontamination methods. The Commission incourages licensees to employ quantified analyses to define ALARA, but their use is not required. One reason for this is that many ALARA procedures simply reflect sound operating practice and do not lend themselves to a numerical analysis. Another reason is that cost-benefit analyses could have a cost associated with obtaining the necessary information and carrying out the analysis that may exceed the monetary value of the dose reduction. Thus, the quantitative optimization analysis would be expected to be used primarily in situations where both the costs of control and the resultant benefits were not only quantifiable, but also appreciable compared to the cost of performing the analysis.

#### Subpart C--Occupational Dose Limits

Section 20.201 Occupational Dose Limits for Adults.

Comment: Elimination of the 5(N - 18) age-provated cumulative dose limit and the adoption of the 5-rem annual effective dose limit. Most commenters favored this change noting that licensees have generally succeeded in keeping doses below 5 rems per year for the past few years and, therefore, are already meeting the new limit.

Comment: Lifetime dose limits. A few commenters believe that there should be a limit on the cumulative total dose that can be received by any individual in a lifetime.

Response: The Commission considered the use of a lifetime dose limit but rejected it. The EPA had proposed such a limit (100 rems) in its proposed Federal Guidance on Occupational Radiation Exposure (46 FR 7836, January 23, 1981) but withdrew it.

If the magnitude of the annual dose is limited, there is a <u>de fact</u>o limitation of the lifetime dose that can be received. The Commission believes that such a <u>de facto</u> lifetime limit is preferable to an actual cumulative lifetime

dose limit because the cumulative limit could act to limit employability. This, in turn, raises questions concerning the right of an individual to pursue employment in a chosen profession. If an individual were to deplete the "rose bank" provided by a lifetime dose limit, it might be difficult to obtain future employment using ionizing radiation.

Comment: Quarterly dose limit. A number of commenters noted that the ICRP system of dose limitation does not have quarterly or other limits covering periods less than a year. The public comments also noted the possibility of giving rise to two violations for the same event (i.e., the possibility of exceeding both the quarterly and annual dose limits in one event), thereby incurring two penalties.

Response: The quarterly limit (only for deep-dose equivalent) had been retained in the proposed rule as a result of suggestions received from several groups during the development of the rule. The primary protection function of retaining a quarterly limit was to reduce the potential for receiving several high doses within a relatively short period of time. However, there is not much of a radiobiological significance between 10 rems (two 5-rem doses) and 6 rems (two 3-rem doses) received in a short time period. One consideration is the employability of a worker who has exceeded the dose limit. A worker who exceeded the 5-rem annual dose limit might have to work in a job not involving radiation for a year (or take part in a planned special exposure) instead of only a calendar quarter if a quarterly dose was used.

Final Rule: In order to maintain compatibility with the ICRP and to eliminate the possibility of double violations, the quarterly limit has not been kept and only annual limits are stated.

Comment: Eye dose limit. Some commenters questioned the 15-rem (0.15sievert) eye limit used in the proposed rule noting that ICRP Publication No. 26 contains a recommended value of 0.3 sieverts (30 rems). Response: The ICRP recommended a reduction in the limit for the eye to 0.15 sieverts (15 rems) at their Brighton, England, meeting in 1980.<sup>16</sup> This was done because the ICRP concluded that, for a lifetime of occupational exposure at the former 0.3-sievert (30-rem) limit, some opacities in the lense of the eye might be produced that could develop to the point of causing deterioration of vision (even without further radiation exposure). In most situations, the limits for the deep-dose equivalent and the shallow-dose equivalent to the skin should ensure that the eye dose limit is also met. Consequently, the reduction from 30 rems to 15 rems is not expected to have a significant impact on either health protection or control cost.

Comment: Parameters defining the shallow-dose equivalent ("skin dose"). The proposed rule would have established a dose limit for the skin of 50 rems averaged over 10 square centimeters (10 cm<sup>2</sup>). There were several comments concerning the scientific basis for this area. Some commenters suggested othe. surface areas, such as 15 cm<sup>2</sup>, as being better suited to measurement conditions. Proponents of the larger areas generally favored these areas because of their compatibility with either contamination survey practices or with the physical size of survey instrument detector probes.

One set of comments prepared by the developer of the NRC's VARSKIN computer program for skin dose calculation (comment letter No. 262 in the NRC Public Document Room) contains a well-documented discussion of the selection of an appropriate area over which to average the skin dose. These comments conclude that  $1 \text{ cm}^2$  is a more appropriate area than either 10 cm<sup>2</sup> or 100 cm<sup>2</sup>.

Response: ICRP Publication 26 contains two recommendations for such areas: a 100-cm<sup>2</sup> area and a 1-cm<sup>2</sup> area, the larger area being associated with routine monitoring for skin contamination and the smaller area being associated with

<sup>16</sup> International Commission on Radiological Protection, "Statement and Recommendations of the 1980 Brighton Meeting," <u>Annuals of the ICRP 4(3/4)</u> Oxford, England: Pergamon Press (1980). (Available for sale from Pergamon Press, Inc., Elmsford, NY 10523.)

accident dose evaluation. After reviewing these comments and various recommendations regarding skin dose measurements, the Commission has decided to use the smaller area of 1 cm<sup>2</sup> for routine skin dose evaluations. The 1-cm<sup>2</sup> area is consistent with the prior recommendations in NCRP Report No.  $39^{17}$  and ICRP Publication No.  $9^{18}$  as well as the smaller area recommended in ICRP Publication No. 26.

Within the past several years, there have been instances where very small (5-250 µm) "hot" particles of fuel or activated corrosion products have been discovered in reactor facilities, on workers or their clothing, and, in a few isolated cases, in worker's vehicles or homes. These particles are generally too large to pose a significant risk from inhalation, but are capable of producing intense beta-radiation doses over very small areas of the skin. The principal hazard appears to be skin ulceration if the particles remain localized on the skin surface. The primary uncertainty associated with evaluating the hazard of these small particles is determining the skin area or tissue volume to which the dose is to be computed (or even whether "dose" is the most appropriate indicator of the hazard). The NRC requested the National Council on Radiation Protection and Measurements (NCRP) to look into the hot particle issue and make recommendations. The NCRP's recommendations have been published in NCRP Report No. 10619 and use a criterion based upon the number of radioactive disintegrations that have occurred (uCi-hours) rather than dose. The NRC staff is reviewing these recommendations and has issued an Information Notice on a modified enforcement policy for hot particles.

Final Rule: This revision of Part 20 specifies an area of 1 cm<sup>2</sup> for skin dose evaluations.

- 18 International Commission on Radiological Protection, "Recommendations of the International Commission on Radiological Protection (adopted September 17, 1965)," ICRP Publication No. 9 (1966), page 6, paragraph 28. (Available for sale from Pergamon, Press, Inc., Inc., Elmsford, NY 10523.)
- 19 National Council on Radiation Protection and Measurements, "Limit for Exposure to 'Hot Particles' on the Skin," NCRP Report No. 106 (December 31, 1989). (Available for sale from the NCRP, Bethesda, MD 20814.)

<sup>17</sup> National Council on Radiation Protection and Measurements, "Basic Radiation Protection Criteria," NCRP Report No. 39 (January 15, 1971), page 79, paragraph 207. (Available for sale from the NCRP, Bethesda, MD 20814.)

Comment: Effective dose equivalent for external exposure. The most prevalent comment concerning the effective dose equivalent is the restriction in the proposed rule of the risk-weighted organ dose "effective dose" concept to internal doses without permitting a similar approach to be employed for external doses. There were several comments that noted the desirability of using organ weighting factors for external doses.

Response: The ICRP and NCRP recommendations and the 1987 Federal guidance on occupational radiation exposure in principle permit the use of external weighting factors. However, none of the principal standard-setting organizations has included specific recommendations for the use of weighting factors for external dose.

The application of weighting factors also entails calculation of organ doses instead of whole-body doses from external radiation. One component of this calculation is estimation of the attenuation of the radiation as a function of the depth of the organ in the body. There are practical problems in the determination of the type and energies of the radiation involved and of the orientation of the individual with respect to the source of the radiation that have to be considered in making such calculations. Therefore, application of weighting factors for external exposures will be evaluated on a case-by-case basis until more guidance and additional weighting factors (such as for the head and the extremities) are recommended.

Final Rule: External doses to the head, trunk (including male gonads), arms above the elbow, or legs above the knee are to be treated as whole-body doses. For the purpose of weighting the external whole-body dose (for adding it to the internal dose), a single weighting factor,  $w_T = 1.0$ , has been specified. The use of other weighting factors for external exposure may be approved on a case-by-case basis upon request to the NRC.

Comment: Allowance for exposure after limits are exceeded. Commenters noted that allowance of an additional 1 rem per quarter dose limit for a worker who had already exceeded the 5-rem annual limit might be counterproductive. Workers who remain under the annual limit, and whose dose was X rems, would be
constrained to receive (5 - X) rems, whereas workers who received more than 5 rems in the first quarter could be allowed an additional 4 rems (1 rem in each of the four quarters). One commenter suggested that this could provide an incentive for individuals who are approaching the dose limit to deliberately exceed the limit and thereby protect their employability by taking advantage of the extra dose allowance available to those who have exceeded the limits. Another commenter believed that such a blanket authorization to exceed the limits was inappropriate and preferred prior NRC review of the use of these extra doses on a case-by-case basis.

Response: The purpose of the dose allowance was to protect the worker's employability after having received a dose above the dose limits. Although intentionally getting additional exposure might be in the worker's interest for employability reasons, such an action would not be in the worker's interest with respect to health protection. Licensees having workers with critical skills who are approaching the dose limits early in the year or workers who have received an accidental overexposure should consider use of the planned special exposure (§ 20.206) to permit continued employment.

Final Rule: The allowance of an additional 1 rem per quarter following an exposure in excess of the limits has been deleted.

# Section 20.202 Compliance with Requirements for Summation of Internal and External Doses.

Comment: Implementation burden. Many commenters felt that the burden of adding external and internal doses was substantial, particularly as most licensees would be faced with either external exposure situations or internal dose situations, but not both.

Response: The NRC staff disagrees that there will be a substantial recordkeeping burden because this summation will be required only if both the internal dose and the external dose are each likely to exceed 10 percent of the dose limit. Thus, in most situations, as noted in the comments, only one component will be required in the required and, consequently, summation of internal and external doses will not be required. Final Rule: The requirement remains that the committed effective dose equivalent and the deep-dose equivalent should be summed to give the total effective dose equivalent. However, this summation need only be performed if both components are required to be monitored (i.e., exceed 10% of an applicable dose limit). If the summation of doses is not required, then the limit applies to the component (internal or external) that is measured. The NRC is planning to issue additional guidance in the form of a regulatory guide before the effective date of the revised Part 20. This guide will be on procedures to be used in estimating committed effective dose equivalents and deep-dose equivalents and guidance on when internal and external doses have to be summed.

Comment: Use of individual metabolic or dosimetric data. Several commenters thought that the proposed rule required the use of specific metabolic and dosimetric parameters for the exposed individual. One commenter also thought that the use of such parameters would "invalidate the stochastic approach of the regulation, which presumes that the effects of radiation exposure at these levels are statistical in nature."

Response: It was not intended that licensees would be required to collect and use specific metabolic or dosimetric information on exposed individuals for use in dose assessments. The intent was to permit the use of personal data for dose assessment when such data were available. The use of parameters that are more appropriate for a particular exposed individual than those assumed for the "Reference Man" should improve the accuracy of the dose estimate for that individual. This is unrelated to the concept of stochastic health effects.

The statistical nature of the potential stochastic effects of low doses of ionizing radiation does not require that the associated dose estimates be based on Reference Man doses. However, it is necessary to resort to populationaveraged dose-to-risk conversion factors as there are no health risk coefficients available for specific individuals.

[Monitoring thresholds and thresholds for summation of internal and external dose -- see discussion under § 20.502] Note: Section 20.202(c) states that: "The assigned deep-dose equivalent and shallow-dose equivalent must be for the part of the body receiving the highest exposure." This requirement is intended to apply primarily to situations where there are steep gradients in the radiation dose rate, depending upon location within the facility and spatial orientation of the worker's body. For example, good practice for a worker in a nuclear power plant who is reaching up into a radioactive steam generator would be to wear at least two personnel dosimeters: one to monitor the extremity dose (worn on the finger or wrist) and one to monitor the whole-body dose (worn on the upper arm). For routine monitoring in relatively homogeneous radiation fields, special consideration to identify the actual "highest" exposed area would not be required.

Section 20.203 Determination of External Dose from Airborne Radioactive Material.

Comment: This could be read to require that the air concentration be measured at two locations. This section appears to require that the air concentration be measured at the location of the individual and at the point of maximum concentration in the cloud. The regulation should emphasize the reliance on personnel dosimeters or other monitoring devices.

Response and Final Rule: Section 20.203 has been shortened considerably. The revised section emphasizes the use of survey instruments and personnel monitoring devices to evaluate the external dose. The remaining technical guidance from this section in the proposed rule will be incorporated into a regulatory guide.

Section 20.204 Determination of Internal Exposure.

Comment: Interim dose calculation factors and parameters. Because the existing Part 20 is based on  $ICRP-2^{20}$  dosimetry and metabolic models and the

<sup>20</sup> International Commission on Radiological Protection, "Report of Committee II on Permissible Dose for Internal Radiation," ICRP Publication No. 2 (1959). (Available for sale from Pergamon, Press, Inc., Inc., Elmsford, NY 10523.

revised Part 20 employs the ICRP- $30^{21}$  dose parameters, there was concern regarding whether the more recent ICRP-30 parameters should be used, particularly when the value is to be compared with the intake limits in the present Part 20.

Response: The NRC is planning to issue a regulatory guide that will address the use of bioassay measurements for determining compliance with Part 20. Appropriate parameters for calculating organ doses from radionuclide intakes can be found in ICRP-30 and its supplements. Dose factors in Federal Guidance Report #11<sup>22</sup> are also acceptable for use in calculating occupational exposures. However, the effective dose equivalent factors in Federal Guidance Report #11 do not employ a rounding method suggested in ICRP-30. For this reason, the dose factors in Report #11 may be be slightly higher (10-20 percent) than the effective dose factors that correspond to the ALIs and DACs in both the revised Part 20 and Report #11. These dose factors would be more restrictive (give slightly higher doses for the same intake) than dose factors computed using the ICRP-30 roundoff procedure, but they can be used for evaluating compliance with Part 20.

Section 20.205 [deleted] Further Provisions -- Internal Exposure Involving Radionuclides with Very Long Effective Half-Lives.

Comment: Exemption for long-lived radionuclides and the use of the committed dose equivalent concept. The use of the concept of a "committed dose equivalent" drew numerous comments. This approach entails assigning to the year of intake the future internal dose (the "committed dose equivalent" over 50 years) from radionuclides taken into the body during that year. The proposed rule (in § 20.205) allowed an exemption from the use of committed dose equivalents for several long-lived radionuclides.

International Commission on Radiological Protection, "Limits for Intakes of Radionuclides by Workers," ICRP Publication No. 30. (Available for sale from Pergamon, Press, Inc., Inc., Elmsford, NY 10523.)
Environmental Protection Agency, Federal Guidance Report No. 11, "Limiting

<sup>22</sup> Environmental Protection Agency, Federal Guidance Report No. 11, "Limiting Values of Radionuclide Intake and Air Concentration, and Dose Conversion Factors for Inhalation, Submersion and Ingestion." USEPA Report EPA-520/ 1-88-020 (September 1988). (Available from the USEPA, Office of Radiation Programs, 401 M Street, S.W., Washington, DC 20460.)

Many of the commenters objected to having to assign the future 50-year dose to a single year. Others suggested that variable integration periods be allowed instead of one fixed 50-year value. One argument offered in support of either of these positions is that many adult workers would not normally be expected to live long enough to accrue the full 50-year committed dose equivalent. Commenters pointed out that while pre-exposure controls (such as the annual limits on intake and the derived air concentrations) should be based upon the committed dose equivalent concept for planning and control, the use of controls based upon limiting the annual effective dose equivalent rate (rather than using the committed dose equivalent) might be preferable for post-exposure management following actual radionuclide intakes.

It was also noted that there were several additional nuclides that had similar half-lives and retention characteristics but were not included in the proposed exception. Among these were cobalt, strontion, and americium. The approach in the proposed rule was characterized as appearing to place almost complete emphasis on the control of the work environment rather than on the assessment and control of the individual worker.

Response: The concept of dost commitment is not nev; this concept has been used as the basis for controlling internal doses since the late 1950s when ICRP Publication No.  $2^{20}$  and the present 10 CFR Part 20 were published. However, the term "committed dose equivalent" applied to future doses from internal emitters initially appeared in 1977 in ICRP Publication No. 26.<sup>1</sup>

The concentration limits for air and water in Appendix B to the existing Part 20 were based upon concentrations which, if continually inhaled (for air) or ingested (for water) over a 50-year period, would produce a dose rate in the "relitical organ" in the 50th year that was numerically equal to the annual organ dose limit. For certain radionuclides that slowly approached a constant body burden, primarily those radionuclides that have both long radiological halflives and long biological clearance half-times, the limiting organ dose rate is not reached by the 50th year. For shorter-lived radionuclides and those that are rapidly removed from the body, equilibrium may be attained more rapidly and the limiting annual organ dose rate could persist over many years.

The limiting <u>dose rate</u> in the 50th year from a constant intake of a radionuclide each year over a 50-year period is numerically equal to the <u>total dose</u> integrated over the 50-year period from a single year's intake of the same magnitude. Therefore, controlling the integrated future ("committed") dose for each year's radionuclide intake also controls the annual dose rate in the 50th year to be within the dose limit.

It was noted that use of limits to annual doses in some cases would not ensure that doses in future years would be within limits. The example of the ingrowth of americium-241 from plutonium-241 was cited in which, even if the initial annual dose from plutonium-241 were within the limit, the ingrowth of the radiologically more significant americium-241 would lead to doses higher that the limits in subsequent years.

There are only a few radionuclides that would not attain an equilibrium level (and a constant annual organ dose rate) within time periods of less than 50 years. The use of the committed dose equivalent, rather than controlling internal dose on the basis of annual dose, substantially overestimates annual doses only for those radionuclides that do not reach an equilibrium level in the body early in the working lifetime. These radionuclides are primarily the long-lived radionuclides for which the exemptions of § 20.205 in the proposed rule were intended. Radionuclides (such as cobalt-60, strontium-90, ard americium-241) that were easily measured at airborne concentrations or body burdens below the DAC and ALI values were not included in the list of exempted radionuclides because an exemption was not believed to be necessary for them.

The annual limits on intake and derived air concentrations are used mainly for pre-exposure control rather than post-exposure dose assessment so that finetuning these values to specific ages or adjusting them for factors such as the length of the period over which the committed dose is evaluated or to differences in individual organ sizes (as were suggested) is not warranted for occupational dose assessment. The use of age-dependent committed dose factors as suggested by some commenters would add needless complexity to the assessment of internal doses and cannot be justified on the basis of the availability of information on either age-dependent metabolic parameters or age-dependent radiobiological risk information.

The use of an annual dose limitation system, even with a reduction in the allowable dose limit from 5 rems to 3 rems such as in the proposed § 20.205, does not provide a limitation on the lifetime radiation dose or risk equivalent to that provided by the committed dose limitation system of this final rule for all classes of workers. Although long-term workers would be protected to the same degree under either the annual or committed dose systems, short-term or temporary workers could get somewhat higher lifetime doses under a dose limitation system based on limiting only individual annual dose. Furthermore, it is neither reasonable nor practical to expect future employers to take special measures to control radiation dose to workers who transfer because a previous employer, working under annual organ dose limits, permitted intakes that would result in future dose rates that are appreciable fractions of the allowable dose limits. Such a practice would not be fair to workers whose future employability may be limited because of the additional restrictions a new employer would have to put on their exposure, or to future employers of these workers who may have to assess internal doses from residual body burdens of internal radionuclides in order to show compliance. The annual dose system also requires a complex bookkeeping effort because the annual dose limit for each worker depends upon the worker's pre-existing body burden of radioactive materials.

Final Rule. For the reasons discussed above, the Commission has decided not to adopt proposed § 20.205 and the exemptions for certain long-lived radionuclides for the final rule. The use of the committed dose equivalent will be applied uniformly to all radionuclides, regardless of half-life. The Commission recognizes that the removal of this exemption, combined with the lowering of the airborne concentration limits for several radionuclides (notably thorium and uranium), could impact on the current and future facilities that use these materials. Licensees that are affected by these changes may request an extension of the implementation time in order to make the necessary modifications to comply with the revised limits as they relate to long-lived radionuclides identified in the proposed § 20.205. In addition, licensees should note the flexibility provided in the revised rule for more accurate dose assessments to be made that might show that additional controls were not required in order to meet the dose limits. Specifically, § 20.204 allows the use of actual particle-size distributions and physicochemical characteristics of airborne particulates to define a site-specific derived air concentration to be used in lieu of the generic values in Appendix B. Such adjustments result in the use of more precise dose estimates because of a better characterization of the actual exposure conditions. Although these adjustments might permit higher airborne radionuclide concentration limits to be used, the same degree of health protection would exist because the radiation dose (and risk) would remain the same. This section also allows for whole-body counting or bioassay measurements to determine the behavior of radioactive materials in the individual and the use of these data to calculate internal doses. A 7-month delay between a bioassay or retention measurement and recording of the associated dose is also permitted in order to make confirmatory measurements.

The Commission recognizes that alternative methods may be identified in the future that might achieve the same degree of lifetime risk limitation for both short-term and long-term workers as the dose system recommended by the ICRP, the Radiation Protection Guidance to Federal Agencies for Occupational Exposure, and adopted in the current and revised 10 CFR Part 20. The Commission further believes that, to be acceptable, such alternatives should not result in an ad-verse impact on worker employability or result in undue recordkeeping or excessive monitoring requirements for the future employers of transferring workers.

Section 20.206 Planned Special Exposures.

Comment: The use of planned special exposures could result in lifetime cumulative doses greater than those doses formerly permitted under the 5(N - 18)formula. One commenter noted that the new regulatory scheme, including planned special exposures, allowed a higher total lifetime dose than was permitted using the 5(N - 18) formula. The calculation presumes a working lifetime of 47 years (starting at age 18 and ending at age 65). Under the revised Part 20, the lifetime limiting dose would be 260 rems (5 rems per year)(47years) + 5(5 rems) (planned special exposures) = 235 + 25 = 260 rems). Under the 5(N - 18) formula, at age 65 (N = 65), the cumulative dose would be 5(47) = 235 rems. The comment further noted that the NCRP recommended [in NCRP Report No. 91] a cumulative dose limit of 1 rem x age; the Department of Energy has proposed a 100-rem lifetime dose limit, and the ICRP at its 1984 Stockholm meeting inferred a goal of 1 rem per year. Other commenters noted that, because of the potential lifetime dose including the planned special exposure, the claim on page 51, FR 1121 (Table 5), of the proposed rule that "Individuals receiving highest exposure will be reduced" is unjustified and incorrect.

Response: The analysis of maximum doses discussed above is overly simplified because it assumes that there are individuals who will be exposed at the allowable dose limit every year of their working lifetime. Under the old 5(N - 18) formula, the unused portion of the dose limit (the difference between the actual dose received and 5 rems) became part of a "dose bank" that could be drawn on in later years (at a rate up to 3 rems per quarter or 12 rems per year). This "dose bank," which is inherent in the age-prorated formula of 5(N - 18), does not exist with the straight annual dose limit. If the worker's exposure is under the 5-rem annual dose limit, there is no way to recapture the difference for use in future years. Consequently, the average annual dose (for the more highly exposed workers) associated with new Part 20 is expected to be less than under the former rule.

As noted above (see Response under § 20.201 Occupational Dose Limits), the Commission considered the use of a lifetime dose limit but rejected it.

Comment: Planned special exposures should not be limited to external exposures but should also be permitted for internal exposures. Several commenters noted that it was inconsistent to treat internal and external doses as equivalent by summing them and then restricting planned special exposures to only external doses. Commenters also pointed out that the total effective dose equivalent (TEDE) could be minimized in some cases if some external doses were reduced at the expense of incurring some internal doses.

Response: The Commission agrees that restricting the use of planned special exposures to only external doses would be inconsistent with the ALARA principle and the presumed equivalence of internal and external doses inherent in the revised Part 20. Consequently, the requirements have been modified so that internal doses may be included in planned special exposures in order that the total dose (TEDE) can be controlled in keeping with ALARA. Comment: The annual dose allowed in a planned special exposure does not agree with the recommendations of the ICRP. A few commenters thought that the allowable annual dose from planned special exposures should be 10 rems as stated in the ICRP recommendations. Other commenters agreed with the NRC's modification to reduce the annual dose for planned special exposures to 5 rems.

Response: The NRC has intentionally reduced the dose allowed in any year from a planned special exposure from the 10-rem value proposed by the ICRP to 5 rems. The lifetime total limit from planned special exposures of 25 rems remains the same as the ICRP recommendation. The Commission believes that it would be better to distribute the dose over the lifetime more evenly than to permit a large portion of the cumulative dose to be received within a small period of time. In this sense it should be recalled that the planned special exposure is in addition to the normal dose limits. Under the Part 20 condition, it would be theoretically possible to get a 10-rem dose in 1 year, 5 rems from a planned special exposure and 5 rems from routine operation. This is roughly equivalent to the 12 rems (3 rems/quarter) that could be received under the present Part 20 limitations using the 5(N - 18) formula. The initial ICRP proposal would have permitted a 15-rem dose in 1 year, 10 rems from planned special exposures and 5 rems from routine operation.

Comment: Subtraction of emergency doses. Some commenters suggested that doses received under emergency conditions, up to a lifetime total of 25 rems, not be subtracted from the lifetime allowance for planned special exposures. It was also suggested that the employability of the individual might be jeopardized if the dose "bank" were depleted.

Response: The NRC has not officially sanctioned the 25-rem "forgivable" emergency dose that has been recommended by some organizations for a once-in-alifetime dose that would not be counted against an individual's lifetime dose. Consequently, all doses received as a result of occupational exposure must be recorded in an individual worker's record.

The Commission believes that planned special exposures will be used infrequently so that the lack of a dose bank for some individuals would not be a major drawback to their employability. Comment: The time period for notifying exposed individuals of their dose is too short. A number of commenters thought that the 15-day period for notifying exposed individuals of their exposure from a planned special exposure was too short. Some commenters noted that most NRC reporting requirements provide a 30-day, not a 15-day, period. Other commenters suggested that the 15-day period could give the impression [to the worker] that an inordinate risk was involved when that was not the case.

Response: The 15-day period for notification was intended to be unique and to further emphasize that "planned special exposures" were indeed "special." However, the Commission has extended the time period for notification of the individual from 15 days to 30 days to allow licensees additional time to estimate internal exposures that are now permitted in the revised rule to be part of a planned special exposure. The requirement to notify the NRC (see § 20.1204) that a planned special exposure has taken place is also 30 days.

Comment: Doses received during a planned special exposure that do not exceed the dose limits for normal operation should not have to be recorded as planned special exposures or be subtracted from the lifetime planned special exposure limit. A few commenters expressed concern that exposures during planned special exposures that did not result in doses to an individual in excess of the occupational annual dose limits would nevertheless have to be reported separately and subtracted from the individual's lifetime allotment for planned special exposures.

Response: The intent of the planned special exposure was that it would be used infrequently in circumstances where the elimination of the 5(N - 18) lifetime cumulative limit might create a severe handicap to the licensee's operations. Being able to switch doses between planned special exposures and routine dose limits would tend to encourage the use of planned special exposures as the licensee would have nothing to lose by using the planned special exposure. This is contrary to the Commission's intent that the planned special exposures be restricted to "special" situations. Once a licensee decides to conduct a planned special exposure, all of the unique limitations, reporting, and recordkeeping requirements are to apply, even if the doses actually received fall within the dose limits for routine operations. Final Rule: The provisions of planned special exposures have been extended to include internal exposures, and the reporting time to the individuals involved has been changed to 30 days to allow sufficient time for analysis of internal dose.

Section 20.207 Occupational Dose Limits for Minors.

Comment: Exposure of Minors. One commenter stated that minors should not be exposed to radiation because they do not meet the criteria for occupational radiation exposure. The commenter argued that minors are not trained regarding radiation protection, do not derive a benefit from employment, and would require the preparation of an NRC Form 4 if they were workers.

Response: Allowing minors to be occupationally exposed to radiation was permitted in the present Part 20 (§ 20.104). All individuals, including minors, who enter a restricted area are required (10 CFR 19.12) to be instructed as to the risks involved. Minors who are employed receive salaries and other associated benefits of employment so that there does not appear to be a major difference in this respect from other workers. Furthermore, licensees are required under the existing and revised Part 20 rules to maintain the same exposure re cords for minors as for adults.

An alternative to this procedure would be to exclude minors completely from radiation-related work. This does not appear to be desirable as the monetary, experience, and educational banefits that may accrue to the minor appear to outweigh the small incremental risk involved (particularly considering the reduced dose limits applied to minors). Consequently, no change has been made from the proposed rule.

Section 20.208 Dose to an Embryo/Fetus.

Comment: Biological basis for lower dose limits for pregnant women. There were comments that cited older studies and recommendations for dose limits for the embryo/fetus that are considerably higher than 0.5 rem. These comments questioned the biological basis for the 0.5 rem dose limit for the embryo/ fetus in the proposed rule.

Response: The biological effects of ionizing radiation upon the embryo/ fetus are summarized in Regulatory Guide 8.13.23 The limit of 0.5 rem during the entire gestation period is based upon a recommendation by the NCRP in 1977. 24 The International Commission on Radiological Protection (ICRP-26)1 recommended 0.3 times the annual dose limit or 15 mSv (1.5 rems) over the full gestation period and 5 mSv (0.5 rem) in the first 2 months of pregnancy. More detailed information can be found in publications of the NCRP, 24 ICRP, 25 UNSCEAR, 26 and the OECD/NEA. 27

Final Rule. The limit for the embryo/fetus of a declared pregnant woman is 0.5 rem over the entire gestation period. There is also an admonition that the licensee avoid substantial variation above the average monthly exposure rate that would comply with the 0.5-rem limit. These conditions are consistent with the Federal guidance on occupational radiation exposure and with the recommendations of the NCRP in NCRP Report No. 916.

Comment: Licensee's Responsibilities to Protect the Embryo/Fetus of an Undeclared Pregnant Woman. Several commenters raised the question of whether the licensee had any responsibility for protecting the embryo/fetus of an obviously pregnant female employee who had not formally declared her pregnancy to the employer.

<sup>23</sup> 

U.S. Nuclear Regulatory Commission, "Instructions Concerning Prenatal Radiation Exposure," Regulatory Guide 8.13, Rev. 2, December 1987. National Council on Radiation Protection and Measurements, "Review of Radiation Dose Limit for Embryo and Fetus in Occupationally Exposed Women," NCRP Report No. 53 (1977). (Available for sale from the NCRP, 7910 Wood-24 mont Avenue, Suite 800, Bethesa: MD 20814-3095.)

International Commission on Radiological Protection, "Developmental Effects of Irradiation on the Brain of the Embryo and Fetus," Annals of the ICRP 25 16 4) (1986). (Available for sale from Pergamon, Press, Inc., Inc., Elmsiord, NY 10523.)

United Nations Scientific Committee on the Effects of Atomic Radiation 26 (UNSCEAR), Genetic and Somatic Effects of Ionizing Radiation Sales Section, United Nations, NY 1986, particularly Chapter III, Biological Effects of Pre-natal Irradiation."

<sup>27</sup> Organization for Economic Cooperation and Development/Nuclear Energy Agency, "The Biological Basis for the Control of Prenatal Irradiation," OECD/NEA, Paris, France (1988).

Response: It is the fundamental responsibility of the pregnant worker to decide when or whether she will formally declare her condition to her employer. This position is derived from court rulings concerning a pregnant woman's rights regarding termination of the pregnancy. Having a woman formally declaring her pregnancy to her employer derives from legal, not health protection, considerations. If she chooses not to declare her pregnancy, the licensee will not be required under the Commission's regulations to limit her dose to the 0.5-rem limit.

Undeclared pregnant women are protected under the NRC regulations for all workers. The normal occupational dose limits would still be in effect and would have to be complied with, and the dose would also have to be kept "as low as is reasonably achievable." In addition, as part of her initial employment, the woman should have received instructions in radiation protection (10 CFR 19.12), and she should have been provided with a copy of Regulatory Guide 8.13.

It might be prudent for a licensee to remind a pregnant, but undeclared, worker of the special limit for protection of the embryo/fetus of a declared pregnant woman and to provide another copy of Regulatory Guide 8.13 to her. However, if the licensee has previously provided this information to the employee, it is not a Commission requirement that it be done again. If the requirements referred to in the previous paragraph have been fulfilled, the licensee will not be cited for a violation of the Commission's regulations if the estimated dose to the embryo/fetus of an undeclared pregnant woman exceeds the 0.5-rem limit, even if the worker's pregnant state seems obvious.

Response: Section 161c. of the Atomic Energy Act gives NRC the authority to require such information to be provided by the worker. However, such a requirement could be considered to be discriminatory and an invasion of personal privacy. It would also be unenforceable because the woman and her physician know when she knew of the pregnancy and patient-doctor communications are privileged. Infringement on personal privacy is also a drawback that applies to requiring the female worker to supply information concerning her "fertility" or "infertility."

Comment: Estimation of Dose to the Embryo/Fetus. The assignment to the embryo/fetus of a dose equal to the dose to the declared pregnant woman was questioned. For example, would it be reasonable to assign to the embryo/fetus a dose based upon the dose received by the woman's shoulder or head?

Commenters also indicated that licensees should be permitted to employ factors other than a factor of 2 and take into account shielding of the embryo/ fetus by maternal organs and the placenta in evaluating the external dose component of the embryo/fetus.

Response: The concept used in the proposed rule of relating the dose to the embryo/fetus to the dose received by the mother has been modified. The final rule permits direct calculation of the dose to the embryo/fetus. This was done so that the use of more accurate dose assessments would not be precluded by the rule. The infernal dose to the embryo/fetus may or may not be directly proportional to the dose received by the mother.

A forthcoming regulatory guide will provide guidance on methods for calculating the dose to the embryo/fetus. For interim assessments of the dose to the embryo/fetus, it may be assumed that the dose to the embryo/fetus from external radiation and from radionuclides in the body that are relatively uniformly distributed, such as cesium-137 and compounds of tritium and carbon-14 that are not organically bound, is the same as the dose to the mother since under these circumstances the same energy would be deposited per gram of tissue in both the mother and the fetus. For external gamma irradiation, the assumption that the dose to the fetus is the same as to the mother should be conservative (yield calculated doses that are somewhat higher than the actual doses determined by more precise evaluations).

Permitting calculations of the embryo/fetal dose using reduction factors for attenuation within the body of the mother would entail knowledge of the energy spectra of the incident radiation. As noted previously (Response for § 20.201), photon spectral measurements, although technically feasible, are not currently required by the Commission and are considered to be beyond the scope of routine radiation protection survey measurements. The small amount of reduction in the calculated dose afforded by such attenuation corrections

would be secondary in importance compared to uncertainties due to body orientation, partial-body exposure from collimated beams of radiation, and the radiobiological sensitivity of the embryo/fetus.

In situations where the use of a single dose measurement would be inappropriate for both the woman and the embryo/fetus, a solution would be to monitor the two doses separately.

Comment: Additional Dose Increment Allowed to Pregnant Women Beyond the Dose Limits. The rationale was requested by a few commenters for permitting an extra 0.05 rem (0.5 millisievert) beyond the 0.5-rem (5 millisieverts) dose limit to an embryo/fetus.

Response: The small additional dose is intended to apply in situations where the embryo/fetus has accumulated a substantial fraction of the dose limit or has already exceeded the limit before the woman formally declares herself to be a "declared pregnant woman." If the incremental 0.05-rem dose were not available, a woman having already received a dose in excess of the 0.5-rem limit might not be able to be further employed in a radiation-related job. The licensee could be in "instant noncompliance" as the embryo/fetus dose limit could have been exceeded before the licensee was aware that it was applicable (i.e., before the woman declared her pregnancy). Thus, the small incremental 0.05-rem dose provides a means of ensuring continued employment for the woman and also removes the threat of inadvertent noncompliance on the part of the licensee. The additional risk posed by this incremental dose to the embryo/ fetus is small compared to the potential risk from the overall 0.5-rem dose limit.

Final Rule: The final rule corrects an anomaly in the proposed rule regarding the application of the additional 0.05-rem incremental dose. In the proposed rule, the additional 0.05-rem dose was available if the embryo/ fetal dose limit had been exceeded prior to the woman's declaration of pregnancy (even if the dose were 0.501 rem). However, the additional 0.05-rem dose increment would not have been available if the embryo/fetal dose were less than the

0.5-rem limit (even if the dose were as much as 0.499 rem). There is no significant difference in risk between 0.551 (0.501 + 0.05) rem and 0.549 (0.439 + 0.05) rem. This provision would have resulted in unnecessary penalties to both the licensee and the declared prednant woman. In the final rule, the 0.05-rem dose increment is available as an additional dose if the embryo/fetal dose at the time of declaration is greater than 0.45 rem (0.45 = 0.5 - 0.05).

### Subpart D--Radiation Dose Limits for Individual Members of the Public

Section 20.301 Dose Limits for Individual Members of the Public.

Comment: NRC should defer changes to limits for the general public until the EPA issues revised Federal guidance. The EPA suggested that NRC not modify its radiation limits for protection of the general public until EPA prepares revised Federal guidance on dose limits applicable to the general public (the recently issued Federal guidance applied only to occupational radiation protection).

Response: Although it would be desirable to use Federal guidance as a basis for the revision of the limits for the public, the Commission believes that Part 20 needs to be based on a consistent set of principles and concepts rather than having its standards for workers using one dose limitation system and its standards for the general public using an entirely different (and out-moded) system. The latest Federal guidance does not address radiation exposure of the general public and, although the NRC staff is represented on an EPA Task Group which is developing draft Federal guidance on doses to members of the general public, the Commission has chosen not to defer these limits until this Task Group has completed drafting the guidance and EPA makes recommendations to the President for its issuance. The Commission's intent to address these limits was noted explicitly in the statement of considerations that accompanied the proposed rule (51 FR 1118, Section XXVIII).

Comment: Facilities that are subject to other lower standards should not have to demonstrate compliance with the 0.1-rem limit ["reference level"]. Several commenters expressed concern that additional efforts would be required to demonstrate compliance with the proposed 0.1-rem "reference level." For licensees that were already subject to the 0.025-rem (25-millirem) limits of EPA's 40 CFR Part 190, this appeared to be an unnecessary burden.

Response: The concept that 0.1 rem represents a "Reference Level" has been eliminated. The 0.1-rem value in the final rule represents the primary dose limit for protection of the public. This change from the proposed rule reflects the clarifications by the ICRP (see Section II.A.) regarding the usage of the 0.1-rem and 0.5-rem recommended dose levels. This change does not represent a major change from the proposed rule. Many commenters had indicated a belief tilat, because of the reporting and control requirements associated with the 0.1-rem reference level, it already represented a de facto limit.

Demonstration of compliance with the limits in 40 CFR Part 190 or with the design objectives of Appendix I to 10 CFR Part 50 will be deemed to demonstrate compliance with the 0.1-rem dose limit for most licensed facilities. Power reactor licensees that comply with Appendix I may also have to demonstrate that they are within the 0.025-rem limit in 40 CFR Part 190. Demonstration of compliance with the limits of 40 CFR Part 190 will be considered to demonstrate compliance with the 0.1-rem limit. For uranium mills, it will be necessary to show that the dose from radon and its daughters, when added to the dose calculated for 40 CFR Part 190 compliance, does not exceed 0.1 rem.

The dose rate limit of 2 millirems in any 1 hour from § 20.105(b)(1) of the present Part 20 was omitted in the proposed rule but has been reinstated in the revised rule. The reason for this is that this limit provides a more readily measurable quantity than the 100 millirem per year value and can be more easily verified by short-term measurements.

Ċ

Comment: Inclusion of doses from other licensed or unlicensed radiation sources. Many commenters expressed an opinion that the dose should not be allinclusive and should not include fallout from nuclear weapons tests, transportation of radioactive Naterial, or other sources of radiation not under the control of the licensee.

Response: The new lower dose limit for members of the general public (which was described as a "reference level" in the proposed rule) applies only to doses from radiation and radioactive materials under the licensee's control. The EPA's generally applicable environmental radiation limit for nuclear power operations (40 CFR Part 190) does apply to the total dose from all sources within the uranium fuel cycle. However, in its practical implementation, the sources would have to be located within a few miles of each other for the combined dose contributions to be significantly different from the dose from either facility alone.

The definition of "natural background" has been replaced by "background radiation," which includes natural background, global fallout, and radon not associated with licensed material. This clarifies sources of radiation and radionuclides that can be excluded from evaluations of the dose from licensed activities.

Comment: Differentiation of limits for long-term operation and for shorterterm transient operation. A number of commenters noted that ICRP-26 described the 0.1 rem (1 mSv) per year value as intended to be an average goal for longterm operation but that 0.5 rem (5 mSv) was intended as the primary annual dose limit for members of the public. Some commenters suggested that a lifetime dose limit be established for members of the public.

Response: As noted above in Section II.A., the ICRP has modified its interpretation in the ICRP statement issued following their 1985 Paris meeting.<sup>2</sup> so that the primary standard is 1 mSv (0.1 rem) per year. This clarification of ICRP philosophy is reflected in Part 20 by the change of the 0.1 rem per year value from a "reference level" in the proposed rule to a primary limit in the final rule.

Final Rule: It should be emphasized that the 0.1 rem per year limit in Pert 20 is not intended to be applied as a long-term average goal: it is an annual limit. As a matter of practicality, long-term (or lifetime) dose limits for members of the public cannot be implemented unless each year's dose is kept within the long-term goal. Doses to individuals in the general public are not usually monitored directly (locations rather than individuals in the offsite environment are monitored). As individuals may change residency and there is no reporting or tracking system, lifetime doses to specific individuals in the general population are very difficult to determine.

The 0.5 rem per year limit is available only upon specific application to and approval by the Commission (see § 20.301(c)). A 0.5-rem value has been retained in order to apply to transient situations and to alleviate the immediate need to redesign or reshield existing facilities that were designed to meet the former 0.5-rem limit. The 0.5-rem limit is intended to be applied primarily to temporary situations where operation of a facility, or the person's exposure to radiation and radioactive emissions, is not expected to result in doses above 0.1 rem over long periods of time. For design of new installations, the 0.1-rem limit should be used. However, existing facilities may apply for NRC approval to use the 0.5-rem limit while more complete evaluation of the need for any additional modifications is performed.

The Commission is aware that some categories of licensees, such as uranium mills and <u>in situ</u> uranium mining facilities, may experience difficulties in determining compliance with the revised values in Appendix B, Table 2, for radionuclides such as radon-222. Provision has been made for licensees to use air and water concentration limits for protection of members of the general public that are different from those in Appendix B, Table 2, if the licensee can demonstrate that the physicochemical properties of the effluent justify such modification and the revised value is approved by the NRC. For example, uranium mill licensees could, under this provision, adjust the Table 2 value for radon (with daughters) to take into account the actual degree of equilibrium present in the environment. This provision permits (upon NRC approval) the use of concentration limits for members of the general public that better represent actual exposure conditions. This is similar to the allowance for use of modified derived air concentrations (with Commission approval) in § 20.204(c)(3).

In both situations, licensees would be permitted to propose radionuclide concentration limits for their facility that reflect actual properties of the effluents rather than using the generic concentration-to-dose assumptions associated with Appendix B values. These adjustments tailor the concentration limits to specific conditions, provide the same limitation of dose, and do not permit any greater risk even though the adjusted concentration limits (for members of the general public or for workers) may be higher than the Appendix B generic values.

Use of this provision, applied to the percentage of radionuclide equilibrium existing in radioactive decay chains, could provide a factor of 2 or 3 upward change in the appropriate air concentration limit. In addition, the licensee can demonstrate compliance by calculating the dose to the nearest resident rather than meeting the air concentration limit at the site boundary. This should provide an additional factor of 2 or 3 allowance. Lastly, if the 0.1-rem effective dose limit still cannot be met, the licensee can apply to NRC under § 20.301(c) for permission to use a temporary 0.5 rem per year limit rather than the 0.1 rem per year limit. Section 20.301(c) of the revised rule requires that, in order to receive permission for use of this higher dose limit, the licensee has to specify (1) the need for and expected duration of the higher value, (2) their program to assess and control doses, and (3) procedures to control doses to be ALARA. These options used singularly or in combination coupled with process or operational modifications of these facilities is expected to provide sufficient flexibility to enable most uranium recovery facilities to comply with the provisions of the revised 10 CFR Part 20.

Section 20.303 [Reserved].

The former 0.1-rem "Reference Level" and the EPA Standard for Nuclear Power Operations that were in this section in the proposed rule are included as primary limits for members of the public in § 20.301 of the final rule. Section 20.304 [Deleted] De Minimis Level and Collective Dose Evaluations.

Comment: Adoption of a threshold for calculating collective (population) doses. The proposed § 20.304 would have allowed licensees to disregard doses to individuals that were less than 1 millirem per year when evaluating collective (population or "person-rem") doses. A major criticism of this section was the narrowness of its scope. The section pertained only to a change in the calculational methodology for estimating collective doses and would not have permitted unrestricted release of any materials or equipment.

Most comments from people and organizations within the nuclear power and radiation applications industry favored this measure as an initial step toward developing more general "below regulatory concern" (BRC) levels. Several commenters thought that NRC acknowledgment of the concept of a BRC level was more important than the specific proposal to truncate collective dose calculations. Many commenters thought that a generic BRC level would limit unnecessary expenditure of resources that would otherwise have to be spert to control inconsequential risks.

There were also a number of comments that were not in favor of either the proposed collective dose cutoff or the more general application of the concept of below regulatory concern. A few commenters expressed opinions that it did not appear feasible to arrive at a universal <u>de minimis</u> level because the level that would appear to be truly insignificant to most people would be too low to result in any appreciable saving to the industry. There also were comments that noted that the proposed collective dose cutoff could cause large numbers of potential adverse health effects to be overlooked if they resulted from small radiation doses delivered to very large numbers of people. Many commenters, both pro and con regarding the adoption of a BRC level, thought that a threshold value for collective dose should also be developed. A few commenters noted that the focus of the more generic BRC concept tended to be for single licensees and that it might be necessary to consider the impacts from multiple licensees.

Many of the commenters who supported a generic BRC concept did not agree with the numerical value (0.001 rem per year) proposed for the cutoff, believing it to be too low. An explanation for this opinion was that if 0.001 rem represented an insignificant level of risk, then all larger doses might be perceived as representing "significant" levels of risk. A value of 0.010 rem was noted by several commenters as being a more suitable value and still represented an inconsequential risk.

Response: The Commission agrees that "Below Regulatory Concern" levels would be useful and has issued policy statements on the application of the concept of below regulatory concern with regard to waste disposal ("Radioactive Waste Below Regulatory Concern," Federal Register of August 29, 1986 (51 FR 30839)) and a general policy statement on below regulatory concern was announced on June 27, 1990, and was subsequently published in the Federal Register on July 3, 1990 (55 FR 27522). The general policy statement establishes the framework for the Commission to formulate rules and licensing decisions to exempt certain practices involving small quantities of radioactive materials from some or all regulatory controls. The BRC policy statement sets forth criteria for protection of both individuals (individual dose criteria) and population groups (a collective dose criterion).

In order to ensure that any computational changes reflect the policy that evolves from the effort to develop generic BRC policy, the Commission removed the threshold for truncating collective doses (§ 20.304) from Part 20 and has included such a threshold in the generic BRC policy statement. This deletion is also consistent with comments that noted that this section described a method for calculating a quantity (collective dose) that was not required to be calculated by Part 20 and comments that such details of calculations would be better in a regulatory guide rather than in a regulation.

#### Subpart E--[Reserved]

#### Subpart F -- Surveys and Monitoring

Section 20.501 Surveys.

Comment: Accreditation of Personnel Monitoring Processors. There were a number of comments concerning the desirability of requiring accreditation of personnel dosimetry processors.

It was also noted that the National Voluntary Laboratory Accreditation Program (NVLAP) does not provide accreditation for doses delivered to the lens of the eye, a depth equivalent to approximately 0.3 centimeter (an areal density of 300 milligrams per square centimeter). The only tissue depth equivalents that are accredited at this time are 1.0 centimeter (the deep-dose equivalent) and 0.007 centimeter (the shallow or "skin" dose equivalent).

Response: The issuance of a dosimetry accreditation requirement or "NVLAP Rule" overlapped the Part 20 rulemaking. Because this issue was the subject of a recent separate NRC rulemaking, issues concerning the desirability of such a program were considered and addressed in the rulemaking on accreditation. No revision from the dosimeter processor accreditation rule (52 FR 4601) has been made, and the final Part 20 rule incorporates the final form of the accreditation rule.

As noted in the discussion of the "eye dose equivalent" in Section XI, "Standards for Occupational Exposure of Individuals," of the proposed Part 20 rule, the Commission believes that compliance with the eye dose limit will be generally ensured by compliance with the deep-dose limit. Consequently, the lack of accreditation for this depth should not have a major impact on the degree of protection of the eye.

Comment: The accreditation requirement requires the use of a commercial dosimetry service.

Response: This is an incorrect interpretation of the dosimetry accreditation rule (52 FR 4601). That rule, which is incorporated into the revised Part 20, states that the dosimetry processor must be accredited. It is possible for licensees that provide their own dosimetry services to be accredited.

Comment: Lack of specificity in monitoring requirements. Commenters noted that the monitoring requirements, both in the present Part 20 and in the proposed rule, were general and imprecise.

Response: Many portions of Part 20 are not very specific and detailed because Part 20 contains the NRC's general radiation protection requirements and applies to all classes of licensees, including large power reactors, universities, and medical institutions as well as small radionuclide and sealed source users. Because of this breadth of application, the requirements in Part 20 cannot be very detailed for any one type of facility. However, the requirements in Part 20 are designed to provide the framework for all licensees and to establish provisions that the NRC considers to be fundamental to basic radiation protection.

## Section 20.502 Conditions Requiring Individual Monitoring of External and Internal Occupational Dose.

Comment: Monitoring Thresholds. A number of commenters questioned the rationale for the lack of agreement of the thresholds in the proposed rule for monitoring external doses (10 percent of the annual limits) and for requiring monitoring of internal doses (30 percent of the annual limit). It was frequently mentioned that starting to require monitoring at 30 percent of the dose limit could result in overlooking dose f 1.5 rems (30 percent of 5 rems). The 1.5-rem value would have been above the limits for minors and for the embryo/fetus (0.5 rem) and was characterized as being a rather substantial fraction of the deep-dose equivalent limit. In this connection, it was also noted that the possibility existed, when large external doses were expected, of exceeding a total effective dose equivalent limit of 5 rems because the licensee was not aware of the internal dose contribution.

Some commenters thought that the monitoring thresholds would be understood more easily if they were expressed as doses instead of percentages.

Response: The unequal thresholds for requiring monitoring of internal doses (30 percent of the dose limit) and external doses (10 percent of the dose limit) were originally set because of the difficulties in performing low-level bioassay analyses of alpha-emitting radionuclides at fuel fabrication and other facilities where actinides may be prevalent. (Bioassays for the radionuclides most commonly found at nuclear power reactors were viewed as generally being able to meet the 10 percent threshold set for external doses.) In situations such as bioassay for alpha-emitting radionuclides, it may be difficult to detect 10 percent of the ALI or 10 percent of the dose limit by bioassay measurements on excreta.

The monitoring threshold is a predetermined level of anticipated dose for carrying out bioassay procedures and does not represent a required level of detection sensitivity. If, by a reasonable analysis of the working environment, it appears that a worker is likely to inhale radioactive materials at concentrations that could produce an annual committed effective dose equivalent of 0.5 rem (10 percent of the 5-rem limit) or more, then that worker's intake should be monitored using measurements of exposure (e.g., estimates of DAC-hours based upon measured air concentrations) or intake (such as by whole-body counting or other bioassay technique) or by measurements of both exposure and intake. Whether the actual doses received were in excess of 10 percent of the limits could only be determined from these subsequent measurements.

The monitoring thresholds are specified as percentages of the dose limits rather than as doses because the thresholds apply to several different dose limits: the total effective dose equivalent, the eye dose equivalent, and the shallow-dose equivalent.

Final Rule: The threshold for monitoring internal doses has been dropped from 30 percent of the dose limit to 10 percent of the limit. This provides consistency in the internal and external monitoring requirements. The Commission acknowledges that, in some cases, particularly bicassay measurements of transuranic elements, it may not be feasible to actually confirm such levels

by bioassay. However, the monitoring threshold is not a requirement on the capability of the measurement. Average airborne radionuclide concentrations and the expected time of exposure can be used to estimate radionuclide intakes and the need for bioassay or other monitoring methods.

The Commission intends to issue a regulatory guide on the procedures to be used in estimating committed effective dose equivalents and deep-dose equivalents and guidance on when they have to be summed.

Comment: Evaluation of radionuclide intakes for respirator wearers. Several commenters mentioned that internal dose monitoring, such as bioassays, should not be required solely because respiratory protection devices were used. The rationale given by the commenters was that the requirement provides a negative incentive for using respirators and is, therefore, counter to ALARA operating practices.

Response: The requirement (in § 20.502(b)(3)of the proposed rule) for bioassays for anyone using respiratory protection has been dropped. The Commission agrees that such a requirement might be a disincentive for using respirators as part of an ALARA effort. There is, however, a requirement (in § 20.703) for bioassays to be conducted, as appropriate, as part of a respiratory protection program. Whether bioassays are necessary for a particular individual will depend upon whether that individual could have exceeded 10 percent of the annual limit on intake (ALI) or was exposed to airborne radionuclide concentrations in excess of the monitoring threshold. An evaluation of internal dose would be required if there were a potential for exceeding 10 percent of an annual limit on intake (0.1 ALI), whether or not a respirator is worn.

[Note: Because the requirement for performing bioassays for a particular individual has been separated from the wearing of a respirator, the concentrations to be used for evaluating monitoring thresholds are those of the ambient atmosphere before credit is taken for respiratory protective factors. One of the purposes of such bioassays is to confirm the effectiveness of the respiratory protection being provided. If bioassay were made dependent upon the corrected air concentration (after dividing by the protection factor), it would be

equivalent to assuming that the intended protection factor was correct without further verification.]

## Subpart G--Control of Exposure from External Sources in Restricted Areas

Sections 20.601, 20.602, and 20.603 Control of Access to High and Very High Radiation Areas.

Comment: Inapplicability of requirements to nuclear power reactors. Many commenters indicated that the proposed requirements for control of entry into very high radiation areas could not be applied to nuclear power reactors because of the number and size of potential "very high radiation areas" and the physical inability to restrict access to these areas. Similarly, interlocks that can result in the withdrawal or cessation of the radiation source may be unworkable in nuclear power reactors. Several commenters proposed incorporating requirements for power reactors that are similar to reactor license conditions in reactor technical specifications.

Response: The Commission recognizes that the detailed requirements applicable to large irradiators that were formerly in § 20.203(c)(6) should be in a specific regulation dealing with these facilities rather than in Part 20. For this reason, these detailed requirements will be placed in a future Part 36 of Title 10 which is being issued for public comment and applies specifically to irradiators. At the time that that rule is made effective, the Commission will transfer these requirements from Part 20 to Part 36. In the meantime, the NRC staff will issue a regulatory guide that provides more specific detailed guidance for nuclear power reactors on high and verv high radiation areas.

Comment: Choice of Dose Rate Defining a "Very High Radiation Area." Several commenters believed that the 500 rads per hour dose rate that defines a "very high radiation area" was too high, noting the proximity of this value to the median lethal dose  $(LD_{50})$  for acute radiation exposures. Alternative values, such as 1 rem per hour at 30 centimeters, were proposed. Response: The seriousness of this dose rate was a factor in its adoption. The 500 rads per hour value appears in the previous 10 CFR 20.203(c)(6) as a criterion for additional access controls for irradiators (similar in scope to the requirements of § 20.603 in the final rule). However, the previous Part 20 did not use a unique designation such as the "very high radiation area" designation used in the proposed and revised Part 20 rules. The difference between the 1 rem per hour definition of a "very-high" radiation area used in reactor technical specifications and the 500 rads per hour definition used in the revised Part 20 is discussed in a regulatory guide currently being prepared.

Comment: Meaning of "direct surveillance." Several commenters thought that the term "direct surveillance" used in the proposed § 20.601 could be interpreted to require stationing an observer at the entrance to the "high" or "very high" radiation areas.

Response: The final rule permits "...continuous direct or electronic surveillance over a high radiation area that is capable of preventing unauthorized entry..." This removes the burden of having to station a person in or near a "radiation area," but requires interlocks or electronic locks so that the remotely located observer may prevent entry into the area when necessary.

Final Rule: The section on very high radiation areas has been divided into two sections. Section 20.602 provides a general requirement for restricting access to such areas. This general requirement applies to all very high radiation areas, regardless of the type of licensed operation, including those at nuclear power reactors. A second, more detailed, set of requirements applies only to large gamma irradiators. This section, § 20.603, restates requirements for irradiators that are in § 20.203(c)(6) of the present 10 CFR Part 20.

## Subpart H -- Respiratory Protection and Controls to Restrict Internal Exposure - Restricted Areas

Sections 20.701 and 20.702 Use of Process or Other Engineering Controls and Use of Other Controls.

Comment: "Use of other controls." Commenters suggested that, if workers could be exposed to concentrations of radioactive materials greater than 1 derived air concentration, ALARA should be applied to the total of internal and external doses (to the total effective dose equivalent). It was noted that this condition was included in the Federal Guidance on Occupational Radiation Exposure.

Response: Modifications have been made in the final rule to permit ALARA considerations to apply to the total effective dose equivalent rather than just the internal dose portion.

Comment: Some commenters indicated that the use of respirators should be permitted even if their use would not be able to reduce airborne concentrations below 1 DAC. They noted that this would be consistent with the ALARA philosophy.

Response: Section 20.702 has been rewritten to clarify the intent that the concentration of 1 DAC is not a cutoff on the voluntary use of respirators but is intended to be the point where some corrective action (including, but not limited to, the use of respirators) by the licensee would be required when the use of ventilation and process controls cannot further reduce the airborne concentrations of radioactive materials.

Section 20.703 Use of Individual Respiratory Protection Equipment.

Comment: The proposed rule permits low estimates but not high estimates of intake to be corrected. Commenters noted the the proposed rule (§ 20.703(a)(1)) was not balanced as correction of intake estimates based upon dividing DAC-hours by the respirator protection factor and was only permitted if the initial estimate was later shown (by bioassay results) to have been low.

Response: The rule has been modified so that corrected estimates of actual intake can be used in records in place of earlier estimated intakes, regardless of whether the change would result in an increase or in a decrease in the intake estimate.

Comment: NRC should provide a recommended minimum acceptable standard for determining an individual's physical fitness for respirator use. Part 20 requires that a physician determine that an individual worker is physically able to wear a respirator. NRC should, therefore, provide guidance to the physician on minimum standards for wearing respirators.

Response: The NRC policy is that the decision as to medical fitness has been, and continues to be, left to the physician; i.e., the medical doctor should decide what constitutes minimum health standards for respirator wearers. Furthermore, the requirements may vary, depending on the respirator used and physical situations, such as the type of work to be performed, which are outside the scope of Part 20. Licensees desiring more guidance should obtain ANSI Standard Z03.6(1984), "For Respiratory Protection -- Respirator Use --Physical Qualifications For Personnel," which was developed as an industry consensus standard that provides definitive guidance to "identify the responsibilities of the physician, the employee, and management in determining the employee's ability to use a respirator."

Comment: NRC should permit a health professional to certify physical capability to use a respirator rather than requiring a physician to perform each required certification. The proposed rule requires that a physician annually certify a worker's physical suitability for using a respirator. This should be broadened to permit any qualified health professional, acting under a physician's orders, to perform the actual certification rather than requiring a doctor to do this.

Response: As noted in the previous response, the decision on the physical ability of an individual to wear a respirator is a subjective judgment that, in the Commission's opinion, requires the decisionmaker to have a medical degree. The Commission notes that this annual certification could easily be included in an annual physical checkup.

Comment: The selection of respirator protection factors based upon "average concentrations" and not "peak airborne concentrations" is an improvement. The proposed rule, unlike the previous Part 20, permitted protection factors to be applied to the time-averaged air concentration rather than the peak air concentration.

Response: Despite some favorable comments on this change, the Commission has determined that the use of the average airborne concentration may not provide an adequate margin for health protection and, in the final rule, has reverted to the use of the anticipated peak concentration.

Final Rule: The proposed rule has been modified to require a respiratory protection program when respiratory protection devices are being used to limit intakes, whether or not credit is taken for respiratory protection factors. Allowance has been made for use of respirators that do not provide protection factors that would keep exposures below the derived air concentrations if (and only if) such use would keep the total effective dose equivalent ALARA.

# Section 20.704 Further Restrictions on the Use of Respiratory Protection Equipment.

Comment: Section 20.704 should be deleted. This section, which states that the Commission may impose additional conditions on respirator use, is not necessary because § 20.1302 permits the NRC to place additional requirements on a licensee.

Response: Although the commenters are correct that § 20.1302 gives the Commission general authority to impose additional requirements on licensees, the Commission believes that the restatement of this policy in a section pertaining specifically to respiratory protection is desirable. As roted by the comments, this section does not create any additional requirement not otherwise contained in the regulations.

Final Rule: The requirements contained in the proposed rule are retained.

# Subpart I -- Storage and Control of Licensed Material

Sections 20.801 and 20.802 Security of Stored Material and Control of Material Not in Storage.

h

Comment: Definition of "secure." Saveral commenters requested a definition of the term "secure," which they felt was vague and did not provide an indication of the required licensee action.

Response: The phrase has been rearranged and now reads "secure from unauthorized removal or access," which is similar to the wording in the provides Part 20. This should provide sufficient clarification of what was intended by "secure." Comment: Unnecessary restrictions on research. One commenter thought that the requirement to secure small quantities of radioactive materials when they are not in use would interfere with university research.

Response: The Commission believes that locking radiotracer laboratories when they are not being used is a small nuisance compared to the consequences of unauthorized access to or theft of the radioactive materials, which could result in contamination of unrestricted areas or exposure of individuals, as well as having to report a loss of licensed material to the NRC.

#### Subpart J--Precautionary Procedures

Section 20.901 Caution Signs.

Comment: Black should be permitted as an acceptable color for the radiation warning symbol. Several commenters requested that the color black should also be allowed to be used on signs and for stenciling on packages. The fading of magenta inks in sunlight and the use of black for marking international shipments were cited as supporting this position.

Response: The Commission believes that, although the "magenta-on-yellow" color scheme has provided a unique warning of possible radiation hazards, black-on-yellow would also be acceptable. The fading of the magenta color as cited above may reduce the visibility of the sign with time. Because of the cost impacts if existing warning signs had to be replaced, the Commission is permitting the use of black in addition to continued approval of magenta and purple, rather than as a required replacement.

Final Rule: This section has been modified to add black as an acceptable color for the radiation warning symbol.

Section 20.902 Posting Requirements.

Comment: The terms "Caution" and "Danger" are not used consistently. Commenters noted that "Caution" or "Danger" could be used on signs for "Radiation Areas," "High Radiation Areas," and "Very High Radiation Areas" despite the considerable variation in the hazards that might exist in these different areas.

Response and Final Rule: The Commission agrees that the terms "Caution" and "Danger" should be used in a more consistent manner. The final rule permits only the term "Caution" to be used in "Radiation Areas." "Caution" or "Danger" may be used in "High Radiation Areas," since it covers a considerable range from 0.1 rem per hour to over 500 rads per hour. Only "Grave Danger" may be used in "Very High Radiation Areas." This should provide more emphasis on the use of "Danger," the importance of which might have been diminished by its prior applicability to the lower haz; "Radiation Area." "Caution" is inappropriate for use in "very high radiation areas" because of the potential hazard.

Comment: There should be a requirement to post all "restricted areas" whether or not it is a radiation or an airborne radioactivity area.

Response: The objective of posting is to warn personnel of a potential hazard. A "restricted area," <u>per se</u>, does not warrant such a warning. There is nothing to prevent a licensee from posting a notice designating a "restricted area," but such action is not required.

Comment: The definition of "airborne radioactivity area" would require tracking of employee "stay times" (time spent in the area). The second option to the definition of "airborne radioactivity area" would require performing surveys of airborne activity and tracking the time spent by workers in the area. The present rule would have only necessitated the survey. Response: There are two alternative definitions of an "airborne radioactivity area"; only the second one would require consideration of stay times. This second option does not require posting in areas that have low occupancy times and airborne radioactivity concentrations between 0.3 and 1.0 times the applicable DACs.

Comment: Areas containing only noble gases should not require posting as "airborne radioactivity areas." The hazard associated with such areas is primarily from external radiation.

Response: The DACs in Appendix 8 that apply to noble gases (and define an "airborne radioactivity area") are based upon submersion doses; therefore, the relationship remains valid. It should be noted that, because some shortlived noble gases have particulate daughters (such as <sup>88</sup>Rb and <sup>138</sup>Cs), the warning denoted by posting as an "airborne radioactivity area" may still be required.

Comment: There is no evident need to post all rooms containing 10 times the Appendix C levels. The requirement to post a caution sign in rooms that store ten times the Appendix C concentrations is unwarranted. There was some concern noted that such posting could deter firefighters or other emergency workers from entering an otherwise safe area, and increased damages could result.

Response: Complete dispersion of 10 times the Appendix C activities could produce air concentrations for some radionuclides in excess of the occupational DACs. For example, if 10 times the Appendix C quantities were dispersed in a 1,000 cubic foot (10 ft. x 10 ft. x 10 ft.) room, the resulting concentrations would be 35 times the DAC for organic carbon-14, 58 times the DAC for cesium-137, about 18 times the DACs for iodine-131 and tritium (water vapor), and approximately 6 times the DAC for technetium-99m. These appear to be sufficiently large to justify a posting requirement, particularly to caution firefighters in case of a fire.
Comment: The posting requirement should not be applied to sealed sources, such as gauges. Posting the entrances to areas having radioisotopic gauges could require multiple postings in large buildings.

Response: Posting is only required at entrances to the room containing the source and only when the dose rate at 30 centimeters would exceed 0.005 rem (0.05 mSv) in any hour (§ 20.903(c)) unless areas outside the room warrant posting as "radiation areas" and are already posted.

Section 20.903 Exceptions to Posting Requirements.

Comment: The proposed rule omits the past exemption for posting rooms containing only packages prepared for transportation.

Response: The Commission believes that there should be posting of these areas because there is no restriction on the length of time that packages may remain in a room. If the packages contain only small quantities of radioactive materials, then posting of the room would still be exempted under the remaining exemptions. The term "prepared for transportation" does include packages that are intended to be carried in a "sole use" vehicle. Such packages are permitted to have higher allowable dose rates than those specified in DOT (or NRC) limits for general shipment.

Final Rule: The exception for posting areas containing packages prepared for transportation has not been reinstated.

Comment: The requirement for a person in attendance would be unworkable in a hospital. The requirement (in lieu of posting the room containing a radiotherapy patient) for a person in attendance in order to prevent entry was interpreted as requiring a 24-hour escort for each radiotherapy patient.

Response: The intent was to generally require posting of therapy patients' rooms. (As noted in one of the comments, the dose rate from patients even with

diagnostic nuclear medicine treatments might exceed dose rates of 0.002 rem per hour.) The intent of "in attendance" would be satisfied by a duty nurse at a nursing station, providing that the station was in sight of the entrance to the patient's room.

Section 20.904 Labeling Containers.

Comment: There is no way to meet the requirement to label containers in some nuclear power plants or in hot cells. It is difficult to mark the detailed information on a container in some areas of a plant or in hot cells.

Response: Section 20.905 contains exceptions to the labeling requirements that take care of the problem noted by the commenter.

[Note: For the purpose of this section, "Mixed Fission Products" and "Fission and Activation Products" may be regarded as radionuclides, provided that the total activity is also specified. Designations as to the process stream or location sampled or type of sample (e.g., "primary coolant") may also be helpful as an additional designation of the potential hazard.]

Section 20.905 Exemptions to Labeling Requirements.

Comment: The proposed rule omits existing exemptions for packages containing only exempt quantities and those containing less than 10 mCi or less of tritium, iodine-125, carbon-14, and sulfur-35.

Response: While these sources pose little external hazard from gamma radiation, the quantities could be a potential internal hazard if the package were ruptured and the contents were released. Consequently, some warning remains appropriate.

Comment: The proposed rule omitted the existing exemption from labeling for packages labeled for shipment in accordance with DOT requirements.

Response and Final Rule: The exemption for DOT-labeled packages has been restored because the Commission agrees that the DOT labeling is sufficient to denote the presence of radioactive materials and provide an indication of any potential hazard. Quantities and concentrations not requiring DOT labels would not warrant an NRC labeling requirement. (See § 20.905(d).)

Section 20.906 Procedures for Handling Packages.

Comment: The requirement to monitor all packages is unnecessary. The requirement to monitor all incoming packages containing radioactive materials is unnecessary and in large installations creates a substantial monitoring burden.

Response: This requirement has been reevaluated and modified in order to reduce the burden.

Final Rule: Section 20.906 in the final rule requires incoming packages to be monitored when: (1) they are labeled as containing radioactive materials acco ng to DOT regulations, or (2) when a package is damaged or leaking. The first provision would reinstate the exemption from monitoring for shipments of small quantities of radioactive materials that would not require DOT labeling.

Comment: The requirement to survey external surfaces of packages is unnecessary. Several commenters with extensive experience in monitoring packages noted that externa' contamination was rarely if ever present and that wipe tests are time-consuming both to make the smears and to count them.

Response: Experience in the shipment of thousands of packages each year has been very good. However, potential problems with leaking packages during transit warrant continued monitoring upon receipt to ensure that leaking packages are found and reported. Appropriate action can then be taken to determine the extent of contamination in transport vehicles and storage areas in order to limit the consequences and avoid recurrence. However, an exemption from the

contamination survey requirement has been provided for special form (sealed) sources that are being moved to and from work sites in licensee owned or operated vehicles. This partially restores an exemption from the package survey requirements in the existing Part 20 (§ 20.205(b)(iii)) for all special form sources.

The Commission believes that restoring this exemption will not result in any additional hazard. An external radiation survey of the package is still required. The primary purpose of this external survey of sealed sources is to ensure that the source is still properly secured and shielded after transporting it.

Final Rule: The requirement to monitor external surfaces of packages has been retained and applies to the two classes of packages for which surveys are required (labeled "radioactive" and damaged or leaking). A partial exemption to sealed sources transported for field use has been reinstated because of the difficulty in making field measurements of surface contamination and because the transporting vehicle is not in general commerce.

Comment: The requirement to monitor packages within 3 hours is unwarranted. This requirement would be difficult to meet for several types of licensees, some of which do not have a full-time health physics staff person.

Response: Licensees receiving labeled packages of radioactive materials to which this requirement applies are expected to have available persons who are qualified to perform such monitoring. However, the person monitoring the package need not be a board-certified health physicist.

Final Rule: The 3-hour period in the surrent Part 20 (§ 20.205(b)(1)(5)) has been retained except if the package is received after normal working hours.

#### Subpart K--Waste Disposal

Section 20.1001 General Requirements.

Comment: Decay in storage as a disposal option. Many commenters noted favorably the addition of "decay in storage" as an allowed waste disposal option. Several commenters, however, did not believe that the option, as expressed in the proposed rule, was particularly helpful.

Response: Technically, the "decay in storage" option has always been available to a licensee since the license permitted possession of the radioactive materials and these materials naturally underwent radioactive decay. The option was formally included in the proposed and final rules because the list of disposal options is exclusive and there have been questions as to whether this was allowed under the previous Part 20. It should be noted that this option does not allow material that has "decayed in storage" to be released to unrestricted areas unless it meets the requirements of one of the other allowed forms of waste disposal in Part 20, or the requirements if § 35.92, "Decay-in-Storage," of 10 CFR Part 35, or the specific requirements given in any NRC or Agreement State license conditions.

The NRC staff considered adding a separate "Disposal by Docay in Storage" option with specific criteria for unrestricted release of material after decay. These criteria are commonly included in source and byproduct material licenses. However, the provisions included in 10 CFR 35.92 and certain specific license conditions pertain to relatively short-lived radionuclides and are neither appropriate nor applicable to other classes of licenses, such as those issued under Part 50. Also, when evaluated for a specific licensed activity, it is possible to consider existing pathways of exposure and to establish specific criteria for decay.

General criteria in a rule would need to be sufficiently conservative to take into account all reasonably conceivable pathways, thereby reducing the applicable level from what would be permitted in a case-by-case evaluation. Final Rule: The final rule has been modified to explicitly list "decayin-storage" as an authorized form of disposal. Section 20.1001 has been modified to incorporate the requirements that were in § 20.1002(b) of the proposed rule. These provisions require NRC licenses for persons who receive wastes containing licensed radioactive materials for treatment, for treatment or disposal by incineration, decay-in-storage, or disposal in facilities licensed under Part 60 or Part 61.

Section 20.1003 Disposal by Release into Sanitary Sewerage.

Comment: Removal of allowance for disposal of "dispersible wastes." A number of commenters felt that the restriction of wastes released to sanitary sewers to soluble wastes would have an adverse impact on certain licensees that, under the previous rule, had disposed of "dispersible" but insoluble radioactive materials. In particular, the practice was mentioned of grinding up animal carcasses with subsequent sewer disposal of the ground residue. This practice is permitted by the previous Part 20 but would not have been permitted under the proposed rule.

Response: In the final rule, the Commission has modified the conditions in the proposed rule for disposal of radioactive wastes into sanitary sever systems so that "dispersible biological materials" may continue to be disposed of by release to sanitary sewers. This means of disposal is advantageous compared with other alternatives for disposal of this type of biological material.

The prohibition on disposal of insoluble materials via the sanitary sewer was intended to prevent disposal via sanitary sewers of material in which the radioactive material is primarily in an insoluble form. Such materials may accumulate in the sewer system, in the sewer treatment plants, and in the sewer sludge.

Final Rule: The final rule permits disposal into sanitary sewers of: (1) radionuclides in soluble form or (2) radionuclides in readily dispersible biological material, provided that the limits in Appendix B. Table 3, on the

average monthly concentrations and the limits in § 20.1003(a)(4) on the total activity released annually are met. The revised rule no longer permits the disposal of nonbiological insoluble materials because of potential reconcentration of these materials in the sanitary sewer system, sewage treatment plants, and sewage sludge. This prohibition for insoluble materials is the reason why there are no values listed in Table 3 of Appendix B for insoluble materials.

Comment: The rationale for the reduction in the limits for sewer disposal is not explained. The concentration limits for radionuclides released to sanitary sewer systems in the proposed rule have been reduced by a factor of 10 from the former rule. This reduction did not appear to take into account the dilution afforded from multiple users of the sewer system. Commenters indicated that they thought that this reduction would increase the amount of material that would have to be disposed of via a low-level radioactive waste burial site and could result in increased radiation doses to workers having to package this material.

Response: The assumption noted by many commenters that radionuclides discharged into sanitary sewer systems are not ingested is not necessarily true because water in large lake or river systems may be recycled. The dilution afforded by having multiple users of a sewer system can be offset in part because there can also be several users that discharge radioactive wastes into the same sewer system. The revised Part 20 rule permits a higher concentration limit for discharges into sanitary sewers than for other liquid effluent releases of radioactive materials, but has lower concentration limits than were formerly allowed for sewage. In view of past contamination incidents (involving cobalt-60 and americium-241) and the reduction in the dose limit for members of the public, the Commission believes that continuation of the higher limits is no longer desirable.

The NRC has under way a study of the dose pathways associated with disposal of radioactive materials via sanitary sewers. This study will help clarify the potential for human exposure.

Comment: The exemption on disposal of human excreta should be removed. Hospitals should have to comply with the same regulations as other licensees.

Response: Disposal into a sanitary sewer system (which was designed specifically to handle this type of waste) is the preferred method of disposal because of the other health considerations in handling human excreta in addition to radiation protection. This exemption is in the current Part 20.

Section 20.1004 Treatment or Disposal by Incineration.

Comment: Relaxation of specific NRC authorization for incineration. A number of comments questioned the need for the existing requirement that incineration of radioactive materials requires specific prior NRC approval (except for small quantities of tritium and carbon-14, which are specifically exempted). These commenters noted that the source of the released material (from an incinerator stack or from a fume hood vent) should not be the basis of requiring specific prior NRC approval of incineration while permitting general effluent releases.

Response: Relaxation of the prior approval requirement for incineration was considered in connection with the revision of Part 20. The requirement for prior NRC approval of incineration remains in the revised Part 20 because the acceptability of incineration as a disposal option, except for exempted quantities of radioactive materials, must be determined on a site-specific basis considering (1) incinerator design to safely dispose of hazardous materials, (2) the variable nature of the material to be burned both in terms of isotopic composition and activity, and (3) because many of these incinerators can be located in urban areas, special calculational methods may be required to assess doses to people located near these facilities.

Final Rule: Disposal by incineration still requires specific approval by the Commission (or Agreement State) whether done only for wastes from the licensed facility or whether done for wastes received from other licensees. Section 20.1005 Disposal of Specific Wastes.

Comment: There should be a definition of ALARA for solid wastes. Many commenters suggested the need for ALARA or exempt quantities of radioactive material in solid wastes so that very low-level solid wastes could be disposed of without regard to their radioactivity.

Response: The Commission agrees that such levels would be useful and has developed a policy statement regarding levels of dose and risk that can be used to determine that specific practices involve radiation hazards that are Below Regulatory Concern (BRC). This policy statement was published in the Federal Register on July 3, 1990 (55 FR 27522). The BRC policy statement provides a comprehensive policy that will establish a disciplined and consistent framework for all future Commission exemption decisions. This includes potential application to rulemaking or licensing actions for disposal of slightly contaminated solid radioactive wastes. The Commission is developing a program for implementing the BRC policy separate from this Part 20 rulemaking.

Section 20.1006 Transfer for disposal and manifests.

Comment: This section should not \_\_ 'n Part 20.

Response and Final Rule: This section is in Part 20 because it relates to the radiation protection aspects of low-level waste shipments.

#### 

Final R is section has a counterpart in the present Part 20 and in the proposed rule (§ 20.1005) stating that meeting Part 20 requirements does not remove the responsibility of licensees, when disposing of licensed radioactive materials, from meeting the requirements of other applicable Federal, State, and local regulations applicable to toxic or hazardous wastes. The advisory statement in the final rule has been expanded to cover all methods of waste disposal. This section of the rule is advisory and is not intended to imply that NRC will take enforcement action for violations of other environmental protection regulations issued under statutes other than the Atomic Energy Act.

## Subpart L--Records

Standardization of Record Retention Requirements.

Final Rule: Records directly pertaining to effluents released to the general environment, waste disposal, and doses received by individuals are to be kept until the "Commission terminates each pertinent license requiring the record." Other record retention requirements in this subpart generally have been modified to be for 3 years after the record is made. This change is in conformance with the final rule published in the Federal Register of May 27, 1988 (53 FR 19240) on record retention requirements for other parts of the NRC regulations. This change provides for consistent record retention requirements throughout the NRC regulations in Title 10 of the Code of Federal Regulations.

# Section 20.1101 General Requirements.

Comment: The units used in records should be limited to those commonly in use: the rad, the rem, and the curie. Some commenters thought that the use of SI units (gray, sievert, and becquerel) should not be allowed.

Response and Final Rule: The Commission agrees that the use of "special units," the rad, the rem, and the curie, is preferable at this time. This will avoid any difficulties arising from trying to implement both a new regulation and new units. This will reduce potential problems in records and reports that could result from some licensees using the "SI units" and some using the older "special units." The final rule requires the use of the "special units" instead of the "SI units." See the discussion of this topic under § 20.4 Units. Section 20.1102 Records of Radiation Protection Programs.

Comment: Added implementation turden associated with requirements for formal radiation programs. A number of commenters thought that the requirement to have a formal ALARA program would result in substantial increased costs due to additional recordkeeping, procedural requirements, and quality assurance requirements.

Response: As discussed under § 20.101, these provisions have been modified to require ALARA as one part of a licensee's radiation protection program. The adoption of requirements for licensees to have a formal radiation protection program was not intended to cause large implementation costs. Much of the cost associated with the recordkeeping requirements in the proposed rule was a result of the ALARA documentation requirements. These recordkeeping requirements have been reduced in the final rule by deleting specific reference to documenting ALARA actions. Specific types of records will be developed by each licensee as part of its radiation protection program. Therefore, this section contains gene.al recordkeeping requirements associated with the radiation protection program.

Comment: The recordkeeping burden for small licensees requires a commitment of resources that is not commensurate with the risk. (II Section XXXVI of the proposed rule (51 FR 1121-1122), NRC specifically requested comments on the magnitude of the impact of the proposed rule on small licensees and requested suggestions on how these impacts could be reduced.) Quite a few commenters expressed their belief that the proposed rule will require more extensive monitoring and recordkeeping efforts than were required by the existing Part 20. Several commenters suggested that the NRC explore possible exemptions or exclusions for academic licensees and other users of small quantities of licensed material. Other commenters expressed the view that the protection of public health for both the worker and the general public should be the same regardless of the size or economic resources of the licensee.

Response: Because of the changes to reduce the recordke hing burden discussed in response to the preceding comment and because the basic requirement in § 20.101 calls for effort "... commensurate with the scope and extent of licer.sed activities ...," the Commission has not made further exemptions or exclusions from the recordkeeping requirements in this section for certain types of licensees.

Section 20.1104 Determination of Prior Occupational Dose.

Comment: Medical and academic licensees would have difficulty in complying with the requirement to determine prior exposures. The transitory nature of personnel in these facilities would make meeting these requirements very costly. Doses to employees are small fractions of the limits so that such costs would be difficult to justify.

Response: The requirement to determine dose received in the current year implements the annual dose limits. The requirement to attempt to obtain records of lifetime cumulative doses follows one of the provisions of the guidance to Federal agencies on occupational radiation protection. Efforts to obtain prior exposure histories are only required for workers who are required to be monitored under § 20.502. Determination of prior doses received during planned special exposures or doses in excess of the annual limits are required only for workers who will be used in planned special exposures.

Comment: The recording of "fictitious" radiation doses should be avoided. The present and proposed rules state that, when information is not available regarding the dose received for a specific period, the licensee should assume that the dose received was at the dose limit. Several commenters thought that this was inappropriate. Some commenters mentioned that this practice might be nonconservative as it would tend to overestimate the dose used in any epidemiological studies of radiation effects, thereby resulting in an underestimate of the risk associated with a unit radiation dose.

Response and Final Rule: The final rule has been modified so that it does not require any assumed dose value to be recorded in case of incomplete prior dose histories. Only the lack of data must be recorded for periods where there is no information. However, for the current year, where there are missing data, an assumption is to be made for establishing administrative controls: the portion of the dose limit remaining for the current year is reduced by 1.25 rems for each calendar quarter for which information is missing. (The values for other limits, such as the shallow dose equivalent or eye dose equivalent should be reduced by a one-quarter of their annual limit for each unreported quarter.) The licensee must note the absence of this information on the employee's record but should not enter the assumed dose value as part of the employee's permanent dose record. For example, an employee who had prior radiation working experience joins (impany X on July 1st but does not have the prior radiation records. This employee's dose should be limited to 2.5 rems (5 rems = 2(1.25) = 2.5 rems) until such time as the records are obtained.

Comment: There should be a quarterly dose limit to cover workers whose records have not been received from a former employer. A 0.5-rem dose might be appropriate for this purpose.

Response: If data were missing for all four quarters (employment commenced late in the fourth calendar quarter), then the employee could not be exposed to radiation above the level for a member of the general public. However, this limit is 0.1 rem per year not 0.5 rem.

Section 20.1105 Records of Planned Special Exposures.

See discussion under § 20.1204.

Section 10.1106 Records of Individual Monitoring Results.

Comment: NRC should not require reporting or recording of cumulative dose. A number of commenters noted that the ICkP system of dose limitation is based [as one of the principles] on controlling annual doses. Consequently, they questioned the need for recording cumulative doses.

Comment: The proposed rule does not require recording annual doses as listed in the 1987 Federal occupational guidance.

Response: "Annual dose" is specified in the guidance and is the same as the annual deep-dose equivalent for external doses. However, "annual dose" is not required to be recorded by the revised Part 20 for internal doses. This is consistent with an exception noted in fcotnote 5 to the Federal guidance (Federal Register of January 27, 1977; 52 FR 2832):

> "When these conditions on intake of radioactive materials have been satisfied [i.e., meeting the committed dose limits], it is not necessary to assess contributions from such intakes to annual doses in future years, and, as an operational procedure, such doses may be assigned to the year of intake for the purpose of assessing compliance."

Paragraph 20.1106(b) -- See discussion under § 20.1204.

Comment: The recordkeeping requirement in the proposed § 20.1106(d)(2) would require that all records begin at the beginning of a calendar year. This would create an unnecessary hardship on dosimeter processors since they could not stagger the dosimeter changeover schedules to provide a more uniform work-load distribution.

Response and Final Rule: The term "year" replaces the term "calendar year" in § 20.3 and permits the licensee to define the year to begin anytime in January. A licensee may change the starting date, provided that the change is made at the beginning of the year and provided that no day is omitted and no day is included twice in consecutive years.

Comment: The requirement in § 20.1105(e) for each licensee to keep a copy of the dosimeter processor's accreditation certificate creates an undue burden on commercial processors. Commercial dosimeter processors would have to print and distribute thousands of their certificates so that each user had a copy.

Response: The proposed rule contained a requirement for the licensee to maintain a copy of the dosimetry processing accreditation certificate issued to the processor providing dosimetry services to the licensee. This requirement, which was in the proposed dosimetry accreditation rule, was considered unnecessary and was dropped as a requirement in the final version of that rule. Consequently, it has been deleted from revised Part 20. Licensees who provide their own dosimeter processing services do have to maintain a copy of their NVLAP accreditation certificate for inspection.

Comment: The NRC should consider a "traveling dose history" that can move with the worker. This was suggested, particularly for transient workers and for workers employed concurrently by two employers. The master record will reside with the current employer and would have to be transmitted by the worker to a new employer.

Response: Because the NRC can only regulate its licensees and has no authority over individual workers, the recordkeeping and transmittal requirements for dose histories are placed on the licensee and not on the worker. The concept of a "passport" incorporating security and dosimetry data has been used successfully in Japan and elsewhere. The requirements for determination of prior exposures that are in § 20.1104 provide a similar record to a "moving history," but this would have to be updated by each new employer.

Concurrent employment with two (or more) employers requires special attention so that the combined doses from both employers would not exceed the dose limits. When two employers are aware of such concurrent employment, the simplest expedient to achieve this goal is for them to agree that the dose limit they will use for this employee in the individual programs is less than one-half of the NRC dose limits (the fraction of the dose limit allocated to each employer might also be determined on the basis of the relative amount of time worked at each location).

The problem of dual employment is more of a problem when the employee has not confided in the employer. The licensee is required to ascertain the employment and dose record for the current year for new employees (§ 20.1104). If the employee deliberately falsifies this information, the licensee would not know of concurrent employment and the licensee would not be penalized for combined doses from both employers that exceeded the dose limits. If a current employee takes on additional outside radiation work without informing the employer, the employer should not be penalized. It should be noted that, under the new reporting requirements in § 20.1206, individual dose records will be required to be submitted to the NRC for all workers for those categories of licensees formerly subject to § 20.407, including nuclear power reactor licensees.

Final Rule: Section 20.1106 has been modified in order to separate the requirement for keeping a record from the format of the record. A clarification has been added that the dose information on an embryo/fetus be kept with the mother's dose record.

Section 20.1107 Records of Dose to Individual Members of the Public.

Comment: Reporting requirements for exceeding "reference levels." The proposed rule contained requirements for reporting exposures in excess of the "reference levels" for doses to members of the general public. Many commenters thought that this was excessive because this was not an actual regulatory limit.

Response: The 100 millirems per year "reference level for doses to members of the general public in the proposed rule has been incorporated as the dose limit in the final rule for members of the general public so that the associated recording and reporting requirements now pertain to a regulatory limit.

Final Rule: Section 20.1107 has been broadened in scope from "effluents" to pertain to records of all estimates of doses received by individual members of the public. Doses to members of the public are calculated from measurements of direct radiation, and radionuclides in effluents, and the environment rather than as measurements pertaining to a a particular individual. This difference in method of dose assessment from the more direct measurements used for occupational exposure does not imply any lessening of requirements for keeping ade-quate records of effluents released to unrestricted areas.

Section 20.1108 Records of Waste Disposal.

Final Rule: Section 20.1108 is unchanged from the proposed rule.

Section 20.1109 Records of Testing Entry Control Devices for Very High Radiation Areas.

Final Rule: Section 20.1109 contains an addition to the proposed rule for keeping records of tests of entry control devices for very high radiation areas. This addition is based upon a requirement in § 20.203(c)(6) of the present Part 20. Section 20.1110 Form of Records.

Comment: NRC should allow computerized recordkeeping systems to handle records. A few licensees suggested that NRC allow "electronic" recordkeeping systems and provide guidance for their use.

Response: The Commission agrees that there is great value in the use of "electronic media." There are a growing number of licensees that are using computer information networks for retaining and transmitting radiation dose histories and other worker-related information among different facilities.

Final Rule: The final Part 20 expands the definition of "record" to include "electronic media." The use of electronic media requires authentication and the prevention of alteration or loss of the records. As with existing requirements for paper records, the electronic media must be capable of producing a legible copy of the record.

# Subpart M--Reports

Section 20.1201 Reports of Theft or Loss of Licensed Material.

Comment: The term "substantial exposure" in § 20.1201(a) should be defined. The requirement to report the loss of radiation sources capable of producing "substantial exposure" needs to be more precise.

Response: The term "substantial exposure" has been replaced by a specific designation of the activity of lost source that requires immediate reporting to the Commission. This quantity is 1,000 times the Appendix C activity levels. For sealed sources of cobalt-60, cesium-137, or iridium-192, this activity would produce a dose of around 25 rems at 1 foot over a 30-day period (25 rems is the worker dose that requires immediate Comm. Ssion notification). Although somewhat similar doses may be projected from inhalation of dispersible material, the exact exposure conditions would have to be known in order to make a valid activity-to-dose relationship.

Final Rule: The final rule now contains specific activity criteria for immediate reporting rather than the vague term, "substantial exposure."

Comment: The quantity for reporting the loss of a source is too low (too high). The reportable quantity of 10 times the Appendix C activity values appeared to some commenters to be overly restrictive; others thought that all lost or missing radiation sources should be reported.

Response: The specified 30-day reporting level is a compromise between having higher reporting levels and having a requirement that all lost or missing sources be reported. Further, the report permits review of the circumstances involved including any lack of security of materials or weakness in the licensee's control program that may be unrelated to the sources being stolen or lost, but may be pertinent in avoiding recurrent theft or loss.

Final Rule: The activity levels in Appendix C for some long-lived radionuclides have been increased from those specified in the proposed rule. This increase means that the loss of milligram quantities of natural uranium will no longer have to be reported.

Comment: A 30-day telephone report should not be required concomitant with a written report. Sections 20.1201(a)(1)(ii) and 20.1201(b) both call for a 30-day report; the first requires a telephone report and the latter section requires a written report.

Response and Final Rule: The rule has been revised to clarify that the written reports required by § 20.1201(b) are to be submitted within 30 days of the telephone notification required by § 20.1201(a), rather than both being within 30 days of learning of the theft or loss.

Comment: The rule should provide for a "grace period" before having to report a lost source to NRC. Commenters noted that, in many instances, a source "lost" in transit eventually turns up. Some specified period, such as 7 days, should be permitted before a "lost" source would have to be reported to the NRC.

Response: The rule contains two notification requirements: the one for immediate notification only pertains to those sources that exceed 1,000 times the Appendix C activity levels. The second notification requirement pertains to sources that exceed 10 times the activity levels in Appendix C and that are still missing after 30 days. This provides a grace period of 30 days for reporting the loss of most sources.

Section 20.1202 Notification of Incidents.

Comment: The requirements for immediate notification of NRC are too low. Some commenters thought that the doses associated with the requirements for immediate reporting to NRC (five times the respective annual limits) would not produce any discernible harmful effects to the individual to warrant immediate reporting.

Response: Doses of the order of 25 rems (5 times the 5-rem annual dose limit) can produce discernible biological effects in the body in the form of chromosome aberrations and changes in the white blood cell populations. Although the majority of these effects are temporary, they could be discerned. However, irrespective of the potential for discernible effects, doses at these levels represent a major breakdown in the licensee's control over the radioactive material, and the Commission believes that it is important that NRC be promptly notified so that it can take actions, if necessary, to limit further consequences.

Final Rule: The final rule retains the previous reporting requirement.

Comment: Immediate reporting should be required if there is any potential for dose reduction. The Environmental Protection Agency (EPA) suggested that incidents always be reported if there is the potential for significantly reducing public doses through protective actions. It is believed by the EPA that this would occur at doses significantly less than those of the proposed reporting criteria.

Response: The incident reporting levels and response times have been selected to limit attention to the more potentially serious events without the entire NRC emergency response network being activated unduly for events involving only small quantities of radioactive materials. For most cases, it is expected that the licensee would have initiated any necessary remedial measures.

Comment: Immediate and 24-hour notification requirements should be suspended in the case of a declared emergency at a nuclear power plant. Commenters felt that any emergency at a nuclear power plant will involve onsite NRC staff and that stopping emergency activities to make the Part 20 incident reports could be a burden on the licensee.

Response and Final Rule: These reports are particularly easy to make for nuclear power reactors (the reactor operator merely has to pick up the dedicated NRC telephone line to get the NRC Operations Center). There are certain functions of the NRC (such as activating the NRC Incident Response Plan) that require that NRC be notified; therefore, this notification requirement has been retained.

Section 20.1203 Reports of Exposures, Radiation Levels, and Concentrations.

Comment: There is no requirement for reporting doses that exceed the limit for protection of the embryo/fetus in § 20.208.

Response and Final Rule: A requirement has been added to the final rule in § 20.1203(a)(2)(iii).

Comment: The identifiers required in § 20.1203(b)(2) for the embryo/fetus should be those of the mother. As the fetus has no date of birth and no Social Security account number, those of the mother should be used.

Response and Final Rule: A footnote to this effect has been added to § 20.1203.

Comment: Reports of exceeding the 0.1-rem reference level should not be required. A number of commenters noted that the 0.1-rem reference level was not a limit and, therefore, exceeding it should not necessitate a report to the NRC.

Response: As a result of changes in the ICRP interpretation of the 0.1-rem level and the former 0.5-rem dose limit, the 0.1-rem level is now the recommended limit. Consequently, 0.1 rem is the primary limit applicable to members of the general public and reports arr justified when it is exceeded.

Comment: Smaller licensees, such as nuclear medicine facilities, should be exempted from the reporting requirements of § 20.1203. Licensees are required to report concentrations in unrestricted areas that exceed 10 times any applicable limit in the license. Because some nuclear medicine units use the room air volume for dilution, calculated concentrations exceeding 10 times the Appendix B limits might frequently occur. This would require either more frequent reporting to NRC or use of more sophisticated atmospheric dispersion models.

Response: The reporting requirements are very similar to those in the previous Part 20. Part 35 of the Commission's regulations, which deals with medical applications, covers the medical use of noble gases and in § 35.205(a) limits airborne concentrations to the 10 CFR Part 20 Appendix B concentrations. Experience has not indicated large numbers of reports of such limits being exceeded.

## § 20.1204 Reports of Planned Special Exposures

Comment: The licensee should not have to file a separate report to NRC for Planned Special Exposures. Several commenters objected to having to file these separate reports each time a Planned Special Exposure is carried out. This was viewed as representing a reporting requirement for operating within the NRC regulations. It was suggested that this information be included in the employee's records without reporting to NRC.

Response: Because of the newness of the concept, the NRC wishes to monitor carefully the use of the Planned Special Exposures. Further, while the Planned Special Exposures are provided in the final rule, its use does represent a situation in which the licensee is operating outside of the normal dose limits, and of which the Commission should be aware.

Comment: Period for reporting planned special exposures. Several commenters noted that the 15-day period for reporting planned special exposures is shorter than the 30-day period usually allowed for similar reports.

Response: The reporting period of a planned special exposure has been increased from 15 days to 30 days to be more consistent with other reporting requirements.

Section 20.1206 Reports of Individual Monitoring.

Comment: Could the requirement for the reporting of individual exposures be construed as an invasion of privacy? Some commenters believed that requiring the reporting of individual doses rather than a statistical summary might constitute an invasion of personal privacy.

Response: The Commission does not believe that submission of individual dose data constitutes an invasion of privacy. Such data have been reported to the NRC routinely in the termination reports for some time. Such information will be protected in accordance with the Privacy Act and will be restricted, as it has been in the past, to use by NRC officials, NRC contractors, or qualified scientific investigat.rs. Instructions on protecting this information appear in § 20.1106(d).

Comment: If the radiation exposure data are collected into a central repository, would the NRC by the proper place for it? One commenter felt that the radiation exposure data might be better maintained by an agency whose charter encompasses the analysis of the data for estimates of risk.

Response: Arguments might be made for other agencies having the lead role in the storage and analysis of those data; however, it is the NRC that has the statutory authority to require that these data be collected. Although the Parc 20 recordkeeping requirements are intended primarily to fulfill NRC's information needs for regulation, the NRC has continuing contacts with agencies that have expertise in conducting epidemiological studies (such as the National Cancer Institute of the National Institutes of Health and the Office of Health and Safety of the Department of Energy) to ensure that the Part 20 reporting and recordkeeping requirements do not lose information that would be vital to carrying out studies of this type.

Comment: The total collective (person-rem) dose should be reported. It was felt by one commenter that NRC should require the total collective dose to be reported so that the numbers used in NUREG-0473 (NRC's annual summary of occupational radiation doses) will be the same as those calculated by the licensee.

Response: The reason for a possible discrepancy between a licensee's estimate of the collective dose to workers and the estimate published by the NRC has been that the licensee may sum the actual individual doses and the NRC estimate is based upon the statistical summary rather than the actual individual dose reports. Such differences should be reduced in the future because NRC will also be using dose information for individuals. The final rule requires licensees who previously submitted the dose summaries to report the individual dose data to NRC. Both collective dose calculations should then be using the same data base.

Comment: The termination report required in § 20.1207 should (or should not) be replaced with an annual report for all personnel monitored. Some commenters felt that an annual report just to the NRC should replace the present requirement for a termination report. Other commenters felt that annual reports to the NRC of doses to individuals constituted a considerably larger burden than did a statistical summary. Some commenters, who disagree with filing an annual report to the NRC, were in favor of giving such an annual dose summary to the worker. Other commenters suggested that all licensees be required to submit an annual report to NRC on each monitored individual.

Response: The reporting of individual monitoring data will help track doses to individuals who are exposed at several facilities during any given year and whose total dose would be underreported by statistical reports prepared at each work site. Such information is shown at the present time only by analysis of the termination reports.

Licensees who were previously required to file both annual statistical summaries and termination reports with the NRC will, instead, submit annual dose reports to NRC for all workers for whom monitoring was required under § 20.502. A copy of the annual report to NRC could also be given to the individual worker in order to satisfy the revised reporting requirement in § 19.13 of 10 CFR Part 19. Although this may entail some additional burden to licensees, the use of "electronic media" for recordkeeping might in fact reduce overall costs. It is intended that large employers (such as nuclear power reactor licensees) would submit an electronic copy of their dose reports in a prescribed format to the NRC in lieu of paper copies of individual records.

#### Subpart N--Exemptions and Additional Requirements

Section 20.1301 Applications for Exemptions.

Comment: NRC should make the issuance of exemptions a matter of public record. Several commenters felt that the issuance of any exemptions under this section should require public notice and comment. The EPA stated that exemptions could adversely affect its ability to control radionuclides under the Safe Drinking Water Act. Response: The NRC has issued few exemptions under this longstanding provision and has not exempted anyone from the dose limits for a worker or for a member of the public. The Safe Drinning Water Act was not intended to control effluents and, although radionuclide concentrations at downstream water supplies are routinely calculated as part of litensing evaluations, the licensee must meet the Part 20 concentration limits at the effluent release point, not at the drinking water intake after dilution occurs.

#### Appendix A

Comment: The protection factor for air-purifying respirators with particulate elements is too low. The listed protection factor for air-purifying respirators with particulate filters is 50, whereas both ANSI Z88.2 and the OSHA regulations in 29 CFR Part 134 use 100.

Response: The NRC never endorsed ANSI Z88.2-1980, whereas the OSHA regulations generally follow ANSI standards. The current NRC-allowed protection factors (PFs) are based upon research conducted by the Los Alamos National Laboratory (LANL). These recommendations included a PF of 50 for full face respirators, based on experimental data on actual testing of personnel using respirators under carefully controlled conditions. In actual use, there is essentially no difference between a PF of 50 versus a PF of 100, so that there should be little or no real impact on field use of respirators or on operations at nuclear facilities that would result from using the higher protect on factor.

Comment: Several respiratory equipment specifications in Appendix A should be applicable only for areas that are "immediately dangerous to life and health." Footnotes "h" and "i" contain specifications for air flow rates and flow calibration and a requirement for standby rescuers to be available when using supplied-air suits. These were felt to be unneeded considering that, if the air flow failed, the person could withstand a small exposure to the airborne radionuclides while exiting the area after removing the protective hood. Response: The supposition that conditions "immediately dangerous to life and health" do not exist is not always correct. Failure of an airline in supplied-air suits may be considered as "immediately dangerous to life and health" because there is an acute danger of suffocation if the air supply is interrupted and the hood cannot be removed by the wearer. Rapid recovery of and assistance to the individual in the supplied-air suit necessitates the presence of a preequipped rescuer.

#### Appendix B

General comments: Most of the comments from radiation protection professionals favored the adoption of the ICRP-26/ICRP-30 annual limits on intake and the derived air concentrations. Comments from private citizens were against adoption of the ICRP values because the majority of the values would increase (as stated in Section XXIX of the proposed rule, 51 FR 1120).

Response: From an occupational protection standpoint, the changes that result from adoption of the ICRP risk-based approach lead to higher limiting intake values than in the previous Part 20. These increases result from the increase in the allowable ceiling for organ doses. The values that served as the basis for calculating the concentration limits used in the former Part 20 were organ dose limits of 5, 15, and 30 rems. The new concentration limits are based upon the effective (weighted) organ dose or upon the nonstochast's limit that forms an organ dose ceiling when the stochastic risk is not limiting. These changes increase the limiting annual organ doses (when only one organ is irradiated) for those doses that are limited by the stochastic (effective dose) limit from 5 rems to 20 rems for the gonads, from 15 rems to 32 rems per year for the breast, and from 15 rems to 42 rems for the lung. Limiting doses to other organs increase from the former 15- and 30-rem values to the 50-rem nonstochastic limit.

The former ICRP-2 "critical organ" concept based the limiting intake upon controlling the dose rate to the organ receiving the highest dose rate (the "critical organ") The doses to organs other than the critical organ did not

have to be avaluated, even if these doses were close to the estimated dose to the critical organ. The new ICRP-26/30 system evaluates the doses to the major organs and the six remaining organs that receive the next highest doses. These doses are then multiplied by the appropriate weighting factors ( $w_T$ ) and are summed to give a risk-weighted "effective dose." The roncentration limits that are based upon this newer ICRP approach reflect the doses to all principal organs that are irradiated, not just the one organ that receives the highest dose as was done in the former Part 20.

Many of the comments from private citizens do not appear to reflect the proposed rule because many of the comments objected to raising the limits for radionuclide concentrations applicable to the general public. As noted in the discussion of Appendix B in the notice of proposed rulemaking (Section XXIX, 51 FR 1119-1120), the concentration limits for members of the public were based upon a "reference level" dose (now the dose limit for members of the general public) of 0.1 rem per year and incorporated an additional factor of 2 reduction (Proposed Appendix B; 51 FR 1145) for age-dependency and combined air and water intakes. Thus, the concentration limits for the public reflect a reduction in their basis from a whole-body annual dose o. w.5 rem in the former Part 20 to 0.05 rem in the proposed and final rules.

The concentration limits for individual radionuclides may be higher or lower for members of the general public in unrestricted areas in the final Appendix B than in the former tables because of changes that occurred in the intervening 25 years in the metabolic and other parameters used to calculate internal doses. These changes are reflected in ICRP Publication 30 and its supplements and amendments. However, these changes are a result of changes in the scientific techniques and parameters used in calculating doses and do not reflect an increase in the allowable dose limits, which, in reality, have been decreased in the revised Part 20.

Comment: NRC should consider deleting Table 2 from Appendix B. The concentration limits in Appendix B do not provide adequate protection of children and infants because they do not take into account a pend icy in a proper manner. Compliance with the dose limits, rather with chese concentration limits, should be required.

Response: The use of the effective dose equivalent concept reduces the importance of age-dependent intake-to-dose factors. Age dependency is of primary importance in calculating organ doses. Those organs for which age dependency is important, such as the thyroid gland, are of lesser importance because of lower  $w_T$  values (for the thyroid, for example,  $w_T = 0.03$ ) used to calculate the effective dose. A factor of 2 is included in the calculation of concentration limits for release to air and water, which, in part, accounts for age dependency. In addition, the Commission believes that there is a lack of detailed age-dependent metabolic data for all but the most common radionuclides that will inhibit such attempts to increase the precision of the dose estimates.

Many smaller licensees routinely use concentrations and the Appendix B tables in order to demonstrate compliance. The use of concentration limits for determining compliance is a well-established practice that is economical for many of the smaller licensees. Despite the growing availability of simplified dose assessment models, the Commission is continuing to accept the use of concentrations to demonstrate compliance with the dose limits.

Comment: The Appendix B tables fail to account for the chemical toxicity of natural and low-enriched uranium. This fails to take into account the possible kidney (renal) damage associated with the chemical toxicity.

Response: There is a separate limit for uranium intake that is based upon the chemical toxicity. This limit was expressed as footnote 3 to Appendix B, page 1199 of the January 9, 1986 notice of proposed rulemaking and also as § 20.204(i) on page 1131. In the revised rule, it still appears as footnote 3 in Appendix B, but the limit also has been moved up in the text to the section on dose limits and now appears as § 20.201(e).

Comment: The limits for occupational and nonoccupational exposure to radon-222 and its particulate daughters do not appear to be consistent with the airborne concentration limits for other radionuclides in terms of risk.

Response: The occupational concentration limits for radon-222 are based on the existing Federal guidance, which is 4 WLM (4 Working-Level Months) per year. The annual limit on intake (ALI) is stated as 100  $\mu$ Ci or 4 working-level months. The derived air concentration (DAC) in Part 20 for occupational exposure to radon-222 of 3 x 10<sup>-8</sup> is equivalent to 0.33 working levels (this equivalence is also given in the Appendix B table). The concentration limit for members of the general public is a factor of 300 lower and, like the other airborne concentration limits, represents an effective dose of 0.05 rem per year.

Comment: Concentration limits for tritium omit chemical forms other than for tritiated water vapor.

Response: As there is expected to be no occupational intake via oral ingestion, and most of the commonly used organic tritiated compounds are not volatile, inhalation and transpiration through the skin are the principal pathways of exposure. Different intake limits would apply to hydrogen gas (HT or  $T_2$ ) and tritiated water vapor(HTO). The HT or  $T_2$  gas is rapidly converted to HTO by isotopic exchange and oxidation (both in air and in the body) so that specifying a submersion dose limit for HT would understate the actual radio-logical impact. Comparison with other derived limits for other chemical forms shows that the use of the concentration limits for HTO provides an adequate level of protection for most of the other chemical forms.

Comment: No concentration limits are listed for natural thorium. There are limits for natural uranium, but corresponding concentration limits for natural thorium are not given. The isotopic composition of thorium can vary somewhat with different ores and with different times after chemical separation.

Response: A licensee should use the thorium-232 value or, if a more precise value is desired, use the procedure for mixtures in Appendix B applied to the actual isotopic concentrations present.

Comment: The derived air concentrations for the general public are not always 0.1 times the occupational values.

Response: The limits for the general public are calculated solely from the stochastic risks. This differs from ICRP, which would use a "capping" organ dose limit of 5 rems (0.1 x the nonstochastic limit of 50 rems) in deriving the organ dose limit for organs that are limited by the nonstochastic risk. If there is a threshold for nonstochastic effects for the worker at 50 rems, it would also apply to a member of the public. Rather than applying a factor of 10 reduction to a nonstochastic value, the limiting stochastic (effective) dose was used to calculate the concentration limits for the general public. Values are not based on the nonstochastic risk for members of the public, even if they were the basis for the calculation of the DACs and ALIs for the worker. This difference in method of calculation accounts for the lack of a consistent ratio between worker DACs and effluent limits for the public.

### Appendix C

Comment: The reduction from 100  $\mu$ Ci to 0.001  $\mu$ Ci for thorium values will require posting of areas where thoriated-nickel machine parts are used.

Response: On the basis of specific activity considerations, the existing 100  $\mu$ Ci limit has been retained for long-lived radionuclides (half-lives longer than 10<sup>8</sup> years) such as thorium-232, which would require several grams of material to produce the stated activity level. Because this is based on half-life, two isotopes may be treated differently, e.g., uranium-235 which does not meet the half-life criterion has an Appendix C value of 0.001  $\mu$ Ci, and uranium -238 which does meet the criterion has a value of 100  $\mu$ Ci.

Appendix D contains the NRC Regional Office addresses and telephone numbers.

### Appendix E [Reserved]

Final Rule: The calculational guidelines and equations that appeared in Appendix E are being incorporated into a regulatory guide on summation of internal and external doses. This will make it easier to revise and clarify the calculational methods without having to resort to formal rulemaking. (Note: NRC routinely issues regulatory guides for public comment before making them final.)

#### Appendix F

[Note: Appendix F is derived directly from requirements inserted by the Part 61 rulemaking proceeding on low-level radioactive waste disposa! sites. These requirements were in § 20.311 of the existing 10 CFR Part 20. Because these requirements are relatively recent, they were not modified in the Part 20 revision. The Commission is considering revisions to the manifest requirements in a rulemaking separate from the Part 20 rulemaking.]

Appendix G

No comments on Appendix G were received.

## VII. Conforming Amendments

Accompanying the revised rule are amendments to other parts of Chapter I that update citations to 10 CFR Part 20 that are found in these other regulations. Two amendments are particularly important as they go beyond updating cross-reference citations. One amendment to Appendix C to 10 CFR Part 2 updates and modifies the examples of the severity levels associated with violations of 10 CFR Part 20. Because Appendix C relates to the administrative policy of the Commission and because the listed violations are used as examples of different severity levels and are not all inclusive, the Commission does not believe that solicitation of public comment is required before these are issued in final form.

The second major change to other parts is the requirement to provide all workers with information on their radiation doses. This modification was made to conform to the 1987 Federal guidance on occupational radiation exposure. Formerly, Part 19 required licensees to furnish such a report at least annually upon the request of the worker. The change deletes the words "upon request." Public comment is not being solicited on this change as the comments were requested in the proposed rule (Section XXVII, 51 FR 1118) on the option of requiring reports to individual workers. (These comments are discussed with regard to § 20.1106.) Part 19 has been revised to require licensees to advise each worker at least annually of the worker's dose recorded pursuant to § 20.1106.

## VIII. Finding of No Significant Environmental Impact: Availability

The Commission has determined under the National Environmental Policy Act of 1969, as amended and the Commission's regulations in Subpart A of 10 CFR Part 51 that this rule is not a major Federal action significantly affecting the quality of the human environment and therefore an environmental impact statement is not required. The revised 10 CFR Part 20 changes the level for protection of the general public from an implicit limit of 0.5 rem per year to an explicit limit of 0.1 rem per year. There are also numerous changes in airborne and water radionuclide concentration limits. These changes result from changes in the models and parameters used to estimate the radiation dose associated with intake of a radionuclide. Some of the concentration limits for the general public in this revision are higher or lower than present concentration limits; and some are similar to the present limits.

Despite the changes in the dose and concentration limits, the Commission believes that issuance of the final Part 20 rule will not have a major impact on the environment. The primary basis for this conclusion is that, in addition to 10 CFR Part 20, there are other regulations that govern allowable doses to members of the public and that remain unchanged by the changes to Part 20. These other regulations include Appendix I to 10 CFR Part 50, 10 CFR Part 60, and 10 CFR Part 61, the EPA's generally applicable environmental standards in 40 CFR Part 190 and the National Emission Standards for Hazardous Air Pollutants (NSSHAP) in 40 CFR Part 61 Subpart I. These standards set limits or design objectives (Appendix I) for releases of radioactive material to the general environment that are generally more restrictive than the dose limits in Part 20. Consequently, since these more restrictive standards remained essentially

unchanged by the Part 20 revision, the level of public protection and the associated environmental impact are not changed appreciably from those associated with the current rule and the aforenamed regulations.

The environmental assessment and finding of no significant impact on which this determination is based are available for inspection at the NRC Public Document Room, 2120 L Street NW (Lower-Level), Washington, DC 20555. Single copies of the environmental assessment and finding of no significant impact are available from Harold T. Peterson, Jr., Nuclear Regulatory Commission, NL/S-139, Washington, DC 20555, Telephone: (301)492-3640.

#### IX. Paperwork Reduction Act Statement

This final rule amends information collection requirements that are subject to the Paperwork Reduction Act of 1980 (44 U.S.C. 3501, et seq.). These information collection requirements in this final rule have not been reviewed by the Office of Management and Budget (OMB), but will be submitted by NRC for approval by OMB. These information collection requirements will not become effective until approved by OMB. The OMB approval will be published in the Federal Register.

Public reporting burden for this collection of information is estimated to average 33 hours per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Information and Records Management Branch (MNBB 7714), U.S. Nuclear Regulatory Commission, Washington, DC 20555, and to the Desk Officer, Office of Information and Regulatory Affairs, NEOB-3019 (3150-0014, 3150-0044, 3150-0005, and 3150-0006), Office of Management and Budget, Washington, DC 20503.

### X. Revised Regulatory Analysis

The Commission has issued a final regulatory analysis for this regulation. This revised analysis was based on the draft regulatory analysis as modified to account for the changes from the proposed rule resulting from public comments on both the proposed rule and the staff's revised rule in SECY-88-315 and supplemental papers. Copies of both the draft and final regulatory analysis are available for inspection and copying for a fee in the NRC Public Document Room. (See Address.)

### XI. Final Regulatory Flexibility Analysis

In accordance with the Regulatory Flexibility Act of 1980, 5 U.S.C. 605(b), the Commission has prepared a regulatory flexibility analysis that indicated the revised rule will apply to all NRC licensees. The NRC has approximately 7,500 licensees, approximately one-quarter of which are classified as small entities. (Note: Agreement States, which implement comparable regulations under Section 274 of the Atomic Energy Act of 1954, as amended, have about 16,000 licensees of which a comparable number are assumed to be small entities.) The types of small entities that would be affected by this rule include physicians, small hospitals, small laboratories, industrial applications in small industries, radiographers, and well loggers.

Copies of the draft and final regulatory analysis are available for inspection and copying, for a fee, in the NRC Public Document Room. (See Address.)

### XII. Backfit Analysis

A final backfit analysis has been prepared for this rule and may be examined and copied for a fee in the Commission's Public Document Room (see Address). For the reasons stated in this backfit analysis, the Commission believes that the reductions in allowable dose limits that are embodied in the revised Part 20 constitute substantial increases in the protection of public health and safety. Although current practice, including the philosophy of keeping radiation exposures as low as is reasonably achievable (ALARA), generally has kept radiation exposures well below the existing limits, the reductions in the allowable dose limits ensure that such doses will also remain low in the future.

In addition to the quantifiable safety benefits accruing from dose reductions and other improvements in the revised Part 20, there are several qualitative factors that support issuing the Part 20 revision. One of the main qualitative factors is that it is necessary to revise the 30-year-old existing Part 20 to ensure that the NRC regulations reflect the current state of radiation protection science. Any future revisions in dose limits recommended by ICRP or NCRP would undoubtedly be based upon the 1977 ICRP and 1987 NCRP recommendations and, therefore, would be more easily incorporated into the framework of the revised Part 20 than in the framework of the current Part 20. Other qualitative factors include: maintaining consistency with international radiation protection factors, keeping the radiation protection requirements consistent with current risk assessment methodologies, and having the NRC's standards conform to Federal radiation protection guidance.

Based upon the conclusions in the final backfit analysis, the revised Part 20 provides a substantial increase in public health and safety compared to current standards, including a determination that, when the quantitative and qualitative safety benefits of the revision are considered, the costs of implementing the revised Part 20 are justified, the Commission finds that the requirements of the "Backfit Rule" (§ 50.109) are satisfied and that the Part 20 revision should be issued as final rule.

The Commission is adopting the final rule based on the conclusions of this analysis that the rule provides for a substantial increase in the overall protection of the public health and safety and that the direct and indirect costs of its implementation are justified in terms of the quantitative and qualitative benefits associated with the rule. The Commission notes, however, that, even had the analysis not concluded that the revised Part 20 provides a substantial increase in the overall public health and safety, it could have gone forward
with the rule because the changes made to Part 20 also amount to a redefinition of the level of adequate protection and the backfit rule's substantial increase in protection and cost justification standards do not apply to a redefinition of adequate protection.

# XIII. Additional Views of Commissioner Curtiss

I approve the revisions to 10 CFR Part 20 and related changes to other regulations as outlined in SECY-88-315 and SECY-89-267, subject to the modifications discussed below.

<u>Backfit</u>: I have examined the proposed Part 20 amendments from the standpoint of whether and, if so, how the backfit rule should apply to this particular rulemaking. The nature and effects of the proposed changes to Part 20 lead me to the conclusion that the proposed amendments, in essence, would redefine what is necessary for adequate protection of the public health and safety in the radiation protection area. Thus, while I believe that we should apply the backfit rule to this Part 20 rulemaking effort, I also believe that this rulemaking constitutes a redefinition of adequate protection as described in 10 CFR § 50.109(a)(4)(iii) and that the usual backfit analysis and cost-benefit balancing are therefore not required in this instance.

On the question of whether such an approach would require this rule to be renoticed for further public comment, I have concluded that there was ample indication in the notice of proposed rulemaking that the Commission is rethinking its radiation protection standards across-the-board in this Part 20 rulemaking. Moreover, this initiative was explained in a manner that could logically be construed to encompass the approach to backfitting described above. Of particular importance, the notice of proposed rulemaking itself seems to indicate that the Commission is contemplating an action that would redefine what is necessary for adequate protection in the radiation protection area. For example, the notice states that:

105

[T]he Nuclear Regulatory Commission (NRC) is proposing a major revision of its regulations in 10 CFR Part 20 which provide the requirements for the protection of individuals who are exposed . . . to ionizing radiation from routine activities . . . which are licensed by the NRC. . . . The intent of the revision is to improve NRC radiation protection standards by reflecting developments in the principles that underlie radiation protection and advances in related sciences that have occurred since the promulgation of 10 CFR Part 20 nearly thirty years ago. . . . The expected result of promulgating and implementing the proposed revised rule is an improved rule that provides better assurance of protection; establishes a clear health protection basis for limits and other regulatory actions taken to protect public health; applies to all licensees in a consistent manner; and reflects current information on health risk, dosimetry, and radiation protection practices and experiences.

51 Fed. Reg. 1092 (January 9, 1986).

with regard to existing Part 20 standards, the Commission noted that:

[i]n promulgating these standards, the AEC emphasized "that the standards are subject to change with the development of new knowledge, with significant increase in the average exposure of the whole population to radiation and with further experience in the administration of the Commission's regulatory program." Consistent with this emphasis. the proposed revision reflects new knowledge, increased uses of radiation and generation of radiation sources. and experience gained during the past twenty years. . . [Earlier] revisions [to the existing Part 20] have not kept the regulations in accord with more recent recommendations of scientific organizations . . . to improve overall protection and establish a clear health risk rationale. . . . [T]he central thrust of the revision [is] to ensure that radiation protection is adequate and defendable when judged by good protection practices and contemporary standards.

51 Fed. Reg. 1093, 1094 (citations omitted).

In discussing the benefits of the proposed rulemaking, the Commission indicated that:

[t]he proposed revision to Part 20 includes numerous changes required to bring the radiation protections standards into accord with current defendable [sic] scientific knowledge, and to reflect contemporary scientific and philosophical approaches to protection against radiation. . . The Commission anticipates that promulgating and implementing the proposed rule will result in a regulation that provides better assurance of protection, establishes a clear health protection basis for limits, applies to all licensees, including small entities, in a consistent manner, and reflects current information on health risk, dosimetry, and radiation protection practices and experiences.

### 51 Fed. Reg. 1120, 1122.

Consistent with all of these statements on the nature of the proposed changes to Part 20, a supplemental notice of proposed rulemaking that requested comments on a proposed backfit analysis indicated that:

> [T]his is the first complete revision of these regulations in over 25 years. This revision will bring the Commission's radiation protection standards into accord with current recommendations of the International Commission on Radiological Protection (ICRP).

The proposed revision to 10 CFR Part 20 [is] intended to:

- Update the quarter-century-old 10 CFR Part 20 to incorporate advances in science and new concepts of radiation protection methodology and philosophy;
- Implement pending Federal radiation guidance on occupational radiation protection;
- Implement the principal current dose-limiting recommendations of the ICRP;
- Incorporate the ICRP "effective dose equivalent" concept;
- e. Update the limits on airborne radionuclide intakes, effluent releases and doses from inhaled or ingested radionuclides using up-to-date metabolic models and dose factors; and

f. Require that licensees have programs for keeping radiation exposures "as low as is reasonably achievable" (ALARA).

51 Fed. Reg. 30870, 30871 (August 29, 1986).

Overall, these various characteristics of the purpose, intent, and nature of the proposed changes to Part 20 lead to the conclusion that the Commission is, in fact, rethinking its radiation protection standards. For these reasons, I believe that the notice adequately describes the nature and substance of the proposed rule changes and that renoticing to further reflect a Commission judgment that the proposed changes constitute a redefinition of adequate protection is not necessary.

Implementation date: I would have preferred a common implementation date of January 1, 1994 for both NRC and Agreement State licensees to allow adequate time for all licensees to implement the revised Part 20 on the same schedule.

1

.

# XIV. List of Subjects

Part 20 - Byproduct material, licensed material, nuclear materials, nuclear power plants and reactors, occupational safety and health, packaging and containers, penalty, radiation protection, reporting and recordkeeping requirements, special nuclear material, source material, waste treatment and disposal.

Parts 2, 19, 20, 31, 32, 34, 35, 39, 40, 50, and 61 - Radiation protection.

Under the authority of the Atomic Energy Act of 1954, as amended, the Energy Reorganization Act of 1974, as amended, and 5 U.S.C. 552 and 553, the following amendments to 10 CFR Parts 2, 19, 20, 31, 32, 34, 35, 39, 40, 50, and 61 are published as a document subject to codification. 1. 10 CFR Part 20 is revised to read as follows:

PART 20 - STANDARDS FOR PROTECTION AGAINST RADIATION

SUBPART A--GENERAL PROVISIONS

# Section

- 20.1 Purpose.
- 20.2 Scope.
- 20.3 Definitions.
- 20.4 Units of radiation dose.
- 20.5 Units of radioactivity.
- 20.6 Interpretations.
- 20.7 Communications.
- 20.8 Implementation.
- 20.9 Reporting, recording, and application requirements: OMB approval.

SUBPART B--RADIATION PROTECTION PROGRAMS

20.101 Radiation protection programs.

SUBPART C--OCCUPATIONAL DOSE LIMITS

- 20.201 Occupational dose limits for adults.
- 20.202 Compliance with requirements for summation of external and internal doses.
- 20.203 Determination of external dose from airborne radioactive material.
- 20.204 Determination of internal exposure.

20.205 [Reserved]

20 206 Planned special exposures.

20.207 Occupational dose limits for minors.

20.208 Dose to an embryo/fetus.

# SUBPART D--RADIATION DOSE LIMITS FOR INDIVIDUAL MEMBERS OF THE PUBLIC

- 20.301 Dose limits for individual members of the public.
- 20.302 Compliance with dose limits for individual members of the public.

# SUBPART E--[RESERVED]

# SUBPART F -- SURVEYS AND MONITORING

- 20.501 General.
- 20.502 Conditions requiring individual monitoring of external and internal occupational dose.

# SUBPART G--CONTROL OF EXPOSURE FROM EXTERNAL SOURCES IN RESTRICTED AREAS

- 20.602 Control of access to very high radiation areas.
- 20.603 Control of access to very high radiation areas irradiators.

# SUBPART H--RESPIRATORY PROTECTION AND CONTROLS TO RESTRICT INTERNAL EXPOSURE IN RESTRICTED AREAS

- 20.701 Use of process or other engineering controls.
- 20.702 Use of other controls.
- 20.703 Use of individual respiratory protection equipment.
- 20.704 Further restrictions on the use of respiratory protection equipment.

# SUBPART I--STORAGE AND CONTROL OF LICENSED MATERIAL

- 20.801 Security of stored material.
- 20.802 Control of material not in storage.

# SUBPART J--PRECAUTIL ARY PROCEDURES

- 20.901 Caution signs.
- 20.902 Posting requirements.
- 20.903 Exceptions to posting requirements.
- 20.904 Labeling containers.
- 20.905 Exemptions to labeling requirements.
- 20.906 Procedures for receiving and opening packages.

# SUBPART K--WASTE DISPOSAL

20.1001	General	requi	rements.
---------	---------	-------	----------

- 20.1002 Method for obtaining approval of proposed disposal procedures.
- 20.1003 Disposal by release into sanitary sewerage.
- 20.1004 Treatment or disposal by incineration.
- 20.1005 Disposal of specific wastes.
- 20.1006 Transfer for disposal and manifests.
- 20.1007 Compliance with environmental and health protection regulations.

# SUBPART L -- RECORDS

- 20.1101 General provisions.
- 20.1102 Records of radiation protection programs.
- 20.1103 Records of surveys.
- 20.1104 Determination of prior occupational dose.
- 20.1105 Records of planned special exposures.
- 20.1106 Records of individual monitoring results.

- 20.1107 Records of dose to individual members of the public.
- 20.1108 Records of waste disposal.
- 20.1109 Records of testing entry control devices for very high radiation areas.
- 20.1110 Form of records.

# SUBPART M--REPORTS

- 20.1201 Reports of theft or loss of licensed material.
- 20.1202 Notification of incidents.
- 20.1203 Reports of exposures, radiation levels, and concentrations of radioactive material exceeding the limits.
- 20.1204 Reports of planned special exposures.
- 20.1205 [Reserved]
- 20.1206 Reports of individual monitoring.

# SUBPART N--EXEMPTIONS AND ADDITIONAL REQUIREMENTS

- 20.1301 Applications for exemptions.
- 20.1302 Additional requirements.

#### SUBPART O--ENFORCEMENT

20.1401 Violations.

#### APPENDICES

Appendix A Protection factors for respirators.

Appendix B Annual limits on intake (ALIs) and derived air concentrations (DACs) of radionuclides for occupational exposure; effluent concentrations; concentrations for release to sewerage. Appendix C Quantities of licensed material requiring labeling.

Appendix D United States Nuclear Regulatory Commission Regional Offices.

Appendix E [Reserved]

Appendix F Requirements for low-level waste transfer for disposal at land disposal facilities and manifests.

AUTHORITY: Secs. 53, 63, 65, 81, 103, 104, 161, 182, 186, 68 Stat. 930, 933, 935, 936, 937, 948, 953, 955, as amended (42 U.S.C. 2073, 2093, 2095, 2111, 2133, 2134, 2201, 2232, 2236): secs. 201, as amended, 202, 206, 88 tat. 1242, as amended, 1244, 1246 (42 U.S.C. 5841, 5842, 5546). For the purposes of sec. 223, 68 Stat. 958, as amended (42 U.S.C. 2273), §§ 20.102, 20.201 - 20.204, 20.206, 20.207, 20.208, 20.301, 20.302, 20.501, 20.502, 20.601(a) and (d), 20.602, 20.603, 20.701, 20.704, 20.801, 20.802, 20.901(a), 20.902, 20.904, 20.906, 20.1001, 20.1002, 20.1003, 20.1004, 20.1005(b) - (d), 20.1006, 20.1101 - 20.1110, 20.1201 - 20.1206, and 20.1301 are issued under sec. 161b., 68 Stat. 948 (42 U.S.C. 2201(b)) and § 20.1106(d) is issued under the Privacy Act of 1974, Pub.L. 93-579, 5 U.S.C. 552a: and §§ 20.102(a)(2) and (4), 20.204(c), 20.206(g) and (h), 20.904(c)(4), 20.905(c) and (d), 20.1005(c), 20.1006(b) - (d), 20.1101 - 20.1103, 20.1104(b) -(d), 20.1105 - 20.1108, and 20.1201 - 20.1207 are issued under sec. 1610, 68 Stat. 950, as amended (42 U.S.C. 2201(o)).

#### SUBPART A--GENERAL PROVISIONS

#### § 20.1 Purpose.

(a) The regulations in this part establish standards for protection against ionizing radiation resulting from activities conducted under licenses issued by the Nuclear Regulatory Commission. These regulations are issued under the Atomic Energy Act of 1954, as amended, and the Energy Reorganization Act of 1974, as amended. (b) It is the purpose of the regulations in this part to control the receipt, possession, use, transfer, and disposal of licensed material by any licensee in such a manner that the total dose to an individual (including doses resulting from licensed and unlicensed radioactive material and from radiation sources other than background radiation) does not exceed the standards for protection against radiation prescribed in the regulations in this part. However, nothing in this part shall be construed as limiting actions that may be necessary to protect health and safety.

# § 20.2 Scope.

The regulations in this part apply to persons licensed by the Commission to receive, possess, use, transfer, or dispose of byproduct, source, or special nuclear material or to operate a production or utilization facility under Parts 30 through 35, 39, 40, 50, 60, 61, 70, or 72 of this chapter. The limits in this part do not apply to doses due to background radiation, to exposure of patients to radiation for the purpose of medical diagnosis or therapy, or to voluntary participation in medical research programs.

# § 20.3 Definitions.

As used in this part:

"Absorbed dose" means the energy imparted by ionizing radiation per unit mass of irradiated material. The units of absorbed dose are the rad and the gray (Gy).

"Act" means the Atomic Energy Act of 1954 (42 U.S.C. 2011 et seq.), as amended.

"Activity" is the rate of disintegration (transformation) or decay of radioactive material. The units of activity are the curie (Ci) and the becquerel (Bq).

"Adult" means an individual 18 or more years of age.

"Airborne radioactive material" means radioactive material dispersed in the air in the form of dusts, fumes, particulates, mists, vapors, or gases.

"Airborne radioactivity area" means a room, enclosure, or area in which airborne radioactive materials, composed wholly or partly of licensed material, exist in concentrations--

(1) In excess of the derived air concentrations (DACs) specified in Appendix B, or

(2) To such a degree that an individual present in the area without respiratory protective equipment could exceed, during the hours an individual is present in a week, an intake of 0.6 percent of the annual limit on intake (ALI) or 12 DAC-hours.

"Annual limit on intake" (ALI) means the derived limit for the amount of radioactive material taken into the body of an adult worker by inhalation or ingestion in a year. ALI is the smaller value of intake of a given radionuclide in a year by the reference man that would result in a committed effective dose equivalent of 5 rems (0.05 Sv) or a committed dose equivalent of 50 rems (0.5 Sv) to any individual organ or tissue. (ALI values for intake by ingestion and by inhalation of selected radionuclides are given in Table 1, Columns 1 and 2, of Appendix B.)

"ALARA" (acronym for "as low as is reasonably achievable") means making every reasonable effort to maintain exposures to radiation as far below the dose limits in this part as is practical consistent with the purpose for which the licensed activity is undertaken, taking into account the state of technology, the economics of improvements in relation to state of technology, the economics of improvements in relation to benefits to the public health and safety, and other societal and socioeconomic considerations, and in relation to utilization of nuclear energy and licensed materials in the public interest. "Background radiation" means radiation from cosmic sources; naturally occurring radioactive materials, including radon in concentrations or levels commonly found in structures or the environment; and global fallout as it commonly exists in the environment from the testing of nuclear explosive devices. "Background radiation" does not include radiation from source, byproduct, or special nuclear materials regulated by the Commission.

"Bioassay" (radiobioassay) means the determination of kinds, quantities or concentrations, and, in some cases, the locations of radioactive material in the human body, whether by direct measurement (in vivo counting) or by analysis and evaluation of materials excreted or removed from the human body.

"Byproduct material" means --

(1) Any radioactive material (except special nuclear material) yielded in, or made radioactive by, exposure to the radiation incident to the process of producing or utilizing special nuclear material; and

(2) The tailings or wastes produced by the extraction or concentration of uranium or thorium from ore processed primarily for its source material content, including discrete surface wastes resulting from uranium solution extraction processes. Underground ore bodies depleted by these solution extraction operations do not constitute "byproduct material" within this definition.

"Class" (or "lung class" or "inhalation class") means a classification scheme for inhaled material according to its rate of clearance from the pulmonary region of the lung. Materials are classified as D, W, or Y, which applies to a range of clearance half-times: for Class D (Days) of less than 10 days, for Class W (Weeks) from 10 to 100 days, and for Class Y (Years) of greater than 100 days.

"Collective dose" is the sum of the individual doses received in a given period of time by a specified population from exposure to a specified source of radiation.

"Commission" means the Nuclear Regulatory Commission or its duly authorized representatives.

"Committed dose equivalent"  $(H_{T,50})$  means the dose equivalent to organs or tissues of reference (T) that will be received from an intake of radioactive material by an individual during the 50-year period following the intake.

"Committed effective dose equivalent" ( $H_{E,50}$ ) is the sum of the products of the weighting factors applicable to each of the body organs or tissues that are irradiated and the committed dose equivalent to these organs or tissues ( $H_{E,50} = \Sigma w_T H_{T,50}$ ).

"Controlled area" means an area, outside of a restricted area but inside the site boundary, access to which can be limited by the licensee for any reason.

"Declared pregnant woman" means a woman who has voluntarily informed her employer, in writing, of her pregnancy and the estimated date of conception.

"Deep-dose equivalent" (H<sub>d</sub>), which applies to external whole-body exposure, is the dose equivalent at a tissue depth of 1 cm (1000 mg/cm<sup>2</sup>).

"Department" means the Department of Energy established by the Department of Energy Organization Act (Pub. L. 95-91, 91 Stat. 565, 42 U.S.C. 7101 et seq.) to the extent that the Department, or its duly authorized representatives, exercises functions formerly vested in the U.S. Atomic Energy Commission, its Chairman, members, officers, and components and transferred to the U.S. Energy Research and Development Administration and to the Administrator thereof pursuant to sections 104 (b), (c), and (d) of the Energy Reorganization Act of 1974 (Pub. L. 93-438, 88 Stat. 1233 at 1237, 42 U.S.C. 5814) and retransferred to the Secretary of Energy pursuant to section 301(a) of the Department of Energy Organization Act (Pub. L. 95-91, 91 Stat. 565 at 577-578, 42 U.S.C. 7151).

"Derived air concentration" (DAC) means the concentration of a given radionuclide in air which, if breathed by the reference man for a working year of 2,000 hours under conditions of light work (inhalation rate 1.2 cubic meters of air per hour), results in an intake of one ALI. DAC values are given in Table 1, Column 3, of Appendix B. "Derived air concentration-hour" (DAC-hour) is the product of the concentration of radioactive material in air (expressed as a fraction or multiple of the derived air concentration for each radionuclide) and the time of exposure to that radionuclide, in hours. A licensee may take 2,000 DAC-hours to represent one ALI, equivalent to a committed effective dose equivalent of 5 rems (0.05 Sv).

"Dose" or "radiation dose" is a generic term that means absorbed dose, dose equivalent, effective dose equivalent, committed dose equivalent, committed effective dose equivalent, or total effective dose equivalent, as defined in other paragraphs of this section.

"Dose equivalent"  $(H_T)$  means the product of the absorbed dose in tissue, quality factor, and all other necessary modifying factors at the location of interest. The units of dose equivalent are the rem and sievert (Sv).

"Dosimetry processor" means an individual or an organization that processes and evaluates individual monitoring equipment in order to determine the radiation dose delivered to the equipment.

"Effective dose equivalent" (H<sub>E</sub>) is the sum of the products of the dose equivalent to the organ or tissue (H<sub>T</sub>) and the weighting factors (w<sub>T</sub>) applicable to each of the body organs or tissues that are irradiated (H<sub>E</sub> =  $\Sigma w_T H_T$ ).

"Embryo/fetus" means the developing human organism from conception until the time of birth.

"Entrance or access point" means any location through which an individual could gain access to radiation areas or to radioactive materials. This includes entry or exit portals of sufficient size to permit human entry, irrespective of their intended use.

"Exposure" means being exposed to ionizing radiation or to radioactive material.

"External dose" means that portion of the dose equivalent received from radiation sources outside the body.

"Extremities" means hand, elbow, arm below the elbow, foot, knee, and leg below the knee.

"Eye dose equivalent" applies to the external exposure of the lens of the eye and is taken as the dose equivalent at a tissue depth of 0.3 centimeter  $(300 \text{ mg/cm}^2)$ .

"Generally applicable environmental radiation standards" means standards issued by the Environmental Protection Agency (EPA) under the authority of the Atomic Energy Act of 1954, as amended, that impose limits on radiation exposures or levels, or concentrations or quantities of radioactive material, in the general environment outside the boundaries of locations under the control of persons possessing or using radioactive material.

"Government agency" means any executive department, commission, independent establishment, corporation wholly or partly owned by the United States of America, which is an instrumentality of the United States, or any board, bureau, division, service, office, officer, authority, administration, or other establishment in the executive branch of the Government.

"Gray" [See § 20.4].

"High radiation area" means an area, accessible to individuals, in which radiation levels could result in an individual receiving a dose equivalent in excess of 0.1 rem (1 mSv) in 1 hour at 30 centimeters from the radiation source or from any surface that the radiation penetrates.

"Individual" means any human being.

119

"Individual monitoring" means--

 The assessment of dose equivalent by the use of devices designed to be worn by an individual;

(2) The assessment of committed effective dose equivalent by bioassay (see Bioassay) or by determination of the time-weighted air concentrations to which an individual has been exposed, i.e., DAC-hours; or

(3) The assessment of dose equivalent by the use of survey data.

"Individual Monitoring Devices" ("individual monitoring equipment") means devices designed to be worn by a single individual for the assessment of dose equivalent such as film badges, thermoluminescent dosimeters (TLDs), pocket ionization chambers, and personal ("lapel") air sampling devices.

"Internal dose" means that portion of the dose equivalent received from radioactive material taken into the body.

"License" means a license issued under the regulations in Parts 30 through 35, 39, 40, 50, 60, 61, 70, or 72 of this chapter.

"Licensee" means the holder of a license.

"Licensed material" means source material, special nuclear material, or byproduct material received, possessed, used, transferred or disposed of under a general or specific license issued by the Commission.

"Limits" (dose limits) means the permissible upper bounds of radiation doses.

"Lost or missing licensed material" means licensed material whose location is unknown. It includes material that has been shipped but has not reached its destination and whose location cannot be readily traced in the transportation system.

"Member of the public" means an individual in a controlled or unrestricted area. However, an individual is not a member of the public during any period in which the individual receives an occupational dose. "Minor" means an individual less than 18 years of age.

"Monitoring" (radiation monitoring, radiation protection monitoring) means the measurement of radiation levels, concentrations, surface area concentrations or quantities of radioactive material and the use of the results of these measurements to evaluate potential exposures and doses.

"Nonstochastic effect" means health effects, the severity of which varies with the dose and for which a threshold is believed to exist. Radiation-induced cataract formation is an example of a nonstochastic effect.

"NRC" means the Nuclear Regulatory Commission or its duly authorized representatives.

"Occupational dose" means the dose received by an individual in a restricted area or in the course of employment in which the individual's assigned duties involve exposure to radiation and to radioactive material from licensed and unlicensed sources of radiation, whether in the possession of the licensee or other person. Occupational dose does not include dose received from background radiation, as a patient from medical practices, from voluntary participation in medical research programs, or as a member of the general public.

#### "Person" means ---

(1) Any individual, corporation, partnership, firm, association, trust, estate, public or private institution, group, Government agency other than the Commission or the Department of Energy (except that the Department shall be considered a person within the meaning of the regulations in 10 CFR Chapter 1 to the extent that its facilities and activities are subject to the licensing and related regulatory authority of the Commission under Section 202 of the Energy Reorganization Act of 1974 (88 Stat. 1244), the Uranium Mill Tailings Radiation Control Act of 1978 (92 Stat. 3021), the Nuclear Waste Policy Act of 1982 (96 Stat. 2201), and Section 3(b)(2) of the Low-Level Radioactive Waste Policy Amendments Act of 1985 (99 Stat. 1842)), any State or any political subdivision of or any political entity within a State, any foreign government or nation or any political subdivision of any such government or nation, or other entity; and

(2) Any legal successor, representative, agent, or agency of the foregoing.

"Planned special exposure" means an infrequent exposure to radiation, separate from and in addition to the annual dose limits.

"Public dose" means the dose received by a member of the public from exposure to radiation and to radioactive material released by a licensee, or to another source of radiation either within a licensee's controlled area or in unrestricted areas. It does not include occupational dose or doses received from background radiation, as a patient from medical practices, or from voluntary participation in medical research programs.

"Quality Factor" (Q) means the modifying factor (listed in Tables 1 and 2 of § 20.4) that is used to derive dose equivalent from absorbed dose.

"Quarter" means a period of time equal to one-fourth of the year observed by the licensee (approximately 13 consecutive weeks), providing that the heginning of the first quarter in a year coincides with the starting date of the year and that no day is omitted or duplicated in consecutive quarters.

"Rad" [See § 20.4].

"Radiation" (ionizing radiation) means alpha particles, beta particles, gamma rays, x-rays, neutrons, high-speed electrons, high-speed protons, and other particles capable of producing ions. Radiation, as used in this part, does not include non-ionizing radiation, such as radio- or microwaves, or visible, infrared, or ultraviolet light.

"Radiation area" means an area, accessible to individuals, in which radiation levels could result in an individual receiving a dose equivalent in excess of 0.005 rem (0.05 mSv) in 1 hour at 30 centimeters from the radiation source or from any surface that the radiation penetrates. "Reference man" means a hypothetical aggregation of human physical and physiological characteristics arrived at by international consensus. These characteristics may be used by researchers and public health workers to standardize results of experiments and to relate biological insult to a common base.

"Rem" [See § 20.4].

"Respiratory protective device" means an apparatus, such as a respirator, used to reduce the individual's intake of airborne radioactive materials.

"Restricted area" means an area, access to which is limited by the licensee for the purpose of protecting individuals against undue risks from exposure to radiation and radioactive materials. Restricted area does not include areas used as residential quarters, but separate rooms in a residential building may be set apart as a restricted area.

"Sanitary sewerage" means a system of public sewers for carrying off waste water and refuse, but excluding sewage treatment facilities, septic tanks, and leach fields owned or operated by the licensee.

"Shallow-dose equivalent" (H<sub>s</sub>), which applies to the external exposure of the skin or an extremity, is taken as the dose equivalent at a tissue depth of 0.007 centimeter (7 mg/cm<sup>2</sup>) averaged over an area of 1 square centimeter.

"Sievert" [See § 20.4].

"Site boundary" means that line beyond which the land or property is not owned, leased, or otherwise controlled by the licensee.

"Source material" means--

(1) Uranium or thorium or any combination of uranium and thorium in any physical or chesical form; or

(2) Ores that contain, by weight, one-twentieth of 1 percent (0.05 percent), or more, of uranium, thorium, or any combination of uranium and thorium. Source material does not include special nuclear material.

123

"Special nuclear material" means-

(1) Plutonium, uranium-233, uranium enriched in the isotope 233 or in the isotope 235, and any other material that the Commission, pursuant to the provisions of section 51 of the Act, determines to be special nuclear material, but does not include source material; or

(2) Any material artificially enriched by any of the foregoing but does not include source material.

"Stochastic effects" means health effects that occur randomly and for which the probability of the effect occurring, rather than its severity, is assumed to be a linear function of dose without threshold. Hereditary effects and cancer incidence are examples of stochastic effects.

"Survey" means an evaluation of the radiological conditions and potential hazards incident to the production, use, transfer, release, disposal, or presence of radioactive material or other sources of radiation. When appropriate, such an evaluation includes a physical survey of the location of radioactive material and measurements or calculations of levels of radiation, or concentrations or quantities of radioactive material present.

"Total Effective Dose Equivalent" (TEDE) means the sum of the deep-dose equivalent (for external exposures) and the committed effective dose equivalent (for internal exposures).

"Unrestricted area" means an area, access to which is neither limited nor controlled by the licensee.

"Uranium fuel cycle" means the operations of milling of uranium ore, chemical conversion of uranium, isotopic enrichment of uranium, fabrication of uranium fuel, generation of electricity by a light-water-cooled nuclear power plant using uranium fuel, and reprocessing of spent uranium fuel to the extent that these activities directly support the production of electrical power for public use. Uranium fuel cycle does not include mining operations, operations at waste disposal sites, transportation of radioactive material in support of these operations, and the reuse of recovered non-uranium special nuclear and byproduct materials from the cycle. "Very high radiation area" means an area, accessible to individuals, in which radiation levels could result in an individual receiving an absorbed dose in excess of 500 rads (5 grays) in 1 hour at 1 meter from a radiation source or from any surface that the radiation penetrates. [Note: At very high doses received at high dose rates, units of absorbed dose (e.g., rads and grays) are appropriate, rather than units of dose equivalent (e.g., rems and sieverts).]

"Week" means 7 consecutive days starting on Sunday.

"Weighting factor,"  $w_T$ , for an organ or tissue (T) is the proportion of the risk of stochastic effects resulting from irradiation of that organ or tissue to the total risk of stochastic effects when the whole body is irradiated uniformly. For calculating the effective dose equivalent, the values of  $w_T$  are:

Ongen on	and the subscription of a product of the
Tissue	₩ <sub>T</sub>
Gonads	0.25
Breast	0.15
Red bone marrow	0.12
Lung	0.12
Thyroid	0.03
Bone surfaces	0.03
emainder	0.30 <sup>a</sup>
hole Body	1.00 <sup>b</sup>

<sup>a</sup> 0.30 results from 0.06 for each of 5 "reminder" organs (excluding the skin and the lens of the eye) that receive the highest doses.

<sup>b</sup> For the purpose of weighting the external whole body dose (for adding it to the internal dose), a single weighting factor,  $w_T = 1.0$ , has been specified. The use of other weighting factors for external exposure will be approved on a case-by-case basis until such time as specific guidance is issued. "Whole body" means, for purposes of external exposure, head, trunk (including male gonads), arms above the elbow, or legs above the knee.

"Working level" (WL) is any combination of short-lived radon daughters (for radon-222: polonium-218, lead-214, bismuth-214, and polonium-214; and for radon-220: polonium-216, lead-212, bismuth-212, and polonium-212) in 1 liter of air that will result in the ultimate emission of 1.3 x  $10^5$  MeV of potential alpha particle energy.

"Working level month" (WLM) means an exposure to 1 working level for 170 hours (2,000 working hours per year/12 months per year = approximately 170 hours per month).

"Year" means the period of time beginning in January used to determine compliance with the provisions of this part. The licensee may change the starting date of the year used to determine compliance by the licensee provided that the change is made at the beginning of the year and that no day is omitted or duplicated in consecutive years.

# § 20.4 Units of radiation dose.

(a) As used in this part, the units of radiation dose are:

"Gray" (Gy) is the SI unit of absorbed dose. One gray is equal to an absorbed dose of 1 Joule/kilogram (100 rads).

"Rad" is the special unit of absorbed dose. One rad is equal to an absorbed dose of 100 ergs/gram or 0.01 joule/kilogram (0.01 gray).

"Rem" is the special unit of any of the quantities expressed as dose equivalent. The dose equivalent in rems is equal to the absorbed dose in rads multiplied by the quality factor (1 rem = 0.01 sievert).

"Sievert" is the SI unit of any of the quantities expressed as dose equivalent. The dose equivalent in sieverts is equal to the absorbed dose in grays multiplied by the quality factor (1 Sv = 100 rems).

(b) As used in this part, the quality factors for converting absorbed dose to dose equivalent are shown in Table 1.

Qu	ality Factor	Absorbed Dose Equal to a Unit Dose Equivalent	
TYPE OF RADIATION	(Q)		
X-, gamma, or beta radiation	1	1	
Alpha particles, multiple-charged particles, fission fragments and heavy particles of unknown charge	20	0.05	
Neutrons of unknown energy	10	0.1	
High-energy protons	10	0.1	

			TABLE 1		
QUALITY	FACTORS	AND	ABSORBED	DOSE	EQUIVALENCIES

(c) If it is more convenient to measure the neutron fluence rate than to determine the neutron dose equivalent rate in rems per hour or sieverts per hour, as provided in paragraph (b) of this section, 1 rem (0.01 Sv) of neutron radiation of unknown energies may, for purposes of the regulations in this part, be assumed to result from a total fluence of 25 million neutrons per square centimeter incident upon the body. If sufficient information exists to estimate the approximate energy distribution of the neutrons, the licensee may use the fluence rate per unit dose equivalent or the appropriate Q value from Table 2 to convert a measured tissue dose in rads to dose equivalent in rems.

-		25.1		per -	10
- 12	n	ю і		6 C	9.
- 81	n	21	ini i	£	6.
				-	

	Neutron	Quality	Fluence per Unit
	Energy (MeV)	Factor <sup>a</sup> (Q)	Dose Equivalent <sup>b</sup> (neutrons cm- <sup>2</sup> rem- <sup>1</sup> )
(thermal)	$2.5 \times 10^{-8}$ $1 \times 10^{-7}$ $1 \times 10^{-6}$ $1 \times 10^{-5}$ $1 \times 10^{-4}$ $1 \times 10^{-3}$ $1 \times 10^{-2}$ $1 \times 10^{-1}$ $5 \times 10^{-1}$ $1$ $2.5$ $5$ $7$ $10$ $14$ $20$ $40$ $60$ $1 \times 10^{2}$ $2 \times 10^{2}$ $3 \times 10^{2}$ $4 \times 10^{2}$	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	$\begin{array}{c} 980 \times 10^{6} \\ 980 \times 10^{6} \\ 810 \times 10^{6} \\ 810 \times 10^{6} \\ 840 \times 10^{6} \\ 980 \times 10^{6} \\ 1010 \times 10^{6} \\ 170 \times 10^{6} \\ 29 \times 10^{6} \\ 27 \times 10^{6} \\ 23 \times 10^{6} \\ 23 \times 10^{6} \\ 24 \times 10^{6} \\ 17 \times 10^{6} \\ 16 \times 10^{6} \\ 16 \times 10^{6} \\ 16 \times 10^{6} \\ 16 \times 10^{6} \\ 19 \times 10^{6} \\ 16 \times 10^{6} \\ 14 \times 10^{6} \\ $

MEAN QUALITY FACTORS, Q, AND FLUENCE PER UNIT DOSE EQUIVALENT FOR MONOENERGETIC NEUTRONS

<sup>a</sup>Value of quality factor (Q) at the point where the dose equivalent is maximum in a 30-cm diameter cylinder tissue-equivalent phantom.

<sup>b</sup>Monoenergetic neutrons incident normally on a 30-cm diameter cylinder tissue-equivalent phantom.

# § 20.5 Units of radioactivity.

For the purposes of this part, activity is expressed in the special unit of curies (Ci) or in the SI unit of becquerels (Bq), or their multiples, or disintegrations (transformations) per unit of time.

(a) One becquerel = 1 disintegration per second  $(s^{-1})$ .

(b) One curie =  $3.7 \times 10^{10}$  disintegrations per second =

 $3.7 \times 10^{10}$  becquerels =  $2.22 \times 10^{12}$  disintegrations per minute.

# § 20.6 Interpretations.

Except as specifically authorized by the Commission in writing, no interpretation of the meaning of the regulations in this part by an officer or employee of the Commission other than a written interpretation by the General Counsel will be recognized to be binding upon the Commission.

# § 20.7 Communications.

Unless otherwise specified, communications or reports concerning the regulations in this part should be addressed to the Executive Director for Operations, U.S. Nuclear Regulatory Commission, Washington, DC 20555. A communication, report, or application may be delivered in person to the Commission's offices at 2120 L Street, NW. (Lower Level), Washington, DC 20037, or 11555 Rockville Pike, Rockville, MD 20852.

# § 20.8 Implementation.

(a) Licensees shall implement the provisions of this part on or before January 1, 1993. If a licensee chooses to implement the provisions of this part prior to January 1, 1993, the licensee shall implement all provisions of this part not otherwise exempted by paragraph (d) of this section, and shall provide written notification to either the Director of the Office of Nuclear Materials Safety and Safeguards or the Director of the Office of Nuclear Reactor Regulation, as appropriate, that the licensee is adopting early implementation of this part. Until January 1, 1993, or until the licensee notifies the Commission of early implementation of the provisions of this part, compliance will be required with the 10 CFR Part 20 in the Code of Federal Regulations on January 1, 1991.

(b) After the time the licensee implements this part, the applicable section of this part shall be used in lieu of any section of this part in effect on or before January 1, 1991 that is cited in license conditions or technical specifications, except as specified in paragraphs (c), (d) and (e) of this section. If the requirements of this part are more restrictive than the existing license condition, then the licensee shall comply with this part unless exempted by paragraph (d) of this section.

129

(c) Any existing license condition or technical specification that is more restrictive than this part remains in force until there is a technical specification change, license amendment, or license renewal.

(d) If a license condition or technical specification exempts a licensee from a provision of Part 20 in effect on or before January 1, 1991, it also exempts the licensee from the corresponding provision of this part.

(e) If no section in this part corresponds to the provisions of Part 20 in effect prior to January 1, 1991 cited in a license condition, a license condition based on Part 20 in effect on or before January 1, 1991 remains in force until either there is a technical specification change, license amendment, or license renewal that modifies or removes this condition.

# § 20.9 Reporting, recording, and application requirements: OMB approval.

(a) The Nuclear Regulatory Commission will submit the information collection requirements contained in this part to the Office of Management and Budget for approval as required by the Paperwork Reduction Act of 1980 (44 U.S.C. 3501 et seq.). The information collection requirements in this part will not become effective until OMB clearance is obtained.

(b) The information collection requirements contained in this part appear in §§ 20.101, 20.202, 20.204, 20.206, 20.301, 20.501, 20.601, 20.603, 20.703, 20.901, 20.902, 20.904, 20.906, 20.1002, 20.1004, 20.1006, 20.1102, 20.1103, 20.1104, 20.1105, 20.1106, 20.1107, 20.1108, 20.1109, 20.1110, 20.1201, 20.1202, 20.1203, 20.1204, 20.1206, and Appendix F.

#### SUBPART B--F DIATION PROTECTION PROGRAMS

#### § 20.101 Radiation protection programs.

(a) Each licensee shall develop, document, and implement a radiation protection program commensurate with the scope and extent of licensed activities and sufficient to ensure compliance with the provisions of this part. (See § 20.1102 for recordkeeping requirements relating to these programs.)

(b) The licensee shall use, to the extent practicable, procedures and engineering controls based upon sound radiation protection principles to achieve occupational doses and doses to members of the public that are as low as is reasonably achievable (ALARA).

(c) The licensee shall periodically (at least annually) review the radiation protection program content and implementation.

#### SUBPART C--OCCUPATIONAL DOSE LIMITS

# § 20.201 Occupational dose limits for adults.

(a) The licensee shall control the occupational dose to individual adults, except for planned special exposures under § 20.206, to the following dose limits.

(1) An annual limit, which is the more limiting of--

(i) The total effective dose equivalent being equal to 5 rems(0.05 Sv); or

(ii) The sum of the deep-dose equivalent and the committed dose equivalent to any individual organ or tissue other than the lens of the eye being equal to 50 rems (0.5 Sv).

(2) The annual limits to the lens of the eye, to the skin, and to the extremities which are:

(i) An eye dose equivalent of 15 rems (0.15 Sv), and

(ii) A shallow-dose equivalent of 50 rems (0.50 Sv) to the skin or to each of the extremities.

(b) Doses received in excess of the annual limits, including doses received during accidents, emergencies, and planned special exposures, must be subtracted from the limits for planned special exposures that the individual may receive during the current year (see § 20.206(e)(1)) and during the individual's lifetime (see § 20.206(e)(2)).

(c) The assigned deep-dose equivalent and shallow-dose equivalent must be for the part of the body receiving the highest exposure. The deep-dose equivalent, eye dose equivalent and shallow-dose equivalent may be assessed from surveys or other "adiation measurements for the purpose of demonstrating compliance with the occupational dose limits, if the individual monitoring device was not in the region of highest potential exposure, or the results of individual monitoring are unavailable.

(d) Derived air concentration (DAC) and annual limit on intake (ALI) values are presented in Table 1 of Appendix B and may be used to determine the individual's dose (see § 20.1106) and to demonstrate compliance with the occupational dose limits.

(e) In addition to the annual dose limits, the licensee shall limit the soluble uranium intake by an individual to 10 milligrams in a week in consideration of chemical toxicity (see footnote 3 of Appendix B).

(f) The licensee shall reduce the dose that an individual may be allowed to receive in the current year by the amount of occupational dose received while employed by any other person (see § 20.1104(e)).

# § 20.202 Compliance with requirements for summation of external and internal doses.

(a) If the licensee is required to monitor under both §§ 20.502(a) and (b), the licensee shall demonstrate compliance with the dose limits by summing external and internal doses. If the licensee is required to monitor only under § 20.502(a) or only under § 20.502(b), then summation is not required to demonstrate compliance with the dose limits. The licensee may demonstrate compliance with the requirements for summation of external and internal doses by meeting one of the conditions specified in paragraph (b) of this section and the conditions in paragraphs (c) and (d) of this section. (Note: The dose equivalents for the lens of the eye, the skin, and the extremities are not included in the summation, but are subject to separate limits.)

(b) <u>Intake by inhalation</u>. If the only intake of radionuclides is by inhalation, the total effective dose equivalent limit is not exceeded if the sum of the deep-dose equivalent divided by the total effective dose equivalent limit, and one of the following, does not exceed unity:

(1) The sum of the fractions of the inhalation ALI for each radionuclide, or

(2) The total number of derived air concentration-hours (DAC-hours) for all radionuclides divided by 2,000, or

(3) The sum of the calculated committed effective dose equivalents to all significantly irradiated organs or tissues (T) calculated from bioassay data using appropriate biological models and expressed as a fraction of the annual limit.

(c) <u>Intake by oral ingestion</u>. If the occupationally exposed individual also receives an intake of radionuclides by oral ingestion greater than 10 percent of the applicable oral ALI, the licensee shall account for this intake and include it in demonstrating compliance with the limits.

(d) <u>Intake through wounds or absorption through skin</u>. The licensee shall evaluate and, to the extent practical, account for intakes through wounds or skin absorption. (NOTE: The intake through intact skin has been included in the calculation of DAC for hydrogen-3 and does not need to be further evaluated.)

# § 20.203 Determination of external dose from airborne radioactive material.

Licensees shall, when determining the dose from airborne radioactive material, i clude the contribution to the deep-dose equivalent, eye dose equivalent, and shallow-dose equivalent from external exposure to the radioactive cloud (see Appendix B, footnotes 1 and 2).

NOTE: Airborne radioactivity measurements and DAC values should not be used as the primary means to assess the deep-dose equivalent when the airborne radioactive material includes radionuclides other than noble gases or if the

An organ or tissue is deemed to be significantly irradiated if, for that organ or tissue, the product of the weighting factors,  $w_T$ , and the committed dose equivalent,  $H_{50}$ , per unit intake is greater than 10 percent of the maximum weighted value of  $H_{50}$  (i.e.,  $w_T H_{50,T}$ ) per unit intake for any organ or tissue.

cloud of airborne radioactive material is not relatively uniform. The determination of the deep-dose equivalent to an individual should be based upon measurements using instruments or individual monitoring devices.

#### § 20.204 Determination of internal exposure.

(a) For purposes of assessing dose used to determine compliance with occupational dose equivalent limits, the licensee shall, when required under § 20.502, take suitable and timely measurements of --

(1) Concentrations of radioactive materials in air in work areas; or

- (2) Quantities of radionuclides in the body; or
- (3) Quantities of radionuclides excreted from the body; or
- (4) Combinations of these measurements.

(b) Unless respiratory protective equipment is used, as provided in § 20.703, or the assessment of intake is based in bioassays. the licensee shall assume that an individual inhales radioactive material at the airborne concentration in which the individual is present.

(c) When specific information on the physical and biochemical properties of the radionuclides taken into the body or the behavior or the material in an individual is known, the licensee may--

(1) Use that information to calculate the committed effective dose equivalent, and, if used, the licensee shall document that information in the individual's record; and

(2) Upon prior approval of the Commission, adjust the DAC or ALI values to reflect the actual physical and chemical characteristics of airborne radioactive material (e.g., aerosol size distribution or density); and

(3) Separately assess the contribution of fractional intakes of Class D,W, or Y compounds of given radionuclide (see Append x B) to the committed effective dose equivalent.

(d) If the licensee chooses to assess intakes of Class Y material using the measurements given in § 20.204(a)(2) or (3), the licensee may delay the recording and reporting of the assessments for periods up to 7 months, unless otherwise required by §§ 20.1202 or 20.1203, in order to permit the licensee to make additional measurements basic to the assessments.

(e) If the identity and concentration of each radionuclide in a mixture are known, the fraction of the DAC applicable to the mixture for use in calculating DAC-hours must be either--

(1) The sum of the ratios of the concentration to the appropriate DAC value (e.g., D, W, Y) from Appendix B for each radionuclide in the mixture; or

(2) The ratio of the total concentration for all radionuclides in the mixture to the most restrictive DAC value for any radionuclide in the mixture.

(f) If the identity of each radionuclide in a mixture is known, but the concentration of one or more of the radionuclides in the mixture is not known, the DAC for the mixture must be the most restrictive DAC of any radionuclide in the mixture.

(g) When a mixture of radionuclides in air exists, licensees may disregard certain radionuclides in the mixture if--

(1) The licensee uses the total activity of the mixture in demonstrating compliance with the dose limits in § 20.201 and in complying with the monitoring requirements in § 20.502(b), and

(2) The concentration of any radionuclide disregarded is less than 10 percent of its DAC, and

(3) The sum of these percentages for all of the radionuclides disregarded in the mixture does not exceed 30 percent.

(h)(1) In order to calculate the committed effective dose equivalent, the licensee may assume that the inhalation of one ALI, or an exposure of 2,000 DAC-hours, results in a committed effective dose equivalent of 5 rems (0.05 Sv) for radionuclides that have their ALIs or DACs based on the committed effective dose equivalent.

(2) When the ALI (and the associated DAC) is determined by the nonstochastic organ dose limit of 50 rems (0.5 Sv), the intake of radionuclides that would result in a committed effective dose equivalent of 5 rems (0.05 Sv) (the stochastic ALI) is listed in parentheses in Table 1 of Appendix B. In this case, the licensee may, as a simplifying assumption, use the stochastic ALIs to determine committed effective dose equivalent. However, if the licensee uses the stochastic ALIs, the licensee must also demonstrate that the limit in § 20.201(a)(1)(ii) is met. § 20.205 [Reserved]

§ 20.206 Planned special exposures.

A licensee may authorize an adult worker to receive doses in addition to and accounted for separately from the doses received under the limits specified in § 20.201 provided that each of the following conditions is satisfied--

(a) The licensee authorizes a planned special exposure only in an exceptional situation when alternatives that might avoid the higher exposure are unavailable or impractical.

(b) The ficensee (and employer if the employer is not the licensee) specifically authorizes the planned special exposure, in writing, before the exposure occurs.

(c) Before a planned special exposure, the licensee ensures that the individuals involved are--

(1) Informed of the purpose of the planned operation;

(2) Informed of the estimated doses and associated potential risks and specific radiation levels or other conditions that might be involved in performing the task; and

(3) Instructed in the measures to be taken to keep the dose ALARA considering other risks that may be present.

(d) Prior to permitting an individual to participate in a planned special exposure, the licensee ascertains prior doses as required by § 20.1104(b) during the lifetime of the individual for each individual involved.

(e) Subject to § 20.201(b), the licensee does not authorize a planned special exposure that would cause an individual to reclive a dose from all planned special exposures and all doses in excess of the limits to exceed--

(1) The numerical values of any of the dose limits in § 20.201(a) in any year; and

(2) Five times the annual dose limits in § 20.201(a) during the individual's lifetime.

(f) The licensee maintains records of the conduct of a planned special exposure in accordance with § 20.1105 and submits a written report in accordance with § 20.1204.

(g) The licensee records the best estimate of the dose resulting from the planned special exposure in the individual's record and informs the individual, in writing, of the dose within 30 days from the date of the planned special exposure. The dose from planned special exposures is not to be considered in controlling future occupational dose of the individual under § 20.201(a) but is to be included in evaluations required by § 20.206(d) and (e).

# § 20.207 Occupational dose limits for minors.

The annual occupational dose limits for minors are 10 percent of the annual dose limits specified for adult workers in § 20.201.

#### § 20.208 Dose to an embryo/fetus.

(a) The licensee shall ensure that the dose to an embryo/fetus during the entire pregnancy, due to occupational exposure of a declared pregnant woman, does not exceed 0.5 rem (5 mSv). (For recordkeeping requirements, see § 20.1106.)

(b) The licensee shall make efforts to avoid substantial variation above a uniform monthly exposure rate to a declared pregnant woman so as to satisfy the limit in paragraph (a) of this section.

- (c) The dose to an embryo/fetus shall be taken as the sum of--
- (1) The deep-dose equivalent to the declared pregnant woman; and
- (2) The dose to the embryo/fetus from radionuclides in the embryo/fetus and radionuclides in the declared pregnant woman.

(d) If the dose to the embryo/fetus is found to have exceeded 0.5 rem (5 mSv), or is within 0.05 rem (0.5 mSv) of this dose, by the time the woman declares the pregnancy to the licersee, the licensee shall be deemed to be in compliance with paragraph (a) of this section if the additional dose to the embryo/fetus does not exceed 0.05 rem (0.5 mSv) during the remainder of the pregnancy.

# SUBPART D--RADIATION DOSE LIMITS FOR INDIVIDUAL MEMBERS OF THE PUBLIC

# § 20.301 Dose limits for individual members of the public.

(a) Each licensee shall conduct operations so that --

(1) The total effective dose equivalent to individual members of the public from the licensed operation does not exceed 0.1 rem (1 mSv) in a year, exclusive of the dose contribution from the licensee's disposal of radioactive material into sanitary sewerage in accordance with § 20.1003, and

(2) The dose in any unrestricted area from external sources does not exceed 0.002 rem (0.02 mSv) in any one hour.

(b) If the licensee permits members of the public to have access to controlled areas, the limits for members of the public continue to apply to those individuals.

(c) A licensee or license applicant may apply for prior NRC authorization to operate up to an annual dose limit for an individual member of the public of 0.5 rem (5 mSv). The licensee or license applicant shall include the following information in this application:

 Demonstration of the need for and the expected duration of operations in excess of the limit in paragraph (a) of this section;

(2) The licensee's program to assess and control dose within the 0.5 rem(5 mSv) annual limit; and

(3) The procedures to be followed to maintain the dose as low as is reasonably achievable.

(d) In addition to the requirements of this part, a licensee subject to the provisions of EPA's generally applicable environmental radiation standards in 40 CFR Part 190 shall comply with those standards.

(e) The Commission may impose additional restrictions on radiation levels in unrestricted areas and on the total quantity of radionuclides that a licensee may release in effluents in order to restrict the collective dose.

138

§ 20.302 Compliance with dose limits for individual members of the public.

(a) The licensee shall make or cause to be made, as appropriate, surveys of radiation levels in unrestricted and controlled areas and radioactive materials in effluents released to unrestricted and controlled areas to demonstrate compli. Ice with the dose limits for individual members of the public in § 20.301.

(b) A licensee shall show compliance with the annual dose limit in § 20.301 by--

(1) Demonstrating by measurement or calculation that the total effective dose equivalent to the individual likely to receive the highest dose from the licensed operation does not exceed the annual dose limit; or

(2) Demonstrating that --

(i) The annual average concentrations of radioactive material released in gaseous and liquid effluents at the boundary of the unrestricted area do not exceed the values specified in Table 2 of Appendix B; and

(ii) If an individual were continually present in an unrestricted area, the dose from external sources would not exceed 0.002 rem (0.02 mSv) in an hour and 0.05 rem (0.5 mSv) in a year.

(c) Upon approval from the Commission, the licensee may adjust the effluent concentration values in Appendix B, Table 2, for members of the public, to take into account the actual physical and chemical characteristics of the effluents (e.g., aerosol size distribution, solubility, density, radioactive decay equilibrium, chemical form).

# SUBPART E--[RESERVED]

## SUBPART F -- SURVEYS AND MONITORING

#### § 20.501 General.

(a) Each licensee shall make or cause to be made, surveys that --

(1) May be necessary for the licensee to comply with the regulations in this part; and

(2) Are reasonable under the circumstances to evaluate --

(i) The extent of radiation levels; and

(ii) Concentrations or quantities of radioactive waterial; and

(iii) The potential radiological hazards that could be present.

(b) The licensee shall ensure that instruments and equipment used for quantitative radiation measurements (e.g., dose rate and effluent monitoring) are calibrated periodically for the radiation measured.

(c) All personnel dosimeters (except for direct and indirect reading pocket ionization chambers and those dosimeters used to measure the dose to the extremities) that require processing to determine the radiation dose and that are used by licensees to comply with § 20.201, with other applicable provisions of this chapter, or with conditions specified in a license must be processed and evaluated by a dosimetry processor--

(1) Holding current personnel dosimetry accreditation from the National Voluntary Laboratory Accreditation Program (NVLAP) of the National Institute of Standards and Technology; and

(2) Approved in this accreditation process for the type of radiation or radiations included in the NVLAP program that most closely approximates the type of radiation or radiations for which the individual wearing the dosimeter is monitored.

# § 20.502 <u>Conditions requiring individual monitoring of external and</u> internal occupational dose.

Each licensee shall monitor exposures to radiation and radioactive material at levels sufficient to demonstrate compliance with the occupational dose limits of this part. As a minimum--

(a) Each licensee shall monitor occupational exposure to radiation and shall supply and require the use of individual monitoring devices by--

(1) Adults likely to receive, in 1 year from sources external to the body, a dose in excess of 10 percent of the limits in § 20.201(a),

(2) Minors and declared pregnant women likely to receive, in 1 year from sources external to the body, a dose in excess of 10 percent of any of the applicable limits in §§ 20.207 or 20.208, and
(3) Individuals entering a high or very high radiation area.

(b) Each licensee shall monitor (see § 20.204) the occupational intake of radioactive material by and assess the committed effective dose equivalent to---

 Adults likely to receive, in 1 year, an intake in excess of 10 percent of the applicable ALI(s) in Table 1, Columns 1 and 2, of Appendix B; and

(2) Minors and declared pregnant women likely to receive, in 1 year, a committed effective dose equivalent in excess of 0.05 rem (0.5 mSv).

## SUBPART G--CONTROL OF EXPOSURE FROM EXTERNAL SOURCES IN RESTRICTED AREAS

## § 20.601 Control of access to high radiation areas.

(a) The licensee shall ensure that each entrance or access point to a high radiation area has one or more of the following features--

(1) A control device that, upon entry into the area, causes the level of radiation to be reduced below that level at which an individual might receive a deep-dose equivalent of 0.1 rem (1 mSv) in 1 hour at 30 centimeters from the radiation source or from any surface that the radiation penetrates;

(2) A control device that energizes a conspicuous visible or audible alarm signal so that the individual entering the high radiation area and the supervisor of the activity are made aware of the entry; or

(3) Entryways that are locked, except during periods when access to the areas is required, with positive control over each individual entry.

(b) In place of the controls required by paragraph (a) of this section for a high radiation area, the licensee may substitute continuous direct or electronic surveillance that is capable of preventing unauthorized entry.

(c) A licensee may apply to the Commission for approval of alternative methods for controlling access to high radiation areas.

(d) The licensee shall establish the controls required by paragraphs (a) and (c) of this section in a way that does not prevent individuals from leaving a high radiation area.

(e) Control is not required for each entrance or access point to a room or other area that is a high radiation area solely because of the presence of radioactive materials prepared for transport and packaged and labeled in accordance with the regulations of the Department of Transportation provided that--

(1) The packages do not remain in the area longer than 3 days; and

(2) The dose rate at 1 meter from the external surface of any package es not exceed 0.01 rem (0.1 mSv) per hour.

(f) Control of entrance or access to rooms or other areas in hospitals is not required solely because of the presence of patients containing radioactive material, provided that there are personnel in attendance who will take the necessary precautions to prevent the exposure of individuals to radiation or radioactive material in excess of the limits established in this part and to operate within the ALARA provisions of the licensee's radiation protection program.

#### § 20.602 Control of access to very high radiation areas.

In addition to the requirements in § 20.601, the licensee shall institute additional measures to ensure that an individual is not able to gain unauthorized or inadvertent access to areas in which radiation levels could be encountered at 500 rads (5 grays) or more in 1 hour at 1 meter from a radiation source or any surface through which the radiation penetrates.

# § 20.603 Control of access to very high radiation areas - irradiators.

(a) Each area in which there may exist radiation levels in excess of 500 rads (5 grays) in 1 hour at 1 meter from a sealed radioactive source<sup>2</sup> that is used to irradiate materials must meet the following requirements.

(1) Each entrance or access point must be equipped with entry control devices which--

(i) Function automatically to prevent any individual from inadvertently entering the area when very high radiation levels exist.

(ii) Permit deliberate entry into the area only after a control device is actuated that causes the radiation level within the area, from the sealed source, to be reduced below that at which it would be possible for an individual to receive a deep-dose equivalent in excess of 0.1 cm (1 mSv) in 1 hour; and

(iii) Prevent operation of the source if the source would produce radiation levels in the area that could result in a deep-dose equivalent to an individual in excess of 0.1 rem (1 mSv) in 1 hour.

(2) Additional control devices must be provided so that, upon failure of the entry control devices to function as required by paragraph (a)(1) of this section--

(i) The radiation level within the area, from the sealed source, is reduced below that at which it would be possible for an individual to receive a deep-dose equivalent in excess of 0.1 rem (1 mSv) in 1 hour; and

(ii) Conspicuous visible and audible alarm signals are generated to make an individual attempting to enter the area aware of the hazard and at least one other authorized individual, who is physically present, familiar with the

١

<sup>2</sup> This section applies to radiation from byproduct, source, or special nuclear materials that are used in sealed sources in non-self-shielded irradiators. This section does not apply to radioactive sources that are used in teletherapy, in radiography, or in completely self-shielded irradiators in which the source is both stored and operated within the same shielding radiation barrier and, in the designed configuration of the irradiator, is always physically inaccessible to any individual and cannot create high levels of radiation in an area that is accessible to any individual. This section also does not apply to sources from which the radiation is incidental to some other use or to nuclear reactorgenerated radiation.

activity, and prepared to render or summon assistance, aware of the failure of the entry control devicea.

(3) The licensee shall provide control devices so that, upon failure or removal of physical radiation barriers other than the source's shielded storage container--

(i) The radiation level from the source is reduced below that at which it would be possible for an individual to receive a deep-dose equivalent in excess of 0.1 rem (1 mSv) in 1 hour; and

(ii) Conspicuous visible and audible alarm signals are generated to make potentially affected individuals aware of the hazard and the licensee or at least one other individual, who is familiar with the activity and prepared to render or summon assistance, aware of the failure or renainal of the physical barrier.

(4) When the shield for the stored source is a liquid, the licensee shall provide means to monitor the integrity of the shield and to signal, automatically, loss of adequate shielding.

(5) Physical radiation barriers that comprise permanent structural components, such as walls, that have no credible probability of failure or removal in ordinary circumstances need not meet the requirements of paragraphs (a)(3) and
 (4) of this section.

(6) Each area must be equipped with devices that will automatically generate conspicuous visible and audible alarm signals to alert personnel in the area before the source can be put into operation and in sufficient time for any individual in the area to operate a clearly identified control device, which must be installed in the area and which can prevent the source from being put into operation.

(7) Each area must be controlled by use of such administrative procedures and such devices as are necessary to ensure that the area is cleared of personnel prior to each use of the source.

(8) Each area must be checked by a radiation measurement to ensure that, prior to the first individual's entry into the area after any use of the source, the radiation level from the source in the area is below that at which it would be possible for an individual to receive a deep-dose equivalent in excess of 0.1 rem (1 mSv) in 1 hour. (9) The entry control devices required in paragraph (a)(1) of this section must have been tested for proper functioning (see § 20.1109 for recordkeeping requirements).

(i) Testing must be conducted prior to initial operation with the source of radiation on any day (unless operations were continued uninterrupted from the previous day); and

(ii) Testing must be conducted prior to resumption of operation of the source of radiation after any unintended interruption; and

(iii) The licensee shall submit and adhere to a schedule for periodic tests of the entry control and warning systems.

(10) The licensee may not conduct operations, other than those necessary to place the source in safe condition or to effect repairs on controls, unless control devices are functioning properly.

(11) Entry and exit portais that are used in transporting materials to and from the irradiation area, and that are not intended for use by individuals, must be controlled by such devices and administrative procedures as are necessary to physically protect and warn against inadvertent entry by any individual through these portals. Exit portals for processed materials must be equipped to detect and signal the presence of any loose radiation sources that are carried toward such an exit and to automatically prevent loose radiation sources from being carried out of the area.

(b) Persons holding licenses or applicants for licenses for radiation sources that are within the purview of paragraph (a) of this section and that will be used in a variety of positions or in locations, such as open fields or forests, that make it impracticable to comply with certain requirements of paragraph (a) of this section, such as those for the automatic control of radiation levels, may apply to the Director, Office of Nuclear Material Safety and Safeguards, U.S. Nuclear Regulatory Commission, Washington, DC 20555, for approval of the use of alternative safety measures. Any alternative safety measures must provide a degree of personnel protection at least equivalent to those specified in paragraph (a) of this section. At least one of the alternative measures must include an entry-preventing interlock control based on a measurement of the radiation that ensures the absence of high radiation levels before an individual can gain access to the area where such radiation sources are used. (c) The entry control devices required by paragraphs (a) and (b) of this section must be established in such a way that no individual will be prevented from leaving the area.

## SUBPART H--RESPIRATORY PROTECTION AND CONTROLS TO RESTRICT INTERNAL EXPOSURE IN RESTRICTED AREAS

## § 20.701 Use of process or other engineering contr. 1s.

The licensee shall use, to the extent practicable, process or other engineering controls (e.g., containment or ventilation) to control the concentrations of radioactive material in air.

## § 20.702 Use of other controls.

When it is not practicable to apply process or other engineering controls to control the concentrations of radioactive material in air to values below those that define an airborne radioactivity area, the licensee shall, consistent with maintaining the total effective dose equivalent ALARA, increase monitoring, and limit intakes by one or more of the following means:

- (a) Control of access;
- (b) Limitation of exposure times;
- (c) Use of respiratory protection equipment; or
- (d) Other controls.

# § 20.703 Use of individual respiratory protection equipment.

(a) If the licensee uses respiratory protection equipment to limit intakes pursuant to § 20.702--

(1) The licensee shall use only respiratory protection equipment that is tested and certified or had certification extended by the National Institute for Occupational Safety and Health/Mine Safety and Health Administration (NIOSH/MSHA).

(2) If the licensee wishes to use equipment that has not been tested or certified by NIOSH/MSHA, has not had certification extended by NIOSH/MSHA, or for which there is no schedule for testing or certification, the licensee shall submit an application for authorized use of that equipment, including a demonstration by testing, or a demonstration on the basis of reliable test information, that the material and performance characteristics of the equipment are capable of providing the proposed degree of protection under anticipated conditions of use.

(3) The licensee shall implement and maintain a respiratory protection program that includes--

(i) Air sampling sufficient to identify the potential hazard, permit proper equipment selection, and estimate exposures;

(ii) Surveys and bioassays, as appropriate, to evaluate actual intakes;

(iii) Testing of respirators for operability immediately prior to each use;

(iv) Written procedures regarding selection, fitting, issuance, maintenance, and testing of respirators, including testing for operability immediately prior to each use; supervision and training of personnel; monitoring, including air sampling and bioassays; and recordkeeping; and

(v) Determination by a hysician prior to initial fitting of respirators, and at least every 12 most's thereafter, that the individual user is physically able to use the respiratory protection equipment.

(4) The licensee shall issue a written policy statement on respirator usage covering--

(i) The use of process or other engineering controls, instead of respirators;

(ii) The routine, nonroutine, and emergency use of respirators; and

(iii) The periods of respirator use and relief from respirator use.

(5) The licensee shall advise each respirator user that the user may leave the area at any time for relief from respirator use in the event of equipment malfunction, physical or psychological distress, procedural or communication failure, significant deterioration of operating conditions, or any other conditions that might require such relief. (6) The licensee shall use equipment within limitations for type and mode of use and shall provide proper visual, communication, and other special capabilities (such as adequate skin protection) when needed.

(b) In estimating exposure of individuals to airborne radioactive materials, the licensee may make allowance for respiratory protection equipment used to limit intakes pursuant to § 20.702, provided that the following conditions, in addition to those in § 20.703(a), are satisfied:

(1) The licensee selects respiratory protection equipment that provides a protection factor (see Appendix A) greater than the multiple by which peak concentrations of airborne radioactive materials in the working area are expected to exceed the values specified in Appendix B, Table 1, Column 3. If the selection of a respiratory protection device with a protection factor greater than the peak concentration is inconsistent with the goal specified in § 20.702 of keeping the total effective dose equivalent ALARA, the licensee may select respiratory protection equipment with a lower protection factor only if such a selection would result in keeping the total effective dose equivalent that is inhaled when respirators are worn may be initially estimated by dividing the average concentration in air, during each period of uninterrupted use, by the protection factor. If the exposure is later found to be greater than estimated, the corrected value must be used; if the exposure is later found to be less than estimated, the corrected value may be used.

(2) The licensee shall obtain authorization from the Commission before assigning respiratory protection factors in excess of those specified in Appendix A. The Commission may authorize a licensee to use higher protection factors on receipt of an application that --

(i) Describes the situation for which a need exists for higher protection factors, and

(ii) Demonstrates that the respiratory protection equipment provides these higher protection factors under the proposed conditions of use.

(c) The licensee shall use as emergency devices only respiratory protection equipment that has been specifically certified or had certification extended for emergency use by NIOSH/MSHA. (d) The licensee shall notify, in writing, the Director of the appropriate NRC Regional Office listed in Appendix D at least 30 days before the date that respiratory protection equipment is first used under the provisions of either § 20.703(a) or (b).

# § 20.704 Further restrictions on the use of respiratory protection equipment.

The Commission may impose restrictions in addition to those in §§ 20.702, 20.703, and Appendix A to--

 (a) Ensure that the respiratory protection program of the licensee is adequate to limit exposures of individuals to airborne radioactive materials; and

(b) Limit the extent to which a licensee may use respiratory protection equipment instead of process or other engineering controls.

#### SUBPART I--STORAGE AND CONTROL OF LICENSED MATERIAL

#### § 20.801 Security of stored material.

The licensee shall secure from unauthorized removal or access licensed materials that are stored in controlled or unrestricted areas.

## § 20.802 Control of material not in storage.

The licensee shall control and maintain constant surveillance of licensed material that is in a controlled or unrestricted area and that is not in storage.

#### SUBPART J -- PRECAUTIONARY PROCEDURES

§ 20.901 Caution signs.

(a) <u>Standard radiation symbol</u>. Unless otherwise authorized by the Commission, the symbol prescribed by this part shall use the colors magenta, or purple, or black on yellow background. The symbol prescribed by this part is the three-bladed design:



RADIATION SYMBOL

(1) Cross-hatched area is to be magenta, or purple, or black, and

(2) The background is to be yellow.

(b) Exception to color requirements for standard radiation symbol. Notwithstanding the requirements of paragraph (a) of this section, licensees are authorized to label sources, source holders, or device components containing sources of licensed materials that are subjected to high temperatures, with conspicuously etched or stamped radiation caution symbols and without a color requirement.

(c) Additional information on signs and labels. In addition to the contents of signs and labels prescribed in this part, the licensee may provide, on or near the required signs and labels, additional information, as appropriate, to make individuals aware of potential radiation exposures and to minimize the exposures.

## § 20.902 Posting requirements.

(a) Posting of radiation areas. The licensee shall post each radiation area with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, RADIATION AREA."

(b) <u>Posting of high radiation areas</u>. The licensee shall post each high radiation area with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, HIGH RADIATION AREA" or "DANGER, HIGH RADIATION AREA."

(c) Posting of very high radiation areas. The licensee shall post each very high radiation area with a conspicuous sign or signs bearing the radiation symbol and words "GRAVE DANGER, VERY HIGH RADIATION AREA."

(d) <u>Posting of airborne radioactivity areas</u>. The licensee shall post each airborne radioactivity area with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, AIRBORNE RADIOACTIVITY AREA" or "DANGER, AIRBORNE RADIOACTIVITY AREA."

(e) <u>Posting of areas or rooms in which licensed material is used</u> <u>or stored</u>. The licensee shall post each area or room in which there is used or stored an amount of licensed material exceeding 10 times the quantity of such material specified in Appendix C with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, RADIOACTIVE MATERIAL(S)" or "DANGER, RADIOACTIVE MATERIAL(S)."

#### § 20.903 Exceptions to posting requirements.

(a) A licensee is not required to post caution signs in areas or rooms containing radioactive materials for periods of less than 8 hours, if each of the following conditions is met:

(1) The materials are constantly attended during these periods by an individual who takes the precautions necessary to prevent the exposure of individuals to radiation or radioactive materials in excess of the limits established in this part; and

(2) The area or room is subject to the licensee's control.

(b) Rooms or other areas in hospitals that are occupied by patients are not required to be posted with caution signs pursuant to § 20.902 provided that--

(1) The patient is being treated with sealed sources or has been treated with unsealed radioactive material in quantities less than 30 millicuries, or the measured dose rate at 1 meter from the patient is less than 0.005 rem (0.05 mSv) per hour; and

(2) There are personnel in attendance who will take the necessary precautions to prevent the exposure of individuals to radiation or radioactive material in excess of the limits established in this part and to operate within the ALARA provisions of the licensee's radiation protection program.

(c) A room or area is not required to be posted with a caution sign because of the presence of a sealed source provided the radiation level at 30 centimeters from the surface of the source container or housing does not exceed 0.005 rem (0.05 mSv) per hour.

§ 20.904 Labeling containers.

(a) The licensee shall ensure that each container of licensed material bears a durable, clearly visible label bearing the radiation symbol and the words "CAUTION, RADIOACTIVE MATERIAL" or "DANGER, RADIOACTIVE MATERIAL." The label must also provide sufficient information (such as the radionuclide(s) present, an estimate of the quantity of radioactivity, the date for which the

activity is estimated, radiation levels, kinds of materials, and mass enrichment) to permit individuals handling or using the containers, or working in the vicinity of the containers, to take precautions to avoid or minimize exposures.

(b) Each licensee shall, prior to removal or disposal of empty uncontaminated containers to unrestricted areas, remove or deface the radioactive material label or otherwise clearly indicate that the container no longer contains radioactive materials.

§ 20.905 Exemptions to labeling requirements.

A licensee is not required to label ---

(a) Containers holding licensed material in quantities less than the quantities listed in Appendix C; or

(b) Containers holding licensed material in concentrations less than those specified in Table 3 of Appendix B; or

(c) Containers attended by an individual who takes the precautions necessary to prevent the exposure of individuals in excess of the limits established by this part; or

(d) Containers when they are in transport and packaged and labeled in accordance with the regulations of the Department of Transportation,  $^{3}$  or

(e) Containers that are accessible only to individuals authorized to handle or use them, or to work in the vicinity of the containers, if the contents are identified to these individuals by a readily available written record (examples of containers of this type are containers in locations such as water-filled canals, storage vaults, or hot cells). The record must be retained as long as the containers are in use for the purpose indicated on the record; or

<sup>3</sup> Labeling of packages containing radioactive materials is required by the Department of Transportation (DOT) if the amount and type of radioactive material exceeds the limits for an excepted quantity or article as defined and limited by DOT regulations 49 CFR 173.403(m) and (w) and 173.421-424.

(f) Installed manufacturing or process equipment, such as reactor components, piping, and tanks.

§ 20.906 Procedures for receiving and opening packages.

(a) Each licensee who expects to receive a package containing quantities of radioactive material in excess of a Type A quantity, as defined in § 71.4 and Appendix 7 to Part 71 of this chapter, shall make arrangements to receive==

(1) The package when the carrier offers it for delivery; or

(2) Notification of the arrival of the package at the carrier's terminal and to take possession of the package expeditiously.

(b) Each licensee shall monitor the external surfaces of a package known to contain radioactive material for radioactive contamination and radiation levels if the package--

(1) Is labeled as containing radioactive material; or

(2) Has evidence of potential contamination, such as packages that are crushed, wet, or damaged.

(c) The licensee shall perform the monitoring required by paragraph (b) of this section as soon as practicable after receipt of the package, but not later than 3 hours after the package is received at the licensee's facility if it is received during the licensee's normal working hours, or not later than 3 hours from the beginning of the next working day if it is received after working hours.

(d) The licensee shall immediately notify the final delivery carrier and, by telephone and telegram, mailgram, or facsimile, the Administrator of the appropriate NRC Regional Office listed in Appendix D when=-

(1) Removable radioactive surface contamination exceeds the limits of § 71.87(i) of this chapter; or

(2) External radiation levels exceed the limits of § 71.47 of this chapter.

(e) Each licensee shall--

(1) Establish, maintain, and retain written procedures for safely opening packages in which radioactive material is received; and

(2) Ensure that the procedures are followed and that due consideration is given to special instructions for the type of package being opened.

10

(f) Licensees transferring special form sources in licensee-owned or licensee-operated vehicles to and from a work site are exempt from the contamination monitoring requirements of paragraph (b), but are not exempt from the survey requirement in paragraph (b) for measuring radiation levels that is required to ensure that the source is still properly lodged in its shield.

#### SUBPART K--WASTE DISPOSAL

§ 20.1001 General requirements.

(a) A licensee shall dispose of licensed material only--

(1) By transfer to an authorized recipient as provided in § 20.1006 or in the regulations in Parts 30, 40, 60, 61, 70, or 72 of this chapter; or

(2) By decay in storage; or

(3) By release in effluents within the limits in § 20.301; or

(4) As authorized under §§ 20.1002, 20.1003, 20.1004, or 20.1005.

(b) A person must be specifically licensed to receive waste containing licensed material from other persons for:

(1) Treatment prior to disposal; or

(2) Treatment or disposal by incineration; or

(3) Decay in storage; or

(4) Disposal at a land disposal facility licensed under Part 61 of this chapter; or

(5) Disposal at a geologic repusitory under Part 60 of this chapter.

#### § 20.1002 Method for obtaining approval of proposed disposal procedures.

A licensee or applicant for a license may apply to the Commission for approval of proposed procedures, not otherwise authorized in the regulations in this chapter, to dispose of licensed material generated in the licensee's activities. Each application shall include:

(a) A description of the waste containing licensed material to be disposed of, including the physical and chemical properties important to risk evaluation, and the proposed manner and conditions of waste disposal; and

(b) An analysis and evaluation of pertinent information on the nature of the environment; and

(c) The nature and location of other potentially affected licensed and unlicensed facilities; and

(d) Analyses and procedures to ensure that doses are maintained ALARA and within the dose limits in this part.

§ 20.1003 Disposal by release into sanitary sewerage.

(a) A licensee may discharge licensed material into sanitary sewerage if each of the following conditions is satisfied:

 The material is readily soluble (or is readily dispersible biological material) in water; and

(2) The quantity of licensed or other radioactive material that the licensee releases into the sewer in 1 month divided by the average monthly volume of water released into the sewer by the licensee does not exceed the concentration listed in Table 3 of Appendix B; and

(3) If more than one radionuclide is released, the following conditions must also be satisfied:

(i) The licensee shall determine the fraction of the limit in Table 3 of Appendix B represented by discharges into sanitary sewerage by dividing the actual monthly average concentration of each radionuclide released by the licensee into the sewer by the concentration of that radionuclide listed in Table 3 of Appendix B; and

(ii) The sum of the fractions for each radionuclide required by paragraph(a)(3)(i) of this section does not exceed unity; and

(4) The total quantity of licensed and other radioactive material that the licensee releases into the sanitary sewerage system in a year does not exceed 5 curies (185 GBq) of hydrogen=3, 1 curie (37 GBq) of carbon=14, and 1 curie (37 GBq) of all other radioactive materials combined.

(b) Excreta from individuals undergoing medical diagnosis or therapy with radioactive material is not subject to the limitations contained in paragraph(a) of this section.

#### § 20.1004 Treatment or disposal by incineration.

A licensee may treat or dispose of licensed material by incineration only in the amounts and forms specified in § 20.1005 or as specifically approved by the Commission pursuant to § 20.1002.

#### § 20.1005 Disposal of specific wastes.

(a) A licensee may dispose of the following licensed material as if it were not radioactive:

(1) 0.05 microcurie (1.85 kBq), or less, of hydrogen-3 or carbon-14 per gram of medium used for liquid scintillation counting; and

(2) 0.05 microcurie (1.85 kBq), or less, of hydrogen=3 or carbon=14 per gram of animal tissue, averaged over the weight of the entire animal.

(b) A licensee may not dispose of tissue under paragraph (a)(2) of this section in a manner that would permit its use either as food for humans or as animal feed.

(c) The licensee shall maintain records in accordance with § 20.1108.

#### § 20.1006 Transfer for disposal and manifests.

(a) The requirements of this section and Appendix F are designed to control transfers of low-level radioactive waste intended for disposal at a land disposal facility (as defined in Part 61 of this chapter), establish a manifest tracking system, and supplement existing requirements concerning transfers and recordkeeping for those wastes.

(b) Each shipment of radioactive waste intended for disposal at a licensed land disposal facility must be accompanied by a shipment manifest as specified in Section I of Appendix F.

(c) Each shipment manifest must include a certification by the waste generator as specified in Section II of Appendix F.

(d) Each person involved in the transfer for disposal and disposal of waste, including the waste generator, waste collector, waste processor, and disposal facility operator, shall comply with the requirements specified in Section III of Appendix F.

#### § 20.1007 Compliance with environmental and health protection regulations.

Nothing in this subpart relieves the licensee from complying with other applicable Federal, State, and local regulations governing any other toxic or hazardous properties of materials that may be disposed of under this subpart.

#### SUBPART L--RECORDS

#### § 20.1101 General provisions.

(a) Each licensee shall use the units: curie, rad, rem, including multiples and subdivisions, and shall clearly indicate the units of all quantities on records required by this part.

(b) The licensee shall make a clear distinction among the quantities entered on the records required by this part (e.g., total effective dose equivalent, shallow-dose equivalent, eye dose equivalent, deep-dose equivalent, committed effective dose equivalent).

#### § 20.1102 Records of radiation protection programs.

(a) Each licensee shall maintain records of the radiation protection program, including:

(1) The provisions of the program; and

(2) Audits and other reviews of program content and implementation.

(b) The licensee shall retain the records required by paragraph (a)(1) of this section until the Commission terminates each pertinent license requiring the record. The licensee shall retain the records required by paragraph (a)(2) of this section for 3 years after the record is made.

#### § 20.1103 Records of surveys.

(a) Each licensee shall maintain records showing the results of surveys and calibrations required by §§ 20.501 and 20.906(b). The licensee shall retain these records for 3 years after the record is made. (b) The licensee shall retain each of the following records until the Commission terminates each pertinent license requiring the record:

(1) Records of the results of surveys to determine the dose from external sources and used, in the absence of or in combination with individual monitoring data, in the assessment of individual dose equivalents; and

(2) Records of the results of measurements and calculations used to determine individual intakes of radioactive material and used in the assessment of internal dose; and

(3) Records showing the results of air sampling, surveys, and bioassays required pursuant to § 20.703(a)(3)(i) and (ii); and

(4) Records of the results of measurements and calculations used to evaluate the release of radioactive effluents to the environment.

#### § 20.1104 Determination of prior occupational dose.

(a) For each individual who may enter the licensee's restricted or controlled area and is likely to receive, in a year, an occupational dose requiring monitoring pursuant to § 20.502, the licensee shall--

 Determine the occupational radiation dose received during the current year; and

(2) Attempt to obtain the records of lifetime cumulative occupational radiation dose.

(b) Prior to permitting an individual to participate in a planned special exposure, the licensee shall determine--

 The internal and external doses from all previous planned special exposures; and

(2) All doses in excess of the limits (including doses received during accidents and emergencies) received during the lifetime of the individual.

(c) In complying with the requirements of paragraph (a) of this section, a licensee may --

(1) Accept, as a record of the occupational dose that the individual received during the current year, a written signed statement from the individual, or from the individual's most recent employer for work involving radiation exposure, that discloses ... nature and the amount of any occupationa? dose that the individual may have received during the current year; (2) Accept, as the record of lifetime cumulative radiation dose, an up-to-date NRC Form 4, or equivalent, signed by the individual and countersigned by an appropriate official of the most recent employer for work involving radiation exposure, or the individual's current employer (if the individual is not employed by the licensee); and

(3) Obtain reports of the individual's dose equivalent(s) from the most recent employer for work involving radiation exposure, or the individual's current employer (if the individual is not employed by the licensee) by telephone, telegram, electronic media, or letter. The licensee shall request a written verification of the dose data if the authenticity of the transmitted report cannot be established.

(d) The licensee shall record the exposure history, as required by paragraph (a) of this section, on NRC Form 4, or other cleat and legible record, of all the information required on that form.<sup>4</sup> The form or record must show each period in which the individual received occupational exposure to radiation or radioactive material and must be signed by the individual who received the exposure. For each period for which the licensee obtains reports, the licensee shall use the dose shown in the report in preparing NRC Form 4. For any period in which the licensee does not obtain a report, the licensee shall place a notation on NRC Form 4 indicating the periods of time for which data are not available.

(e) If the licensee is unable to obtain a complete record of an individual's current and previously accumulated occupational dose, the licensee shall assume---

<sup>4</sup> Licensees are not required to reevaluate the separate external dose equivalents and internal committed dose equivalents or intakes of radionuclides assessed under the regulations in this part in effect before January 1, 1993. Further, occupational exposure histories obtained and recorded on NRC Form 4 before January 1, 1993, would not have included effective dose equivalent, but may be used in the absence of specific information on the intake of radionuclides by the individual.

(1) In establishing administrative controls under § 20.201(f) for the current year, that the allowable dose limit for the individual is reduced by 1.25 rems (12.5 mSv) for each quarter for which records were unavailable and the individual was engaged in activities that could have resulted in occupational radiation exposure; and

(2) That the individual is not available for planned special exposures.

(f) The licensee shall retain the records on NRC Form 4 or equivalent until the Commission terminates each pertinent license requiring this record. The licensee shall retain records used in preparing NRC Form 4 for 3 years after the record is made.

#### § 20.1105 Records of planned special exposures.

(a) For each use of the provisions of § 20.206 for planned special exposures, the licensee shall maintain records that describe--

 The exceptional circumstances requiring the use of a planned special exposure; and

(2) The name of the management official who authorized the planned special exposure and a copy of the signed authorization; and

(3) What actions were necessary; and

(4) Why the actions were necessary; and

(5) How doses were maintained ALARA; and

(6) What individual and collective doses were expected to result, and the doses actually received in the planned special exposure.

(b) The licensee shall retain the records until the Commission terminates each pertinent license requiring these records.

#### § 20.1106 Records of individual monitoring results.

 (a) <u>Recordkeeping requirement</u>. Each licensee shall maintain records of doses received by all individuals for whom monitoring was required pursuant to § 20.502, and records of doses received during planned special exposures, accidents, and emergency conditions. These records<sup>5</sup> must include, when applicable--

(1) The deep-dose equivalent to the whole body, eye dose equivalent, shallow-dose equivalent to the skin, and shallow-dose equivalent to the extremities; and

(2) The estimated intake or body burden of radionuclides (see § 20.202); and

(3) The committed effective dose equivalent assigned to the intake or body burden of radionuclides; and

(4) The specific information used to calculate the committed effective dose equivalent pursuant to § 20.204(c); and

(5) The total effective dose equivalent when required by § 20.202; and

(6) The total of the deep-dose equivalent and the committed dose to the organ receiving the highest total dose.

(b) <u>Recordkeeping frequency</u>. The licensee shall make entries of the records specified in paragraph (a) of this section at least annually.

(c) <u>Recordkeeping format</u>. The licensee shall maintain the records specified in paragraph (a) of this section on NRC Form 5, in accordance with the instructions for NRC Form 5, or in clear and legible records containing all the information required by NRC Form 5.

(d) <u>Privacy protection</u>. The records required under this section should be protected from public disclosure because of their personal privacy nature. These records are protected by most State privacy laws and, when transferred to the NRC, are protected by the Privacy Act of 1974, Pub.L. 93-579, 5 U.S.C. 552a, and the Commission's regulations in 10 CFR Part 9.

(e) The licensee shall maintain the records of dose to an embryo/fetus with the records of dose to the declared pregnant woman. The declaration of pregnancy shall also be kept on file, but may be maintained separately from the dose records.

(f) The licensee shall retain each required form or record until the Commission terminates each pertinent license requiring the record.

5 Assessments of dose equivalent and records made using units in effect before January 1, 1993, need not be changed.

#### § 20.1107 Records of dose to individual members of the public.

 (a) Each licensee shall maintain records sufficient to demonstrate compliance with the dose limit for individual members of the public (see § 20.301).

(b) The licensee shall retain the records required by paragraph (a) of this section until the Commission terminates each pertinent license requiring the record.

#### § 20.1108 Records of waste disposal.

(a) Each licensee shall maintain records of the disposal of licensed materials made under §§ 20.1002, 20.1003, 20.1004, 20.1005, Part 61, and disposal by burial in soil, including burials authorized before January 28, 1981.

(b) The licensee shall retain the records required by paragraph (a) of this section until the Commission terminates each pertinent license requiring the record.

## § 20.1109 <u>Records of testing entry control devices for very high</u> radiation areas.

(a) Each licensee shall maintain records of tests made under § 20.603(a)(9) on entry control devices for very high radiation areas. Times records must include the date, time, and results of each such test of function.

(b) The licensee shall retain the records required by paragraph (a) of this section for 3 years after the record is made.

<sup>6</sup> A previous § 20.304 permitted burial of small quantities of licensed materials in soil before January 28, 1981, without specific Commission authorization.

#### § 20.1110 Form of records.

10

00

Each record required by this part must be legible throughout the specified retention period. The record may be the original or a reproduced copy or a microform provided that the copy or microform is authenticated by authorized personnel and that the microform is capable of producing a clear copy throughout the required retention period. The record may also be stored in electronic media with the capability for producing legible, accurate, and complete records during the required retention period. Records, such as letters, drawings, and specifications, must include all pertinent information, such as stands, initials, and signatures. The licensee shall maintain adequate safeguards against tampering with and loss of records.

#### SUBPART M--REPORTS

## § 20.1201 Reports of thef: or loss of licensed material.

(a) Telephone reports.

(1) Each licensee shall report by tel. hone as follo -:

(i) Immediately after its occurrence becomes known to the licensee, any lost, stolen, or missing licensed material in an aggregate quantity equal to or greater than 1,000 times the quantity specified in Appendix C under such circumstances that it appears to the licensee that an exposure could result to persons in unrestricted areas; or

(ii) Within 30 days after the occurrence of any lost, stolen, or missing licensed material becomes known to the licensee, all licensed material in a quantity greater than 10 times the quantity specified in Appendix C that is still missing at this time.

(2) Reports must be made as follows:

(i) Licensees having an installed Emergency Notification System shall make the reports to the NKC Operations Center in accordance with § 50.72 of this chapter, and

(ii) All other licensees shall make reports to the NRC Operations Center.

(b) Written reports.

(1) Each licensee required to make a report under paragraph (a) of this section shall, within 30 days after making the telephone report, make a written report setting forth the following information:

(i) A description of the licensed material involved, including kind, quantity, and chemical and physical form; and

(ii) A description of the circumstances under which the loss or theft occurred; and

(iii) A statement of disposition, or probable disposition, of the licensed material involved; and

(iv) Exposures of individuals to radiation, circumstances under which the exposures occurred, and the possible effective dose equivalent to persons in unrestricted areas; and

(v) Actions that have been taken, or will be taken, to recover the material; and

(vi) Procedures or measures that have been, or will be, adopted to ensure against a recurrence of the loss or theft of lilensed material.

(2) Reports must be made as follows:

Ø 1

(i) For holders of an operating license for a nuclear power plant, the events included in paragraph (b) of this section must be reported in accordance with the procedures described in § 50.73(b), (c), (d), (e), and (g) of this chapter and must include the information required in paragraph (b)(1) of this section, and

(ii) All other licensees shall make reports to the Administrator of the appropriate NRC Regional Office listed in Appendix D.

(c) A duplicate report is not required under (b) if the licensee is also required to submit a report pursuant to \$\$ 30.55(c), 40.64(c), 50.72, 50.73. 70.52, 73.27(b), 73.67(e)(3)(vi), 73.67(g)(3)(iii), 73.71, or 150.19(c) of chapter.

(d) Subsequent to filing the written report, the licensee shall also report any additional substantive information on the loss or theft within 30 days after the licensee learns of such information.

(e) The licensee shall prepare any report filed with the Commission pursuant to this section so that names of individuals who may have received exposure to radiation are stated in a separate and detachable part of the report.

#### § 20.1202 Notification of incidents.

(a) <u>Immediate notification</u>. Notwithstanding any other requirements for notification, each licensee shall immediately report any event involving byproduct, source, or special nuclear material possessed by the licensee that may have caused or threatens to cause any of the following conditions--

(1) An individual to receive--

(i) A total effective dose equivalent of 25 rems (0.25 Sv) or more; or

(ii) An eye dose equivalent of 75 rems (0.75 Sv) or more; or

(iii) A shallow-dose equivalent to the skin or extremities of 250 rads (2.5 Gy) or more; or

(2) The release of radioactive material, inside or outside of a restricted area, so that, had an individual been present for 24 hours, the individual could have received an intake five times the occupational annual limit on intake (the provisions of this paragraph do not apply to locations where personnel are not normally stationed during routine operations, such as hot-cells or process enclosures); or

(3) A loss of 1 working week or more of the operation of any facilities affected; or

(4) Damage to property in excess of \$200,000.

(b) <u>Twenty-four hour notification</u>. Each licensee shall, within 24 hours of discovery of the event, report any event involving loss of control of licensed material possessed by the licensee that may have caused, or threatens to cause, any of the following conditions:

(1) An individual to receive, in a period of 24 hours--

(i) A total effective dose equivalent exceeding 5 rems (0.05 Sv); or

(ii) An eye dose equivalent exceeding 15 rems (0.15 Sv); or

(iii) A shallow-dose equivalent to the skin or extremities exceeding 50 rems (0.5 Sv); or

(2) The release of radioactive material, inside or outside of a restricted area, so that, had an individual been present for 24 hours, the individual could have received an intake in excess of one occupational annual limit on intake (the provisions of this paragraph do not apply to locations where personnel are not normally stationed during routine operations, such as hot-cells or process enclosures); or (3) A loss of 1 day or more of the operation of any facilities affected; or

(4) Damage to property in excess of \$2,000.

(c) The licensee shall prepare any report filed with the Commission pursuant to this section so that names of individuals who have received exposure to radiation or radioactive material are stated in a separate and detachable part of the report.

(d) Reports made by licensees in response to the requirements of this section must be made as follows:

(1) Licensees having an installed Emergency Notification System shall make the reports required by paragraphs (a) and (b) of this section to the NRC Operations Center in accordance with § 50.72; and

(2) All other licensees shall make the reports required by paragraphs (a) and (b) of this section by telephone to the NRC Operations Center and by telegram, mailgram, or facsimile to the Administrator of the appropriate NRC Regional Office listed in Appendix D.

(e) The provisions of this section do not include doses that result from planned special exposures, that are within the limits for planned special exposures, and that are reported under § 20.1204.

## § 20.1203 <u>Reports of exposures, radiation levels, and concentrations of</u> radioactive material exceeding the limits.

(a) <u>Reportable events</u>. In addition to the notification required by
 § 20.1202, each licensee shall submit a written report within 30 days after
 learning of any of the following occurrences:

(1) Any incident for which notification is required by § 20.1202; or

(2) Doses in excess of any of the following:

(i) The occupational dose limits for adults in § 20.201; or

(ii) The occupational dose limits for a minor in § 20.207; or

(iii) The limits for an embryo/fetus of a declared pregnant woman in § 20.208; or

(iv) The limits for an individual member of the public in § 20.301; or

(v) Any applicable limit in the license; or

(3) Levels of radiation or concentrations of radioactive material in--

(i) A restricted area in excess of any applicable limit in the license; or

(ii) An unrestricted area in excess of 10 times any applicable limit set forth in this part or in the license (whether or not involving exposure of any individual in excess of the limits in § 20.301); or

(4) For licensees subject to the provisions of EPA's generally applicable environmental radiation standards, levels of radiation or releases of radioactive material in excess of those standards, or of license conditions related to those standards.

(b) Contents of reports.

(1) Each report required by paragraph (a) of this section must describe the extent of exposure of individuals to radiation and radioactive material, including, as appropriate:

(i) Estimates of each individual's dose; and

(ii) The levels of radiation and concentrations of radioactive material involved; and

(iii) The cause of the elevated exposures, dose rates, or concentrations; and

(iv) Corrective steps taken or planned to ensure against a recurrence, including the schedule for achieving conformance with applicable limits, generally applicable environmental standards, and associated license conditions.

(2) Each report filed pursuant to paragraph (a) of this section must include for each individual<sup>7</sup> exposed: the name, Social Security account number, and date of birth. The report must be prepared so that this information is stated in a separate and detachable part of the report.

(c) For holders of an operating license for a nuclear power plant, the occurrences included in paragraph (a) of this section must be reported in accordance with the procedures described in § 50.73(b), (c), (d), (e), and (g) of this chapter and must also include the information required by paragraph (b) of this section. Occurrences reported in accordance with § 50.73 of this chapter need not be reported by a duplicate report under paragraph (a) of this section.

7

With respect to the limit for the embryo/fetus (§ 20.208), the identifiers should be those of the declared pregnant woman.

(d) All licensees, other than those holding an operating license for a nuclear power plant, who make reports under paragraph (a) of this section shall submit the report in writing to the U.S. Nuclear Regulatory Commission, Document Control Desk, Washington, DC 20555, with a copy to the appropriate NRC Regional Office listed in Appendix D.

## § 20.1204 Reports of planned special exposures.

The licensee shall submit a written report to the Administrator of the appropriate NRC Regional Office listed in Appendix D within 30 days following any planned special exposure conducted in accordance with § 20.206, informing the Commission that a planned special exposure was conducted and indicating the date the planned special exposure occurred and the information required by § 20.1105.

## § 20.1205 [Reserved].

## § 20.1206 Reports of individual monitoring.

(a) This section applies to each person licensed by the Commission to--

(1) Operate a nuclear reactor designed to produce electrical or heat energy pursuant to § 50.21(b) or § 50.22 of this chapter or a testing facility as defined in § 50.2 of this chapter; or

(2) Possess or use byproduct material for purposes of radiography pursuant to Parts 30 and 34 of this chapter; or

(3) Possess or use at any one time, for purposes of fuel processing, fabricating, or reprocessing, special nuclear material in a quantity exceeding 5,000 grams of contained uranium-235, uranium-233, or plutonium, or any combination thereof pursuant to Part 70 of this chapter; or

(4) Possess high-level radioactive waste at a geologic repository operations area pursuant to Part 60 of this chapter; or

(5) Possess spent fuel in an independent spent fuel storage installation (ISFSI) pursuant to Part 72 of this chapter; or

(6) Receive radioactive waste from other persons for disposal under Part 61 of this chapter; or

(7) Possess or use at any time, for processing or manufacturing for distribution pursuant to Parts 30, 32, 33 or 35 of this chapter, byproduct material in quantities exceeding any one of the following quantities:

	Quantity of Radionuclide <sup>a</sup> in curies		
Cesium-137	1		
Cobalt-60 Gold-198	1100		
Iodine-131	1		
Krypton-85	1,000		
Promethium-147	10		
Technetium-99m	1,000		

<sup>a</sup>The Commission may require as a license condition, or by rule, regulation, or order pursuant to § 20.1302, reports from licensees who are licensed to use radionuclides not on this list, in quantities sufficient to cause comparable radiation levels.

(b) Each licensee in a category listed in paragraph (a) of this section shall submit an annual report of the results of individual monitoring carried out by the licensee for each individual for whom monitoring was required by § 20.502 during that year. The licensee may include additional data for individuals for whom monitoring was provided but not required. The licensee shall use Form NRC 5 or electronic media containing all the information required by Form NRC 5.

(c) The licensee shall file the report required by § 20.1206(b), covering the preceding year, on or before April 30 of each year. The licensee shall submit the report to the Director, Office of Nuclear Regulatory Research, Nuclear Regulatory Commission, Washington, DC 20555.

## SUBPART N--EXEMPTIONS AND ADDITIONAL REQUIREMENTS

#### § 20.1301 Applications for exemptions.

The Commission may, upon application by a licensee or upon its own initiative, grant an exemption from the requirements of the regulations in this part if it determines the exemption is authorized by law and would not result in undue hazard to life or property.

### § 20.1302 Additional requirements.

The Commission may, by rule, regulation, or order, impose requirements on a licensee, in addition to those established in the regulations in this part, as it deems appropriate or necessary to protect health or to minimize danger to life or property.

#### SUBPART O--ENFORCEMENT

#### § 20.1401 Violations.

(a) The Commission may obtain an injunction or other court order to prevent a violation of the provisions of--

(1) The Atomic Energy Act of 1954, as amended;

(2) Title II of the Energy Reorganization Act of 1974, as amended; or

(3) A regulation or order issued pursuant to those Acts.

(b) The Commission may obtain a court order for the payment of a civil penalty imposed under Section 234 of the Atomic Energy Act:

(1) For violations of --

(i) Sections 53, 57, 62, 63, 81, 82, 101, 103, 104, 107, or 109 of the Atomic Energy Act of 1954, as amended;

(ii) Section 206 of the Energy Reorganization Act;

(iii) Any rule, regulation, or order issued pursuant to the sections specified in paragraph (b)(1)(i) of this section;

(iv) Any term, condition, or limitation of any license issued under the sections specified in paragraph (b)(1)(i) of this section.

(2) For any violation for which a license may be revoked under Section 186 of the Atomic Energy Act of 1954, as amended.

(c) Any person who willfully violates a provision of the Atomic Energy Act or regulation or order issued under the requirements of that Act may be guilty of a crime and, upon conviction, be punished by fine or imprisonment or both, as provided by law.

# APPENDIX A

# PROTECTION FACTORS FOR RESPIRATORS<sup>a</sup>

		Protection Factors			Tested & Certified Equipment
	Description <sup>b</sup>	Modes <sup>C</sup>	Particu- lates only	Particu- lates, gases, & vapors <sup>e</sup>	National Institute for Occupational Safety and Health / Mine Safety and Health Administration tests for permissibility
Ι.	AIR-PURIFYING RESPIRATORS				
	Facepiece, half-mask <sup>g</sup> Facepiece, full Facepiece, half-mask full, or hood	NP NP PP	10 50 1000		30 CFR Part 11, Subpart K.
II.	ATMOSPHERE-SUPPLYING RESPIRATORS				
	1. Air-line respirator				
	Facepiece, half-mask Facepiece, half-mask Facepiece, full Facepiece, full Facepiece, full Hood Suit	CF DF DDF CF		1000 5 2000 5 2000 h i	30 CFR Part 11, Subpart J. J
	<ol> <li>Self-contained breathing apparatus (SCBA)</li> </ol>				
	Facepiece, full Facepiece, full Facepiece, full Facepiece, full	D PD RD RP		10,000 <sup>k</sup> 50 5,000 <sup>1</sup>	30 CFR Part 11, Subpart H.
III.	COMBINATION RESPIRATORS				
	Any combination of air- purifying and atmosphere- supplying respirators		Protection factor for type and mode of operation as listed above		30 CFR Part 11, §11.63(b).

The second states of

#### FOOTNOTES

- a. For use in the selection of respiratory protective devices to be used only where the contaminants have been identified and the concentrations (or possible concentrations) are known.
- b. Only for shaven faces and where nothing interferes with the seal of tight-fitting facepieces against the skin. (Hoods and suits are excepted.)
- c. The mode symbols are defined as follows:
  - CF = continuous flow
  - D = demand
  - NP = negative pressure (i.e., negative phase during inhalation)
  - PD = pressure demand (i.e., always positive pressure)
  - PP = positive pressure
  - RD = demand, recirculating (closed circuit)
  - RP = pressure demand, recirculating (closed circuit)
- d.1. The protection factor is a measure of the degree of protection afforded by a respirator, defined as the ratio of the concentration of airborne radioactive material outside the respiratory protective equipment to that inside the equipment (usually inside the facepiece) under conditions of use. It is applied to the ambient airborne concentration to estimate the concentrations inhaled by the wearer according to the following formula:

Concentration inhaled = Ambient airborne concentration Protection factor

- 2. The protection factors apply:
  - (a) Only for individuals trained in using respirators and wearing properly fitted respirators that are used and maintained under supervision in a well-planned respiratory protective program.
  - (b) For air-purifying respirators only when high efficiency particulate filters (above 99.97% removal efficiency by thermally generated 0.3 µm dioctyl phthalate (DOP) test or equivalent) are used in atmospheres not deficient in oxygen and not containing radioactive gas or vapor respiratory hazards.

- (c) No adjustment is to be made for the use of sorbents against radioactive material in the form of gases or vapors.
- (d) For atmosphere-supplying respirators only when supplied with adequate respirable air. Respirable air shall be provided of the quality and quantity required in accordance with NIOSH/MSHA certification (described in 30 CFR Part 11). Oxygen and air shall not be used in the same apparatus.
- e. Excluding radioactive contaminants that present an absorption or submersion hazard. For tritium oxide, approximately one-third of the intake occurs by absorption through the skin so that an overall protection factor of less than 2 is appropriate when atmospheresupplying respirators are used to protect against tritium oxide. If the protection factor for a device is 5 the effective protection factor for tritium is about 1.4; for devices with protection factors of 10 the effective factor for tritium oxide is about 1.7, and for devices with protection factors of 100 or more the effective factor for tritium oxide is about 1.9. Air-purifying respirators are not suitable for protection against tritium oxide. See also footnote i concerning supplied-air suits.
- Canisters and cartridges shall not be used beyond service-life limitations.
- g. Under-chin type only. This type of respirator is not satisfactory for use where it might be possible (e.g., if an accident or emergency were to occur) for the ambient airborne concentrations to reach instantaneous values greater than 10 times the pertinent values in Table 1, Column 3 of Appendix B of this part. This type of respirator is not suitable for protection against plutonium or other high-toxicity materials. The mask is to be tested for fit prior to use, each time it is donned.

174

h.1. Equipment shall be operated in a manner that ensures that proper air flow-rates are maintained. A protection factor of no more than 1000 may be utilized for tested-and-certified supplied-air hoods when a minimum air flow of 6 cubic feet (0.17 cubic meters) per minute is maintained and calibrated airline pressure gauges or flow measuring devices are used. A protection factor of up to 2000 may be used for tested and certified hoods only when the air flow is maintained at the manufacturer's recommended maximum rate for the equipment, this rate is giller than 6 cubic feet (0.17 cubic meters) per minute, and calibrated airline pressure gauges or flow measuring devices are used.

2. The design of the supplied-air hood or helmet (with a minimum flow of 6 cfm (0.17 m<sup>3</sup> per minute) of air) may determine its overall efficiency and the protection it provides. For example, some hoods aspirate contaminated air into the breathing zone when the wearer works with hands-c --head. This aspiration may be overcome if a short cape-like extension to the hood is worn under a coat or overalls. Other limitations specified by the approval agency shall be considered before using a hood in certain types of atmospheres (see footnote i).

- i. Appropriate protection factors shall be determined, taking into account the design of the suit and its permeability to the contaminant under conditions of use. There shall be a standby rescue person equipped with a respirator or other apparatus appropriate for the potential hazards and communications equipment whenever supplied-zir suits are used.
- j. No approval schedules are currently available for this equipment. Equipment is to be evaluated by testing or on the basis of reliable test information.

175

- k. This type of respirator may provide greater protection and be used as an emergency device in unknown concentrations for protection against inhalation hazards. External radiation hazards and other limitations to permitted exposure, such as skin absorption, must be taken into account in such circumstances.
- Quantitative fit testing shall be performed on each individual and no more than 0.02% leakage is allowed with this type of apparatus. Perceptible outward leakage of gas from this or any positive pressure self-contained breathing apparatus is unacceptable because service life will be reduced substantially. Special training in the use of this type of apparatus shall be provided to the wearer.

Note 1: Protection factors for respirators as may be approved by the U.S. Bureau of Mines/National Institute for Occupational Safety and Health (NIOSH), according to applicable approvals for respirators for type and mode of use to protect against airborne radionuclides, may be used to the extent that they do not exceed the protection factors listed in this table. The protection factors listed in this table may not be appropriate to circumstances where chemical or other respiratory hazards exist in addition to radioactive hazards. The selection and use of respirators for such circumstances should take into account applicable approvals of the U.S. Bureau of Mines/NIOSH.

Note 2: Radipactive contaminants for which the concentration values in Table 1, Column 3 of Appendix B of this part are based on internal dose due to inhalation may, in addition, present external exposure hazards at higher concentrations. Under these circumstances, limitations on occupancy may have to be governed by external dose limits.
#### APPENDIX B

# ANNUAL LIMITS ON INTAKE (ALIS) AND DERIVED AIR CONCENTRATIONS (DACS) OF RADIONUCLIDES FOR OCCUPATIONAL EXPOSURE; EFFLUENT CONCENTRATIONS; CONCENTRATIONS FOR RELEASE TO SEWERAGE

#### Introduction

For each radionuclide Table 1 indicates the chemical form which is to be used for selecting the appropriate ALI or DAC value. The ALIs and DACs for inhalation are given for an aerosol with an activity median aerodynamic diameter (AMAD) of 1  $\mu$ m and for three classes (D,W,Y) of radioactive material, which refer to their retention (approximately days, weeks or years) in the pulmonary region of the lung. This classification applies to a range of clearance halftimes for 0 of less than 10 days, for W from 10 to 100 days, and for , greater than 100 days. Table 2 provides concentration limits for airborne and liquid effluents released to the general environment. Table 3 provides concentration limits for discharges to sanitary sewer systems.

#### Notation

The values in Tables 1, 2, and 3 are presented in the computer "E" notation. In this notation a value of 6E-02 represents a value of 6 x  $10^{-2}$  or 0.06, 6E+2 represents 6 x  $10^{2}$  or 600, and 6E+0 represents 6 x  $10^{0}$  or 6.

#### Table 1 "Occupational"

Note that the columns in Table 1 of this appendix captioned "Oral Ingestion ALI," "Inhalation ALI," and "DAC," are applicable to occupational exposure to radioactive material.

The ALIs in this appendix are the annual intakes of given radionuclide by "Reference Man" which would result in either (1) a committed effective dose equivalent of 5 rems (stochastic ALI) or (2) a committed dose equivalent of 50 rems to an organ or tissue (non-stochastic ALI). The stochastic ALIs were derived to result in a risk, due to irradiation of organs and tissues, comparable to the risk associated with deep dose equivalent to the whole body of 5 rems. The derivation includes multiplying the committed dose equivalent to an organ or tissue by a weighting factor,  $w_T$ . This weighting factor is the

proportion of the risk of stochastic effects resulting from irradiation of the organ or tissue, T, to the total risk of stochastic effects when the whole body is irradiated uniformly. The values of  $w_T$  are listed under the definition of weighting factor in § 20.3. The non-stochastic ALIs were derived to avoid non-stochastic effects, such as prompt damage to tissue or reduction in organ function.

A value of  $J_{T} = 0.06$  is applicable to each of the five organs or tissues in the "remainder" category receiving the highest dose equivalents, and the dose equivalents of all other remaining tissues may be disregarded. The following parts of the GI tract -- stomach, small intestine, upper large intestine, and lower large intestine -- are to be treated as four separate organs.

Note that the dose equivalents for extremities (hands and forearms, feet and lower legs), skin, and lens of the eye are not considered in computing the committed effective dose equivalent, but are subject to limits that must be met separately.

When an ALI is defined by the stochastic dose limit, this value alone, is given. When an ALI is determined by the non-stochastic dose limit to an organ, the organ or tissue to which the limit applies is shown, and the ALI for the stochastic limit is shown in parentheses. (Abbreviated organ or tissue designations are used:

> LLI wall = lower large intestine wall; St. wall = stomach wall; Blad wall = bladder wall; and Bone surf = bone surface.)

The use of the ALIs listed first, the more limiting of the stochastic and non-stochastic ALIs, will ensure that non-stochastic effects are avoided and that the risk of stochastic effects is limited to an acceptably low value. If, in a particular situation involving a radionuclide for which the nonstochastic ALI is limiting, use of that non-stochastic ALI is considered unduly conservative, the licensee may use the stochastic ALI to determine the committed effective dose equivalent. However, the licensee shall also ensure that the 50-rem dose equivalent limit for any organ or tissue is not exceeded by the sum of the external deep dose equivalent plus the internal committed dose to that organ (not the effective dose). For the case where there is no external dose contribution, this would be demonstrated if the sum of the fractions of the nonstochastic ALIs (ALI<sub>ns</sub>) that contribute to the committed dose equivalent to the organ receiving the highest dose does not exceed unity (i.e.,  $\Sigma$  (intake (in  $\mu$ Ci) of each radionuclide/ALI<sub>ins</sub>)  $\leq$  1.0). If there is an external deep dose equivalent contribution of H<sub>d</sub> then this sum must be less than 1 - (H<sub>d</sub>/50) instead of being  $\leq$  1.0.

Note that the dose equivalents for extremities (hand and forearms, feet and lower legs), skin, and lens of the are are not considered in computing the committed effective dose equivalent, but are subject to limits that must be met separately.

The derived air concentration (DAC) values are derived limits intended to control chronic occupational exposures. The relationship between the DAC and the ALI is given by:

DAC = ALI(in  $\mu$ Ci)/(2000 hours per working year x 60 minutes/hour x 2 x 10<sup>4</sup> ml per minute) = [ALI/2.4 x 10<sup>9</sup>]  $\mu$ Ci/ml,

where 2 x 10<sup>4</sup> ml per minute is the volume of air breathed per minute at work by "Reference Man" under working conditions of "light work."

The DAC values relate to one of two modes of exposure: either external submersion or the internal committed dose equivalents resulting from inhalation of radioactive materials. Derived air concentrations based upon submersion are for immersion in a semi-infinite cloud of uniform concentration and apply to each radionuclide separately.

The ALI and DAC values relate to exposure to the single radionuclide named, but also include contributions from the in-growth of any daughter radionuclide produced in the body by the decay of the parent. However, intakes that include both the parent and daughter radionuclides should be treated by the general method appropriate for mixtures.

The value of ALI and DAC do not apply directly when the individual both ingests and inhales a radionuclide, when the individual is exposed to a mixture of radionuclides by either inhalation or ingestion or both, or when the individual is exposed to both internal and external irradiation ( see § 20.202). When an individual is exposed to radioactive materials which fall under several of the translocation classifications (i.e., Class D, Class W, or Clas Y) of the same radionuclide, the exposure may be evaluated as if it were a mixture of different radionuclides.

It should be noted that the classification of a compound as Class D, W, or Y is based on the chemical form of the compound and does not take into account the radiological half-life of different radioisotopes. For this reason, values are given for Class D, W, and Y compounds, even for very short-lived radionuclides.

#### Table 2

The columns in Table 2 of this appendix captioned "Effluents," "Air," and "Water," are applicable to the assessment and control of dose to the public, particularly in the implementation of the provisions of § 20.302. The concentration values given in Columns 1 and 2 of Table 2 are equivalent to the radionuclide concentrations which, if inhaled or ingested continuously over the course of a year, would produce a total effective dose equivalent of 0.05 rem (50 millirem or 0.5 millisieverts)

Consideration of non-stochastic limits has not been included in deriving the air and water effluent concentration limits because non-stochastic effects are presumed not to occur at the dose levels established for individual members of the public. For radionuclides, where the non-stochastic limit was governing in deriving the occupational DAC, the stochastic ALI was used in deriving the corresponding airborne effluent limit in Table 2. For this reason, the DAC and airborne effluent limits are not always proportional as they were the previous Appendix B. The air concentration values listed in Table 2, Column 1, were derived by one of two methods. For those radionuclides for which the stochastic limit is governing, the occupational stochastic inhalation ALI was divided by 2.4 x  $10^9$ , relating the inhalation ALI to the DAC, as explained above, and then divided by a factor of 300. The factor of 300 includes the following components: a factor of 50 to relate the 5-rem annual occupational dose limit to the 0.1-rem limit for members of the public, a factor of 3 to adjust for the difference in exposure time and the inhalation rate for a worker and that for members of the public; and a factor of 2 to adjust the occupational values (derived for adults) so that they are applicable to other age groups.

For those radionuclides for which submersion (external dose) is limiting, the occupational DAC in Table 1, Column 3, was divided by 219. The factor of 219 is composed of a factor of 50, as described above, and a factor of 4.38 relating occupational exposure for 2,000 hours per year to full-time exposure (8,760 hours per year). Note that an additional factor of 2 for age considerations is not warranted in the submersion case.

The water concentrations were derived by taking the most restrictive occupational stochastic oral ingestion ALI and dividing by 7.3 x  $10^7$ . The factor of 7.3 x  $10^7$  (ml) includes the following components: the factors of 50 and 2 described above and a factor of 7.3 x  $10^5$  (ml) which is the annual water intake of "Reference Man."

Note 2 of this appendix provides groupings of radionuclides which are applicable to unknown mixtures of radionuclides. These groupings (including occupational inhalation ALIs and DACs, air and water effluent concentrations and sewerage) require demonstrating that the most limiting radionuclides in successive classes are absent. The limit for the unknown mixture is defined when the presence of the one of the listed radionuclides cannot be definitely excluded as being present either from knowledge of the radionuclide composition of the source or from actual measurements.

### Table 3 "Sewer Disposal"

The monthly average concentrations for release to sanitary sewers are applicable to the provisions in § 20.1003. The concentration values were derived by taking the most restrictive occupational stochastic oral ingestion ALI and dividing by  $7.3 \times 10^6$  (ml). The factor of  $7.3 \times 10^6$  (ml) is composed of a factor of  $7.3 \times 10^6$  (ml), the annual water intake by "Reference Man," and a factor of 10, such that the concentrations, if the sewage released by the licensee were the only source of water ingested by a reference man during a year, would result in a committed effective dose equivalent of 0.5 rem.

## LIST OF ELEMENTS

Γ

	Ato	mic		Ato	mic
Name	Symbol	Number	Name	Symbol	Number
Actinium	Ac	89	Mercury	Ha	80
Aluminum	Al	13	Molybdenum	Mo	42
Americium	Am	95	Neodymium	Nd	60
Antimony	Sh	51	Neptupium	NO	00
Argon	Ar	18	Nickol	NP	93
Arcenic	Ac	10	Nichium	NI	28
Actatino	M5	33	NIODIUM	ND	41
Ranium	Re	65	Osmium	Us	76
Parkalium	ba	50	Palladium	Pd	46
Penullium	DK	97	Phosphorus	P	15
Deryllium	be	4	Platinum	Pt	78
Dismuth	81	83	Plutonium	Pu	94
bromine	Br	35	Polonium	Po	84
Cadmium	Cd	48	Potassium	K	19
Calcium	Ca	20	Praseodymium	Pr	59
Californium	Cf	98	Promethium	Pm	61
Carbon	C	6	Protactinium	Pa	91
Cerium	Ce	58	Radium	Ra	88
Cesium	Cs	55	Radon	Rn	86
Chlorine	C1	17	Rhenium	Re	75
Chromium	Cr	24	Rhodium	Rh	45
Cobalt	Co	27	Rubidium	Rb	37
Copper	Cu	29	Ruthenium	Ru	44
Curium	Cm	96	Samarium	Sm	62
Dysprosium	DV	66	Scandium	Sc	21
Einsteinium	Es	99	Selenium	So	24
Erbium	Er	68	Silicon	Ci i	14
Europium	Eu	63	Silver	51	14
Fermium	Fm	100	Sodium	ng	4/
Fluorine	F	9	Stroptium	na	11
Francium	Fr	07	Scroncrum	Sr	38
Gadolinium	Gd	64	Tantalum	5	10
Gallium	Ga	21	Tantalum	la	/3
Germanium	Go	22	Tellinetium	IC	43
Gold	Au	52	Tellurium	le	52
Hafnium	NU UF	79	Terbium	TD	65
Holmium	He	67	Inallium	11	81
Hudnogon	HO	6/	Inorium	Th	90
Indium	n	1	Thullum	Tm	69
Indian	10	49	Tin	Sn	50
Todine	1	53	Titanium	Ti	22
Inigium	Ir	17	Tungsten	W	74
Iron	Fe	26	Uranium	U	92
Krypton	Kr	36	Vanadium	V	23
Lanthanum	La	57	Xenon	Xe	54
Lead	Pb	82	Ytterbium	Yb	70
Lutetium	Lu	71	Yttrium	Y	39
Magnesium	Mg	12	Zinc	Zn	30
Manganese	Mn	25	Zirconium	Zr	40
Mendelevium	Md	101			

			Occ	Table 1 Supational N	Values	Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration ) (µCi/ml)
			Col. 1 Oral	Co1 2	Col. 3	Col. 1	ble 2 luent trations Col. 2 Water (µCi/ml) 1E-3 1 tir and in the 6E-4 6 - - 2E-5 2 - - 6E-3 6	Monthly
			Ingestion Inhalation		Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-2 the body to HTO. 6E-3 - 2E-4 - 6E-2 - 3E-4			
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/m1)	Air (µCi∕ml)	Water (µCi/ml)	Concentratio (µCi/ml)
1	Hydrogen-3	Water, DAC includes skin	8F+đ	9E+4	25-5	15-7	15-2	15-2
		Gas (HT or T.) Submersion	1 lise above	Values as 1	and T ovi	dize in a	ir and in t	the body to UTO
		das (in or 12), submersion	USE above	raiues as i	in and 12 OX	ioize in al	in and in t	the body to HTU
4	Beryllium-7	W, all compounds except those given for Y	4E+4	2E+4	9E-6	3E-8	6E-4	6E-3
		Y, oxides, halides, and						
		nitrates		2E+4	8E-6	3E-8	-	-
4	Beryllium-10	W, see <sup>7</sup> Be	1E+3 LLI wall	2E+2	6E-8	2E-10	-	-
		입장 같은 바다 이 것이 같은 것이 없다.	(1E+3)	-	-	-	2F-5	2F-4
		Y, see 'Be		1E+1	6E-9	2E-11	-	-
	2	그렇게요? 많이 그는 다 숨긴다.						
6	Carbon-11-	Monoxide	249212131	1E+6	5E-4	2E-6	-	
		Dioxide		6E+5	3E-4	9E-7	-	1.7
		compounds	46+5	4E+5	2E-4	6E-7	6E-3	6E-2
6	Carbon-14	Monoxide		2E+6	7F-4	2F-6		
		Dioxide		2E+5	9E-5	3E-7	-	
		Compounds	2E+3	2E+3	1F-6	3F-9	35-5	3E-4

			Occ	Table 1 upational	Values	Tal Eff Concent	ble 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inha ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/ml)
9	Fluorine-18 <sup>2</sup>	D, fluorides of H, Li, Na, K, Rb, Cs, and Fr	5E+4 St,wall	7E+4	3E-5	1E-7	-	-
		W, fluorides of Be, Mg, Ca, Sr, Ba, Ra, Al, Ga, In, Tl, As, Sb, Bi, Fe, Ru, Os, Co, Ni, Pd, Pt, Cu, Ag, Au, Zn, Cd, Hg, Sc, Y, Ti, Zr, V, Nb, Ta, Mn, Tc, and Re	(5E+4)	9E+4	4E-5	1E-7	7E-4	7E-3
		Y, lanthanum fluoride		8£+4	3E-5	1E-7	-	2
11	Sodium-22	D, all compounds	4E+2	6E+2	3E-7	9E-10	6E-6	6E-5
11	Sodium-24	D, all compounds	4E+3	5E+3	2E-6	7E-9	5E-5	5E-4
12	Magnesium-28	D, all compounds except those given for W	7E+2	2E+3	7E-7	2E-9	9E-6	9E-5
		W, oxides, hydroxides, carbides, halides, and nitrates	-	16+3	5E-7	2E-9	-	
13	Aluminum-26	D, all compounds except those given for W	4E+2	6E+1	3E-8	98-11	6E-6	6E-5
		W, oxides, hydroxides, carbides, halides, and nitrates		9E+1	4E-8	1E-10		

Table 3 Releases to ions Sewers	Table 2 Effluent Concentrations	/alues	Table 1 upational Va	0cc			
I. 2 Manthlu	Col. 1 Col. 2	Col. 3	Co1. 2	Col. 1			
Average		lation	Inhala	Indestion			
ter Concertration Ci/ml) (µCi/ml)	Air Water (µCi/ml) (µCi/ml)	DAC (µCi/ml)	ALI (µCi)	ALI (µCi)	Class	Radionuclide	Atomic No.
				T and	D, all compounds except	Silicon-31	14
4 1E-3	4E-8 1E-4	1E-5	3E+4	9E+3	those given for W and Y		
					W axides hydroxides		
and set to a set	5E-8 -	1E-5	3E+4		carbides, and nitrates		
	4E-8 -	1E-5	3E+4		Y, aluminosilicate glass		
-	3E-10 -	1E-7	2E+2	2E+3 LLI wall	D, see <sup>31</sup> Si	Silicon-32	14
5 4E-4	- 4E-5	-	. 영화	(3E+3)	31		
	2E-10 -	5E-8	1E+2	방법이 같은 것	W, see 31Si		
	7E-12 -	2E-9	5E+0	영화 중 영화	Y, see Si		
					D, all compounds except	Phosphorus-32	15
6 9E-5	1E-9 9E-6	4E-7	9E+2	6E+2	phosphates given for W		
					W, phosphates of $Zn^{2+}$ , $S^{3+}$ , $Mg^{2+}$ , $Fe^{3+}$ , $Bi^{3+}$ .		
	5E-10 -	2E-7	4E+2		and lanthanides		
	15.0	AF. C	05+3	65+2	D see 32	Phosphorus-33	15
5 8E-4	1E-8 8E-5	46-0	35+3	-	W see 32p	inospiloras 00	
	5E-8 - 4E-8 - 3E-10 - - 2E-10 - 7E-12 - 1E-9 9E-6 5E-10 - 1E-8 8E-1 4E-9 -	1E-5 1E-5 1E-7 - 5E-8 2E-9 4E-7 2E-7 4E-6 1E-6	3E+4 3E+4 2E+2 - 1E+2 5E+0 9E+2 4E+2 8E+3 3E+3	- 2E+3 LLI wall (3E+3) - 6E+2 -	W, oxides, hydroxides, carbides, and nitrates Y, aluminosilicate glass D, see <sup>31</sup> Si W, see <sup>31</sup> Si Y, see <sup>31</sup> Si D, all compounds except phosphates given for W W, phosphates of Zn <sup>2+</sup> , S <sup>3+</sup> , Mg <sup>2+</sup> , Fe <sup>3+</sup> , Bi <sup>3+</sup> , and lanthanides D, see <sup>32</sup> p W, see <sup>32</sup> p W, see <sup>32</sup> p	Silicon-32 Phosphorus-32 Phosphorus-33	14 15 15

			0cc	Table 1Table 2Occupational ValuesEffluentConcentrations				Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml)
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhal ALI (µCi)	DAC (µCi/ml)	Air (µCi∕ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
16	Sulfur-35	Vapor	-	1E+4	6E-6	2E-8	-	-
		D, sulfides and sulfates except those given for W	6E+3 LLI wall (8E+3)	2E+4	7E-6 -	2E-8	- 1E-4	- 1E-3
		W, elemental sulfur, sulfides of Sr, Ba, Ge, Sn, Pb, As, Sb, Bi, Cu, Ag, Au, Zn, Cd, Hg, W, and Mo. Sulfates of Ca, Sr, Ba, Ra, As, Sb, and Bi	-	2E+3	9E-7	3E-9	-	
17	Chlorine-36	D, chlorides of H, Li, Na, K, Rb, Cs, and Fr	2E+3	2E+3	1E~6	3E-9	2E-5	2E-4
		W, chlorides of lantha- nides, Be, Mg, Ca, Sr, Ba, Ra, Al, Ga, In, Tl, Ge, Sn, Pb, As, Sb, Bi, Fe, Ru, Os, Co, Rh, Ir, Ni, Pd, Pt, Cu, Ag, Au, Zn, Cd, Hg, Sc, Y, Ti, Zr, Hf, V, Nb, Ta, Cr, Mo, W, Mn, Tc, and Re		2E+2	1E-7	3E-10		

			Table 1 Occupational Values			Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-3 - 5E-3 - - 4E-5 6E-4 9E-4
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 1 Col. 2 Air Water (µCi/ml) (µCi/ml) 5E-8 - - 3E-4 5E-8 -	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Monthly Average Concentration (µCi/ml) - 3E-3 - 5E-3 -
17	Chlorine-38 <sup>2</sup>	D, see <sup>36</sup> C1	2E+4 St. wall	4E+4	2E-5	6E-8	-	-
		W see 36	(3E+4)	- 5F+4	25-5	- 6F-8	3E-4	3E-3
17	Chlorine-39 <sup>2</sup>	D, see <sup>36</sup> C?	2E+4 St. wall	5E+4	2E-5	7E-8	-	-
		36	(4E+4)	-	-	-	5E-4	5E-3
		W, see <sup>So</sup> Cl		6E+4	2E-5	8E-8	-	
18	Argon-37	Submersion <sup>1</sup>		-	1E+0	6E-3	-	-1.1.196
18	Argon-39	Submersion <sup>1</sup>		-	2E-4	8E-7	-	
18	Argon-41	Submersir	667 - C	-	3E-6	1E-8	-	-
19	Potassium-40	D, all compounds	3E+2	4E+2	2E-7	6E-10	4E-6	4E-5
19	Potassium-42	D, all compounds	5E+3	5E+3	2E-6	7E-9	6E-5	6E-4
19	Potassium-43	D, all compounds	6E+3	9E+3	4E-6	1E-8	9E-5	9E-4
19	Potassium-44 <sup>2</sup>	D, all compounds	2E+4 St. wall	7E+4	3E-5	9E-8	-	-
			(4E+4)	-	-	-	5E-4	5E-3
19	Potassium-45 <sup>2</sup>	D, all compounds	3E+4 St_wall	1E+5	5E-5	2E-7	-	-
			(5E+4)	-		-	7E-4	7E-3

**-**'I'

1

			0cc	Table 1 cupational Va	lues	Tal Eff Concent	ble 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration ) (µCi/ml) - 6E-4 2E-4
			Col. 1 Oral Ingestion	Col. 2 Inbala	Col. 2 Col. 3		Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi∕ml)	Water (µCi/ml)	Monthly Average Concentration (µCi/ml) - 6E-4 2E-4 1E-4 1E-3 7E-5 5E-4 1E-4 1E-4
20	Calcium-41	W, all compounds	3E+3 Bone surf	4E+3 Bone surf	2E-6	-	-	-
			(4E+3)	(4E+3)		5E-9	6E-5	6E-4
20	Calcium-45	W, all compounds	2E+3	8E+2	4E-7	1E-9	28-5	2E-4
20	Calcium-47	W, all compounds	8E+2	9E+2	4E-7	1E-9	1E-5	1E-4
21	Scandium-43	Y, all compounds	7E+3	2E+4	9E-6	3E-8	1E-4	1F-3
21	Scandium-44m	Y, all compounds	5E+2	7E+2	3E-7	1E-9	7F-6	75-5
21	Scandium-44	Y, all compounds	4E+3	1E+4	5E-6	2F-8	55-5	55-4
21	Scandium-46	Y, all compounds	9E+2	2E+2	1F-7	3E-10	16-5	15-4
21	Scandium-47	Y, all compounds	2E+3 LLI wall	3E+3	1E-6	4E-9	-	-
			(3E+3)		-	-	4E-5	4E-4
21	Scandium-48	Y, all compounds	8E+2	1E+3	6E-7	2E-9	1E-5	1E-4
21	Scandium-49 <sup>2</sup>	Y, all compounds	2E+4	5E+4	2E-5	8E-8	3E-4	3E-3
22	Titanium-44	D, all compounds except those given for W and Y	3E+2	1E+1	5E-9	2E-11	4E-6	4E-5
		W, oxides, hydroxides, carbides, halides, and nitrates		25.13				
			1. A.	32+1	16-8	4E-11	-	-
		Y, SrTiO <sub>3</sub>		6E+0	2E-9	8E-12	-	

			Table 1 Occupational Values			Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-3 - 4E-3 - 9E-5 - 1E-2 - 8E-4
			Col. 1 Oral	Col. 2 Col. 3		Col. 1 C	ble 2 luent itrations Col. 2 Water (µCi/ml) 1E-4 - - 4E-4 - 9E-6 - 1E-3 - 1E-3 -	Monthly
		그는 것이 많은 것을 위한 것을 하는 것을 알았다.	Ingestion	Inhala	tion			Average
No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Pr Concentration i/m1) (µCi/m1) I IE-3 - - - - - - - - - - - - -
22	Titanium-45	D, see 44Ti	9E+3	3E+4	16-5	3F-8	1F-4	1F-3
		W. see 44Ti	-	4F+4	1F-5	5E-8	-	1L J
		Y, see <sup>44</sup> Ti	-	3E+4	1E-5	4E-8	-	-
23	Vanadium-47 <sup>2</sup>	D, all compounds except						
		those given for W	3E+4 St wall	8E+4	3E-5	1E-7	-	ABLE-2 ABLE-3 ABLE-3 ABLE-3 ABLE-4 ABLE-3 ABLE-4 ABLE-4 ABLE-3 ABLE-4 ABLE-3 ABLE-4 ABLE-3 ABLE-4 ABLE-3
			(3E+4)	-	-	-	4E-4	4E-3
		W, oxides, hydroxides, carbides, and halides		1E+5	4E-5	1E-7	_	
23	Vanad um-49	D coo 47.	65.12	15.0				
2.5	Valiau I din 40	W SPE 47V	0272	1673	5E-7	2E-9	9E-6	9E-5
		**, 500		OL+2	SE-7	96-10		-
23	Vanadium-49	D, see <sup>47</sup> V	7E+4	3E+4	1E-5	-	-	2
			LLI wall	Bone surf				
		47	(9E+4)	(3E+4)	-	5E-8	1E-3	1E-2
		W, see 'V		2E+4	8E-6	2E-8	-	
24	Chromium-48	D, all compounds except						
		those given for W and Y	6E+3	1E+4	5E-6	2E-8	8E-5	8E-4
		W, halides and nitrates		7E+3	3E-6	1E-8	-	-
		Y, oxides and hydroxides		7E+3	3E-6	1E-8	-	-
24	Chromium-49 <sup>2</sup>	D. see 48cr	35+4	85+4	45-5	15-7	45.4	15.3
		W SPP 48Cr	-	1645	AE-E	15-7	46.4	41-3
		Y see 48 Cr		06+1	46-5	10-7		
		1, 000 01		JE+4	46-0	11-1	-	-

			Table 1 Occupational Values			Tat Effi Concent	ole 2 luent crations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 5E-3 - - 5E-3 - 1E-4 - 7E-3
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Ingestion ALI (µCi)	n Inhalation				Average
Atomic No.	Radionuclide	Class		ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
24	Chromium-51	D. see 48 Cr	4F+4	55+4	25-5	65-0	EE . A	
		W see 48 Cr	46.14	25+4	16-6	00-8	56-4	5E~3
		Y, see <sup>48</sup> Cr		2E+4	8F-6	3E-8	-	2
25	Manganese-51 <sup>2</sup>	D, all compounds except				02.0		
		those given for W W, oxides, hydroxides,	2E+4	5E+4	2E-5	7E-8	3E~4	3E-3
		halides, and nitrates	. 물로운 것이	6E+4	3E-5	8E-8	-	
25	Manganese~52m <sup>2</sup>	D, see <sup>51</sup> Mn	3E+4 St. wall	9E+4	4E-5	1E-7	-	- 1752
		51	(4E+4)		-	-	55-4	55-2
		W, see <sup>31</sup> Mn	-	1E+5	4E-5	1E-7	-	-
25	Manganeco-52	D 500 51mg	75.0					
2.5	Hangarese SE	b, see 51m	12+2	112+3	5E-7	2E-9	1E-5	1E-4
		w, see Mn		9E+2	4E-7	1E-9	-	-
25	Manganese-53	D, see <sup>51</sup> Mn	5E+4	1E+4 Bone surf	5E-6	-	7E-4	7E-3
		61	-	(2E+4)		35-8		
		W, see <sup>SI</sup> Mn		1E+4	5E-6	2E-8	1	2
25	Manganese-54	D, see <sup>51</sup> Mn W, see <sup>51</sup> Mn	2E+3	9E+2 8E+2	4E-7	1E-9	3E-5	3E-4
				OL Z	SE-1	16-9		1
25	Manganese-56	D, see <sup>51</sup> Mn W, see <sup>51</sup> Mn	55+3	2E+4 2E+4	6E-6 9E-6	2E-8 3E-8	7E-5	7E-4

			Table 1 Occupational Values			Tab Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-4 - 1E-3 - 1E-4 - 4E-6 - 2E-4
			Col. 1 Oral Ingestion	Col. 2 Inhal	Col. 3	Col. 1	ble 2 luent trations Col. 2 Water (µCi/ml) IE-5 - IE-4 - IE-5 - 4E-7 - 2E-5 - 6E-6 - 6E-5	Monthly
Atomic No.	Radionuclide	flass	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/mi)	Water (µCi/ml)	Concentration (µCi/ml)
26	Iron-52	D, all compounds except those given for W	9E+2	3E+3	1E-6	4E-9	1E-5	1E-4
		W, oxides, hydroxides, and halides		2E+3	1E-6	3E-9	-	-
26	Iron-55	D, see <sup>52</sup> Fe W, see <sup>52</sup> Fe	9E+3	2E+3 4E+3	8E-7 2E-6	3E-9 6E-9	1E-4	1E-3
26	Iron-59	D, see <sup>52</sup> Fe , see <sup>52</sup> Fe	8E+2	3E+2 5E+2	1E-7 2E-7	5E-10 7E-10	18-5	16-4
26	Iron-60	D, see <sup>52</sup> Fe W, see <sup>52</sup> Fe	3E+1 -	6E+0 2E+1	3E-9 8E-9	9E-12 3E-11	46-7	4E-6
27	Cobalt-55	W, all compounds except those given for Y	1E+3	3E+3	1E-6	4E-9	2E-5	2E-4
		Y, oxides, hydroxides, halides, and nitrates		3E+3	18-6	42-9	-	
27	Cobalt-56	W, see <sup>55</sup> Co Y, see <sup>55</sup> Co	4E+2	3E+2 2E+2	1E-7 8E-8	4E-10 3E-10	6E-6	6E-5
27	Cobalt-57	W, see <sup>55</sup> Co Y, see <sup>55</sup> Co	4E+3	3E+3 7E+2	17-6 3F-7	4E-9 9E-10	6E-5	68-4

			0cc	Table 1 upational \	/alues	Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 8E-3
			6.	Cə1. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/ml)
27	Cobalt-58m	W, see <sup>55</sup> Co Y, see <sup>55</sup> Co	6E+4 -	9E+4 6E+4	4E-5 3E-5	1E-7 9E-8	8E-4	8E-3
27	Cobalt-58	W, see <sup>55</sup> Co Y, see <sup>55</sup> Co	1E+3	1E+3 7E+2	5E-7 3E-7	2E-9 1E-9	2E-5	2E-4
27	Cobalt-60m <sup>2</sup>	W, see <sup>55</sup> Co	1E+6 St_wall	i	2E-3	6E-6	-	
		Y, see <sup>55</sup> Co	(1E+6)	- 3E+6	- 1E-3	- 4E-6	2E-2	2E-1
27	Cobalt-60	W, see <sup>55</sup> Co Y, see <sup>55</sup> Co	2E+2	2E+2 3E+1	7E-8 1F-8	2E-10 5E-11	3E-6	3E-5
27	Cobalt-61 <sup>2</sup>	W, see <sup>55</sup> Co Y, see <sup>55</sup> Co	2E+4	6E+4 6E+4	3E-5	9E-8	3E-4	3E-3
27	Cobalt-62m <sup>2</sup>	₩, see <sup>55</sup> Co	4E+4 St wall	2E+5	76-5	2E-7	-	-
		Y, see <sup>55</sup> Co	(5E+4)	- 2E+5	- 6E-5	- 2E-7	7E-4	7E-3
28	Nickel-56	D, all compounds except those given for W	1E+3	2E+3	8E-7	3E-9	25-5	25-4
		W, oxides, hydroxides,		15.2				
		Vapor	-	1E+3	5E-7	2E-9 2E-9	2	-
28	Nickel-57	D, see <sup>56</sup> Ni W, see <sup>56</sup> Ni Vapor	2E+3 -	5E+3 3E+3	2E-6 1E-6	7E-9 4E-9	2E-5	2E-4

			Table 1 Occupational Values			Tal Eff Concent	ble 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-3 - 1E-3 - 6E-5 - - 4E-3
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Ingestion	Inhal	lation			Monthly Average Concentration (µCi/ml) 3E-3 - 1E-3 -
Atomic No.	Radionuclide	Class	ALI (µCi)	ALĬ (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
28	Nickel-59	D. SPE SONI	2E+4	4E+3	2E-6	5E-9	3E-4	3E-3
~~~	intenet av	W. see <sup>56</sup> Ni	-	7E+3	3E-6	1E-8	-	-
		Vapor		2E+3	8E-7	3E-9	-	-
28	Nickel-63	D see 56Ni	95+3	2E+3	7F-7	2F-9	1F-4	16-3
20	MICKET 00	W see 56Ni	-	3E+3	15-6	4F-9	-	-
		Vapor	-	8E+2	3E-7	1E-9	-	-
28	Nickel-65	D see 56Ni	8F+3	2F+4	16-5	35-8	1E-4	1E-3
2.0	HICKET 00	W see 56Ni	-	3F+4	1F-5	4F-8	-	-
		Vapor	-	2E+4	7E-6	2E-8	-	-
28	Nickel-66	D, see <sup>56</sup> Ni	4E+2 LLI wall	2E+3	7E-7	2E-9	-	
		5.6	(5E+2)	-	-	-	6E-6	6E-5
		W, see DN1	-	6E+2	3E-7	9E-10	-	-
		Vapor		3E+3	1E-6	4E-9	-	-
29	Copper-60 <sup>2</sup>	D, all compounds except						
		those given for W and Y	3E+4 St. wall	9E+4	4E-5	1E-7	-	-
			(3E+4)	-	-	+	4E-4	4E-3
		W, sulfides, halides, and nitrates	-	1E+5	5E-5	2E-7	-	-
		Y, oxides and hydroxides	-	1E+5	4E-5	1E-7	4	2.1
20	Conser-61	0 60 60	15+4	35+4	15-5	45-0	25-4	25-2
	copper or	W see 60 Cu	-	AE+A	28-5	65-9	20.4	20-3
		Y SPP 60CH		AE+A	16-5	55-9		
		r, see cu		45.14	16 3	76-0		

 $\square$ 

П

1

.

			0cc	Table 1 upational N	Values	Tal Eff Concent	ble 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 2E-3 - 6E-4 - 2E-4 - 3E-3 5E-5 6E-4 8E-3 8E-4 1E-4 - 9E-3
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (pCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
29	Copper-64	D, see 60Cu	1E+4	3E+4	1F-5	4E-8	2E-4	2E-3
		W, see 60Cu		2E+4	1E-5	3E-8		-
		Y, see <sup>OC</sup> Cu	1.5	2E+4	9E-6	3E-8	1	여행 사람이 같다.
29	Copper-67	D see 60 CH	5E+3	8F+3	35-6	1F-8	6F-5	6F-4
	asphere of	W SPP 60CH	-	5E+3	2E-6	7E-9	-	-
		Y, see <sup>60</sup> Cu	-	5E+3	2E-6	6E-9		¥
30	Zinc-62	Y, all compounds	1E+3	3E+3	1E-6	4E-9	2E-5	2E-4
30	Zinc-63 <sup>2</sup>	Y, all compounds	2E+4 St. wall	7E+4	3E-5	9E-8		-
			(3E+4)	- 10 C	가는 데요?	-	3E-4	3E-3
30	Zinc-65	Y, all compounds	4E+2	3E+2	1E-7	4E-10	5E-6	5E-5
30	Zinc-69m	Y, all compounds	4E+3	7E+3	3E-6	1E-8	6E-5	6E-4
30	Zinc-69 <sup>2</sup>	Y, all compounds	6E+4	1E+5	6E-5	2E-7	8E-4	8E-3
30	Zinc-71m	Y, all compounds	6E+3	2E+4	7E-6	25-8	8E-5	8E-4
30	Zinc-72	Y, all compounds	1E+3	1E+3	5E-7	2E-9	18-5	1E-4
31	Gallium-65 <sup>2</sup>	D, all compounds except those given for W	5E+4	2E+5	7E-5	2F-7		
			St. wall					
		W, oxides, hydroxides,	(6E+4)	-		Ĩ	9E-4	9E-3
		nitrates		25+5	95-5	25-7		
				21.3	OC 3	36-1		The second se

			Table 1 Occupational Values		Table 2 Effluent Concentrations		Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-4 - 1E-3 - 2E-3 - 2E-3 - 2E-4 - 7E-4 - 3E-3
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi∕ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
31	Gallium-66	D, see <sup>65</sup> Ga W, see <sup>65</sup> Ga	1E+3 -	4E+3 3E+3	1E-6 1E-6	5E-9 4E-9	1E-5 -	1E-4
31	Gallium-67	D, see <sup>65</sup> Ga W, see <sup>65</sup> Ga	7E+3	1E+4 1E+4	6E-6 4E-6	2E-8 1E-8	1E-4	16-3
31	Gallium-68 <sup>2</sup>	D, see <sup>65</sup> Ga W, see <sup>65</sup> Ga	2E+4	4E+4 5E+4	2E-5 2E-5	6E-8 7E-8	2E-4	2E-3
31	Gallium-70 <sup>2</sup>	D, see <sup>65</sup> Ga	5E+1 St. wall	2E+5	7E-5	2E-7		
		W, see <sup>65</sup> Ga	(7E+4)	2E+5	- 8E-5	- 3E-7	1E-3 -	1E-2
31	Gallium-72	D, see <sup>65</sup> Ga W, see <sup>65</sup> Ga	1E+3	4E+3 3E+3	1E-6 1E-6	5E-9 4E-9	2E-5	2E-4
31	Gallium-73	D, see <sup>65</sup> Ga W, see <sup>65</sup> Ga	5E+3	2E÷4 2E+4	6E-6 6E-6	2E-8 2E-8	7E-5	7E-4
32	Germanium-66	D, all compounds except those given for W	2E+4	3E+4	1E-5	4E-8	3E-4	3E-3
		W, oxides, sulfides, and halides	_	2E+4	8E-6	3E-8		

			Table 1Table 2Occupational ValuesEffluentConcentrations				ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-3 - 6E-4 - 2E-3 - 7E-2 - 9E-3 - 1E-3	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-3 - 6E-4 - 2E-3 - 7E-2 - 9E-3 - 1E-3 - 3E-3 -		
Atomic	Radionuclide	Class	Ingestion	Inhalation				Average		
No.	Nouronact rue	01033	μCi)	ALI (µCi)	UAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)		
32	Germanium-67 <sup>2</sup>	D, see <sup>66</sup> Ge	3E+4 St. wall	9E+4	4E-5	1E-7	-	-		
		66	(4E+4)	-	-	-	6F-4	6F-3		
		W, see Ge		1E+5	4E-5	1E-7	-	-		
32	Germanium-68	D. see 66 Ge	5F+3	4F+3	25-6	55-0	65-5			
		W, see <sup>bb</sup> Ge	-	1E+2	4E-8	1E-10	- JO	bt-4		
32	Germanium-69	D see 66	15+4	25.4						
	oct marriant 05	W, see 66Ge	112+4	2E+4 9E+3	6E-6	2E-8	2E-4	Releases to Sewers 2 Monthly Average Concentration ml) (µCi/ml) - 6E-3 - 6E-4 - 2E-3 - 7E-2 - 9E-3 - 1E-3 - 3E-3 -		
		EE.		02+3	32-6	1E-8	-	-		
32	Germanium-71	D, see 66Ge	5E+5	4E+5	2E-4	6F-7	7F-3	75-2		
		W, see Ge		4E+4	2E-5	6E-8	-	-		
32	Germanium-752	0 500 6600	45.4							
	ocrimati un 75	U, SEE GE	4±+4 \$* wall	8E+4	3E-5	1E-7	-	-		
			(7F+4)	_			05.4	05.0		
		W, see <sup>66</sup> Ge	-	8E+4	4E-5	1F-7	96-4	96-3		
32	Cormanium- 77	66								
J	Germannum-//	U, see 650e	9E+3	1E+4	4E-6	1E-8	1E-4	2 Monthly Average Concentration ml) (µCi/ml) - 6E-3 - 6E-4 - 2E-3 - 7E-2 - 9E-3 - 1E-3 - 3E-3 -		
		m, see de		6E+3	2E-6	8E-9	-	-		
32	Germanium-78 <sup>2</sup>	D, see <sup>66</sup> Ge	2E+4 St. wall	2E+4	9E-6	3E-8	-	-		
		66	(2E+4)	-	-	-	35-4	Sewers Monthly Average Concentratio (µCi/ml) - 6E-3 - 6E-4 - 2E-3 - 7E-2 - 9E-3 - 1E-3 - 3E-3 -		
		W, see Ge	-	2E+4	9E-6	3F-8	-	36-3		

			Table 1 Occupational Values			Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-3 2E-3 5E-4 1E-4 1E-3 2E-4 1E-4 1E-4 1E-4 - 6E-4
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-3 2E-3 5E-4 1E-4 1E-4 1E-3 2E-4 1E-4 1E-4 1E-3 2E-4 1E-3 1E-3
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhal ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
33	Arsenic-69 <sup>2</sup>	W, all compounds	3E+4 St. wall	1E*5	5E-5	2E-7	-	-
			(4E+4)	-	-	-	6E-4	62-3
33	Arsenic-70 <sup>2</sup>	W, all compounds	1E+4	5E+4	2E-5	7E-8	2E-4	2E-3
33	Arsenic-71	W, all compounds	4E+3	5E+3	2E-6	6E-9	5E-5	5E-4
33	Arsenic-72	₩, all compounds	9E+2	1E+3	6E-7	2E-9	1E-5	1E-4
33	Arsenic-73	W, all compounds	8E+3	2E+3	7E-7	2E-9	1E-4	1E-3
33	Arsenic-74	W, all compounds	1E+3	8E+2	38-7	1E-9	2E-5	2E-4
33	Arsenic-76	W, all compounds	1E+3	1E+3	6E- ;	2E-9	1E-5	1E-4
33	Arsenic-77	₩, all compounds	4E+3	5E+3	2E-6	7E-9	~	- 1.
			(5E+3)	14 C			6E-5	6E-4
33	Arsenic-78 <sup>2</sup>	W, all compounds	8E+3	2E+4	9E-6	3E-8	1E-4	1E-3
34	Selenium-70 <sup>2</sup>	D, all compounds except those given for W	1E+4	4E+4	2E-5	5E-8	1E-4	1E-3
		W, oxides, hydroxides, carbides, and elemental Se	_	4E+4	2E-5	6E-8		

				Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Co1. 1	Col. 2	Table 3 Releases to Sewers Munthly Average Concentration (µCi/ml) 4E-3 - 7E-5 - 8E-5 - 3E-3 - 1E-2 - 4E-2
Atomic	Dadianualida	63	Ingestion	Inhal	ation			Average
No.	Radionacii de	The class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
34	Selenium-73m <sup>2</sup>	D, see <sup>70</sup> Se W, see Se	3E+4 -	2E+5 1E+5	6E-5 6E-5	2E-7 2E-7	4E-4	4E-3
34	Selenium-73	D, see <sup>70</sup> Se W, see <sup>Se</sup>	3E+3 -	1E+4 2E+4	5E-6 7E-6	2E-8 2E-8	4E-5	4E-4
34	Selenium-75	D, see <sup>70</sup> Se W, see <sup>Se</sup>	5E+2	7E+2 6E+2	3E-7 3E-7	1E-9 8E-10	7E-6	7E-5
34	Selenium-79	D, see <sup>70</sup> Se W, see <sup>70</sup> Se	6E+2	8E+2 6E+2	3E-7 2E-7	1E-9 8E-10	8E-6	8E-5
34	Selenium-81m <sup>2</sup>	D, see 70Se W, see 70Se	2E+4	7E+4 7E+4	3E-5 3E-5	9E-8 1F-7	3E-4	3E-3
34	Selenium-81 <sup>2</sup>	D, see <sup>70</sup> Se	6E+4 St. wall	2E+5	9E-5	3E-7	-	-
		W, see <sup>70</sup> Se	(8E+4)	- 2F+5	- 1E-A	25-7	1E-3	1E-2
34	Selenium-83 <sup>2</sup>	D, see 70 W, see 70 Se	3E+4 -	1E+5 1E+5	SE-5 SE-5	2E-7 2E-7	4E-4	4E-3

			0cc	Table 1 Table Occupational Values Efflu Concentra		Table 2 Effluent Concentrations		Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-3 - 5E-3 - 5E-3 - 5E-4
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-3 - 5E-3 - 5E-4 - 2E-3
		선생님은 영상에서 가지 않는 것이 없는 것이 없다.	Ingestion	Inhal	ation			
Atomic No.	Kadionuciide	LIASS	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	
35	Bromine-74m <sup>2</sup>	D bromides of H Li						
		Na, K, Rb, Cs, and Fr	1E+4 St. wall	4 <u>r</u> +4	2E-5	5E-8	-	-
			(2E+4)	-	-		3F-4	3F-3
		W, bromides of lantha- nides, Be, Mg, Ca, Sr, Ba, Ra, Al, Ga, In, Tl, Ge, Sn, Pb, As, Sb, Bi, Fe, Ru, Os, Co, Rh, Ir, Ni, Pd, Pt, Cu, Ag, Au, Zn, Cd, Hg, Sc, Y, Ti, Zr, Hf, V, Nb, Ta, Mn, Tc, and Re		4E+4	2E-5	6F-8	_	
35	Bromine-74 <sup>2</sup>	D, see <sup>74m</sup> Br	2E+4	7E+4	3E-5	1E-7	_	
			St. wall					
		74m	(4E+4)	-	-	-	5E-4	5E-3
		W, see Br	-	8E+4	4E-5	1E-7	-	-
35	Bromine-75 <sup>2</sup>	D, see <sup>74m</sup> Br	3E+4 St. wall	5E+4	2E-5	7E-8	-	-
		74m	(4E+4)	-	-	-	SE-4	5E-3
		W, see ''Br	-	5E+4	2E-5	7E-8	-	-
25	Bromine-76	D	15.0					
	oromine /o	W see 74mBr	4673	St+3 A5+3	25-6	78-9	5E-5	5E-4
		.,		4213	22-0	65-3		-
35	Bromine-77	D, see 74mBr	2E+4	2E+4	1E-5	3F-8	2F-4	2E-3
		W, see Br	-	2E+4	8E-6	3F-R	-	-

			0cc	Table 1 Occupational Values			ble 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Co1. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-3 - 1E-2 - 4E-4 - 9E-3 - - 4E-3 -
Atomic	Radionuclide	Class	ALT	ALL DAC		Air	Water	Average
No.			(µCi)	(µCi)	(µCi/ml)	(µCi/m1)	(µCi/ml)	(µCi/ml)
35	Bromine-80m	D, see 74mBr 74mBr	2E+4	2E+4	7E-6	2E-8	3E-4	3E-3
	김 씨는 아이들은 것이 같아.	w, see or	그는 것이 집 같이 봐.	16+4	65-6	2E-8	-	
35	6romine-80 <sup>2</sup>	D, see <sup>74m</sup> Br	5E+4 St wall	2E+5	8E-5	3E-7	-	-
			(9F+4)	_		-	15-3	15-2
		W, see <sup>/4m</sup> Br		2E+5	9E-5	3E-7	-	-
35	Brominu-82	D con 74mpr	35.2	45.2	AF 6			
	or om the oc	W see 74mBr	36+3	46+3	ZE-6	65-9	4E-5	4E-4
		-,		4673	25-0	58-9		-
35	Bromine-83	D, see <sup>/4m</sup> Br	5E+4 St. wall	6E+4	3E-5	9E-8	-	-
		74-	(7E+4)			_	9E-A	05-30
		W, see "Br		6E+4	3E-5	9E-8	-	-
35	8romine-84 <sup>2</sup>	D, see <sup>74m</sup> Br	2E+4 St. wall	6E+4	2E-5	8E-8	-	-
		74m	(3E+4)	-	-	-	4F-4	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-3 - 1E-2 - 4E-4 - 9E-3 - - 4E-3 -
		W, see Br		6E+4	3E-5	9E-8	-	-
35	Krypton-74 <sup>2</sup>	Submersion <sup>1</sup>	-		3E-6	1E-8	_	-
36	Krypton-76	Submersion <sup>1</sup>			9E-6	4E-8	1	
36	Krypton-77 <sup>2</sup>	Submersion <sup>1</sup>		-	4E-6	2E-8	-	-
36	Krypton-79	Submersion <sup>1</sup>		-	2E-5	75-8	21	
36	Krypton-81	Submersion <sup>1</sup>		-	7E-4	3E-6	1	_

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/m])
36	Krypton-83m <sup>2</sup>	Submersion <sup>1</sup>	1	-	1E-2	5E-5	-	
36	Krypton-85m	Submersion <sup>1</sup>		-	2E-5	1E-7		-
36	Krypton-85	Submersion <sup>1</sup>		-	1E-4	7E-7	-	-
36	Krypton-87 <sup>2</sup>	Submersion <sup>1</sup>		-	5E-6	2E-8		-
36	Krypton-88	Submersion		-	2E-6	9E-9	-	-
37	Rubidium-79 <sup>2</sup>	D, all compounds	4E+4 St. wall	1E+5	5E-5	2E-7	-	-
	2		(6E+4)		1. 6.	-	8E-4	8E-3
37	Rubidium-81m <sup>C</sup>	D, all compounds	2E+5 St. wall	3E+5	1E-4	5E-7	-	-
27			(32+5)	-		-	4E-3	4E-2
31	Rubidium-81	C, all compounds	4E+4	5E+4	2E-5	7E-8	5E-4	5E-3
37	Rubidium-82m	D, all compounds	1E+4	2E+4	7E-6	2E-8	2E-4	2E-3
37	Rubidium-83	D, all compounds	6E+2	1E+3	4E-7	1E-9	9E-6	98-5
37	Rubidium-84	D, all compounds	5E+2	8E+2	3E-7	1E-9	7E-6	7E-5
37	Rubidium-86	D, all compounds	5E+2	8E+2	3E-7	18-9	7E-6	7E-5
37	Rubidium-87	D, all compounds	1E+3	2E+3		2E-9	1E-5	1E-4

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	ALI (uCi)	DAC (uCi/ml)	Air (uCi/ml)	Water (uCi/ml)	Average Concentratio
37	Rubidium-88 <sup>2</sup>	0, all compounds	2E+4 St. wall	6E+4	3E-5	9E-8	-	
			(3E+4)			-	4E-4	48-3
37	Rubidium-89 <sup>2</sup>	D, all compounds	4E+4 St. wall	15+5	6E-5	2E-7	-	-
			(6E+4)		-	-	9E-4	9E-3
38	Strontium-80 <sup>2</sup>	D, all soluble compound except SrTiO <sub>3</sub>	ls 4E+3	1E+4	5E-6	2E-8	6E-5	6E-4
		Y, all insoluble compounds and ${\rm SrTi0}_3$		1E+4	5E-6	2E-8	_	
38	Strontium-81 <sup>2</sup>	D, see 80 Y, see 80 Sr	2E+4	8E+4 8E+4	3E-5 3E-5	1E-7 1E-7	3E-4	3E-3
38	Strontium-82	D, see <sup>80</sup> Sr	2E+2	4E+2	2E-7	6E-10	-	-
		Y, see <sup>80</sup> Sr	(2E+2)	- 9E+1	- 4E-8	- 1E-10	3E-6	3E-5
38	Strontium-83	D, see $\frac{80}{80}$ Sr Y, see Sr	2E+3	7E+3 4E+3	3E-6 1E-6	1E-8 5E-9	3E-5	3E-4
38	Strontium-85m <sup>2</sup>	D, see $\frac{80}{80}$ Sr Y, see $\frac{80}{5}$ Sr	2E+5	6E+5 8E+5	3E-4 4E-4	9E-7	3E-3	3E-2
38	Strontium-85	D, see $\frac{80}{80}$ Sr Y, see $\frac{80}{5}$ r	3E+3	3E+3 2F+3	1E-6 6E-7	4E-9	4E-5	4E-4
38	Strontium-87m	D, see $\begin{array}{c} 80\\80\\sr\\Y, see \end{array}$ Sr	4E+4	1E+5 2E+5	5E-5 6E-5	2E-7 2E-7	6E-4	68-3

			Table 1 Occupational Values			Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 8E-5 - 5E-6 - 2E-4 - 3E-3 - 2E-4 - 3E-3 - 3E-3
			Col. 1 Oral Ingestion	Col. 2 Inhala	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 8E-5 - 5E-6 - 2E-4 - 3E-3 - 2E-4 - 3E-3 - 1E-4
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
38	Strontium-89	D, see <sup>80</sup> Sr	5E+2 LLI wall	9E+2	4E-7	1E-9	-	-
		Y, see <sup>80</sup> Sr	(6E+2)	- 1E+2	- 6E-8	- 2E-10	8E-6 -	8E-5
38	Strontium-90	0, see <sup>80</sup> Sr	3E+1 Bone surf	2E+1 Bone surf	8E-9	-	-	-
		Y, see <sup>80</sup> Sr	(4E+1) -	(2E+1) 4E+0	- 2E-9	3E-11 6E-12	5E-7	5E-6
38	Strontium-91	D, see $\frac{80}{80}$ Sr Y, see Sr	2E+3	6E+3 4E+3	2E-6 1E-6	8E-9 5E-9	2E-5	2E-4
38	Strontium-92	D, see 80 Y, see	3E+3	9E+3 7E+3	4E-6 3E-6	1E-8 9E-9	4E-5	4E-4
39	Yttrium-86m <sup>2</sup>	W, all compounds except those given for Y	2E+4	6E+4	2E-5	8E-8	3E-4	3E-3
		Y, oxides and hydroxides		5E+4	2E-5	8E-8		
39	Yttrium-86	W, see 86my Y, see 86my	1E+3 -	3E+3 3E+3	1E-6 1E-6	5E-9 5E-9	2E-5	2E-4
39	Yttrium-87	W, sec 86my Y, see 86my	2E+3 -	3E+3 3E+3	1E-6 1E-6	5E-9 5E-9	3E-5	3E-4
39	Yttrium-88	W, see $\frac{86m_Y}{86m_Y}$ Y, see	1E+3 -	3E+2 2E+2	1E-7 1E-7	3E-10 3E-10	1E-5	1E-4
39	Yttrium-90m	W, see $\frac{86m_Y}{86m_Y}$	8E+3	1E+4 1E+4	5E-6	2E-8 2E-8	1E-4	1E-3

			0cc	Table 1 Occupational Values			ble 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 7E-5 - 2E-2 - 8E-5 - 4E-4 - 2E-4 - 4E-3 - 7E-3
			Ingestion Inhalation			Avecage		
Atomic No.	Radionuclide	Class	ALĨ (µCi)	ALI (µCi)	DAC (µCi/m?)	Air (pCi/mł)	Water (µCi/ml)	Concentration (µCi/ml)
39	Yttrium-90	W, see <sup>86m</sup> y	4E+2 LLI wall	7E+2	3E-7	9E-10	-	
		96	(5E+2)				75-6	75-5
		Y, see oomy	그 가슴이 물길	6E+2	3E-7	9E-10	-	-
20	Vttrium-01m2	85m.,		1.2.2.1.1.1				
33	ICCI IGH-3TW	w, see 86m,	1E+5	2E+5	1E-4	3E-7	2E-3	2E-2
		r, see r		2E+5	7E-5	2E-7	- 10 A A A A A A A A A A A A A A A A A A	2010 Sec. 20
39	Yttrium-91	W, see <sup>86m</sup> Y	5E+2 LLI wall	2E+2	7E-8	2E-10		
		86m	(6E+2)				RE-6	Monthly Average Concentration (µCi/m1) - 7E-5 - 2E-2 - 8E-5 - 4E-4 - 2E-4 - 4E-4 -
		Y, see Y		1E+2	5E-8	2E-10	-	- OL -
39	Yttrium-92	W, see $\frac{86m_{Y}}{86m_{Y}}$	3E+3	9E+3 8E+3	4E-6 3E-6	1E-8	4E-5	4E-4
		96-		OL.S	36 0	16-9		
39	Yttrium-93	W, see per	1E+3	3E+3	15-6	45-9	25-5	25.4
		Y, see Y	÷	2E+3	1E-6	35-9	26.3	28-4
39	Yttrium-94 <sup>2</sup>	₩, see <sup>86m</sup> γ	2E+4	8E+4	3E-5	1E-7	-	
		한 것 같은 것 같은 것 같은 것 같이 없다.	(3E+A)					
		Y, see <sup>86m</sup> Y	(30.14)	95+3	25.5		4E-4	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 7E-5 - 2E-2 - 8E-5 - 4E-4 - 2E-4 - 4E-3 - 7E-3
	2	- 20		OC.4	36-5	3.E-1		
39	Yttrium-95 <sup>2</sup>	₩, see <sup>com</sup> Y	4E+4 St. wall	2E+5	6E-5	2E-7	-	-
		86m	(5E+4)	18. S. P. M.	<ul> <li>- 111112</li> </ul>	-	7F-4	Releases to Sewers Monthly Average Concentratio (µCi/ml) - 7E-5 - 8E-5 - 8E-5 - 2E-4 - 2E-4 - 4E-4 - 7E-3 - 7E-3
		Y, see Y		1E+5	6E-5	2E-7	-	-

			0ce	Table 1 cupational Va	lues	Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 2E-4 - 5E-4 - 2E-4 - 4E-4 - 2E-4 - 2E-4
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	
			Ingestion Inhalation				Average	
Atomic No.	Radionuclide	Class	ALĬ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
40	Zirconium-86	D. all compounds except						
		those given for W and Y	1E+3	4E+3	2E-6	6E-9	2E-5	2E-4
		W. cxides, hydroxides.						
		halides, and nitrates		3E+3	1E-6	4E-9	-	
		Y, carbide		2E+3	1E-6	3E-9		-
40	7irconium-88	D 500 8675	45.42	25.12	05.0	35. 10		
10	LITCOTTUM 00	U, See 8627	4673	21+2	91-8	3E-10	58-5	5E-4
		Y, see 86Zr	2.101.0	3E+2	2E-7 1E-7	/E-10 4E-10	2	-
					2.C /	46 10		
40	Zirconium-89	D, see 867r	2F+3	4F+3	1E-6	55-0	25-5	25-4
		W. see 857r		2E+3	15-6	35-0	21. 3	20-4
		Y, see <sup>86</sup> Zr		2E+3	1E-6	3E-9	-	-
40	7irconium 93	D see 867r	15+3	0422	25-0			
	211001110. 30	0, 300 21	Bone surf	Bone surf	35-3			
		00	(3E+3)	(2E+1)	-	2F-11	45-5	AF-A
		W, see <sup>oo</sup> Zr	-	2E+1	1E-8	-	-	-
				Bone surf				
		20	-	(6E+1)	-	9F-11	-	_
		Y, see Zr	10 - 10 C	6E+1	2E-8	-		-
				Bone surf				Average Concentration (μCi/ml) 2E-4 - 5E-4 - 2E-4 - 4E-4 - 2E-4 - 2E-4
				(7E+1)	-	9E-11	-	- 1
40	Tirconium-05	0 600 8670	15.0	17.0				
	Enconium 33	u, see Li	16+3	It+2 Recent	5E-8	-	2E-5	2E-4
			<ul> <li>200 Table 1</li> </ul>	cone surt				
		W 500 867		(3E+Z)	-	4E-10	-	-
		, see 867	14 D.A. Marson	46+2	2E-1	5E-10	uent rationsReleases to SewersCol. 2Monthly Average Concentration (µCi/ml)Water (µCi/ml)2E-52E-52E-45E-55E-42E-52E-42E-52E-42E-52E-42E-52E-4 <td< td=""><td>-</td></td<>	-
		r, see Zr		3E+2	16-7	4E-10	-	-

			0cc	Table 1 upational N	/alues	Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
		방법 이 방법 이 있는 것이 있는 것이 없다.	Ingestion	Inhal	lation			Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/m1)
40	Zirconium-97	D. see 86Zr	6E+2	2F+3	8F-7	3E-9	9:-6	95-5
		W. see orZr		1E+3	6F-7	2F-9	-	-
		Y, see <sup>Br</sup> Zr		1E+3	5E-7	2E-9	-	-
41	Nichium-882	W all compands event						
	NTOD TOM GO	those given for Y	SE+4 St wall	2E+5	9E-5	3E-7	-	-
			(7F+4)	12.00		-	15-2	15.0
		Y, oxides and hydroxides		2E+5	9E-5	3E-7	10 3	1C-2
41	Nichi - 002	88						
41	NIODIUM-89	W, see 88ND	1E+4	4E+4	2E-5	6E-8	1E-4	1E-3
	(00 min)	r, see ND		4E+4	2E-5	5E-8	-	-
41	Niobium-89	W. see 88Nb	5F+3	2F+4	85-6	9-35	75.5	75.4
	(122 min)	Y, see <sup>88</sup> Nb	-	2E+4	6E-6	2E-8	76-5	72-4
41	Nichium-00	88						
41	NTOD FUIL 90	W, see 88 <sup>ND</sup>	16+3	3E+3	1E-6	4E-9	1E-5	1E-4
		r, see ND		2E+3	1E-6	3E-9	-	
41	Niobium-93m	W, see <sup>88</sup> Nb	9E+3 LLI wall	2E+3	8E-7	3E-9	-	*
		00	(1E+4)	-	-		25-4	25-2
		Y, see <sup>88</sup> Nb	-	2E+2	7E-8	2E-10	26-4	££-3
41	Nichium 04							
41	N10D1Um-94	w, see 88 <sup>ND</sup>	9E+2	2E+2	8E-8	3E-10	1E-5	1E-4
		T, SEE ND		2E+1	6E-9	2E-11	-	-
41	Niobium-95m	W, see <sup>88</sup> Nb	2E+3 LLI wall	3E+3	18-6	4E-9	-	-
		99	(2E+3)	-		-	38-5	35-4
		Y, see Nb	-	2E+3	9E-7	3F-9	-	-

	Radionuclide	Class	0cc	lalues	Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Co1. 3	Col. 1	Col. 2	Monthly Average Concentratio (uCi/ml)
Atomic No.			Ingestion ALI (µCi)	Inhal ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	
41	Niobium-95	W, see <sup>88</sup> Nb Y, see <sup>88</sup> Nb	2E+3	1E+3 1E+3	5E-7 5E-7	2E-9 2E-9	3E-5	3E-4
41	Niobium-96	W, see <sup>88</sup> Nb Y, see <sup>88</sup> Nb	1E+3	3E+3 2E+3	1E-6 1E-6	4E-9 3E-9	2E-5	2E-4
41	Niobium-97 <sup>2</sup>	W, see <sup>88</sup> Nb Y, see Nb	2E+4	8E+4 7E+4	3E-5 3E-5	1E-7 1E-7	3E-4	3E-3
41	Niobium-98 <sup>2</sup>	W, see <sup>88</sup> Nb Y, see <sup>88</sup> Nb	1E+4	5E+4 5E+4	2E-5 2E-5	8E-8 7E-8	2E-4	2E-3
42	Molybdenum-90	D, all compounds except those given for Y	2E+3	7E+3	3E-6	1E-8	3E-5	3E-4
		Y, oxides, hydroxides, and $MoS_2$		5E+3	2E-6	6E-9	-	-
42	Molybdenum-93m	D, see <sup>90</sup> Mo Y, see <sup>90</sup> Mo	4E+3	2E+4 1E+4	7E-6 6E-6	2E-8 2E-8	6E-5 -	6E-4
42	Molybdenum-93	D, see $90_{Mo}$ Y, see $90_{Mo}$	4E+3	5E+3 2E+2	2E-6 8E-8	8E-9 2E-10	5E-5	5E-4
42	Molybdenum-99	D, see <sup>90</sup> Mo	1E+3 LLI wall	3E+3	1E-6	4E-9	-	-
		Y, see <sup>90</sup> Mo	(1E+3) -	- 1E+3	- 6E-7	- 2E-9	2E-5	2E-4 -
42	Molybdenum-101 <sup>2</sup>	D, see <sup>90</sup> Mo	4E+4 St. wall	1E+5	6E-5	2E-7	-	-
		Y, see <sup>90</sup> Mo	(5E+4) -	- 1E+5	- 6E-5	- 2E-7	7E-4	7E-3

			0cc	Table i cupational V	alues	Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly Average Concentration (uCi/ml)
			Ingestion	Inhala	ation			
Atomic No.	Radionuclide	Class	ALĬ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	
12	Tachasting 02-2	0						
43	recimet rum-93m	U, all compounds except those given for W	7E+4	2E+5	6E-5	2E-7	1E-3	1E-2
		W axides hydroxides						
		halides, and nitrates		3E+5	1E-4	4E-7	-	-
43	Technetium-93	D see 93m	25+4	75.4				
	recimectum 33	W see 93m Tc	35+4	15+5	35-5	18-7	4E-4	4E-3
	같은 물건 것 같은 것.	w, see it		16+3	46-0	IE-/		-
43	Technetium-94m <sup>2</sup>	D, see 93mTc	2F+4	AE+A	25-5	65.0	25.4	AF. 4
		W, see <sup>93m</sup> Tc	-	6E+4	2F-5	RF-R	35-4	3E-3
		0.2-				or o		
43	Technetium-94	D, see gamTc	9E+3	2E+4	8E-6	3F-8	1F-4	15-3
		W, see Tc		2E+4	1E-5	3E-8	-	*
43	Technot ium-05m	D						
15	recuner rum-30W	W sop 93mTc	42+3	5E+3	2E-6	8E-9	5E-5	5E-4
		", see it	이야 한 말 같이.	2E+3	8E-1	3E-9	-	-
43	Technetium-95	D. SPP 93mIC	15+4	25+4	05.0	25.0		
		W, see <sup>93m</sup> Tc	-	25+4	95-6	31-8	1E-4	1E-3
				22.14	05-0	35-8		-
43	Technetium-96m <sup>2</sup>	D, see 93 ic	2E+5	3E+5	1E-4	AE-7	25-2	05.0
		W, see <sup>93m</sup> Tc		2E+5	1E-4	35-7	20-3	25-1
<u>z 18</u> .		02-		~~ ~	10.4	JE-/		
43	Technetium-96	D, see 02-Tc	2E+3	3E+3	1F-6	55-9	25-5	35-4
		W, see <sup>55m</sup> Tc		2E+3	9E-7	3E-9	-	32-4
4.2	T 1 11 11	93m						
43	lechnetium-97m	D, see Tc	5E+3	7E+3	3E-6	-	6E-5	6F-4
				St. wall				
		93m-	17 B B B B B	(7E+3)		1E-8	-	-
		w, see Ic		1E+3	5E-7	2E-9	-	A

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1	Col. 2 Col. 3 Inhalation		Col. 1 Col	Col. 2	Monthly
			Ingestion					Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/m1)
43	Technetium-97	D. see 93mTc	4E+4	5E+4	2E-5	7E-8	5E-4	5E-3
15		W, see <sup>93m</sup> Tc		6E+3	2E-6	8E-9	-	-
43	Technetium-98	D see 93m	15+3	2E+3	7E-7	2E-9	1E-5	1E-4
+3	recime crum 50	W, see 93mTc		3E+2	1E-7	4E-10	-	-
43	Technotium-99m	D see 93m	8F+4	25+5	6E-5	2E-7	1E-3	1E-2
43	rechnecium 55m	W, see 93mTc	-	2E+5	1E-4	3E-7	-	-
13	Tachaatium-00	D con 93m	4F+3	55+3	2F-6	_	6F-5	6F-4
43	recime trum 33	D, see ic	46.5	St. wall	ee o		OL J	OL 4
		말 이번 방법에서 집에 가지?		(6E+3)	-	8E-9	-	-
		W, see <sup>93m</sup> Tc	같은 아이지	7E+2	3E-7	9E-10	-	-
43	Technotium-1012	D 500 93m	QE+4	35+5	1F-4	5F-7	_	
43	recimectum 101	b, see ic	St wall	32.5	16 4	JE /		
			(1E+5)	-	-	-	2F-3	2E-2
		W, see <sup>93m</sup> Tc	-	4E+5	2E-4	5E-7	+	-
12	Technotium 1042	D	25+4	75+4	25-5	15-7		
+3	rechner rum- 104	D, see ic	Ct wall	15.14	SE S	TE-1		
			(3E+A)	-		_	45-4	45-3
		W, see <sup>93m</sup> Tc	-	9E+4	4E-5	1E-7	-	-
	Puthanium-042	D all compands arcent						
**	Auchen rum-34	those given for W and Y	2E+4	4E+4	2E-5	6E-8	2E-4	2E-3
		N. K. Mider		15.4	ar. r	05.0		
		w, nailoss	. E 14 S	02+4	35-5	95-8		
		1, oxides and hydroxides		01+4	25-2	8E-8		-

Atomic No.				Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
	Radionuclide	dionuclide Class	Col. 1 Oral	Col. 2 Col. 3		Col. 1	Col. 2	
			Ingestion AL1 (µCi)	Inhal ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (uCi/ml)
44	Ruthenium-97	D. SPP 94Ru	8E+3	2E+A	95-5	25-0	15.4	15.2
		W SPP 94Ru	-	15+4	55-6	25-0	15-4	1E-3
		Y, see Ru	-	1E+4	5E-6	2E-8	-	2
44	Puthonium-102	942	AF					
44	Kuchentum-103	U, see 94 Ru	2E+3	2E+3	7E-7	2E-9	3E-5	3E-4
		w, see 94 <sup>Ku</sup>		1E+3	4E-7	1E-9	-	
		r, see Ru		6E+2	3E-7	9E-10	-	
44	Ruthenium-105	D. see 94Ru	5F+3	1F+4	6F-6	25-9	75.6	75.4
		W. see S.Ru	-	1E+4	6E-6	20-0	15-2	/E-4
		Y, see <sup>94</sup> Ru	-	1E+4	5E-6	2E-8	2	-
44	Ruthenium-106	D, see <sup>94</sup> Ru	2E+2 LLI wall	9E*1	4E-8	1E-10	-	-
		04	(2E+2)	-	1 a	-	35-6	2-36
		W, see orRu	-	5E+1	2F-8	8F-11	50 0	36-3
		Y, see <sup>34</sup> Ru	-	1E+1	5E-9	2E-11	-	-
45	Rhodium-99m	D all compounds except						
		those given for W and Y	28+4	6E+4	2E-5	8E-8	2E-4	2E-3
		W, halides	-	8F+4	35-5	15-7		
		Y, oxides and hydroxides	-	7E+4	3E-5	9E-8	-	2
45	Rhodium-99	D coo 99mpt	05.0					
1.5	torourum 33	W see 99mph	2E+3	3E+3	16-6	4E-9	32-5	3E-4
		V, see 99mph		ZE+3	9E-7	3E-9	-	-
		r, see Kn	-	2E+3	8E-7	3E-9	-	-

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Co1. 3	Col. 1	Col. 2	Monthly
			Ingestion	Inhal	lation			Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (uCi/ml)	Concentration (µCi/ml)
45	Rhodium-100	D, see 99mRh	2E+3	5E+3	2E-6	7E-9	2E-5	2E-4
		W, see gamRh	영상 이 동안에서 가지	4E+3	2E-6	6E-9	-	-
		Y, see <sup>35m</sup> Rh	이 아파 가지가?	4E+3	2E-6	5E-9	-	-
45	Rhodium-101m	D, see 99mRh	6F+3	1F+4	5E-6	2F-8	8E-5	95-4
		W. see 99mRh		8F+3	4E-6	1E-8	-	00.4
		Y, see 99mRh	그 그 옷을 깨끗했는	8E+3	3E-6	1E-8	-	-
45	Rhodium-101	D see 99mph	25+3	55+2	25-7	75-10	35-55	25-4
		W see 99mRh	-	8E+2	25-7	15-0	36-3	36-4
		Y, see <sup>99m</sup> Rh		2E+2	6E-8	2E-10	-	2
45	Phodium-102m	D coo 99mph	15.2		AF 7	75.10		
***	KHOUTUM 102M	D, See Ri	10+3	24-36	2E-1	/E-10	-	
			(1E+2)	1. S.			ar . r	
		W coo 99mph	(10+3)	45.0	05 7		28-5	71-1
		V see 99mph		45+2	21-1	5E-10		-
		i, see Rii		16+2	5E-8	2E-10	-	5
45	Rhodium-102	D. see 99mRh	6F+2	QE+1	4E-8	15-10	05-5	05-5
	ALL	W. see 99mRh	-	2E+2	7E-8	25-10	05-0	C-30
		Y, see 99mRh		6E+1	2E-8	8E-11	-	2
**	n:	99m-						
45	Khodium-103m	D, see 99mRh	4E+5	1E+6	5E-4	2E-6	6E-3	6E-2
		W, see 99m Rh		1E+6	5E-4	2E-6	-	-
		Y, see Rh		1E#6	5E-4	2E-6	-	
45	Rhodium-105	D, see 99mRh	4E+3	1F+4	55-6	2F-8	1.1.1.1	
			LLI wall	~	JE U	ee o		
		00-	(4F+3)	-	-	-	55-5	55-4
		W, see on Rh	-	6F+3	35-6	9E-9	56.5	36-4
		Y, see 99mRh	-	6E+3	25-6	85-9		
				0 L 0	LL U	OL J		
				Table 1 Occupational Values			ble 2 luent trations	Table 3 Releases to Sewers
--------	--------------------------	--------------------------	----------------	--------------------------------	-----------------	-----------------	----------------------------	-------------------------------------------------------------------------------------------------------------------------------------------
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic	Radionuclide	Class	ingestion	15001	Inhalation			Average
No.	nuo i onaci nac	61033	(µCi)	(µCi)	UAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (uCi/ml)
45	Rhodium-106m	D Sep 99mpb	8E+3	25+4	15-5	AF 0	15.4	15.2
		W see 99mph	01.13	AE +A	10-5	46-8	11-4	12-3
		V coo 99mph	이 그렇게 많다.	45+4	28-5	51-8	-	-
		, see an		45+4	16-2	5E-8	-	-
45	Rhodium-107 <sup>2</sup>	D see 99mph	75+4	25.45	15.4			
	terrourium 201	o, see an	7E+4	2640	IE-4	31-1		-
			SL. Wall					
		W coo 99mph	(96+4)	-	-		1E-3	1E-2
		W, see 99mon	-	32+5	1E-4	4E-7	-	*
		1, see Kn	1121112	3E+5	1E-4	3E-7	-	-
46	Palladium-100	D, all compounds except						
		those given for W and Y	1E+3	1E+3	6E-7	2E-9	2E-5	2E-4
		W. nitrates	-	15+3	55-7	25.0		
		Y, exides and hydroxides	-	15+2	55-7	28-9	-	-
		, on see one nyeron rues		16+3	DE-1	2E-9	-	-
46	Palladium-101	D. see 100 Pd	15+4	75+8	15.5	FF 0		
		W see 100pd	12.14	35+4	16-5	SE-8	2E-4	2E-3
		Y see 100pd		35+4	16-5	5E-8	-	-
		r, see ru	이 아들은 보이는	35+4	15-2	4E-8	-	*
46	Palladium-103	D see 100pd	6543	65.2	25.6			
	Torreation 200	0, 500 10	111	6E+3	35-6	9E-9	-	-
			(75.12)					
		W con 100pd	(12+3)	-		÷	1E-4	1E-3
		W, see 100 <sup>P0</sup>		4E+3	2E-6	6E-9	-	Releases to Sewers Monthly Average Concentration (µCi/ml) IE-3 - 2E-4 - 2E-4 - IE-3 - 5E-3 -
		r, see Pd		4E+3	1E-6	5E-9	-	with 1997
46	Palladium-107	D 100pd						
10	(d) (d) (d) (10) (10)	u, see Pa	3E+4	2E+1	9E-6	-	-	-
			LLI wall	Kidneys				
		100-	(4E+4)	(2E+4)	-	3E-8	5E-4	58-3
		W, see 100Pd		7E+3	3E-6	1E-8	-	-
		Y, see Pd		4F+2	2F-7	6E-10		

			Table 1Table 2Occupational ValuesEffluentConcentrations				Table 3 Releases to Sewers Monthly Average Concentration (µCi/m1) 3E-4 - 9E-3 - 5E-3 - 4E-3 - 3E-3	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-4 - 9E-3 - 5E-3 - 4E-3 - 3E-3 - 4E-3 - 4E-4
			Ingestion Inhala		ation			Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
46	Palladium-109	D. see 100rd	2F+3	6F+3	35-6	QE-9	35-5	35-4
		W see 100Pd	-	55+3	25-6	05-0	52 5	JC 4
		Y, see <sup>100</sup> Pd		5E+3	2E-6	6E-9	2	1
47	5:1							
4/	511ver-102	D, all compounds except						
		those given for W and Y	5E+4 St. wall	2E+5	8E-5	2E-7	-	-
			(6E+4)	-		-	9E-4	9E-3
		W, nitrates and sulfides	-	2E+5	9E-5	3E-7	-	-
		Y, oxides and hydroxides	-	2E+5	8E-5	3E-7	4	
47	2002	102.						
4/	511ver-103	D, see 102Ag	4E+4	1E+5	4E-5	1E-7	5E-4	5E-3
		W, see 162Ag	-	1E+5	5E-5	2E-7	-	-
		Y, see YAg	-	1E+5	5E-5	2E-7	-	-
47	Silver-104m <sup>2</sup>	0 500 102	25+4	05+4	AT . F			
	Contract a contract	W co. 102Ag	3214	35+4	46-5	11-1	41-4	4E-3
		V SUD 102AG		16+3	52-5	ZE-/	-	-
	그는 아이들을 많다.	i, she ng		16+2	5E-5	2E-1	*	
47	Silver-104 <sup>2</sup>	D see 102Ag	28+4	75+5	25-5	15.7		ar
		W see 102Ag	-	1545	55-5	16-7	SE 4	3E-3
		Y, see <sup>102</sup> Ag	-	16+5	6E-5	2E-7 2E-7	-	2
		102				Sec. 1.		
47	Silver-105	D, see 102Ag	3E+3	1E+3	4E-7	1F-9	4F-5	QE-4
		W, see 102Ag	-	2E+3	7E-7	2F-9	-	-
		Y, see TOZAg	-	2F+3	7F-7	25-9	12.11.11.11.1	and the second second

			Table 1 Occupational Values			Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-4 - 9E-3 - 9E-5 - 6E-5 - - 2E-4 - 4E-4
	김 씨는 것이 많이 많이 봐.		Ingestion	Inhalation				Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
47	Silver-106m	D, see 102Ag	8E+2	7F+2	35-7	15-9	15-5	16-4
		W, see 102Ag		9F+2	45-7	16-0	IL J	16-4
		Y, see <sup>102</sup> Ag	그는 작품을 못했다.	9E+2	4E-7	1E-9	-	-
47	Silver-106 <sup>2</sup>	D, see <sup>102</sup> Ag	6E+4 St. wall	2E+5	8E-5	3E-7	-	-
			(6F+4)			1. St. 11 - 1	05-4	05.0
		W. SPP 102Ag		25+5	05-5	35.7	96-4	96-3
		Y, see <sup>102</sup> Ag		2E+5	8E-5	3E-7 3E-7	2	2
47	Silver-108m	D, see 102 W, see 102Ag Y, see 102Ag Y, see Ag	6E+2 -	2E+2 3E+2 2E+1	8E-8 1E-7 1E-8	3E-10 4E-10 3E-11	9E-6 -	9E-5
47	Silver-110m	D 102.						
**	211461 - TTOW	U, see 102Ag	5E+2	1E+2	5E-8	2E-10	6E-6	6E-5
		w, see 102 <sup>Ag</sup>		2E+2	8E-8	3E-10	-	-
		r, see Ag		9E+1	4E-8	1E-10	-	-
47	Silver-111	D, see <sup>102</sup> Ag	9E+2 LLI wall	2E+3 Liver	6E-7	-	-	-
		102	(1E+3)	(2E+3)	-	2E-9	25-5	26-4
		W, see 102Ag		9E+2	4F-7	15-9	-	20-4
		Y, see TOZAg		9E+2	4E-7	1E-9	-	-
47	Silver-112	D, see 102Ag	3E+3	8E+3	3E-6	1E-8	4E-5	4E-4
		W, see 102Ag		1E+4	4E-6	1E-8	-	-
		r, see Ag		9E+3	4E-6	1E-8	-	-

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 4E-3 - 3E-3 - - 3E-3 - - 6E-5 -
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/ml)
47	Silver-115 <sup>2</sup>	D, see <sup>102</sup> Ag	3E+4 St. wali	98+5	4E-5	18-7	-	-
		102	(3E+4)	-	-	-	4E-4	4E-3
		W, see 102Ag	-	9E+4	4E-5	1E-7	-	-
		Y, see <sup>102</sup> Ag		8E+4	3E-5	1E-7	-	
48	Cadmium-104 <sup>2</sup>	D, all compounds except those given for W and Y	2E+4	7E+4	3E-5	9E-8	3E-4	3E-3
		W, sulfides, halides, and nitrates		1E+5	5E-5	2E-7	-	-
		Y, oxides and hydroxides		1E+5	5E-5	2E-7	-	-
48	Cadmium-107	D, see 104Cd	. 5+4	5E+4	2E-5	8F-8	3F-4	35-3
		W, see 104Cd	-	6E+4	2E-5	8F-8	-	-
		Y, see 104Cd	-	5E+4	2E-5	7E-8	-	Table 3 Releases to Sewers Monthly Average Concentratio (µCi/ml) - 4E-3 - 3E-3 - - 6E-5 -
48	Cadmium-109	D, see <sup>104</sup> Cd	3E+2 Kidneys	4E+1 Kidneys	1E-8	-	-	-
		104	(4E+2)	(5£+1)	-	7E-11	6E-6	6E-5
		W, see <sup>10</sup> Cd		1E+2 Kidneys	5E-d	-	-	*
		104	-	(1E+2)	-	2E-10	12	-
		Y, see Cd		1E+2	5E-8	2E-10	-	-

			000	Table 1 Occupational Values			ble 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 5E-6 - - 4E-6 - - 4E-6 - - 4E-5 - - 1E-4 - - 6E-4
Atomic	Padionuclida	Class	Ingestion	Inhal	ation	on		Average
No.	Radionacii de	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
48	Cadmium-113m	D, see <sup>104</sup> Cd	2F+1	2E+0	1F-9	_		
			Kidneys	Kidneys	10 5			
		104	(4E+1)	(4E+0)	-	5E-12	5F-7	58-6
		W, see <sup>104</sup> Cd		8E+0 Kidneys	4E-9	-	-	-
		104	1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 - 1998 -	(1F+1)	-	2E-11	_	
		Y, see <sup>104</sup> Cd		1E+1	5E-9	2E-11	=	-
48	Cadmium-113	D, see <sup>104</sup> Cd	2F+1	2E+0	9F-10			19. The 19.
			Kidnevs	Kidneys	JC 10			
		104	(3E+1)	(3F+0)		5E-12	15-7	AT-C
		W, see <sup>104</sup> Cd		8E+0 Kidneys	3E-9	-	-	-
		104		(1E+1)	-	2E-11	_	
		Y, see <sup>104</sup> Cd		1E+1	6E-9	2E-11	-	-
48	Cadmium-115m	D, see <sup>104</sup> Cd	3E+2	5E+1 Kidneys	2E-8	-	4E-6	4E-5
		104		(8E+1)	-	1E-10	-	-
		W, see 104Cd	-	1E+2	5E-8	2E-10	1.1	-
		Y, see <sup>10</sup> Cd		1E+2	6E-8	2E-10	-	-
48	Cadmium-115	D, see <sup>104</sup> Cd	9E+2	1E+3	6E-7	2E-9	-	
			(1F+3)	_			15.6	15.4
		W, see 104Cd	(AC 0)	1F+3	5E-7	25-9	16-2	16-4
		Y, see <sup>104</sup> Cd	-	1E+3	6E-7	2E-9	-	-
48	Cadmium-117m	D. see 104	5E+3	15+4	EE-C	25.0		
		W, see 104Cd	52.5	25+4	75-6	25-0	01-3	bt-4
		Y, see 104 Cd	경제 등 유지 상태의 역	16+4	65-6	25-8		
				10.44	05-0	ZE-8	-	

			0cc	/alues	Tat Eff Concent	ole 2 luent trations	Table 5 Releases to Sewers Monthly Average Concentration (µCi/ml) bE-4 - 3E-3 - 7E-4 - 6E-4 - 2E-2 - 7E-3 -	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) bE-4 - 3E-3 - 2E-3 - 7E-4 - 6E-4 - 2E-2 - 7E-3 - 5E-5 -
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	
Atomic	Padionuclide	Class	Ingestion ALL	ALT DAC		Air	Water	Concentration
No.	Radionacinae	61855	(µCi)	(µCi)	(µCi/ml)	(µCi/ml)	(µCi/m1)	(µCi/ml)
48	Cadmium-117	D see 104 d	58+3	1F+4	5F-6	2F-8	6E-5	bF-4
10	CCONTON III	W see 104Cd	-	2E+4	7E-6	2E-8	-	*
		Y, see <sup>104</sup> Cd		1E+4	6E-6	2E-8	-	-
49	Indium-109	D. all compounds except						
		those given for W	2E+4	4E+4	2E-5	6E-8	3E-4	3E-3
		W. oxides, hydroxides.						
		halides, and nitrates	88 <b>-</b>	6E+4	3E-5	9E-8	-	-
49	Indium-110 <sup>2</sup>	0, see 109In	2E+4	4E+4	2E-5	6E-8	2E-4	2E-3
	(69.1 min)	W, see <sup>109</sup> In		6E+4	2E-5	8E-8	-	F
49	Indium-110	0. see 109 In	5E+3	2F+4	7E-6	2E-8	7E-5	7F-4
	(4.9 h)	W, see <sup>109</sup> In	C	2E+4	8E-6	3E-8	-	-
49	Indium-111	D, see 109In	4E+3	6E+3	3E-6	95-9	6.5	6F-4
		W, see <sup>109</sup> In		6E+3	3E-6	9E-9	-	-
49	Indium-112 <sup>2</sup>	D. see 109	2E+5	6E+5	3E-4	9E-7	2F-3	2E-2
		W, see <sup>109</sup> In	-	7E+5	3E-4	1E-6	-	-
49	Indium-113 <sup>2</sup>	D. see 109 In	5F+4	1F+5	6E-5	2F-7	(F-A	75-3
		W, see <sup>109</sup> In	-	2E+5	8E-5	3E-7	-	-
19	Indium-114m	D. see <sup>109</sup> In	3E+2	6F+1	3E-8	9F-11		2
			LLI wall					
		109.	(4E+2)	-	-	-	5E-6	5E-5
		W, see In		1E+2	4E-8	1E-10	-	-
19	Indium-115m	D, see 109In	1E+4	4E+4	2E-5	6E-8	2E-4	2E-3
		W, see 105 In	-	5E+4	2E-5	7E-8	-	-

•

•

				Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col 1 Oral	Col. 2	Co1. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 5E-6 - 3E-3 - 2E-3 - 8E-3 - 7E-3 - 5E-4
Atomic No.	Radionuclide	Class	Ingestion AL1 (µCi)	Inhal ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
49	Indium-115	D, see <sup>109</sup> In W, see In	4E+1	1E+0 5E+0	6E-10 2E-9	2E-12 8E-12	5E-7	5E-6
49	Indium-116m <sup>2</sup>	D, see <sup>109</sup> In W, see <sup>109</sup> In	2E+4	8E+4 1E+5	3E-5 5E-5	1E-7 25-7	3E-4	3E~3
49	Indium-117m <sup>2</sup>	D, see $109_{109}$ In W, see In	1E+4	3E+4 4E+4	1E-5 2E-5	5E-8 6E-8	2E-4	2E-3
49	Indium-117 <sup>2</sup>	D, see <sup>109</sup> In W, see <sup>109</sup> In	6E+4	2E+5 2E+5	7E-5 9E-5	2E-7 3F-7	8E-4	8E-3
49	indium-119m <sup>2</sup>	D, see <sup>109</sup> In	4E+4 St. wall	1E+5	5E-5	2E-7	-	-
		W, see <sup>109</sup> In	(5E+4) -	- 1E+5	- 6E-5	- 2E-7	7E-4	7E-3
50	Tin-110	D, all compounds except those given for W	4E+3	1E+4	5E-6	2E-8	5E-5	5E-4
		W, sulfides, oxides, hydroxides, halides, nitrates, and stannic						
50	Tin-111 <sup>2</sup>	D, see 110 <sub>Sn</sub> W, see Sn	7E+4	2E+5 3E+5	5E-6 9E-5 1E-4	2E-8 3E-7 4F-7	1E-3	1E-2

			Table 1 Occupational Values			Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-4 - 3E-4 - 5E-4 - 5E-4 - 8E-4 - 7E-3	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-4 - 3E-4 - 5E-4 - 5E-4 - 8E-4 - 7E-3 - 9E-5	
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)	
50	Tin-113	D, see <sup>110</sup> n	2E+3	1E+3	5E-7	2E-9	-	-	
		W, see <sup>110</sup> Sn	(2E+3)	- 5E+2	- 2E-7	- 8E-10	35-5	3E-4	
50	Tin-117m	D, see <sup>110</sup> Sn	2E+3	1E+3 Rone surf	5E-7		-		
		W, see <sup>110</sup> Sn	(2E+3)	(2E+3) 1E+3	- 6E-7	3E-9 2E-9	3E-5 -	3E-4	
50	Tin-119m	D, see <sup>110</sup> Sn	3E+3	2E+3	1E-6	3E-9	-	-	
		W, see <sup>110</sup> Sn	(4E+3)	- 1E+3	- 4E-7	- 1E-9	6E-5	6E-4 -	
50	Tin-121m	D, see <sup>110</sup> Sn	3E+3	9E+2	4E-7	1E-9	-	-	
		W, see <sup>110</sup> Sn	(4E+3)	- 5E+2	- 2E-7	- 8E-10	5E-5 -	5E-4	
50	Tin-121	D, see <sup>110</sup> Sn	6E+3 LLI wall	2E+4	6E-6	2E-8	-	-	
		₩, see <sup>110</sup> Sn	(6£+3) -	- 1E+4	- 5E-6	- 2E-8	8E-5	8E-4	
50	Tin-123m <sup>2</sup>	0, see <sup>110</sup> Sn W, see <sup>110</sup> Sn	5E+4	1E+5 1E+5	5E-1 6E-5	2E-7 2E-7	7E-4	7E-3	
50	Tin-123	D, see <sup>110</sup> Sn	5E+2	6E+2	3E-7	9E-10	-	-	
		W, see <sup>110</sup> Sn	(6E+2)	- 2E+2	- 7F-8	- 2E-10	9E-6	9E-5	

			0cc	/alues	Table 2 Effluent Concentrations		Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-5 - 4E-5 - 9E-4 - 1E-3 - 1E-2 - 3E-3 - 1E-2 - 9E-3
			Ingestion	Ingestion Inhalation				Average
No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/m1)
50	Tin-125	D, see <sup>110</sup> Sn	4E. LLI wall	9E+2	42-7	1E-9	-	-
		110	(5E+2)	1. A.		-	65-6	65-5
		W, see <sup>110</sup> Sn		4E+2	<i>₹</i> -7	5E-10	-	-
50	Tin-126	D coo 110co	25.0		A second second			
50	1111 120	W see 110 cm	32+2	6E+1	2E-8	8E-11	4E-6	4E-5
		w, see Sh		/2+1	3E-8	9E-11	-	
50	Tin-127	D see 110 sp	75+2	25.14	05.0			
		W see 110 Sp	1643	25+4	85-6	31-8	ar-2	9E-4
		w, see su		21+4	8E-6	3E-8	-	-
50	Tin-128 <sup>2</sup>	0. see 110 Sp	91.12	35+4	15.5	45 0		- 6E-5 - 9E-4 - 1E-3 - 1E-2
		W, see <sup>110</sup> Sn	-	AE+4	16-5	46-8	11-4	1E-3
				41.14	16-2	3E~8		-
51	Antimony-115 <sup>2</sup>	D, all compounds except those given for W	8 <u>E</u> +4	2E+5	1E-4	3E-7	1E-3	1E-2
		W, oxides, hydroxides, halides, sulfides,						
		surfaces, and mitrates		3E+5	1E-4	4E-7	-	-
51	Antimony-116m <sup>2</sup>	D see 115sh	25+4	75.0	25.5			
		W. see 115sb	2014	1645	31-5	11-1	3E-4	3E-3
		,		1640	6E-3	22-1		-
51	Antimony-116 <sup>2</sup>	D, see <sup>115</sup> Sb	7E+4	3E+5	1E-4	4E-7		
			St. wall					
		и 115-г.	(9E+4)			-	1E-3	1E-2
		w, see SD		3E+5	1E-4	5E-7	-	-
51	Antimony-117	D con 115ch	75.0	AF				
	including 117	W see 115sb	/1+4	26+5	9E-5	3E-7	9E-4	9E-3
		n, see 50		32+5	1E~4	4E-7	-	

			0cc	Table 1Table 2Occupational ValuesEffluentConcentrations				Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 7E-4 - 2E-3 - 2E-2 - 1E-4 - 1E-4 - 3E-2 - 7E-5 - 3E-4 -	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 7E-4 - 2E-3 - 2E-2 - 1E-4 - 1E-4 - 3E-2 - 7E-5 - 3E-4 - 9E-3		
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)		
51	Antimony-118m	D, see 115 W, see Sb	5E+3 -	2E+4 2E+4	8E-6 9E-6	3E-8 3E-8	7E-5	7E-4		
51	Antimony-119	D, see 115 W, see Sb	1E+4	5E+4 3E+4	2E-5 1E-5	6E-8 4E-8	2E-4	2E-3		
51	Antimony-120 <sup>2</sup> (16 min)	D, see <sup>115</sup> Sb	1E+5 St. wall	4E+5	2E-4	6E-7	-			
		W, see <sup>115</sup> Sb	(2E+5)	- 5E+5	- 2E-4	- 7E-7	2E-3	2E-2		
51	Antimony-120 (5.76 d)	D, see 115 W, see 115 Sb	9E+2	2E+3 1E+3	9E-7 5E-7	3E-9 2E-9	1E-5	1E-4		
51	Antimony-122	D, see <sup>115</sup> Sb	7E+2 LLI wall	2E+3	1E-6	3E-9	-	-		
		W, see <sup>115</sup> Sb	(8E+2)	- 1E+3	- 4E-7	- 2F-9	18-5	1E-4		
51	Antimony-124m <sup>2</sup>	D, see 115 W, see 115 Sb	2E+5	8E+5 6E+5	4E-4 2E-4	1E-6 8E-7	3E-3	3E-2		
51	Antimony-124	D, see <sup>115</sup> Sb W, see <sup>115</sup> Sb	5E+2	9E+2 2E+2	4E-7 1E-7	1E-9 3E-10	7E-6	7E-5		
51	Antimony-125	D, see <sup>115</sup> <sub>Sb</sub> W, see <sup>115</sup> <sub>Sb</sub>	2E+3	2E+3 5E+2	1E-6 2E-7	3E-9 7E-10	3E-5	3E-4		
51	Antimony-126m <sup>2</sup>	D, see $^{115}{ m Sb}$	5E+4 St. wall	2E+5	8E-5	3E-7	-			
		W, see <sup>115</sup> Sb	(7E+4)	- 2E+5	- 8E-5	- 3E-7	9E-4	9E-3		

			Table 1Table 2Occupational ValuesEffluentConcentrations				Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 7E-5 - 1E-4 - 1E-4 - 2E-4 - 3E-3 - 2E-3 -	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly Average Concentration (µCi/ml) 7E-5 - 1E-4 - 1E-2 - 2E-4 - 3E-3 - 2E-3 -
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/m1)	Concentration (µCi/ml)
51	Antimony-126	D, see 115 W, see Sb	5E+2 -	1E+3 5E+2	5E-7 2E-7	2E-9 7E-10	7E-6 -	78-5
51	Antimony-127	D, see <sup>115</sup> Sb	7E+2 LLI wall	2E+3	9E-7	3E-9	-	
		W, see <sup>115</sup> Sb	(8E+2)	- 9E+2	- 4E-7	- •E-9	1E-5 -	Able 3 Releases to Sewers 2 Monthly Average Concentration (µCi/ml) 7E-5 - 1E-4 - 1E-4 - 2E-4 - 3E-3 - 2E-3 -
51	Antimony-128 <sup>2</sup> (10.4 min)	D, see <sup>115</sup> Sb	8E+4 St. wall	4E+5	2E-4	5E-7	-	-
		W, see <sup>115</sup> Sb	(1E+5) -	- 4E+5	- 2E-4	- 6E-7	1E-3 -	1E-2 -
51	Antimony-128 (9.01 h)	D, see <sup>115</sup> <sub>Sb</sub> W, see <sup>115</sup> <sub>Sb</sub>	1E+3 -	4E+3 3E+3	2E-6 1E-6	6E-9 5E-9	25-5	2E-4
51	Antimony-129	D, see 115Sb W, see 115Sb	3E+3 -	9E+3 9E+3	4E-6 4E-6	1E-8 1E-8	4E-5	4E-4
51	Antimony-130 <sup>2</sup>	D, see <sup>115</sup> Sb W, see <sup>115</sup> Sb	2E+4	6E+4 8E+4	3E-5 3E-5	9E-8 1E-7	3E-4	3E-3
51	Antimony-131 <sup>2</sup>	D, see <sup>115</sup> Sb	1E+4 Thyroid	2E+4 Thyroid	1E-5	-	-	
		W, see <sup>115</sup> Sb	(2E+4) -	(4E+4) 2E+4 Thyroid	- 1E-5	6E-8	2E-4 -	2E-3 -
			김 김희는 것 같아요. 가	(4E+4)	-	6E-8	-	200

			000	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) IE-3 - IE-4 - IE-4 - 2E-4 - 2E-4 -
			Ingestion	Inhala	tion			Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	(µCi/ml)	Air (μCi/ml)	(µCi/ml)	(µCi/ml)
52	Tellurium-116	D, all compounds except those given for W	8E+3	2E+4	9E-6	3E-8	1E-4	1E-3
		W, oxides, hydroxides, and nitrates		3E+4	1E-5	4E-8	-	-
52	Tellurium-121m	D, see <sup>116</sup> Te	5E+2 Bone surf	2E+2 Bone surf	8E-8		-	-
		116	(7E+2)	(4E+2)	-	5E-10	1E-5	1E-4
		W, see Tre	-	4E+2	2E-7	6E-10	-	
		n 116-	25.2	45.13	25.6	CF 0	45.5	AF . A
52	Tellurium-121	W, see 116Te	3E#3 -	3E+3	1E-6	4E-9	46-5	45-4
52	Tellurium-123m	D, see <sup>116</sup> Te	6E+2 Bone surf	2E+2 Bone surf	9E-/	-	-	-
		116	(1E+3)	(5E+2)	-	8E-10	1E-5	1E-4
		W, see <sup>11b</sup> Te	-	5E+2	2E-7	8E-10	-	-
52	Tellurium-123	0, see <sup>116</sup> Te	5E+2 Bone surf	2E+2 Bone surf	8E-8	-	-	
		116	(1E+3)	(5E+2)	-	7E-10	2E-5	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-3 - - 1E-4 - 1E-4 - 2E-4 - 2E-4 -
		W, see <sup>110</sup> Te	-	4E+2 Bone surf	2E-7	~	-	
			전통 전 것 같아?	(1E+3)	-	2E9	-	
52	Tellurium-125m	D, see <sup>116</sup> Te	1E+3 Bone surf	4E+2 Bone surf	2E-7	-	-	-
		116	(1E+3)	(1E+3)	-	1E-9	2E-5	Releases to Sewers Monthly Average Concentratio (µCi/ml) IE-3 IE-4 - 4E-4 - IE-4 - 2E-4 - 2E-4 -
		W, see <sup>110</sup> Te		7E+2	3E-7	1E-9	-	4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1

				Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhala ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
52	Tellurium-127m	D, see <sup>116</sup> Te	6E+2	3E+2 Bone surf	1E-7	-	9E-6	9E-5
		W, see <sup>116</sup> Te		(4E+2) 3E+2	- 1E-7	6E-10 4E-10	-	2
52	Tellurium-127	D, see <sup>116</sup> Te W, see <sup>116</sup> Te	7E+3	2E+4 2E+4	9E-6 7E-6	3E-8 2E-8	1E-4 -	1E-3 -
52	Tellurium-129m	D, see <sup>116</sup> Te W, see <sup>116</sup> Te	5E+2	6E+2 2E+2	3E-7 1E-7	9E-10 3E-10	7E-6	7E-5
52	Tellurium-129 <sup>2</sup>	D, see <sup>116</sup> Te W, see <sup>116</sup> Te	3E+4	6E+4 7E+4	3E-5 3E-5	9E-8 1E-7	4E-4	4E-3
52	Tellurium-131m	D, see <sup>116</sup> Te	3E+2 Thuraid	4E+2 Thyroid	2E-7	-	-	-
		W, see <sup>116</sup> Te	(6E+2)	(1E+3) 4E+2 Thyroid	- 2E-7	2E-9 -	8E-6 -	8E-5 -
	2	116		(9E+2)		1E-9	-	
52	Tellu ium-131	D, see Te	3E+3 Thyroid	5E+3 Thyroid	2E-6		- -	
		W, see <sup>116</sup> Te	(6E+3) -	(1E+4) 5E+3 Thyroid	- 2E-6	2E-8 -	8E-5 -	8E-4 -
				(1E+4)	-	2E-8	-	

1.25

			0c	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 1 Col. 2 Col. 3 Oral		Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 9E-5 - - 9E-4 - - 4E-3 - - 3E-3 - - 2E-3
1. S. S. S.		Class	Ingestion	Inhal	ation	A		Average
Atomic No.	Radionuclide		ALI (µCi)	ALI (µCi)	(µCi/ml)	Air (µCi/ml)	(µCi/ml)	(µCi/ml)
52	Tellurium-132	D, see <sup>116</sup> Te	2E+2 Ibyroid	2E+2 Thyroid	9E-8	-	-	-
			(7E+2)	(8-+2)		1F-9	9F-6	9F-5
		W, see <sup>116</sup> Te	-	2E+2 Thyroid	9E-8	-	-	-
				(6E+2)	-	9E-10	-	
52	Tellurium-133m <sup>2</sup>	D, see <sup>116</sup> Te	3E+3 Thyroid	5E+3 Thyroid	2E-6	-	-	
			(6E+3)	(1E+4)	-	2E-8	9E-5	9E-4
		W, see <sup>116</sup> Te		5E+3 Thyroid	2E-6	-	-	
			성장은 친구 같이 나가서	(1E+4)	-	2E-8	-	-
52	Tellurium-133 <sup>2</sup>	D, see <sup>116</sup> Te	1E+4 Thursid	2E+4 Thursd	9E-6	-	-	-
			(35+4)	(65+4)	_	85-8	AE-A	45-3
		W, see <sup>116</sup> Te		2E+4 Thyroid	9E-6	-	-	-
				(6E+4)	-	8E-8	-	-
52	Tellurium-134 <sup>2</sup>	D, see <sup>116</sup> Te	2E+4 Thyroid	2E+4 Thyroid	1E-5	-	-	-
			(2E+4)	(5E+4)	-	7F-8	3F-4	35-3
		W, see <sup>116</sup> Te	-	2E+4 Thyroid	1E-5	-	-	-
				(5E+4)	-	7E-8	-	-
53	Iodine-120m <sup>2</sup>	D, all compounds	1E+4 Thyroid	2E+4	9E-6	3E-8	-	-
			(1E+4)	-		-	2E-4	2E-3

.

.

			Table 1 Occupational Values			Tal Eff Concent	ble 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic	Radionuclide	Class	ALI (uCi)	ALI (uCi)	DAC (uCi/ml)	Air (uCi/ml)	Water (uCi/ml)	Concentration (uCi/ml)
53	Indine-120 <sup>2</sup>	D all compounds	4F+3	9E+3	4E-6	-	-	-
55	Tourne 120	o, arr compounds	Thyroid (8E+3)	Thyroid (1E+4)	-	2E-8	1E-4	1E-3
53	Iodine-121	D, all compounds	1E+4 Thyroid	2E+4 Thyroid	8E-6	-	- 24	-
			(3E+4)	(5E+4)	-	7E-8	4E-4	4E-3
53	Iodine-123	D, all compounds	3E+3 Thyroid	6E+3 Thyroid	3E-6	-	-	
			(1E+4)	(2E+4)	-	2E-8	1E-4	1E-3
53	Iodine-124	D, all compounds	5E+1 Thyroid	8E+1 Thyroid	3E-8	-	-	
			(2E+2)	(3E+2)	-	4E-10	2E-6	2E-5
53	Iodine-125	D, all compounds	4E+1 Thyroid	6E+1 Thyroid	3E-8	-	-	-
			(1E+2)	(2E+2)	8 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	3E-10	2E-6	2E-5
53	lodine-126	D, all compounds	2E+1 Thyroid	4E+1 Thyroid	1E-8	-	-	-
			(7E+1)	(1E+2)	-	2E-10	1E-6	1E-5
53	Iodine-128 <sup>2</sup>	D, all compounds	4E+4 St. wall	1E+5	5E-5	2E-1	-	-
			(6E+4)	-	-	-	8E-4	8E-3
53	Iodine-129	D, all compounds	5E+0 Thyroid	9E+0 Thyroid	4E-9	-	-	-
			(2E+1)	(3E+1)	-	4E-11	2E-7	2E-6

			Table 1 Occupational Values		alues	Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2 Col. 3		Col. 1 Col. 2		Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	ALI (µCi)	ation DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/m1)
53	Iodine-130	D, all compounds	4E+2 Thyroid	7E+2 Thyroid	3E-7	-	-	-
			(1E+3)	(2E+3)	-	3E-9	2E-5	2E-4
53	Iodine-131	D, all compounds	3E+1 Thyroid	5E+1 Thyroid	2E-8	-	-	-
			(9E+1)	(2E+2)	-	2E-10	1E-6	1E-5
53	Iodine-132m <sup>2</sup>	D, all compounds	4E+3 Thyroid	8E+3 Thyroid	48-6	-	-	-
			(1E+4)	(2E+4)	-	3E-8	1E-4	1E-3
53	Iodine-132	D, all compounds	4E+3 Thyroid	8E+3 Thyroid	3E-6	-	-	-
			(9E+3)	(1E+4)	-	2E-8	1E-4	1E-3
53	Iodine-133	D, all compounds	1E+2 Thyroid	3E+2 Thyroid	1E-7	-	-	-
			(5E+2)	(9E+2)	-	1E-9	7E-6	7E-5
53	Iodine-134 <sup>2</sup>	D, all compounds	2E+4 Thyroid	5E+4	2E-5	6E-8	-	-
			(3E+4)	-	-	-	4E-4	4E-3
53	Iodine-135	D, all compounds	8E+2 Thyroid	2E+3 Thyroid	7E-7	-	-	- 4
			(3E+3)	(4E+3)	-	6E-9	3E-5	3E-4

			0cc	Table 1 Occupational Values			ble 2 luent trations	Table 3 Releases to Sewers Monthly
			Col. 1 Oral Ingestion	Col. 1 Col. 2 Col. 3 Oral Ingestion Inhalation		Col. 1 Col. 2		
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
54	Xenon-120 <sup>2</sup>	Submersion <sup>1</sup>	-	-	1E-5	4E-8	-	-
54	Xenon-121 <sup>2</sup>	Submersion <sup>1</sup>	-	-	2E-6	1E-8	-	- 1
54	Xenon-122	Submersion <sup>1</sup>			7E-5	3E-7	-	-
54	Xenon-123	Submersion <sup>1</sup>		-	6E-6	3E-8	-	-
54	Xenon-125	Submersion <sup>1</sup>		-	2E-5	7E-8	-	-
54	Xenon-127	Submersion <sup>1</sup>	-	-	18-5	6E-8	-	
54	Xenon-129m	Submersion <sup>1</sup>		-	2E-4	9E-7	-	- 11
54	Xenon-131m	Submersion <sup>1</sup>		-	4E-4	2E-6	-	-
54	Xenon-133m	Submersion <sup>1</sup>	-	-	1E-4	6E-7	-	-
54	Xenon-133	Submersion <sup>1</sup>	-	-	1E-4	5E-7	-	
54	Xenon-135m <sup>2</sup>	Submersion		-	9E-6	4E-8	-	
54	Xenon-135	Submersion <sup>1</sup>	1	-	1E-5	7E-8	2	
54	Xenon-138 <sup>2</sup>	Submersion <sup>1</sup>	-	-	4E-6	2E-8	-	2
55	Cesium-125 <sup>2</sup>	D, all compounds	5E+4 St. wall	1E+5	6E-5	2E-7		
			(92+4)	-			18-3	1E-2
15	Cesium-127	D, all compounds	5E+4	9E+4	4E-5	1E-7	9E-4	9E-3

"CI

....

2.8

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-3 - 1E-2 3E-3 4E-4 - 2E-2 9E-6 1E-2 9E-6 1E-2 1E-4 6E-5 1E-5 - 4E-3 8E-4
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhal ALI (µCi)	lation DAC (µCi/ml)	Air (µCi/m1)	Water (µCi/ml)	Average Concentration (µCi/ml)
55	Cesium-129	D, all compounds	2E+4	3E+4	1E-5	5E-8	3E-4	3E-3
55	Cesium-130 <sup>2</sup>	D, all compounds	6E+4 St. wall	2E+5	8E-5	3E-7	-	-
			(1E+5)		-		1E-3	1E-2
55	Cesium-131	D, all compounds	2E+4	3E+4	1E-5	4E-8	3E-4	3E-3
55	Cesium-132	D, all compounds	3E+3	4E+3	2E-6	6E-9	4E-5	4E-4
55	Cesium-134m	D, all compounds	1E+5 St. wall	1E+5	6E-5	2E-7	-	
			(1E+5)		-	-	2E-3	2E-2
55	Cesium-134	D, all compounds	7E+1	1E+2	4E-8	2E-10	9E-7	9E-6
55	Cesium-135m <sup>2</sup>	D, all compounds	1E+5	2E+5	8E-5	3E-7	1E-3	1E-2
55	Cesium-135	D, all compounds	7E+2	1E+3	5E-7	2E-9	1E-5	1E-4
55	Cesium-136	D, all compounds	4E+2	7E+2	3E-7	9E-10	6E-6	6E-5
55	Cesium-137	D, all compounds	1E+2	2E+2	6E-8	2E-10	1E-6	1E-5
55	Cesium-138 <sup>2</sup>	D, all compounds	2E+4 St. wall	6E+4	2E-5	8E-8	-	-
			(3E+4)	-	-	-	4E-4	4E-3
56	Barium-126 <sup>2</sup>	D, all compounds	6E+3	2E+4	6E-6	2E-8	8E-5	8E-4

**П** 1 **П** 

			Table 1 Occupational Values			Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/m1)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
56	Barium-128	D, all compounds	5E+2	2E+3	7E-7	2E-9	7E-6	7E-5
56	Barium-131m <sup>2</sup>	D, all compounds	4E+5 St. wall	1E+6	6E-4	2E-6	-	-
			(5E+5)		1.00	: 1993 - 1993	7E-3	7E-2
56	Barium-131	D, all compounds	3E+3	8E+3	3E-6	1E-8	4E-5	4E-4
56	Barium-133m	D, all compounds	2E+3 LLI wall	9E+3	4E-6	1E-8	-	-
			(3E+3)	1997			4E-5	4E-4
56	Barium-133	D, all compounds	2E+3	7E+2	3E-7	9E-10	2E-5	2E-4
56	Barium-135m	D, all compounds	3E+3	1E+4	5E-6	2E-8	4E-5	4E-4
56	Barium-139 <sup>2</sup>	D, all compounds	1E+4	3E+4	1E-5	4E-8	2E-4	2E-3
56	Barium-140	D, all compounds	5E+2	1E+3	6E-7	2E-9	-	-
			(6E+2)	-			8E-6	8E-5
56	Barium-141 <sup>2</sup>	D, all compounds	2E+4	7E+4	3E-5	1E-7	3E-4	3E-3
56	Barium-142 <sup>2</sup>	D, all compounds	5E+4	1E+5	6E-5	2E-7	7E-4	7E-3
57	Lanthanum-131 <sup>2</sup>	D, all compounds except those given for W	5E+4	1E+5	5E-5	2E-7	6E-4	6E-3
		W, oxides and hydroxides		2E+5	7E-5	2E-7	-	-

			Table 1 Occupational Values		Values	Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
	0.11		Ingestion	Inhal	lation			Average
No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
57	Lanthanum-132	D, see 131	3E+3	1F+4	4F-6	1F-8	4F-5	AF-A
		W, see <sup>131</sup> La	이 아이들 것이 같아.	18+4	5E-6	2E-8	-	-
57	Lanthanum-135	D. see 131	4F+A	15+5	AE-5	15-7	55-4	55.3
		W, see <sup>131</sup> La	-	9E+4	4E-5	1E-7	JE-4	56-3
57	Lanthanum-137	D, see <sup>131</sup> La	1E+4	6E+1	3E-8	-	2E-4	2E-3
		121	승규가 누구 같다.	(7E+1)		1E-10	2.00	
		W, see <sup>131</sup> La		3E+2	1E-7	-	-	
				(3E+2)	-	4E-10	-	
57	Lanthanum-138	D. see 131	9F+2	4E+0	15-0	55-12	15.5	15.4
		W, see <sup>131</sup> La	-	1E+1	6E-9	2E-11	-	16-4
57	Lanthanum-140	D. see 131	6E+2	15+2	65-7	25.0	05.0	0F F
		W, see <sup>131</sup> La	-	1E+3	5E-7	2E-9 2E-9	98-0	9E-5
57	Lanthanum-141	D see 131	45+3	65+30	AF . C	35.0		
		W, see <sup>131</sup> La	-	1E+4	42~6 5E-6	1E-8 2E-8	5E-5	5E-4
57	Lanthanum-1422	D coo 131	05.12	05.4				
	Editoriarian 142	W, see 131La	0000	2E+4 3E+4	9E-6 1E-5	3E-8	1E-4	1E-3
57	Lanthania 142	n			AL 5	JL U		
57	Lanthanum-143	D, see La	4E+4 St wall	1E+5	4E-5	1E-7	-	-
		131	(4E+4)	-	-	-	5F-4	Monthly Average Concentration (µCi/ml) 4E-4 - 5E-3 - 2E-3 - 1E-4 - 5E-4 - 1E-3 - 5E-3 - 5E-3 -
		W, see La		9E+4	4E-5	1E-7	1	-

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic	Dadianualida	61	Ingestion	stion Inhalation				Average
No.	Radionuciide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
58	Cerium-134	W. all compounds except						
		those given for Y	5E+2 LLI wall	7E+2	3E-7	1E-9	-	-
		V oxidas hudrovidos	(6E+2)		-	-	8E-6	8E-5
		and fluorides		7E+2	3E-7	9E-10	-	-
58	Cerium-135	W, see 134Ce	2F+3	4F+3	2F-6	55-0	25-5	25.4
		Y, see <sup>134</sup> Ce	-	4E+3	1E-6	5E-9	-	2E-4
58	Cerium-137m	W, see <sup>134</sup> Ce	2E+3 LLI wall	4E+3	2E-6	6E-9	-	-1
		134	(2E+3)	Sector 14	8 ( <b>-</b> ) (-) (-)	-	3E-5	3F-4
		Y, see <sup>15</sup> Ce	영향 관 문화	4E+3	2E-6	5E-9	12000	-
58	Cerium-137	W, see 134 134Ce	5E+4	1E+5	6E-5	2E-7	7E-4	7F-3
		Y, see <sup>10</sup> Ce		1E+5	5E-5	2E-7	-	-
58	Cerium-139	W. see 134Ce	5F+3	8E+2	25-7	15.0	75.5	
		Y, see <sup>134</sup> Ce	-	7E+2	3E-7	9F-10	72-5	7E-4
58	Cerium-141	₩, see <sup>134</sup> Ce	2E+3	7E+2	3E-7	1E-9	_	
			LLI wall					- 2E-4 - 3E-4 - 7E-3 - 7E-4 - 3E-4
		Y, see <sup>134</sup> Ce	(20+3)	- 6E+2	- 2F-7	- 8F-10	3E-5	3E-4
58	Cerium-143	W, see <sup>134</sup> Ce	1E+3 LLI wall	2E+3	8E-7	3E-9	-	-
		134	(1E+3)	-	-	-	2E-5	2F-4
		Y, see Ce		2E+3	7E-7	2E-9	-	-

			Table 1 Occupational Values			Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral Ingestion	Col. 2 Inhal	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALĨ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
58	Cerium-144	W, see <sup>134</sup> Ce	2E+2 LLI wall	3E+1	1E-8	4E-11	-	-
		124	(3E+2)			-	3F-6	38-5
		Y, see <sup>134</sup> Ce		1E+1	6E-9	2E-11	-	-
59	Praseodymium-136 <sup>2</sup>	W, all compounds except those given for Y	5E+4 St. wall	2E+5	1E-4	3E-7	-	-
		Y. oxides, hydroxides	(72+4)			1.1	1E-3	1E-2
		carbides, and fluorides	1848 B. C. S.	2E+5	9E-5	3E-7	_	
59	Praseodymium-137 <sup>2</sup>	W, see <sup>136</sup> Pr Y, see <sup>136</sup> Pr	4E+4	2E+5 1E+5	6E-5 6E-5	2E-7 2E-7	5E-4	5E-3
59	Praseodymium-138m	W, see <sup>136</sup> Y, see <sup>136</sup> Pr	1E+4 -	5E+4 4E+4	2E-5 2E-5	8E-8 6E-8	1E-4	1E-3
59	Praseodymium-139	W, see <sup>136</sup> Pr Y, see <sup>136</sup> Pr	4E+4	1E+5 1E+5	5E-5 5E-5	2E-7 2E-7	6E-4	6E-3
59	Praseodymium-142m	W, see <sup>136</sup> pr Y, see <sup>136</sup> pr	8E+4 -	2E+5 1E+5	7E-5 6E-5	2E-7 2E-7	1E-3	1E-2
59	Praseodymium-142	W, see <sup>136</sup> pr Y, see <sup>136</sup> pr	1E+3 -	2E+3 2E+3	9E-7 8E-7	3E-9 3E-9	1E-5	1E-4

			0cc	Table 1 Supational N	Values	Table 2 Effluent Concentrations		Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 2E-4 - 6E-3 - 4E-4 - 1E-2 - 2E-3 - 2E-3 - 3E-4	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly		
Atomic No.	Radionuclide	Class	ALI (uCi)	ALI (uCi)	DAC (uCi/ml)	Air (uCi/ml)	Water	Average Concentration		
59	Praseodymium-143	W, see <sup>136</sup> Pr	9E+2	8E+2	3E-7	1E-9	-	- (pc (7 m))		
	Praseodymium-143 W Praseodymium-144 <sup>2</sup> W Y Praseodymium-145 W	V can 136	(1E+3)			-	2E-5	2E-4		
		1, see Pr		7E+2	3E-7	9E-10	-			
59	Praseodymium-144 <sup>2</sup>	W, se <sup>136</sup> Pr	3E+4 St. wall	1E+5	5E-5	2E-7	-	-		
		136	(4E+4)			-	6E-4	6F-3		
		Y, see Pr		1E+5	5E-5	2E-7	-	1		
59	Praseodymium-145	W, see 136pr	3E+3	9E+3	4E-6	1E-8	4E-5	4E-4		
	이 사람은 사람들을	r, see Pr		8E+3	3E-6	1E-8	-			
59	Praseodymium-147 <sup>2</sup>	W, see <sup>136</sup> Pr	5E+4 St. wall	2E+5	8E-5	3E-7	-	-		
		136	(8E+4)		-		1F-3	1F-2		
		Y, see Pr	1. <b>*</b>	2E+5	8E-5	3E-7	-	-		
60	Neodymium-136 <sup>2</sup>	W, all compounds except								
		chose given for Y	1E+4	6E+4	2E-5	8E-8	2E-4	2E-3		
		Y, oxides, hydroxides,								
		carbides, and fluorides	-	5E+4	2E-5	8E-8		2.0		
60	Neodymium-138	W son 136 <sub>Nd</sub>	05.0							
	neo dymronii 150	Y. See 136 Nd	26+3	62+3	3E-6	9E-9	3E-5	3E-4		
		100		5E+3	2E-6	/E-9	-	5 (1) (1)		
60	Necdymium-139m	W, see 136Nd	5E+3	2E+4	7E-6	2F-8	7E-5	75-4		
		Y, see Nd		1E+4	6E-6	2E-8	-	-		

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentratio (µCi/ml) 1E-2 - 2E-2 - 2E-4 - 9E-3 - 8E-3 - 7E-4 - 2E-4
			Ingestion	Inhalation				Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI DAC (µCi) (µCi/ml)		Air Water (µCi/ml) (µCi/ml)		Concentration (µCi/ml)
60	Neodymium-139 <sup>2</sup>	W, see <sup>136</sup> Nd Y, see <sup>136</sup> Nd	9E+4 -	3E+5 3E+5	1E-4 1E-4	5E-7 4E-7	1E-3 -	1E-2
60	Neodymium-141	W, see <sup>136</sup> Nd Y, see <sup>136</sup> Nd	2E+5	7E+5 6E+5	3E-4 3E-4	1E-6 9E-7	2E-3	2E-2 -
60	Neodymium-147	W, see <sup>136</sup> Nd	1E+3 LLI wall	9E+2	4E-7	1E-9	-	
		Y, see <sup>136</sup> Nd	(1E+3) -	- 8E+2	- 4E-7	- 1E-9	2E-5	2E-4 -
60	Neodymium-149 <sup>2</sup>	W, see <sup>136</sup> Nd Y, see <sup>136</sup> Nd	1E+4	3E+4 2E+4	1E-5 1E-5	4E-8 3E-8	1E-4 -	1E-3
60	Neodymium-151 <sup>2</sup>	W, see <sup>136</sup> Nd Y, see <sup>136</sup> Nd	7E+4 -	2E+5 2E+5	8E-5 8E-5	3E-7 3E-7	9E-4 -	9E-3
61	Promethium-141 <sup>2</sup>	W, all compounds except those given for Y	5E+4 St. wall	2E+5	8E-5	3E-7	-	-
		Y, oxides, hydroxides, carbides, and fluorides	(6E+4) -	- 2E+5	- 7E-5	- 2E-7	8E-4	8E-3
61	Promethium-143	W, see <sup>141</sup> Pm Y, see <sup>141</sup> Pm	5E+3 -	6E+2 7E+2	2E-7 3E-7	8E-10 1E-9	7E-5	7E-4
61	Promethium-144	W, see <sup>141</sup> Pm Y, see <sup>141</sup> Pm	1E+3	1E+2 1E+2	5E-8 5E-8	2E-10 2E-10	2E-5	2E-4

Vania

-

÷.

	Padionuclida		0cc	Table 1 Occupational Values			ble 2 luent trations	Table 3 Releases to Sewers	
			Col Ora Inc	Col. 1 Oral	1 Col. 2 Col. 3		Col. 1	Col. 2	Monthly
Atomic		61000		Ingestion			Air	Water	Concentration
No.	Radionucifue	01035		(µCi)	(µCi)	(µCi/ml)	(µCi/ml)	(µCi/ml)	(µCi/m1)
61	Promethium-145	W, see	141 <sub>Pm</sub>	1E+4	2E+2 Bone surf	7E-8	-	1E-4	1E-3
				16 - 2 전 4 2 2 2 2 2 2 2	(2E+2)	-	3E-10	-	-
		Y, see	141 <sub>Pm</sub>		2E+2	8E-8	3E-10	-	-
61	Promethium-146	W. see	141 <sub>Pm</sub>	2E+3	5E+1	2E-8	7E-11	2E-5	2E-4
(C. s.)		Y, see	141 <sub>Pm</sub>		4E+1	2E~8	6E-11	-	-
61	Promethium-147	W, see	141 <sub>Pm</sub>	4E+3 LLI wall	1E+2 Bone sur?	5E-8	-	-	
			141	(5E+3)	(2E+2)	-	3E-10	78-5	7E-4
		Y, see	141Pm	-	1E+2	6E-8	2E-10	-	
61	Promethium-148m	W. see	141 <sub>Pm</sub>	7E+2	3E+2	1F-7	4E-10	16-5	1F-4
		Y, see	141 <sub>Pm</sub>		3E+2	1E-7	5E-10	-	-
61	Promethium-148	W, see	141 <sub>Pm</sub>	4E+2 LLI wall	5E+2	2E-7	8E-10	-	-
			141	(5E+2)	-	-	-	7E-6	7E-5
		Y, see	141Pm		5E+2	2E-7	7E-10	-	-
61	Promethium-149	W, see	141 <sub>Pm</sub>	1E+3 LLI wall	2E+3	8E-7	3E-9	-	-
			141	(1E+3)	-	-	-	2E-5	2E-4
		Y, see	141Pm		2E+3	8E-7	2E-9	-	÷ 244
61	Promethium-150	W, see	141 Pm	5E+3	2E+4	8E-6	3E-8	7E-5	7E-4
		Y, see	141Pm		2E+4	7E-6	2E-8	-	-
61	Promethium-151	W. see	141 <sub>Pm</sub>	2E+3	4E+3	1E-6	5F-9	2E-5	25-4
		Y, see	141 Pm		3E+3	1E-6	4E-9	-	-

				Table 1 Occupational Values			Table 2 Effluent Concentrations	
				Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Ingestion	Inhalation				Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	(µCi/ml)
62	Samarium-141m <sup>2</sup>	W, all compounds	3E+4	1E+5	4E-5	1E-7	4E-4	4E-3
62	Samarium-141 <sup>2</sup>	W, all compounds	5E+4 St wall	2E+5	8E-5	2E-7		-
			(6E+4)			-	8E-4	8E-3
62	Samarium-142 <sup>2</sup>	W, all compounds	8E+3	3E+4	1E-5	4E-8	18-4	1E-3
62	Samarium-145	W, all compounds	6E+3	5E+2	2E-7	7E-10	8E-5	8E-4
62	Samarium-146	W, all compounds	1E+1 Bone surf	4E+2 Bone surf	1E-11	-	-	-
			(3E+1)	(6E-2)	-	9E-14	3E-7	3E-6
62	Samarium-147	W, all compounds	2E+1 Rone surf	4E+2 Bone surf	2E-11	-	-	-
			(3E+1)	(7E-2)	-	1E-13	4E-7	4E-6
62	Samarium-151	W, all compounds	1E+4	1E+2 Rone surf	4E-8	-	-	-
			(1E+4)	(2E+2)	-	2E-10	2E-4	2E-3
62	Samarium-153	W, all compounds	2E+3	3E+3	1E-6	4E-9	-	-
			(2E+3)	-	-	-	3E-5	3E-4
52	Samarium-155 <sup>2</sup>	W, all compounds	6E+4 St wall	2E+5	9E-5	3E-7	-	-
			(8E+4)		-	-	1E-3	1E-2

· · · •

			0cc	Table 1 Occupational Values			ole 2 luent crations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhala ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
62	Samarium-156	W, all compounds	5E+3	9E+3	4E-6	1E-8	7E-5	7E-4
63	Europium-145	W, all compounds	2E+3	2E+3	8E-7	3E-9	2E-5	2E-4
63	Europium-146	W, all compounds	1E+3	1E+3	5E - 7	2E-9	1E-5	1E-4
63	Europium-147	W, all compounds	3E+3	2E+3	7E-7	2E-9	4E-5	4E-4
63	Europium-148	W, all compounds	1E+3	4E+2	1E-7	5E-10	1E-5	1E-4
63	Europium-149	₩, all compounds	1E+4	3E+3	1E-6	4E-9	2E-4	2E-3
63	Europium-150 (12.62 h)	W, all compounds	3E+3	8E+3	4E-6	1E-8	42-5	4E-4
63	Europium-150 (34.2 y)	₩, all compounds	8E+2	2E+1	8E-9	3E-11	1E-5	1E-4
63	Europium-152m	W, all compounds	3E+3	6E+3	3E-6	9E-9	4E-5	4E-4
63	Europium-152	W, all compounds	8E+2	2E+1	1E-8	3E-11	1E-5	1E-4
63	Europium-154	W, all compounds	5E+2	2E+1	8E-9	3E-11	7E-6	7E-5
63	Europium-155	W, all compounds	4E+3	9E+1 Bone surf	4E-8		5E-5	5E-4
			김 동일 전 것	(1E+2)	-	2E-10	-	-
63	Europium-156	W, all compounds	6E+2	5E+2	2E-7	6E-10	8E-6	8E-5

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
63	Europium-157	W, all compounds	2E+3	5E+3	2E-6	7E-9	3E-5	3E-4
63	Europium-158 <sup>2</sup>	W, all compounds	2E+4	6E+4	2E-5	8E-8	3E-4	3E-3
64	Gadolinium-145 <sup>2</sup>	D, all compounds except those given for W	5E+4 St. wall	2E+5	6E-5	2E-7		-
		W, oxides, hydroxides, and fluorides	(5E+4)	- 2E+5	- 7E-5	- 2E-7	6E-4	6E-3
64	Gadolinium-146	D, see <sup>145</sup> <sub>145</sub> Gd W, see <sup>145</sup> <sub>Gd</sub>	1E+3	1E+2 3E+2	5E-8 1E-7	2E-10 4E-10	2E-5	2E-4
64	Gadolinium-147	D, see $145_{Gd}$ W, see $145_{Gd}$	2E+3	4E+3 4E+3	2E-6 1E-6	6E-9 5E-9	3E-5	3E-4
64	Gadolinium-148	D, see <sup>145</sup> Gd	1E+1 Bone surf	8E+3 Bone surf	3F-12	-	-	-
		W, see <sup>145</sup> Gd	(2E+1)	(2E+2) 3E-2 Bone surf	- 1E-11	2E-14 -	3E-7 -	3E-6 -
			가 가장 말한 것	(6E-2)		8E-14	-	-
64	Gadolinium-149	D, see <sup>145</sup> Gd W, see <sup>145</sup> Gd	3E+3	2E+3 2E+3	9E-7 1E-6	3E-9 3E-9	4E-5	4E-4
64	Gadolinium-151	D, see <sup>145</sup> Gd	6E+3	4E+2 Bone surf	2E-7	-	9E-5	9E-4
		W, see <sup>145</sup> Gd	-	(6E+2) 1E+3	- 5E-7	9E-10 2E-9	-	2

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2 Col. 3		Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/ml)
64	Gadolinium-152	D, see <sup>145</sup> Gd	2E+1 Bone surf	1E-2 Bone surf	4E-12	-	-	-
		W, see <sup>145</sup> Gd	(3E+1)	(2E-2) 4E-2 Bone surf	- 2E-11	3E-14 -	4E-7 -	4E-6
				(8E-2)	41.13	1E-13	- 1 A	-
64	Gadolinium-153	D, see <sup>145</sup> Gd	5E+3	1E+2 Bone surf	6E-8	-	6E-5	6E-4
		W, see <sup>145</sup> Gd		(2E+2) 6E+2	- 2E-7	3E-10 8E-10	-	-
64	Gadolinium-159	D, see <sup>145</sup> Gd W, see <sup>145</sup> Gd	3E+3	8E+3 6E+3	3E-6 2E-6	1E-8 8E-9	4E-5	4E-4
65	Terbium-147 <sup>2</sup>	W, all compounds	9E+3	3E+4	1E-5	5E-8	1E-4	1E-3
65	Terbium-149	W, all compounds	5E+3	7E+2	3E-7	1E-9	7E-5	7E-4
65	Terbium-150	W, all compounds	5E+3	2E+4	9E-6	3E-8	7E-5	7E-4
65	Terbium-151	W, all compounds	4E+3	9E+3	4E-6	1E-8	5E-5	5E-4
65	Terbium-153	W, all compounds	5E+3	7E+3	3E-6	1E-8	7E-5	7E-4
65	Terbium-154	W, all compounds	2E+3	4E+3	2E-6	6E-	25-5	2E-4
65	Terbium-155	W, all compounds	6E+3	8E+3	3E-6	1E-8	8E-5	8E-4
65	Terbium-156m	W, all compounds	2E+4	3E+4	1E-5	4E-8	2E-4	2E-3

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2 Col. 3		Col. 1 Col. 2		Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
65	Terbium-156m (24.4 h)	W, all compounds	7E+3	8E+3	3E-6	1E-8	1E-4	1E-3
65	Terbium-156	W, all compounds	1E+3	1E+3	6E-7	2E-9	1E-5	1E-4
65	Terbium-157	W, all compounds	5E+4 LLI wall	3E+2 Bone surf	1E-7	-	-	-
			(5E+4)	(6E+2)	-	8E-10	7E-4	7E-3
65	Terbium-158	W, all compounds	1E+3	2E+1	8E~9	3E-11	2E-5	2E-4
65	Terbium-160	W, all compounds	8E+2	2E+2	9E-8	3E-10	1E-5	1E-4
65	Terbium-161	W, all compounds	2E+3 LLI wall	2E+3	7E-7	2E-9	-	-
			(2E+3)	-	-	-	3E-5	38-4
66	Dysprosium-155	W, all compounds	9E+3	3E+4	1E-5	4E-8	1E-4	1E-3
66	Dysprosium-157	W, all compounds	2E+4	6E+4	3E-5	9E-8	3E-4	3E-3
66	Dysprosium-159	W, all compounds	1E+4	2E+3	1E-6	3E-9	2E-4	2E-3
66	Dysprosium-165	W, all compounds	1E+4	5E+4	2E-5	6E-8	2E-4	2E-3
66	Dysprosium-166	W, all compounds	6E+2 LLI wall	7E+2	3E-7	1E-9	-	
	1. 1. C. M. M. M.		(8E+2)		-	-	1E-5	1E-4
67	Holmium-155 <sup>2</sup>	W, all compounds	4E+4	2E+5	6E-5	2E-7	6E-4	6E-3

## Table 1 Table 2 Table 3 Releases to Effluent Occupational Values Concentrations Sewers Col. 1 Col. 2 Col. 3 Col. 1 Col. 2 Monthly Oral Ingestion Average Inhalation Concentration ALI Water Atomic Radionuclide Class. ALI DAC Air (µCi) (µCi/ml) (µCi/ml) (µCi/ml) (µCi/ml) (µCi) Holmium-1572 W, all compounds 3E+5 1E+6 5E-4 2E-6 4E-3 4E-2

No.

67

67	Holmium-159~	W, all compounds	2E+5	1E+6	4E-4	1	3E-3	3E-2	
67	Holmium-161	W, all compounds	1E+5	4E+5	2E-4	6E-7	1E-3	1E-2	
67	Holmium-162m <sup>2</sup>	W, all compounds	5E+4	3E+5	1E-4	4E-7	7E-4	7E-3	
67	Holmium-162 <sup>2</sup>	W, all compounds	5E+5 St. wall	2E+6	1E-3	3E-6			
			(8E+5)			1. A. A. A. A.	1E-2	1E-1	
67	Holmium-164m <sup>2</sup>	W, all compounds	1E+5	3E+5	1E-4	4E-7	1E-3	1E-2	
(7	Holmium-164 <sup>2</sup>	W, all compounds	2E+5 St. wall	6E+5	3E-4	9E-7	-	-	
			(2E+5)	-		-	3E-3	3E-2	
67	Holmium-166m	W, all compounds	6E+2	7E+0	3E-9	9E-12	9E-6	9E-5	
67	Holmium-156	W, all compounds	9E+2	2E+3	7E-7	2E-9	-	-	
			(9E+2)			-	1E-5	1E-4	
67	Holmium-167	W, all compounds	2E+4	6E+4	2E-5	8E-8	2E-4	2E-3	
68	Erbium-161	W, all compounds	2E+4	6E+4	3E-5	9E-8	2E-4	2E-3	
68	Erbium-165	W, all compounds	6E+4	2E+5	8E-5	3E-7	9E-4	9E-3	

			0c	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
68	Erbium-169	W, all compounds	3E+3 LLI wall	3E+3	1E-6	4E-9	-	-
			(4E+3)			-	5E-5	5E-4
68	Erbium-171	W, all compounds	4E+3	1E+4	4E-6	1E-8	5E-5	5E-4
68	Erbium-172	W, all compounds	1E+3 LLI wall	1E+3	6E-7	2E-9	-	
			(1E+3)			-	2E-5	2E-4
69	Thulium-162 <sup>2</sup>	W, all compounds	7E+4 St. wall	3E+5	1E-4	4E-7	-	-
			(7E+4)		-	-	1E-3	1E-2
69	Thulium-166	W, all compounds	4E+3	1E+4	6E-6	2E-8	6E5	6E-4
69	Thulium-167	W, all compounds	2E+3	2E+3	8E-7	3E-9	-	-
			(2E+3)			-	3E-5	3E-4
69	Thulium-170	W, all compounds	8E+2	2E+2	9E-8	3E-10	-	-
			(1E+3)			-	1E-5	1E-4
69	Thulium-171	W, all compounds	1E+4 LLI wall	3E+2 Bone surf	1E-7		- 1	
			(1E+4)	(6E+2)	-	8E-10	2E-4	2E-3
69	Thulium-172	W, all compounds	7E+2 LLI wall	1E+3	5E-7	2E-9	3	-
			(8E+2)		-	-	1E-5	1E-4

			0cc	/alues	Table 2 Effluent Concentrations		Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhal ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi∕ml)	Average Concentratio (µCi/m1)
69	Thulium-173	W, all compounds	4E+3	1E+4	5E-6	2E-8	68-5	6F-4
69	Thulium~175 <sup>2</sup>	W, all compounds	7E+4 St. wall	3E+5	1E-4	4E-7	-	-
			(9E+4)		-	-	1E-3	1E-2
70	Ytterbium-162 <sup>2</sup>	W, all compounds except	75+4	35+5	15-4	45.7	15.0	
		chose green for f	10.4	32+5	10-4	48-7	it-3	1E-2
		Y, oxides, hydroxides, and fluorides		3E+5	1E-4	4E-7	-	
70	Ytterhium-166	W see 162vh	15.2	25.2	05.7			
	Coccestian 100	Y, see 162Yb	-	2E+3 2E+3	8E-7 8E-7	35-9	2E-5	2E-4
70		162		22.5	OL /	36-9		
/0	Ytterbium-16/~	W, see 162Vb	3E+5	8E+5	3E-4	1E-6	4E-3	4E-2
		t, see 1D		7E+5	3E-4	1E-6		. <del>.</del>
70	Ytterbium-169	W, see 162yb	2F+3	8E+2	45-7	15-0	0F F	AF 4
		Y, see <sup>162</sup> Yb	-	7E+2	3E-7	16-9	2E-5	ZE-4
70		162			52.7	11 5		
70	Ytterbium-175	W, see <sup>102</sup> Yb	3E+3 LLI wall	4E+3	1E-6	5E-9	-	-
		v 162 <sub>v</sub>	(3E+3)	-	-	-	4E-5	4E-4
		1, see 1b		3E++3	1E-6	5E-9	-	-
70	Ytterbium-177 <sup>2</sup>	W, see 162yb	2F+4	55+1	25-5	75-0	25.4	AF . 3
		Y, see <sup>162</sup> Yb	-	SEAN	28-5	65-8	26-4	2E-3
-	2	162			er s	02 0		
/0	Ytterbium-178 <sup>2</sup>	W, see 162Yb	1E+4	4E+4	2E-5	6E-8	2E-4	2E-3
		Y, see Yb	-	4E+4	2E-5	5E-8	-	-

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Ingestion Inhalation		ation			Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
71	Lutetium-169	W, all compounds except those given for Y	3E+3	4E+3	2E-6	6E-9	3E-5	3E-4
		Y, oxides, hydroxides, and fluorides		4E+3	2E-6	6E-9	-	-
73	1.1.1.	169,	15+2	25+3	05-7	25-0	25-5	25-4
/1	Lutetium-1/U	Y, see 169Lu	-	2E+3 2E+3	8E-7	3E-9		-
71	Lutetium-171	W. see 169	2E+3	2E+3	8E-7	3E-9	3E-5	3E-4
		Y, see <sup>169</sup> Lu		2E+3	8E-7	3E-9		-
71	lutotium-172	W see 169	15+3	1613	55-7	2F-9	16-5	15-4
11	Lucecium 172	Y, see 169Lu	-	1E+3	5E-7	2E-9	-	-
71	Lutetium-173	W, see <sup>169</sup> Lu	5E+3	3E+2 Bone surf	1E-7		7E-5	7E-4
		169	- B. C. 199	(5E+2)	-	6E-10	-	-
		Y, see <sup>105</sup> Lu		3E+2	1E-7	4E-10	-	
71	Lutetium-174m	W, see <sup>169</sup> Lu	2E+3 LLI wall	2E+2 Bone surf	1E-7	-	-	-
		169	(3E+3)	(3E+2)		5E-10	4E-5	4E-4
		Y, see Lu	요즘 가 가 가 있는	2E+2	9E-8	3E-10	-	
71	Lutetium-174	W, see <sup>169</sup> Lu	5E+3	1E+2 Bone surf	5E-8	-	7E-5	7E-4
		169	상 비행하는 것을	(2E+2)	-	3E-10	-	4.00
		Y, see <sup>105</sup> Lu	나 가지 않으셨다.	2E+2	6E-8	2E-10	-	-
71	Lutetium-176m	W see 169	8F+3	3F+4	1F-5	3F-8	15-4	1F-3
		Y, see <sup>169</sup> Lu	-	2E+4	9E-6	3E-8	-	-

1

**T**I

			000	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µCi)	Inhala ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/m1)
71	Lutetium-176	W, see <sup>169</sup> Lu	7E+2	5E+0 Bone surf	2E-9	-	1E-5	1E-4
		Y, see <sup>169</sup> Lu		(1E+1) 8E+0	- 3E-9	2E-11 1E-11	-	1
71	Lutetium-177m	W, see <sup>169</sup> Lu	7E+2	1E+2 Bone surf	5E-8	-	1E-5	1E-4
		Y, see <sup>169</sup> Lu	-	(1E+2) 8E+1	- 3E-8	2E-10 1E-10	2	2
71	Lutetium-177	W, see <sup>169</sup> Lu	2E+3	2E+3	9E-7	3E-9	-	-1
		Y, see <sup>169</sup> Lu	(3E+3)	- 2E+3	- 9E-7	- 3E-9	41:-5	4E-4
71	Lutetium-178m <sup>2</sup>	W, see <sup>169</sup> Lu	5E+4 St. wall	2E+5	8E-5	3E-7	-	-
		Y, see <sup>169</sup> Lu	(6E+4)	- 2E+5	- 7E-5	- 2E-7	8E-4	85 3 -
71	Lutetium-173 <sup>2</sup>	₩, see <sup>169</sup> Lu	4E+4 St. wall	1E+5	5E-5	2E-7	-	-
		Y, see <sup>169</sup> Lu	(4E+4) -	- 1E+5	- 5E-5	- 2E-7	6E-4 -	6E-3
71	Lutetium-179	W, see 169 Y, see 169 Lu	۲+3	2E+4 2E+4	8E-6 6E-6	3E-8 3E-8	9E-5	9E-4

	Radionuclide		Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 F leases to ers
			Col. 1 Oral	Cel. 2	Col. 3	Col. 1	Col. 2	Me anly
			Ingestion	Inhalation				Average
No.	Radionuciide	Class	ALI (µCi)	ALI (µCi)	UAC (µCi/ml)	Air (µCi/mì)	Water (µCi/ml)	Concentratio (µCi/ml)
72	Hafnium-170	D, all compounds except those given for W	3E+3	6E+3	2E-6	8E-5	4E-5	4E-4
		W, oxides, hydroxides, carbides, and nitrates	-	5E+3	2E-6	6E-9	_	-
72	Hafnium-172	D, see <sup>170</sup> Hf	1E+3	9E+0 Bone surf	4E-9	-	2E-5	2E-4
		170	-	(2E+1)	-	3E-11	-	-
		W, see ""Hf	-	4E+1 Rone surf	2E-8	-	-	-
			-	(6E+1)	-	8E-11		- 21.0
72	Hafnium-173	D see 170Hf	5E+3	15+4	55-6	25-9	75-5	75-4
		W, see <sup>170</sup> Hf	-	1E+4	5E-6	2E-8	-	-
72	Hafnium-175	D, see <sup>170</sup> Hf	3E+3	9E+2 Bone surf	4E-7	-	4E-5	4E-4
		170	-	(1E+3)	-	18-9	-	-
		W, see <sup>170</sup> Hf	-	1E+3	5E-7	2E-9	-	-
72	Hafnium-177m <sup>2</sup>	D. see 170	2F+4	65+4	25-5	9E-9	25-1	25-2
		W, see <sup>1/0</sup> Hf	-	9E+4	4E-5	1E-7		-
72	Hafnium-178m	D, see <sup>170</sup> Hf	3E+2	1E+0 Bone surf	5E-10	-	31:-6	3E-5
		170	-	(2E+0)	-	3E-12		-
		W, see ""Hf		5E+0 Bone surf	2E-9	-	-	-
			-	(9E+0)	-	1E-11	-	+
			0c	Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-4 - - 2E-4 - 5E-3 - 5E-5 -		
---------------	---------------------------	--------------------------	----------------	------------------------	----------------------------	----------------------------------------------------------------------------------------------------------------------------------------------	-------------------	------------------------------------------------------------------------------------------------------------------------------
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Indestion	Inhala	alation			Average
Atomic No.	Radionuclide	Class	ALĨ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
72	Hafnium-179m	D, see <sup>170</sup> Hf	1E+3	3E+2 Bone surf	1E-7	-	1E-5	1E-4
				(6E+2)	-	8F-10	-	10 C C
		W, see <sup>170</sup> Hf		6E+2	3E-7	8E-10	-	
72	Hafnium-180m	D see 170Hf	75+3	25+4	05-6	25.0		
		W, see <sup>170</sup> Hf	-	3E+4	1E-5	3E-8 4E-8	11-4	18-3
70		170				12.0		
12	Hatnium-181	D, see THF	1E+3	2E+2 Rope surf	7E-8	-	2E-5	2E-4
		170		(4E+2)	_	6E-10	_	Table 3 Releases to Sewers 
		W, see <sup>170</sup> Hf		4E+2	2E-7	6E-10	-	2
72	Hafnium-182m <sup>2</sup>	D See 170 <sub>Hf</sub>	AEAA	05+4	AF . F			
		W, see 170Hf	-	3E+4 1E+5	46-5	11:-/	5E-4	5E-3
7.0		170			02 5	25-1		-
12	Hafnium-182	D, see THF	2E+2	8E-1	3E-10	-	-	Monthly Average Concentratio (µCi/ml) 1E-4 - 1E-3 - 5E-3 - 5E-5 - 3E-3 - 3E-4 -
			Bone surf	Bone surf				
		M 170mc	(4E+2)	(2E+0)	-	2E-12	5E-6	5E-5
		w, see hi		3E+0	1E-9	-	-	· · · · · · · · · · · · · · · · · · ·
			-	(7F+0)	-	15-11		
10		170				10-11		
12	Hafnium-183~	D, see 170Hf	2E+4	5E+4	2E-5	6E-8	3F-4	3F-3
		W, see Hf	-	6E+4	2E-5	8E~8	-	-
72	Hafnium-184	D coo 170 ur	25.2	25.2		1.1.1.1.1.1.1		
	LIGHTIGH LOT	W see 170 Hf	22+3	82+3	3E-6	16-8	3E-5	3E-4
		n, see 11		02+3	35-6	9E-9	-	-

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table Relea Sewe
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (uCi/ml)
73	Tantalum-1722	W all compounds except						
	Tancarum 172	those given for Y	4E+4	1E+5	5E-5	2E-7	5E-4	5E-3
		Y, elemental Ta, oxides, hydroxides, halides, carbides, nitrates,						
		and nitrides	4.796 34.95	1E+5	4E-5	1E-7	-	
73	Tantalum-173	W, see 172 Y, see 172 Ta	7E+3	2E+4 2E+4	8E-6 7E-6	3E-8 2E-8	9E-5	9E-4
73	Tantalum-174 <sup>2</sup>	W, see <sup>172</sup> Ta Y, see <sup>172</sup> Ta	3E+4	1E+5 9E+4	4E-5 4E-5	1E-7 1E-7	4E-4	4E-3
73	Tantalum-175	W, see 172 Y, see 172 Ta	6E+3	2E+4 3E+4	7E-6 6E-6	2E-8 2E-8	8E-5	8E-4
73	Tantalum-176	W, see <sup>172</sup> Ta Y, see <sup>172</sup> Ta	4E+3	1E+4 1E+4	5E-6 5E-6	2E-8 2F-8	5E-5	5E-4
73	Tantalum-177	W. see <sup>172</sup> Ta Y, see <sup>172</sup> Ta	1E+4 -	2E+4 2E+4	8E-6 7E-6	3E-8 2F-8	2E-4	2E-3
73	Tantalum-178	W, see 172Ta Y, see 172Ta	2E+4	9E+4 7E+4	4E-5 3E-5	1E-7 1E-7	2E-4	2E-3
73	Tantalum-179	W, see 172Ta Y, see 172Ta	2E+4	5E+3 9E+2	2E-6 4E-7	8E-9 1E-9	3E-4	3E-3
73	Tantalum-180m	W, see 172 Y, see 172 Ta	2E+4	7E+4 6E+4	3E-5 2E-5	9E-8 8E-8	3E-4	3E-3

			0cc	Table 1 upational V	alues	Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic	Radionuclide	Class	Ingestion	ALT DAC				Monthly Average Concentration (µCi/ml) 2E-4 - 3E-2 - 1E-4 - 2E-4 - 2E-4 -
No.	Regenere	01035	(µCi)	(µCi)	(µCi/ml)	(µCi/ml)	<pre>water (µCi/ml)</pre>	(µCi/ml)
73	Tantalum-180	W, see 172 172Ta	1E+3	4E+2	2E-7	6E-10	2E-5	2E-4
		Y, see ''Ta		2E+1	12-8	3E-11	-	-
73	Tantalum-182m <sup>2</sup>	W, see <sup>172</sup> Ta	2E+5 St. wall	5E+5	2E-4	8E-7	-	-
		172	(2E+5)	-	-	-	3F-3	3E-2
		Y, see 172Ta	-	4E+5	2E-4	6E-7	-	-
73	Tantalum-182	W see 172	95+2	25.2	15.3			
/ 5	Turreation 202	Y. SPP 172Ta	00-+2	36+2	11-7	5E-10	18-5	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 2E-4 - 3E-2 - 1E-4 - 3E-4 - 4E-3 - 1E-2 - 1E-2 - 1E-3
73	Tantalum-183	W, see <sup>172</sup> Ta	9E+2 LLI wal	1E+3	5E-7	2E-9	-	-
		v 172.	(1E+3)		-	-	2E-5	2E-4
		r, see la	-	1E+3	4E-7	12-9	-	-
73	Tantalum-184	W, see 172Ta	2E+3	5E+3	2E-6	8E-9	3E-5	3E-4
		i, see ia		51+3	2E-6	7E-9	- T	
73	Tantalum-185 <sup>2</sup>	W, see 172 Ta	3F+4	7F+4	35.5	15-7		
		Y, see <sup>1/2</sup> Ta	-	6F+4	38-5	16-1	45-4	4E-3
73	Tantalum-186 <sup>2</sup>	W, see <sup>172</sup> Ta	5E+4 St. wall	2E+5	1E-4	3E-7	-	-
		172	(7E+4)	-	_		15-2	15-2
		Y, see TreTa		2E+5	9E-5	3E-7	-	-
74	Tungsten-176	D, all compounds	1E+4	5E+4	2E-5	7E-8	1E-4	1E-3
74	Tungsten-177	D, all compounds	2E+4	9E+4	4E-5	1E-7	3E-4	38-3

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCî/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/ml)
74	Tungsten-178	D, all compounds	5E+3	2E+4	8E-6	3E-8	7E-5	7E-4
74	Tungsten-179 <sup>2</sup>	D, all compounds	5E+5	2E+6	7E-4	2E-6	7E-3	7E-2
74	Tungsten-181	D, all compounds	2E+4	3E+4	1E-5	5E-8	2E-4	2E-3
74	Tungsten-185	D, all compounds	2E-3 LLI wall	7E+3	3E-6	9E-9	-	-
			(3E+3)	-	-	-	4E-5	4E-4
74	Tungsten-187	D, all compounds	2E+3	9E+3	4E-6	1E-8	3E-5	3E-4
74	Tungsten-188	D, all compounds	4E+2 LLI wall	1E+3	5E-7	2E-9	-	-
			(5E+2)	-	-	-	7E-6	7E-5
75	Rhenium-177 <sup>2</sup>	D, all compounds except	05+4	25.45	15.4	45.2		
		chose given for #	St. wall	35+3	11-4	46-1		Releases to Sewers Monthly Average Concentration (µCi/ml) 7E-4 7E-2 2E-3 - 4E-4 3E-4 - 7E-5 - 2E-2 - - 1E-2 - IE-2 - 7E-4
			(1E+5)	-	-	-	2E-3	2E-2
		W, oxides, hydroxides, and nitrates	-	4E+5	1E-4	5E-7	-	_
75	Rhenium-178 <sup>2</sup>	D, see <sup>177</sup> Re	7E+4 St. wall	3E+5	1E-4	4E-7	-	-
			(1E+5)	-	-	-	1E-3	1E-2
		W, see Re	-	3E+5	1E-4	4E-7	-	-
75	Rhenium-181	D. see 177 Re	5E+3	9F+3	AF-F	15-9	75-5	75 . 4
		W, see 1/ Re	-	9E+3	4E-6	16-8		72-4

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewars
			Col. 1 Oral	Co1. 2	Col. 0	Col. 1	Col. 2	Monthly
Atomic	Dadiasuslide	61	Ingestion	Inhalation				Average
No.	Radionuciide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
75	Rhenium-182 (12.7 h)	D, see 177 W, see 177 Re	7E+3	1E+4 2E+4	5E-6 6E-6	2E-8 2E-8	9E-5	9E-4
75	Rhenium-182 (64.0 h)	D, see 177 W, see 177 Re	1E+3	2E+3 2E+3	1E-6 9E-7	3E-9 3E-9	2E-5	2E-4
75	Rhenium-184m	D, see 177 W, see 177 Re	2E+3	3E+3 4E+2	1E-6 2E-7	4E-9 6E-10	3E-5	3E-4
75	Rhenium-184	D, see 177 W, see Re	2E+3	4E+3 1E+3	1E-6 6E-7	5E-9 2E-9	3E-5	3E-4
75	Rhenium-186m	D, see <sup>177</sup> Re	1E+3 St. wall	2E+3 St. wall	7E-7			
		W, see <sup>177</sup> Re	(2E+3)	(2E+3) 2E+2	- 6E-8	3E-9 2E-10	2E-5	2E-4
75	Rhenium-186	D, see 177 W, see 177 Ra	2E+3	3E+3 2E+3	1E-6 7E-7	4E-9 2E-9	3E-5	3E-4
75	Rhenium-187	D, see <sup>177</sup> Re	6E+5	8E+5 St. wall	4E-4	-	8E-3	8E-2
		W, see <sup>177</sup> Re	a 1,514	(9E+5) 1E+5	- 4E-5	1E-6 1E-7	-	-
75	Rhenium-188m <sup>2</sup>	D, see 177 W, see 177 Re	8E+4	1E+5 1E+5	6E-5 6E-5	2E-7 2E-7	1E-3	1E-2
75	Rhenium-188	D, see <sup>177</sup> <sub>177</sub> Re W, see <sup>177</sup> Re	2E+3	3E+3 3E+3	1E-6 1E-6	4E-9 4E-9	2E-5	2E-4

			Table 1 Occupational Values			Tat Effi Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 4E-4 - 1E-2 - 3E-4 - 3E-4 - 3E-4 - 1E-2
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	able 2 fluent itrations Col. 2 Water (uCi/ml) 4E-5 1E-3 1 - 3E-5 - 3E-5 - 1E-3 1 - 1E-3 1 - - 2E-4 - 2E-4 - -	Monthly
	S., 45	<b>C1</b>	Ingestion	Inhal	ation			Average
No.	Radionuciide	LIASS	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Monthly Average Concentration (µCi/ml) 4E-4 
75	Rhenium-189	D, see 177Re	3E+3	5E+3	2E-6	7E-9	4F-5	4F-4
		W, see <sup>17</sup> Re	-	4E+3	2E-6	6E-9	-	
70	2 . 2002							
10	USmium-180	D, all compounds except	15.5	AFT				
		those given for w and f	16+5	42+5	22-4	5E-/	1E-3	18-2
		W, halides and nitrates	-	5E+5	2E-4	7E-7	-	-
		Y, oxides and hydroxides	-	5E+5	2E-4	6E-7	-	- 1 C
76	Ocmium-1012	0	15.4	45.4	AF . F			1. S.
10	0.2001.000-101	W see 1800s	11:+4	46+4	2E-5	6E-8	28-4	2E-3
		Y see 1800s		AE+A	25-5	6E-8		
		1, 500 05		46.4	26-3	95-9		-
76	Osmium-182	C, see 1800s	2E+3	6E+3	2E-6	8F-9	38-5	3E-4
		W, see 1800s	-	4E+3	2E-6	6E-9	-	-
		Y, see <sup>100</sup> 0s	-	4E+3	2E-6	6E-9	2	-
76	0	180						
10	0.201.000-792	U, see 1800s	22+3	55+2	2E-7	7E-10	3E-5	3E-4
		W, See 18005		8E+Z	31-1	1E-9	-	4E-4 
		r, see us	-	86+2	3E-1	1E-9	-	
76	Osmium-189m	D. see 180	RF+4	2E+5	1E-4	25-7	16-2	15-2
		W, see 1800s	-	2E+5	95-5	35-7	IE-3	10-2
		Y, see 1800s	-	2E+5	7E-5	2E-7	-	-
76	Ocmium 101m	180,						
10	05m10m-191m	W see 1800s	15+4	32+4	1E-5	4E-8	2E-4	2E-3
		v see 1800s		21-4	8E-6	3E-8	-	-
		r, see us	-	22+4	/E-6	2E-8	-	-

			Table 1 Table 2 Occupational Values Effluent Concentrations			Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-4 - 2E-4 - 8E-5 -						
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly				
Atomic	Radionuclide	Class	Ingestion	on Inhalation		Ingestion Inhalation		tion Inhalation				Monthly Average Concentration (µCi/ml) - 3E-4 - - 2E-4 - 8E-5 -
No.	ind conditioned in the	61835	(µCi)	(µCi)	UAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)				
76	Osmium-191	D, see <sup>180</sup> 0s	2E+3 LLI wall	2E+3	9E-7	3E-9	-	-				
		100	(3E+3)				35-5	35-4				
		W, see 1000s		2E+3	7E-7	2F-9	-	56.4				
		Y, see 1000s		1E+3	6E-7	2E-9	-	2				
76	Osmium-193	D, see <sup>180</sup> 0s	2E+3 LLI wall	5E+3	2E-6	6E-9	-	-				
			(2E+3)	-		121-111-1	25-5	25.4				
		W, seeOs	-	3F+3	15-6	0-35	22-5	25-4				
		Y, see <sup>200</sup> 0s		3E+3	1E-6	4E-9	-	2				
76	Osmium-194	D, see <sup>180</sup> 0s	4E+2 LLI wall	4E+1	2E-8	6E-11	-	-				
		180	(6E+2)	-	-	-	8F-6	8E-5				
		W, see 1800s		6E+1	2E-8	8E-11	-	-				
		Y, see 1000s		8E+0	3E-9	1E-11	-	-				
77	Iridium-182 <sup>2</sup> ·	D, all compounds except										
		those given for W and Y	4E+4 St. wall	1E+5	6E-5	2E-7	-	-				
		W, halides, nitrates,	(4E+4)	-	-	-	6E-4	6E-3				
		and metallic iridium	-	2E+5	6E-5	2E-7	4	-				
		Y, oxides and hydroxides	-	1E+5	5E-5	2E-7	-	_				
77	Inidium-104	D 182.										
	11.101.00.104	0, see 1821r	8E+3	2E+4	1E-5	3E-8	1E-4	1F-3				
		w, see 1821r	-	3E+4	1E-5	5E-8	-					
		r, see ir		3E+4	1E-5	4F-8	-					

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Ingestion	Inhal	lation			Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 7E-4 - 3E-4 - 3E-4 - 3E-4 - 7E-4 - 2E-2 - 1E-3
Atomic No.	Radionuclide	LIASS	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	
77	Iridium-185	0, see 182 101	5E+3	1F+4	58-6	2F-8	78-5	75-4
		W, see 182		1E+4	5E-6	2E-8	-	16.4
		Y, see <sup>182</sup> Ir	-	1E+4	4E-6	1E-8	141.01	2
77	Iridium-186	0 see 182 Ir	25+3	85+3	25-6	15-0	2F F	
		W see 1821r	-	6542	25-6	10-0	36-3	ns Sewers 2 Monthly Average r Concentration (µCi/ml) 7E-4 - 3E-4 - 3E-4 - 3E-4 - 2E-2 - -
		Y, the <sup>102</sup> Ir	-	6E+3	28-6	8E-9	2	2
77	Iridium-187	D see 182	15+4	75+4	15.5	55 A		
	x111010m (07	W see 1821r	16.44	35+4	16-5	55-8	1E-4	16-3
		Y, see <sup>182</sup> Ir	~	3E+4	16-5	4E-8 4E-8	2	<u>.</u>
77	Iridium-199	D con 182	15.2					
· ·	TI IGIGH TOO	W see 1821	2E+3	52+3	28-6	6E-9	3E-5	3E-4
		v, see 1821		46+3	11-6	5E-9		18 I I I I I I I I I I I I I I I I I I I
		r, see ir		3E+3	16-6	5E-9	-	÷
77	Iridium-189	D, see <sup>182</sup> Ir	5E+3 LLI wall	5E+3	2E-6	7E-9	÷	-
		182	(5E+3)	-		-	76-5	7E-4
		W, see 182Ir	-	4E+3	2E-6	5E-9	1400 - N	27 C
		Y, see Ir		4E+3	1E-6	5E-9	- 11 A	동안 전 등
77	Iridium-190m <sup>2</sup>	D, see 182	2E+5	2E+5	RE-5	35-7	25-2	25-2
		W, see 102Ir		2E+5	9E-5	35-7	22.3	20-2
		Y, see <sup>102</sup> Ir		2E+5	8E-5	3E-7	-	-
77	Iridium-190	D, see 182	15+3	QE+2	45-7	15-0	15.5	
		W, see 182 Ir	-	1E+3	45-7	16-9	1E-2	16-4
		Y, see <sup>182</sup> 1r		QE+2	46-7	15-9		
				SETE	46-1	16-9	-	-

			0cc	Table 1 supational N	Values	Tal Eff Concen	ble 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	ble 2 luent trations Col. 2 Water (µCi/ml) 4E-5 - - 1E-5 - - 1E-5 - - 1E-4 - 2E-4 - 2E-4 - 2E-4 - 2E-5 1E-4 - 5E-5	Monthly
			Ingestion	Inha	lation			Average
No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	UAC (µCi/ml)	Air (µCi/ml)	<pre>water (µCi/ml)</pre>	(µCi/m])
77	Iridium-192m	D, see 182 1021r	3E+3	9E+1	4E-8	1E-10	4E-5	4E-4
		W, see 102Ir		2E+2	9E-8	3E-10	-	-
		Y, see <sup>182</sup> Ir	-	2E+1	6E-9	2E-11	-	-
77	Iridium-192	D con 182 r	QE+2	35+2	15-7	45-10	16-5	16-4
· · · · ·	1110108 1.52	W coo 1821	JL'L	AE42	26-7	6E-10	16 5	11 4
		Y, see <sup>182</sup> Ir		2E+2	9E-8	3E-10	-	-
77	Inidium-104m	D 182	65.12	05+1	45-0	15.10	NF (	05.5
"	1110100-1940	U, See 18217	0545	25+2	46-8	16-10	96-0	9E-3
		Y, see 1821r		1E+2	4E-8	1E-10	-	-
		182						
11	Iridium-194	D, see 182 Ir	1E+3	3E+3	1E-6	4E-9	1E-5	1E-4
		W, see 1821r		2E+3	9E-7	3E-9	-	-
		Y, see Ir		2E+3	8E-7	3E-9		-
77	Iridium-195m	D, see 182 Ir	8E+3	2E+4	16-5	3F-8	1F-4	1E-3
		W, see 182Ir		3E+4	16-5	41-8	-	-
		Y, see <sup>182</sup> Ir		2E+4	9E-6	3E-8	-	-
77	Iridium-195	D see 182	15+4	AE+A	25-5	65-0	25-4	25-2
	ATTOTON 135	W see 1821	11.4	55+4	25-5	75-9	21-4	20-3
		Y, see <sup>182</sup> Ir	-	4E+4	2E-5	6E-8	-	2
78	Platinum-186	D, all compounds	1E+4	4E+4	2E-5	5E-8	2E-4	2E-3
78	Platinum-188	D, all compounds	2E+3	2E+3	7E-7	28-9	2E-5	2E-4
78	Platinum-189	D, all compounds	1E+4	3E+4	15-5	4E-8	1E-4	1E-3
78	Platinum-191	D, all compounds	4E+3	8E+3	4E-6	1E-8	SE-5	5E-4

1.5%

		0cc	/alues	Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 4E-4 - 6E-3 - 3E-4 2E-3 4E-4 7E-3 2E-4 1E-3 -	Table 3 Releases to Sewers
		Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Radionuclide	Class	ALI	ALI	DAC	Air	Water	Monthly Average Concentration (µCi/ml) - 4E-4 - 6E-3 - 3E-4 2E-3 4F-4
		(µCi)	(µCi)	(µCi/ml)	(µCi/ml)	(µCi/ml)	(µCi/ml)
Platinum-193m	D, all compounds	3E+3 \LI wall	6E+3	3E-6	8E-9	-	-
		(3E+4)	아이 가지?		-	4E-5	4E-4
Platinum-193	D, all compounds	4E+4 LLI wall	2E+4	1E-5	3E-8		
		(5E+4)		-	-	6E-4	6E-3
Platinum-195m	D, all compounds	2E+3 LLI wall	4E+3	2E-6	6E-9	-	-
		(2E+3)	~		*	3E-5	3E-4
Platinum-197m <sup>2</sup>	D, all compounds	2E+4	4E+4	2E-5	6E-8	2E-4	2E-3
Platinum-197	D, all compounds	3E+3	1E+4	4E-6	1E-8	4t-5	4E-4
Platinum-199 <sup>2</sup>	D, all compounds	5E+4	1E+5	6E-5	2E-7	7E-4	7E-3
Platinum-200	D, all compounds	1E+3	3E+3	1E-6	58-9	2E-5	2E-4
Gold-193	D, all compounds except those given for W and Y	9E+3	3E+4	1E-5	4E-8	1E-4	IE-3
	W, halides and citrates	-	2E+4	9E-6	3E-8	-	2
	Y, oxides and hydroxides		2E+4	8E-6	3E-8	-	
Gold-194	D, see 193Au	3E+3	8E+3	3E-6	1E-8	4E-5	48-4
	Y, see 193Au	10119874	56+3	28-6	88-9		-
	Radionuclide Platinum-193m Platinum-193 Platinum-195m Platinum-197m <sup>2</sup> Platinum-199 <sup>2</sup> Platinum-200 Gold-193 Gold-194	RadionuclideClassPlatinum-193mD, all compoundsPlatinum-193mD, all compoundsPlatinum-193D, all compoundsPlatinum-195mD, all compoundsPlatinum-197m <sup>2</sup> D, all compoundsPlatinum-197D, all compoundsPlatinum-197D, all compoundsPlatinum-197D, all compoundsPlatinum-193D, all compoundsPlatinum-193D, all compoundsSold-193D, all compounds except those given for W and YW, halides and citratesY, oxides and hydroxidesGold-194D, see 193 W, see 193 Au Y, see 193	RadionuclideClassCol. 1 Oral Ingestion ALI (µCi)Platinum-193mD, all compounds3E+3 'LI wall (3E+4)Platinum-193D, all compounds4E+4 LLI wall (SE+4)Platinum-195mD, all compounds2E+3 LLI wall (SE+4)Platinum-197mD, all compounds2E+3 LLI wall (ZE+3)Platinum-197m2D, all compounds2E+4Platinum-197m2D, all compounds3E+3Platinum-197D, all compounds3E+3Platinum-1992D, all compounds3E+3Platinum-200D, all compounds1E+3Gold-193E, all compounds except those given for W and Y9E+3W, halides and mitrates-Y, oxides and hydroxides-Y, oxides and hydroxides-Gold-194D, see 193Au W, see 193Au W, see 193Au L3E+3 -	Table 1 Occupational VRadionuclideClass $\overline{\begin{array}{c} Col. 1 \\ 0ral \\ Ingestion \\ ALI \\ (\mu^{Ci}) \end{array}} \\ \hline \\ \hline \\ Col. 1 \\ Oral \\ Ingestion \\ ALI \\ (\mu^{Ci}) \end{array} \\ \hline \\ \hline \\ \hline \\ Platinum-193 \\ Platinum-193 \\ D, all compounds \\ \hline \\ Platinum-193 \\ D, all compounds \\ \hline \\ \hline \\ Platinum-195 \\ D, all compounds \\ \hline \\ \hline \\ Platinum-195 \\ D, all compounds \\ \hline \\ \hline \\ Platinum-197 \\ D, all compounds \\ \hline \\ Platinum-199 \\ D, all compounds \\ \hline \\ Platinum-200 \\ D, all compounds \\ \hline \\ Platinum-200 \\ D, all compounds \\ \hline \\ \hline \\ Platinum-200 \\ D, all compounds \\ \hline \\ \hline \\ Platinum-200 \\ D, all compounds \\ \hline \\ \hline \\ Platinum-200 \\ D, all compounds \\ \hline \\ $	Table 1 Occupational ValuesRadionuclideClassCol. 2 Oral Ingestion ALI $\mu(ci)$ Col. 2 $\mu(ci)$ Col. 3 Inhalation ALI $\mu(ci)$ Platinum-193mD, all compounds $3E+3$ $UII wall(3E+4)6E+3 3E-5UII wall(3E+4)Platinum-193D, all compounds4E+4UII wall(3E+4)2E+4 1E-5UII wall(2E+3)Platinum-193D, all compounds2E+3UII wall(2E+3)4E+4 2E-6UII wall(2E+3)Platinum-197mD, all compounds2E+4UII wall(2E+3)4E+4 2E-5-Platinum-197mD, all compounds2E+4UII wall(2E+3)4E-6-Platinum-197mD, all compounds3E+3E+41E-6Platinum-197D, all compounds1E+33E+31E-6Gold-193D, all compounds exceptthose given for W and YW, halides and mitrates-2E+4 2E+44E-6Gold-194D, see \frac{193}{19}AuW, see \frac{193}{4}AuW, see \frac{193}{4}Au3E+3- 8E+3- 8E+3--$	Table 1 DCCUPAtional Values         Tat Eff Concrete Col. 1 Ocil. 1 Oral Ingestion Indelation All (pCi)         Table 1 Def Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Concrete Co	Table 1 Decupational Value         Table 2 Efficient Concurtations           Radionuclide         Cols 1 Oral Innalation ALI (µCi)         Cols 2 Cols 3         Cols 2 Cols 3         Cols 2 Cols 2           Radionuclide         Class         Cols 2 (µCi)         Cols 2 (µCi)         Cols 3         Cols 1         Cols 2 Cols 3           Platinum-193m         D, all compounds         3£+3 (11 wall (3E+4)         Cel 3         3E-6         8E-9         -           Platinum-193         D, all compounds         3£+3 (11 wall (3E+4)         Cel 3         3E-6         8E-9         -           Platinum-193         D, all compounds         2E+3 (LII wall (2E+3)         2E+4         1E-5         3E-8         -           Platinum-197m         D, all compounds         2E+3 (LII wall (2E+3)         4E+4         2E-5         6E-8         2E-4           Platinum-197m <sup>2</sup> D, all compounds         2E+4         4E+4         2E-5         6E-8         2E-7         7E-4           Platinum-197         D, all compounds         3E+3         3E+4         1E+5         6E-5         2E-7         7E-4           Platinum-197         D, all compounds         1E+3         3E+4         1E-5         4E-8         1E-4

			0cc	Table 1 cupational (	Values	Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Cel. 3	Col. 1	lie 2 luent rations Col. 2 Water (μCi/ml) 7E-5 1E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5 2E-5	Monthly
			Ingestion	Inhalation				Average
No.	Radionaclide	CTASS	ALI (µCi)	ALI (µCi)	(µCi/ml)	(µCi/ml)	water (μCi/ml)	(pCi/ml)
79	Gold-195	D, see 193Au	5E+3	1E+4	5E-6	2E-8	7E-5	7E-4
		W, see 193Au	이 아이 아이는 아이 가지?	1E+3	6E-7	2E-9	-	-
		Y, see <sup>193</sup> Au		4E+2	28-7	66-10	-	-
79	Gold-198m	D see 193Au	15+3	3E33	15-6	05-30	16-5	15-4
,,,	GOIG TOWN	W see 193Au	11.5	1E+3	SE-7	2E-9		Table 3 Releases to Sewers Monthly Average Concentratic (µCi/ml) 7E-4 - 1E-4 - 2E-4 - 2E-4 - 2E-4 - 4E-3 - 1E-2 -
		Y, see 193Au		1E+3	5E-7	2E-9	-	-
79	Cold-109	D can 193	16+3	AE+3	25-6	55-0	25-5	25-4
	0016 130	W 500 193Au	-	25+3	8F-7	31-0	20 3	25-4
		Y, see <sup>193</sup> Au		2E+3	7E-7	2E-9	-	
79	Gold-199	D, see <sup>193</sup> Au	3E+3	9E+3	45-6	1E-8	-	-
			(3E+3)	-		-	48-5	Releases to Sewers Monthly Average Concentration (µCi/ml) 7E-4 - 1E-4 - 2E-4 - 2E-4 - 4E-4 - 1E-2 - 1E-2
		W, see 193Au		4E+3	2E-6	6F-9	-	-
		Y, see <sup>193</sup> Au		4E+3	2E-6	58-9	-	-
79	Gold-200m	D see 193	1F+3	<b>4</b> E+2	16-6	55-9	25-5	25-4
	GOTO LOOM	W see 193Au	-	35+3	16-6	AF-0	26-3	20-4
		Y, see <sup>193</sup> Au		2E+4	15-6	3E-9	-	
79	Cold-2002	D con 193	25+4	CELA	25-5	05.0		15.3
/ 3	0010 200	W, see 193Au	3544	0000	35-3	95-6	42-4	46-3
		Y, see 193Au		7E+4	36-5	1E-7 1E-7	-	-
70	Cald 2012	0 193,	75.4	25.5	0.F. F.			
13	G010-231	u, see au	/E+4 St. (2013)	26+5	9E-5	31-1	-	-
			(OELA)	1.1.1			15.2	15.0
		W see 193	(32.74)	25.45	15-4	25-7	16-3	16-2
		V con 193		2015	05 5	35-7		
		r, see Au		26+3	96-5	SE-1	-	-

instant of

			0cc	Table 1 Occupational Values			ole 2 luent crations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Co1. 3	Col. 1	Col. 2	Monthly
Atomic	Radionuclide	Class	Ingestion	Inhal	ation		Makau	Average
No.	Radionacii rae	01055	(µCi)	(µCi)	(µCi/ml)	(µCi/ml)	(µCi/ml)	Average Concentration (µCi/m1) - 6E-4 4E-4 - 3E-3 2E-3 -
80	Mercury-193m	Vapor		8E+3	4E-6	1E-8	-	-
		Organic D	4E+3	1E+4	5E-6	2E-8	6E-5	6E-4
		D, sulfates	3E+3	9E+3	4E-6	18-8	4E-5	4E-4
		W, oxides, hydroxides, halides, nitrates, and						
		sulfides	-	8E+3	3E-6	1E-8	-	-
80	Mercury-193	Vapor		3E+4	1E-5	4F-8	_	_
		Organic D	2E+4	6E+4	3E-5	9E-8	38-4	3E-3
		0 coo 193mua	25+4	45.14	05 F			
		W see 193mHg	2574	45+4	25-0	6E-8	ZE-4	2E-3
		w, see ing		41.14	2C-3	02-8	-	
80	Mercury-194	Vapor		3E+1	1E-8	4F-11	-	
		Organic D	2E+1	3E+1	1E-8	4E-11	2E-7	2E-6
		D see 193mHa	8E+2	AE+1	25-9	65.11	15.6	
		W, see <sup>193m</sup> Hg	-	1E+2	5E-8	2E-10	-	16-4
80	Mancunu-105m	Vanan		45.0				
00	Hercury 195m	Organic D	-	42+3	2E-6	6E-9	-	-
		organic b	35+3	0E+3	3E-6	8E-3	48-5	4E-4
		D, see 193mHg	2E+3	5F+3	2E-6	7F-9	35-5	25-4
		W, see <sup>193m</sup> Hg		4E+3	2E-6	5E-9	-	-
80	Mercury-195	Vapor	11.000	3F+4	15-5	AE-9		
		Organic D	2E+4	5F-4	2F-5	6E-8	25-4	25-2
		193m			CL 3	000	26.4	20-3
		D, see 193mHg	1E+4	4E+4	1E-5	5E-8	2E-4	2E-3
		W, see Hg		3E+4	1E-5	5E-8	-	

			Table 1 Occupational Values			Tat Effi Concent	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 5E-4 4E-4 - - 9E-4 8E-4 - - 1E-2 8E-3 - 7E-5 3E-4 -
		Class	Ingestion	Inhal	lation			Average
Atomic No.	Radionuclide		ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
80	Mercury-197m	Vapor	_	5F+3	2F-6	7E-9	-	
		Organic D	4E+3	9E+3	4E-6	1E-8	5E-5	5E-4
		D coo 193m	25.12	75.2	25.5			
		u, see 193m.	3E+3	12+3	3E-6	1E-8	4E-5	4E-4
		w, see Hg		5E+3	2E-6	7E-9	-	-
80	Mercury-197	Vapor		8E+3	4F-6	1F-8	_	
		Organic D	7E+3	1E+4	6E-6	2E-8	9E-5	9E-4
		D 193m.	15.0					
		U, see 193m	6E+3	12+4	5E-6	2E-8	8E-5	Monthly Average Concentration ) (µCi/ml) 
		w, see Hg		9E+3	4E-6	1E-8	-	+
80	Mercury-199m <sup>2</sup>	Vapor		RE+A	25-5	15-7		
		Organic D	65+4	25+5	75-5	25-7	-	-
		organity b	St. wall	22+5	75-5	2E-1	-	-
		100	(1E+5)	-	-		15-2	Monthly Average Concentration ) (µCi/m1) 
		D, see 193mHg	6F+4	1F+5	65-5	25-7	10-3	11-2
		W, see <sup>193m</sup> Hg	-	2E+5	7E-5	28-7	86-4	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 5E-4 4E-4 - - 9E-4 8E-4 - - 1E-2 8E-3 - 7E-5 3E-4 - - 1E-2
00								
80	Mercury-203	Vapor		8E+2	4E-7	1E-9	- S	14
		Organic D	5E+2	8E+2	3E-7	1E-9	7E-6	7E-5
		D soo 193mu	25+3	15.3				5E-4 4E-4 - 9E-4 8E-4 - 1E-2 8E-3 - 7E-5 3E-4
		W soo 193mus	22+3	16+3	5E-/	ZE-9	38-5	3E-4
		w, see ng	그는 가지 않는 것이 없다.	16+3	5E-7	5E-3	-	-
61	Thallium-194m <sup>2</sup>	D, all compounds	5E+4 St. wall	2E+5	6E-5	2E-7	-	-
			(7E+4)	-		-	1E-3	Monthly Average Concentration (µCi/ml) - 5E-4 4E-4 - - 9E-4 8E-4 - - 1E-2 8E-3 - 7E-5 3E-4 - - 7E-5 3E-4

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/m1) - 4E-2 9E-3 1E-2 4E-3 3E-3 9E-3 1E-3 9E-3 1E-3 2E-3 5E-4 2E-4 8E-3 4E-3 4E-3 4E-3 4E-3
Atomic No.	Radionuclide	Class	Ingestion ALT (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentratio (µCi/ml)
81	Thallium-194 <sup>2</sup>	D, all compounds	3E+5 St. wall	6E+5	2E-4	8E-7	-	-
			(3E+5)			-	4E-3	4E-2
81	Thallium-195 <sup>2</sup>	D, all compounds	6E+4	1E+5	58-5	2E-7	9E-4	9E-3
81	Thallium-197	D, all compounds	7E+4	1E+5	5E-5	2E-7	1E-3	1E-2
81	Thallium-198m <sup>2</sup>	D, all compounds	3E+4	5E+4	2E-5	8E-8	4E-4	4E-3
81	Thallium-1/38	D, all compounds	2E+4	3E+4	1E-5	5E-8	3E-4	3E-3
81	Thallium-199	D, all compounds	6E+4	8E+4	4E-5	1E-7	9E-4	9E-3
81	Thallium-200	D, all compounds	8E+3	1E+4	5E-6	2E-8	1E-4	1E-3
81	Thallium-201	D, all compounds	2E+4	25+4	9E-6	3E-8	2E-4	2E-3
81	Thallis-202	D, all compounds	4E+3	5E+3	2E-6	7E-9	5E-5	5E-4
81	Thall'um 204	D, all compounds	2E+3	2E+3	9E-7	3F-9	28-5	2E-4
82	Lead-195m <sup>2</sup>	D, all compounds	6E+4	2E+5	8E-5	3E-7	8E-4	8E-3
82	Lead-198	D, all compounds	3E+4	6E+4	38-5	9E-8	4F-4	4F-3
82	Lead-199 <sup>2</sup>	D, all compour.is	2E+4	7E+4	3E-1	1E-7	3E-4	3E-3
82	Lead-200	D, all compounds	3E+3	6E+3	3E-6	9E-9	4E-5	4E-4
82	Lead-201	D, all compounds	7E+3	2E+4	8E-6	3E-8	16-4	1E-3

			Table 1 Tab Occupational Values Efflic Concent			ble 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-3 2E-5 7E-4 5E-4 3E-3 - 1E-7 2E-5 1E-7 2E-5 1E-3 4E-3	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 1E-3 2E-5 7E-4 5E-4 3E-3 - 1E-7 2E-3 - 2E-5 1E-3 4E-3 - 2E-5
			Ingestion	Inhala	ation	8. m i i		
Atomic No.	Racionu.lide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
82	Lead-202m	D, all compounds	9E+3	3E+4	1E-5	4E-8	1E-4	1E-3
82	Lead-202	D, all compounds	1E+2	5E+1	2E-8	7E-11	2E-6	2E-5
82	Lead-203	D, all compounds	5E+3	9E+3	4E-6	1E-8	7E-5	7E-4
82	Lead-205	D, all compounds	4E+3	1E+3	6E-7	2E-9	5E-5	5E-4
82	Lead-209	D, all compounds	2E+4	6E+4	2E-5	8E-8	3E-4	3E-3
82	Lead-210	D, all compounds	6E+1 Bone surf	2E+1 Bone surf	1E-10	- 1	-	-
			(1E+0)	(4E-1)		6E-13	1E-8	1E-7
82	Lead-211 <sup>2</sup>	D, all compounds	1E+4	6E+2	3E-7	9E-10	2E-4	2E-3
82	Lead-212	D, all compounds	8E+1	3E+1	1E-8	5E-11	-	
			(1E+2)	한 같이 가 관람이 있다.			2E-6	28-5
82	Lead-214 <sup>2</sup>	D, all compounds	9E+3	8E+2	3E-7	1E-9	1E-4	1E-3
03	Picmuth-2002	D. situates	25.4	05.4				
0.3	Dismoth 200	W, all other compounds	- -	1E+5	46-5	16-7	42-4	45-3
0.3	0:	200						
83	Bismuth-201	V, see 200 <sup>B1</sup> W, see 200 <sup>B1</sup>	1E+4	3E+4 4F+4	1E-5 2E-5	4E-8 5E-8	2E-4	2E-3
1.1	2	200						
83	Bismuth-202°	D, see 200Bi	1E+4	4E+4	2E-5	6E-8	2E-4	2E-3
		w, see 81		8E+4	31-5	1E-/	-	

IN DI IL

			0cc	Tab Effi Concent	Table 3 Releases to Sewers			
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-4 - 9E-5 - 1E-4 - 8E-6 - 1E-4 - 7E-4 - 1E-3
0.817	그렇게 걸고 물건 것		Ingestion	Inhalation				Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
83	Bismuth-203	D. see 200Bi	2F+3	7E+3	35-6	95-9	35-5	75-4
		W, see <sup>200</sup> Bi	-	6E+3	3E-6	9E-9	-	-
83	Rismuth-205	D see 200 <sub>Ri</sub>	15+3	35+3	15-6	25-0	25.5	25
	Dromotin 200	W, see <sup>200</sup> Bi	-	1E+3	5E-7	2E-9	-	
83	Rismuth-206	0 see 200 <sub>Bi</sub>	6542	15.42	65.7	25.0	AF . C	
	01580001 200	W, see 200Bi	-	9E+2	4E-7	1E-9	92-6	9E-5
83	Rismuth-207	D see 200 <sub>Ri</sub>	1542	25+3	75.7	25.0	15.5	
		W, see <sup>200</sup> Bi	-	4E+2	1E-7	5E-10	16-5	11-4
83	Bismuth-210m	D. see <sup>200</sup> Bi	4F+1	55+0	25-9			
			Kidneys	Kidneys	22 3			
		N con 200ps	(6E+1)	(6E+0)	-	9E-12	8E-7	8E-6
		w, see DI		/t-1	3E-10	9E-13	-	-
83	Bismuth-210	D, see <sup>200</sup> Bî	8E+2	2E+2 Kidneys	1E-7	-	1E-5	1E-4
		200		(4E+2)	-	5E-10	-	-
		W, see Bi		35+1	1E-8	4E-11	-	-
83	Bismuth-2122	D, see 200Bi	5E+3	2E+2	15-7	35-10	75-5	75-4
		W, see <sup>200</sup> Bi		3E+2	1E-7	4E-10	-	-
83	Bismuth-213 <sup>2</sup>	D, see 200Bi	7E+3	3E+2	1E-7	4E-10	1E-4	1E-3

			Table 1Table 2Occupational ValuesEffluentConcentrations				Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-3 - 3E-3 - 3E-3 - 1E-3 - 4E-7 - 8E-4 - 2E-5	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Co1. 2	Monthly	
			Ingestion	Inhala	ation			Average	
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi∕ml)	Concentratio (µCi/ml)	
83	Bismuth-214 <sup>2</sup>	D, see <sup>200</sup> Bi	2E+4 St. wall	8E+2	3E-7	1E-9	-	-	
		230	(2E+4)	-			3F-4	Releases to Sewers Monthly Average Concentration ) (µCi/ml) - 3E-3 - 3E-3 - 3E-3 - 3E-3 - 4E-7 - 8E-4 - 2E-5 -	
		W, see <sup>200</sup> Bi	-	9E-2	4F-7	1F-9		56 5	
84	Polonium-203 <sup>2</sup>	D, all compounds except those given for W	3E+4	6E+4	3E-5	9E-8	3E-4	3E-3	
		W, oxides, hydroxides, and nitrates		9E+4	45-5	1E-7	_	_	
0.4	Delesium 2002	203							
04	POTOTTUR-205	U, see 203 Po	2E+4	4E+4	2E-5	5E-8	3E-4	3E-3	
		W, see Po		7E+4	3E-5	1E-7	-	-	
84	Polonium 207	203							
04	P01011011-207	U, see 203 PO	8E+3	3E+4	1E-5	3E-8	1E-4	1E-3	
		w, see Po	-	3E+4	16-5	4E-8	-	-	
9.6	Polonium-210	D							
04	L0100108-510	U, see 203 Po	3E+0	6E-1	3E-10	9E-13	4E-8	4E-7	
		w, see Po		6E-1	3E-10	9E-13	-	-	
85	Actation-2072	D balides	65.0						
0.5	hacacine 207	U, nalides	bE+3	3E+3	1E-6	4E-9	8E-5	8E-4	
				2E+3	9E-7	3E-9	-	(A) ( ) ( ) ( ) ( )	
85	Astatine-211	D halider	15.0	05-1					
00	Houde HIC CIT	w	16+5	8E+1	3E-8	1E-10	2E-6	2E-5	
				5E+1	ZE-8	8E-11	-	-	
86	Radon-220	With daughtens							
00	10001 220	with uaughters							
		With daughtens		ZE+4	7E-6	2E-8	~	Sewers Monthly Average Concentration (μCi/ml) - 3E-3 - 3E-3 - 3E-3 - 3E-3 - 3E-3 - 3E-3 - 3E-3 - 3E-5 - 2E-5 - - -	
		with daughters							
		present	-	2E+1	9E-9	3E-11	-	-	
			(or le	12 working vel months)	(or 1.0 working level)				

			0cc	Table 1Table 2TableOccupational ValuesEffluentReleasConcentrationsSewers		Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - - 3E-4 8E-5 - 1E-6 - 2E-6 - 2E-6 -		
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-4 8E-5 - 1E-6 - 2E-6 - 2E-6 - 6E-7 -
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/ml)
86	Radon-222	With daughters removed		1E+4	4E-6	1E-8	-	-
		present	- (or le	1E+2 4 working evel months)	3E-8 (or 0.33 working level)	1E-10		-
87	Francium-222 <sup>2</sup>	9, all compounds	2E+3	5E+2	2E-7	6E-10	3E-5	3E-4
87	Francium-223 <sup>2</sup>	D, all compounds	6E+2	8E+2	3E-7	1E-9	8E-6	8E-5
88	Radium-223	W, all compounds	5E+0 Bone surf	7E-1	3E-10	9E-13	-	-
			(9E+0)	-	-	-	1E-7	18-6
88	Radium-224	W, all compounds	8E+0 Bone surf	2E+0	7E-10	2E-12	-	-
			(2E+1)		-	-	2E-7	2E-6
88	Radium-225	W, all compounds	8E+0 Bone surf	7E-1	3E-10	9E-13	-	-
			(2E+1)	-	-	-	2E-7	2E-6
88	Radium-226	W, all compounds	2E+0 Bone surf	6E-1	3E-10	9E-13	-	- 1
			(5E+0)	-	-	-	8-36	6E-7
88	Radium-227 <sup>2</sup>	W, all compounds	2E+4 Bone surf	1E⊹4 Bone surf	6E-6	-	-	-
			(2E+4)	(2E+4)	-	3E-8	3E-4	3E-3

			Table 1Table 2Occupational ValuesEffluentConcentrations				Table 1Table 2Table 3Occupational ValuesEffluentReleases toConcentrationsSewers		Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-7 - 3E-4 - 7E-6 - - 2E-5 -
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 6E-7 - 3E-4 - 7E-6 - 7E-6 - 5E-8 -	
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/m1)	Water (µCi/ml)		
88	Radium-228	W, all compounds	2E+0 Bone surf	1E+0	5E-10	2E-12	-	-	
			(4E+0)	-	-	-	6E-8	6E-7	
89	Actinium-224	D, all compounds except							
		those given for W and Y	2E+3 LLI wall	3E+1 Bone surf	1E-8	-	-	-	
			(2E+3)	(4E+1)	-	5E-11	3E-5	3E-4	
		W, halides and nitrates	-	5E+1	28-8	7E-11	-		
		Y, oxides and hydroxides	-	5E+1	2E-8	6E-11	-		
89	Actinium-225	D, see <sup>224</sup> Ac	5E+1 LLI wall	3E-1 Bone surf	1E-10	-	-	-	
		224.	(5E+1)	(5E-1)	-	7E-13	7E-7	7E-6	
		w, see 224 AC	-	6E-1	3E-10	9E-13	-	2	
		r, see Ac	(C. 1997)	6E-1	3E-10	9E-13	-	-	
89	Actinium-226	D, see <sup>224</sup> Ac	1E+2 LLI wall	3E+0 Bone surf	1E-9	-	-	-	
		224	(1E+2)	(4F+0)	-	55-12	25.5	AF	
		W, see 224Ac	-	5E+0	25-9	75-12	25-0	28-5	
		Y, see 224 Ac	-	5E+0	2E-9	6F-12	2	2	
00	A-1: : 007	224				OL AL		Table 3 Releases to Sewers Monthly Average Concentratio (µCi/ml) - 6E-7 - 3E-4 - 7E-6 - - 2E-5 - 5E-8 -	
0.9	ACTINIUM-22/	D, see Ac	2E-1 Bone surf	4E-4 Bone surf	2E-13	-	-	-	
		224.	(4E-1)	(8E-4)	-	1F-15	5E-9	55-0	
		W, see Ac	-	2E-3 Bone surf	7E-13	-	-	-	
		224	-	(3E-3)	+	4F-15	-		
		Y, see Ac	-	4E-3	2E-12	6E-15	-	-	

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Co1. 2	Col. 3	Col. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) 3E-4 - - - - - - - - 2E-5 - - 2E-6 - - - 2E-7 -
			Ingestion	Inhala	tion			Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
89	Actinium-228	D, see <sup>224</sup> Ac	2E+3	9E+0 Bone surf	4E-9	-	3E-5	3E-4
		224	-	(2E+1)	-	2E-11	-	-
		W, see <sup>224</sup> Ac		4E+1 Bone surf	2E-8	-	-	-
		224		(6E+1)		8E-11	-	-
		Y, see <sup>224</sup> Ac	-	4E+1	2E-8	6E-11	-	-
90	Thorium-226 <sup>2</sup>	W, all compounds except						
		those given for Y	5E+3 St. wall	2E+2	6E-8	2E-10	-	-
			(5E+3)		-	-	7E-5	7E-4
		Y, oxides and hydroxides	-	1E+2	6E-8	2E-10	-	-
90	Thorium-227	W SPP 226Th	15+2	35-1	15-10	55-12	25.0	AF. 1
		Y, see <sup>226</sup> Th	-	3E-1	1E-10	5E-13	26-0	26-5
90	Thorium-228	W, see <sup>226</sup> Th	6E+0	1E-2	4E-12	-	1	
			Bone surf	Bone surf				
		226	(1E+1)	(2E-2)	-	3E-14	2E-7	2E-6
		r, see th		2E-2	7E-12	2E-14	-	-
90	Thorium-229	W, see <sup>226</sup> Th	6E-1 Boug surf	9E-4	4E-13	-	4.11	-
		225	(1E+0)	(2F-3)	_	25-15	25-0	25.7
		Y, see <sup>226</sup> Th	-	2E-3 Bone surf	1E-12	-	-	-
			-	(3E-3)	-	4E-15	-	-

			0c	Table 1 cupational Va	lues	Tat Eff Concent	ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - IE-6 - 5E-4 - 5E-7 - 5E-5 - 5E-4 - 2F-4
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	
		그는 것 같은 것은 것 같은 것 같이 많이 많이 많이 했다.	Ingestion	Inhalation				Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	(µCi/ml)	(µCi/ml)	(µCi/ml)	(µCi/ml)
90	Thorium-230	W, see 226Th	4E+0	6E-3	3E-12	-	-	-
			Bone surf	Bone surf				
		226	(9E+0)	(2E-2)	-	2E-14	16-7	1E-6
		Y, see 220		2E-2 Bone surf	6E-12	-	-	-
			19 <b>-</b>	(2E-2)		3E-14	-	-
00	Than : 221	N 226 Th	4542	65+3	25-6	0-30	55-5	55-4
90	INOTIUM-231	Y, see 226Th	46+3	6E+3	3E-6	9E-9	-	56-4
		226						
90	Thorium-232	W, see 220Th	7E-1	1E-3	5E-13	-	-	· · · · · · · · · · · · · · · · · · ·
			Bone surf	Bone surf				
		226	(2E+0)	(3E-3)	-	4E-15	3E-8	3E-7
		Y, see 220Th		3E-3 Rone surf	16-32	-	-	
			-	(4E-3)	-	6E-15	-	-
00	Thursday 224	22672	25.12	05.0	or o	25.10		
90	Inorium-234	w, see in	35.72	25+2	85-9	36-10	-	
			(AE+2)				er e	er. e
		V con 226 Th	(45+2)	25+2	65-8	2E-10	56-90	5E-5
	사망 옷을 가져야 할 때?	1, see 11		er.e	UL O	CL IV		
91	Protactinium-2272	W. all compounds except						
F		those given for Y	4E+3	1E+2	5E-8	2E-10	5E-5	5E-4
		Y, oxides and hydroxides	-	1E+2	4E-8	1E-10	-	- 1
01	Destaction 200	M 227 pa	15.2	15.1	F.F. 0		AF . F	AF 4
21	FIGUACEINTUM-228	m, see ra	TE+2	Rone surf	26-3		26-2	26-4
				(2E+1)	-	3E-11	2.1.1.1	-
		¥ 226p		25.2	F.F. 6	01. A.L.		

.

			0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Co1. 2	Col. 3	Co'. 1	Col. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 1E-4 - 6E-8 - 2E-4 - - 2E-4 - - 2E-4
			Ingestion	Inhalation				Average
Atomic No.	Radionuclide	Class	ALĬ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
91	Protactinium-230	W, see <sup>227</sup> Pa	6E+2 Bone surf	5E+0	2E-9	7E-12	-	•
		영애에는 것으로 이야한다.	(9E+2)		1	-	15-5	15-4
		Y, see <sup>227</sup> Pa	-	4E+0	1E-9	5E-12	-	-
91	Protactinium-231	W, see <sup>227</sup> Pa	2E-1 Bone surf	2E-3 Bone surf	6E-13	-	-	-
			(5F-1)	(45-3)		6E-15	6E-9	65-9
		Y, see <sup>226</sup> Pa	(0)	4E-2 Bone surf	2E-12	-	-	-
				(6E-3)	-	8E-15	-	*
91	Protactinium-232	W, see <sup>227</sup> Pa	1E+3	2E+1 Bone surf	9E-9	-	2E-5	2E-4
		227	-	(6E+1)	-	8E-11	-	-
		Y, see <sup>227</sup> Pa		6E+1 Bone surf	2E-8	-	-	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 1E-4 - 6E-8 - 2E-4 - 2E-4 - 3E-4
			-	(7E+1)		1E-10	-1 12 N	* 11 d
91	Protactinium-233	W, see <sup>227</sup> Pa	1E+3 LLI wall	7E+2	3E-7	1E-9	-	-
		227	(2E+3)		-		2E-5	2F-4
		Y, see 22'Pa		6E+2	2E-7	8E-10	(#1))) (*	
91	Protactinium-234	W, see 227 227Pa	2E+3	8E+3	3E-6	1E-8	3E-5	3E-4
		r, see Pa		7E+3	3E-6	9E-9	-	-

			Table 1Table 1Occupational ValuesEffluenConcentrat				ole 2 luent trations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 8E-7 - - 6E-4 - - EE-7 - - 3E-6
			Col. 1 0:31	1 Col. 2 Col. 3 estion Inhalation		Col. 1	Col. 2	Monthly
Atomic	Dedisourlide	61	Ingestion					Average
No.		Class	(µCi)	(µCi)	UAL (μCi/ml)	Air (µCi/ml)	Water (µCi/ml)	(µCi/ml)
92	l.anium-230	D, $UF_6$ , $UO_2F_2$ , $UO_2(NO_3)_2$	4E+0 Bone surf	4E-1 Bone surf	2E-10	-	-	-
			(6E+0)	(6E-1)	-	8E-13	8E-8	8E-7
		W, UO <sub>3</sub> , UF <sub>4</sub> , UCl <sub>4</sub>	-	4E-1	1E-10	5E-13	-	
		Y, U0 <sub>2</sub> , U <sub>3</sub> 0 <sub>8</sub>	-	3E-1	1E-10	4E-13	-	-
92	Uranium-231	D, see <sup>230</sup> U	4E+3 LLI wall	8E+3	3E-6	1E-8	-	-
		000	(4E+3)	-	-		6E-5	6F-4
		W, see 230U	-	6E+3	2E-6	8F-9	-	-
		Y, see 2300		5E+3	2E-6	6E-9		
92	Uranium-232	D, see <sup>230</sup> U	2E+0 Bone surf	2E-1 Bone surf	98-11	-	-	-
		230	(4E+0)	(4E-1)	-	6E-13	6E-8	65-7
		W, see 230U	-	4E-1	2E-10	5E-13	-	-
		Y, see 2000		8E-3	3E-12	1E-14	-	-
92	Uranium-233	D, see <sup>230</sup> U	1E+1 Bone surf	1E+0 Bone surf	5E-10	-	-	-
		230	(2E+1)	(2E+0)		3E-12	3E-7	3F-6
		W, see 230U	-	7E-1	3E-10	1E-12	-	1.1
		Y, see 2000		4E-2	2E-11	5E-14	-	-
92	Uranium-234 <sup>3</sup>	D, see <sup>230</sup> U	1E+1 Bone surf	1E+0 Bone surf	5E-10	-	-	-
		230	(2E+1)	(2E+0)	-	3E-12	3E-7	3E-6
		W, see 230	-	7E-1	3E-10	1E-12	-	-
		Y, see	-	4E-2	28-11	5E-14		· •

			Table 1Table 2Occupational ValuesEffluentConcentrati			le 2 uent Pations	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-6 - - 3E-6 - - 3E-6 - - 3E-6 - -	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Co1. 2	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-6 - - 3E-6 - - 3E-6 - - 3E-6 - - 3E-6 - - - 3E-6 - - - 2E-4 - - 2E-4 -
			Ingestion	Inhala	tion			
Atomic No.	Radionur, de	Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	
92	Uranium-235 <sup>3</sup>	D, see <sup>230</sup> U	1E+1 Bone surf	1E+0 Bone surf	6E-10	-	*	-
			(2E+1)	(2E+0)	-	3E-12	3E-7	3E-6
		W, see 230		8E-1	3E-10	1E-12	-	-
		Y, see 230U		4E-2	2E-11	6E-14	-	-
92	Uranium-236	D, see <sup>230</sup> U	1E+1 Rope surf	1E+0 Bone surf	5E-10	-	-	-
			(2E+1)	(2E+0)		35-12	35-7	35-6
		W SPP 230	(22-1)	8F-1	35-10	15-12	-	50.0
		Y, see 2300	이 같은 것 같은 것 같이 없다.	4E-2	2E-11	6E-14	-	-
92	Uranium-237	D, see <sup>230</sup> U	2E+3 LLI wall	3E+3	1E-c	4E-9	-	
		230	(2E+3)	-	-	-	3E-5	3E-4
		W, see 230	-	2E+3	7E-7	2E-9	-	-
		Y, see U	-	2E+3	6E-7	2E-9	-	-
92	Uranium-238 <sup>3</sup>	D, see <sup>230</sup> U	1E+1 Bone surf	1E+0 Bone surf	6E-10	-	-	-
		230	(2E+1)	(2E+0)	-	3E-12	3E-7	Table 3 Releases to Sewers Monthly Average Concentration (µCi/ml) - 3E-6 - - 3E-6 - - 3E-6 - - 3E-6 - - - 3E-6 - - - 3E-6 - - - 2E-4 - - - 2E-4 - -
		W, see 230		8E-1	3E-10	1E-12	-	A
		Y, see 2000		4E-2	2E-11	6E-14	-	
92	Uranium-239 <sup>2</sup>	D, see 230	7E+4	2E+5	8E-5	3E-7	9F-4	QF-3
		W, see 230U		2E+5	7E-5	2E-7	-	-
		Y, see 2300		2E+5	6E-5	28-7	-	-
92	Uranium-240	D. see 230	15+3	45+3	25-6	55-0	25-5	25-4
		W. see 230	-	36+3	15-6	AE-0	20-5	25-4
		Y. see 230		25+3	15-6	35-0		
		· · · · · ·		22.3	10 0	36 3		-

				Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
		그는 아이에서 주말한 것이라.	Ingestion ALI (µCi)	Inhalation				Average
No.	Radionuclide	Class		ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
92	Uranium-natural <sup>3</sup>	D, see <sup>230</sup> U	1E+1 Bone surf	1E+0 Bone surf	5E-10	-	-	-
		220	(2E+1)	(2E+0)	-	3E-12	3E-7	3E-6
		W, see 230U		8E-1	3E-10	9E-13	-	-
		Y, see 230U		5E-2	2E-11	9E-14	-	-
93	Neptunium-232 <sup>2</sup>	W, all compounds	18+5	2E+3 Bone surf	7E-7	-	2E-3	2E-2
			전 같은 것 같아.	(5E+2)	-	6E-9	-	
93	Neptunium-233 <sup>2</sup>	W, all compounds	8E+5	3E+6	1E-3	4E-6	1E-2	18-1
93	Neptunium-234	W, all compounds	2E+3	3E+3	1E-6	4E-9	3E-5	3E-4
93	Neptunium-235	W, all compounds	2E+4 LLI wall	8E+2 Bone surf	3E-7	-	-	-
			(2E+4)	(1E+3)	-	2E-9	3E-4	3E-3
93	Neptunium-236 (1.15E+5 y)	W, all compounds	3E+0 Bone surf	2E-2 Bone surf	9E-12	-	-	-
			(6E+0)	(5E-2)	-	8E-14	9E-8	9E-7
93	Neptunium-236 (22.5 h)	W, all compounds	3E+3 Bone surf	3E+1 Rone surf	1E-8	-	-	-
			(4E+3)	(7E+1)	-	1E-10	5E-5	5E-4
93	Neptunium-237	W, all compounds	5E-1 Bone surf	4E-3 Bone surf	2E-12	-	-	-
			(1E+0)	(1E-2)	-	1E-14	2E-8	2F-7

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.	Radionuclide	Class	Ingestion ALI (µci)	Inhala ALI (µCi)	tion DAC (µCi/m1)	Air (µCi/ml)	Water (µCi/ml)	Average Concentration (µCi/m1)
93	Neptunium-238	W, all compounds	1E+3	6E+1 Bone surf	3E-8	-	2E-5	2E-4
			-	(2E+2)	-	2E-10	-	-
93	Neptunium-239	W, all compounds	2E+3 LLI wall	2E+3	9E-7	3E-9	-	-
			(2E+3)	-	-	-	2E-5	2E-4
93	Neptunium-240 <sup>2</sup>	W, all compounds	2E+4	8E+4	3E-5	1E-7	3E-4	3E-3
94	Plutonium-234	W, all compounds						
		except PuO <sub>2</sub> Y, PuO <sub>2</sub>	8E+3	2E+2 2E+2	9E-8 8E-8	3E-10 3E-10	1E-4 -	1E-3
94	Plutonium-235 <sup>2</sup>	W, see 234 Pu	9E+5	3E+6	1E-3	4E-6	1E-2	1E-1
		Y, see <sup>254</sup> Pu	-	3E+6	1E-3	3E-6	-	-
94	Plutonium-236	W, see <sup>234</sup> Pu	2E+0 Bone surf	2E-2 Bone surf	8E-12	-	-	-
		234-	(4E+0)	(4E-2)		5E-14	6E-8	6E-7
		Y, see Pu		4E-2	2E-11	6E-14	-	-
94	Plutonium-237	W, see <sup>234</sup> Pu Y, see <sup>234</sup> Pu	1E+4	3E+3 3E+3	1E-6	5E-9 4E-9	2E-4	2E-3

			0cc	Table 1 Occupation⊾i Values			ole 2 luent trations	Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Co1 2	Monthly
			Ingestion	Inhala	tion			Average
Atomic No.	Radioruclide	Class	ALĪ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air Water C (µCi/ml) (µCi/ml) (	Concentration (µCi/ml)	
94	Plutonium-238	W, see <sup>234</sup> Pu	9E-1 Bone surf	7E-3 Bone surf	3E-12	-	-	-
		224	(2E+0)	(1E-2)	-	2E-14	2F-8	2F-7
		Y, see <sup>234</sup> Pu		2E-2	8E-12	2E-14	-	-
94	Plutonium-239	W, see <sup>234</sup> Pu	8E-1 Bone surf	6E-3 Bone surf	3E-12	-	-	-
		234-	(1E+0)	(1E-2)	-	2E-14	2E-8	2E-7
		Y, see Pu		2E-2 Bone 5 rf	7E-12	-	-	-
			-	(2E-2)	-	2E-14	-	-
94	Plutonium-240	W, see <sup>234</sup> Pu	8E-1 Bone surf	6E-3 Bone surf	3E-12	-	-	2
		234	(1E+0)	(1E-2)	-	2E-14	2E-8	2E-7
		Y, see Pu	-	2E-2	7E-12	-	-	-
				Bone surf				
				(2E-2)	-	2E-14	-	-
94	Plutonium-241	W, see <sup>234</sup> Pu	4E+1 Bone surf	3E-1 Bone surf	1E-10	-	-	1 ( ) ( )
		234	(7E+1)	(6E-1)	-	8F-13	15-6	15-5
		Y, see <sup>234</sup> Pu		8E-1 Bone surf	3E-10	-	-	-
				(1E+0)		1E-12	-	1.1

	Radionuclide		0cc	Table 1 Occupational Values			ole 2 luent trations	Table 3 Releases to Sewers	
			Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly	
				Ingestion	Inhalation				Average
No.		Class	lass	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/ml)
94	Pluton:um-242	W, see	234 <sub>Pu</sub>	8E-1 Bone surf	7E-3 Bone surf	3E-12	-	-	-
			224	(1E+0)	(1E-2)	-	2E-14	2E-8	2E-7
		Y, see	<sup>234</sup> Pu		2E-2 Bone surf	7E-12	-	-	
				이 아이는 아이들이 같아?	(2E-2)	-	2E-14	-	-
94	Plutonium-243	W SPO	234 <sub>p11</sub>	2F+4	ec+4	25-5	55-0	25-4	25-2
	ridcorran 245	Y, see	234Pu	-	4E+4	2E-5	52-8	-	25-3
94	Plutonium-244	W, see	234 <sub>Pu</sub>	8E-1	7E-3	3E-12	-	-	_
				Bond surf	Bone surf				
		Y SPP	234 <sub>pu</sub>	(2E+0)	(1E-2) 2E-2	- 7E-12	2E-14	2E-8	2E-7
					Bone surf	11 12			
					(2E-2)	-	2E-14	-	-
94	Plutonium-245	W coo	234 <sub>p.</sub> ,	25+2	55.12	25.6	CE 0	AF . F	
	Fidconitan 245	Y SPP	234 Pu	2E+3	DE+3 AF+3	25-6	6E-9	3E-5	3E-4
		.,	224		46.3	20 0	02-3		
94	Plutonium-246	W, see	<sup>234</sup> Pu	4E+2	3E+2	1E-7	4E-10	-	-
				LLI wall					
		V	234 <sub>pu</sub>	(4E+2)	-	35.7	45.30	6E-6	65-5
		1, 200	ru		35+2	16-1	45-10		
95	Americium-237 <sup>2</sup>	W, all (	compounds	8E+4	3E+5	1E-4	4E-7	1E-3	1E-2
95	Americium 238 <sup>2</sup>	W, all d	compounds	4E+4	3E+3	1E-6	-	5F-4	5F-3
		1. 20. 162			Gone surf				52.5
					(6E+3)	-	9E-9	-	-

			Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
			Col. 1 Oral	Col. 2	Cci. 3	Col. 1	Col. 2	Monthly
			Ingestion	Inhala	tion	in .		Average
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (µCi)	(µCi/ml)	Air (μCi/m')	(µCi/ml)	(µCi/ml)
95	Americium-239	W, all compounds	5E+3	1E+4	5E-6	2E-8	7E-5	7E-4
95	Americium-240	W, all compounds	2E+3	3E+3	1E-6	4E-9	3E-5	3E-4
95	Americium-241	W, all compounds	8E-1	6E-3	3E-12	- 148		
			(1E+0)	(1E-2)		2E-14	2E-8	2E-7
95	Americium-242m	W, all compounds	8E-1	6E-3 Rong cunf	3E-12	동원이		-
			(1E+0)	(1E-2)	94 S.S.	2E-14	2E-8	2E-7
95	Americium-242	W, all compounds	4E+3	8E+1	4E-8		5E-5	5E-4
				(9E+1)	-	1E-10		
95	Americium-243	₩, all compounds	8E-1	.2-3 Rana cunf	3E-12	- 19		
			(1E+0)	(1E-2)	anaga	2E-14	2E-8	2E-7
95	Americium-244m <sup>2</sup>	W, all compounds	6E+4	4E+3	2E-6	6.5		-
			(8E+4)	(7E+3)		1E-8	1E-3	1E-2
95	Americium-244	W, all compounds	3E+3	21 · ?	8E-8		4E-5	4E-4
			이 감독 관심되는	(3E+2)	9 - S	4E-10	-	
95	Americium-245	W, all compounds	3E+4	SE+4	3E-5	18-7	4E-4	4E-3

1

-

•

\*\*

				Table 1 Occupational Values				Table 3 Releases to Sewers
	Radionuclide		Col. 1 Oral	Col. 2	°o1. 3	Col. 1 Col. 2	Monthly Average	
			Ingestion	Inhala	tion			
Atomic No.		Class	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/m1)
95	Americium-246m <sup>2</sup>	W, all compounds	5E+4 St. wa`l	2E+5	8E-5	3E-7	-	-
			(6E+4)		-	-	8E-4	8E-3
95	Americium-246 <sup>2</sup>	W, all compounds	3E+4	18+5	4E-5	1E-7	4E-4	4E-3
96	Curium-238	W, all compounds	2E+4	1E+3	5E-7	2E-9	2E-4	2E-3
96	Curium-240	W, all compounds	6E+1 Bone surf	6E-1 Bone surf	2E-10	-	-	-
			(8E+1)	(6E-1)	-	9E-13	1E-6	1E-5
96	Curium-241	W, all compounds	1E+3	3E+1 Bone surf	1E-8	-	2E-5	2E-4
				(42+1)	-	5E-11	-	-
96	Curium 242	W, all compounds	3E+1 Bone surf	3E-1 Bone surf	18-10	-	-	-
			(5E+1)	(3E-1)	-	4E-13	7E-7	7E-6
96	Curium-243	W, all compounds	LE+0	9E-3 Bone surf	4E-12	-	-	-
			(2E+0)	(2E-2)	-	2E-14	3E-8	3E-7
96	Curium-244	W, all compounds	1E40 Bone surf	1E-2 Bone surf	5E-12	-	-	-
			(3E+0)	(2E-2)	-	3E-14	3E-8	3E-7
96	Curium-245	W, alı compounds	7E-1	6E-3	3E-12	-	-	-
			(1E+0)	(1E-2)	-	2E-14	2E-8	2E-7

				Table 1 Occupational Values				Table 3 Releases to Sewers
				Col. 2	Col. 3	Col. 1 Col. 2		Monthly
Atomic No.	Radionuclide	Class	ALI (µCi)	ALI (uCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratic (µCi/ml)
96	Curium-246	₩, all compounds	7E-1 Bone surf	6E-3 Bone surf	3E-12	-	-	-
			(1E+0)	(1E-2)	-	2E-14	2E-8	2E-7
96	Curium-247	W, all compounds	8E-1 Bone surf	6E-3 Bone surf	3E-12	-	-	-
			(1E+0)	(1E-2)	-	2E-14	2E-8	2E-7
96	Curium-248	₩, all compounds	2E-1 Bone surf	2E-3 Bone surf	7E-13	7	-	-
			(4E-1)	(3E-3)	-	4E-15	5E-9	5E-8
96	Curium-249 <sup>2</sup>	W, all compounds	5E+4	2E+4 Bone surf	7E-6	-	7E-4	7E-3
			-	(3E+4)	-	4E-8	-	-
96	Curium-250	W, all compounds	4E-2 Bone surf	3E-4 Bone surf	1E-13	-	-	-
			(6E-2)	(5E-4)	-	8E-16	9E-10	9E-9
97	Berkelium-245	₩, all compounds	2E+3	1E+3	5E-7	2ē-9	3E-5	3E-4
97	Berkelium-246	W, all compounds	3E+3	3E+3	1E-6	4. 9	4E- "	4E-4
97	Berkelium-247	W, all compounds	5E-1 Bone surf	4E-3 Bone surf	2E-12	-	-	-
			(1E+0)	(9E-3)	-	1E-14	2E-8	2E-7
97	Berkelium-249	W, all compounds	2E+2 Bone surf	2E+0 Bone surf	7E-10	-	-	-
			(5E+2)	(4E+0)	-	5E-12	6E-6	6E-5

			0cc	Table 1 cupational Va	lues	Tab Effi Concent	ole 2 luent craticns	Table 3 Releases to Sewers
		onuclide Class	Col. 1 Ural	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
Atomic No.			Ingestion	Inhala	tion			Average
	Radionuclide		ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentration (µCi/m1)
97	Berkelium-250	W, all compounds	9E+3	3E+2 Bone surf	1E-7	-	1E-4	1E-3
			1 - 21.	(7E+2)	-	1E-9	-	4
98	Californium-244 <sup>2</sup>	W. all compounds except						
	역가 경험했는	those given for Y	3E+4 St. wall	6E+2	2E-7	8E-10	-	-
			(3E+4)	-	-	-	48-4	4E-3
		<sup>v</sup> , oxides and hydroxides	-	6E+2	2E-7	8E-10	-	
98	Californium-246	W. see 244	4F+2	9F+0	4F-9	1E-11	55-6	55-5
		Y, see <sup>244</sup> Cf	-	9E+0	4E-9	1E-11	-	-
98	Californium-248	W, see <sup>244</sup> Cf	8E+0 Bone surf	6E-2 Bone surf	3E-11	-	-	-
		244	(2E+1)	(1E-1)	-	2E-13	2E-7	2E-6
		Y, see ~ Cf	-	1E-1	4E-11	1E-13	-	-
98	Californium-249	W, see <sup>244</sup> Cf	5E-1 Bone surf	4E-3 Bone surf	2E-12	-	-	-
		244	(1E+0)	(9E-3)	-	1E-14	2E-8	2E-7
		Y, see <sup>244</sup> Cf	-	1E-2 Bone surf	4E-12	-	-	
			-	(1E-2)	-	2E-14	-	
98	Californium-250	W, see <sup>244</sup> Cf	1E+0 Bone surf	9E-3 Bone surf	4E~ 42	-	-	-
		244	(2E+0)	(2E-2)	-	3E-14	3E-8	3E-7
		Y, see Cf Cf	-	3E-2	1E-11	4E-14	-	

			Table 1 Occupational Values			Tat Effi Concert	ole 2 luent trations	Table 3 Releases to Sewers
	Radionuclide		Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly
			Ingestion	Inhala	tion			Average
Atomic No.		Class	ALĪ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
98	Californium-251	W, see <sup>244</sup> Cf	5E-1 Bone surf	4E-3 Bone surf	2E-12	-	-	-
		244	(1E+0)	(9E-3)	41.000	1F-14	2F-8	25-7
		Y, see <sup>244</sup> (		1E-2 Bone surf	4E-12		-	-
			-	(1E-2)		2E-14	- 10 C	-
98	Californium-252	W, see <sup>244</sup> Cf	2E+0 Bone surf	2E-2 Bone surf	8E-12	-	-	-
			(5E+0)	(4F-2)	12.00	55-14	75-0	75.7
		Y, see <sup>244</sup> Cf	-	3E-2	18-11	5E-14	76-0	-
98	Californium-253	W, see <sup>244</sup> Cf	2E+2 Bone surf	2E+0	8F-10	3E-12	-	-
		244	(4E+2)			-	5E-6	5E-5
		Y, see Cf		2E+0	7E-10	2E-12	-	- 11. 12. 13.
98	Californium-254	W, see <sup>244</sup> Cf Y, see <sup>244</sup> Cf	2E+0 -	2E-2 2E-2	9E-12 7E-12	3E-14 2E-14	3E-8	3E-7
99	Einsteinium-250	W, all compounds	4E+4	5E+2 Bone surf	2E-7		6E-4	6E-3
				(1E+3)		2E-9	12.00	그는 그는 그는 것
99	Einsteinium-251	W, all compounds	7E+3	9E+2 Bone surf	4E-7	-	1E-4	1E-3
				(1E+3)	-	2E-9	-	
99	Einsteinium-253	W, all compounds	2E+2	1E+0	6E-10	2E-12	2E-6	2E-5

			000	Table 1 Occupational Values			Table 2 Effluent Concentrations		
	Radionuclide			Col. 1 Oral Ingestion	Col. 2 Inhala	Col. 3	Col. 1 Col. 2		Monthly Average
Atomic No.		Class	5	ALĬ (µCi)	ALĪ (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (µCi/ml)	Concentratio (µCi/ml)
99	Einsteinium-254m	W, all (	compounds	3E+2 LLI wall	1E+1	4E-9	1E-11	-	-
				(3E+2)	-	-	-	4E-6	4E-5
99	Einsteinium-254	W, all (	compounds	8E+0 Bone surf	7E-2 Bone surf	3E-11	-	-	-
				(2E+1)	(1E-1)	-	2E-13	2E-7	2E-6
100	Fermium-252	W, all (	compounds	5E+2	1E+1	5E-9	2E-11	6E-6	6E-5
100	Fermium-253	W, all d	compounds	1E+3	1E+1	4E-9	1E-11	1E-5	1E-4
100	Fermiur 4	W, all c	compounds	3E+3	9E+1	4E-8	1E-10	4E-5	4E-4
100	Fermium-205	∀, all c	compounds	5E+2	2E+1	9E~9	3E-11	7E-6	7E-5
100	Fermium-257	W, 311 c	compounds	2E+1 Bong surf	2E-1 Bone surf	7E-11	-	-	-
				(4E+1)	(2E-1)	-	3E-13	5E-7	5E-6
101	Mendelevium-257	₩, all c	compounds	7E+3	8E+1 Bone surf	4E-8	-	1E-4	1E-3
					(9E+1)	-	1E-10	-	-
101	Mendelevium-258	W, all c	compounds	37+1 Bone surf	2E-1 Bone surf	1E-10	-	-	-
				(5E+1)	(3E-1)	-	5E-13	6E-7	6E-6

.

		Table 1 Occupational Values				ole 2 luent trations	Table 3 Releases to Sewers
		Cel. 1 Oral Ingestion	Col. 2 Col. 3		Col. 1	Col. 2	Monthly
No.	Radionuclide Class	ALĪ (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi/ml)	Water (, Ci/ml)	Concentration (µCi/ml)
-	Any single radionuclide not listed above with decay mode other than alpha emission or spontaneous fis- sion and with radioactive half- life less than 2 hours Submersic. <sup>1</sup>	-	2E+2	1E-7	1E-9	-	
-	Any single radionuclide not listed above with decay mode other than alpha emission or spontaneous fis- sion and with radioactive half- life greater than 2 hours	-	2E-1	1E-10	1E-12	1E-8	1F-7
-	Any single radionuclide not listed above that decays by alpha emission or spontaneous fission, or any min- ture for which either the identity or the concentration of any radio- nuclide in the mixture is not						
	known	-	4E-4	2E-13	1E-15	2E-9	2E-8

## FOOTNOTES:

<sup>1</sup>"Submersion" means that values given are for submersion in a hemispherical semi-infinite cloud of airborne material.

<sup>2</sup>These radionuclides have radiological half-lives of less than 2 hours. The total effective dose equivalent received during operations with these radionuclides might include a significant contribution from external exposure. The DAC values for all radionuclides, other than those designated Class "Submersion," are based upon the committed effective dose equivalent doe to the intake of the radionuclide into the body and do NOT include potentially significant contributions to equivalent from external exposures. The licensee may substitute 1E-7 µCi/ml for the listed DAC to account the submersion dose prospectively, but should use individual monitoring devices or other radiation measuring instruments that measure external exposure to demonstrate compliance with the limits. (See § 20.203.)

<sup>3</sup>For soluble mixtures of U-238, U-234, and U-235 in air, chemical toxicity may be the limiting factor (see § 20.201(e)). If the percent by weight (enrichment) of U-235 is not greater than 5, the concentration value for a 40-hour workweek is 0.2 milligrams uranium per cubic meter of air average. In y enrichment, the product of the average concentration and time of exposure during a 40-hour workweek shal exceed 8E-3 (SA) µCi-hr/ml, where SA is the specific activity of the uranium inhaled. The specific active or natural uranium is 6.77E-7 curies per gram U. The specific activity for other mixtures of U-238, U-235, and U-234, if not known, shall be:

SA = 3.6E-7 curies/gram U U-depleted

 $SA = [0.4 + 0.38 \text{ (enrichment)} + 0.0034 \text{ (enrichment)}^2] E-6$ , enrichment > 0.72

where enrichment is the percentage by weight of U-235, expressed as percent.
#### NOTE:

 If the identity of each radionuclide in a mixture is known but the concentration of one or more of the radionuclides in the mixture is not known, the DAC for the mixture shall be the most restrictive DAC of any radionuclide in the mixture.

2. If the identity of each radionuclide in the mixture is not known, but it is known that certain radionuclides specified in this appendix are not present in the mixture, the inhalation ALI, DAC, and effluent and sewage concentrations for the mixture are the lowest values specified in this appendix for any radionuclide that is not known to be absent from the mixture; or

	Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
Radionuclide	Col. 1 Oral	Col. 2	Col. 3	Col. 1	Cc1. 2	Monthly
	Ingestion ALI (µCi)	n <u>Inhalation</u> ALI DAC (µCi) (µCi/ml)		Air Water (µCi/m1) (µCi/m1)		Average Concentration (µCi/ml)
If it is known that Ac-227-D and Cm-250-W are not present		7E-4	3E-13			-
If, in addition, it is known that Ac-227-W,Y, Th-229-W,Y, Th-230-W, Th-232-W,Y, Pa-231-W,Y, Np-237-W, Pu-239-W, Pu-240-W, Pu-242-W, Am-241-W, Am-242m-W, Am-243-W, Cm-245-W, Cm-246-W, Cm-247-W, Cm-248-W, 8k-247-W, Cf-249-W, and Cf-251-W are not present		7E-3	3E-12			
If, in addition, it is known that Sm-146-W, Sm-147-W, Gd-148-D,W, Gd-152-D,W, Th-228-W,Y, Th-230-Y, U-232-Y, U-233-Y, U-234-Y, U-235-Y, U-236-Y, U-238-Y, Np-236-W, Pu-236-W,Y. Pu-238-W,Y, Pu-239-Y, Pu-240-Y, Pu-242-Y, Pu-244-W,Y, Cm-243-W, Cm-244-W, Cf-248-W, Cf-249-Y, Cf-250-W,Y, Cf-251-Y, Cf-252-W,Y, and Cf-254-W,Y are not present		7E-2	3E-11			
If, in addition, it is known that Pb-210-D, Bi-210m-W, Po-210-D,W, Ra-223-W, Ra-225-W, Ra-226-W, Ac-225-D,W,Y, Th-227-W,Y, U-230-D,W,Y, U-232-D,W, Pu-241-W, Cm-240-W, Cm-242-W, Cf-248-Y, Es-254-W, Fm-257-W, and Md-258-W are not present		75-1	3E-10			
			OL 10			

	Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
	Col. 1 Oral	Col. 2	Col. 3	Col. 1	Col. 2	Monthly Average
Radionuclide	ALI (µCi)	ALI (µCi)	DAC (µCi/ml)	Air (µCi∕mì)	Water (µCi/ml)	Concentration (µCi/ml)
If, in addition, it is known that Si-32-Y, Ti-44-Y, Fe-60-D, Sr-90-Y, Zr-93-D, Cd-113m-D, Cd-113-D, In-115-D,W, La-138-D, Lu-176-W, Hf-178m-D,W, Hf-182-D,W, Bi-210m-D, Ra-224-W, Ra-228-W, Ac-226-D,W,Y, Pa-230-W,Y, U-233-D,W, U-234-D,W, U-235-D,W, U-236-D,W, U-238-D,W, Pu-241-Y, Bk-249-W, Cf-253-W,Y, and Es-253-W are not present	-	7E+0	3E-9		-	
If it is known that Ac-227-D,W,Y, Th-229-W,Y, Th-232-W,Y, Pa-231-W.Y, Cm-248-W, and Cm-250-W are not present -	-	-		1E-14	-	-
If, in addition, it is known that Sm-146-W, Gd-148-D,W, Gd-152-D, Th-228-W,Y, Th-230-W,Y, U-232-Y, U-233-Y, U-234-Y, U-235-Y, U-236-Y, U-238-Y, U-Nat-Y, Np-236-W, Np-237-W, Pu-236-W,Y, Pu-238-W,Y, Pu-239-W,Y, Pu-240-W,Y, Pu-242-W,Y, Pu-244-W,Y, Am-241-W, Am-242m-W, Am-243-W, Cm-243-W. Cm-244-W, Cm-245-W, Am-243-W, Cm-243-W. Cm-244-W, Cm-245-W, Cm-246-W, Cm-247-W, Bk: 47-W, Cf-249-W,Y, Cf-250-W,Y, Cf-251-W,Y, Cf-252-W,Y, and Cf-254-W,Y 're not present	-		-	1E-13		
in addition, it is known that Sm-147-W, -152-W, Pb-210-D, Bi-210m-W, Po-210-D,W, -223-W, Ra-225-W, Ra-226-W, Ac-225-D,W,Y, -227-W,Y, U-230-D,W,Y, U-232-D,W, U-Nat-W, u-241-W, Cm-240-W, Cm-242-W, Cf-248-W,Y, Es-254-W, Fm-257-W, and Md-258-W are not						
present	1	-		1E-12	- 11 V.	

	Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Releases to Sewers
Radionuclida	C.1. 1 Oral Ingestion ALI (uCi)	Col. 2 Inhal ALI (uCi)	Col. 3 ation DAC (uCi/ml)	Col. 1 Air (uCi/ml)	Col. 2 Water (uCi/ml)	Monthly Average Concentration (uCi/ml)
If, in addition it is known that Fe-60, Sr-90, Cd-113m, Cd-113, in-115, I-129, Cs-134, Sm-145, Sm-147, Gd-148, Gd-152, Hg-194 (organic), Bi-210m, Ra-223, Ra-224, Ra-225, Ac-225, Th-228, Th-230, U-233, U-234, U-235, U-236, U-238, U-Nat, Cm-242, Cf-248, Es-254, Fm-257, and Md-258 are not present			-	-	1E-6	1E-5

If a mixture of radionuclides consists of uranium and its daughters in one dust (30 µm AMAD particle distribution assumed) prior to chemical separation of the uranium from the one, the following values may be used for the DAC of the mixture: 6E-11 µCi of gross alpha activity from uranium-238, uranium-234, thorium-230, and radium-226 per milliliter of air; 3E-11 µCi of natural uranium per milliliter of air; or 45 micrograms of natural uranium per cubic meter of air.

3.

4. If the identity and concentration of each radionuclide in a mixture are known, the limiting values should be derived as follows: determine, for each radionuclide in the mixture, the ratio between the concentration present in the mixture and the concentration otherwise established in Appendix B for the specific radionuclide when not in a mixture. The sum of such ratios for all of the radionuclides in the mixture may not exceed "1" (i.e., "unity").

Example: If radionuclides "A," "B," and "C" are present in concentrations  $C_A$ ,  $C_B$ , and  $C_C$ , and if the applic applic backs are DAC<sub>A</sub>, DAC<sub>B</sub>, and DAC<sub>C</sub>, respectively, then the concentrations shall be limited so that the following relationship exists:

$$\frac{C_A}{DAC_A} + \frac{C_B}{DAC_B} + \frac{C_C}{DAC_C} \leq 1$$

## APPENDIX C

# QUANTITIES\* OF LICENSED MATERIAL REQUIRING LABELING

Radionuclide	Quantity (µCi)	Radionuclide	Quantity (µCi)
Hydrogen-3	1,000	Chromium-48	1.000
Beryllium-7	1,000	Chromium 49	1,000
Beryllium-10	1	Chromium-51	1,000
Carbon-11	1,000	Manganese=51	1,000
Carbon-14	1,000	Manganese=52m	1,000
Fluorine-18	1,000	Manganese 52m	1,000
Sodium-22	10	Manganese 52	1 000
Sodium-24	100	Mangaporo-E4	1,000
Magnes 1um=28	100	Manganese-54	100
Aluminum-26	10	Manganese-bb	1,000
Silicon=31	1 000	100-52	100
Silicon=32	1,000	1ron-55	100
Phosphorus=32	10	Iron-59	10
Phoenhonue=33	100	Iron-60	1
Sulfur= 35	100	Cobalt-55	100
Chloning-26	100	Cobalt-56	10
Chloning=20	1 000	Cobalt-57	100
Chlonine-38	,000	Cobalt-58m	1,000
Angene 20	1,000	Cobalt-58	1.00
Argon- 39	1,000	Cobalt-60m	1,000
Argon=41	1,00(	Cobalt-60	1
Potassium-40	100	Cobalt-61	1,000
Potassium-42	1,000	Cobalt-62m	2,000
Potassium-43	1,000	Nickel-56	100
Polassium-44	1,000	Nickel-57	100
Potassium=45	1,000	Nickel-59	100
Calcium-41	100	Nicke <sup>1</sup> =63	100
Calcium-45	100	Nickel-65	1,000
Calcium-47	100	Nickel-66	10
Scandium-43	1,000	Copper-60	000
Scandium-44m	100	Copper-61	1,000
Scandium-44	100	Copper-64	1,000
Scandium-46	10	Copper=67	1,000
Scandium-47	100	7inc-62	1,000
Scandium-48	100	Zinc-63	1 000
Scandium-49	1.000	7inc-65	1,000
Titanium-44	1	7inc=60m	10
Titanium-45	1.000	7100-69	100
Vanadium-47	1,000	7100-71-	1,000
Vanadium-48	100	21nc=/1m	1,000
Vanadium-40	1 000	21nc-72	100
variau i uni 49	1,000	Gallium-65	1.000

n,

ø

Radionuclide	Quantity (µCi)	Radionuclide	Quantity (µCi)
Gallium-66	100	Krypton-85	1.000
Gallium+67	1.000	Krypton-87	1,000
Gallium-68	1,000	Krypton-88	1,000
Gallium-70	1,000	Rubidium-79	1,000
Callium-72	100	Rubidium-81m	1,000
Gallium-73	1.000	Rubidium-81	1 000
Germanium-66	1,000	Rubidium-82m	1,000
Germanium-67	1,000	Rubidium-83	100
Germanium-68	10	Rubidium-84	100
Germanium-69	1.000	Rubidium-86	100
Germanium-71	1,000	Rubidium-87	100
Germanium-75	1,000	Rubidium-88	1 000
Germanium-77	1,000	Rubidium-89	1,000
Germanium-78	1,000	Strontium-80	100
Arsenic=69	1,000	Strontium-81	1 000
Arsenic-70	1,000	Strontium-83	100
Arsenic-71	100	Strontium-85m	1 000
Arsenic-72	100	Strontium-85	100
Arsenic=73	100	Strontium-87m	1 000
Arsenic-74	100	Strontium-89	10
Arsenic-76	100	Strontium-90	10
Arsenic-77	100	Strontium-91	100
Arsenic-78	1.000	Strontium-92	100
Selenium-76	1.000	Yttrium-86m	1 000
Selenium-73m	1.000	Yttrium-86	100
Selenium-73	100	Yttrium-87	100
Selenium-75	100	Yttrium-88	10
Sclenium-79	100	Yttrium-SOm	1 000
Seleni.m-81m	1.000	Yttrium-90	1,000
Seleniun-81	1.000	Yttrium-91m	1 000
Selenium-83	1.000	Yttrium-91	2,000
Bromine-74m	1,000	Yttrium-92	100
Bromine-74	1,000	Yttrium-93	100
Bromine-75	1.000	Yttrium-94	1 000
Bromine-76	100	Yttrium-95	1,000
Bromine-77	1,000	7irconium-86	100
Bromine-80m	1,000	Zirconium-88	10
Bromine-80	1,000	Zirconium-89	100
Bromine-82	100	Zirconium-93	100
Bromine-83	1.000	Zirconium-95	10
Bromine-34	1,000	irconium-97	100
Krypton-74	1.000	Niobium-88	1 000
Krypton-76	1,000	Niobium-89	1,000
Krypton-77	1,000	(66 min)	1 000
Krypton-79	1,000	Niobium-89	1,000
Krypton-81	1,000	(122 min)	1 000
Krypton-83m	1,000	Niobium-90	100
Krypton-85m	1,000	Nichium-93m	100

Radionuclide	Quantity (µCi)	Radionuclide	Quantity (µCi)
Niobium-94	1	Silver-104	1.000
Niobium-95m	100	Silver-105	100
Niobium-85	100	Silver-106m	100
Niobium-96	100	Silver-106	1.000
Niobium-97	1,000	Silver-108m	1
Niobium-98	1,000	Silver-110m	10
Molybdenum-90	100	Silver-111	100
Molybdenum-93m	100	Silver-112	100
Molybdenum-93	10	Silver-115	1 000
Molybdenum-99	100	Cadmium-104	1,000
Molybdenum-101	1,000	Cadmium-107	1,000
Technetium-93m	1.000	Cadmium-109	1,000
Technetium-93	1,000	Cadmium-113m	0.1
Technetium-94m	1,000	Cadmium-113	100
Technetium-94	1.000	Cadmium-115m	10
Technetium-96m	1,000	Cadmium-115	100
Technetium-96	100	Cadmium-117m	1 000
Technetium-97m	100	Cadmium-117	1,000
Technetium-97	1 000	Indium-100	1,000
Technetium-98	10	Indium-110m	1,000
Technetium-99m	1 000	(60 1m)	1 000
Technotium-99	100	(og.im) Indium-110m	1,000
Technetium-101	1 000		1 000
Technetium-104	1,000	(4.91) Todium-717	1,000
Ruthenium-94	1,000	Indium-111	100
Ruthenium-97	1,000	Indium-112	1,000
Rut. nium-103	100	Indium-113m	1,000
Ruthenium-105	1 000	Indium-114m	10
Ruthenium-106	1,000	Indium-115m	1,000
Rhodium-99m	1 000	Indium-115	100
Rhodium-99	100	Indium-116m	1,000
Rhodium-100	100	Indium-11/m	1,000
Rhodium-101m	1 000	Indium-11/	1,000
Rhodium-101	1,000	10010m-119m	1,000
Rhouium-102m	10	110-110	100
Rhodium~102	10	110~111	1,000
Rhodium-103m	1 000	110-113	100
Rhodium-105	1,000	110-11/m	100
Rhodium-106m	1 000	11n-119m	100
Rhodium-107	1,000	11n=121m	100
Palladium-100	1,000	110-121	1 000
Pulladium-101	1 000	1)n=123m	1,000
Palladium=102	1,000	110-123	10
Palladium-107	100	110-125	10
Palladium-100	100	110-126	10
Silver-102	100	110-127	1,000
Silver-102	1,000	lin-128	1,000
Silver-103	1,000	Antimony-115	1,000
511Ver-104m	1,000	Artimony=116m	1,000

.

---

1

-

Radionuclide	Quantity ( $\mu$ Ci)	Radionuclide	Quantity (µCi)
Antimony-116	1,000	Iodine-131	1
Antimony-117	1,000	Iodine-132m	100
Antimony-118m	1,000	Iodine-132	100
Antimony-119	1,000	Iodine-133	10
Antimony-120		Iodine-134	1.000
(16m)	1,000	Iodine-135	100
Antimony-120		Xenon-120	1,000
(5.76d)	100	Xenon-121	1,000
Antimony-122	100	Xenon-1:	1,000
Antimony-124m	1,000	Xenon-123	1,000
Antimony-124	10	Xenon=125	1,000
Antimony-125	100	Xenon=127	1,000
Antimony-126m	1.000	Xenon=129m	1,000
Antimony-126	100	Xenop-131m	1,000
Antimony-127	100	Xenon-133m	1,000
Antimony-128		Xenon=133	1,000
(10.4m)	1.000	Xenon=135m	1,000
Antimony-128	1,000	Xenon=125	1,000
(9.01h)	100	Venone 139	1,000
Antimony-129	100	Contine 125	1,000
Antimony-130	1 000	Cestum-125	1,000
Antimony-131	1,000	Cestum-127	1,000
Tellurium-116	1,000	Cestum-129	1,000
Tellurium-121m	10	Cestum-130	1,000
Tellurium-121	100	Cestum-131	1,000
Tellurium-123m	100	Cestum-132	100
Tellurium-123	100	Cesium-134m	1,000
Tellurium-125m	100	Lesium-134	10
Tellurium-127m	10	Cesium-135m	1,000
Tellurium-127	1 600	Cesium-135	100
Tellurium=120m	1,000	Cesium-136	10
Tellurium-120	1 000	Cesium-137	10
Tellunium-131m	1,000	Cesium-138	1,000
Tollumium-121	10	Barium-126	1,000
Tellunium-122	100	Barium-128	100
Tellumium-132	10	Barium-131m	1,000
Tellumium=133m	100	Barium-131	100
Tellumium-133	1,000	Barium-133m	100
Teriurium~134	1,000	Barium-133	100
Iodine-120m	1,000	Barium-135m	100
Iodine-120	100	Barium-139	1,000
locine~121	1,000	Barium-140	100
1001ne-123	100	Barium-141	1,000
1001ne=124	10	Barium-142	1,000
lodine-125	1	Lanthanum-131	1,000
Iodine-126	1	Lanthanum-132	100
lodine-128	1,000	Lanthanum-135	1,000
lodine-129	1	Lanthanum-137	10
lodine-130	10	Lanthanum-138	100

inst and

.

Radionuclide	Quantity (µCi)	Radionuclide	Quantity (µCi)
Lanthanum-140	100	Samarium-153	100
Lanthanum-141	100	Samarium-155	1.000
Lanthanum-142	1,000	Samarium-156	1,000
Lanthanum-143	1,000	Europium-145	100
Cerium-134	100	Europium 146	100
Cerium-135	100	Europium-147	100
Cerium-137m	100	Europium-148	10
Cerium-137	1,000	Europium-149	100
Cerium-139	100	Europium=150	100
Cerium-141	100	(12,62b)	100
Cerium-143	100	Europium=150	100
Cerium-144	100	(34 20)	
Praseodymium-136	1 000	Europium-152m	100
Praseodymium-137	1,000	Europium-152	100
Praseodymium-138m	1,000	Europium-152	1
Praseodymium-139	1,000	Europium-154	10
Praseodymium-142m	1,000	Europium-155	10
Praseodymium-142	1,000	Europium-156	100
Praseodymium-142	100	Europium-15/	100
Praseodymium-143	1 000	Europium-158	1,000
Pracoodymium 144	1,000	Gadolinium-145	1,000
Pracoodumium-140	100	Gadolinium-146	10
Noodumium-126	1,000	Gadolinium-147	100
Neodymium-130	1,000	Gadolinium-148	0.001
Neodymium-138	100	Gadolinium-149	100
Neodymium-139m	1,000	Gadolinium-151	10
Neodym1um-139	1,000	Gadolinium-152	100
Neodym1um-141	1,000	Gadolinium-153	10
Neodym1um-147	100	Gadolinium-159	100
Neodym1um-149	1,000	Terbium-147	000
Neodym1um-151	1,000	Terbium-149	100
Promethium-141	1,000	Terbium-150	1,000
Promethium-143	100	Terbium-151	100
Prometnium-144	10	Terbium-153	1,000
Promethium-145	10	Terbium-154	100
Promethium-146	1	Terbium-155	1,000
Promethium-147	10	Terbium-156m	
Promethium-148m	10	(5.0h)	1,000
Promethium-148	10	Terbium-156m	
Promethium-149	100	(24.4h)	1,000
Promethium-150	1,000	Terbium-156	100
Promethium-151	100	Terbium-157	10
Samarium-141m	1,000	Terbium-158	1
Samarium-141	1,000	Terbium-160	10
Samarium-142	1,000	Terbium-161	100
Samarium-145	100	Dysprosium-155	1.000
Samarium-146	1	Dysprosium-157	1.000
Samarium-147	100	Dysprosium-159	100
Samarium-151	10	Dysprosium-165	1,000

# (beunitno)) ) XIONE99A

 and the second se	A REAL PROPERTY OF A REAL PROPER	the second s	En balletik fan fil yn en en haar die Kreen die Kreek kerden af die kerden in die kerden in die die personen.
00G'T	£71-muinî6H	100	0ysprosium-166
100	STI-muinteH	000'T	SSI-muimioH
1,000	mTTI-ruinteH	J'000	721-muimfoH
0.1	m871-muinteH	J'000	est-muimfoH
JO	meTi-muinteH	000'T	191-muimioH
J'000	mOSI-muinteH	000'T	mSði-muimfoH
JO	181-muinteH	1'000	291-Wniwi-162
000'T	mS&I-muinteH	000'τ	MAAL-MUTM,OH
T.0	S81-muinteH	000'T	\$9T-WHIWLOH
000'T	681-muinteH	τ	W99T-WNLWLOH
OOT	481-muinteH	100	99T-WniwioH
000'T	STI-muletneT	000'T	L9T-WDIWIOH
000'T	ETI-mulstnsT	000'T	Erbium-161
J'000	ATI-muletneT	000'T	591-WDIQUA
J'000	271-mufetnsT	100	69T-Wnigua
100	Jantalum 176	100	T/T-Wnigua
J'000	TTanteiner	100	Erbium-172
J'000	871-mulsinsT	000°T	[µn]ium-162
100	eri-muleinel	JOOT	99T-Wniinui
1'000	mOS1-mufstns1	100	L9T-Wniinui
00T	081-mu[stns]	OT	0/T-Wniinyi
J'000	mS&I-mufetneT	JO	T/T-Wniinyi
JO	S81-muletneT	JOOT	Thulium-172
OOL	581-muleineT	JOOT	ELT-MUIINAT
OOT	481-mulsinsT	000°T	ST-muiluit
1,000	28[-mu⊺stnsſ	000'T	Ytterbium-162
000'T	7arta1um_186	JOO	7fferbium-166
000'T	371-nstapnuT	000'T	752-muidressy
000'T	771-nstepul	DOT	Ytterbium-169
000°T	871-natepnuT	JOOT	Ytterbium-175
000'T	671-nstepul	J '000	Ytt-muidressy
1,000	181-nstepul	000°T	871-muidressy
00T	281-nacepunT	OOT	69T-Wnileiny
TOOT	781-nstepul	DOT	0/I-muijejuj
OT	881-nstepul	JOOT	TLT-Whileing
J'000	771-muinada	DOT	7/T-WNIJaanj
000'T	871-muinadя	JO	ELT-muijejuj
000'T	181-muinada	50	Wb/T-Wnileirj
	S81-muinada	OT	t/T-wnijeinj
1'000	(42.21)	000'T	W9/T-Wnijəini
	S81-muinada	100	9/T-Wnileinj
300	(40.43)	OT	W//T-Wnileann
OT	Rhenium-184m	TCO	//T-wniienn
100	Rhenium-184	000'T	WR/T_Wnijeani
OT	Rhenium-186m	000'T	8/T-Wnileann
J00	81 -muriner	000'T	6/T_Wnijajni
000 L	781-murnany	TOO	O/T_BDIUIPU
non'r			CCC - milita a ch

Radionuclide	Quantity (µCi)	Radionuclide	Quantity (µCi)
Rhenium-188	100	Mercury-194	1
Rhenium-189	100	Mercury-195m	100
Osmium-180	1,000	Mercury-195	1.000
Osmium-181	1,000	Mercury-197m	100
Osmium-182	100	Mercury-197	1.000
Osmium-185	100	Mercury-199%	1,000
Osmium-189m	1,000	Mercury-203	100
Osmium-191m	1,000	Thallium-194m	1,000
Osmium-191	100	Thallium-194	1,000
Osmium-193	100	Thallium-195	1,000
Osmium-194		Thallium-197	1,000
Iridium-182	1.000	Thallium-198m	1,000
Iridium-184	1,000	Thallium-198	1,000
Iridium-185	1,000	Thallium-193	1,000
Iridium-186	100	Thalli m= 200	1,000
Irigium-187	1 000	Thallium-200	1,000
Iridium-188	100	Thallium-202	1,000
Iridium-18S	100	Thallium-202	100
Iridium-190m	1 000	Load-105m	100
Iridium-190	100	Lead-195m	1,000
Iridium-192m	100	Lead-198	1,000
Iridium-192	10	Lead-199	1,000
Iridium-194m	10	Lead-200	100
Iridium-194	100	Lead-201	1,000
Iridium-195m	1 000	Lead-202m	1,000
Iridium-195	1,000	Lead-202	10
Platinum=186	1,000	Lead-203	1,000
Platinum-188	100	Lead-205	100
Platinum-189	1 000	Lead-209	1,000
Platinum-191	1,000	Lead-210	0.01
Platinum-193m	100	Lead-211	100
Platinum~193	1 000	Lead-212	1
Platinum=195m	1,000	Lead-214	100
Platinum=197m	1 000	Bismuth-200	1,000
Platinum-197	1,000	Bismuth-201	1,000
Platinum=199	1 000	Bismuth-202	1,000
Platinum 200	1,000	Bismuth-203	100
Gold-193	1 000	Bismuth-205	100
Gold-194	1,000	Bismuth-206	100
Gold-105	100	Bismuth-207	10
Gold-199m	10	Bismuth-210m	0.1
Gold-198	100	Bismuth-210	1
Gold-199	100	Bismuth-212	10
Gold=200m	100	Bismuth-213	10
Gold=2000	100	Bismuth-214	100
Gold=200	1,000	Polonium-203	1,000
Monoumur 202-	1,000	Polonium-205	1,000
Monoury-193m	100	Polonium-207	1,000
mercury=193	1,000	Polonium-210	0.1

Radionuclide	Quantity (µCi)	Radionuclide	Quantity (µCi)
Astatine-207	100	Neptunium-234	100
Astatine-211	10	Neptunium-235	100
Radon-220	1	Neptunium-236	200
Radon-222		$(1, 15 \times 10^5 \vee)$	0.001
Francium-222	100	Neptupium-236	0.001
Francium-223	100	(22.56)	
Radium-223	100 1	Nontunium-227	1 001
Radium-224	0.1	Neptunium-239	1.001
Radium-225	0.1	Neptunium 230	100
Padium-226	0.1	Neptunium-235	100
Radi m= 227	1 000	Neptunium-240	1,000
Padi ima 220	1,000	Plutonium-234	10
Nau ull-220	0.1	Plutonium-235	1,000
Ac infum-224	1	Plutonium-236	0.001
Ac inium-225	0.01	Plutonium-237	100
A tinium-226	0.1	Plutonium-238	0.001
Actinium-227	0.001	Plutonium-239	0.001
Actinium-228	1	Plutonium-240	0.001
Thorium-226	10	Plutonium-241	0.01
Thorium-227	0.01	Plutonium-242	0.001
Thorium-228	0.001	Plutonium-243	1,000
Thorium-229	0.001	Plutonium-244	0.001
Thorium-230	0.001	Plutonium-245	100
Thorium-231	100	Americium-237	1.000
Thorium-232	100	Americium-238	100
Thorium-234	10	Americium-239	.00
Thorium-natural	100	Americium-240	100
Protactinium-227	10	Americium-241	0 001
Protactinium-228	1	Americium-242m	0.001
Protactinium-230	0.1	Americium-242	10
Protactinium-231	0.001	Americium-243	0 001
Premactinium-232	1	Americium-24Am	1 )
Prr actinium-233	100	Americium-244	10
Protactinium-234	100	Americium-245	1 000
Uranium-230	0.01	Amonicium-246m	1,000
Uranium-231	100	Anonicium-246	1,000
Uranium-232	0 001	Currisme 230	1,000
Uranium-233	0.001	Cumium-240	100
livanium-234	0.001	Curium-24C	0.1
linanium-225	0.001	Curium-241	1
Upanium-226	0.001	Curium-242	0.01
Unanium-227	200	Curium-243	0.001
Unanium-220	100	Curium-244	0.001
Unanium-238	100	Curium-245	0.001
Unanium-239	1,000	Curium-246	0.001
Uranium-240	100	Curium-247	0.001
Uranium-natural	100	Curium-248	0.001
Neptunium-232	100	Curium-249	1,000
Neptunium-233	1,000	Berkelium-245	100

Radionuclide	Quantity (µCi)	Radionuclide	Quantity	(µCi)
Berkelium~246	100	Finsteinium-250	100	*****
Berkelium-247	0.001	Finsteinium-251	100	
Berkelium-249	0.1	Finsteinium-253	0 1	
Berkelium-250	10	Finsteinium 254m	2.1	
Californium-244	100	Fineteinium=254	0.01	
Californium-246	1	Formium-252	0.01	
Californium-248	0 01	Formium-252	1	
Californium-249	0.001	Fermium-254	10	
Californium-250	0.001	Fermium-255	10	
Californium-251	0.001	Formium=257	0.01	
Californium-252	0.001	Mendelevium-257	10.01	
Californium-253	0.1	Mendelevium-250	10 01	
Californium-254	0.001	Hende Tev Tull 200	0.01	
Any alpha emitting		Any radionuclide		
radionuclide not		other than alpha		
listed above or		emitting radionuclides		
mixtures of alpha		not listed above, or		
emitters of unknown		mixtures of beta		
composition	0.001	emiiters of unknown		
		composition	0.01	

NOTE: For purposes of §§ 20.902(e), 20.905(a), and 20.1201(a) where there is involved a combination of radionuclides in known amounts, the limit for the combination should be derived as follows: determine, for each radionuclide in the combination, the ratio between the quantity present in the combination and the limit otherwise established for the specific radionuclide when not in combination. The sum of such ratios for all radionuclides in the combination may not exceed "1" (i.e., "unity").

The quantities listed above were derived by taking 1/10th of the most restrictive ALI listed in Table 1, Columns 1 and 2, of Appendix B of this part, rounding to the nearest factor of 10, and arbitrarily constraining the values listed between 0.001 and 1,000  $\mu$ Ci. Values of 100  $\mu$ Ci have been assigned for radionuclides having a radioactive half-life in excess of 10<sup>9</sup> years (except rhenium, 1000  $\mu$ Ci) to take into account their low specific activity.

297

### APPENDIX D

UNITED STATES NUCLEAR REGULATORY COMMISSION REGION "FFICES

	Address	Telephone (24 hours)
Region I: Connecticut, Deleware, District of Columbia, Maine, Maryland, Massachusetts, New Hampshire, New Jersey, New York, Pennsylvania, Rhode Island, and Vermont.	USNRC, Region I 475 Allendale Road King of Prussid, PA 19406	(215) 337-5000, (FTS) 346-5000.
Region II: Alabama, Florida, Georgia, Kentucky, Mississippi, North Carolina, Puerto Rico, South Carolina, Tennessee, Virginia, Virgin Islands, and West Virginia.	USNRC, Region II 101 Marietta Street, NW Suite 2900 Atlanta, GA 30323	(404) 331-4503, (FTS) 841-4503.
Region III: Illinois, Indiana, Iowa, Michigan, Minnesota, Missouri, Ohio, and Wisconsin.	USNRC, Region III 799 Roosevelt Road Glen Ellyn, IL 60137	(708) 790-5500, (FTS) 388-5500.
Region IV: Arkansas, Colorado, Idaho, Kansas, Louisiana, Montana, Nc raska, New Mexico, North Dakota, Oklahoma, South Dakota, Texas, Utah, and Wyoming.	USNRC, Region IV 611 Ryan Plaza Drive Suite 1000 Arlington, TX 76011	(817) 860-8100 (FTS) 728-8100.
Region IV: Field Office	USNRC, Region IV Uranium Recovery Field Office 730 Simms Street, Suite 100a Golden, CO 80401 Mail: P.O. Box 25325 Denver, CO 80225	(303) 236-2805. (FTS) 776-2805.
Region V: Alaska, Arizona, California, Hawaii, Nevada, Oregon, Washington, and U.S. territories and possessions in the Pacific.	USNRC, Region V 1450 Maria Lane Suite 210 Walnut Creek, CA 94596	(415) 943-3700, (FTS) 463-3700.

#### APPENDIX E [RESERVED]

### APPENDIX F

## REQUIREMENTS FOR LOW-LEVEL-WASTE TRANSFER FOR DISPOSAL AT LAND DISPOSAL FACILITIES AND MANIFESTS

### I. MANIFEST

The shipment manifest shall contain the name, address, and telephone number of the person generating the waste. The manifest shall also include the name, address, and telephone number or the name and EPA hazardous waste identification number of the person transporting the waste to the land disposal facility. The manifest must also indicate as completely as practicable: a physical description of the waste, the volume, radionuclide identity and quantity, the total radioactivity, and the principal chemical form. The solidification agent must be specified. Waste containing more than 0.1% chelating agents by weight must be identified and the weight percentage of the chelating agent estimated. Wastes classified as Class A, Class B, or Class C in § 61.55 of this chapter must be clearly identified as such in the manifest. The total quantity of the radionuclides <sup>3</sup>H, <sup>14</sup>C, <sup>99</sup>Tc, and <sup>129</sup>I must be shown. The manifest required by this paragraph may be shipping papers used to meet Department of Transportation or Environmental Protection Agency regulations or requirements of the receiver, provided all the required information is included. legible \_notocopies.

#### II. CERTIFICATION

The waste generator shall include in the shipment manifest a certification that the transported materials are properly classified, described, packaged, marked, and labeled and are in proper condition for transportation according to the applicable regulations of the Department of Transportation and the Commission. An authorized representative of the waste generator shall sign and date the manifest.

III. CONTROL AND TRACKING

A. Any generating licensee who transfers radioactive waste to a land disposal facility or = licensed waste collector shall comply with the requirements in paragraphs A.1 through 8 of this section. Any generating licensee who transfers waste to a licensed waste processor who treats or repackages waste shall comply with the requirements of paragraphs A.4 through 8 of this section. A licensee shall:

 Prepare all wastes so that the waste is classified according to § 61.55 of this chapter and meets the waste characteristics requirements in § 61.56 of this chapter;

2. Label each package of waste to identify whether it is Class A waste, Class B waste, or Class C waste, in accordance with § 61.55 of this chapter;

3. Conduct a quality control program to ensure compliance with §§ 61.55 and 61.56 of this chapter, the program must include management evaluation of audits;

 Prepare shipping manifests to meet the requirements of sections I and II of this appendix;

5. Forward a copy of the manifest to the intended recipient, at the time of shipment, or deliver to a collector at the time the waste is collected, obtaining acknowledgment of receipt in the form of a signed copy of the manifest or equivalent documentation from the collector;

6. Include one copy of the manifest with the shipment;

 Retain a copy of the manifest and documentation of acknowledgment of receipt as the record of transfer of licensed material as required by Parts 30, 40, and 70 of this chapter; and

8. For any shipments or any part of a shipment for which acknowledgment of receipt has not been received within the times set forth in this section, conduct an investigation in accordance with paragraph E of this appendix.

300

B. Any waste collector licensee who handles only prepackaged waste shall:

 Acknowledge receipt of the waste from the generator within 1 week of receipt by returning a signed copy of the manifest or equivalent documentation.,

2. Prepare a new manifest to reflect consolidated shipments; the new manifest shall serve as a listing or index for the detailed generator manifests. Copies of the generator manifests shall be a part of the new manifest. The waste collector may prepare a new manifest without attaching the generator manifests, provided the new manifest contains for each package the information specified in section I of this appendix. The collector licensee shall certify that nothing has been done to the waste that would invalidate the generator's certification;

 Forward a copy of the new manifest to the land disposal facility operator at the time of shipment;

4. Include the new manifest with the shipment to the disposal site;

 Retain a copy of the manifest and documentation of acknowledgment of receipt as the record of transfer of licensed material as required by Parts 30,
40, and 70 of this chapter, and retain information from generator manifest until disposition is authorized by the Commission; and

6. For any chipments or any part of a shipment for which acknowledgment of receipt is not received within the times set forth in this section, conduct an investigation in accordance with paragraph E of this section.

C. Any licensed waste processor who treats or repackages wastes shall:

 Acknowledge receipt of the waste from the generator within 1 week of receipt by returning a signed copy of the manifest or equivalent documentation;

2. Prepare a new manifest that meets the requirements of sections I and II of this appendix. Preparation of the new manifest reflects that the processor is responsible for the waste;

Prepare all wastes so that the waste is classified according to
§ 61.55 of this chapter and meets the waste characteristics requirements in
§ 61.56 of this chapter;

4. Label each package of waste to identify whether it is Class A waste, Class B waste, or Class C waste, in accordance with §§ 61.55 and 61.57 of this chapter;  Conduct a quality control program to ensure compliance with §§ 61.55 and 61.56 of this chapter. The program shall include management evaluation of audits;

6. Forward a copy of the new manifest to the disposal site operator or waste collector at the time of shipment, or deliver to a collector at the time the waste is collected, obtaining acknowledgment of receipt in the form of a signed copy of the manifest or equivalent documentation by the collector;

7. Include the new manifest with the shipment;

8. Retain copies of original manifests and new manifests and documentation of acknowledgment of receipt as the record of transfer of licensed material required by Parts 30, 40, and 70 of this chapter; and

9. For any shipment or part of a shipment for which acknowledgment is not received within the times set forth in this section, conduct an investigation in accordance with paragraph E of this section.

D. The land disposal facility operator shall:

1. Acknowledge receipt of the waste within 1 week of receipt by returning a signed copy of the manifest or equivalent documentation to the shipper. The shipper to be notified is the licensee who last possessed the waste and transferred the waste to the operator. The returned copy of the manifest or equi alent documentation shall indicate any discrepancies between materials listed on the manifest and materials received.

2. Maintain copies of all completed manifests or equivalent documentation until the Commission authorizes their disposition; and

3. Notify the shipper (i.e., the generator, the collector, or processor) and the Administrator of the nearest Commission Regional Office listed in Appendix D to this part when any shipment or part of a shipment has not arrived within 60 days after the advance manifest was received.

E. Any shipment or part of a shipment for which acknowledgment is not received within the times set forth in this section must:

1. Be investigated by the shipper if the shipper has not received notification or receipt within 20 days after transfer; and

2. Be traced and reported. The investigation shall include tracing the shipment and filing a report with the nearest Commission Regional Office listed

in Appendix D to this part. Each licensee who conducts a trace investigation shall file a written report with the appropriate NRC Regional Office within 2 weeks of completion of the investigation.

### JONFORMING AMENDMENTS

The following amendments to other parts of Chapter I of Title 10 generally update citations to 10 CFR Part 20 that are found in these other parts of the NRC regulations. Two amendments are particularly important as they go beyond updating cross-reference citations. The amendment to 10 CFR Part 2 Appendix C updates and modifies the examples of the severity levels associated with violations of 10 CFR Part 20. Because Appendix C relates to administrative policy of the Commission and because the listed violations are used as examples of different severity levels and are not all-inclusive, the Commission is issuing these Part 2 amendments in final form without public comment. The other important change in the conforming amendments is the deletion of "upon request" from § 19.13(b). This has the effect of requiring annual dose reports to all workers rather than only upon a request by the worker. This change conforms to the 1987 Federal Radiation Guidance from the President.

### PART 2- RULES OF PRACTICE

The authority citation for Part 2 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, a. amended (42 U.S.C. 2201); sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

3. Supplement IV -- Severity Categories of Appendix C to 10 CFR Part 2 is amended to read as follows:

Appendix C--General Statement of Policy and Procedures for NRC Enforcement Actions

### Health Physics 10 CFR Part 2015

A. Severity I -- Violations involving for example:

1. Single radiation exposure of a worker in excess of 25 rems total effective dose equivalent, 75 rems to the lens of the eye, or 250 rads to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Single radiation exposure of the embryo/fetus of a declared pregnant woman in excess of 2.5 rems total effective dose equivalent;

3. Single radiation exposure of a minor in excess of 2.5 rems total effective dose equivalent, 7.5 rems to the lens of the eye, or 25 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Annual exposure of a member of the public in excess of 2.5 rems total effective dose equivalent;

5. Release of radioactive material to an unrestricted area at concentrations in excess of 50 times the limits for members of the public in Appendix B, Table 2, of 10 CFR Part 20; or

6. Disposal of licensed material in quantities or concentrations in excess of 10 times the limits of 10 CFR 20.1003.

B. Severity II -- Violations involving for example:

1. Single radiation exposure of a worker in excess of 10 rems total effective dose equivalent, 30 rems to the lens of the eye, or 100 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

<sup>&</sup>lt;sup>15</sup> Personnel overexposures and associated violations, incurred during a lifesaving effort, will be treated on a case-by-case basis.

 Single radiation exposure of the embryo/fetus of a declared pregnant woman in excess of 1.0 rem total effective dose equivalent;

3. Single radiation exposure of a minor in excess of 1 rem total effective dose equivalent; 3.0 rems to the lens of the eye, or 10 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Annual exposure of a member of the public in excess of 0.5 rem total effective dose equivalent;

5. Release of radioactive material to an unrestricted area at concentrations in excess of 10 times the limits for members of the public in Appendix B, Table 2, of 10 CFR Part 20;

 Disposal of licensed material in quantities or concentrations in excess of five times the limits of 10 CFR 20.1003; or

 Failure to make an immediate notification as required by 10 CFR 20.1202(a)(1) or (a)(2).

C. Severity III -- Violations involving for example:

1. Single radiation exposure of a worker in excess of 5 rems total effective dose equivalent, 15 rems to the lens of the eye, or 50 rems to the skin of the whole body or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Single radiation exposure of the embryo/fetus of a declared pregnant woman in excess of 0.5 rem total effective dose equivalent;

3. Single radiation exposure of a minor in excess of 0.5 rem total effective dose equivalent; 1.5 rems to the lens of the eye, or 5 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Worker exposure above regulatory limits when such exposure reflects a programmatic (rather than an isolated) weakness in the radiation control program;

5. Annual exposure of a member of the public in excess of 0.1 rem total effective dose equivalent (except when operation up to 0.5 rem a year has been approved by the Commission under § 20.301(c));

6. Release of radioactive material to an unrestricted area at concentrations in excess of two times the limits for members of the public in Appendix B to 10 CFR Part 20 (except when operation up to 0.5 rem a year has been approved by the Commission under § 20.301(c));

 Failure to make a 24-hour notification required by 10 CFR 20.1202(b) or an immediate notification required by 10 CFR 20.1201(a)(1)(i);

8. Substantial potential for exposures or releases in excess of the applicable limits in 10 CFR Part 20 whether or not such exposure or release occurs (e.g., operation of a radiation facility with a nonfunctioning interlock system or entry into high radiation areas, such as under reactor vessels or in the vicinity of exposed radiographic sources, without having performed an ade-quate survey);

9. Improper disposal of licensed material not covered in Severity Levels I or II,

10. Release for unrestricted use of contaminated or radioactive material or equipment that poses a realistic potential for exposure of the public to levels or doses exceeding the annual dose limits for member of the public, or that reflects a programmatic (rather than an isolated) weakness in the radiation control program;

Conduct of licensee activities by a technically unqualified person;
or

12. Significant failure to control licensed material.

D. Severity IV -- Violations involving for example:

1. Exposures in excess of the limits of 10 CFR 20.201, 20.207, or 20.208 not constituting Severity Level I, II, or III violations;

2. Release of radioactive material to an unrestricted area at concentrations in excess of the limits for members of the public in Appendix B to 10 CFR Part 20 (except when operation up to 0.5 rem a year has been approved by the Commission under § 20.301(c));

3. A radiation dose rate in an unrestricted or controlled area in excess of 0.002 rem in any 1 hour (2 millirem/hour) or 50 millirems in a year;

4. Failure to maintain and implement radiation programs to keep radiation exposures as low as is reasonably achievable;

5. Doses to a member of the public in excess of any EPA generally applicable environmental radiation standards, such as 40 CFR Part 190;

Failure to make the 30-day notification required by 10 CFR
20.1201(a)(1)(ii) or 20.1203(a);

7. Failure to make a timely written report as required by 10 CFR 20.1201(b), 20.1204, or 20.1206; or

8. Any other matter that has more than a minor safety, health, or environmental significance.

E. Severity V -- Violations that are of a minor safety, health, or environmental significance.

PART 19 - NOTICES, INSTRUCTIONS, AND REPORTS TO WORKERS; INSPECTIONS

 The authority citation for Part 19 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); sec. 201, 88 Stat. 1242, as amended (42 U.S.C. 5841).

5. Section 19.3 is amended by revising paragraph (e) to read as follows:

\*

### § 19.3 Definitions.

\*

×

(e) "Restricted area" means an area, access to which is limited by the licensee for the purpose of protecting individuals against undue risks from exposure to radiation and radioactive materials. Restricted area does not include areas used as residential quarters, but separate rooms in a residential building may be set apart as a restricted area.

6. In § 19.13, paragraph (d) is amended by changing the reference to "\$20.405 and § 20.408" to read "§\$20.1202, 20.1203, 20.1204, or 20.1206" and by revising paragraphs (b), (c), and (e) to read as follows:

### § 19.13 Notifications and reports to individuals.

(b) Each licensee shall advise each worker annually of the worker's dose as shown in records maintained by the licensee pursuant to § 20.1106.

(c) At the request of a worker formerly engaged in licensed activities controlled by the licensee, each licensee shall furnish to the worker a report of the worker's exposure to radiation or radioactive material for each year the worker was required to be monitored under § 20.502. Such report shall be furnished within 30 days from the time the request is made, or within 30 days after the exposure of the individual has been determined by the licensee, whichever is later. This report shall cover the period of time that the worker's activities involved exposure to radiation from radioactive materials licensed by the Commission and shall include the dates and locations of licensed activities in which the worker participated during this period.

(e) At the request of a worker who is terminating employment with the licensee that involved exposure to radiation or radioactive materials, during the current year, each licensee shall provide at termination to each such worker, or to the worker's designee, a written report regarding the radiation dose received by that worker from operations of the licensee during the current year or fraction thereof. If the most recent individual monitoring results are not available at that time, a written estimate of the dose shall be provided together with a clear indication that this is an estimate.

# PART 30 - RULES OF GENERAL APPLICABILITY TO DOMESTIC LICENSING OF BYPRODUCT MATERIAL

7. The authority citation for Part 30 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

### § 30.51 [Amended].

8. In § 30.51(c)(4), the reference to "§ 20.401(c)" is changed to read "§ 20.1108."

# PART 31 - GENERAL DOMESTIC LICENSES FOR BYPRODUCT MATERIAL

9. The authority citation for Part 31 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

### § 31.5 [Amended].

10. In § 31.5(c)(10), the reference to '§§ 20.402 and 20.403" is changed to read "§§ 20.1201 and 20.1202."

#### § 31.7 [Amended].

11. In § 31.7(b), the reference to "§§ 20.402 and 20.403" is changed to read "§§ 20.1201 and 20.1202."

### § 31.10 [Amended].

12. In § 31.10(b)(1) the reference to "§ 20.301" is changed to read "§ 20.1001."

13. In § 31.10(b)(3) the reference to "§§ 20.301, 20.402, and 20.403" is changed to read "§§ 20.1001, 20.1201, and 20.1202."

### § 31.11 [Amended]

14. In § 31.11(c)(5), the reference to "§ 20.301" is changed to read "§ 20.1001."

15. In § 31.11(f), the reference to "§§ 20.301, 20.402, and 20.403" is changed to read "§§ 20.1001, 20.1201, and 20.1202."

## PART 32 - SPECIFIC DOMESTIC LICENSES TO MANUFACTURE OR TRANSFER CERTAIN ITEMS CONTAINING BYPRODUCT MATERIAL

16. The authority citation for Part 32 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242 as amended (42 U.S.C.5841).

17. Section 32.51 is amended by revising paragraphs (a)(2)(ii) and (c) to read as follow:

§ 32.51 Byproduct material contained in devices for use under § 31.5;

requirements for license to manufacture or initially transfer.

(a) \* \* \*

(2) \* \* \*

(ii) Under ordinary conditions of handling, storage, and use of the device, the byproduct material contained in the device will not be released or inadvertently removed from the device, and it is unlikely that any person will receive in any period of 1 year a dose in excess of 10 percent of the annual limits specified in § 20.201(a) of this chapter; and

\* \* \* \* \*

(c) In the event the applicant desires that the general licensee under § 31.5 of this chapter, or under equivalent regulations of an Agreement State, be authorized to install the device, collect the sample to be analyzed by a specific licensee for leakage of radioactive material, service the device, test the on-off mechanism and indicator, or remove the device from installation, the applicant shall include in the application written instructions to be followed by the general licensee, estimated calendar quarter doses associated with such activity or activities, and the bases for these estimates. The submitted information must demonstrate that performance of this activity or activities by an individual untrained in radiological protection, in addition to other handling, storage, and use of devices under the general license, is unlikely to cause that individual to receive in a year a dose in excess of 10 percent of the annual limits specified in § 20.201(a) of this chapter.

§ 32.61 [Amended].

18. In § 32.61(d), the reference to "§ 20.203(a)" is changed to read "§ 20.901(a)."

§ 32.71 [Amended].

19. In § 32.71(c)(2), the reference to "§ 20.203(a)(1)" is changed to read "§ 20.901(a)."

20. In § 32.71(e), the reference to "§ 20.301" is changed to read "§ 20.1001."

## PART 34 - LICENSES FOR RADIOGRAPHY AND RADIATION SAFETY REQUIREMENTS FOR RADIOGRAPHIC OPERATIONS

21. The authority citation for Part 34 continues to read in part as follows:

<u>Authority</u> : Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

### § 34.29 [Amended]

22. In § 34.29(a), the reference to "§ 20.203(c)(2)(11), (2)(11), or (4)" is changed to read "§ 20.601(a)(2), (a)(3), or (b)."

### § 34.41 [Amended].

23. In § 34.41(a), the reference to "§ 20.203(c)(2)" is changed to read "§ 20.601(a)(1), (a)(2), or (a)(3)."

## § 34.42 [Amended].

24. In § 34.42, the reference to "§ 20.204(c)" is changed to read "§ 20.903" and the reference to "§ 20.203(b) and (c)(1)" is changed to read "§ 20.902(a) and (b)."

## PART 35 - MEDICAL USE OF BYPRODUCT MATERIAL

25. The authority citation for Part 35 continues to read as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

### § 35.92(a).

26. Change reference to "§ 20.301" to "§ 20.1001."

### § 35.315(a)(8)

27. Change reference to "§ 20.401(c)(1)" to "§ 20.1106(a)."

### \$ 35.415.

28. Change reference to "§ 20.105(b)" to "§ 20.301(a)."

### § 35.630(a)(1).

29. Change reference to "National Bureau of Standards" to "National Institute of Standards and Technology."

### § 35.630(a)(2).

30. Change reference to "National Bureau of Standards" to "National Institute of Standards and Technology."

### § 35.641(a)(2)(i).

31. Change reference to "§ 20.101" to "§ 20.201."

### § 35.641(a)(2)(ii).

32. Change reference to "§ 20.105(b)" to "§ 20.301."

### § 35.641(b)(2).

33. Change reference to "§ 20.501" to "§ 20.1301."

## § 35.643 (a).

34. Change reference to "§ 20.105(b)" to "§ 20.301."

### § 35.643(a)(1).

35. Change reference to "§ 20.105(b)" to "§ 20.301."

### § 35.630(a)(2).

36. Change reference to "§ 20.105(a)" to "§ 20.301(c)."

37. Change reference to "§ 20.105(b)" to "§ 20.301(a)."

PART 39 - LICENSES AND RADIATION SAFETY REQUIREMENTS FOR WELL-LOGGING

38. The authority citation for Part 39 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

39. In § 39.15(a)(5)(iii)(B), the reference to "§ 20.203" is changed to read ""§ 20.901(a)."

40. In § 39.31(a)(1), the reference to "§ 20.203" is changed to read "§ 20.901(a).

41. In § 39.31(a)(2), the reference to "§ 20.203" is changed to read "§ 20.901(a)."

42. In § 39.77(b), the reference to "§§ 20.402, 20.403, and 20.405" is changed to read "§§ 20.1201 and 20.1203."

PART 40 - DOMESTIC LICENSING OF SOURCE MATERIAL

43. The authority citation for Part 40 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948 as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C. 5841).

44. Section § 40.34 is amended by revising paragraph (a)(2) to read as follows:

## § 40.34 Special requirements for issuance of specific licenses.

(a) \* \* \*

(2) The applicant submits sufficient information relating to the design, manufacture, prototype testing, quality control procedures, labeling or marking, proposed uses, and potential hazards of the industrial product or device to provide reasonable assurance that possession, use, or transfer of the depleted uranium in the product or device is not likely to cause any individual to receive in any period of 1 year a radiation dose in excess of 10 percent of the annual limits specified in § 20.201(a) of this chapter; and

### § 40.61 [Amended].

45. In § 40.61(c)(4), the reference to "§ 20.401(c)" is changed to read "§ 20.1108."

### Appendix A to Part 40.

46. In the Introduction to Appendix A, the reference to "§ 20.1(c)" is changed to read "§ 20.3."

## PART 50 - DOMESTIC LICENSING OF PRODUCTION AND UTILIZATION FACILITIES

47. The authority citation for Part 50 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

48. Section 50.34 is amended by revising paragraph (f)(2)(viii) to read as follows:

\* \*

(f) \* \* \*

(2) \* \* \*

(viii) Provide a capability to promptly obtain and analyze samples from the reactor coolant system and containment that may contain TID-14844 source term radioactive materials without radiation exposures to any individual exceeding 5 rems to the whole body or 50 rems to the extremities. Materials to be analyzed and quantified include certain radionuclides that are indicators of the degree of core damage (e.g., noble gases, iodines and cesiums, and nonvolatile isotopes), hydrogen in the containment atmosphere, dissolved gases, chloride, and boron concentrations. (II.B.3)

49. In § 50.36a(a), the reference to "§ 20.106" is changed to read "§ 20.301" and paragraph (b) is revised to read as follows:

# § 50.36a Technical specifications on effluents from nuclear power reactors.

(b) In establishing and implementing the operating procedures described in paragraph (a) of this section, the licensee shall be guided by the following considerations: Experience with the design, construction, and operation of nuclear power reactors indicates that compliance with the technical specifications described in this section will keep average annual releases of radioactive material in effluents and their resultant committed effective dose equivalents at small percentages of the values specified in § 20.301 of this chapter and in the operating license. At the same time, the licensee is permitted the flexibility of operation, compatible with considerations of health and safety, to assure that the public is provided a dependable source of power even under unusual operating conditions which may temporarily result in releases higher than such small percentages, but still within the dose values specified in § 20.301 of this chapter and in the operating license. It is expected that in using this operational flexibility under unusual operating conditions, the licensee will exert its best efforts to keep levels of radioactive material in effluents as low as is reasonably achievable. The guides set out in Appendix I provide numerical guidance on limiting conditions for operation for light-watercooled nuclear power reactors to meet the requirement that radioactive materials in effluents released to unrestricted areas be kept as low as is reasonably achievable.

50. In § 50.72 in paragraph (a), Footnote 1, the reference to "§ 20.205, § 20.403" is changed to read "§ 20.906, § 20.1202," and paragraphs (b)(2)(iv) (A) and (B) are revised to read as follows:

§ 50.72 Immediate notification requirements for operating nuclear power

reactors.

\*

- (b) \* \* \*
- (2) \* \* \*

(iv) (A) Any airborne radioactive release that results in concentrations in unrestricted areas that exceed 20 times the applicable concentration specified in Appendix B, Table 2, Column 1, of Part 20 of this chapter, when averaged over a time period of 1 hour.

(B) Any liquid effluent release that exceeds 20 times the applicable concentration specified in Appendix B, Table 2, Column 2, of Part 20 of this chapter at the point of entry into the receiving waters (i.e., unrestricted area) for all radionuclides except tritium and dissolved noble gases, when averaged over a time period of 1 hour. (Immediate notifications made inder this paragraph also satisfy the requirements of paragraphs (a)(2) and (b)(2) of § 20.1202 of this chapter.)

51. Section 50.73 is amended by revising paragraphs (a)(2)(viii)(A) and (B) and (ix) to read as follows:

- § 50.73 Licensee event report system.
  - (a) \* \* \* (2) \* \* \*

(viii)(A) Any airborne radioactivity release that exceeded 20 times the applicable concentrations specified in Appendix B, Table 2, Column 1, of Part

20 of this chapter in unrestricted areas, when averaged over a time period of 1 hour.

(B) Any liquid effluent release that exceeded 20 times the applicable concentrations specified in Appendix B, Table 2, Column 2, of Part 20 of this chapter at the point of entry into the receiving water (i.e., unrestricted area) for all radionuclides except tritium and dissolved noble gases, when averaged over a time period of 1 hour.

(ix) Reports submitted to the Commission in accordance with paragraph (a)(2)(viii) of this section also meet the effluent release reporting requirements of § 20.1203(a)(3) of this chapter.

\*

## PART 61 - LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE

52. The authority citation for Part 61 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

§ 61.52 [Amended].

53. In § 61.52(a)(6), the reference to "§ 20.105" is changed to read "§§ 20.301 and 20.302."

## PART 70 - DOMESTIC LICENSING OF SPECIAL NUCLEAR MATERIAL

54. The authority citation for Part 70 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

§ 70.51 [Amended].

55. In § 70.51(b)(6), the reference to "§ 20.401(c)" is changed to read "§ 20.1108."

Dated at Rockville, Maryland, this \_\_\_\_\_ day of \_\_\_\_\_ 1990. For the Nuclear Regulatory Commission.

> Samuel J. Chilk, Secretary of the Commission.

ENCLOSURE B MARKED COPIES OF THE SRM

Enclosure B
## KEY TO CHANGES TO TEXT OF REVISED PART 20 FEDERAL REGISTER NOTICE

# (REFLECTS CHANGES FROM THE TEXT FROM APPENDICES 3, 4, AND 5 OF SECY-88-315)

KEY: = Addition; [ ] = Deletion; tr = same wording but transposed order Vertical lines in margin = large insertion or replacement of text

Margin Code	Reference for Change
267 p.6	SECY-89-267, Enclosure 3, page 6
237 # 2	SECY-90-237, Enclosure A, item 2
SRM # 4	Staff Requirements Memorandum, July 30, 1990, item # 4
KCVS p.3 KCMU p.45	Chairman Carr's Votesheet, page 3 Chairman Carr's Mark-up, page 45
RoVS p. 3 RoMU p. 2	Commissioner Rogers' Votesheet page 2 Commissioner Rogers' Mark-Up page 2
ReVS p. 3 ReMU p 2	Commissioner Remick's Votesheet page 2 Commissioner Remick's Mark-up page 2
JCVS p. 3. JCAC	Commissioner Curtiss's Votesheet p. 3 Commissioner Curtiss's Additional Comments
RuR	Change recommended by Division of Rules and Records
CGC	Change recommended by the Office of General Counsel
IRM	Change recommended by the Office of Information Resource Management
EDC	Editorial Change to improve clarity
EDS	Editorial change to improve style and grammar

Cross-Indices to Documents Used for Quality Assurance:

- NUREG/BR-0095, "Checklist for Preparation and Review of Federal Register Rulemaking Documents"
- 2. Staff Requirements Memorandum of July 30, 1990 on SECY-89-247/SECY-88-315/ SECY-90-237
- Markups Accompanying the Votesheets of Chairman Carr, Commission Rogers, and Commissioner Remick, and Commissioner Curtiss's additional comments.
- 4. Enclosure 6 of SECY-88-315, "Changes from Proposed Rule" per SRM # 7.

UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON D.C. 20555

July 30, 1990

ACTION - Beckjord, RES

Cvs: Taylor Sniezek Thompson Blaha Murley, NRR Bernero, NMSS Jordan, AEOD Scroggins, OC HPeterson, RES DMeyer, ADM BShelton, IRM

OFFICE OF THE SECRETARY

> James M. Taylor Executive Director for Operations

William C. Parler General Counsel Bate Samuel J. Chilk, Secretary

FROM:

SUBJECT:

MEMORANDUM FOR:

SECY-89-267/SECY-88-315/SECY-90-237 - REVISION OF 10 CFR PART 20 - STANDARDS FOR PROTECTION AGAINST RADIATION

This is to advise you that the Commission (with Chairman Carr and Commissioners Rogers, Curtiss, and Remick agreeing except as noted below) has approved the proposed revisions of 10 CFR Part 20 as presented in SECY-88-315, revised in SECY-89-267 and SECY-90-237, and subject to the modifications listed below.

Following staff completion of the following items the rule should be returned for final Commission review, affirmation and publication in the Federal Register. Publication of the new Part 20 should be accompanied by appropriate efforts to disseminate information about the rule to licensees, other Federal agencies, States, the Congress and the public.

(Resu/OGC))

(SECY Suspense: 9/90)

(RES)

The Commission (with Chairman Carr and Commissioners Rogers and Remick 1. approving) has agreed that publication of the rule changes can be supported under the backfit rule as described below.

The Federal Register Notice should incorporate the staff's summary of the Pages revised backfit analysis based on a finding that the revisions to Part 20 104provide for a substantial increase in safety. The analysis should 105 conclude with the following paragraph:

"The Commission is adopting the final rule based on the conclusion page of this analysis that the rule provides for a substantial increase '105 in the overall protection of the public health and safety and that 2nd 9 the direct and indirect costs of its implementation are justified in terms of the quantitative and qualitative benefits associated with

NOTE: THIS SRM, THE SUBJECT SECY PAPER, AND THE VOTE SHEETS OF COMMISSIONERS CURTISS, REMICK, AND ROGERS WILL BE MADE PUBLICLY AVAILABLE WHEN THE FEDERAL REGISTER NOTICE IS FUBLISHED. 111.1.200

Tima 7.31.90 Tima 8.30 A

the rule. The Commission would note, however, that, even had the analysis not concluded that revised Part 20 provides a substantial increase in the overall protection of the public health and safety, polar it could have gone forward with the rule because the changes made to 2 mol 97 Part 20 also amount to a redefinition of the level of adequate protection and the backfit rule's cost justification standard does not apply to a redefinition of adequate rotection."

Commissioner Curtiss believes that this rulemaking constitutes a redefinition of adequate protection. He believes that the Statement of Consideration and the backfit discussion should be modified as necessary to reflect this determination; he does not believe that the Commission should attempt to justify the revision as a substantial increase in the overall protection of the public health and safety. His comments on this are included in his vote project sheet and are attached hereto.

- The revision of Part 20 should become effective on January 1, 1992 and the staff should complete, to the maximum extent practicable, development 2. of the necessary regulatory guidance documents by January 1, 1991. Early completion of the guidance, at least in draft form, should provide time for licensees to review and comment on the guidance and to develop and implement the measures necessary to comply with the new Part 20 by the Retery to effective date. In preparing regulatory guidance, the staff should Regal mory ensure that it provides for the same flexibilities that have been incorporated into the rule, particularly in the areas of (1) determining & rde CON test compliance with the occupational dose limits involving internally deposited radionuclides and (2) establishing site-specific effluent limits in air and water considering physical and environmental characteristics that influence potential doses to members of the public. The language in the Statement of Consideration, the rule, and the apply only within an envelope of equivalent safety and protection (i.e., 2 sertime. Part 20 provides flexibility in how the dose calculations are performed 2 sertime. guidance documents should clearly emphasize that these flexibilities but in all cases, unless specifically exempted, the dose limits in prof work Sections 20.201 and 20.301 apply). The language in Enclosure 3, pages 2, for public 4 and 8 of SECV-20-267 should also be removed to remove that flowing it is the public 4, and 8 of SECY-89-267 should also be revised to ensure that flexibility is clearly and correctly reflected. =302 = 040 bettern; E304 = page 54 For; E308 (rate - Heribility me gains; The discussion in the Statement of Consideration (pg. 13 of Enclosure Mile
  - 3. The discussion in the Statement of Consideration (pg. 13 of Enclosure (MRC 3 to SECY-88-315) allows licensees to make pen and ink changes to their (Argorow) licenses to reflect these revisions to Part 20. Language should be added (10) to the rule itself in 10 CFR 20.8 to authorize the pen and ink changes in the 520.8
  - 4. The Statement of Consideration should be expanded to clarify the impact, if any, of the change in dose limits for members of the public from 500 does of the to 100 millirem/yr when conforming the general license design standards champed in Parts 32 and 40 (see conforming amendment in Enclosure 5 to SECY-88-315 on pages 144 and 147). The staff should ensure that the they are conforming revisions are consistent with the current intent. Champed neulor regume
  - 5. The Federal Register Notice should be updated as appropriate to reflect m partition for the Commission's recent decision on the Below Regulatory Concern Policy of examinent Statement. <math>page 15 PP(2) page 2-5

6. Staff should clarify the purpose of the rule and the definition of natural background radiation in the Statement of Consideration and the rule, in regard to consideration of sources of radiation exposure that fall outside of the scope of Part 20 (e.g., fallout, NARM, x-rays) (pages 1 6 and 13 of Enclosure 4 of SECY-88-315).

- The attached modifications should be considered for incorporation into 7. the Federal Register Notice. Comments not incorporated should be identified within the final rulemaking package. In addition to the inserts provided in SECY-90-237, the notice should be reviewed to assure that no further changes are needed to reflect the national and international radiation protection developments that have occurred singe ABDS 9-14 the text was prepared (i.e. BEIR IV and V, UNSCEAR 1988, ICRP's 1990 recommendations, and NCRP's Report No. 106). Finally, the Notice should reger have a final quality control check, including use of Enclosure 6 of SECY-88-315, to be sure that issues raised in the Statement of compoletes Consideration are answered and that all significant changes between the A typichang see proposed and final rules are discussed.
- 8. The Commission understands that the rationale for the staff's preferred risk coefficient of 5E-4 is that while 4E-4 remains a good working coefficient for occupational exposures, the greater susceptibility of fetuses and children makes 5E-4 a better number for the population as a whole. This should be made plain in the Statement accompanying the rule.

On p. 5 of Enclosure A to SECY-90-237, the discussion notes that the range of fatal cancer risk from lower doses in UNSCEAR-88 (.7E-4 to 3.5E-4) is .6 to 5 times higher than the 1977 ICRP risk value of 1.25E-4. The correct relationship of 3.5 to 1.25 is about 3 times, not 5.

Finally the second paragraph of the discussion of the 1990 ICRP recommendations should be revised to state the following:

"Until the final ICRP recommendations are published, and the need for further revisions in NRC standards established, the Commission believes it would be advisable to proceed with promulgation of the proposed dose limits, rather than deferring a reduction of the existing limits to a future rulemaking. The Commission plans to review the comments of the professional community and others on the ICRP recommendations and ICRP's response to them. In addition, the Commission will review the recommendations of other expert bodies, such as the National Council on Radiation Protection and Measurements, and participate in the deliberations of the Committee for Interagency Radiation Research and Policy Coordination on the need for further revision of the occupational radiation protection standards after the ICRP recommendations are published."

The rule and the Statement of Consideration should be revised to incorporate the additional changes described in Enclosure A to SECY-90-237, as modified in the attached markings. (These changes partially implement items 3 and 7 listed above.)

3

page 19

bottom

PAge 12

Changel

10 2.8

time

ENCLOSURE 6 TO SECY-88-315

SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE ANNOTATION is where discussed in Statement. NOTE: \* denotes major change from proposed rule.

General Provisions

\$ 20.1 Purpose - sentence added to instruct licensees to take necessary actions in an emergency to protect life and property regardless of Part 20 requirements. This sentence conveys the essence of proposed § 20.9 which would have dispensed licensee from all of Part 20 in an emergency.

§ 20.2 Scope - essentially unchanged. p. 19

- § 20.3 Definitions extensively rewritten and reganized. Put into pure alphabetical order rather than being grouped (e.g. "monitoring terms"). See Statement for lists of new, modified and deleted definitions. pages 19-21
- \* § 20.101 Description of ICRP Dose Limitation System and linear, nonthreshold dose-effect assumption moved to Statement of Considerations, proposed § 20.102 moved to § 20.101 and emphasis changed from ALARA programs to ALARA within the context of an overal! Radiation Protection Program. Position on ALARA somewhat less rigid and prescriptive than proposed rule, but more of a requirement than the hortatory "licensees should" in the present rule.

pages 3-4, 24-28

# SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

Occupational Protection

- § 20.201 New terms introduced for "total effective dose equivalent," "eye dose equivalent" and "shallow dose equivalent" to distinguish in rule types of doses. Conditions of measurement (e.g. tissue depth) moved into definitions of terms. Uranium chemical toxicity limit moved into § 20,201 from § 20.204.
- \* Quarterly dose limit of 3 rem deleted. page 29
  - § 20.202 Clarified to set forth when summation of internal and external doses is required, but numerical monitoring thresholds are now only pages 35, 58-41 in § 20.502.
  - § 20.203 (Submersion Dose) is greatly reduced and advisory, nonregulatory text is deleted. page 35
  - § 20.204 Ability of licensee to modify DACs and ALIs based upon local exposure conditions (particle size, solubility, etc.) now requires prior NRC approval.
- \* § 20.205 Deleted as it is philosophically contrary to existing and past Commission procedures and practices for controlling exposures. pages 37 - 4/
- \* § 20.206 Changed to permit use of internal dose in Planned Special page 42 Exposures in order to keep the total effective dose equivalent ALARA.
  - § 20.207 Extends use of 10 % of adult dose limit for minors from deeppage 45 dose equivalent to eye dose and shallow dose for skin and extremities.
  - § 20.208 Made requirement to keep dose to embryo/fetus uniform into a proge 46 separate paragraph (provides consistency with NCRP 0.05 rem/month). Deleted advisory information on Class Y compounds which will be put into Regulatory Guide.

ENCLOSURE 6

SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

Public Protection

- \* § 20.301 and § 20.303 have been combined. Proposed "Reference Level" of 0.1 rem per year has been made the primary limit for members of the public. (Many commenters felt that the "reference level" was a page 51 de facto limit as it had the same recorrokeeping and reporting pro- 3rd A visions as a limit). This change reflects the 1985 clarification by the ICRP regarding the status of the public dose limits.
- \* The 2 millirem per hour limit for doses in unrestricted areas has been reinstated. (Useful for compliance evaluations.)

A 0.5-rem annual limit is available upon NRC approval if the need can be justified. (Primarily intended to provide for teletherapy and similar facilities whose shielding was designed to meet a 0.5-rem annual dose limit for unrestricted areas.)

Revision also clarifies that EPA generally-applicable environmental radiation standards are "limits" not "reference levels."

§ 20.302 is a new section that includes information on methods for complying with § 20.301 limits. Includes some material, from former § 20.303. discussed in concerned and a sector \$ 20.301

§ 20.303 [Deleted - see discussion on § 20.301 and § 20.302]. progr 53

\$ 20.304 [Deleted, pages 55-57]

SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

Surveys & Monitoring

- § 20.501 Modified to incorporate provisions of final oosimeter processor accreditation rule (adopted by NRC after publication of proposed Part 20 rule.
- \* § 20.502 Monitoring threshold for internal doses reduced from 30% of dose limit to 10% of limit. (Many commenters thought that monitoring should begin before a 1.5-rem dose could reached, e.g. would have exceeded dose limit for protection of the embryo-fetus) The 10% level was set in consideration that DAC-hours could be used to determine compliance rather than bioassay.

Removed counterproductive requirement that bioassay was required for any individual wearing a respirator. (It is required if the individual would be exposed to concentrations exceeding 10% of the Derived Air Concentrations and as part of a respiratory protection program.) PRGCS 60-61

Radiation Areas, High Radiation Areas and Very-High Radiation Areas

- § 20.601 (High Radiation Areas) added clarification that dose measure- rule ment applies at the "external surface of any package."
- \* § 20.602 rewritten to provide a general requirement for all licensees to restrict access to "very high radiation areas." (Nuclear power reactors have a technical specification (license condition) that differs from the wording of the proposed rule. This change defers to this technical specification for power reactors.)
- \* § 20.503 Detailed requirements for large irradiators reinstated 'rum' § 20.203g of present Part 20 but rewritten to improve clarity and organization. New § 20.1103 recordkeeping requirement added to conform with requirement to produce record.

ongeb

CHANGES FROM PROPOSED RULE

# SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

#### Respiratory Protection

§ 20.701 Not significantly changed.

- \* § 20.702 Modifies protection goal to limit intakes consistent with maintaining total effective dose equivalent ALARA rather than just minimizing internal exposure. (The previous text was noted in several comments as an example of unequal treatment of internal and external doses under old rule.)
- \* § 20.703 Requirement for having a respiratory protection program is now linked to use of respiratory protection equipment to protect workers rather than use of numerical protection factors from Appendix A. page 65 Written procedures are now required for monitoring, air sampling, and bioassay analyses.
- As suggested by comments, revision now permits changes in estimates intakes and dose in either direction (upwards or downwards) based upon bioassay results. (Proposed rule only permitted upward page 64 revisions.)
- \* § 20.703(c) Requirement reinstated from present rule that emergency breathing apparatus be NIOSH/MSHA certified.
- \* § 20.704(a) Requires that licensees subject to OSHA requirements cumply Deletax with OSHA requirements for respiratory protection (Many of which are peralready incorporated in Appendix A). This does not constitute a new Comm. When requirement as the licensee is already required (under OSHA regular -er tions) to conform. However, it saves the NRC staff the effort to Curtres' conduct rulemaking to place similar conditions in Part 20 (such as conditions for the use of respirators) and also gives the NRC the capability to cite against noncompliance with the OSHA requirements.

# SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

§ 20.801, § 20.802 Minor word changes to clarify meaning of "secure."

Posting Requirements

§20.901 [Unchanged]

- \$20.902 Removed use of "DANGER" from signs denoting "Radiation Areas." It was felt that this would induce complacency and detract from importance of warning for "high" and "very-high" radiation areas. Very-high radiation areas (greater than 500 rads/hour) warrant "GRAVE DANGER." Pages 67-68
- § 20.903 Present exemption from area posting for packages prepared for shipment and labeled in accord with DOT rules was deleted in proposed rule and continues to be deleted program. To topo
- § 20.905 Exemption from package labeling for packages labeled in accord with DOT requirements was reinstated from present rule. (Deleted in Proposed Rule.) page 71 outters
- \* § 20.906 (Picking up and receiving packages) Significantly changed to conform with revised DOT regulations and to conform with revised 10 CFR Part 71. PAGES 72-73
- Requirements for monitoring incoming packages now restricted to:

   (1) packages expressly labeled externally as containing radio
  - active materials; or
  - (2) packages showing obvious damage or that are leaking.

page 72

SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

Waste Disposal Requirements

- § 20.1001 Minor change requires licensing of persons receiving wastes from others for decay in storage. pages 73 bottom, 74
- § 20.1002 No major changes.
- § 20.1003 Capability to dispose of "readily dispersible biological material" is restored to allow research laboratories that dispose of test animals by grinding them up followed by disposal into sanitary sewer systems. Proposed rule restricted sewer disposal to soluble materials; present rule permits both "soluble and readily dispersible materials" to be disposed of in sewers. Restriction was imposed to eliminate contamination of sewage treatment plants by metallic radionuclides (e.g., americium-241).

Higher concentration limits for sewer disposal than for other liquid effluents are based upon anticipated dilution by other users of the sewer system (non-radioactive effluents). There has been a long-stand-3rd) ing NRC staff position that such disposal does not consitute "disposal into a sanitary sewer system" because disposal into a rafe sanitary sewer system operated for the sole use of the licensee does not provide such dilution. The new definition of a "sanitary sewer system" does not encompass such licensee-owned systems and also

§ 20.1004, § 20.1005, and § 20.1006 are relatively unchanged from proposed rule.

SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

Recordkeeping

General change to a record re ention period of 3 years to conform to NRC record retention policy.

\* § 20.1101 Requires licensee to use "special units" (rad, rem, and 10 Age 79 curies).

Licensee must also clearly designate what type of dose is being used, e.g., effective dose equivalent, organ dose equivalent, shallow dose rule equivalent.

\* § 20.1102 Major change to reflect emphasis on Radiation Protection Programs in § 20.101. Place ALARA recordkeeping in perspective and greatly reduces paperwork burden that would have resulted from the PAges 29 60tom proposed rule.

§ 20.1103 no major changes.

§ 20.1104 Eliminated proposed rule requirement for maintaining dosimetry # 83-84 processor accreditation certificate (except for in-house processing) Comments requested change and it conforms to final accreditation vule

rule.

§ 20.1105 no major changes.

§ 20.1106 Split out format of records from requirement to keep record.

Clarified use of mother's records and social security account number for embryo/fetus.

§ 20.1107 Expanded recordkeeping requirement from "effluents" to include all data related to assessment of doses to members of the public start

§ 20.1108 no major changes.

- § 20.1109 New section contains requirements for records of entry poor 86 interlock tests required by § 20.603.
- \* § 20.1110 New section contains requirements on the form of records. Now permits use of "electronic media" providing that the method used is capable of producing a clear legible record and that means are provided for prevention of loss or alteration. proge 86

CHANGES FROM PROPOSED RULE

ENCLOSURE 6

Minor

SECTION BY SECTION SUMMARY OF CHANGES FROM PROPOSED RULE

Reporting Requirements

- § 20.1201 Clarified 30-day written, report is 30 days after telephone report. page 88 2mo 2 3rd 96
- § 20.1202 Reinstated monetary loss and lost time criteria for incident notification (in present rule; removed in proposed rule).

Changed NRC organization to be notified from Regional Office to NRC Operations Center.

- § 20.1203 Separated criteria for reportable events from requirements for contents of reports.
- \$ 20.1204 Extended reporting period of a planned special exposure from 15 to 30 days. page 44 fop
- § 20.1205 [Reserved] (Was on reports of exceeding the "Reference Level" for members of the public. A requirement for reports of exceeding dose limits for members of the public is in § 20.1203.) deletion page 5/ 60 How Q
- \* § 20.1206 (combined with § 20.1207) Eliminates termination and annual statistical reports in favor of individual dose reports from those licensees that were required (old § 20.408) to submit the termination and statistical reports.

Exceptions §§ 20.1301, 20.1302 No major changes.

Enforcement § 20.1401 Minor editorial changes.

CHANGES FROM PROPOSED RULE 9

ENCLOSURE 6

RuR

OGC

#### NUCLEAR REGULATORY COMMISSION

10 CFR PART 20 and 10 CFR Parts 2, 19, 31, 32, 34, 35, 39, 40, 50, and 61

#### RIN 3150 - AA38

Standards for Protection Against Radiation

AGENCY: Nuclear Regulatory Commission.

ACTION: Final rule.

SUMMARY: The Nuclear Regulatory Commission (NRC) is revising its standards for protection against ionizing radiation. This action is necessary to incorporate updated scientific information and to reflect changes in the basic philosophy of radiation protection. The revision conforms the Commission's regulations to the Presidential Radiation Protection Guidance to Federal Agencies for Occupational Exposure and to recommendations of national and international radiation protection organizations.  $\int$ 

EFFECTIVE DATE: This regulation becomes effective January 2, 1991. However, licensees may delay implementation of this rule until January 1, 1993. (See § 20.8 of the rule and Part V., "Implementation," of the preamble.)

ADDRESS: Copies of documents relating to the January 9, 1986 proposed rule (51 FR 1092) or this document may be examined and copied for a fee in the Commission's Public Document Room at 2120 L Street NW (Lower-Level), Washington, DC 20555.  $\Box$ 

FOR FURTHER INFORMATION CONTACT: Harold T. Peterson, Jr., Division of Regulatory Applications, Office of Nuclear Regulatory Research, U.S. Nuclear Regulatory Commission, Washington, D.C. 20555, Telephone (301)492-3640.

SUPPLEMENTARY INFORMATION:

#### I. Introduction

A. Purpose of the Revision

The purpose of this revision of 10 CFR Part 20 is to modify the NRC's radiation protection standards to reflect developments in the principles and scientific knowledge underlying radiation protection that have occurred since Part 20 was originally issued more than 30 years ago. These developments not only include updated scientific information on radiunuclide uptake and metabolism, but also reflect changes in the basic philosophy of radiation protection. Incorporation of these changes will ensure that Part 20 continues to provide adequate protection of public health and safety.

It is also the purpose of this revision to implement the 1987 Presidential guidance on occupational radiation exposure (see Section II.D). The <u>Atomic</u> <u>Energy Commission (AEC)</u> and the NRC have followed past Federal radiation protection guidance, and conformance with the guidance is viewed by the Commission as being necessary to ensure that NRC licensees are using levels of protection comparable to those used by Federal agencies.

The AEC and the NRC have generally followed the basic radiation protection recommendations of the International Commission on Radiological Protection (ICRP) and its U.S. counterpart, the National Council on Radiation Protection and Measurements (NCRP), in formulating basic radiation protection standards. In 1977, ICRP issued revised recommendations for a system of radiation dose limitation. This system, which was described in ICRP Publication 26,<sup>1</sup> introduced a number of significant modifications to calling concepts and recommendations of the ICRP and the NCRP that are now being incorporated in the NRC regulations. In particular, this revision of Part 20 puts into practice recom

1 Recommendations of the International Commission on Radiological Protection, January 13, 1977, ICRP Publication No. 26, (1977). (Available for sale from Pergamon Press, Elmsford, NY 10523.)

mendations from ICRP Publication 26 and subsequent ICRP publications. The Federal radiation protection guidance signed by the President on January 20, 1987, is also based upon the ICRP 1977 recommendations in ICRP Publication 26.

In adopting the basic tenets of the ICRP system of dose limitation, the Nuclear Regulatory Commission <u>recognizes</u> that, when application of the dose limits is combined with the principle of keeping all radiation exposures "as low as is reasonably achievable," the degree of protection could be significantly greater than from relying upon the dose limits alone.

B. Fundamental Radiation Protection Principles

The radiation protection standards in this part are based upon the assumptions that--

 Within the range of exposure conditions usually encountered in radiation work, there is a linear relationship, without threshold, between dose and probability of stochastic health effects (such as latent cancer and genetic effects) occurring;

(2) The severity of each type of stochastic health effect is independent of dose; and

(3) Nonstochastic (nonrandom) radiation-induced health effects can be prevented by limiting exposures so that doses are below the thresholds for their induction.

The first assumption, the linear nonthreshold dose-effect relationship, implies that the potential health risk is proportional to the dose received and that there is an incremental health risk associated with even very small doses, even radiation doses much smaller than doses received from naturally occurring radiation sources. These health risks, such as cancer, are termed stochastic because they are statistical in nature; i.e., for a given level of dose, not every person exposed would exhibit the effect. The second assumption means that when a stochastic effect is induced, the severity of the effect is not related to the radiation dose received. The third assumption implies that there are effects, termed nonstochastic effects, for which there is an apparent threshold; i.e., a dose level below which the effect is unlikely to occur. An

3

KCM

example of a nonstochastic effect is the formation of radiation-induced cataracts of the eyes.

The above assumptions are necessary because it is generally impossible to determine whether or not there are any increases in the incidence of <u>disease</u> at very low doses and low dose rates, particularly in the range of doses to members of the general public resulting from NRC-licensed activities. It is firmly established, both from animal studies and human epidemiological studies (such as those of the radium dial painters, radiologists, and the atomic bomb survivors) that there is an increased incidence of certain cancers associated with radiation exposure at high doses and high dose rates. However, whether these effects occur at very low doses and, if they occur, whether their occurrence is linearly proportional to dose are not firmly established. <u>This creates considerable uncertainty in the magnitude of the risk at low doses and low dose rates.</u> There is no clear human evidence of radiation-induced genetic damage to the children of irradiated parents. Such effects are inferred from studies of mice and nonmammalian species (e.g., fruit flies).

In the absence of convincing evidence that there is a dose threshold or that low levels of radiation are beneficial, the Commission believes that the assumptions regarding a linear non threshold dose-effect model for cancers and genetic effects and the existence of thresholds only for certain nonstochastic effects remain appropriate for formulating radiation protection standards and planning radiation protection programs.

C. Background

Standards for radiation protection were originally issued by the former  $\Box R_{o}M_{H}$ AEC in the late 1950s (22 FR 548, January 29, 1957) and republished in 1960. These standards have been modified since that time by a series of amendments relating to specific issues., however, no complete revision of Part 20 has been made since the original standards were issued.

Federal Register of March 20, 1980 (45 FR 18023). This ANPRM requested

odit

Ston the

REVS

comments on possible topics that should be revised in a proposed revision of Part 20. The responses received to this announcement were considered in the formulation of the proposed revision.

During the development of this rule, early comments from licensees, labor unions, public interest groups, other Federal agencies, and scientific organizations were solicited, discussed, and considered in formulating the proposed rule. In addition, the NRC staff has benefited from its participation in several public meetings held by the Environmental Protection Agency (EPA) in connection with the guidance for occupational radiation exposure. The revised Part 20 and the Federal guidance on occupational exposure were developed in parallel and are both based primarily on the ICRP recommendations. The comments made in these EPA-sponsored meetings and those received by EPA on the draft guidance published by EPA in the January 23, 1981 Federal Register (46 FR 7836) were reviewed by the NRC staff and considered in preparing the proposed Part 20.

The NRC published the proposed revision of the 10 CFR Part 20 rule in the January 9, 1986 Federal Register (51 FR 1092). More than 800 sets of public comments were received on the proposed revision. The public comments on the proposed revision were categorized, analyzed, and taken into account in developing the final rule. The principal public comments and the NRC staff responses to them are discussed in Section VI.

## II. Developments Since the Proposed Revision Was Issued

A. ICRP 1985 Paris Meeting

In March 1985, the International Commission on Radiological Protection (ICRP) held a meeting in Paris, France, to review the work of the various ICRP task groups and committees. One of the outcomes of this meeting was an ICRP

statement<sup>2</sup> that the ICRP intended the principal dose limit for members of the general public to be 1 millisievert (100 millirems) in a year, rather than 5 millisieverts (500 millirems). This clarification has been taken into account for the limits adopted for members of the public in the final rule and is discussed more fully in the discussion on § 20.301.

A second recommendation of the ICRP made at that time concerned the appropriate quality factor for converting the absorbed dose from neutrons (in rads or grays) to a dose equivalent (in rems or sieverts). The ICRP statement recommended increasing the quality factor for high-energy neutrons by a factor of 2. The quality factor for fast neutrons, for example, would be increased from 10 to 20. This change has the effect of doubling the apparent biological effectiveness of high-energy neutrons. For reasons explained in the discussion of quality factors (see the discussion of § 20.4), the NRC has not adopted this recommendation in this revision of Part 20.

# B. ICRP 1987 Washington Meeting

The primary focus of the statement issued by the ICRP following the 1987 meeting in Washington<sup>3</sup> was ICRP Publication No. 48.<sup>4</sup> That publication discussed higher transfer factors for transport of certain transuranic elements across the intestinal walls. These higher fractional absorption factors have been incorporated in revisions to the <u>annual limits on intake</u> (ALIs) and <u>derived air</u>. <u>concentrations</u> (DACs) in Appendix B of the final rule. The changes resulting from the use of these revised factors would not change either the ingestion or inhalation ALIs for plutonium in the oxide or nitrate forms, but would lower

Romu

- International Commission on Radiological Protection, "Statement from the 1985 Paris Meeting of the [ICRP]," British Journal of Radiology, Vol. 58, page 910: 1985, also Health Physics, 48(6): 828-829 (June 1985)
- 3 International Commission on Radiological Protection, "ICRP Statement from 1987 Washington Meeting," Health Physics 53(3): 335-342 (1987).
- 4 International Commission on Radiological Protection, "The Metabolism of Plutonium and Related Elements," ICRP Publication No. 48, (<u>Available for</u> sale from Pergamon Press, Elmsford, NY 10523.) (1986).

KCMU p7

the ALIs for other compounds or mixtures by a factor of 10. The transfer factor for the gut transfer of neptunium was found to be an order of magnitude lower than the value used in ICRP-30 and, consequently, the ingestion ALI can be increased by almost an order of magnitude. The transfer factors for americium, curium, and californium were found to be a factor of 2 higher than the ICRP-30 value so the ingestion ALIs are reduced by a factor of 2. Parameters applicable to inhalation ALIs and DACs are less affected than the ingestion ALIs as the transfer from the gastrointestinal (GI) tract to the blood for these radionuclides generally is less significant than transfer from the lung to the blood.

C. ICRP 1987 Como Meeting

Following its 1987 meeting in Como, Italy, the ICRP issued a statement<sup>5</sup> that reviewed the existing estimates of the biological risks of ionizing radiation and, in particular, the preliminary data from the reanalysis of the Hiroshima-Nagasaki atomic bomb followup studies. Reanalysis of these data indicated that the risks from gamma radiation are approximatel, a factor of 2 higher than previous estimates for the general population and are also higher, but by a smaller factor, for workers. The ICRP concluded in 1987 that this information alone was "not considered sufficient at that time to warrant a change in the dose limits for occupational exposure and, for the general population, the increase in risk indicated by the new data is not considered to require an immediate change in the recommended dose limits, following the reduction by the ICRP (in 1985) in the principal limit from 5 to 1 mSv in a year (from sources other than medical and natural background radiation)." The ICRP also noted that the potential higher risks indicated by the reanalysis of the atomic bomb data should not be a major consideration as the dose limits should not be of primary importance in controlling doses if the principle of keeping radiation exposures "as low as is reasonably achievable" is being practiced. This position has since been modified by the ICRP 1990 Statement (see Section I below).

5 International Commission on Radiological Protection, "Statement from the 1987 Como Meeting of the [ICRP]," Health Physics, 54(1): 125-132 (1988).

# D. Federal Radiation Protection Guidance on Occupational Exposure

On January 20, 1987, President Reagan approved revised guidance to Federal agencies for occupational radiation protection. This guidance, which was published in the Federal Register (52 FR 2822; January 27, 1987), generally adopts the philosophy and methodology of ICRP Publications 26 and 30. The Part 20 revision was developed in parallel with the development of the guidance. Because of this parallel development, the proposed Part 20 rule conformed with the draft Federal guidance available at the time the proposed Part 20 rule was written. However, because of changes made to both the draft guidance and the draft Part 20 revision, there were a few differences between the guidance in its final published form and the proposed Part 20 revision. As discussed in the respective sections below, changes to the proposed rule have been made in order to implement the final version of the Federal guidance.

E. NCRP Report No. 91

On June 1, 1987, the National Council on Radiation Protection and Measurements (NCRP) issued a report<sup>6</sup> containing updated NCRP recommendations for radiation protection limits. These recommendations replace recommendations published in 1971. The majority of these recommendations are in accord with the 1977 recommendations of the ICRP and, consequently, were already reflected in the proposed Part 20 rule. There are, however, several NCRP recommendations that were not in the ICRP-26 recommendations. These NCRP recommendations are:

(1) A general "guideline" that the cumulative effective dose equivalent to a worker should not exceed 1 times the worker's age in years; i.e.,  $1 \times N$  instead of the former 5(N - 18) formula;

6 National Council on Radiation Protection and Measurements (NCRP), "Recommendations on Limits for Exposure to Ionizing Radiation," NCRP Report No. 91 (June 1, 1987). (Available for sale from the NCRP, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)

(2) Use of committed effective dose equivalent for planning purposes and the use of annual (rather than committed) doses for post-(internal) exposure control;

(3) A monthly dose limit as well as a limit on total gestation dose to the embryo/fetus;

(4) Adoption of a 0.1-rem (1 mSv) effective dose equivalent limit for exposure of the general public with the condition that the "site operator" assess the total exposure to the most exposed individual if estimated or measured exposures exceed 25 percent of this limit (2f millirems or 0.25 mSv per year):

(5) The use of "reference levels" set up by the radiation user below the regulatory limit;;

(6) A Negligible Individual Ris: Level of 1 millirem (0.01 mSv) per year. This level is the "...average annual excess risk of fatal health effects attributable to irradiation, below which further effort to reduce radiation exposure to the individual is unwarranted" (NCRP No.91, p. 43).

These NCRP recommendations were issued after publication of the proposed Part 20 rule and, consequently, there has not been an opportunity for public comment on them. For this reason, these NCRP recommendations are not being adopted in the revised Part 20 rule at this time.

F. The 1988 Report of the United Nations Scientific Committee on the Effects of Atomic Radiation (UNSCEAR-88).

The United Nations Scientific Committee on the Effects of Atomic Radiation has analyzed data on the sources and effects of atomic radiation and published a series of reports containing summaries of the sources of radiation, the doses received by workers and members of the general public from these sources, and an analysis of the potential health risks from exposure to ionizing radiation. 237 #8 5RM1 #7

The latest report in this series is the 1988 report. The 1988 report<sup>7</sup> contains more recent information on the health risks of ionizing radiation determined from a reevaluation of the Jata on the survivors of the Hiroshima-Nagasaki atomic bombings. Sased upon these data, the radiation risk at high doses and high dose rates is estimated to be 7.1 x 10"4 fatal health effects per rad (0.071 effects per gray). For estimating the risk from radiation doses below 100 rads, the UNSCEAR report recommended that a dose rate reduction factor be applied to account for the reduced effectiveness of lower doses and lower dose rates. This would lead to an estimated risk of fatality of between (0.7 to 3.5) x  $10^{-4}$  health effects per rad for low doses such as those encountered in routine occupational exposure and the even lower doses that might be received by members of the general public from NRC- (or Agreement State) licensed activities. The fatal cancer risk value associated with the 1977 ICRP recommendations, 1 is 1.25 x 10<sup>-4</sup> (the proposed Part 20 rule, 51 FR 1102, January 9, 1986) so that the risks as estimated by the 1988 UNSCEAR report for low doses are between 0.6 to 2.8 times higher than the earlier ICRP estimate. The implications of the increased risk are discussed in Section II.I.

237 #8 SRM2 #7

G. The 1988 Report of the National Academy of Sciences' Committee on the Biological Effects of Ionizing Radiation (BEIR-IV)<sup>8</sup>

The 1988 BEIR-IV report supplements the 1980 BEIR-III report by providing a more detailed analysis of the risks from internal alpha-emitting radionuclides to complement the emphasis of the BEIR-III report on gamma and beta radiation. Revised risk estimates are given for intakes of radon, radium, polonium, thorium, uranium, and higher transuranic elements (e.g., plutonium).

 7 United Nations Scientific Committee on the Effects of Ionizing Radiation (UNSCEAR), "Sources, Effects and Risks of Ionizing Radiation, 1988 Report to the General Assembly, Sales Section, United Nations, NY 10017 (1988)
 8 National Academy of Sciences-National Research Council, Committee on the Biological Effects of Ionizing Radiation "Health Risks of Radon and Other Internally Deposited Alpha-Emitters, (BEIR-IV)," National Research Council, National Academy Press, Washington, DC 20418 (1988).

#8 SRM #7

The radionuclide given the greatest emphasis in the BEIR-IV report is radon (radon-222), the gaseous decay product of radium-2°6. The radon dose conversion factor in the BEIR-IV report for exposure conditions representative of those of the general public is consistent with the value used to derive the airborne effluent concentration limit for radon-222 in Appendix B, Table 2 of the revised 10 CFR Part 20.

H. The 1990 Report of the National Academy of Sciences' Committee on the Biological Effects of Ionizing Radiation (BEIR-V)<sup>9</sup>

The BEIR-V report is another comprehensive reevaluation of the health risks of radiation exposure based upon the revised dose estimates for the survivors of the atomic bombings of Hiroshima and Nagasaki. The BEIR-V report gives risk estimates for leukemia and non-leukemia (solid cancers) that are about two to five times higher than the estimates in the 1980 BEIR-III report. The BEIR-V report gives the following factors as the principal reasons for this increase: (1) use of different dose-response and risk projection models, (2) revised estimates of the doses to the individual survivors of the atomic bombings in Japan, and (3) improved epidemiological data from additional years of followup studies since the BEIR-III was completed in 1980.

The BEIR-V Committee uses the linear dose response model and the relative risk projection model to extrapolate the fatal tumor risk to future periods. The relative risk projection model assumes the risk to be proportional to the natural cancer incidence, which generally increases with age. Because of this dependence on age, the relative risk model generally predicts higher future (lifetime) risks than the absolute risk model which employs a constant added risk per year with increasing age. Estimates are given of the risk as a func-

9 National Academy of Sciences-National Research Council, Committee on the Biological Effects of Ionizing Radiation, "Health Effects of Exposure to Low Levels of Ionizing Radiation, (BEIR-V)," National Research Council, National Academy Press, Washington, DC 20418 (1990).

tion of the time since the exposure occurred and the age and sex of the exposed person. The BEIR-V report, like the UNSCEAR-88 report, indicates that a reduction factor should be applied to the risk estimates derived from nigh doses and dose rates in order to apply them to low dose and low dose-rate situations. Although neither the BEIR-V report nor the UNSCEAR-88 report recommends a specific value for this factor, both reports indicate that this factor should be greater than 2, (larger reduction factors would give a lower risk per unit dose). Assuming a factor of 2 reduction in the risk estimates derived from high doses and high dose rates, BEIR-V would give a lifetime risk of a radiation-induced cancer fatality of about 4 x 10<sup>-4</sup> fatal cancers/rem (0.04 per sievert) for workers and 5 x 10"4 per rem (0.05 per sievert) for the general population, the higher value for the public being associated with the higher sensitivity and the longer period of elevated risk associated with the younger ages present in the general population. The value of 5 x  $10^{-4}$  is three times as large as the recommended value in the 1980 BEIR-III report and four times as large as the estimate in the 1977 ICRP-261 report (see Section II.F).

The BEIR-V report also summarized the data on the frequency of severe mental retardation found in the children of Hiroshima and Nagasaki atomic bomb survivors. These children were exposed in utero at gestational ages of 8-15 weeks and the risk of severe mental retardation during this period is about 4 x  $10^{-3}$  per rem with less of a risk at other gestational ages.

The estimates of genetic effects to the offspring of irradiated individuals remained similar to those in the 1972 BEIR-I and 1980 BEIR-III reports. As radiation-induced inherited abnormalities have not been observed directly in humans, estimates of genetic effects have been based primarily upon experimental studies with mice. These studies suggest that it would take a dose of about 100 rads to double the natural frequency of genetically transmitted diseases.

I. ICRP 1990 Recommendations

On June 22, 1990, the International Commission on Radiological Protection issued a press release indicating that it would issue revised recommendations

to carnents on SECY-91 237

SRME

#8 Soe man

for radiation protection based upon the newer studies of radiation risks (such as those described in Sections F, G, and H above). The press release indicated that the ICRP would recommend a reduction in the occupational dose limit from an equivalent of 5 rems p : year to an average o' rems per year with some allowance for year-to-year flexibility. The ICRP uose limit for long-term exposure of members of the general public would remain equivalent to the level adopted in this revision of Part 20, 0.1 rem per year.

The Nuclear Regulatory Commission does not believe that additional reductions in the dose limits are urgently required by the latest radiation risk estimates. Few individuals in either the work force or in the general public are exposed at or near the limits, and most of these will not be exposed at such levels over long periods of time. Due to the practice of ALARA ("as low as is reasonably achievable"), the average radiation dose to occupationally exposed individuals is well below the limits in either the existing or revised Part 20 and also below the changes being considered by the ICRP. For example, in 1987 about 97 per cent of the workers in nuclear power plants, industrial radiography, reactor fuel fabrication, and radioisotope manufacturing, four of the industries having the highest potential for occupational radiation exposures, were below an annual dose of 2 rems so that an immediate reduction in the occupational dose limits would result in only a small reduction in the population dose and in the potential health impact. Although the risk per unit dose is higher than previously thought, individual annual exposures averaged over a lifetime in the highest exposed groups in the working population appear to be about 2-3 rems per year (50-60 % of the 5-rem annual limit). Therefore, a factor of 2 increase in the risk per unit dose would result in estimated potential risks associated with actual lifetime exposures that are comparable to the previous risk estimate applied to an assumed lifetime exposure of 5 rems per year.

As a result of the application of the ALARA philosophy to effluent release standards in Appendix I to 10 CFR Part 50 for nuclear power reactors and EPA's 40 CFR Part 190 for the uranium fuel cycle, doses from radioactive effluents from fuel cycle facilities are already much less than the 0.1 rem per year standard in the revised Part 20. The 0.1 rem per year remains as the level recommended by the ICRP for protection of the general public. #8

Rovs

237 #8 SRM1

#8

Until the final ICRP recommendations are published, and the need for further revisions in NRC standards established, the Commission believes it would be advisable to proceed with the promulgation of the proposed dose limits, rather than deferring the dose reductions that are already associated with the revised Part 20 rule. The Commission will carefully review the final recommendations of the International Commission on Radiological Protection, the comments of the scientific community and others on these recommendations, and the ICRP response to these comments. In addition, the Commission staff will review the recommendations of other expert bodies, such as the National Council on Radiation Protection and Measurements, and participate in the deliberations of the U.S. Committee on Radiation Research and Policy Coordination and any interagency task force convened by the Environmental Protection Agency to consider revised Federal radiation guidance. Any future reductions in the dose limits by the Commission would be the subject of a future rulemaking proceeding.

### III. Issues Being Resolved Separately

As noted in the above discussion. there are several areas where the Commission believes a better scientific consensus is needed before adopting values different from those in the present Part 20. There are also several areas where issues raised in the public comments (see Section V) are being resolved in other NRC rulemaking proceedings because of either their scope, complexity, or timing. The following issues are being or will be resolved in other NRC rulemaking proceedings:

(1) Establishment of "Below Regulatory Concern (BRC)" levels (related to <u>de minimis</u> levels and a negligible level of risk). On June 27, 1990, the Commission announced the issuance of a policy statement on Below Regulatory Concern, which was subsequently published in the Federal Register on July 3, 1990 (55 FR 27522). This policy statement establishes the framework for the Commission to formulate rules and licensing decisions to exempt certain practices involving small quantities of radioactive materials from some or all regulatory controls. The BRC policy statement sets forth criteria for protection of both individuals (individual dose criteria) and population groups (a collective dose criterion).

23?4 SRM2 #8

Itg

(2) Limits for decommissioning of nuclear facilities and for residual radioactive contamination. This is being actively pursued by both the NRC staff, which is developing criteria for residual contamination of soils and structures, which is one aspect of the implementation of the Below Regulatory Concern Policy, and by NRC staff participation on an EPA Interagency Task Force on Residual Radioactivity.

(3) Limits and calculational procedures for dealing with the "hot particle" issue (small particles found in nuclear reactors that, because of their high activity and small size, produce high localized doses to skin). The NRC notes that the National Council on Radiation Protection and Measurements (NCRP) has recently issued new recommendations regarding "hot particles" in NCRP Report No. 106, "Limit for Exposure to 'Hot Particles' On the Skin," December 31, 1989. A modified NRC enforcement policy statement with regard to the "hot particle issue" was published in the July 51, 1990 Federal Register (55 FR 31113). The NCRP report, together with a forthcoming ICRP report on the biological effects of skin irradiation and other technical analyses will be considered in a future rulemaking to set limits for skin irradiation.

(4) Modification of NRC incident notification requirements. A modification of the incident notification requirements was issued for public comment on May 14, 1990 (55 FR 19890). If this proposal is adopted as a final rule, it would modify both the existing Part 20 and this revision.

(5) Publicatio a separate rule for large irradiators. A new Part 36 is being proposed \_\_ublic comment. The detailed requirements for irradiators presently in the revised Part 20 (§ 20.603) will eventually be deleted and replaced by the provisions incorporated in the new Part 36.

There are also additional areas where the scientific basis is not yet resolved sufficiently to justify a change from current practice. These two areas require better scientific consensus on the appropriate position: (1) The need for and impact of a lifetime cumulative dose limit of 1 rem per year of age and (2) quality factors, especially for neutrons, low-energy beta-emitters, and high-energy gamma photons. These issues will be reconsidered as consensus positions are reached by the scientific community.

# IV. Need for Additional Regulatory Guidance

The Commission recognizes that the incorporation of many new concepts into Part 20 will require additional guidance and explanation on their application to practical problems in radiation protection. The Commission also notes the desirability of having such additional guidance available at the same time that the final rule is issued in offective form. However, it was impractical, both for reasons of scheduling and availability of resources, for these guides to be developed concurrently with Part 20. Some of the regulatory guides being developed or revised to assist in the implementation of the revised Part 20 are: up dated titles list

- (1) Content of Radiation Protection Programs at Nuclear Power Plants;
- (2) Interpretation of Bioassay Measurements (Draft Regulatory Guide 8.9, Revision 1),
- (3) Criteria and Procedures for Summation of Internal and External Occupational Doses,
- (4) Acceptable Criteria for Planned Special Exposures and for Satisfying Documentation Requirements;
- (5) Methods and Parameters for Calculating the Dose to the Embryo/Fetus:
- (6) Instructions for Recording and Reporting Occupational Radiation Exposures (includes NRC Forms 4 and 5).

The Commission has instructed the staff to have these and other draft guides published for public comment early in 1991 so that there will be ample time to incorporate public comments into the final guides prior to the effective date of the revised Part 20 rule which is January 1, 1993.

SRM: #3

SRM:

#### V. Implementation and Existing License Conditions

Section 20.8 of the rule provides that NRC licensees must implement the Part 20 rule on or before January 1, 1993. Licensees that adopt the provisions of this rule prior to the required implementation date are required to notify the NRC. Early implementation may benefit applicants for new licenses or license renewals as they could avoid having to adopt and implement one version of Part 20 for only a short period of time prior to the required implementation date of this revision. Licensees choosing early implementation must adopt the entire revised Part 20. Compliance will be required with the version of 10 CFR Part 20 codified in the Code of Federal Regulations on January 1, 1991 until January 1, 1993 or until the licensee notifies the Commission of early implementation of the revised Part 20.

License conditions and reactor technical specifications may contain citations to portions of the existing 10 CFR Part 20. After adoption of the revised Part 20 by the licensee or after January 1, 1993, the applicable section of the revised Part 20 that corresponds to the same topic should be used in lieu of any section of the Part 20 in effect on or before January 1, 1991 that is cited in the technical specifications or license conditions. When there is no corresponding section in the revised Part 20 to these cited provisions, the current license condition based on the Part 20 in effect prior to January 2, 1991 shall remain in force until there is a technical specification change or license amendment or renewal. If a license condition or technical specification exempted a licensee from a provision of Part 20, it will be assumed to also exempt the licensee from the applicable provision of the revised Part 20. If the license condition or technical specification is more restrictive than the revised Part 20, it shall remain in force until it is modified by a technical specification change or license amendment or renewal.

The NRC will issue a regulatory guide that provides the section and paragraph identifiers in the revised Part 20 and the corresponding sections or paragraphs in the earlier Part 20. This document will issued shortly after the publication of this rule and will enable licensees to locate sections of the revised Part 20 that correspond to sections of the earlier Part 20 cited in license conditions and technical specifications.

ED

VI. Summary of Public Comments and Changes from Proposed Rule

The purpose of this section is to respond to comments raised on the proposed rule and to explain and highlight the changes made to the proposed rule. This section presents, for each paragraph or section of the rule, the principal public comments on the proposed rule, an NRC staff response to the comments (where appropriate), and a summary of the principal changes that were made to the proposed rule. This section has been arranged so that it corresponds to the structure of the rule. Although it follows the format of the final rule, the following text is not intended to create any additional requirement not already in the regulatory text.  $\sum \sum_{i=1}^{n} e^{-i\pi i \pi i x_i} e^{-i\pi i \pi i x_i} e^{-i\pi i \pi i x_i} e^{-i\pi i x_i} e^{-i\pi i x_i}$ 

Subpart A -- General Provisions

Section 20.1 Purpose.

Final Rule. A new sentence was added to convey the intent of the former § 20.9 in the proposed rule (which has been removed) that the regulations in Part 20 should not hinder a licensee's actions to protect health and safety in the event of an emergency. It is the Commission's intent that the regulations be observed to the extent practicable during emergencies, but that conformance with the regulations should not hinder any actions that are necessary to protect public health and safety such as lifesaving or maintaining confinement of radioactive materials.

In this regard, the Commission notes that the Federal guidance on occupational radiation protection states that those dose standards only apply to normal operating conditions. The Commission believes that the dose limits for normal operation should remain the primary guidelines in emergencies. However, the Commission also recognizes that, in an emergency, operations that do not conform to the regulations may have to be carried out to achieve the high-priority tasks of worker, public, and facility protection. The purpose of the addition to this section is to assure licensees that their first priority should be to carry out those actions that are necessary to protect workers and the public from radiation exposure, to perform lifesaving activities, to prevent or limit the spread of radioactive contamination or the release of radioactive materials to the environment, and to preserve an adequate margin of safety. In evaluating KCAL any ensuing violations and their severity, the Commission will consider on a case-by-case basis any extenuating circumstances.

### Section 20.2 Scope.

Final Rule. The statement of scope remains essentially the same as in the proposed rule except that "background radiation" has replaced "natural background." This change was made to include residual global fallout and ambient radon levels within the definition of "background."

### Section 20.3 Definitions.

General. Because of the large number of comments that dealt primarily with wording changes or that questioned the need for or the use of a particular definition, the individual comments will not be discussed separately. However, these comments did result in substantial revisions to many of the definitions that appeared in the proposed rule. Those definitions that were added, modified, or deleted as a result of the public comments are listed below.

<u>Comment:</u> Differentiation among different kinds of dose equivalents. The potential for confusion among different dose equivalents was noted. Commenters noted that effective dose equivalents, committed effective dose equivalents, and doses to the lens of eye, skin, or extremities were all expressed in units of rems or sieverts and may be difficult to distinguish from one another.

Response: In the final rule the NRC staff has applied unique names for these previously undesignated quantities including: eye dose equivalent, shallow-dose equivalent (skin), shallow-dose equivalent (extremities), and total effective dose equivalent. The ICRP did not give these quantities specific names. The use of characteristic names is intended to reduce confusion in using these units. In this regard, it should be noted that the licensee is required to designate, in a clear and unambiguous manner, the quantities that are being recorded (see paragraph 20.1101(b)).

Final Rule. All the important definitions in the revised rule have been collected into one section, f20.3 Definitions. Unlike the proposed rule, which employed groups of related terms ("Area Terms," "Dose Terms," "Monitoring Terms," etc.), all the definitions in the final rule are listed in strict alphabetical order. This organization also avoids the presence of "local definitions" that appear only in a specific section of the regulation.

1. New Terms. The following definitions have been added to the final rule. These definitions have been added to clarify the meaning of the terms:

- a. "Activity"
- b. "Background radiation"
- c. "Derived air concentration-hours" ("DAC-hours")
- d. "Dosimetry processor"
- e. "Entrance or access point"
- f. "Generally applicable environmental standard"
- g. "Individual monitoring device"
- h. "Quality factor"
- i. "Sanitary sewerage"
- j. "Total effective dose equivalent (TEDE)."
- []

12 CMUN 16

SRM1 #6

2. Revised Definitions. The following definitions have been revised or modified from the definition used in the proposed rule:

- a. "Absorbed dose"
- b. "Annual limit on intake"
- c. "Class"
- d. "Committed dose equivalent"
- e. "Committed effective dose equivalent"
- f. "Derived air concentration"
- g. "Dose equivalent"

st8

- h. "Effective dose e uivalent"
- i. "Embryo/fetus"
- j. "Eye dose equivalent"
- k. "Member of the public"
  - "Nonstochastic"
- m. "Person"
- n. "Planned special exposure"
- o. "Quarter"
- p. "Survey"
- q. "Weighting factor"
- r. "Working level"
- s. "Year"

Noturn background SRM2

3. Definitions and terms deleted. Two definitions were deleted because the terms no longer appear in the rule: "Collective effective dose equivalent" and "Roentgen." "Natura! background" has been replaced by "Background radiation." JPP 12

# Section 20.4 Units of Radiation Dose.

Comment: Choice of the system of units. Several commenters expressed a preference for retaining the older "special" units (the curie, rad, and rem) rather than allowing the use of the newer SI unit: Reasons cited for retaining the fact system included: present widespread use and licensee familiarity, potential for misunderstandings with the newer units, the need for worker retraining (particularly while learning the new ICRP system of dose limitation), and the costs associated with changing recordkeeping systems. A smaller number of comments favored changing over to the SI units: becquerels, grays, and sieverts.

Response: Although both the "special units" and the SI units appear in the text of Part 20 (to increase the familiarity of licensees with the SI units), the Commission has decided that adoption of the SI units at this time is not necessary. The Commission recognizes that the new terms and methodological approaches in the revised Part 20 are complex and that imposition of the SI

system of units on top of this complexity would further increase the potential for confusion. Consequently, at the present time, the recordkeeping, reporting, and notification requirements require the use of the "special units," the rad, the rem, and the curie. However, as the national move to metrication continues, as anticipated in Section 5164 of the Omnibus Trade and Competitiveness Act of 1988 (P.L. 100-418), at some later time there may be amendments to Part 20 that would require the use of SI units only (becquerels, grays, and sieverts).

Final Rule. The final Part 20 rule includes the International System of Units (SI units) for distance, area, and volume. The older "special units" are retained for activity (curie), absorbed dose (rad), and dose equivalent (rem`.

Comment: Quality factors for neutrons. The quality factor is the conversion factor between the absorbed dose (rads) and the dose equivalent (rems). Several publications<sup>2,3,4,7,8</sup> have recommended changes in neutron quality factors that are a factor of 2 higher than those in proposed Part 20. These changes would raise the quality factor for fast neutrons from 10 to 20.

Response: Increases in the quality factor for neutrons are suggested by some animal experimental data on the relative biological effectiveness (RBE) of neutrons. However, there appears to be considerable uncertainty as to whether the data actually demonstrate an increase in the hazard of neutrons. Because the RBE is defined as a ratio of doses to produce equivalent biological effects, it is not clear whether the apparent increase in the neutron RBE is due to the increased effectiveness of neutrons or whether it actually results from the decreased effectiveness of the reference gamma radiation at low doses.

<sup>7</sup> International Commission on Radiological Units and Measurements, "The Quality Factor in Radiation Protection " ICRU Report No. 40 (1986). (Available for sale from ICRU Publications, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)

<sup>8</sup> International Commission on Radiological Protection, "Data for Use in Protection Against External Radiation," ICRP Publication No. 51 (January 1988). (Available for sale from Pergamon Press, Elmsford, NY 10523.)
Final Rule. The NRC has decided not to revise the neutron quality factor at this time but to defer any change until there is greater scientific consensus on the most appropriate value. A major consideration underlying this decision is that neutron exposures at most NRC-licensed facilities are currently small and the potential increase of a factor of 2 would not have a major health or regulatory impact.

The decision to defer any change is consistent with recommendations of the Committee on Interagency Radiation Research and Policy Coordination (CIRRPC) of the Office of Science and Technology Policy that there should not be a revision of the value of the neutron quality factor at this time without more study. This position is also reflected in papers from the United Kingdom National Radiological Protection Board (UKNRPB)<sup>9</sup> and a statement on the neutron quality factor from the British Committee on Radiation Units and Measurements.<sup>10</sup>

Comment: Table of neutron quality factors. Several commenters questioned the accuracy and timeliness of the table of neutron quality factors and fluence rates (to give dose equivalents of 1 rem) that appeared in the proposed rule. Some commenters suggested that there were more appropriate tables published by the NCRP or ICRP.

Response: The tables in the proposed and revised rules were taken from NCRP Report No 38<sup>11</sup> and are appropriate for the neutron dose equivalent at a soft tissue depth of 1 centimeter (which is the depth specified for the deter-

Report. Pub Is

<sup>9</sup> J.A. Dennis, "The Relative Biological Effectiveness of Neutron Radiation and Its Implications for Quality Factor and Dose Limitation," Nuclear Energy 20(2). 133-149 (1987).

British Committee on Radiation Units and Measurements (BCRU), "Memorandum from the BCRU: Effective Quality Factor for Neutrons," Physics in Medicine and Biology 31 (7):797-799 (1986).

<sup>11</sup> National Council on Radiation Protection and Measurements, "Protection Against Neutron Radiation," NCRP Report No. 38 (January 1971). (Available for sale from the NCRP, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)

mination of the deep dose equivalent). There are newer tables from the ICRP, but these tables incorporate the factor of 2 increase in the neutron quality factor. (See the preceding discussion of the neutron quality factor.)

Subpart B -- Radiation Protection Programs

Section 20.101 Radiation Protection Programs ["As Low As Is Reasonably Achievable" (ALARA)] (f 20.102 of the Proposed Rule).

<u>Comment</u>: The concept of ALARA is a philosophical principle of radiation protection and, as such, it should not be made into a regulatory requirement. A primary objection to changing the status of ALARA from the hortatory suggestion in the current Part 20 ("licensees should") to a mandatory requirement ("licensees shall") is that there are no guidelines (except for light-waterreactor (LWR) effluents) as to what constitutes ALARA. Because of the subjective nature of an "ALARA level," there are problems in the retrospective evaluation of licensee performance by NRC inspectors and, at least in one case, interpretations by the courts concerning whether the levels achieved were truly "as low as is reasonably achievable."

Response: There were a number of comments that expressed similar concerns regarding the proposed implementation of "ALARA." The emphasic on ALARA actions has been revised from detailed requirements to document all ALARA actions to a requirement to have a radiation protection program that includes measures to keep doses and intakes "as low as is reasonably achievable." This shift is to emphasize that the ALARA concept is intended to be an operating principle rather than an absolute minimization of exposures.

Comment: Any requirement for ALARA should include a lower bound. Many commenters felt that there should be a "floor" for ALARA necessary.

Response: The Commission agrees that there would be advantages to estab-Tishing such a "floor," below which efforts to further reduce doses would not be necessary. An NRC policy statement on "Below Regulatory Concern" was announced on June 27, 1990, and published in the Federal Register on July 3, 1990 (55 FR 27522). The "Below Regulatory Concern" levels in that policy statement KCM4 delineate criteria below which additional licensee actions to further reduce p2; doses would not be required. Specific rulemaking actions will be carried out to define operational thresholds for particular classes of activities such as disposal of very low-level contaminated materials. The BRC policy statement provides a framework for evaluating these case-specific actions. (See also p2) discussion on § 20.304.)

Comment: Compliance with "ALARA-based" standards should constitute being ALARA. Several commenters supported the statement in the proposed Part 20 (§ 20.102(b)) that compliance with EPA's 40 CFR Part 190 and with Appendix I to 10 CFR Part 50 should constitute de facto compliance with the requirement to keep LWR effluents ALARA. Environmental Protection Agency (EPA) comments did not support this view.

Response: Appendix I to 10 CFR Part 50 defines ALARA levels of radioactive materials in LWR effluents. If the design objectives of Appendix I are met, it constitutes a demonstration that the effluents are ALARA and no additional effort is required to reduce the effluent levels. Although the EPA interprets 40 CFR Part 190 as an "ALARA-based" standard, it also believes that 40 CFR Part 190 constitutes an upper bound, not a lower bound, to ALARA efforts.<sup>12</sup> Consequently, compliance with 40 CFR Part 190 is not in itself sufficient to demonstrate that releases and doses are ALARA.

As Appendix I to Part 50 defines ALARA design objectives that constitute ALARA effluent levels, meeting these levels is sufficient to demonstrate ALARA

<sup>12</sup> Letter of January 7, 1986, from Sheldon Meyers, Director, Office of Radiation Programs, Office of Air and Radiation, U.S. Environmental Protection Agency, to Robert B. Minogue, Director, Office of Nuclear Regulatory Research, U.S. Nuclear Regulatory Commission. (This letter is reproduced as Enclosure B to the comments of the Environmental Protection Agency on the 10 CFR Part 20 revision (Docket PR-19, 20, 30 et al., 50 FR 51992, Comment # 769).)

effluent releases. In order for light-water reactors to demonstrate that doses from both effluents and direct radiation are ALARA, it is necessary to demonstrate that effluents meet the design objectives of Appendix I to 10 CFR Part 50, that direct radiation from onsite sources (gamma radiation from external radwaste tanks and turbine generators ("turbine shine")) is also ALARA, and that the total dose to any member of the public is within the numerical standards in 40 CFR Part 190. Meeting these conditions will constitute sufficient evidence that offsite doses from LWRs are ALARA and in conformance with both Appendix I and 40 CFR Part 190.

Comment: The NRC should establish "reference levels" in its rules. One commenter thought that the NRC should have "reference levels" for licensee action in Part 20.

Response: The Commission recognizes that licensees generally establish their own lower "reference levels" in order to keep from reaching and exceeding the Commission's formal dose limits. Based upon the public comments on the reference level for exposure of members of the public, which was in the proposed § 20.303, this approach would not be favored by a majority of licensees. Several commenters viewed the reference level for the dose to members of the public as being applied exactly as if it were a limit. Consequently, if the NRC were to specify generic reference levels for licensee action, the impact might be similar to lowering the magnitude of the dose limits. The Commission believes that the use of the ALARA philosophy is a preferable means to keep exposures well below the limits established by the Commission.

Final Rule. The final rule establishes a requirement for all licensees to have a radiation protection program that includes provisions for keeping radiation doses ALARA. It is expressly intended that the level of this program and efforts to document it are commensurate with the size of the licensed facility and the potential hazards from radiation exposure and the intake of radioactive materials. The requirement for a radiation protection program is not new; it was discussed in the proposed rule (under ALARA) and is consistent with requirements in Part 33 (§§ 33.13, 33.14, and 33.15), Part 34 (§ 34.11), Part 35 (§§ 35.20-35.31), and Part 40 (§ 40.32) of the NRC regulations, with the information requested in Chapter 12 of Regulatory Guide 1.70, "Standard Format and Content of Safety Analysis Reports for Nuclear Power Plants," and with the conditions in most licenses issued by the Commission. The extent of this program and requirements for written records and procedures for operating the program are intended to be commensurate with the scope and potential hazards associated with the licensee's activities. The Commission recognizes the need to provide guidance on the scopes of radiation protection programs and such guidance will be prepared in the form of Regulatory Guides.

The Commission continues to emphasize the importance of the ALARA concept to an adequate radiation protection program. In order to strengthen this concept, the Commission has adopted a requirement that all licensees include provisions for maintaining radiation doses and intakes of radioactive materials as low as is reasonably achievable as part of their radiation protection programs. Compliance with this requirement will be judged on whether the licensee has incorporated measures to track and, if necessary, to reduce exposures and not whether exposures and doses represent an absolute minimum or whether the licensee has used all possible methods to reduce exposures. This shift in emphasis should reduce potential problems of retrospective evaluation of licensee performance under admittedly subjective criteria. However, the licensee should be able to demonstrate that periodic reviews of performance have been made and that efforts have been made to achieve ALARA. As noted above, the level of effort expended on the radiation protection programs should reflect the magnitude of the potential exposures, both the magnitude of average and maximum individual doses and, in facilities with large numbers of employees, collective (population) doses. A nuclear power reactor licensee would be expected to have a considerably larger program than a licensee with only small sealed sources.

The Commission has not adopted a requirement that a numerical cost-benefit analysis (optimization analysis) be used to demonstrate ALARA. The quantitative

approach is useful for those situations where both costs and benefits (dose reduction) can be quantitated, such as in shielding design or analysis of decontamination methods. The Commission encourages licensees to employ quantified analyses to define ALARA, but their use is not required. One reason for this is that many ALARA procedures simply reflect sound operating practice and do not lend themselves to a numerical analysis. Another reason is that cost-benefit analyses could have a cost associated with obtaining the necessary information and carrying out the analysis that may exceed the monetary value of the dose reduction. Thus the quantitative optimization analysis would be expected to be used primarily in situations where both the costs of control and the resultant benefits were not only quantifiable, but also appreciable compared to the cost of performing the analysis.

#### Subpart C--Occupational Dose Limits

Section 20.201 Occupational Dose Limits for Adults.

<u>Comment</u>: Elimination of the 5(N - 18) age-provated cumulative dose limit and the adoption of the 5-rem annual effective dose limit. Most commenters favored this change noting that licensees have generally succeeded in keeping doses below 5 rems per year for the past few years and, therefore, are already meeting the new limit.

<u>Comment</u>: Lifetime dose limits. A few commenters believe that there should be a limit on the cumulative total dose that can be received by any individual in a lifetime.

Response: The Commission considered the use of a lifetime dose limit but rejected it. The EPA had proposed such a limit (100 rems) in its proposed Federal Guidance on Occupational Radiation Exposure (46 FR 7836, January 23, 1981) but withdrew it.

If the magnitude of the annual dose is limited, there is a <u>de</u> <u>fact</u>o limitation of the lifetime dose that can be received. The Commission believes that

such a <u>de facto</u> lifetime limit is preferable to an actual cumulative lifetime dose limit because the cumulative limit could act to limit employability. This, in turn, raises questions concerning the right of an individual to pursue employment in a chosen profession. If an individual were to deplete the "dose bank" provided by a lifetime dose limit, it might be difficult to obtain future employment using ionizing radiation.

<u>Comment:</u> Quarterly dose limit. A number of commenters noted that the ICRP system of dose limitation does not have quarterly or other limits covering periods less than a year. The public comments also noted the possibility of giving rise to two violations for the same event (i.e., the possibility of exceeding both the quarterly and annual dose limits in one event), thereby incurring two penalties.

Response: The quarterly limit (only for deep-dose equivalent) had been retained in the proposed rule as a result of suggestions received from several groups during the development of the rule. The primary protection function of retaining a quarterly limit was to reduce the potential for receiving several high doses within a relatively short period of time. However, there is not much of a radiobiological significance between 10 rems (two 5-rem doses) and 6 rems (two 3-rem doses) received in a short time period. One consideration is the employability of a worker who has exceeded the dose limit. A worker who exceeded the 5-rem annual dose limit might have to work in a job not involving radiation for a year (or take part in a planned special exposure) instead of only a calendar quarter if a quarterly dose was used.

Final Rule. In order to maintain compatibility with the ICRP and to eliminate the possibility of double violations, the quarterly limit has not been kept and only annual limits are stated.

<u>Comment:</u> Eye dose limit. Some commenters questioned the 15-rem (0.15sievert) eye limit used in the proposed rule noting that ICRP Publication No. 26 contains a recommended value of 0.3 sieverts (30 rems).

Response: The ICRP recommended a reduction in the limit for the eye to 0.15 sieverts (15 rems) at their Brighton, England, meeting in 1980.<sup>13</sup> This was done because the ICRP concluded inat, for a lifetime of occupational exposure at the former 0.3-sievert (30-rem) limit, some opacities in the tense of the eye might be produced that could develop to the point of causing deterioration of vision (even without further radiation exposure). In most situations, the limits for the deep-dose equivalent and the shallow-dose equivalent to the skin should ensure that the eye dose limit is also met. Consequently, the reduction from 30 rems to 15 rems is not expected to have a significant impact on either health protection or control cost.

<u>Comment:</u> Parameters defining the shallow-dose equivalent ("skin dose"). The proposed rule would have established a dose limit for the skin of 50 rems averaged over 10 square centimeters (10 cm<sup>2</sup>). There were several comments concerning the scientific basis for this area. Some commenters suggested other surface areas, such as 15 cm<sup>2</sup> as being better suited to measurement conditions. Proponents of the larger areas generally favored these areas because of their compatibility with either contamination survey practices or with the physical size of survey instrument detector probes.

One set of comments prepared by the developer of the NRC's VARSKIN computer program for skin dose calculation (comment letter No. 262 in the NRC Public Document Room) contains a well-documented discussion of the selection of an appropriate area over which to average the skin dose. These comments conclude that  $1 \text{ cm}^{\Sigma}$  is a more appropriate area than either 10 cm<sup>2</sup> or 100 cm<sup>2</sup>.

Response: ICRP Publication 26 contains two recommendations for such areas: a  $100-cm^2$  area and a  $1-cm^2$  area, the larger area being associated with routine monitoring for skin contamination and the smaller area being associated with accident dose evaluation. After reviewing these comments and various recommen

13 International Commission on Radiological Protection, "Statement and Recommendations of the 1980 Brighton Meeting," <u>Annuals of the ICRP 4(3/4)</u> Oxford, England: Pergamon Press (1980). (Available for sale from Pergamon Press, Elmsford, NY 10523.)

KCMM

used

adding composition

dennets

"Setto range!"

SRME

#7

hom

COMMENT

dations regarding skin dose measurements, the Commission has decided to use the smaller area of 1 cm<sup>2</sup> for routine skin dose evaluations. The 1-cm<sup>2</sup> area is consistent with the prior recommendations in NCRP Report No.  $39^{14}$  and ICRP Publication No.  $9^{15}$  as well as the smaller area recommended in ICRP Publication No. 26.

Within the past several years, there have been instances where very small (5-250 µm) "hot" particles of fuel or activated corrosion products have been discovered in reactor facilities, on workers or their clothing, and, in a few isolated cases, in worker's vehicles or homes. These particles are generally too large to pose a significant risk from inhalation, but are capable of producing intense beta-radiation doses over very small areas of the skin. The principal hazard appears to be skin ulceration if the particles remain localized on the skin surface. The primary uncertainty associated with evaluating the hazard of these small particles is determining the skin area or tissue volume to which the dose is to be computed (or even whether "dose" is the most appropriate indicator of the hazard). The NRC requested the National Council on Radiation Protection and Measurements (NCRP) to look into the hot particle issue and make recommendations. The NRCP's recommendations have been published in NCRP Report No. 10616 and use a criterion based upon the number of radioactive disintegrations that have occurred (µCi-hours) rather than dose. The NRC staff is reviewing these recommendations and has issued an Information Notice on a modified enforcement policy for hot particles.

Final Rule. This revision of Part 20 specifies an area of  $1 \text{ cm}^2$  for skin dose evaluations.

National Council on Radiation Protection and Measurements, "Basic Radiation Protection Criteria," NCRP Report No. 39 (January 15, 1971), page 79, paragraph 207. (Available for sale from the NCRP, Bethesda, MD 20814.)
International Commission on Radiological Protection, "Recommendations of the International Commission on Radiological Protection (adopted September 17, 1965)," ICRP Publication No. 9 (1966), page 6, paragraph 28. (Available for sale from Pergamon Press, Inc., Elmsford, NY 10523.)
National Council on Radiation Protection and Measurements, "Limit for Exposure to 'Hot Particles' on the Skin," NCRP Report No. 106 (December 31, 1989). (Available for sale from the NCRP, Bethesda, MD 20814.)

14

<u>Comment:</u> Effective dose equivalent for external exposure. The most prevalent comment concerning the effective dose equivalent is the restriction in the proposed rule of the risk-weighted organ dose "effective dose" concept to internal doses without permitting a similar approach to be employed for external doses. There were several comments that noted the desirability of using organ weighting factors for external doses.

Response: The ICRP and NCRP recommendations and the 1987 Federal guidance on occupational radiation exposure in principle permit the use of external weighting factors. However, none of the principal standard-setting organizations has included specific recommendations for the use of weighting factors for external dose.

The application of weighting factors also entails calculation of organ doses instead of whole-body doses from external radiation. One component of this calculation is estimation of the attenuation of the radiation as a function of the depth of the organ in the body. There are practical problems in the determination of the type and energies of the radiation involved and of the orientation of the individual with respect to the source of the radiation that have to be considered in making such calculations. Therefore, application of weighting factors for external exposures will be evaluated on a case-by-case basis until more guidance and additional weighting factors (such as for the head and the extremities) are recommended.

Final Rule. External doses to the head, trunk (including male gonads), arms above the elbow, or legs above the knee are to be treated as whole-body doses. For the purpose of weighting the external whole-body dose (for adding it to the internal dose), a single weighting factor,  $w_T = 1.0$ , has been specified. The use of other weighting factors for external exposure may be approved on a case-by-case basis upon request to the NRC.

<u>Comment:</u> Allowance for exposure after limits are exceeded. Commenters noted that allowance of an additional 1 rem per quarter dose limit for a worker who had already exceeded the 5-rem annual limit might be counterproductive.

Workers who remain under the annual limit, and whose dose was X rems, would be constrained to receive (5 - X) rems, whereas workers who received more than 5 rems in the first quarter could be allowed an additional 4 rems (1 rem in each of the four quarters). One commenter suggested that this could provide an incentive for individuals who are approaching the dose limit to deliberately exceed the limit and thereby protect their employability by taking advantage of the extra dose allowance available to those who have exceeded the limits. Another commenter believed that such a blanket authorization to exceed the limits was inappropriate and preferred prior NRC review of the use of these extra doses on a case-by-case basis.

Response: The purpose of the dose allowance was to protect the worker's employability after having received a dose above the dose limits. Although intentionally getting additional exposure might be in the worker's interest for employability reasons, such an action would not be in the worker's interest with respect to health protection. Licensees having workers with critical skills who are approaching the dose limits early in the year or workers who have received an accidental overexposure should consider use of the planned special exposure (§ 20.206) to permit continued employment.

Final Rule. The allowance of an additional 1 rem per quarter following an exposure in excess of the limits has been deleted.

Section 20.202 Compliance with Requirements for Summation of Internal and External Doses.

<u>Comment:</u> Implementation burden. Many commenters felt that the burden of adding external and internal doses was substantial, particularly as most licensees would be faced with either external exposure situations or internal dose situations, but not both.

Response: The NRC staff disagrees that there will be a substantial recordkeeping burden because this summation will be required only if both the internal dose and the external dose are each likely to exceed 10 percent of the dose

limit. Thus, in most situations, as noted in the comments, only one component will be required to be measured and, consequently, summation of internal and external doses will not be required.

Final Rule. The requirement remains that the committed effective dose equivalent and the deep-dose equivalent should be summed to give the total effective dose equivalent. However, this summation need only be performed if both components are required to be monitored (i.e., exceed 10% of an applicable dose limit). If the summation of doses is not required, then the limit applies to the component (internal or external) that is measured. The NRC is planning to issue additional guidance in the form of a regulatory guide before the effective date of the revised Part 20. This guide will be on procedures to be used in estimating committed effective dose equivalents and deep-dose equivalents and guidance on when internal and external doses have to be summed.

<u>Comment:</u> Use of individual metabolic or dosimetric data. Several commenters thought that the proposed rule required the use of specific metabolic and dosimetric parameters for the exposed individual. One commenter also thought that the use of such parameters would "invalidate the stochastic approach of the regulation, which presumes that the effects of radiation exposure at these levels are statistical in nature."

Response: It was not intended that licensees would be required to collect and use specific metabolic or dosimetric information on exposed individuals for use in dose assessments. The intent was to permit the use of personal data for dose assessment when such data were available. The use of parameters that are more appropriate for a particular exposed individual than those assumed for the "Reference Man" should improve the accuracy of the dose estimate for that individual. This is unrelated to the concept of stochastic health effects.

The statistical nature of the potential stochastic effects of low doses of ionizing radiation does not require that the associated dose estimates be based on Reference Man doses. However, it is necessary to resort to population-averaged dose-to-risk conversion factors as there are no health risk coefficients available for specific individuals.

[Monitoring thresholds and thresholds for summation of internal and external dose -- see discussion under § 20.502]

Note: Section 20.202(c) states that: "The assigned deep-dose equivalent and shallow-dose equivalent must be for the part of the body receiving the highest exposure." This requirement is intended to apply primarily to situations where there are steep gradients in the radiation dose rate, depending upon location within the facility and spatial orientation of the worker's body. For example, good practice for a worker in a nuclear power plant who is reaching up into a radioactive steam generator would be to wear at least two personnel dosimeters: one to monitor the extremity dose (worn on the finger or wrist) and one to monitor the whole-body dose (worn on the upper arm). For routine monitoring in relatively homogeneous radiation fields, special consideration to identify the actual "highest" exposed area would not be required.

# Section 20.203 Determination of External Dose from Airborne Radioactive Material.

<u>Comment:</u> This could be read to require that the air concentration be measured at two locations. This section appears to require that the air concentration be measured at the location of the individual and at the point of maximum concentration in the cloud. The regulation should emphasize the reliance on personnel dosimeters or other monitoring devices.

Response and Final Rule. Section 20.203 has been shortened considerably. The revised section emphasizes the use of survey instruments and personnel monitoring devices to evaluate the external dose. The remaining technical guidance from this section in the proposed rule will be incorporated into a regulatory guide. 267 \$3 Section 20.204 Determination of Internal Exposure.

Comment: Interim dose calculation factors and parameters. Because the existing Part 20 is based on ICRP-217 dosimetry and metabolic models and the revised Part 20 employs the ICRP-30<sup>18</sup> dose parameters, there was concern regarding whether the more recent ICRP-30 parameters should be used, particularly when the value is to be compared with the intake limits in the present Part 20.

Response: The NRC is planning to issue a regulatory guide that will address the use of bioassay measurements for determining compliance with Pari 20. 1033 Appropriate parameters for calculating organ doses from radionuclide intakes can be found in ICRP-30 and its supplements. Dose factors in Federal Guidance 267 Report #11<sup>19</sup> are also acceptable for use in calculating occupational exposures. However, the effective dose equivalent factors in Federal Guidance Report #11 do not employ a rounding method suggested in ICRP-30. For this reason, the dose factors in Report #11 may be be slightly higher (10-20 percent) than the effective dose factors which correspond to the ALIs and DACs in both the revised Part 20 and Report #11. These dose factors would be more restrictive (give slightly higher doses for the same intake) than dose factors computed using the ICRP-30 roundoff procedure, but they can be used for evaluating compliance with Part 20.

E Section on Implementation Delated pour discussion Section 20.205 [deleted] Further Provisions -- Internal Exposure Involving Radionuclides with Very Long Effective Half-Lives.

Comment: Exemption for long-lived radionuclides and the use of the committed dose equivalent concept. The use of the concept of a "committed dose equi-

International Commission on Radiological Protection, "Report of Committee 17 II on Permissible Dose for Internal Radiation," ICRP Publication No. 2 18

(1959). (Available for sale from Pergamon Press, Elmsford, N.Y. 10523.) International Commission on Radiological Protection, "Limits for Intakes of Radionuclides by Workers," ICRP Publication No. 30. (Available for sale from Pergamon Press, Elmsford, N.Y. 10523.) 19

Environmental Protection Agency, Federal Guidance Report No. 11, "Limiting Values of Radionuclide Intake and Air Concentration, and Dose Conversion Factors for Inhalation, Submersion and Ingestion." USEPA Report EPA-520/ 1-88-020 (September 1988). (Available from the USEPA, Office of Radiation Programs, 401 M Street, S.W., Washington, DC 20460.)

valent" drew numerous comments. This approach entails assigning to the year of intake the future internal dose (the "committed dose equivalent" over 50 years) from radionuclides taken into the body during that year. The proposed rule (in § 20.205) allowed an exemption from the use of committed dose equivalents for several long-lived radionuclides.

Many of the commenters objected to having to assign the future 50-year dose to a single year. Others suggested that variable integration periods be allowed instead of one fixed 50-year value. One argument offered in support of either of these positions is that many adult workers would not normally be expected to live long enough to accrue the full 50-year committed dose equivalent.

Commenters pointed out that while pre-exposure controls (such as the annual limits on intake and the derived air concentrations) should be based upon the committed dose equivalent concept for planning and control, the use of controls based upon limiting the annual effective dose equivalent rate (rather than using the committed dose equivalent) might be preferable for post-exposure management following actual radionuclide intakes.

It was also noted that there were several additional nuclides that had similar half-lives and retention characteristics but were not included in the proposed exception. Among these were cobalt, strontium, and americium. The approach in the proposed rule was characterized as appearing to place almost complete emphasis on the control of the work environment rather than on the assessment and control of the individual worker.

Response: The concept of dose commitment is not new; this concept has been used as the basis for controlling internal doses since the late 1950s when ICRP Publication No.  $2^{17}$  and the present 10 CFR Part 20 were published. However, the term "committed dose equivalent" applied to future doses from internal emitters initially appeared in 1977 in ICRP Publication No. 26.<sup>1</sup>

The concentration limits for air and water in Appendix B to the existing Part 20 were based upon concentrations which, if continually inhaled (for air)

or ingested (for water) over a 50-year period, would produce a dose rate in the "critical organ" in the 50th year that was numerically equal to the annual organ dose limit. For certain radionuclides that slowly approached a constant body burden, primarily those radionuclides that have both long radiological half-lives and long biological clearance half-times, the limiting organ dose rate is not reached by the 50th year. For shorter-lived radionuclides and those that are rapidly removed from the body, equilibrium may be attained more rapidly and the limiting annual organ dose rate could persist over many years.

The limiting <u>dose rate</u> in the 50th year from a constant intake of a radionuclide each year over a 50-year period is numerically equal to the <u>total dose</u> integrated over the 50-year period from a single year's intake of the same magnitude. Therefore, controlling the integrated future ("committed") dose for each year's radionuclide intake also controls the annual dose rate in the 50th year to be within the dose limit.

It was noted that use of limits to annual doses in some cases would not ensure that doses in future years would be within limits. The example of the ingrowth of americium-241 from plutonium-241 was cited in which, even if the initial annual dose from plutonium-241 were within the limit, the ingrowth of the radiologically more significant americium-241 would lead to doses higher than the limits in subsequent years.

There are only a few radionuclides that would not attain an equilibrium Tevel (and a constant annual organ dose rate) within time periods of less than 50 years. The use of the committed dose equivalent, rather than controlling internal dose on the basis of annual dose, substantially overestimates annual doses only for those radionuclides that do not reach an equilibrium level in the body early in the working lifetime. These radionuclides are primarily the long-lived radionuclides for which the exemptions of § 20.205 in the proposed rule were intended. Radionuclides (such as cobalt-60, strontium-90, and americium-241) that were easily measured at airborne concentrations or body burdens below the DAC and ALI values were not included in the list of exempted radionuclides because an exemption was not believed to be necessary for them.

267 p2

The annual limits on intake and derived air concentrations are used mainly for pre-exposure control rather than post-exposure dose assessment so that finetuning these values to specific ages or adjusting them for factors such as the length of the period over which the committed dose is evaluated or to differences in individual organ sizes (as were suggested) is not warranted for occupational dose assessment. The use of age-dependent committed dose factors as suggested by some commenters would add needless complexity to the assessment of internal doses and cannot be justified on the basis of the availability of information on either age-dependent metabolic parameters or age-dependent radiobiological risk information.

The use of an annual dose limitation system, even with a reduction in the allowable dose limit from 5 rems to 3 rems such as in the proposed § 20.205, does not provide a limitation on the lifetime radiation dose or risk equivalent to that provided by the committed dose limitation system of this final rule for all classes of workers. Although long-term workers would be protected to the same degree under either the annual or committed dose systems, short-term or temporary workers could get somewhat higher lifetime doses under a dose limitation system based on limiting only individual annual dose. Furthermore, it is neither reasonable nor practical to expect future employers to take special measures to control radiation dose to workers who transfer because a previous employer, working under annual organ dose limits, permitted intakes that would result in future dose rates that are appreciable fractions of the allowable dose limits. Such a practice would not be fair to workers whose future employability may be limited because of the additional restrictions a new employer would have to put on their exposure, or to future employers of these workers who may have to assess internal doses from residual body burdens of internal radionuclides in order to show compliance. The annual dose system also requires e xya lawatia a complex bookkeeping effort because the annual dose limit for each worker depends upon the worker's pre-existing body burden of radioactive materials.

Final Rule. For the reasons discussed above, the Commission has decided not to adopt proposed § 20.205 and the exemptions for certain long-lived radionuclides for the final rule. The use of the committed dose equivalent will be applied uniformly to all radionuclides, regardless of half-life. The Commission

recognizes that the removal of this exemption, combined with the lowering of the airborne concentration limits for several radionuclides (notably thorium and uranium), could impact on the current and future facilities that use these materials. Licensees that are affected by these changes may request an extension of the implementation time in order to make the necessary modifications to comply with the revised limits as they relate to long-lived radionuclides identified in the proposed § 20.205. In addition, licensees should note the flexibility provided in the revised rule for more accurate dose assessments 5RM2#2 + RoVS to be made that might show that additional controls were not required in order to meet the dose limits. Specifically, § 20.204 allows the use of actual particle-size distributions and physiochemical characteristics of airborne particulates to define a site-specific derived air concentration to be used in lieu of the generic values in Appendix B. Such adjustments result in the use of more precise dose estimates because of a better characterization of the actual exposure conditions. Although these adjustments might permit higher airborne radionuclide concentration limits to be used, the same degree of health protection would exist because the radiation dose (and risk) would remain the same. This section also allows for whole-body counting or bioassay measurements to determine the behavior of radioactive materials in the individual and the use of these data to calculate internal doses. A 7-month delay between a bioassay or retention measurement and recording of the associated dose is also permitted in order to make confirmatory measurements.

The Commission recognizes that alternative methods may be identified in the future that might achieve the same degree of lifetime risk limitation for both short-term and long-term workers as the dose system recommended by the ICRP, the Radiation Protection Guidance to Federal Agencies for Occupational Exposure, and adopted in the current and revised 10 CFR Part 20. The Commission further believes that, to be acceptable, such alternatives should not result in an adverse impact on worker employability or result in undue recordkeeping or excessive monitoring requirements for the future employers of transferring workers.

267 p2-3

Section 20.206 Planned Special Exposures.

Comment: The use of planned special exposures could result in lifetime cumulative doses greater than those doses formerly permitted under the 5(N - 18) formula. One commenter noted that the new regulatory scheme, including planned special exposures, allowed a higher total lifetime dose than was permitted using the 5(N - 18) formula. The calculation presumes a working lifetime of 47 years (starting at age 18 and ending at age 65). Under the revised Part 20, the lifetime limiting dose would be 260 rems (5 rems per year)(47years) + 5(5 rems) (planned special exposures) = 235 + 25 = 260 rems). Under the 5(N - 18) formula, at age 65 (N = 65), the cumulative dose would be 5(47) = 235 rems. The comment further noted that the NCRP recommended [in NCRP Report No. 91] a cumulative dose limit of 1 rem x age; the Department of Energy has proposed a 100-rem lifetime dose limit, and the ICRP at its 1984 Stockholm meeting inferred a goal of 1 rem per year. Other commenters noted that, because of the potential lifetime dose including the planned special exposure, the claim on page 51, FR 1121 (Table 5), of the proposed rule that "Individuals receiving highest exposure will be reduced" is unjustified and incorrect.

Response: The analysis of maximum doses discussed above is overly simplified because it assumes that there are individuals who will be exposed at the allowable dose limit every year of their working lifetime. Under the old 5(N - 18) formula, the unused portion of the dose limit (the difference between the actual dose received and 5 rems) became part of a "dose bank" that could be drawn on in later years (at a rate up to 3 rems per quarter or 12 rems per year). This "dose bank," which is inherent in the age-prorated formula of 5(N - 18), does not exist with the straight annual dose limit. If the worker's exposure is under the 5-rem annual dose limit, there is no way to recapture the difference for use in future years. Consequently, the average annual dose (for the more highly exposed workers) associated with new Part 20 is expected to be less than under the former rule.

As noted above (see Response under § 20.201 Occupational Dose Limits), the Commission considered the use of a lifetime dose limit but rejected it.

<u>Comment:</u> Planned special exposures should not be limited to external exposures but should also be permitted for internal exposures. Several commenters noted that it was inconsistent to treat internal and external doses as equivalent by summing them and then restricting planned special exposures to only external doses. Commenters also pointed out that the total effective dose equivalent (TEDE) could be minimized in some cases if some external doses were reduced at the expense of incurring some internal doses.

Response: The Commission agrees that restricting the use of planned special exposures to only external doses would be inconsistent with the ALARA principle and the presumed equivalence of internal and external doses inherent in the revised Part 20. Consequently, the requirements have been modified so internal doses may be included in planned special exposures in order that the total dose (TEDE) can be controlled in keeping with ALARA.

<u>Comment</u>: The annual dose allowed in a planned special exposure does not agree with the recommendations of the ICRP. A few commenters thought that the allowable annual dose from planned special exposures should be 10 rems as stated in the ICRP recommendations. Other commenters agreed with the NRC's modification to reduce the annual dose for planned special exposures to 5 rems.

Response: The NRC has intentionally reduced the dose allowed in any year from a planned special exposure from the 10-rem value proposed by the ICRP to 5 rems. The lifetime total limit from planned special exposures of 25 rems remains the same as the ICRP recommendation. The Commission believes that it would be better to distribute the dose over the lifetime more evenly than to permit a large portion of the cumulative dose to be received within a small period of time. In this sense it should be recalled that the planned special exposure is in addition to the normal dose limits. Under the Part 20 condition, it would be theoretically possible to get a 10-rem dose in 1 year, 5 rems from a planned special exposure and 5 rems from routine operation. This is roughly equivalent to the 12 rems (3 rems/quarter) that could be received under the present Part 20 limitations using the 5(N - 18) formula. The initial ICRP proposal would have permitted a 15-rem dose in 1 year, 10 rems from planned special exposures and 5 rems from planned special exposure is in a special operation.

<u>Comment</u>: Subtraction of emergency doses. Some commenters suggested that doses received under emergency conditions, up to a lifetime total of 25 rems, not be subtracted from the lifetime allowance for planned special exposures. It was also suggested that the employability of the individual might be jeopardized if the dose "bank" were depleted.

Response: The NRC has not officially sanctioned the 25-rem "forgivable" emergency dose that has been recommended by some organizations for a once-in-alifetime dose that would not be counted against an individual's lifetime dose. Consequently, all doses received as a result of occupational exposure must be recorded in an individual worker's record.

The Commission believes that planned special exposures will be used infrequently so that the lack of a dose bank for some individuals would not be a major drawback to their employability.

<u>Comment:</u> The time period for notifying exposed individuals of their dose is too short. A number of commenters thought that the 15-day period for notifying exposed individuals of their exposure from a planned special exposure was too short. Some commenters noted that most NRC reporting requirements provide a 30-day, not a 15-day, period. Other commenters suggested that the 15-day period could give the impression [to the worker] that an inordinate risk was involved when that was not the case.

Response: The 15-day period for notification was intended to be unique and to further emphasize that "planned special exposures" were indeed "special." However, the Commission has extended the time period for notification of the individual from 15 days to 30 days to allow licensees additional time to estimate internal exposures that are now permitted in the revised rule to be part of a planned special exposure. The requirement to notify the NRC (see § 20.1204) that a planned special exposure has taken place is also 30 days. <u>Comment:</u> Doses received during a planned special exposure that do not exceed the dose limits for normal operation should not have to be recorded as planned special exposures or be subtracted from the lifetime planned special exposure limit. A few commenters expressed concern that exposures during planned special exposures that did not result in doses to an individual in excess of the occupational annual dose limits would nevertheless have to be reported separately and subtracted from the individual's lifetime allotment for planned special exposures.

Response: The intent of the planned special exposure was that it would be used infrequently in circumstances where the elimination of the 5(N - 18) lifetime cumulative limit might create a severe handicap to the licensee's operations. Being able to switch doses between planned special exposures and routine dose limits would tend to encourage the use of planned special exposures as the licensee would have nothing to lose by using the planned special exposure. This is contrary to the Commission's intent that the planned special exposures be restricted to "special" situations. Once a licensee decides to conduct a planned special exposure, all of the unique limitations, reporting, and recordkeeping requirements are to apply, even if the doses actually received fall within the dose limits for routing operations.

Final Rule. The provisions of planned special exposures have been extended to include internal exposures, and the reporting time to the individuals involved has been changed to 30 days to allow sufficient time for analysis of  $\mathcal{MCMY}$  internal dose.

Section 20.207 Occupational Dose Limits for Minors.

<u>Comment:</u> Exposure of Minors. One commenter stated that minors should not be exposed to radiation because they do not meet the criteria for occupational radiation exposure. The commenter argued that minors are not trained regarding radiation protection, do not derive a benefit from employment, and would require the preparation of an NRC Form 4 if they were workers.

Response: Allowing minors to be occupationally exposed to radiation was permitted in the present Part 20 (§ 20.104). All individuals, including minors, who enter a restricted area are required (10 CFR 19.12) to be instructed as to the risks involved. Minors who are employed receive salaries and other associated benefits of employment so that there does not appear to be a major difference in this respect from other workers. Furthermore, licensees are required under the existing and revised Part 20 rules to maintain the same exposure records for minors as for adults.

An alternative to this procedure would be to exclude minors completely from radiation-related work. This does not appear to be desirable as the monetary, experience, and educational benefits that may accrue to the minor appear to outweigh the small incremental risk involved (particularly considering the -educed dose limits applied to minors). Consequently, no change has been made from the proposed rule.

## Section 20.208 Dose to an Embryo/Fetus.

<u>Comment:</u> Biological basis for lower dose limits for pregnant women. There were comments that cited older studies and recommendations for dose limits for the embryo/fetus that are considerably higher than 0.5 rem. These comments questioned the biological basis for the 0.5 rem dose limit for the embryo/fetus in the proposed rule.

Response: The biological effects of ionizing radiation upon the embryo/ fetus are summarized in Regulatory Guide 8.13<sup>20</sup> More detailed information can

<sup>20</sup> U.S. Nuclear Regulatory Commission, "Instructions Concerning Prenatal Radiation Exposure," Regulatory Guide 8.13, Rev. 2, December 1987.

be found in publications of the NCRP, 21 ICRP, 22 UNSCEAR, 23 and the OECD/NEA. 24

The limit of 0.5 rem during the entire gestation period is based upon a recommendation by the NCRP in 1977 (see Reference 21). The International Commission on Radiological Protection (ICRP-26) recommended 0.3 times the annual dose limit or 15 mSv (1.5 rems) over the full gestation period and 5 mSv (0.5 rem) in the first 2 months of pregnancy.

Final Rule. The limit for the embryo/fetus of a declared pregnant woman is 0.5 rem over the entire gestation period. There is also an admonition that the licensee avoid substantial variation above the average monthly exposure rate that would comply with the 0.5-rem limit. These conditions are consistent with the Federal guidance on occupational radiation exposure and with the recommendations of the NCRP in NCRP Report No. 91. $^{6}$ 

<u>Comment:</u> Licensee's Responsibilities to Protect the Embryo/Fetus of an Undeclared Pregnant Woman. Several commenters raised the question of whether the licensee had any responsibility for protecting the embryo/fetus of an obviously pregnant female employee who had not formally declared her pregnancy to the employer.

Response: It is the fundamental responsibility of the pregnant worker to decide when or whether she will formally declare her condition to her employer. This position is derived from court rulings concerning a pregnant woman's rights regarding termination of the pregnancy. Having a woman formally declaring her

<sup>21</sup> National Council on Radiation Protection and Measurements, "Review of Radiation Dose Limit for Embryo and Fetus in Occupationally Exposed Women," NCRP Report No. 53 (1971). (Available for sale from the NCRP, 7910 Woodmont Avenue, Suite 800, Bethesda, MD 20814-3095.)
22 International Commission on Radiological Protection. "Developmental Effect."

<sup>22</sup> International Commission on Radiological Protection, "Developmental Effects of Irradiation on the Brain of the Embryo and Fetus," Annals of the ICRP 16 4) (1986). (Available for sale from Pergamon Press Filmsford NY 10522)

 <sup>16 4) (1986). (</sup>Available for sale from Pergamon Press, Elmsford, NY 10523.)
 23 United Nations Scientific Committee on the Effects of Atomic Radiation (UNSCEAR), Genetic and Somatic Effects of Ionizing Radiation Sales Section, United Nations, NY 1986, particularly Chapter III, Biological Effects of Pre-natal Irradiation."

<sup>24</sup> Organization for Economic Cooperation and Development/Nuclear Energy Agency, "The Biological Basis for the Control of Prenatal Irradiation," OECD/NEA, Paris, France (1988).

pregnancy to her employer derives from legal, not health protection, considerations. If she chooses not to declare her pregnancy, the licensee will not be required under the Commission's regulations to limit her dose to the 0.5-rem limit.

Undeclared pregnant women are protected under the NRC regulations for all workers. The normal occupational dose limits would still be in effect and would have to be complied with, and the dose would also have to be kept "as low as is reasonably achievable." In addition, as part of her initial employment, the woman should have received instructions in radiation protection (10 CFR 19.12), and she should have been provided with a copy of Regulatory Guide 8.13.

It might be prudent for a licensee to remind a pregnant, but undeclared, worker of the special limit for protection of the embryo/fetus of a declared pregnant woman and to provide another copy of Regulatory Guide 8.13 to her. However, if the licensee has previously provided this information to the employee, it is not a Commission requirement that it be done again. If the requirements referred to in the previous paragraph have been fulfilled, the licensee will not be cited for a violation of the Commission's regulations if the estimated dose to the embryo/fetus of an undeclared pregnant woman exceeds the 0.5-rem limit, even if the worker's pregnant state seems obvious.

<u>Comment:</u> Requirements on the worker to declare pregnancy information on child-bearing capacity. Some commenters called for a requirement that the employee declare the pregnancy to the employer as soon as it is known to the pregnant woman. Another commenter suggested that two classes of women, "fertile" and "nonfertile," be established with separate dose limits for each class.

Response: Section 161c. of the Atomic Energy Act gives NRC the authority to require such information to be provided by the worker. However, such a requirement could be considered to be discriminatory and an invasion of personal privacy. It would also be unenforceable because the woman and her physician know when she knew of the pregnancy and patient-doctor communications are privileged. Infringement on personal privacy is also a drawback that applies to

267 P3

requiring the female worker to supply information concerning her "fertility" or "infertility."

<u>Comment</u>: Estimation of Dose to the Embryo/Fetus. The assignment to the embryo/fetus of a dose equal to the dose to the declared pregnant woman was questioned. For example, would it be reasonable to assign to the embryo/fetus a dose based upon the dose received by the woman's shoulder or head?

Commenters also indicated that licensees should be permitted to employ factors other than a factor of 2 and take into account shielding of the embryo/ fetus by maternal organs and the placenta in evaluating the external dose component of the embryo/fetus.

Response: The concept used in the proposed rule of relating the dose to the embryo/fetus to the dose received by the mother has been modified. The final rule permits direct calculation of the dose to the embryo/fetus. This was done so that the use of more accurate dose assessments would not be precluded by the rule. The internal dose to the embryo/fetus may or may not be directly proportional to the dose received by the mother.

A forthcuming regulatory guide will provide guidance on methods for calculating the dose to the embryo/fetus. For interim assessments of the dose to the embryo/fetus, it may be assumed that the dose to the embryo/fetus from external radiation and from radionuclides in the body that are relatively uniformly distributed, such as cesium-137 and compounds of tritium and carbon-14 that are not organically bound, is the same as the dose to the mother since under these circumstances the same energy would be deposited per gram of tissue in both the mother and the fetus. For external gamma irradiation, the assumption unai the dose to the fetus is the same as to the mother should be conservative (yield calculated doses that are somewhat higher than the actual doses determined by more precise evaluations).

Permitting calculations of the embryo/fetal dose using reduction factors for attenuation within the body of the mother would entail knowledge of the

energy spectra of the incident radiation. As noted previously (Response for § 20.201), photon spectral measurements, although technically feasible, are not currently required by the Commission and are considered to be beyond the scope of routine radiation protection survey measurements. The small amount of reduction in the calculated dose afforded by such attenuation corrections would be secondary in importance compared to uncertainties due to body orientation, partial-body exposure from collimated beams of radiation, and the radio-biological sensitivity of the embryo/fetus.

In situations where the use of a single dose measurement would be inappropriate for both the woman and the embryo/fetus, a solution would be to monitor the two doses separately.

<u>Comment</u>: Additional Dose Increment Allowed to Pregnant Women Beyond the Dose Limits. The rationale was requested by a few commenters for permitting an extra 0.05 rem (0.5 millisievert) beyond the 0.5-rem (5 millisieverts) dose limit to an embryo/fetus.

Response: The small additional dose is intended to apply in situations where the embryo/fetus has accumulated a substantial fraction of the dose limit or has already exceeded the limit before the woman formally declares herself to be a "declared pregnant woman." If the incremental 0.05-rem dose were not available, a woman having already received a dose in excess of the 0.5-rem limit might not be able to be further employed in a radiation-related job. The licensee could be in "instant noncompliance" as the embryo/fetus dose limit could have been exceeded before the licensee was aware that it was applicable (i.e., before the woman declared her pregnancy). Thus, the small incremental 0.05-rem dose provides a means of ensuring continued employment for the woman and also removes the threat of inadvertent noncompliance on the part of the licensee. The additional risk posed by this incremental dose to the embryo/ fetus is small compared to the potential risk from the overall 0.5-rem dose limit.

Final Rule. The final rule corrects an anomaly in the proposed rule regarding the application of the additional 0.05-rem incremental dose. In the proposed rule, the additional 0.05-rem dose was available if the embryo/ fetal dose limit had been exceeded prior to the woman's declaration of pregnancy (even if the dose were 0.501 rem). However, the additional 0.05-rem dose increment would not have been available if the embryo/fetal dose were less than the 0.5-rem limit (even if the dose were as much as 0.499 rem). There is no significant difference in risk between 0.551 (0.501 + 0.05) rem and 0.549 (0.499 + 0.05) rem. This provision would have resulted in unnecessary penalties to both the licensee and the declared pregnant woman. In the final rule, the 0.05-rem dose increment is available as an additional dose if the embryo/fetal dose at the time of declaration is greater than 0.45 rem (0.45 = 0.5 - 0.05).

### Subpart D--Radiation Dose Limits for Individual Members of the Public

Section 20.301 Dose Limits for Individual Members of the Public.

<u>Comment:</u> NRC should defer changes to limits for the general public until the EPA issues revised Federal guidance. The EPA suggested that NRC not modify its radiation limits for protection of the general public until EPA prepares revised Federal guidance on dose limits applicable to the general public (the recently issued Federal guidance applied only to occupational radiation protection).

Response: Although it would be desirable to use Federal guidance as a basis for the revision of the limits for the public, the Commission believes that Part 20 needs to be based on a consistent set of principles and concepts rather than having its standards for workers using one dose limitation system and its standards for the general public using an entirely different (and outmoded) system. The latest Federal guidance does not address radiation exposure of the general public and, <u>although the NRC staff is represented on an EPA Task</u>. <u>Group which is developing draft Federal guidance on doses to members of the</u> general public, the Commission has chosen not to defer these limits until this KCMM

Task Group has completed drafting the guidance and EPA makes recommendations to the President for its issuance. The Commission's intent to address these limits was noted explicitly in the statement of considerations that accompanied the proposed rule (51 FR 1118, Section XXVIII).

<u>Comment:</u> Facilities that are subject to other lower standards should not have to demonstrate compliance with the 0.1-rem limit ["reference level"]. Several commenters expressed concern that additional efforts would be required to demonstrate compliance with the proposed 0.1-rem "reference level." For licensees that were already subject to the 0.025-rem (25-millirem) limits of EPA's 40 CFR Part 190, this appeared to be an unnecessary burden.

Response: The concept that 0.1 rem represents a "Reference Lovel" has been eliminated. The 0.1-rem value in the final rule represents the primary dose limit for protection of the public. This change from the proposed rule reflects the clarifications by the ICRP (see Section II.A.) regarding the usage of the 0.1-rem and 0.5-rem recommended dose levels. This change does not represent a major change from the proposed rule. Many commenters had indicated a belief that, because of the reporting and control requirements associated with the 0.1-rem reference level, it already represented a <u>de facto</u> limit.

Demonstration of compliance with the limits in 40 CFR Part 190 or with the design objectives of Appendix I to 10 CFR Part 50 will be deemed to demonstrate KCMU compliance with the 0.1-rem dose limit for most licensed facilities. Power reactor licensees that comply with Appendix I may also have to demonstrate that they are within the 0.025-rem limit in 40 CFR Part 190. Demonstration of compliance with the limits of 40 CFR Part 190 will be considered to demonstrate compliance with the 0.1-rem limit. For uranium mills, it will be necessary to show that the dose from radon and its daughters, when added to the dose calculated for 40 CFR Part 190 compliance, does not exceed 0.1 rem.

The dose rate limit of 2 millirems in any 1 hour from § 20.105(b)(1) of the 267 present Part 20 was omitted in the proposed rule but has been reinstated in the revised rule. The reason for this is that this limit provides a more readily

measurable quantity than the 100 millirem per year value and can be more easily verified by short-term measurements.

<u>Comment:</u> Inclusion of doses from other licensed or unlicensed radiation sources. Many commenters expressed an opinion that the dose should not be allinclusive and should not include fallout from nuclear weapons tests, transportation of radioactive material, or other sources of radiation not under the control of the licensee.

Response: The new lower dose limit for members of the general public (which was described as a "reference level" in the proposed rule) applies only to doses from radiation and radioactive materials under the licensee's control. The EPA's generally applicable environmental radiation limit for nuclear power operations (40 CFR Part 190) does apply to the total dose from all sources within the uranium fuel cycle. However, in its practical implementation, the sources would have to be located within a few miles of each other for the combined dose contributions to be significantly different from the dose from either facility alone.

The definition of "natural background" has been replaced by "background radiation," which includes natural background, global fallout, and radon not associated with licensed material. This clarifies sources of radiation and radionuclides that can be excluded from evaluations of the dose from licensed activities.

<u>Comment:</u> Differentiation of limits for long-term operation and for shorterterm transient operation. A number of commenters noted that ICRP-26 described the 0.1 rem (1 mSv) per year value was intended to be an average goal for long-term operation but that 0.5 rem (5 mSv) was intended as the primary annual dose limit for members of the public. Some commenters suggested that a lifetime dose limit be escation of the public.

AH

Response: As noted above in Section II.A., the ICRP has modified its interpretation in the ICRP statement issued following their 1985 Paris meeting." so that the primary standard is 1 mSv (0.1 rem) per year. This clarification of ICRP philosophy is reflected in Part 20 by the change of the 0.1 rem per year value from a "reference level" in the proposed rule to a primary limit in the final rule.

Final Rule. It should be emphasized that the 0.1 rem per year limit in Part 20 is not intended to be applied as a long-term average goal: it is an annual limit. As a matter of practicality, long-term (or lifetime) dose limits for members of the public cannot be implemented unless each year's dose is kept within the long-term goal. Doses to individuals in the general public are not usually monitored directly (locations rather than individuals in the offsite environment are monitored). As individuals may change residency and there is no reporting or tracking system, lifetime doses to specific individuals in the general population are very difficult to determine.

The 0.5 rem per year limit is available only upon specific application to and approval by the Commission (see § 20.301(c)). A 0.5-rem value has been retained in order to apply to transient situations and to alleviate the immediate need to redesign or reshield existing facilities that were designed to meet the REMU former 0.5-rem limit. The 0.5-rem limit is intended to be applied primarily to temporary situations where operation of a facility or the person's exposure to radiation and radioactive emissions, is not expected to result in doses above 0.1 rem over long periods of time. For design of new installations, the 0.1-rem limit should be used. However, existing facilities may apply for NRC approval to use the 0.5-rem limit while more complete evaluation of the need for any additional modifications is performed.

The Commission is aware that some categories of licensees, such as uranium[ mills and in situ uranium mining facilities, may experience difficulties in Pp 4-5 determining compliance with the revised values in Appendix B, Table 2, for radionuclides such as radon-222. Provision has been made for licensees to us air and water concentration limits for protection of members of the general public that are different from those in Appendix B, Table 2, if the licensee

can demonstrate that the physicochemical properties of the effluent justify such modification and the revised value is approved by the NRC. For example, uranium mill licensees could, under this provision, adjust the Table 2 value for radon (with daugnters) to take into account the actual degree of equilibrium present in the environment. This provision permits (upon NRC approval) the use of concentration limits for members of the general public that better represent actual exposure conditions. This is similar to the allowance for use of modified derived air concentrations (with Commission approval) in § 20.204(c)(3). In both situations, licensees would be permitted to propose radionuclide concentration limits for their facility that reflect actual properties of the effluents rather than using the generic concentration-to-dose assumptions associated with Appendix B values. These adjustments tailor the concentration limits to specific conditions, provide the same limitation of dose, and do not permit any greater risk even though the adjusted concentration limits (for members of the general public or for workers) may be higher than the Appendix 8 generic values.

Use of this provision, applied to the percentage of radionuclide equilibrium existing in radioactive decay chains, could provide a factor of 2 or 3 upward change in the appropriate air concentration limit. In addition, the licensee can demonstrate compliance by calculating the dose to the nearest resident rather than meeting the air concentration limit at the site boundary. This should provide an additional factor of 2 or 3 allowance. Lastly, if the 0.1-rem effective dose limit still cannot be met, the licensee can apply to NRC under § 20.301(c) for permission to use a temporary 0.5 rem per year limit rather than the 0.1 rem per year limit. Section 20.301(c) of the revised rule requires that, in order to receive permission for use of this higher dose limit, the licensee has to specify (1) the need for and expected duration of the higher value, (2) their program to assess and control doses, and (3) procedures to control doses to be ALARA. These options used singularly or in combination coupled with process or operational modifications of these facilities is expected to provide sufficient flexibility to enable most uranium recovery facilities to comply with the provisions of the revised 10 CFR Part 20.

267 Pp+

RAMA ROVS F4

Section 20.303 [Reserved].

The former 0.1-rem "Reference Level" and the EPA Standard for Nuclear Power Operations that were in this section in the proposed rule are included as primary limits for members of the public in § 20.301 of the final rule.

Section 20.304 [Deleted] De Minimis Level and Collective Dose Evaluations.

<u>Comment</u>: Adoption of a threshold for calculating collective (population) doses. The proposed § 20.304 would have allowed licensees to disregard doses to individuals that were less than 1 millirem per year when evaluating collective (population or "person"rem") doses. A major criticism of this section was the narrowness of its scope. The section pertained only to a change in the calculational methodology for estimating collective doses and would not have permitted unrestricted release of any materials or equipment.

Most comments from people and organizations within the nuclear power and radiation applications industry favored this measure as an initial step toward developing more general "below regulatory concern" (BRC) levels. Several commenters thought that NRC acknowledgment of the concept of a "BRC" level was more important than the specific proposal to truncate collective dose calculations. Many commenters thought that a generic BRC level would limit unnecessary expenditure of resources that would otherwise have to be spent to control inconsequential risks.

There were also a number of comments that were not in favor of either the proposed collective dose cutoff or the more general application of the concept of "below regulatory concern." A few commenters expressed opinions that it did not appear feasible to arrive at a universal <u>de minimis</u> level, because the level that would appear to be truly insignificant to most people would be too low to result in any appreciable saving to the industry. There also were comments that noted that the proposed collective dose cutoff could cause large numbers of potential adverse health effects to be overlooked if they resulted from small radiation doses delivered to very large numbers of people. Many comment

ters, both pro and con regarding the adoption of a BRC level, thought that a threshold value for collective dose should also be developed. A few commenters noted that the focus of the more generic BRC concept tended to be for single licensees and that it might be necessary to consider the impacts from multiple licensees.

Many of the commenters who supported a generic BRC concept did not agree with the numerical value (0.001 rem per year) proposed for the cutoff, believing it to be too low. An explanation for this opinion was that if 0.001 rem represented an insignificant level of risk, then all larger doses might be perceived as representing "significant" levels of risk. A value of 0.010 rem was noted by several commenters as being a more suitable value and still represented an inconsequential risk.

Response: The Commission agrees that "Below Regulatory Concern" levels would be useful and has issued policy statements on the application of the comcept of "below regulatory concern" with regard to waste disposal ("Radioactive Waste Below Regulatory Concern," Federal Register of August 29, 1986 (51 FR 30839)) and a general policy statement on "Below Regulatory Concern" was announced on June 27, 1990, and was subsequently published in the Federal Register on July 3, 1990 (55 FR 27522). The general policy statement establishes the framework for the Commission to formulate rules and licensing decisions to exempt certain practices involving small quantities of radioactive materials from some or all regulatory controls. The BRC policy statement sets forth criteria for protection of both individuals (individual dose criteria) and population groups (a collective dose criterion).

In order to ensure that any computational changes reflect the policy that evolves from the effort to develop generic BRC policy, the Commission removed the threshold for truncating collective doses (§ 20.304) from Part 20 and has included such a threshold in the generic BRC policy statement. This deletion is also consistent with comments that noted that this section described a method for calculating a quantity (collective dose) that was not required to be calculated by Part 20 and comments that such details of calculations would be better in a regulatory guide rather than in a regulation. <u>Comment</u>: The accreditation requirement requires the use of a commercial dosimetry service.

Response: This is an incorrect interpretation of the dosimetry accreditation rule (52 FR 4601). That rule, which is incorporated into the revised Pari 20, states that the dosimetry processor must be accredited. It is possible for licensees that provide their own dosimetry services to be accredited.

<u>Comment:</u> Lack of specificity in monitoring requirements. Commenters noted that the monitoring requirements, both in the present Part 20 and in the proposed rule, were general and imprecise.

Response: Many portions of Part 20 are not very specific and detailed because Part 20 contains the NRC's general radiation protection requirements and applies to all classes of licensees, including large power reactors, universities, and medical institutions as well as small radionuclide and sealed source users. Because of this breadth of application, the requirements in Part 20 cannot be very detailed for any one type of facility. However, the requirements in Part 20 are designed to provide the framework for all licensees and to establish provisions that the NRC considers to be fundamental to basic radiation protection.

## Section 20.502 Conditions Requiring Individual Monitoring of External and Internal Occupational Dose.

<u>Comment:</u> Monitoring Thresholds. A number of commenters questioned the rationale for the lack of agreement of the thresholds in the proposed rule for monitoring external doses (10 percent of the annual limits) and for requiring monitoring of internal doses (30 percent of the annual limit). It was frequently mentioned that starting to require monitoring at 30 percent of the dose limit could result in overlooking doses of 1.5 rems (30 percent of 5 rems). The 1.5-rem value would have been above the limits for minors and for the embryo/fetus (0.5 rem) and was characterized as being a rather substantial fraction of the

deep-dose equivalent limit. In this connection, it was also noted that the possibility existed, when large external doses were expected, of exceeding a total effective dose equivalent limit of 5 rems because the licensee was not aware of the internal dose contribution.

Some commenters thought that the monitoring thresholds would be understood more easily if they were expressed as doses instead of percentages.

Response: The unequal thresholds for requiring monitoring of internal doses (30 percent of the dose limit) and external doses (10 percent of the dose limit) were originally set because of the difficulties in performing low-level bioassay analyses of alpha-emitting radionuclides at fuel fabrication and other facilities where actinides may be prevalent. (Bioassays for the radionuclides most commonly found at nuclear power reactors were viewed as generally being able to meet the 10 percent threshold set for external doses.) In situations such as bioassay for alpha-emitting radionuclides, it may be difficult to detect 10 percent of the ALI or 10 percent of the dose limit by bioassay measurements on excreta.

The monitoring threshold is a predetermined level of anticipated dose for carrying out bioassay procedures and does not represent a required level of detection sensitivity. If, by a reasonable analysis of the working environment, it appears that a worker is likely to inhale radioactive materials at concentrations that could produce an annual committed effective dose equivalent of 0.5 rem (10 percent of the 5-rem limit) or more, then that worker's intake should be monitored using measurements of exposure (e.g., estimates of DAC-hours based upon measured air concentrations) or intake (such as by whole-body counting or other bioassay technique) or by measurements of both exposure and intake. Whether the actual doses received were in excess of 10 percent of the limits could only be determined from these subsequent measurements.

The monitoring thresholds are specified as percentages of the dose limits rather than as doses because the thresholds apply to several different dose limits: the total effective dose equivalent, the eye dose equivalent, and the shallow-dose equivalent.
Final Rule. The threshold for monitoring internal doses has been dropped from 30 percent of the dose limit to 10 percent of the limit. This provides consistency in the internal and external monitoring requirements. The Commission acknowledges that, in some cases, particularly bioassay measurements of transuranic elements, it may not be feasible to actually confirm such levels by bioassay. However, the monitoring threshold is not a requirement on the capability of the measurement. Average airborne radionuclide concentrations and the expected time of exposure can be used to estimate radionuclide intakes and the need for bioassay or other monitoring methods.

The Commission intends to issue a regulatory guide on the procedures to be used in estimating committed effective dose equivalents and deep-dose equivalents and guidance on when they have to be summed.

<u>Comment:</u> Evaluation of radionuclide intakes for respirator wearers. KCM4 Several commenters mentioned that internal dose monitoring, such as bioassays. 157 should not be <u>required solely because respiratory protection devices were used</u>. The rationale given by the commenters was that the requirement provides a negative incentive for using respirators and is, therefore, counter to ALARA operating practices.

Response: The requirement (in § 20.502(b)(3)of the proposed rule) for bioassays for anyone using respiratory protection has been dropped. The Commission agrees that such a requirement might be a disincentive for using respirators as part of an ALARA effort. There is, however, a requirement (in § 20.703) for bioassays to be conducted, as appropriate, as part of a respiratory protection program. Whether bioassays are necessary for a particular individual will depend a whether that individual could have exceeded 10 percent of the annual limit on intake (ALI) or was exposed to airborne radionuclide concentrations in excess of the monitoring threshold. An evaluation of internal dose would be required if there were a potential for exceeding 10 percent of an annual limit on intake (0.1 ALI), whether or not a respirator is worn.

[Note: Because the requirement for performing lioassays for a particular KCAR individual has been separated from the wearing of a respirator, the concentrations to be used for evaluating monitoring thresholds are those of the ambient atmosphere before credit is taken for respiratory protective factors. One of the purposes of such bioassays is to confirm the effectiveness of the respiratory protection being provided. If bioassay were made dependent upon the corrected air concentration (after dividing by the protection factor), it would be equivalent to assuming that the intended protection factor was correct without further verification. ]

## Subpart G--Control of Exposure from External Sources in Restricted Areas

Sections 20.601, 20.602, and 20.603 Control of Access to High and Very High

Radiation Areas.

Comment: Inapplicability of requirements to nuclear power reactors. Many commenters indicated that the proposed requirements for control of entry into very high radiation areas could not be applied to nuclear power reactors because of the number and size of potential "very high radiation areas" and the physical inability to restrict access to these areas. Similarly, interlocks that can result in the withdrawal or cessation of the radiation source may be unworkable in nuclear power reactors. Several commenters proposed incorporating requirements for power reactors that are similar to reactor license conditions in reactor technical specifications.

Response: The Commission recognizes that the detailed requirements applic-KCAU able to large irradiators that were formerly in § 20.203(c)(6) should be in a 058 specific regulation dealing with these facilities rather than in Part 20. For this reason, these detailed requirements will be placed in a future Part 36 of Title 10 which is being issued for public comment and applies specifically to irradiators. At the time that that rule is made effective, the Commission will transfer these requirements from Part 20 to Part 36. In the meantime, the NRC staff will issue a regulatory guide that provides more specific detailed guidance or nuclear power reactors on high and very high radiation areas.

citation to

<u>Comment</u>: Choice of Dose Rate Defining a "Very High Radiation Area." Several commenters believed that the 500 rads per hour dose rate that defines a "very high radiation area" was too high, noting the proximity of this value to the median lethal dose  $(LD_{50}$  for acute radiation exposures. Alternative values, such as 1 rem per hour at 30 centimeters, were proposed.

Response: The seriousness of this dose rate was a factor in its adoption. The 500 rads per hour value appears in the previous 10 CFR 20.203(c)(6) as a criterion for additional access controls for irradiators (similar in scope to the requirements of § 20.603 in the final rule). However, the previous Part 20 did not use a unique designation such as the "very high radiation area" designation used in the proposed and revised Part 20 rules. The difference between the 1 rem per hour definition of a "very-high" radiation area used in reactor technical specifications and the 500 rads per hour definition used in the revised Part 20 is discussed in a regulatory guide currently being prepared.

<u>Comment</u>: Meaning of "direct surveillance." Several commenters thought that the term "direct surveillance" used in the proposed § 20.601 could be interpreted to require stationing an observer at the entrance to the "high" or "very high" radiation areas.

Response: The final rule permits "...continuous direct or electronic surveillance over a high radiation area that is capable of preventing unauthorized entry..." This removes the burden of having to station a person in or near a "radiation area," but requires interlocks or electronic locks so that the remotely located observer may prevent entry into the area when necessary.

Final Rule. The section on very high radiation areas has been divided into two sections. Section 20.602 provides a general requirement for restricting access to such areas. This general requirement applies to all very high radiation areas, regardless of the type of licensed operation, including those at nuclear power reactors. A second, more detailed, set of requirements applies only to large gamma irradiators. This section, § 20.603, restates requirements for irradiators that are in § 20.203(c)(5) of the present 10 CFR Part 20.

[7590-01]

Subpart H -- Respiratory Protection and Controls Restrict Internal Exposure in Restricted Areas

Sections 20.701 and 20.702 Use of Process or Other Engineering Controls and Use of Other Controls.

<u>Comment:</u> "Use of other controls." Commenters suggested that, if workers could be exposed to concentrations of radioactive materials greater than 1 derived air concentration, ALARA should be applied to the total of internal and external doses (to the total effective dose equivalent). It was noted that this condition was included in the Federal Guidance on Occupational Radiation Exposure.

Response: Modifications have been made in the final rule to permit ALARA considerations to apply to the total effective dose equivalent rather than just the internal dose portion.

<u>Comment</u>: Some commenters indicated that the use of respirators should be permitted even if their use would not be able to reduce airborne concentrations below 1 DAC. They noted that this would be consistent with the ALARA philosophy.

Response: Section 20.702 has been rewritten to clarify the intent that the concentration of 1 DAC is not a cutoff on the voluntary use of respirators but is intended to be the point where some corrective action (including, but not limited to, the use of respirators) by the licensee would be required when the use of ventilation and process controls cannot further reduce the airborne concentrations of radioactive materials.

Section 20.703 Use cf Individual Respiratory Protection Equipment.

<u>Comment:</u> The proposed rule permits low estimates but not high estimates of intake to be corrected. Commenters noted that the the proposed rule

(§ 20.703(a)(1)) was not balanced as correction of intake estimates based upon dividing DAC-hours by the respirator protection factor and was only permitted if the initial estimate was later shown (by bioassay results) to have been low.

Response: The rule has been modified so that corrected estimates of actual intake can be used in records in place of earlier estimated intakes, regardless of whether the change would result in an increase or in a decrease in the intake estimate.

<u>Comment</u>: NRC should provide a recommended minimum acceptable standard for determining an individual's physical fitness for respirator use. Part 20 requires that a physician determine that an individual worker is physically able to wear a respirator. NRC should, therefore, provide guidance to the physician on minimum standards for wearing respirators.

Response: The NRC policy is that the decision as to medical fitness has been, and continues to be, left to the physician; i.e., the medical doctor should decide what constitutes minimum health standards for respirator wearers. Furthermore, the requirements may vary, depending on the respirator used and physical situations, such as the type of work to be performed, which are outside the scope of Part 20. Licensees desiring more guidance should obtain ANSI Standard Z88.6(1984), "For Respiratory Protection -- Respirator Use -Physical Qualifications For Personnel," which was developed as an industry consensus standard that provides definitive guidance to "identify the responsibilities of the physician, the employee, and management in determining the employee's ability to use a respirator."

<u>Comment:</u> NRC should permit a health professional to certify physical capability to use a respirator rather than requiring a physician to perform each required certification. The proposed rule requires that a physician annually certify a worker's physical suitability for using a respirator. This should be broadened to permit any qualified health professional, acting under a physician's orders, to perform the actual certification rather than requiring a doctor to do this. Response: As noted in the previous response, the decision on the physical ability of an individual to wear a respirator is a subjective judgment that, in the Commission's opinion, requires the decisionmaker to have a medical degree. The Commission notes that this annual certification could easily be included in an annual physical checkup.

<u>Comment:</u> The selection of respirator protection factors based upon "average concentrations" and not "peak airborne concentrations" is an improvement. The proposed rule, unlike the previous Part 20, permitted protection factors to be applied to the time-averaged air concentration rather than the peak air concentration.

Response: Despite some favorable comments on this change, the Commission has determined that the use of the average airborne concentration may not provide an adequate margin for health protection and, in the final rule, has reverted to the use of the anticipated peak concentration.

Final Rule. The proposed rule has been modified to require a respiratory protection program when respiratory protection devices are being used to limit intakes, whether or not credit is taken for respiratory protection factors. Allowance has been made for use of respirators that do not provide protection factors that would keep exposures below the derived air concentrations if (and only if) such use would keep the total effective dose equivalent ALARA.

267

Sectic. 20.704 Further Restrictions on the Use of Respiratory Protection Equipment.

<u>Comment</u>: Section 20.704 should be deleted. This section, which states that the Commission may impose additional conditions on respirator use, is not necessary because § 20.1302 permits the NRC to place additional requirements on a licensee.

Response: Although the comments are correct that § 20.1302 gives the Commission general authority to impose additional requirements on licensees, the Commission believes that the restatement of this policy in a section pertaining specifically to respiratory protection is desirable. As noted by the comments, this section does not create any additional requirement not otherwise contained in the regulations.

Final Rule. The requirements contained in the proposed rule are retained.

### Subpart I -- Storage and Control of Licensed Material

# Sections 20.801 and 20.802 Security of Stored Material and Control of Material not in Storage.

<u>Comment:</u> Definition of "secure." Several commenters requested a definition of the term "secure," which they felt was vague and did not provide an indication of the required licensee action.

Response: The phrase has been rearranged and now reads "secure from unauthorized removal or access" similar to the wording in the previous Part 20. This should provide sufficient clarification of what was intended by "secure."

<u>Comment:</u> Unnecessary restrictions on research. One commenter thought that the requirement to secure small quantities of radioactive materials when they are not in use would interfere with university research.

Response: The Commission believes that locking radiotracer laboratories when they are not being used is a small nuisance compared to the consequences of unauthorized access to or theft of the radioactive materials, which could result in contamination of unrestricted areas or exposure of individuals, as well as having to report a loss of licensed material to the NRC.

237

## Subpart J -- Precautionary Procedures

Section 20.901 Caution Signs.

<u>Comment:</u> Black should be permitted as an acceptable color for the radiation warning symbol. Several commenters requested that the color black should also be allowed to be used on signs and for stenciling on packages. The fading of magenta inks in sunlight and the use of black for marking international shipments were cited as supporting this position.

Response: The Commission believes that, although the 'magenta-on-yellow" color scheme has provided a unique warning of possible radiation hazards, black-on-yellow would also be acceptable. The fading of the magenta color as cited above may reduce the visibility of the sign with time. Because of the cost impacts if existing warning signs had to be replaced, the Commission is permitting the use of black in addition to continued approval of magenta and purple, rather than as a required replacement.

Final Rule. This section has been modified to add black as an acceptable color for the radiation warning symbol.

Section 20.902 Posting Requirements.

<u>Comment:</u> The terms "Caution" and "Danger" are not used consistently. Commenters noted that "Caution" or "Danger" could be used on signs for "Radiation Areas," "High Radiation Areas," and "Very High Radiation Areas" despite the considerable variation in the hazards that might exist in these different areas.

Response and Final Rule: The Commission agrees that the terms "Caution" KCMU and "Danger" should be used in a more consistent manner. The final rule permits #65 only the term "Caution" to be used in "Radiation Areas." "Caution" or "Danger" may be used in "High Radiation Areas," since it covers a considerable range from

0.1 rem per hour to over 500 rads per hour. Only "Grave Danger" may be used in "Very High Radiation Areas." This should provide more emphasis on the use of "Danger," the importance of which might have been diminished by its prior applicability to the lower hazard "Radiation Area." "Caution" is inappropriate for use in "very high radiation areas" because of the potential hazard.

<u>Comment</u>: There should be a requirement to post all "restricted areas" whether or not it is a radiation or an airborne radioactivity area.

Response: The objective of posting is to warn personnel of a potential hazard. A "restricted area," per se, does not warrant such a warning. There is nothing to prevent a licensee from posting a notice designating a "restricted area," but such action is not required.

<u>Comment:</u> The definition of "airborne radioactivity area" would require tracking of employee "stay times" (time spent in the area). The second option to the definition of "airborne radioactivity area" would require performing surveys of airborne activity and tracking the time spent by workers in the area. The present rule would have only necessitated the survey.

Response: There are two alternative definitions of an "airborne radioactivity area"; only the second one would require consideration of stay times. This second option does not require posting in areas that have low occupancy times and airborne radioactivity concentrations between 0.3 and 1.0 times the applicable DACs.

<u>Comment:</u> Areas containing only noble gases should not require posting as "airborne radioactivity areas." The hazard associated with such areas is primarily from external radiation.

Response: The DACs in Appendix B that apply to noble gases (and define an "airborne radioactivity area") are based upon submersion doses; therefore, the relationship remains valid. It should be noted that, because some shortlived noble gases have particulate daughters (such as <sup>88</sup>Rb and <sup>188</sup>Cs), the warning denoted by posting as an "airborne radioactivity area" may still be required.

<u>Comment</u>: There is no evident need to post all rooms containing 10 times the Appendix C levels. The requirement to post a caution sign in rooms that store ten times the Appendix C concentrations is unwarranted. There was some concern noted that such posting could deter firefighters or other emergency workers from entering an otherwise safe area, and increased damages could result.

Response: Complete dispersion of ten times the Appendix C activities could produce air concentrations for some radionuclides in excess of the occupational DACs. For example, if ten times the Appendix C quantities were dispersed in a 1,000 cubic foot (10 ft. x 10 ft. x 10 ft.) room, the resulting concentrations would be 35 times the DAC for organic carbon-14, 58 times the DAC for cesium-137, about 18 times the DACs for iodine-131 and tritium (water vapor), and approximately 6 times the DAC for technetium-99m. These appear to be sufficiently large to justify a posting requirement, particularly to caution firefighters in case of a fire.

<u>Comment:</u> The posting requirement should not be applied to sealed sources, such as gauges. Posting the entrances to areas having radioisotopic gauges could require multiple postings in large buildings.

Response: Posting is only required at entrances to the room containing the source and only when the dose rate at 30 centimeters would exceed 0.005 rem (0.05 mSv) in any hour (§ 20.903(c)) unless areas outside the room warrant posting as "radiation areas" and are already posted.

Section 20.903 Exceptions to Posting Requirements.

<u>Comment</u>: The proposed rule omits the past exemption for posting rooms containing only packages prepared for transportation.

Response and Final Rule: The Commission believes that there should be posting of these areas because there is no restriction on the length of time that packages may remain in a room. If the packages contain only small quantities of radioactive materials, then posting of the room would still be exempted under the remaining exemptions. The term "prepared for transportation" does include packages that are intended to be carried in a "sole use" vehicle. Such packages are permitted to have higher allowable dose rates than those specified in DOT (or NRC) limits for general shipment.

Final Rule. The exception for posting areas containing packages prepared for transportation has not been reinstated.

<u>Comment</u>: The requirement for a person in attendance would be unworkable in a hospital. The requirement (in lieu of posting the room containing a radiotherapy patient) for a person in attendance in order to prevent entry was interpreted as requiring a 24-hour escort for each radiotherapy patient.

Response: The intent was to generally require posting of therapy patients' rooms. (As noted in one of the comments, the dose rate from patients even with diagnostic nuclear medicine treatments might exceed dose rates of 0.002 rem per hour.) The intent of "in attendance" would be satisfied by a duty nurse at a nursing station, providing that the station was in sight of the entrance to the patient's room.

Section 20.904 Labeling Containers.

<u>Comment:</u> There is no way to meet the requirement to label containers in some nuclear power plants or in hot cells. It is difficult to mark the detailed information on a container in some areas of a plant or in hot cells. Pesponse: Section 20.905 contains exceptions to the labeling requirements that take care of the problem noted by the commenter.

[Note: For the purpose of this section, "Mixed Fission Products" and "Fission and Activation Products" may be regarded as radionuclides, provided that the total activity is also specified. Designations as to the process stream or location sampled or type of sample (e.g., "primary coolant") may also be helpful as an additional designation of the potential hazard.]

Section 20.905 Exemptions to Labeling Requirements.

<u>Comment</u>: The proposed rule omits existing exemptions for packages containing only exempt quantities and those containing less than 10 mCi or less of tritium, iodine-125, carbon-14, and sulfur-35.

Response: While these sources pose little external hazard from gamma radiation, the quantities could be a potential internal hazard if the package were ruptured and the contents were released. Consequently, some warning remains appropriate.

<u>Comment</u>: The proposed rule omitted the existing exemption from labeling for packages labeled for shipment in accordance with DCT requirements.

Response and Final Rule: The exemption for DOT-labeled packages has been restored because the Commission agrees that the DOT labeling is sufficient to denote the presence of radioactive materials and provide an indication of any potential hazard. Quantities and concentrations not requiring DOT labels would not warrant an NRC labeling requirement. (See § 20.905(d).)

KCAN p70

237

#4

Section 20.906 Procedures for Handling Packages.

<u>Comment:</u> The requirement to monitor all packages is unnecessary. The requirement to monitor all incoming packages containing radioactive materials is unnecessary and in large installations creates a substantial monitoring burden.

Response: This requirement has been reevaluated and modified in order to reduce the burden.

Final Rule. Section 20.906 in the final rule requires incoming packages to be monitored when: (1) they are labeled as containing radioactive materials according to DOT regulations, or (2) when a package is damaged or leaking. The first provision would reinstate the exemption from monitoring for shipments of small quantities of radioactive materials that would not require DOT labeling.

<u>Comment:</u> The requirement to survey external surfaces of packages is innecessary. Several commenters with extensive experience in monitoring packages noted that external contamination was rarely if ever present and that wipe tests are time-consuming both to make the smears and to count them.

Response: Experience in the shipment of thousands of packages each year has been very good. However, potential problems with leaking packages during transit warrant continued monitoring upon receipt to ensure that leaking packages are found and reported. Appropriate action can then be taken to determine the extent of contamination in transport vehicles and storare areas in order to limit the consequences and avoid recurrence. However, an exemption from the contamination survey requirement has been provided for special form (sealed) sources that are being moved to and from work sites in licensee owned or operated vehicles. This partially restores an exemption from the package survey requirements in the existing Part 20 (§ 20.205(b)(iii)) for all special form sources. The Commission believes that restoring this exemption will not result in any additional hazard. An external radiation survey of the package is still required. The primary purpose of this external survey of sealed sources is to ensure that the source is still properly secured and shielded after transporting it.

Final Rule: The requirement to monitor external surfaces of packages has been retained and applies to the two classes of packages for which surveys are required (labeled "radioactive" and damaged or leaking). A partial exemption to sealed sources transported for field use has been reinstated because of the difficulty in making field measurements of surface contamination and because the transporting vehicle is not in general commerce.

237

#4

<u>Comment:</u> The requirement to monitor packages within 3 hours is unwarranted. This requirement would be difficult to meet for several types of licensees, some of which do not have a full-time health physics staff person.

Response: Licensees receiving labeled packages of radioactive materials to which this requirement applies are expected to have available persons who are qualified to perform such monitoring. However, the person monitoring the package need not be a board-certified health physicist.

Final Rule. The 3-hour period in the current Part 20 (§ 20.205(b)(1)(5)) has been retained except if the package is received after normal working hours.

## Subpart K--Waste Disposal

Section 20.1001 General Requirements.

<u>Comment</u>: Decay in storage as a disposal option. Many commenters noted favorably the addition of "decay in storage" as an allowed waste disposal option. Several commenters, however, did not believe that the option, as expressed in the proposed rule, was particularly helpful.

[7590-01]

267

ps

Response: Technically, the "decay in storage" option has always been available to a licensee since the license permitted possession of the radioactive materials and these materials naturally underwent radioactive decay. The option was formally included in the proposed and final rules because the list of disposal options is exclusive and there have been questions as to whether this was allowed under the previous Part 20. It should be noted that this option does not allow material that has "driayed in storage" to be released to unrestricted areas unless it meets the requirements of one of the other ailowed forms of waste disposal in Part 20, or the requirements of § 35.92, "Decay-in-Storage," of 10 CFR Part 35, or the specific requirements given in any NRC or Agreement State license conditions.

The NRC staff considered adding a separate "Disposal by Decay in Storage" option with specific criteria for unrestricted release of material after decay. These criteria are commonly included in source and byproduct material licenses. However, the provisions included in 10 CFR 35.92 and certain specific license rsecy-89conditions pertain to relatively short-lived radionuclides and are neither appropriate nor applicable to other classes of licenses, such as those issued under Part 50. Also, when evaluated for a specific licensed activity, it is possible to consider existing pathways of exposure and to establish specific criteria for decay.

General criteria in a rule would need to be sufficiently conservative to take into account all reasonably conceivable pathways, thereby reducing the applicable level from what would be permitted in a case-by-case evaluation.

Final Rule. The Final Rule has been modified to explicitly list "decayin-storage" as an authorized form of disposal. Section 20.1001 has been modified to incorporate the requirements that were in § 20.1002(b) of the proposed rule. These provisions require NRC licenses for persons who receive wastes containing licensed radioactive materials for treatment, for treatment or disposal by incineration, decay-in-storage, or disposal in facilities licensed under Part 60 or Part 61. Section 20.1003 Disposal by Release into Sanitary Sewerage.

Comment: Removal of allowance for dispusal of "dispersible wastes." A number of commenters felt that the restriction of wastes released to sanitary sewers to soluble wastes would have an adverse impact on certain licensees that, under the previous rule, had disposed of "dispersible" but insoluble radioactive materials. In particular, the practice was mentioned of shinding up animal carcasses with subsequent sewer disposal of the ground residue. This practice is permitted by the previous Part 20 but would not have been permitted under the proposed rule.

LCMU Response: In the final rule, the Commission has modified the conditions in the proposed rule for disposal of radioactive wastes into sanitary sewer systems so that "dispersible biological materials" may continue to be disposed of by release to sanitary sewers. This means of disposal is advantageous compared with other alternatives for disposal of this type of biological material.

The prohibition on disposal of insoluble materials via the sanitary sewer was intended to prevent disposal via sanitary sewers of material in which the radioactive material is primarily in an insoluble form. Such materials may accumulate in the sewer system, in the sewer treatment plants, and in the sewer sludge.

Final Rule: The final rule permits disposal into sanitary sewers of: (1) radionuclides in soluble form or (2) radionuclides in readily dispersible biological material, provided that the limits in Appendix B, Table 3, on the average monthly concentrations and the limits in § 20.1003(a)(4) on the total activity released annually are met. The revised rule no longer permits the disposal of nonbiological insoluble materials because of potential reconcentration of these materials in the sanitary sewer system, sewage treatment plants, and sewage sludge. This prohibition for insoluble materials is the reason why there are no values listed in Table 3 of Appendix B for insoluble materials.

75

267

237

p73

Comment: The rationale for the reduction in the limits for sewer disposal is not explained. The concentration limits for radionuclides released to sanitary sewer systems in the proposed rule have been reduced by a factor of 10 from the former rule. This reduction did not appear to take into account the dijution afforded from multiple users of the sewer system. Commenters indicated that they thought that this reduction would increase the amount of material that would have to be disposed of via a low-level radioactive waste burial site and could result in increased radiation doses to workers having to package this material.

Response: The assumption noted by many commenters that radionuclides dis- KCMM charged into sanitary sewer systems are not ingested is not necessarily true because water in large lake or river systems may be recycled. The dilution afforded by having multiple users of a sewer system can be offset in part because there can also be several users that discharge radioactive wastes into the same sewer system. The revised Part 20 rule permits a higher concentration limit for discharges into sanitary sewers than for other liquid effluent releases of radioactive materials, but has lower concentration limits than were formerly allowed for sewage. In view of past contamination incidents (involving cobalt-60 and americium-241) and the reduction in the dose limit for members of the public, the Commission believes that continuation of the higher limits is no longer desirable.

The NRC has under way a study of the dose pathways associated with disposal of radioactive materials via sanitary sewers. This study will help clarify the potential for human exposure.

Comment: The exemption on disposal of human excreta should be removed. Hospitals should have to comply with the same regulations as other licensees.

Response: Disposal into a sanitary sewer system (which was designed specifically to handle this type of waste) is the preferred method of disposal because of the other health considerations in handling human excreta in addition to radiation protection. This exemption is in the current Part 20.

8000/0

is mour

BIAT

Section 20.1004 Treatment or Disposal by Incineration.

<u>Comment:</u> Relaxation of specific NRC authorization for incineration. A number of comments questioned the need for the existing requirement that incineration of radioactive materials requires specific prior NRC approval (except for small quantities of tritium and carbon-14, which are specifically exempted). These commenters noted that the source of the released material (from an incinerator stack or from a fume hood vent) should not be the basis of requiring specific prior NRC approval of incineration while permitting general effluent releases.

Response: Relaxation of the prior approval requirement for incineration was considered in connection with the revision of Part 20. The requirement for prior NRC approval of incineration remains in the revised Part 20 because the *KCM44* acceptability of incineration as a disposal option, except for exempted quantities of radioactive materials, must be determined on a site-specific basis *p*74 considering: (1) incinerator design to safely dispose of hazardous materials, (2) the variable nature of the material to be burned both <u>in terms</u> of isotopic composition and activity, and (3) because many of these incinerators can be located in urban areas, special calculational methods may be required to assess doses to people located near these facilities.

Final Rule: Disposal by incineration still requires specific approval by the Commission (or Agreement State) whether done only for wastes from the licensed facility or whether done for wastes received from other licensees.

Section 20.1005 Disposal of Specific Wastes.

<u>Comment:</u> There should be a definition of ALARA for solid wastes. Many commenters suggested the need for ALARA or exempt quantities of radioactive material in solid wastes so that very low-level solid wastes could be disposed of without regard to their radioactivity.

[7590-01]

26

rewri ot sugges versit

NCMU

075

KCMU

EXAMPRA

Response: The Commission agrees that such levels would be useful and has developed a policy statement regarding levels of dose and risk that can be used to determine that specific practices involve radiation hazards that are Below Regulatory Concern (BRC). This policy statement was published in the Federal Register on July 3, 1990 (55 FR 27522). The BRC Policy Statement provides a comprehensive policy that will establish a disciplined and consistent framework for all future Commission exemption decisions. This includes potential application to rulemaking or licensing actions for disposal of slightly contaminated solid radioactive wastes. The Commission is developing a program for implementing the BRC policy separate from this Part 20 rulemaking.

# Section 20.1006 Transfer for disposal and manifests.

Comment: This section should not be in Part 20.

Response and Final Rule: This section is in Part 20 because it relates to the radiation protection aspects of low-level waste shipments.

# Section 20.1007 Compliance with Environmental and Health Protection Regulations.

Final Rule. This section has a counterpart in the present Part 20 and in the proposed rule (§ 20.1005) stating that meeting Part 20 requirements does not remove the responsibility of licensees, when disposing of licensed radioactive materials, from meeting the requirements of <u>other</u> applicable Federal, State, and local regulations applicable to toxic or hazardous wastes.

The advisory statement in the final rule has been expanded to cover all methods of waste disposal. This section of the rule is advisory and is not intended to imply that NRC will take enforcement action for violations of other environmental protection regulations issued under statutes other than the Atomic Energy Act.

[7590-01]

#### Subpart L -- Records

Standardization of Record Retention Requirements.

Final Rule. Records directly pertaining to effluents released to the frammy perced. general environment, waste diamonal, and doses received by individuals are to MCAM be kept until the "Commission terminates each pertinent license requiring the M76 record." Other record retention requirements in this subpart generally have been modified to be for 3 years after the record is made. This change is in conformance with the final rule published in the Federal Register of May 27, 1988 (53 FR 19240) on record retention requirements for other parts of the NRC regulations. This change provides for consistent record retention requirements throughout the NRC regulations in Title 10 of the Code of Federal Regulations.

## Section 20.1101 General Requirements.

<u>Comment</u>: The units used in records should be limited to those commonly in use: the rad, the rem, and the curie. Some commenters thought that the use of SI units (gray, sievert, and becquerei) should not be allowed.

Response and Final Rule: The Commission agrees that the use of "special units," the rad, the rem, and the curie, is preferable is time. This will avoid any difficulties arising from trying to implement both a new regulation and new units. This will reduce potential problems in records and reports that could result from some licensees using the "SI units" and some using the older "special units." The final rule requires the use of the "special units" instead of the "SI units." See the discussion of this topic under £ 20.4 Units.

Section 20.1102 Records of Radiation Protection Programs.

<u>Comment</u>: Added implementation burden associated with requirements for formal radiation programs. A number of commenters thought that the require-

PTY

ment to have a forma' ALARA program would result in substantial increased costs due to additional recordkeeping, procedural requirements, and quality assurance KEAN p 71 requirements.

Response: As discussed under § 20.101 these provisions have been modified KCMU to require ALARA as one part of a licensee's radiation protection program. The P76 adoption of requirements for licensees to have a formal radiation protection malition program was not intended to cause large implementation costs. Much of the cost KCMU associated with the recordkeeping requirements in the proposed rule was a result 1077 of the ALARA documentation requirements. These recordkeeping requirements have Frennits KCMU been reduced in the final rule by deleting specific reference to documenting ALARA actions. Specific types of records will be developed by each licensee as part of their radiation protection program. Therefore, this section contains general recordkeeping requirements associated with the radiation protection program.

Comment: The recordkeeping burden for small licensees requires a commitment of resources that is not commensurate with the risk. (In Section XXXVI of the proposed rule (51 FR 1121-1122), NRC specifically requested comments on the magnitude of the impact of the proposed rule on small licensees and requested suggestions on how these impacts could be reduced.) Quite a few commenters expressed their belief that the proposed rule will require more extensive monitoring and recordkeeping efforts than were required by the existing Part 20. Several commenters suggested that the NRC explore possible exemptions or exclusicns for academic licensees and other users of small quantities of licensed material. Other commenters expressed the view that the protection of public health for both the worker and the general public should be the same regardless of the size or economic resources of the licensee.

Response: Because of the changes to reduce the recordkeeping burden dis-RCMH cussed in response to the preceding comment and because the basic requirement NTT in § 20.101 calls for effort "... commensurate with the scope and extent of licensed activities ...," the Commission has not made further exemptions or exclusions from the recordkeeping requirements in this section for certain KCMU types of licensees.

Section 20.1104 Determination of Prior Occupational Dose.

<u>Comment:</u> Medical and academic licensees would have difficulty in complying with the requirement to determine prior exposures. The transitory nature of personnel in these facilities would make meeting these requirements very costly. Doses to employees are small fractions of the limits so that such costs would be difficult to justify.

Response: The requirement to determine dose received in the current year implements the annual dose limits. The requirement to attempt to obtain records of lifetime cumulative doses follows one of the provisions of the guidance to Federal agencies on occupational radiation protection. Efforts to obtain prior exposure histories are only required for workers who are required to be monitored under § 20.502. Determination of prior doses received during planned special exposures or doses in excess of the annual limits are required only for workers who will be used in planned special exposures.

<u>Comment:</u> The recording of "fictitious" radiation doses should be avoided. The present and proposed rules state that, when information is not available regarding the dose received for a specific period, the licensee should assume that the dose received was at the dose limit. Several commenters thought that this was inappropriate. Some commenters mentioned that this practice might be nonconservative as it would tend to overestimate the dose used in any epidemiological studies of radiation effects, thereby resulting in an underestimate of the risk associated with a unit radiation dose.

Response and Final Rule: The final rule has been modified so that it does not require any assumed dose value to be recorded in case of incomplete prior dose histories. Only the lack of data must be recorded for periods where there is no information. However, for the current year, where there are misting data, an assumption is to be made for establishing administrative controls: the portion of the dose limit remaining for the current year is reduced by 1.25 rems for each calendar quarter for which information is missing. (The values for other limits, such as the shallow dose equivalent or eye dose equivalent

[7590-01]

10 80

should be reduced by a one-quarter of their annual limit for each unreported quarter.) The licensee must note the absence of this information on the employee's record but should not enter the assumed dose value as part of the employee's permanent dose record. For example, an employee who had prior radiation working experience joins Company X on July 1st but does not have the prior radiation records. This employee's dose should be limited to 2.5 rems (5 rems - 2(1.25) = 2.5 rems) until such time as the records are obtained.

<u>Comment</u>: There should be a quarterly dose limit to cover workers whose records have not been received from a former employer. A 0.5-rem dose might be appropriate for this purpose.

Response: If data were missing for all four quarters (employment commenced late in the fourth calendar quarter), then the employee could not be exposed to radiation above the level for a member of the general public. However, this limit is 0.1 rem per year not 0.5 rem.

Section 20.1105 Records of Planned Special Exposures.

See discussion under § 20.1204.

Section 10.1106 Records of Individual Monitoring Results.

<u>Comment:</u> NRC should not require reporting or recording of cumulative dose. A number of commenters noted that the ICRP system of dose limitation is based [as one of the principles] on controlling annual doses. Consequently, they questioned the need for recording cumulative doses.

Response: Although the commenters are correct that there is no longer a cumulative dose restriction in Part 20 (such as the former 5(N - 18) formula), the Federal Guidance on Occupational Exposure (see Section II.D.) contains a recommendation that cumulative dose records be maintained and provided to the worker.

P267

pb

<u>Comment</u>: The proposed rule does not require recording annual doses as listed in the 1987 Federal occupational guidance.

Response: "Annual dose" is specified in the guidance and is the same as the annual deep dose equivalent for external doses. However, "annual dose" is not required to be recorded by the revised Part 20 for internal doses. This is consistent with an exception noted in footnote 5 to the Federal guidance (Federal Register of January 27, 1977; 52 FR 2832):

> "When these conditions on intake of radioactive materials have been satisfied [i.e., meeting the committed dose limits], it is not necessary to assess contributions from such intakes to annual doses in future years, and, as an operational procedure, such doses may be assigned to the. year of intake for the purpose of assessing compliance."

Paragraph 20.1106(b) -- See discussion under § 20.1204.

<u>Comment</u>: The recordkeeping requirement in the proposed § 20.1106(d)(2) would require that all records begin at the beginning of a calendar year. This would create an unnecessary hardship on dosimeter processors since they could not stagger the dosimeter changeover schedules to provide a more uniform work-load distribution.

Response and Final Rule: The term "year" replaces the term "calendar year" in § 20.3 and permits the licensee to define the year to begin anytime in January. A licensee may change the starting date, provided that the change is made at the beginning of the year and provided that no day is omitted and no day is included twice in consecutive years.

<u>Comment</u>: The requirement in § 20.1106(e) for each licensee to keep a copy of the dosimeter processor's accreditation certificate creates an undue burden

on commercial processors. Commercial dosimeter processors would have to print and distribute thousands of their certificates so that each user had a copy.

Response: The proposed rule contained a requirement for the licensee to maintain a copy of the dosimetry processing accreditation certificate issued to the processor providing dosimetry services to the licensee. This requirement, which was in the proposed dosimetry accreditation rule, was considered unnecessary and was dropped as a requirement in the final version of that rule. Consequently, it has been deleted from revised Part 20. Licensees who provide their own dosimeter processing services do have to maintain a copy of their NVLAP accreditation certificate for inspection.

<u>Comment</u>: The NRC should consider a "traveling dose history" that can move with the worker. This was suggested, particularly for transient workers and for workers employed concurrently by two employers. The master record will reside with the current employer and would have to be transmitted by the worker to a new employer.

Response: Because the NRC can only regulate its licensees and has no authority over individual workers, the recordkeeping and transmittal requirements for dose histories are placed on the licensee and not on the worker. The concept of a "passport" incorporating security and dosimetry data has been used successfully in Japan and elsewhere. The requirements for determination of prior exposures that are in § 20.1104 provide a similar record to a "moving history," but this would have to be updated by each new employer.

Concurrent employment with two (or more) employers requires special attention so that the combined doses from both employers would not exceed the dose limits. When two employers are aware of such concurrent employment, the simplest expedient to achieve this goal is for them to agree that the dose limit they will use for this employee in the individual programs is less than one-half of the NRC dose limits (the fraction of the dose limit allocated to each employer might also be determined on the basis of the relative amount of time worked at each location).

RAU R83

Comment

The problem of dual employment is more of a problem when the employee has not confided in the employer. The licensee is required to ascertain the employment and dose record for the current year for new employees (§ 20.1104). If the employee deliberately falsifies this information, the licensee would not know of concurrent employment and the licensee would not be penalized for combined doses from both employers that exceeded the dose limits. If a current employee takes on additional outside radiation work without informing the employer, the employer should not be penalized. It should be noted that, under the new reporting requirements in § 20.1206, individual dose records will be required to be submitted to the NRC for all workers for those categories of licensees formerly subject to § 20.407, including nuclear power reactor lice nees.  $\int \int \rho dr \chi C M H$ 

Final Rule. Section 20.1106 has been modified in order to separate the requirement for keeping a record from the format of the record. A clarification has been added that the dose information on an embryo/fetus be kept with the mother's dose record. E

Section 20.1107 Records of Dose to Individual Members of the Public.

<u>Comment:</u> Reporting requirements for exceeding "reference levels." The proposed rule contained requirements for reporting exposures in excess of the "reference levels" for doses to members of the general public. Many commenters thought that this was excessive because this was not an actual regulatory limit.

Response: The 100 millirems per year "reference level" for doses to members of the general public in the proposed rule has been incorporated as the dose limit in the final rule for members of the general public so that the associated recording and reporting require into now pertain to a regulatory limit.

Final Rule. Section 20.1107 has been broadened in scope from "effluents" to pertain to records of all estimates of doses received by individual members of the public. Doses to members of the public are calculated from measurements ED of direct radiation, and radionuclides in effluents, and the environment rather

than as measurements pertaining to a a particular individual. This difference KCAK in method of dose assessment from the more direct measurements used for occupational exposure does not imply any lessening of requirements for keeping adequate records of effluents released to unrestricted areas.

Section 20.1108 Records of Waste Disposal.

Final Rule. Section 20.1108 is unchanged from the proposed rule.

# Section 20.1109 Records of Testing Entry Control Devices for Very High Radiation Areas.

Final Rule. Section 20.1109 contains an addition to the proposed rule for keeping records of tests of entry control devices for very high radiation areas. This addition is based upon a requirement in § 20.203(c)(6) of the present Part 20.

Section 20.1110 Form of Records.

<u>Comment</u>: NRC should allow computerized recordkeeping systems to handle records. A few licensees suggested that NRC allow "electronic" recordkeeping systems and provide guidance for their use.

Response: The Commission agrees that there is great value in the use of "electronic media." There are a growing number of licensees that are using computer information networks for retaining and transmitting radiation dose histories and other worker-related information among different facilities.

Final Rule. The final Part 20 expands the definition of "record" to include "electronic media." The use of electronic media requires authentication and the prevention of alteration or loss of the records. As with existing requirements for paper records, the electronic media must be capable of producing a legible copy of the record.

#### Subpart M--Reports

Section 20.1201 Reports of Theft or Loss of Licensed Material.

<u>Comment:</u> The term "substantial exposure" in § 20.1201(a) should be defined. The requirement to report the loss of radiation sources capable of producing "substantial exposure" needs to be more precise.

Response: The term "substantial exposure" has been replaced by a specific designation of the activity of lost source that requires immediate reporting to the Commission. This quantity is 1,000 times the Appendix C activity levels. For sealed sources of cobalt-60, cesium-137, or iridium-192, this activity would produce a dose of around 25 rems at 1 foot over a 30-day period (25 rems is the worker dose that requires immediate Commission notification). Although somewhat similar doses may be projected from inhalation of dispersible material, the exact exposure conditions would have to be known in order to make a valid activity-to-dose relationship.

Final Rule. The final rule now contains specific activity criteria for immediate reporting rather than the vague term, "substantial exposure."

<u>Comment</u>: The quantity for reporting the loss of a source is too low (too high). The reportable quantity of ten times the Appendix C activity values appeared to some commenters to be overly restrictive; others thought that all lost or missing radiation sources should be reported.

Response: The specified 30-day reporting level is a compromise between having higher reporting levels and having a requirement that all lost or missing sources be reported. Further, the report permits review of the circumstances involved including any lack of security of materials or weakness in the licensee's control program that may be unrelated to the sources being stolen or lost, but may be pertinent in avoiding recurrent theft or loss. Final Rule. The activity levels in Appendix C for some long-lived radionuclides have been increased from those specified in the proposed rule. This increase means that the loss of milligram quantities of natural uranium will no longer have to be reported.

<u>Comment</u>: A 30-day telephone report should not be required concomitant with a written report. Sections 20.1201(a)(1)(ii) and 20.1201(b) both call for a 30-day report; the first requires a telephone report and the latter section requires a written report.

Response and Final Rule. The rule has been revised to clarify that the written reports required by § 20.1201(b) are to be submitted within 30 days of the telephone notification required by § 20.1201(a), rather than both being within 30 days of learning of the theft or loss.

<u>Comment:</u> The rule should provide for a "grace period" before having to report a lost source to NRC. Commenters noted that, in many instances, a source "lost" in transit eventually turns up. Some specified period, such as 7 days, should be permitted before a "lost" source would have to be reported to the NRC.

Response: The rule contains two notification requirements: the one for immediate notification only pertains to those sources that exceed 1,000 times the Appendix C activity levels. The second notification requirement pertains to sources that exceed ten times the activity levels in Appendix C and that are still missing after 30 days. This provides a grace period of 30 days for reporting the loss of most sources.

Section 20.1202 Notification of Incidents.

<u>Comment:</u> The requirements for immediate notification of NRC are too low. Some commenters thought that the doses associated with the requirements for

[7590-01]

×cmq p 84

KCM4 DEG

immediate reporting to NRC (five times the respective annual limits) would not produce any discernible harmful effects to the individual to warrant immediate reporting.

Response: Doses of the order of 25 rems (5 times the 5-rem annual dose limit) can produce discernible biological effects in the body in the form of chromosome aberrations and changes in the white blood cell populations. Although the majority of these effects are temporary, they could be discerned. However, irrespective of the potential for discer<u>nible</u> effects, doses at these levels represent a major breakdown in the licensee's control over the radicactive material, and the Commission believes that it is important that NRC be promptly notified so that it can take actions, if necessary, to limit further consequences.

Final Rule. The final rule retains the previous reporting requirement.

Comment: Immediate reporting should be required if there is any potential for dose reduction. The Environmental Protection Agency (EPA) suggested that incidents always be reported if there is the potential for significantly reducing public doses through protective actions. It is believed by the EPA that this would occur at doses significantly less than those of the proposed reporting criteria.

Response: The incident reporting levels and response times have been selected to limit attention to the more potentially serious events without the entire NRC emergency response network being activated unduly for events involving only small quantities of radioactive materials. For most cases, it is expected that the licensee would have initiated any necessary remedial measures.

Comment: Immediate and 24-hour notification requirements should be suspended in the case of a declared emergency at a nuclear power plant. Commenters felt that any emergency at a nuclear power plant will involve onsite NRC staff and that stopping emergency activities to make the Part 20 incident reports could be a burden on the licensee.

N88

Response and Final Rule. These reports are particularly easy to make for nuclear power reactors (the reactor o trator merely has to pick up the dedicated NRC telephone line to get the NRC Operations Center). There are certain functions of the NRC (such as activating the NRC Incident Response Plan) that require that NRC be notified; therefore, this notification requirement has been retained.

Section 20.1203 Reports of Exposures, Radiation Levels, and Concentrations.

Comment: There is no requirement for reporting doses that exceed the limit for protection of the embryo/fetus in § 20.208.

Response and Final Rule: A requirement has been added to the final rule in El 20.1203(a)(2)(iii). El KCMM

<u>Comment</u>: The identifiers required in § 20.1203(b)(2) for the embryo/fetus should be those of the mother. As the fetus has no date of birth and no Social Security account number, those of the mother should be used.

Response and Final Rule. A footnote to this effect has been added to § 20.1203.

<u>Comment:</u> Reports of exceeding the 0.1-rem reference level should not be required. A number of commenters noted that the 0.1-rem "reference level" was not a limit and, therefore, exceeding it should not necessitate a report to the NRC.

Response: As a result of changes in the ICRP interpretation of the 0.1-rem level and the former 0.5-rem dose limit, the 0.1-rem level is now the recommended limit. Consequently, 0.1 rem is the primary limit applicable to members of the general public and reports are justified when it is exceeded. <u>Comment</u>: Smaller licensees, such as nuclear medicine facilities, should be exempted from the reporting requirements of § 20.1203. Licensees are required to report concentrations in unrestricted areas that exceed ten times any applicable limit in the license. Because some nuclear medicine units use the room air volume for dilution, calculated concentrations exceeding ten times the Appendix B limits might frequently occur. This would require either more frequent reporting to NRC or use of more sophisticated atmospheric dispersion models.

Response: The reporting requirements are very similar to those in the previous Part 20. Part 35 of the Commission's regulations, which deals with medical applications, covers the medical use of noble gases and in § 35.205(a) limits airborne concentrations to the 10 CFR Part 20 Appendix B concentrations. Experience has not indicated large numbers of reports of such limits being exceeded.

Section 20.1204 Reports of Planned Special Exposures.

Comment: The NRC should have to pre-approve planned special exposures. Mores

Response and Final Rule: The Commission has decided not to require pre- 92 approval of planned special exposures. This is, in the Commission's view, con- per sistent with the 1987 Federal Radiation Guidance for Occupational Exposures KCMW because detailed requirements are prescribed in Part 20 for the use of planned p92 special exposures.

<u>Comment:</u> The licensee should not have to file a separate report to NRC for planned special exposures. Several commenters objected to having to file these separate reports each time a planned special exposure is carried out. This was viewed as representing a reporting requirement for operating within the NRC regulations. It was suggested that this information be included in the employee's records without reporting to NRC.

at CA: 4

190

Response: Because of the newness of the concept, the NRC wishes to monitor carefully the use of the planned special exposures. Further, while the planned special exposures are included in the final rule, the use of this concept does represent a situation in which the licensee is operating of side the normal dose limits and of which the Commission should be aware.

<u>Comment</u>: Period for reporting planned special exposures. Several commenters noted that the 15-day period for reporting planned special exposures is shorter than the 30-day period usually allowed for similar reports.

Response: The reporting period of a planned special exposure has been increased from 15 days to 30 days to be more consistent with other reporting requirements.

Section 20.1206 Reports of Individual Monitoring.

<u>Comment:</u> Could the requirement for the reporting of individual exposures be construed as an invasion of privacy? Some commenters believed that requiring the reporting of individual doses rather than a statistical summary might constitute an invasion of personal privacy.

Response: The Commission does not believe that submission of individual dose data constitutes an invasion of privacy. Such data have been reported to the NRC routinely in the termination reports for some time. Such information will be protected in accordance with the Privacy Act and will be restricted, as it has been in the past, to use by NRC officials, NRC contractors, or qualified scientific invest gators. Instructions on protecting this information appear in § 20.1106(d).

<u>Comment</u>: If the radiation exposure data are collected into a central repository, would the the be the proper place for it? One commenter felt that the radiation exposure data might be better maintained by an agency whose charter encompasses the analysis of the data for estimates of risk.

9%

Response: Arguments might be made for other agencies having the lead role in the storage and analysis of those data; however, it is the NRC that has the statutory authority to require that these data be collected. Although the Part 20 recordkeeping requirements are intended primarily to fulfill NRC's information needs for regulation, the NRC has continuing contacts with agencies that have expertise in conducting epidemiological studies (such as the National Cancer Institute of the National Institutes of Health and the Office of Health and Safety of the Department of Energy) to ensure that the Part 20 reporting and recordkeeping requirements do not lose information that would be vital to carrying out studies of this type.

<u>Comment:</u> The total collective (person-rem) dose should be reported. It was felt by one commenter that NRC should require the total collective dose to be reported so that the numbers used in NUREG-0473 (NRC's annual summary of occupational radiation doses) will be the same as those calculated by the licensee.

Response: The reason for a possible discrepancy between a licensee's estimate of the collective dose to workers and the estimate published by the NRC has been that the licensee may sum the actual individual doses and the NRC estimate is based upon the statistical summary rather than the actual individual dose reports. Such differences should be reduced in the future because NRC will be also be using dose information for individuals. The final rule requires licensees who previously submitted the dose summaries to report the individual dose data to NRC. Both collective dose calculations should then be using the same data base.

<u>Comment</u>: The termination report required in § 20.1207 should (cr should not) be replaced with an annual report for all personnel monitored. Some commenters felt that an annual report just to the NRC should replace the present requirement for a termination report. Other commenters felt that annual reports to the NRC of doses to individuals constituted a considerably larger burden than did a statistical summary. Some commenters, who disagree with filing an annual

report to the NRC, were in favor of giving such an annual dose summary to the worker. Other commenters suggested that all licensees be required to submit an annual report to NRC on each monitored individual.

Response: The reporting of individual monitoring data will help track doses to individuals who are exposed at several facilities during any given year and whose total dose would be underreported by statistical reports prepared at each work site. Such information is shown at the present time only by analysis of the termination reports.

Licensees who were previously required to file both annual statistical summaries and termination reports with the NRC will, instead, submit annual dose reports to NRC for all workers for whom monitoring was required under § 20.502. A copy of the annual report to NRC could also be given to the individual worker in order to satisfy the revised reporting requirement in § 19.13 of 10 CFR Part 19. Although this may entail some additional burden to licensees, the use of "electronic media" for recordkeeping might in fact reduce overall costs. It is intended that large employers (such as nuclear power reactor licensees) would submit an electronic copy of their dose reports in a prescribed format to the NRC in lieu of paper copies of individual records.

Section 20.1301 Applications for Exemptions.

Comment: NRC should make the issuance of exemptions a matter of public record. Several commenters felt that the issuance of any exemptions under this section should require public notice and comment. The EPA stated that exemptions could adversely affect its ability to control radionuclides under the Safe Drinking Water Act.

Response: The NRC has issued few exemptions under this longstanding provision and has not exempted anyone from the dose limits for a worker or for a member of the public. <u>The Safe Drinking Water Act was not intended to control</u> <u>effluents and, although radionuclide concentrations at downstream water supplies</u> are routinely calculated as part of licensing evaluations, the licensee must

KCMU

meet the Part 20 concentration limits at the effluent release point, not at the drinking water intake after dilution occurs.

### Appendix A

<u>Comment</u>: The protection factor for air-purifying respirators with particulate elements is too low. The listed protection factor for air-purifying respirators with particulate filters is 50, whereas both ANSI Z88.2 and the OSHA regulations in 29 CFR 134 use 100.

Response: The NRC never endorsed ANSI Z88.2-1980, whereas the OSHA regulations generally follow ANSI standards. The current NRC-allowed protection factors (PFs) are based upon research conducted by the Los Alamos National Laboratory (LANL). These recommendations included a PF of 50 for full face respirators, based on \_perimental data on actual testing of personnel using respirators under carefully controlled conditions. In actual use, there is essentially no difference between a PF of 50 versus a PF of 100, so that there should be little or no real impact on field use of respirators or on operations at nuclear facilities that would result from using the higher protection factor.

<u>Comment:</u> Several respiratory equipment specifications in Appendix A should be applicable only for areas that are "immediately dangerous to life and health." Footnotes "h" and "i" contain specifications for air flow rates and flow calibration and a requirement for standby rescuers to be available when using supplied-air suits. These were felt to be unneeded considering that if the air flow failed, the person could withstand a small exposure to the airborne radionuclides while exiting the area after removing the protective hood.

Response: The supposition that conditions "immediately dangerous to life and health" do not exist is not always correct. Failure of an airline in supplied-air suits may be considered as "immediately dangerous to life and health" because there is an acute danger of suffocation if the air supply is interrupted
and the hood cannot be removed by the wearer. Rapid recovery of and assistance to the individual in the supplied-air suit necessitates the presence of a preequipped rescuer.

#### Appendix B

<u>General comments</u>. Most of the comments from radiation protection professionals favored the adoption of the ICRP-26/ICRP-30 annual limits on intake and the derived air concentrations. Comments from private citizens were against adoption of the ICRP values because the majority of the values would increase (as stated in Section XXIX of the proposed rule, 51 FR 1120).

Response: From an occupational protection standpoint, the changes that result from adoption of the ICRP risk-based approach lead to higher limiting intake values than in the previous Part 20. These increases result from the increase in the allowable ceiling for organ doses. The values that served as the basis for calculating the concentration limits used in the former Part 20 were organ dose limits of 5, 15, and 30 rems. The new concentration limits are based upon the effective (weighted) organ dose or upon the nonstochastic limit that forms an organ dose ceiling when the stochastic risk is not limiting. These changes increase the limiting annual organ doses (when only one organ is irradiated) for those doses that are limited by the stochastic (effective dose) limit from 5 rems to 20 rems for the gonads, from 15 rems to 32 rems per year for the breast, and from 15 rems to 42 rems for the lung. Limiting doses to other organs increase from the former 15- and 30-rem values to the 50-rem nonstochastic limit.

The former ICRP-2 "critical organ" concept based the limiting intake upon controlling the dose rate to the organ receiving the highest dose rate (the "critical organ"). The doses to organs other than the critical organ did not have to be evaluated, even if these doses were close to the estimated dose to the critical organ. The new ICRP-26/30 system evaluates the doses to the major organs and the six remaining organs that receive the next highest doses. These doses are then multiplied by the appropriate weighting factors ( $w_T$ ) and are sum-

med to give a risk-weighted "effective dose." The concentration limits that are based upon this newer ICRP approach reflect the doses to all principal organs that are irradiated, not just the one organ that receives the highest dose as was done in the former Part 20.

Many of the comments from private citizens do not appear to reflect the proposed rule because many of the comments objected to raising the limits for radionuclide concentrations applicable to the general public. As noted in the discussion of Appendix B in the notice of proposed rulemaking (Section XXIX, 51 FR 1119-1120), the concentration limits for members of the public were based upon a "reference level" dose (now the dose limit for members of the general public) of 0.1 rem per year and incorporated an additional factor of 2 reduction (Proposed Appendix B; 51 FR 1145) for age-dependency and combined air and water intakes. Thus the concentration limits for the public reflect a reduction in their basis from a whole-body annual dose of 0.5 rem in the former Part 20 to 0.05 rem in the proposed and final rules.

The concentration limits for individual radionuclides may be higher or lower for members of the general public in unrestricted areas in the final Appendix B than in the former tables because of changes that occurred in the intervening 25 years in the metabolic and other parameters used to calculate internal doses. These changes are reflected in ICRP Publication 30 and its supplements and amendments. However, these changes are a result of changes in the scientific techniques and parameters used in calculating doses and do not reflect an increase in the allowable dose limits, which, in reality, have been decreased in the revised Part 20.

<u>Comment:</u> NRC should consider deleting Table 2 from Appendix B. The concentration limits in Appendix B do not provide adequate protection of children and infants because they do not take into account age-dependency in a proper manner. Compliance with the dose limits, rather than with these concentration limits, should be required.

97

Response: The use of the effective dose equivalent concept reduces the importance of age-dependent intake-to-dose factors. Age-dependency is of primary importance in calculating organ doses. Those organs for which age-dependency is important, such as the thyroid gland, are of lesser importance because of lower  $w_T$  values (for the thyroid, for example,  $w_T = 0.03$ ) used to calculate the effective dose. A factor of 2 is included in the calculation of concentration limits for release to air and water, which, in part, accounts for age-dependency. In addition, the Commission believes that there is a lack of detailed age-dependent metabolic data for all but the most common radionuclides that will inhibit such attempts to increase the precision of the dose estimates.

Many smaller licensees routinely use concentrations and the Appendix B tables in order to demonstrate compliance. The use of concentration limits for determining compliance is a well-established practice that is economical for many of the smaller licensees. Despite the growing availability of simplified dose assessment models, the Commission is continuing to accept the use of concentrations to demonstrate compliance with the dose limits.

<u>Comment:</u> The Appendix B tables fail to account for the chemical toxicity of natural and low-enriched uranium. This fails to take into account the possible kidney (renal) damage associated with the chemical toxicity.

Response: There is a separate limit for uranium intake that is based upon the chemical toxicity. This limit was expressed as footnote 3 to Appendix B, page 1199 of the January 9, 1986 notice of proposed rulemaking and also as § 20.204(i) on page 1131. In the revised rule, it still appears as footnote 3 in Appendix B, but the limit also has been moved up in the text to the section on dose limits and now appears as § 20.201(e).

<u>Comment:</u> The limits for occupational and nonoccupational exposure to radon-222 and its particulate daughters do not appear to be consistent with the airborne concentration limits for other radionuclides in terms of risk.

98

Response: The occupational concentration limits for radon-222 are based on the existing Federal guidance, which is 4 WLM (4 Working-Level Months) per year. The annual limit on intake (ALI) is stated as 100  $\mu$ Ci or 4 working-level months. The derived air concentration (DAC) in Part 20 for occupational exposure to radon-222 of 3 x 10 s is equivalent to 0.33 working levels (this equivalence is also given in the Appendix B table). The concentration limit for members of the general public is a factor of 300 lower and, like the other airborne concentration limits, represents an effective dose of 0.05 rem per year.

<u>Comment</u>: Concentration limits for tritium omit chemical forms other than for tritiated water vapor.

Response: As there is expected to be no occupational intake via oral ingestion, and most of the commonly-used organic tritiated compounds are not volatile, inhalation and transpiration through the skin are the principal pathways of exposure. Different intake limits would apply to hydrogen gas (HT or  $T_2$ ) and tritiated water vapor, HTO. The HT or  $T_2$  gas is rapidly converted to HTO by isotopic exchange and oxidation (both in air and in the body) so that specifying a submersion dose limit for HT would understate the actual radiological impact. Comparison with other derived limits for other chemical forms shows that the use of the concentration limits for HTO provides an adequate level of protection for most of the other chemical forms.

<u>Comment:</u> No concentration limits are listed for natural thorium. There are limits for natural uranium, but corresponding concentration limits for natural thorium are not given. The isotopic composition of thorium can vary somewhat with different ores and with different times after chemical separation.

Response: <u>A licensee should</u> use the thorium-232 value or if a more pre-  $\mathcal{K}$ CAMM cise value is desired, use the procedure for mixtures in Appendix B applied to  $\rho 98$  the actual isotopic concentrations present.

Comment: The derived air concentrations for the general public are not always 0.1 times the occupational values.

Response: The limits for the general public are calculated solely from the stochastic risks. This differs from ICRP, which would use a "capping" organ dose limit of 5 rems (0.1 x the nonstochastic limit of 50 rems) in deriving the organ dose limit for organs that are limited by the nonstochastic risk. If there is a threshold for nonstochastic effects for the worker at 50 rems, it would also apply to a member of the public. Rather than applying a factor of 10 reduction to a nonstochastic value, the limiting stochastic (effective) dose was used to calculate the concentration limits for the general public. Values are not based on the nonstochastic risk for members of the public, even if they were the basis for the calculation of the DACs and ALIs for the worker. This difference in method of calculation accounts for the lack of a consistent ratio between worker DACs and effluent limits for the public.

#### Appendix C

Comment: The reduction from 100 µCi to 0.001 µCi for thorium values will corrected re posting of areas where thoriated-nickel machine require posting of areas where thoriated-nickel machine parts are used.

Response: On the basis of specific activity considerations, the existing 100 µCi limit has been retained for long-lived radionuclides (half-lives longer than 108 years) such as thorium-232, which would require several grams of material to produce the stated activity level. Because this is based on halflife, two isotopes may be treated differently, e.g., uranium-235 which does not meet the half-life criterion has an Apperdix C value of 0.001 µCi, and uranium -238 which does meet the criterion has a value of 100  $\mu\text{Ci.}$ 

Appendix D Contains the NRC Regional Office addresses and telephone numbers.

#### Appendix E [Reserved]

Final Rule. The calculational guidelines and equations that appeared in Appendix E are being incorporated into a regulatory guide on summation of internal and external doses. This will make it easier to revise and clarify the calculational methods without having to resort to formal rulemaking. (Note: NRC routinely issues regulatory guides for public comment before making them final.)

#### Appendix F

[Note: Appendix F is derived directly from requirements inserted by the Part 61 rulemaking proceeding on low-level radioactive waste disposal sites. These requirements were in § 20.311 of the existing <u>10 CFR Part 20</u>. Because these requirements are relatively recent, they were not modified in the Part 20 revision. The Commission is considering revisions to the manifest requirements in a rulemaking separate from the Part 20 rulemaking.]

Appendix G

No comments on Appendix G were received.

### VII. Conforming Amendments

Accompanying the revised rule are amendments to other parts of Chapter I that update citations to 10 CFR Part 20 that are found in these other regulations. Two amendments are particularly important as they go beyond updating cross-reference citations. One amendment to Appendix C of 10 CFR Part 2 updates and modifies the examples of the severity levels associated with violations of 10 CFR Part 20. Because Appendix C relates to administrative policy of the Commission and because the listed violations are used as examples of

different severity levels and are not all inclusive, the Commission does not believe that solicitation of public comment is required before these are issued in final form.

The second major change to other parts is the requirement to provide all workers with information on their radiation doses. This modification was made to conform to the 1587 Federal guidance on occupational radiation exposure. Formerly, Part 19 required licensees to furnish such a report at least annually upon the request of the worker. The change deletes the words "upon request." Public comment is not being solicited on this change as the comments were requested in the proposed rule (Section XXVII, 51 FR 1118) on the option of requiring reports to individual workers. (These comments are discussed with regard to § 20.1106.) Part 19 has been revised to require licensees to advise each worker at least annually of the worker's dose recorded pursuant to § 20.1106.

# VIII. Finding of No Significant Environmental Impact: Availability

The Commission has determined under the National Environmental Policy Act of 1969, as amended and the Commission's regulations in Subpart A of 10 CFR Part 51 that this rule is not a major Federal action significantly affecting the quality of the human environment and therefore an environmental impact statement is not required. The revised 10 CFR Part 20 changes the level for protection of the general public from an implicit limit of 0.5 rems per year to an explicit limit of 0.1 rem per year. There are also numerous changes in airborne and water radionuclide concentration limits. These changes result from changes in the models and parameters used to estimate the radiation dose associated with intake of a radionuclide. Some of the concentration limits for the general public in this revision are higher or lower than present concentration limits; and some are similar to the present limits.

Despite the changes in the dose and concentration limits, the Commission believes that issuance of the final Part 20 rule will not have a major impact on the environment. The primary basis for this conclusion is that, in addition to 10 CFR Part 20, there are other regulations that govern allowable doses to members of the public and that remain unchanged by the changes to Part 20. These other regulations include Appendix I to 10 CFR Part 50, 10 CFR Part 60, and 10 CFR Part 61, the EPA's generally applicable environmental standards in 40 CFR Part 190 and the National Emission Standards for Hazardous Air Pollutants (NESHAP) in 40 CFR Part 61 Subpart I. These standards set limits or design objectives (Appendix I) for releases of radioactive material to the general environment that are generally more restrictive than the dose limits in Part 20. Consequently, since these more restrictive standards remained essentially unchanged by the Part 20 revision, the level of public protection and the associated environmental impact are not changed appreciably from those associated with the current rule and the aforenamed regulations.

The environmental assessment and finding of no significant impact on which this determination is based are available for inspection at the NRC Public Document Room 2120, L Street NW (Lower-Level), Washington, DC 20555. Single copies of the environmental assessment and finding of no significant impact are available from Harold T. Peterson, Jr., Nuclear Regulatory Commission, NL/S-139, Washington, DC 20555, Telephone: (301) 492-3640.

# IX. Paperwork Reduction Act Statement

This final rule amends information collection requirements that are subject to the Paperwork Reduction Act of 1980 (44 U.S.C. 3501, et seq.). These information collection requirements have been submitted for approval by the Office of Management and Budget (OMB). These information collection requirements will not become effective until approved by OMB. The OMB approval will be published in the Federal Register.

Public reporting burden for this collection of information is estimated to average 33 hours per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Information and Records Management Branch (MNBB 7714), U.S. Nuclear Regulatory Commission, Washington, DC 20555, and to the Desk Officer, Office of Information and Regulatory Affairs, NEOB-3019 (3150-0014, 3150-0044, 3150-0005, and 3150-0006), Office of Management and Budget, Washington, DC 20503.

### X. Revised Regulatory Analysis

The Commission has issued a final regulatory analysis for this regulation. This revised analysis was based on the draft regulatory analysis as modified to account for the changes from the proposed rule resulting from public comments on both the proposed rule and the staff's revised rule in SECY-88-315 and supplemental papers. Copies of both the draft and final regulatory analysis are available for inspection and copying for a fee in the NRC Public Document Room. (See Address.)

#### XI. Final Regulatory Flexibility Analysis

In accordance with the Regulatory Flexibility Act of 1980, 5 U.S.C. 605(b), the Commission has prepared a regulatory flexibility analysis that indicated the revised rule will apply to all NRC licensees. The NRC has approximately 7,500 licensees, approximately one-quarter of which are classified as small entities. (Note: Agreement States, which implement comparable regulations under Section 274 of the Atomic Energy Act of 1954, as amended, have about 16,000 licensees of which a comparable number are assumed to be small entities.) The types of small entities that would be affected by this rule include physicians, small hospitals, small laboratories, industrial applications in small industries, radiographers, and well loggers.

Copies of the draft and final regulatory analysis are available for inspection and copying, for a fee, in the NRC Public Document Room. (See Address.)

104

pro.

SEU

### XII. Backfit Analysis

A final backfit analysis has been prepared for this rule and may be examined and copied for a fee in the Commission's Public Document Room (see "Address"). For the reasons stated in this backfit analysis, the Commission believes that the reductions in allowable dose limits that are embodied in the revised Part 20 constitute substantial increases in the protection of public health and safety. Although current practice, including the philosophy of keeping radiation exposures as low as is reasonably achievable (ALARA), generally has kept radiation exposures well below the existing limits, the reductions in the allowable dose limits ensure that such doses will also remain low in the future.

In addition to the quantifiable safety benefits accruing from dose reductions and other improvements in the revised Part 20, there are several qualitative factors that support issuing the Part 20 revision. One of the main qualitative factors is that it is necessary to revise the 30-year-old existing Part 20 to ensure that the NRC regulations reflect the current state of radiation protection science. Any future revisions in dose limits recommended by ICRP or NCRP would undoubtedly be based upon the 1977 ICRP and 1987 NCRP recommendations and, therefore, would be more easily incorporated into the framework of the revised Part 20 than in the framework of the current Part 20. Other qualitative factors include: maintaining consistency with international radiation protection factors, keeping the radiation protection requirements consistent with current risk assessment methodologies, and having the NRC's standards conform to Federal radiation protection guidance.

Basec onclusions in the final backfit analysis, the revised Part 20 pr Substantial increase in public health and safety compared to current s, including a determination that, when the quantitative and qualitative benefits of the revision are considered, the costs of implementing the revised Part 20 are justified, the Commission finds that the requirements of the "Backfit Rule" (§ 50.109) are satisfied and that the Part 20 revision should be issued as final rule.

SRM:

#1

OMM-

add-

OG

SRMA #1 JCA

The Commission is adopting the final rule based on the conclusions of this analysis that the rule provides for a substantial increase in the overall protection of the public health and safety and that the direct and indirect costs of its implementation are justified in terms of the quantitative and qualitative benefits associated with the rule. The Commission would note, however, that, even had the analysis not concluded that the revised Part 20 provides a substantial increase in the overall public health and safety, it could have gone forward with the rule because the changes made to Part 20 also amount to a redefinition of the level of adequate protection and the backfit rule's cost

### XIII. Additional Views of Commissioner Curtiss

I approve the revisions to 10 CFR Part 20 and related changes to other regulations as outlined in SECY-88-315 and SECY-89-267, subject to the modifications discussed below.

<u>Backfit</u>: I have examined the proposed Part 20 amendments from the standpoint of whether and, if so, how the backfit rule should apply to this particular rulemaking. The nature and effects of the proposed changes to Part 20 lead me to the conclusion that the proposed amendments, in essence, would redefine what is necessary for adequate protection of the public health and safety in the radiation protection area. Thus, while I believe that we should apply the backfit rule to this Part 20 rulemaking effort, I also believe that this rulemaking constitutes a redefinition of adequate protection as described in 10 CFR § 50.109(a)(4)(iii) and that the usual backfit analysis and cost-benefit balancing are therefore not required in this instance.

On the question of whether such an approach would require this rule to be renoticed for further public comment, I have concluded that there was ample indication in the notice of proposed rulemaking that the Commission is rethinking its radiation protection standards across-the-board in this Part 20 rulemaking. Moreover, this initiative was explained in a manner that could logically be construed to encompass the approach to backfitting described above. Of particular importance, the notice of proposed rulemaking itself seems to indicate that the Commission is contemplating an action that would redefine what is necessary for adequate protection in the radiation protection area. For example, the notice states that:

[T]he Nuclear Regulatory Commission (NKC) is proposing a major revision of its regulations in 10 CFR Part 20 which provide the requirements for the protection of individuals who are exposed . . . to ionizing radiation from routine activities . . . which are licensed by the NRC. . . . The intent of the revision is to improve NRC radiation protection standards by reflecting developments in the principles that underive radiation protection and advances in related sciences that have occurred since the promulgation of 10 CFR part 20 nearly thirty years ago. . . . The expected result of promulgating and implementing the proposed revised rule is an improved rule that provides better assurance of protection; establishes a clear health protection basis for limits and other regulatory actions taken to protect public health; applies to all licensees in a consistent manner; and reflects current information on health risk, dosimetry, and radiation protection practices and experiences.

51 Fed. Reg. 1092 (January 9, 1986).

With regard to existing Part 20 standards, the Commission noted that:

[i]n promulgating these standards, the ALC emphasized "that the standards are subject to change with the development of new knowledge, with significant increase in the average exposure of the whole population to radiation and with further experience in the administration of the Commission's regulatory program." Consistent with this emphasis, the proposed revision reflects new knowledge, increased uses of radiation and generation of radiation sources,

SRM1#

and experience gained during the past twenty years. . . . [Earlier] revisions [to the existing Part 20] have not kept the regulations in accord with more recent recommendations of scientific organizations . . to improve overall protection and establish a clear health risk rationale. . . . [T]he central thrust of the revision [is] to ensure that radiation protection is adequate and defendable when judged by good protection practices and contemporary standards.

51 Fed. Reg. 1093, 1094 (citations omitted).

In discussing the benefits of the proposed rulemaking, the Commission indicated that:

[t]he proposed revision to Part 20 includes numerous changes required to bring the radiation protections standards into accord with current defendable [sic] scientific knowledge, and to reflect contemporary scientific and philosophical approaches to protection against radiation. . . The Commission anticipates that promulgating and implementing the proposed rule will result in a regulation that provides better assurance of protection, establishes a clear health protection basis for limits, applies to all licensees, including small entities, in a consistent manner, and reflects current information on health risk, dosimetry, and radiation protection practices and experiences.

51 Fed. Reg. 1120, 1122.

Consistent with all of these statements on the nature of the proposed changes to Part 20, a supplemental notice of proposed rulemaking that requested comments on a proposed backfit analysis indicated that:

[T]his is the first complete revision of these regulations in over 25 years. This revision will bring the Commission's radiation protection standards into accord with current recommendations of the International Commission on Radiological Protection (ICRP).

The proposed revision to 10 CFR Part 20 [is] intended to:

- Update the quarter-century-old 10 CFR Part 20 to incorporate advances in science and new concepts of radiation protection methodology and philosophy;
- Implement pending Federal radiation guidance on occupational radiation protection;
- c. Implement the principal current dose-limiting recommendations of the ICRP;
- Incorporate the ICRP "effective dose equivalent" concept;
- e. Update the limits on airborne radionuclide intakes, effluent releases and doses from inhaled or ingested radionuclides using up-to-date metabolic models and dose factors; and
- f. Require that licensees have programs for keeping radiation exposures "as low as is "easonably achievable" (ALARA).

51 Fed. Reg. 30870, 30871 (August 29, 1986).

SR:71#1 JOAC

N . 53 ....

Overall, these various characteristics of the purpose, intent, and nature of the proposed changes to Part 20 lead to the conclusion that the Commission is, in fact, rethinking its radiation protection standards. For these reasons, I believe that the notice adequately describes the nature and substance of the proposed rule changes and that renoticing to further reflect a Commission judgment that the proposed changes constitute a redefinition of adequate protection is not necessary.

TCAC SRMA p1

#### XIV. List of Subjects

Part 20 - Byproduct material, licensed material, nuclear materials, nuclear power plants and reactors, occupational safety and health, packaging and containers, penalty, radiation protection, reporting and recordkeeping requirements, special nuclear material, source material, waste treatment and disposal.

Parts 2, 19, 20, 31, 32, 34, 35, 39, 40, 50, and 61 - Radiation protection.

Under the authority of the Atomic Energy Act of 1954, as amended, the Energy Reorganization Act of 1974, as amended, and 5 U.S.C. 552 and 553, the following amendments to 10 CFR Parts 2, 19, 20, 31, 32, 34, 35, 39, 40, 50, and 61 are published as a document subject to codification.

1. 10 CFR Part 20 is revised to read as follows:

Regulatory Innguest

PART 20 - STANDARDS FOR PROTECTION AGAINST RADIATION

SUBPART A--GENERAL PROVISIONS

### Section

- 20.1 Purpose.
- 20.2 Scope.
- 20.3 Definitions.
- 20.4 Units of radiation dose.

- 20.5 Units of radioactivity.
- 20.6 Interpretations.
- 20.7 Communications.
- 20.8 Implementation.
- 20.9 Reporting, recording, and application requirements: OMB approval

### SUBPART B--RADIATION PROTECTION PROGRAMS

20.101 Radiation protection programs.

### SUBPART C--OCCUPATIONAL DOSE LIMITS

- 20.201 Occupational dose limits for adults.
- 20.202 Compliance with requirements for summation of external and internal doses.
- 20 203 Determination of external dose from airborne radioactive material.
- 20.204 Determination of internal exposure.
- 20.205 [Reserved]
- 20.206 Planned special exposures.
- 20.207 Occupational dose limits for minors.
- 20.208 Dose to an embryo/fetus.

## SUBPART D--RADIATION DOSE LIMITS FOR INDIVIDUAL MEMBERS OF THE PUBLIC

- 20.301 Dose limits for individual members of the public.
- 20.302 Compliance with dose limits for individual members of the public.

### SUBPART E--[RESERVED]

### SUBPART F .- SURVEYS AND MONITORING

- 20.501 General.
- 20.502 Conditions requiring individual monitoring of external and internal occupational dose.

## SUBPART G--CONTROL OF EXPOSURE FROM EXTERNAL SOURCES IN RESTRICTED AREAS

- 20.601 Control of access to high radiation areas.
- 20.602 Control of access to very high radiation areas.
- 20.603 Control of access to very high radiation areas irradiators.

## SUBPART H--RESPIRATORY PROTECTION AND CONTROLS TO RESTRICT INTERNAL EXPOSURE IN RESTRICTED AREAS

- 20.701 Use of process or other engineering controls.
- 20.702 Use of other controls.
- 20.703 Use of individual respiratory protection equipment.
- 20.704 Further restrictions on the use of respiratory protection equipment.

### SUBPART I--STORAGE AND CONTROL OF LICENSED MATERIAL

- 20.801 Security of stored material.
- 20.802 Control of material not in storage.

### SUBPART J--PRECAUTIONARY PROCEDURES

- 20.901 Caution signs.
- 20.902 Posting requirements.

- 20.903 Exceptions to posting requirements.
- 20.904 Labeling containers.
- 20.905 Exemptions to labeling requirements.
- 20.906 Procedures for receiving and opening packages.

#### SUBPART K--WASTE DISPOSAL

- 20.1001 General requirements.
- 20.1002 Method for obtaining approval of proposed disposal procedures.
- 20.1003 Disposal by release into sanitary sewerage.
- 20.1004 Treatment or disposal by incineration.
- 20.1005 Disposal of specific wastes.
- 20.1006 Transfer for disposal and manifests.
- 20.1007 Compliance with environmental and health protection regulations.

#### SUBPART L--RECORDS

- 20.1101 General provisions.
- 20.1102 Records of radiation protection programs.
- 20.1103 Records of surveys.
- 20.1104 Determination of prior occupational dose.
- 20.1105 Records of planned special exposures.
- 20.1106 Records of individual monitoring results.
- 20.1107 Records of dose to individual members of the public.
- 20.1108 Records of waste disposal.
- 20.1109 Records of testing entry control devices for very high radiation areas.
- 20.1110 Form of records.

### SUBPART M--REPORTS

20.1201 Reports of theft or loss of licensed material. 20.1202 Notification of incidents.

- 20.1203 Reports of exposures, radiation levels, and concentrations of radioactive material exceeding the limits.
- 20.1204 Reports of planned special exposures.
- 20.1205 [Reserved]
- 20.1206 Reports of individual monitoring.

#### SUBPART N--EXEMPTIONS AND ADDITIONAL REQUIREMENTS

- 20.1301 Applications for exemptions.
- 20.1302 Additional requirements.

### SUBPART 0--ENFORCEMENT

20.1401 Violations.

#### APPENDICES

Appendix A Protection factors for respirators.

Appendix B Annual limits on intake (ALIs) and derived air concentrations (DACs) of radionuclides for occupational exposure; effluent concentrations; concentrations for release to sewerage.

Appendix C Quantities of licensed material requiring labeling.

Appendix O United States Nuclear Regulatory Commission Regional Offices.

Appendix E [Reserved]

Appendix F Requirements for low-level waste transfer for disposal at land disposal facilities and manifests.

AUTHORITY: Secs. 53, 63, 65, 81, 103, 104, 161, 182, 186, 68 Stat. 930, 933, 935, 936, 937, 948, 953, 955, as amended (42 U.S.C. 2073, 2093, 2095, 2111, 2133, 2134, 2201, 2232, 2236): secs. 201, as amended, 202. 206, 88 Stat. 1242, as amended, 1244, 1246 (42 U.S.C. 5841, 5842, 5546). For the purposes of sec. 223, 68 Stat. 958, as amended (42 U.S.C. 2273), §§ 20.102, 20.201 - 20.204, 20.206, 20.207, 20.208, 20.301, 20.302, 20.501, 20.502, 20.601(a) and (d), 20.602, 20.603, 20.701, 20.704, 20.801, 20.802, 20.901(a), 20.902, 20.904, 20.906, 20.1001, 20.1002, 20.1003, 20.1004, 20.1005(b) = (d), 20.1006, 20.1101 = 20.1110, 20.1201 = 20.1206, and 20.1301 are issued under sec. 161b., 68 Stat. 948 (42 U.S.C. 2201(b)) and § 20.1106(d) is issued under the Privacy Act of 1974, Pub.L. 93=579, 5 U.S.C. 552a: and §§ 20.102(a)(2) and (4), 20.204(c), 20.206(g) and (h), 20.904(c)(4), 20.905(c) and (d), 20.1005(c), 20.1001 = 20.1207 are issued under sec. 1610, 68 Stat. 950, as amended (42 U.S.C. 2201(b)).

### SUBPART A--GENERAL PROVISIONS

### § 20.1 Purpose.

(a) The regulations in this part establish standards for protection against ionizing radiation resulting from activities conducted under licenses issued by the Nuclear Regulatory Commission. These regulations are issued under the Atomic Energy Act of 1954, as amended, and the Energy Reorganization Act of 1974, as amended.

(b) It is the purpose of the regulations in this part to control the receipt, possession, use, transfer, and disposal of licensed material by any licensee in such a manner that the total dose to an individual (including doses resulting from licensed and unlicensed radioactive material and from radiation sources other than background radiation) does not exceed the standards for protection against radiation prescribed in the regulations in this part. However, nothing in this part shall be construed as limiting actions that may be necessary to protect health and safety.

#### § 20.2 Scope.

The regulations in this part apply to persons licensed by the Commission to receive, possess, use, transfer, or dispose of byproduct, source, or special nuclear material or to operate a production or utilization facility under Parts 30 through 35, 39, 40, 50, 60, 61, 70, or 72 of this chapter. The limits in this part do not apply to doses due to background radiation, to exposure of patients to radiation for the purpose of medical diagnosis or therapy, or to voluntary participation in medical research programs.

§ 20.3 Definitions.

As used in this part:

"Absorbed dose" means the energy imparted by ionizing radiation per unit mass of irradiated material. The units of absorbed dose are the rad and the gray (Gy).

"Act" means the Atomic Energy Act of 1954 (42 U.S.C. 2011 et seq.), as amended.

"Activity" is the rate of disintegration (transformation) or decay of radioactive material. The units of activity are the curie (Ci) and the becquerel (Bq).

"Adult" means an individual 18 or more years of age.

"Airborne radioactive material" means radioactive material dispersed in the air in the form of dusts, fumes, particulates, mists, vapors, or gases.

"Airborne radioactivity area" means a room, enclosure, or area in which airborne radioactive materials, composed wholly or partly of licensed material, exist in concentrations--

 In excess of the derived air concentrations (DACs) specified in Appendix B, or

SRM

(2) To such a degree that an individual present in the area without respiratory protective equipment could exceed, during the hours an individual is present in a week, an ir ake of 0.6 percent of the annual limit on intake (ALI) or 12 DAC-hours.

"Annual limit on intake" (ALI) means the derived limit for the amount of radioactive material taken into the body of an adult worker by inhalation or ingestion in a year. ALI is the smaller value of intake of a given radionuclide in a year by the reference man that would result in a committed effective dose equivalent of 5 rems (0.05 Sv) or a committed dose equivalent of 50 rems (0.5 Sv) to any individual organ or tissue. (ALI values for intake by ingestion and by inhalation of selected radionuclides are given in Table 1, Columns 1 and 2, of Appendix B.)

"ALARA" (acronym for "as low as is reasonably achievable") means making every reasonable effort to maintain exposures to radiation as far below the dose limits in this part as is practical consistent with the purpose for which the licensed activity is undertaken, taking into account the state of technology, the economics of improvements in relation to state of technology, the economics of improvements in relation to benefits to the public health and safety, and other societal and socioeconomic considerations, and in relation to utilization of nuclear energy and licensed materials in the public interest.

"Background radiation" means radiation from cosmic sources; naturally occurring radioactive materials, including radon in concentrations or levels commonly found in structures or the environment; and global fallout as it commonly exists in the environment from the testing of nuclear explosive devices. "Background radiation" does not include radiation from source, byproduct, or special nuclear materials regulated by the Commission.

"Bioassay" (radiobioassay) means the determination of kinds, quantities or concentrations, and, in some cases, the locations of radioactive material in the human body, whether by direct measurement (in vivo counting) or by analysis and evaluation of materials excreted or removed from the human body.

p6

"Byproduct material" means -

(1) Any radioactive material (except special nuclear material) yielded in, or made radioactive by, exposure to the radiation incident to the process of producing or utilizing special nuclear material; and

(2) The tailings or wastes produced by the extraction or concentration of uranium or thorium from ore processed primarily for its source material content, including discrete surface wastes resulting from uranium solution extraction processes. Underground ore bodies depleted by these solution extraction operations do not constitute "byproduct material" within this definition.

"Class" (or "lung class" or "inhalation class") means a classification scheme for inhaled material according to its rate of clearance from the pulmonary region of the lung. Materials are classified as D, W, or Y, which applies to a range of clearance half-times: for Class D (Days) of less than 10 days, for Class W (Weeks) from 10 to 100 days, and for Class Y (Years) of greater than 100 days.

"Collective dose" is the sum of the individual doses received in a given period of time by a specified population from exposure to a specified source of radiation.

"Commission" means the Nuclear Regulatory Commission or its duly authorized representatives. 267

"Committed dose equivalent"  $(H_{T,50})$  means the dose equivalent to organs or tissues of reference (T) that w be received from an intake of radioactive material by an individual during the 50-year period following the intake.

"Committed effective dose equivalent"  $(H_{E,50})$  is the sum of the products of the weighting factors applicable to each of the body organs or tissues that are irradiated and the committed dose equivalent to these organs or tissues

 $(H_{E,50} = \Sigma w_T H_{T,50}).$ 

"Controlled area" means an area, outside of a restricted area but inside the site boundary, access to which can be limited by the licensee for any reason.

"Declared pregnant woman" means a woman who has voluntarily informed her employer, in writing, of her pregnancy and the estimated date of conception.

"Deep-dose equivalent" (H<sub>d</sub> which applies to external whole-body exposure, is the dose equivalent at a tissue depth of 1 cm (1000 mg/cm<sup>2</sup>).

"Department" means the Department of Energy established by the Department of Energy Organization Act (Pub. L. 95-91, 91 Stat. 565, 42 U.S.C. 7101 et seq.) to the extent that the Department, or its duly authorized representatives, exercises functions formerly vested in the U.S. Atomic Energy Commission, its Chairman, members, officers, and components and transferred to the U.S. Energy Research and Development Administration and to the Administrator thereof pursuant to sections 104 (b), (c), and (d) of the Energy Reorganization Act of 1974 (Pub. L. 93-438, 88 Stat. 1233 at 1237, 42 U.S.C. 5814) and retransferred to the Secretary of Energy pursuant to section 301(a) of the Department of Energy Organization Act (Pub. L. 95-91, 91 Stat. 565 at 577-578, 42 U.S.C. 7151).

"Derived air concentration" (DAC) means the concentration of a given radionuclide in air which, if breathed by the reference man for a working year of 2,000 hours under conditions of light work (inhalation rate 1.2 cubic meters of air per hour), results in an intake of one ALI. DAC values are given in Table 1, Column 3, of Appendix B.

"Derived air concentration-hour" (DAC-hour) is the product of the concentration of radioactive material in air (expressed as a fraction or multiple of the derived air concentration for each radionuclide) and the time of exposure to that radionuclide, in hours. A licensee may take 2,000 DAC-hours to represent one ALI, equivalent to a committed effective dose equivalent of 5 rems (0.05 Sv).

119

"Dose" or "radiation dose" is a generic term that means absorbed dose, dose equivalent, effective dose equivalent, committed dose equivalent, committed effective dose equivalent, or total effective dose equivalent, as defined in other paragraphs of this section.

"Dose equivalent"  $(H_T)$  means the product of the absorbed dose in tissue, quality factor, and all other necessary modifying factors at the location of interest. The units of dose equivalent are the rem and sievert (Sv).

"Dosimetry processor" means an individual or an organization that processes and evaluates individual monitoring equipment in order to determine the radiation dose delivered to the equipment.

"Effective dose equivalent" (H<sub>E</sub>) is the sum of the products of the dose equivalent to the organ or tissue (H<sub>T</sub>) and the weighting factors (w<sub>T</sub>) applicable to each of the body organs or tissues that are irradiated (H<sub>E</sub> =  $\Sigma w_T H_T$ ).

"Embryo/fetus" means the developing human organism from conception until the time of birth.

"Entrance or access point" means any location through which an individual could gain access to radiation areas or to radioactive materials. This includes entry or exit portals of sufficient size to permit human entry, irrespective of their intended use.

"Exposure" means being exposed to ionizing radiation or to radioactive material.

"External dose" means that portion of the dose equivalent received from radiation sources outside the body.

"Extremities" means hand, elbow, arm below the elbow, foot, knee, and leg below the knee.

"Eye dose equivalent" applies to the external exposure of the lens of the eye and is taken as the dose equivalent at a tissue depth of 0.3 centimeter (300 mg/cmz).

"Generally applicable environmental radiation standards" means standards issued by the Environmental Protection Agency (EPA) under the authority of the Atomic Energy Act of 1954, as amended, that impose limits on radiation exposures or levels, or concentrations or quantities of radioactive material, in the general environment outside the boundaries of locations under the control of persons possessing or using radioactive material.  $\sum 267 \ p \ 6 \ ]$ 

"Government agency" means any executive department, commission, independent establishment, corporation wholly or partly owned by the United States of America, which is an instrumentality of the United States, or any board, bureau, division, service, office, officer, authority, administration, or other establishment in the executive branch of the Government.

"Gray" [See § 20.4]

"High radiation area" means an area, accessible to individuals, in which radiation levels could result in an individual receiving a dose equivalent in excess of 0.1 rem (1 mSv) in § hour at 30 centimeters from the radiation source or from any surface that the radiation penetrates.

"Individual" means any human being.

"Individual monitoring" means:

(1) The assessment of dose equivalent by the use of devices designed to be worn by an individual;

(2) The assessment of committed effective dose equivalent by bioassay (see Bioassay) or by determination of the time-weighted air concentrations to which an individual has been exposed, i.e., DAC-hours; or

(3) The assessment of dose equivalent by the use of survey data.

"Individual Monitoring Devices" ("individual monitoring equipment") means devices designed to be worn by a single individual for the assessment of dose equivalent such as film badges, thermoluminescent dosimeters (TLDs), pocket ionization chambers, and personal ("lapel") air sampling devices. "Internal dose" means that portion of the dose equivalent received from radioactive material taken into the body.

"License" means a license issued under the regulations in Parts 30 through 35, 39, 40, 50, 60, 61, 70, or 72 of this chapter.

"Licensee" means the holder of a license.

"Licensed material" means source material, special nuclear material, or byproduct material received, possessed, used, transferred or disposed of unde: a general or specific license issued by the Commission.

"Limits" (dos: limits) means the permissible upper bounds of radiation doses.

"Lost or missing licensed material" means licensed material whose location is unknown. It includes material that has been shipped but has not reached its destination and whose location cannot be readily traced in the transportation system.

"Member of the public" means an individual in a controlled or unrestricted area. However, an individual is not a member of the public during any period in which the individual receives an occupational dose.

"Minor" means an individual less than 18 years of age.

"Monitoring" (radiation monitoring, radiation protection monitoring) means the measurement of radiation levels, concentrations, surface area concentrations or quantities of radioactive material and the use of the results of these measurements to evaluate potential exposures and doses.

"Nonstochastic effect" means health effects, the severity of which varies with the dose and for which a threshold is believed to exist. Radiation-induced cataract formation is an example of a nonstochastic effect.

"NRC" means the Nuclear Regulatory Commission or its duly authorized representatives.

"Occupational dose" means the dose received by an individual in a restricted area or in the course of employment in which the individual's assigned duties involve exposure to radiation and to radioactive material from licensed and unlicensed sources of radiation, whether in the possession of the licensee or other person. Occupational dose does not include dose received from background radiation, as a patient from medical practices, from voluntary participation in medical research programs, or as a member of the general public.

### "Person" means:

(1) Any individual, corporation, partnership, firm, association, trust, estate, public or private institution, group, Government agency other than the Commission or the Department of Energy (except that the Department shall be considered a person within the meaning of the regulations in 10 CFR Chapter 1 to the extent that its facilities and activities are subject to the licensing and related regulatory authority of the Commission under Section 202 of the Energy Reorganization Act of 1974 (88 Stat. 1244), the Uranium Mill Tailings Radiation Control Act of 1978 (92 Stat. 3021), the Nuclear Waste Policy Act of 1982 (96 Stat. 2201), and Section 3(b)(2) of the Low-Level Radioactive Waste Policy Amendments Act of 1985 (99 Stat. 1842)), any State or any political subdivision of or any political entity within a State, any foreign government or nation or any political subdivision of any such government or nation, or other entity; and

(2) Any legal successor, representative, agent, or agency of the foregoing.

"Planned special exposure" means an infrequent exposure to radiation, separate from and in addition to the annual dose limits.

"Public dose" means the dose received by a member of the public from exposure to radiation and to radioactive material released by a licensee, or to another source of radiation either within a licensee's controlled area or in unrestricted areas. It does not include occupational dose or doses received from background radiation, as a patient from medical practices, or from voluntary participation in medical research programs.

123

"Quality Factor" (Q) means the modifying factor (listed in Tables 1 and 2 of § 20.4) that is used to derive dose equivalent from absorbed dose.

"Quarter" means a period of time equal to one-fourth of the year observed by the licensee (approximately 13 consecutive weeks), providing that the beginning of the first quarter in a year coincides with the starting date of the year and that no day is omitted or duplicated in consecutive quarters.

"Rad" [See § 20.4].

"Radiation" (ionizing radiation) means alpha particles, beta particles, gamma rays, x-rays, neutrons, high-speed electrons, high-speed protons, and other particles capable of producing ions. Radiation, as used in this part, does not include non-ionizing radiation, such as radio- or microwaves, or visible, infrared, or ultraviolet light.

"Radiation area" means an area, accessible to individuals, in which radiation levels could result in an individual receiving a dose equivalent in excess of 0.005 rem (0.05 mSv) in 1 hour at 30 centimeters from the radiation source or from any surface that the radiation penetrates.

"Reference man" means a hypothetical aggregation of human physical and physiological characteristics arrived at by international consensus. These characteristics may be used by researchers and public health workers to standardize results of experiments and to relate biological insult to a common base.

"Rem" [See § 20.4].

"Respiratory protective device" means an apparatus, such as a respirator, used to reduce the individual's intake of airborne radioactive materials.

"Restricted area" means an area, access to which is limited by the licensee for the purpose of protecting individuals against undue risks from exposure to radiation and radioactive materials. Restricted area does not include areas used as residential quarters, but separate room. " residential building may be set apart as a restricted area.

124

"Sanitary sewerage" means a system of public sewers for carrying off waste water and refuse, but excluding sewage treatment facilities, septic tanks, and leach fields owned or operated by the licensee.

"Shallow-dose equivalent" (H<sub>s</sub>), which applies to the external exposure of the skin or an extremity, is taken as the dose equivalent at a tissue depth of 0.007 centimeter (7 mg/cm<sup>2</sup>) averaged over an area of 1 square centimeter.

"Sievert" [See § 20.4].

"Site boundary" means that line beyond which the land or property is not owned, leased, or otherwise controlled by the licensee.

"Source material" means-

(1) Uranium or thorium or any combination of uranium and thorium in any physical or chemical form; or

(2) Ores that contain, by weight, one-twentieth of 1 percent (0.05 percent), or more, of uranium, thorium, or any combination of uranium and thorium. Source material does not include special nuclear material.

"Special nuclear material" means-

(1) Plutonium, uranium-233, uranium enriched in the isotope 233 or in the isotope 235, and any other material that the Commission, pursuant to the provisions of section 51 of the Act, determines to be special nuclear material, but does not include source material; or

(2) Any material artificially enriched by any of the foregoing but does not include source material.

"Stochastic effects" means health effects that occur randomly and for which the probability of the effect occurring, rather than its severity, is assumed to be a linear function of dose without threshold. Hereditary effects and cancer incidence are examples of stochastic effects.

"Survey" means an evaluation of the radiological conditions and potential hazards incident to the production, use, transfer, release, disposal, or presence of radioactive material or other sources of radiation. When appropriate, such an evaluation includes a physical survey of the location of radioactive material and measurements or calculations of levels of radiation, or concentrations or quantities of radioactive material present.

"Total Effective Dose Equivalent" (TEDE) means the sum of the deep-dose equivalent (for external exposures) and the committed effective dose equivalent (for internal exposures).

"Unrestricted area" means an area, access to which is neither limited nor controlled by the licensee.

"Uranium fuel cycle" means the operations of milling of uranium ore, chemical conversion of uranium, isotopic enrichment of uranium, fabrication of uranium fuel, generation of electricity by a light-water-cooled nuclear power plant using uranium fuel, and reprocessing of spent uranium fuel, to the extent that these activities directly support the production of electrical power for public use. Uranium fuel cycle does not include mining operations, operations at waste disposal sites, transportation of radioactive material in support of these operations, and the reuse of recovered non-uranium special nuclear and byproduct materials from the cycle.

"Very high radiation area" means an area, accessible to individuals, in which radiation levels could result in an individual receiving an absorbed dose in excess of 500 rads (5 grays) in 1 hour at 1 meter from a radiation source or from any surface which the radiation penetrates. [Note: At very high doses received at high dose rates, units of absorbed dose (e.g., rads and grays) are appropriate, rather than units of dose equivalent (e.g., rems and sieverts).]

"Week" means 7 consecutive days starting on Sunday.

"Weighting factor",  $w_T$ , for an organ or tissue (T) is the proportion of the risk of stochastic effects resulting from irradiation of that organ or tissue to the total risk of stochastic effects when the whole body is irradiated uniformly. For calculating the effective dose equivalent, the values of  $w_T$  are:

ORGAN DOSE WEIGHTING	FACTORS
Organ or Tissue	w <sub>T</sub>
Gonads	0.25
Breast	0.15
Red bone marrow	0.12
Lung	0.12
Thyroid	0.03
Bone surfaces	0.03
Remainder	0.30 <sup>a</sup>
Whole Body	1.00 <sup>b</sup>

<sup>a</sup> 0.30 results from 0.06 for each of 5 "reminder" organs (excluding the skin and the lens of the eye) that receive the highest doses.

<sup>b</sup> For the purpose of weighting the external whole body dose (for adding it to the internal dose), a single weighting factor,  $w_T = 1.0$ , has been specified. The use of other weighting factors for external exposure will be approved on a case-by-case basis until such time as specific guidance is issued.

"Whole body" means, for purposes of external exposure, head, trunk (including male gonads), arms above the elbow, or legs above the knee.

"Working level" (WL) is any combination of short-lived radon daughters (for radon-222: polonium-218, lead-214, bismuth-214, and polonium-214; and for radon-220: polonium-216, lead-212, bismuth-212, and polonium-212) in 1 liter of air that will result in the ultimate emission of 1.3 x  $10^5$  MeV of potential alpha particle energy.

"Working level month" (WLM) means an exposure to 1 working level for 170 hours (2,000 working hours per year/12 months per year = approximately 170 hours per month).

127

ED

267

pb

"Year" means the period of time beginning in January used to determine compliance with the provisions of this part. The license may change the starting date of the year used to determine compliance by the licensee provided that the change is made at the beginning of the year and that no da, is omitted or duplicated in consecutive years.

### § 20.4 Units of radiation dose.

(a) As used in this part, the units of radiation dose are:

"Gray" (Gy) is the SI unit of absorbed dose. One gray is equal to an absorbed dose of 1 Joule/kilogram (100 rads).

"Rad" is the special unit of absorbed dose. One rad is equal to an absorbed dose of 100 ergs/gram or 0.01 joule/kilogram (0.01 gray).

"Rem" is the special unit of any of the quantities expressed as dose equivalent. The dose equivalent in rems is equal to the absorbed dose in rads multiplied by the quality factor (1 rem = 0.01 sievert).

"Sievert" is the SI unit of <u>any of the quantities expressed as</u> dose equivalent. The dose equivalent in sieverts is equal to the absorbed dose in grays multiplied by the quality factor (1 Sv = 100 rems).

(b) As used in this part, the quality factors for converting absorbed dose to dose equivalent are shown in Table 1.

127

	Quality Factor	Absorbed Dose
TYPE OF RADIATION	(Q)	Equal to a Unit Dosg Equivalent
X-, gamma, or beta radiation	1	1
Alpha particles, multiple-charged particles, fission fragments and heavy particles of unknown charg	ge 20	0.05
Neutrons of unknown energy	10	0.1
High-energy protons	10	0.1

and a second			TABLE 1		
VILLAUS	FACTORS	AND	ABSORBED	DOSE	EQUIVALENCIES

(C) If it is more convenient to measure the neutron fluence rate than to determine the neutron dose equivalent rate in rems per hour or sieverts per hour, as provided in paragraph (b) of this section, 1 rem (0.01 Sv) of neutron radiation of unknown energies may, for purposes of the regulations in this part, be assumed to result from a total fluence of 25 million neutrons per square centimeter incident upon the body. If sufficient information exists to estimate the approximate energy distribution of the neutrons, the licensee may use the fluence rate per unit dose equivalent or the appropriate Q value from Table 2 to convert a measured tissue dose in rads to dose equivalent in rems.

### TABLE 2

neutrons cm=* rem=1)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$

MEAN QUALITY FACTORS, Q, AND FLUENCE PEP UNIT DOSE EQUIVALENT FOR MONOENERGETIC NEUTRONS

# § 20.5 Units of radioactivity.

For the purposes of this part, activity is expressed in the special unit of curies (Ci) or in the SI unit of becquerels (Bq), or their multiples, or disintegrations (transformations) per unit of time.

(a) One becquerel = 1 disintegration per second  $(s^{-1})$ .

(b) One curie = 3.7 x 1010 disintegrations per second =

3.7 x  $10^{10}$  becquerels = 2.22 x  $10^{12}$  disintegrations per minute.

### § 20.6 Interpretations.

Except as specifically authorized by the Commission in writing, no interpretation of the meaning of the regulations in this part by an officer or employee of the Commission other than a written interpretation by the General Counsel will be recognized to be binding upon the Commission.

### § 20.7 Communications.

Unless otherwise specified, communications or reports concerning the regulations in this part should be addressed to the Executive Director for Operations, U.S. Nuclear Regulatory Commission, Washington, DC 20555. A communication, report, or application may be delivered in person to the Commission's offices at 2120 L Street, NW. (Lower Level), Washington, DC 20037, or 11555 Rockville Pike, Rockville, MD 20852.

#### § 20.8 Implementation.

(a) Licensees shall implement the provisions of this part on or before January 1, 1993. If a licensee chooses to implement the provisions of this part prior to January 1, 1993, the licensee shall implement all provisions of this part not otherwise exempted by paragraph (d) of this section, and shall provide written notification to either the Director of the Office of Nuclear Materials Safety and Safeguards or the Director of the Office of Auclear Reactor Regulation, as appropriate, that the licensee is adopting early implementation of this part. Until January 1, 1993, or until the licensee notifies the Commission of early implementation of the provisions of this part, compliance will be required with the version of 10 CFR Part 20 codified in the Code of Federal Regulations on January 1, 1991.

(b) After the time the licensee implements this part, the applicable section of this part shall be used in lieu of any section of this part in effect on or before January 1, 1991 that is cited in license conditions or technical specifications, except as specified in paragraphs (c), (d) and (e) of this section. If the requirements of this part are more restrictive than the existing license condition, then the licensee shall comply with this part unless exempted by paragraph (d) of this section.
SRM

(c) Any existing license condition or technical specification that is more restrictive than this part remains in force until there is a technical specification change, license amendment, or license renewal.

(d) If a license condition or technical specification exempts a licensee from a provision of the version of Part 20 in effect on or before January 1, 1991, it also exempts the licensee from the corresponding provision of this part.

(e) If no section in this part corresponds to the provisions of Part 20 in effect prior to January 1, 1991 cited in a license condition, a license condition based on the version of Part 20 in effect on or before January 1, 1991 remains in force until either there is a technical specification change, license amendment, or license renewal that modifies or removes this condition.

### § 20.9 Reporting, recording, and application requirements: OMB approval.

(a) The Nuclear Regulatory Commission will submit the information collection requirements contained in this part to the Office of Management and Budget for approval as required by the Paperwork Reduction Act of 1980 (44 U.S.C. 3501 et seq.). <u>OME clearance will be obtained prior to January 1</u>, 1993.

(b) The approved information collection requirements contained in this part appear in §§ 20.101, 20.202, 20.204, 20.206, 20.301, 20.501, 20.601, 20.603, 20.703, 20.901, 20.902, 20.904, 20.906, 20.1002, 20.1004, 20.1006, 20.1102, 20.1103, 20.1104, 20.1105, 20.1106, 20.1107, 20.1108, 20.1109, 20.1110, 20.1201, 20.1202, 20.1203, 20.1204, 20.1206, and Appendix F.

#### SUBPART B -- RADIATION PROTECTION PROGRAMS

#### § 20.101 Radiation protection programs.

(a) Each licensee shall develop, document, and in lement a radiation protection program commensurate with the scope and extent of licensed activities

[7590-01]

and sufficient to ensure compliance with the provisions of this part. (See § 20.1102 for recordkeeping requirements relating to these programs.)

(b) The licensee shall use, to the extent practicable, procedures and engineering controls based upon sound radiation protection principles to achieve occupational doses and doses to members of the public that are as low as is reasonably achievable (ALARA).

(c) The licensee shall periodically (at least annually) review the radiation protection program content and implementation.

#### SUBPART C--OCCUPATIONAL DOSE LIMITS

### § 20.201 Occupational dose limits for adults.

(a) The licensee shall control the occupational dose to individual adults, except for planned special exposures under § 20.206, to the following dose limits.

(1) An annual limit, which is the more limiting of--

(i) The total effective dose equivalent being equal to 5 rems (0.05 Sv); or

(ii) The sum of the deep-dose equivalent and the committed dose equivalent to any individual organ or tissue other than the lens of the eye being equal to 50 rems (0.5 Sv).

(2) The annual limits to the lens of the eye, to the skin, and to the extremities which are:

(i) An eye dose equivalent of 15 rems (0.15 Sv), and

(ii) A shallow-dose equivalent of 50 rems (0.50 Sv) to the skin or to each of the extremities.

(b) Doses received in excess of the annual limits, including doses received during accidents, emergencies, and planned special exposures, must be subtracted from the limits for planned special exposures that the individual may receive during the current year (see § 20.206(e)(1)) and during the individual's lifetime (see § 20.206(e)(2)).

(c) The assigned deep-dose equivalent and shallow-dose equivalent must be for the part of the body receiving the highest exposure. The deep-dose equivalent, eye dose equivalent and shallow-dose equivalent may be assessed from surveys or other radiation measurements for the purpose of demonstrating compliance with the occupational dose limits, if the individual monitoring device was not in the region of highest potential exposure, or the results of individual monitoring are unavailable.

(d) Derived air concentration (DAC) and annual limit on intake (ALI) values are presented in Table 1 of Appendix B and may be used to determine the individual's dose (see § 20.1106) and to demonstrate compliance with the occupational dose limits.

(e) In addition to the annual dose limits, the licensee shall limit the soluble uranium intake by an individual to 10 milligrams in a week in consideration of chemical toxicity (see footnote 3 of Appendix B).

(f) The licensee shall reduce the dose that an individual may be allowed to receive in the current year by the amount of occupational dose received while employed by any other person (see § 20.1104(e)).

## \$ 20.202 <u>Compliance with requirements for summation of external and</u> internal doses.

(a) If the licensee is required to monitor under both §§ 20.502(a) and (b), the licensee shall demonstrate compliance with the dose limits by summing external and internal doses. If the licensee is required to monitor only under § 20.502(a) or only under § 20.502(b), then summation is not required to demonstrate compliance with the dose limits. The licensee may demonstrate compliance with the requirements for summation of external and internal doses by meeting one of the conditions specified in paragraph (b) of the section, and the conditions in paragraphs (c) and (d) of this section.

(NOTE: The dose equivalents for the lens of the eye, the skin, and the extremities are not included in the summation, but are subject to separate limits.)

(b) <u>Intake by inhalation</u>. If the only intake of radionuclides is by inhalation, the total effective dose equivalent limit is not exceeded if the sum of the deep-dose equivalent divided by the total effective dose equivalent limit, and one of the following, does not exceed unity:

(1) The sum of the fractions of the inhalation ALI for each radionuclide, or

(2) The total number of derived air concentration-hours (DAC-hours) for all radionuclides divided by 2,000, or

(3) The sum of the calculated committed effective dose equivalents to all significantly irradiated<sup>1</sup> organs or tissues (T) calculated from bioassay data using appropriate biological models and expressed as a fraction of the annual limit.

(c) <u>Intake by oral ingestion</u>. If the occupationally exposed individual also receives an intake of radionuclides by oral ingestion greater than 10 percent of the applicable oral ALI, the licensee shall account for this intake and include it in demonstrating compliance with the limits.

(d) Intake through wounds or absorption through skin. The licensee shall evaluate and, to the extent practical, account for intakes through wounds or skin absorption. (NOTE: The intake through intact skin has been included in the calculation of DAC for hydrogen-3 and does not need to be further evaluated.)

# § 20.203 Determination of external dose from airborne radioactive material.

Licensees shall, when determining the dose from airborne radioactive material, include the contribution to the deep-dose equivalent, eye dose equivalent, and shallow-dose equivalent from external exposure to the radioactive cloud (see Appendix B, footnotes 1 and 2).

An organ or tissue is deemed to be significantly irradiated if, for that organ or tissue, the product of the weighting factors,  $w_T$ , and the committed dose equivalent,  $H_{50}$ , per unit intake is greater than 10 percent of the maximum weighted value of  $H_{50}$  (i.e.,  $w_TH_{50,T}$ ) per unit intake for any organ or tissue.

NUIE. Airborne radioactivity measurements and DAC values should not be used as the primary means to assess the decp-dose equivalent when the airborne radioactive material includes radionuclides other than noble gases or if the cloud of airborne radioactive material is not relatively uniform. The determination of the deep-dose equivalent to an individual should be based upon measurements using instruments or individual monitoring devices.

### § 20.204 Determination of internal exposure .

(a) For purposes of assessing dose used to determine compliance with occupational dose equivalent limits, the licensee shall, when required under § 20.502, take suitable and timely measurements of--

(1) Concentrations of radioactive materials in air in work areas; or

- (2) Quantities of radionuclides in the body; or
- (3) Quantities of radionuclides excreted from the body; or
- (4) Combinations of these measurements.

(b) Unless respiratory protective equipment is used, as provided in § 20.703, or the assessment of intake is based in bioassays, the licensee shall assume that an individual inhales radioactive material at the airborne concentration in which the individual is present.

(c) When specific information on the physical and biochemical properties of the radionuclides taken into the body or the behavior or the material in an individual is known, the licensee may--

(1) Use that information to calculate the committed effective dose equivalent, and, if used, the licensee shall document that information in the individual's record; and

(2) Upon prior approval of the Commission, adjust the DAC or ALI values to reflect the actual physical and chemical characteristics of airborne radioactive material (e.g., aerosol size distribution or density); and

(3) Separately assess the contribution of fractional intakes of Class D,W, or Y compounds of given radionuclide (see Appendix B) to the committed effective dose equivalent.

135

(d) If the licensee chooses to assess intakes of Class Y material using the measurements given in § 20.204(a)(2) or (3), the licensee may delay the recording and reporting of the assessments for periods up to 7 months, unless otherwise required by §§ 20.1202 or 20.1203, in order to permit the licensee to make additional measurements basic to the assessments.

(e) If the identity and concentration of each radionuclide in a mixture are known, the fraction of the DAC applicable to the mixture for use in calculating DAC-hours must be either--

(1) The sum of the ratios of the concentration to the appropriate DAC value (e.g., 0, W, Y) from Appendix B for each radionuclide in the mixture; or

(2) The ratio of the total concentration for all radionuclides in the mixture to the most restrictive DAC value for any radionuclide in the mixture.

(f) If the identity of each radionuclide in a mixture is known, but the concentration of one or more of the radionuclides in the mixture is not known, the DAC for the mixture must be the most restrictive DAC of any radionuclide in the mixture.

(g) When a mixture of radionuclides in air exists, licensees may disregard certain radionuclides in the mixture if--

(1) The licensee uses the total activity of the mixture in demonstrating compliance with the dose limits in § 20.201 and in complying with the monitoring requirements in § 20.502(b), and

(2) The concentration of any radionuclide disregarded is less than 10 percent of its DAC, and

(3) The sum of these percentages for all of the radionuclides disregarded in the mixture does not exceed 30 percent.

(h)(1) In order to calculate the committed effective dose equivalent, the licensee may assume that the inhalation of one ALI, or an exposure of 2,000 DAC-hours, results in a committed effective dose equivalent of 5 rems (0.05 Sv) for radionuclides that have their ALIs or DACs based on the committed effective dose equivalent.

(2) When the ALI (and the associated DAC) is determined by the nonstochastic organ dose limit of 50 rems (0.5 Sv), the intake of radionuclides that would result in a committed effective dose equivalent of 5 rems (0.05 Sv) (the static ALI) is listed in parentheses in Table 1 of Appendix B. In this 267

case, the licensee may, as a simplifying assumption, use the stochastic ALIs to determine committed effective dose equivalent. <u>However, if the licensee</u> <u>uses the stochastic ALIs, the licensee must also demonstrate that the limit</u> <u>in § 20.201(a)(1)(ii) is met.</u>

§ 20.205 [Reserved]

### § 20.206 Planned special exposures.

A licensee may authorize an adult worker to receive doses in addition to 237#5 and accounted for separately from the doses received under the limits specified in § 20.201 provided that each of the following conditions is satisfied--

(a) The licensee authorizes a planned special exposure only in an exceptional situation when alternatives that might avoid the higher exposure are unavailable or impractical.

(b) The licensee (and employer if the employer is not the licensee) specifically authorizes the planned special exposure, in writing, before the ecosure occurs.

(c) Before a planned special exposure, the licensee ensures that the individuals involved arp--

(1) Informed of the purpose of the planned operation;

(2) Informed of the estimated doses and associated potential risks and specific radiation levels or other conditions that might be involved in performing the task; and

(3) Instructed in the measures to be taken to keep the dose ALARA consid- 267 #8 ering other risks that may be present.

(d) Prior to permitting an individual to participate in a planned special exposure, the licensee ascertains prior doses as required by § 20.1104(b) during the lifetime of the individual for each individual involved.

(e) Subject to § 20.201(b), the licensee does not authorize a planned special exposure that would cause an individual to receive a dose from all planned special exposures and all doses in excess of the limits to exceed--

 The numerical values of any of the dose limits in § 20.201(a) in any year; and

267

p8

(2) Five times the annual dose limits in § 20.201(a) during the individual's lifetime.

(f) The licensee maintains records of the conduct of a planned special exposure in accordance with § 20.1105 and submits a written report in accordance with § 20.1204.

(g) The licensee records the best estimate of the dose resulting from the planned special exposure in the individua. s record and informs the individual, in writing, of the dose within 30 days from the date of the planned special exposure. The dose from planned special exposures is not to be considered in controlling future occupational dose of the individual under § 20.201(a) but is to be included in evaluations required by § 20.206(d) and (e).

# § 20.207 Occupational dose limits for minors.

The annual occupational dose limits for minors are 10 percent of the annual dose limits specified for adult workers in § 20.201.

### § 20.208 Dose to an embryo/fetus.

(a) The licensee shall ensure that the dose to an embryo/fetus during the entire pregnancy, due to occupational exposure of a declared pregnant woman, does not exceed 0.5 rem (5 mSv). (For recordkeeping requirements, see § 20.1106.)

(b) The licensee shall make efforts to avoid substantial variation above a uniform monthly exposure rate to a declared pregnant woman so as to satisfy the limit in paragraph (a) of this section.

- (c) The dose to an embryo/fetus shall be taken as the sum of--
- (1) The deep-dose equivalent to the declared pregnant woman; and
- (2) The dose to the embryo/fetus from radionuclides in the embryo/fetus and radionuclides in the declared pregnant woman.

(d) If the dose to the embryo/fetus is found to have exceeded 0.5 rem (5 mSv), or is within 0.05 rem (0.5 mSv) of this dose, by the time the woman

138

JCVS

declares the pregnancy to the licensee, the licensee shall be deemed to be in compliance with paragraph (a) of this section if the additional dose to the embryo/fetus does not exceed 0.05 rem (0.5 mSv) during the remainder of the pregnancy.

SUBPART D--RADIATION DOSE LIMITS FOR INDIVIDUAL MEMBERS OF THE PUBLIC

# § 20.301 Dose limits for individual members of the public.

(a) Each licensee shall conduct operations so that --

(1) The total effective dose equivalent to individual members of the public from the licensed operation does not exceed 0.1 rem (1 mSv) in a year, exclusive of the dose contribution from the licensee's disposal of radioactive material into sanitary sewerage in accordance with § 20.1003, and

(2) The dose in any unrestricted area from external sources does not 267 exceed 0.002 rem (0.02 mSv) in any one hour.

(b) If the licensee permits members of the public to have access to controlled areas, the limits for members of the public continue to apply to those individuals.

(c) A licensee or license applicant may apply for prior NRC authorization to operate up to an annual dose limit for an individual member of the public of 0.5 rem (5 mSv). The licensee or license applicant shall include the following information in this application:

(1) Demonstration of the need for and the expected duration of operations in excess of the limit in paragraph (a) of this section.,

(2) The licensee's program to assess and control dose within the 0.5 rem (5 mSv) annual limit., and

(3) The procedures to be followed to maintain the dose as low as is reasonably achievable.

(d) In addition to the requirements of this part, a licensee subject to the provisions of EPA's generally applicable environmental radiation standards in 40 CFR Part 190 shall comply with those standards.

(e) The Commission may impose additional restrictions on radiation levels in unrestricted areas and on the total quantity of radionuclides that a licensee may release in effluents in order to restrict the collective dose. § 20.302 Compliance with dose limits for individual members of the public.

(a) The licensee shall make or cause to be made, as appropriate, surveys of radiation levels in unrestricted and controlled areas and radioactive materials in effluents released to unrestricted and controlled areas to demonstrate compliance with the dose limits for individual members of the public in § 20.301.

(b) A licensee shall show compliance with the annual dose limit in § 20.301 by--

(1) Demonstrating by measurement or calculation that the total effective dose equivalent to the individual likely to receive the highest dose from the licensed operation does not exceed the annual dose limit., or

(2) Demonstrating that-

(i) The annual average concentrations of radioactive material released in gaseour and liquid effluents at the boundary of the unrestricted area do not exceed the values specified in Table 2 of Appendix B; and

(ii) If an individual were continually present in an unrestricted area, the dose from external sources would not exceed 0.002 rem (0.02 mSv) in an hour and 0.05 rem (0.5 mSv) in a year.

(c) Upon approval from the Commission, the licensee may adjust the effluent concentration values in Appendix B, Table 2, for members of the public, to take into account the actual physical and chemical characteristics of the effluents (e.g., aerosol size distribution, solubility, density, radioactive decay equilibrium, chemical form).  $\sum_{i=1}^{n}$ 

#### SUBPART E--[RESERVED]

#### SUBPART F--SURVEYS AND MONITORING

#### § 20.501 General.

(a) Each licensee shall make or cause to be made, surveys that --

(1) May be necessary for the licensee to comply with the regulations in this part; and

[7590-01]

(2) Are reasonable under the circumstances to evaluate:

(i) The extent of radiation levels; and

(ii) Concentrations or quantities of radioactive material; and

(iii) The potential radiological hazards that could be present.

(b) The licensee shall ensure that instruments and equipment used for quantitative radiation measurements (e.g., dose rate and effluent monitoring) are calibrated periodically for the radiation measured.

(c) All personnel dosimeters (except for direct and indirect reading pocket ionization chambers and those dosimeters used to measure the dose to the extremities) that require processing to determine the radiation dose and that are used by licensees to comply with § 20.201, with other applicable provisions of this chapter, or with conditions specified in a license must be processed and evaluated by a dosimetry processor--

(1) Holding current personnel dosimetry accreditation from the National Voluntary Laboratory Accreditation Program (NVLAP) of the <u>National Institute of</u> <u>Standards and Technology</u>; and

(2) Approved in this accreditation process for the type of radiation or radiations included in the NVLAP program that most closely approximates the type of radiation or radiations for which the individual wearing the dosimeter is monitored.

# § 20.502 <u>Conditions requiring individual monitoring of external and</u> internal occupational dose .

Each licensee shall monitor exposures to radiation and radioactive material at levels sufficient to demonstrate compliance with the occupational dose limits of this part. As a minimum--

(a) Each licensee shall monitor occupational exposure to radiation and shall supply and require the use of individual monitoring devices by--

(1) Adults likely to receive, in 1 year from sources external to the body, a dose in excess of 10 percent of the limits in § 20.201(a),

(2) Minors and declared pregnant women likely to receive, in 1 year from sources external to the body, a dose in excess of 10 percent of any of the applicable limits in §§ 20.207 or 20.208, and

[7590-01]

(3) Individuals entering a high or very high radiation area.

(b) Each licensee shall monitor (see § 20.204) the occupational intake of radioactive material by and assess the committed effective dose equivalent to--

 Adults likely to receive, in 1 year, an intake in excess of 10 percent of the applicable ALI(s) in Table 1, Columns 1 and 2, of Appendix B; and

(2) Minors and declared pregnant women likely to receive, in 1 year, a committed effective dose equivalent in excess of 0.05 rem (0.5 mSv).

### SUBPART G--CONTROL OF EXPOSURE FROM EXTERNAL SOURCES IN RESTRICTED AREAS

### § 20.601 Control of access to high radiation areas.

(a) The licensee shall ensure that each entrance or access point to a high radiation area has one or more of the following features--

(1) A control device which, upon entry into the area, causes the level of radiation to be reduced below that level at which an individual might receive a deep-dose equivalent of 0.1 rem (1 mSv) in 1 hour at 30 centimeters from the  $267\rho^{9}$  radiation source or from any surface that the radiation penetrates;

(2) A control device that energizes a conspicuous visible or audible alarm signal so that the individual entering the high radiation area and the supervisor of the activity are made aware of the entry; or

(3) Entryways that are locked, except during periods when access to the areas is required, with positive control over each individual entry.

(b) In place of the controls required by paragraph (a) of this section for a high radiation area, the licensee may substitute continuous direct or electronic surveillance that is capable of preventing unauthorized entry.

(c) A licensee may apply to the Commission for approval of alternative methods for controlling access to high radiation areas.

(d) The licensee shall establish the controls required by paragraphs (a) and (c) of this section in a way that does not prevent individuals from leaving a high radiation area.

(e) Control is not required for each entrance or access point to a room or other area that is a high radiation area solely because of the presence of radioactive materials prepared for transport and packaged and labeled in accordance with the regulations of the Department of Transportation provided that--

(1) The packages do not remain in the area longer than 3 days; and

(2) The dose rate at 1 meter from the external surface of any package does not exceed 0.01 rem (0.1 mSv) per hour.

(f) Control of entrance or access to rooms or other areas in hospitals is not required solely because of the presence of patients containing radioactive material, provided that there are personnel in attendance who will take the necessary precautions to prevent the exposure of individuals to radiation or radioactive material in excess of the limits established in this part and to operate within the ALARA provisions of the licensee's radiation protection program.

# § 20.602 Control of access to very high radiation areas .

-

In addition to the requirements in § 20.601, the licensee shall institute additional measures to ensure that an individual is not able to gain unauthorized or inadvertent access to areas in which radiation levels could be encountered at 500 rads (5 grays) or more in 1 hour at 1 meter from a radiation source or any surface through which the radiation penetrates.

space intentionally blank as long footnote would not fit with headings and other text

§ 20.603 Control of access to very high radiation areas - irradiators.

(a) Each area in which there may exist radiation levels in excess of 500 rads (5 grays) in 1 hour at 1 meter from a sealed radioactive source<sup>2</sup> that is used to irradiate materials must meet the following requirements.

(1) Each entrance or access point must be equipped with entry control devices which--

(i) Function automatically to prevent any individual from inadvertently entering the area when very high radiation levels exist.

(ii) Permit deliberate entry into the area only after a control device is actuated that causes the radiation level within the area, from the sealed source, to be reduced below that at which it would be possible for an individual to receive a <u>deep-dose equivalent</u> in excess of 0.1 rem (1 mSv) in 1 hour; and

(iii) Prevent operation of the source if the source would produce radiation levels in the area that could result in a deep-dose equivalent to an individual in excess of 0.1 rem (1 mSv) in 1 hour.

(2) Additional control devices must be provided so that, upon failure of the entry control devices to function as required by paragraph (a)(1) of this section--

(i) The radiation level within the area, from the sealed source, is reduced below that at which it would be possible for an individual to receive a <u>deep-dose equivalent</u> in excess of 0.1 rem (1 mSv) in 1 hour; and

(ii) Conspicuous visible and audible alarm signals are generated to make an individual attempting to enter the area aware of the hazard and at least one other authorized individual, who is physically present, familiar with the

This section applies to radiation from byproduct, source, or special nuclear materials that are used in sealed sources in non-self-shielded irradiators. This section does not apply to radioactive sources that are used in teletherapy, in radiography, or in completely self-shielded irradiators in which the source is both stored and operated within the same shielding radiation barrier and, in the designed configuration of the irradiator, is always physically inaccessible to any individual and cannot create high levels of radiation in an area that is accessible to any individual. This section also does not apply to sources from which the radiation is incidental to some other use or to nuclear reactorgenerated radiation.

267 FF 9

67

activity, and prepared to render or summon assistance, aware of the failure of the entry control devicea.

(3) The licensee shall provide control devices so that, upon failure or removal of physical radiation barriers other than the source's shielded storage container--

(i) The radiation level from the source is reduced below that at which it would be possible for an individual to receive a <u>deep-dose equivalent</u> in excess of 0.1 rem (1 mSv) in 1 hour; and

(ii) Conspicuous visible and audible alarm signals are generated to make potentially affected individuals aware of the hazard and the licensee or at least one other individual, who is familiar with the activity and prepared to render or summon assistance, aware of the failure or removal of the physical barrier.

(4) When the shield for the stored source is a liquid, the licensee shall provide means to monitor the integrity of the shield and to signal, automatically, loss of adequate shielding.

(5) Physical radiation barriers that comprise permanent structural components, such as walls, that have no credible probability of failure or removal in ordinary circumstances need not meet the requirements of paragraphs (a)(3) and (4) of this section.

(6) Each area must be equipped with devices that will automatically generate conspicuous visible and audible alarm signals to alert personnel in the area before the source can be put into operation and in sufficient time for any individual in the area to operate a clearly identified control device, which must be installed in the area and which can prevent the source from being put into operation.

(7) Each area must be controlled by use of such administrative procedures and such devices as are necessary to ensure that the area is cleared of personnel prior to each use of the source.

(8) Each area must be checked by a radiation measurement to ensure that, prior to the first individual's entry into the area after any use of the source, the radiation level from the source in the area is below that at which it would be possible for an individual to receive a deep-dose equivalent in excess of 0.1 rem (1 mSv) in 1 hour.

(9) The entry control devices required in paragraph (a)(1) of this section must have been tested for proper functioning (see § 20.1109 for recordkeeping requirements).

(i) Testing must be conducted prior to initial operation with the source of radiation on any day (unless operations were continued uninterrupted from the previous day), and

(ii) Testing must be conducted prior to resumption of operation of the source of radiation after any unintended interruption; and

(iii) The licensee shall submit and adhere to a schedule for periodic tests of the entry control and warning systems.

(10) The licensee may not conduct operations, other than those necessary to place the source in safe condition or to effect repairs on controls, unless control devices are functioning properly.

(11) Entry and exit portals that are used in transporting materials to and from the irradiation area, and that are not intended for use by individuals, must be controlled by such devices and administrative procedures as are necessary to physically protect and warn against inadvertent entry by any individual through these portals. Exit portals for processed materials must be equipped to detect and signal the presence of any loose radiation sources that are carried toward such an exit and to automatically prevent loose radiation sources from being carried out of the area.

(b) Persons holding licenses or applicants for licenses for radiation sources that are within the purview of paragraph (a) of this section and that will be used in a variety of positions or in locations, such as open fields or forests, that make it impracticable to comply with certain requirements of paragraph (a) of this section, such as those for the automatic control of radiation levels, may apply to the Director, Office of Nuclear Material Safety and Safeguards, U.S. Nuclear Regulatory Commission, Washington, DC 20555, for approval of the use of alternative safety measures. Any alternative safety measures must provide a degree of personnel protection at least equivalent to those specified in paragraph (a) of this section. At least one of the alternative measures must include an entry-preventing interlock control based on a measurement of the radiation that ensures the absence of high radiation levels before an individual can gain access to the area where such radiation sources are used.

(c) The entry control devices required by paragraphs (a) and (b) of this section must be established in such a way that no individual will be prevented from leaving the area.

### SUBPART H--RESPIRATORY PROTECTION AND CONTROLS TO RESTRICT INTERNAL EXPOSURE IN RESTRICTED AREAS

### § 20.701 Use of process or other engineering controls.

The licensee shall use, to the extent practicable, process or other engineering controls (e.g., containment or ventilation) to control the concentrations of radioactive material in air.

### § 20 702 Use of other controls.

When it is not practicable to apply process or other engineering controls to control the concentrations of radioactive material in air to values below those which define an airborne radioactivity area, the licensee shall, consistent with maintaining the total effective dose equivalent ALARA, increase monitoring, and limit intakes by one or more of the following means:

- (a) Control of access;
- (b) Limitation of exposure times;
- (c) Use of respiratory protection equipment; or
- (d) Other controls.

# § 20.703 Use of individual respiratory protection equipment.

(a) If the licensee uses respiratory protection equipment to limit intakes pursuant to § 20.702--

(1) The licensee shall use only respiratory protection equipment that is tested and certified or had certification extended by the National Institute for Occupational Safety and Health/Mine Safety and Health Administration (NIOSH/MSHA).

(2) If the licensee wishes to use equipment that has not been tested or certified by NIOSH/MSHA, has not had certification extended by NIOSH/MSHA, or for which there is no schedule for testing or certification, the licensee shall submit an application for authorized use of that equipment, including a demonstration by testing, or a demonstration on the basis of reliable test information, that the material and performance characteristics of the equipment are capable of providing the proposed degree of protection under anticipated conditions of use.

(3) The licensee shall implement and maintain a respiratory protection program that includes--

(i) Air sampling sufficient to identify the potential hazard, permit proper equipment selection, and estimate exposures;

(ii) Surveys and bioassays, as appropriate, to evaluate actual intakes;

(iii) Testing of respirators for operability immediately prior to each use;

(iv) Written procedures regarding selection, fitting, issuance, maintenance, and testing of respirators, including testing for operability immediately prior to each use; supervision and training of personnel; monitoring, including air sampling and bioassays; and recordkeeping; and

(v) Determination by a physician prior to initial fitting of respirators, and at least every 12 months thereafter, that the individual user is physically able to use the respiratory protection equipment.

(4) The licensee shall issue a written policy statement on respirator usage covering--

(i) The use of process or other engineering controls, instead of respirators;

(ii) The routine, nonroutine, and emergency use of respirators; and

(iii) The periods of respirator use and relief from respirator use.

(5) The licensee shall advise each respirator user that the user may leave the area at any time for relief from respirator use in the event of equipment malfunction, physical or psychological distress, procedural or communication failure, significant deterioration of operating conditions, or any other conditions that might require such relief.

148

267

10 9

267

010

(6) The licensee shall use equipment within limitations for type and mode of use and shall provide proper visual, communication, and other special capabilities (such as adequate skin protection) when needed.

(b) In estimating exposure of individuals to airborne radioactive materials, the licensee may make allowance for respiratory protection equipment used to limit intakes pursuant to § 20.702, provided that the following conditions, in addition to those in § 20.703(a), are satisfied:

(1) The licensee selects respiratory protection equipment that provides a protection factor (see Appendix A) greater than the multiple by which peak concentrations of airborne radioactive materials in the working area are expected to exceed the values specified in Appendix B, Table 1, Column 3. If the selection of a respiratory protection device with a protection factor greater than the peak concentration is inconsistent with the goal specified in § 20.702 of keeping the total effective dose equivalent ALARA, the licensee may select respiratory protection equipment with a lower protection factor only if such a selection would result in keeping the total effective dose equivalent ALARA. The concentration of radioactive material in the air that is inhaled when respirators are worn may be initially estimated by dividing the average concentration in air, during each period of uninterrupted use, by the protection factor. If the exposure is later found to be greater than estimated, the corrected value must be used; if the exposure is later found to be less than estimated, the corrected value may be used.

(2) The licensee shall obtain authorization from the Commission before assigning respiratory protection factors in excess of those specified in Appendix A. The Commission may authorize a licensee to use higher protection factors on receipt of an application that--

(i) Describes the situation for which a need exists for higher protection factors, and

(ii) Demonstrates that the respiratory protection equipment provides these higher protection factors under the proposed conditions of use.

(c) The licensee shall use as emergency devices only respiratory protection equipment that has been specifically certified or had certification extended for emergency use by NIOSH/MSHA.

149

(d) The licensee shall notify, in writing, the Director of the appropriate NRC Regional Office listed in Appendix D at least 30 days before the date that respiratory protection equipment is first used under the provisions of either § 20.703(a) or (b).

# § 20.704 <u>Further restrictions on the use of respiratory protection</u> <u>equipment</u>. La] deleted

LoThe Commission may impose restrictions in addition to those in §§ 20.702, 20.703, and Appendix A to--

(a) Ensure that the respiratory protection program of the licensee is adequate to limit exposures of individuals to airborne radioactive materials; and

[2](b) Limit the extent to which a licensee may use respiratory protection equipment instead of process or other engineering controls.

# SUBPART I--STORAGE AND CONTROL OF LICENSED MATERIAL

# § 20.801 Security of stored material.

The licensee shall secure from unauthorized removal or access licensed materials that are stored in controlled or unrestricted areas.

# § 20.802 Control of material not in storage.

The licensee shall control and maintain constant surveillance of licensed material that is in a controlled or unrestricted area and that is not in storage.

237#2

### SUBPART J -- PRECAUTIONARY PROCEDURES

### § 20.901 Caution signs.

(a) <u>Standard radiation symbol</u>. Unless otherwise authorized by the Commission, the symbol prescribed by this part shall use the colors magenta, or purple, <u>or black</u> on yellow background. The symbol prescribed by this part is the three-bladed design:

237

#### RADIATION SYMBOL

(1) Cross-hatched area is to be magenta, or purple, or black, and
(2) The background is to be yellow.

(b) Exception to color requirements for standard radiation symbol. Notwithstanding the requirements of paragraph (a) of this section, licensees are authorized to label sources, source holders, or device components containing sources of licensed materials that are subjected to high temperatures, with conspicuously etched or stamped radiation caution symbols and without a color requirement.

(c) <u>Additional information on signs and labels</u>. In addition to the contents of signs and labels prescribed in this part, the licensee may provide, on or near the required signs and labels, additional information, as appropriate, to make individuals aware of potential radiation exposures and to minimize the exposures.

§ 20.902 Posting requirements.

(a) <u>Posting of radiation areas</u>. The licensee shall post each radiation area with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, RADIATION AREA."

(b) <u>Posting of high radiation areas</u>. The licensee shall post each high radiation area with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, HIGH RADIATION AREA" or "DANGER, HIGH RADIATION AREA."

(c) <u>Posting of very high radiation areas</u>. The licensee shall post each very high radiation area with a conspicuous sign or signs bearing the radiation symbol and words "GRAVE DANGER, VERY HIGH RADIATION AREA."

(d) <u>Posting of airborne radioactivity areas</u>. The licensee shall post each airborne radioactivity area with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, AIRBORNE RADIOACTIVITY AREA" or "DANGER, AIRBORNE RADIOACTIVITY AREA."

(e) <u>Posting of areas or rooms in which licensed material is used</u> <u>or stored</u>. The licensee shall post each area or room in which there is used or stored an amount of licensed material exceeding 10 times the quantity of such material specified in Appendix C with a conspicuous sign or signs bearing the radiation symbol and the words "CAUTION, RADIOACTIVE MATERIAL(S)", or "DANGER, RADIOACTIVE MATERIAL(S)." § 20.903 Exceptions to posting requirements.

(a) A licensee is not required to post caution signs in areas or rooms containing radioactive materials for periods of less than 8 hours, if each of the following conditions is met:

(1) The materials are constantly attended during these periods by an individual who takes the precautions necessary to prevent the exposure of individuals to radiation or radioactive materials in excess of the limits established in this part; and

(2) The area or room is subject to the licensee's control.

(b) Rooms or other areas in hospitals that are occupied by patients are not required to be posted with caution signs pursuant to § 20.902 provided that--

(1) The patient is being treated with sealed sources or has been treated with unsealed radioactive materia: in quantities less than 30 millicuries, or the measured dose rate at 1 meter from the patient is less than 0.005 rem (0.05 mSv) per hour; and

(2) There are personnel in attendance who will take the necessary precautions to prevent the exposure of individuals to radiation or radioactive material in excess of the limits established in this part and to operate within the ALARA provisions of the licensee's radiation protection program.

(c) A room or area is not required to be posted with a caution sign because of the presence of a sealed source provided the radiation level at 30 centimeters from the surface of the source container or housing does not exceed 0.005 rem (0.05 mSv) per hour.

§ 20.904 Labeling containers.

(a) The licensee shall ensure that each container of licensed material bears a durable, clearly visible label bearing the radiation symbol and the words "CAUTION, RADIOACTIVE MATERIAL" or "DANGER, RADIOACTIVE MATERIAL." The label must also provide sufficient information (such as the radionuclide(s) present, an estimate of the

[7590-01]

quantity of radioactivity, the date for which the activity is estimated, radiation levels, kinds of materials, and mass enrichment) to permit individuals handling or using the containers, or working in the vicinity of the containers, to take precautions to avoid or minimize exposures.

(b) Each licensee shall, prior to removal or disposal of empty uncontaminated containers to unrestricted areas, remove or deface the radioactive material label or otherwise clearly indicate that the container no longer contains radioactive materials.

### § 20.905 Exemptions to labeling requirements.

A licensee is not required to label --

(a) Containers holding licensed material in quantities less than the quantities listed in Appendix C; or

(b) Containers holding licensed material in concentrations less than those specified in Table 3 of Appendix B; or

(c) Containers attended by an individual who takes the precautions necessary to prevent the exposure of individuals in excess of the limits established by this part; or

(d) Containers when they are in transport and packaged and labeled in accordance with the regulations of the Department of Transportation,  $^{\rm 3}$  or

(e) Containers that are accessible only to individuals authorized to handle or use them, or to work in the vicinity of the containers, if the contents are identified to these individuals by a readily available written record (examples of containers of this type are containers in locations such as water-filled canals, storage vaults, or hot cells). The record must be retained as long as the containers are in use for the purpose indicated on the record; or

<sup>3</sup> Labeling of packages containing radioactive materials is required by the Department of Transportation (DOT) if the amount and type of radioactive material exceeds the limits for an excepted quantity or article as defined and limited by DOT regulations 49 CFR 173.403(m) and (w), and 173.421-424.

(f) Installed manufacturing or process equipment, such as reactor components, piping, and tanks.

§ 20.906 Procedures for receiving and opening packages.

(a) Each licensee who expects to receive a package containing quantities of radioactive material in excess of a Type A quantity, as defined in § 71.4 and Appendix A to Part 71 of this chapter, shall make arrangements to receive---

(1) The package when the carrier offers it for delivery; or

(2) Notification of the arrival of the package at the carrier's terminal and to take possession of the package expeditiously.

(b) Each licensee shall monitor the external surfaces of a package known to cont in radioactive material for radicactive contamination and radiation levels if the package--

(1) Is labeled as containing radioactive material; or

(2) Has evidence of potential contamination, such as packages that are crushed, wet, or damaged.

(c) The licensee shall perform the monitoring required by paragraph (b) of this section as soon as practicable after receipt if the package, but not later than 3 hours after the package is received at the licensee's facility if it is received during the licensee's normal working hours, or not later than 3 hours from the beginning of the next working day if it is received after working hours.

(d) The licensee shall immediately notify the final delivery carrier and, by telephone and telegram, mailgram, or facsimile, the Administrator of the appropriate NRC Regional Office listed in Appendix D when--

(1) Removable radioactive surface contamination exceeds the limits of § 71.87(i) of this chapter; or

(2) External radiation levels exceed the limits of § 71.47 of this chapter.

(e) Each licensee shall--

(1) Establish, maintain, and retain written procedures for safely opening packages in which radioactive material is received; and

(2) Ensure that the procedures are followed and that due consideration is given to special instructions for the type of package being opened.

237

#4

(f) Licensees transferring special form sources in licensee-owned or licensee-operated vehicles to and from a work site are exempt from the contamination monitoring requirements of (b), but are not exempt from the survey requirement in (b) for measuring radiation levels which is required to ensure that the source is still properly lodged in its shield.

### SUBPART K--WASTE DISPOSAL

§ 20.1001 General requirements.

(a) A licensee shall dispose of licensed material only--

(1) By transfer to an authorized recipient as provided in § 20.1006 or in the regulations in Parts 30, 40, 60, 61, 70, or 72 of this chapter; or

(2) By decay in storage; or

(3) By release in effluents within the limits in § 20.301; or

(4) As authorized under §§ 20.1002, 20.1003, 70.1004, or 20.1005.

(b) A person must be specifically licensed to receive waste containing licensed material from other persons for:

(1) Treatment prior to disposal; or

(2) Treatment or disposal by incineration; or

(3) Decay in storage; or

(4) Disposal at a land disposal facility licensed under Part 61 of this chapter; or

(5) Disposal at a geologic repository under Part 60 of this chapter.

# § 20.1002 Method for obtaining approval of proposed disposal procedures.

A licensee or applicant for a license may apply to the Commission for approval of proposed procedures, not otherwise authorized in the regulations in this chapter, to dispose of licensed material generated in the licensee's activities. Each application shall include:

(a) A description of the waste containing licensed material to be disposed of, including the physical and checical properties important to risk

evaluation, and the proposed manner and conditions of waste disposal; and

(b) An analysis and evaluation of pertinent information on the nature of the environment; and

(c) The nature and location of other potentially affected licensed and unlicensed facilities., and

(d) Analyses and procedures to ensure that doses are maintained ALARA and within the dose limits in this part.

§ 20.1003 Disposal by release into sanitary sewerage.

(a) A licensee may discharge licensed material into sanitary sewerage if each of the following conditions is satisfied:

(1) The material is readily soluble (or is readily dispersible biological material) in water; and

(2) The quantity of licensed or other radicactive material that the licensee releases into the sewer in 1 month divided by the average monthly volume of water released into the sewer by the licensee does not exceed the concentration listed in Table 3 of Appendix B; and

(3) If more than one radionuclide is released, the following conditions must also be satisfied:

(i) The licensee shall determine the fraction of the limit in Table 3 of Appendix B represented by discharges into sanitary sewerage by dividing the actual monthly average concentration of each radionuclide released by the licensee into the sewer by the concentration of that radionuclide listed in Table 3 of Appendix B., and

237 #6

(ii) The sum of the fractions for each radionuclide required by paragraph (a)(3)(i) of this section does not exceed unity; and

(4) The total quantity of licensed and other radioactive material that the licensee releases into the sanitary sewerage system in a year does not exceed 5 curies (185 GBq) of hydrogen-3, 1 curie (37 GBq) of carbon-14, and 1 curie (37 GBq) of all other radioactive materials combined.

(b) Excreta from individuals undergoing medical diagnosis or therapy with radioactive material is not subject to the limitations contained in paragraph (a) of this section.

§ 20.1004 Treatment or disposal by incineration.

A licensee may treat or dispose of licensed material by incineration only in the amounts and forms specified in § 20.1005 or as specifically approved by the Commission pursuant to § 20.1002.

### § 20.1005 Disposal of specific wastes.

(a) A licensee may dispose of the following licensed material as if it were not radioactive:

(1) 0.05 microcurie (1.85 kBq), or less, of hydrogen-3 or carbon-14 per gram of medium used for liquid scintillation counting.

(2) 0.05 microcurie (1.85 kBq), or less, of hydrogen-3 or carbon-14 per gram of animal tissue, averaged over the weight of the entire animal.

(b) A licensee may not dispose of tissue under paragraph (a)(2) of this section in a manner that would permit its use either as food for humans or as animal feed.

(c) The licensee shall maintain records in accordance with § 20.1108.

### § 20.1006 Transfer for disposal and manifests.

(a) The requirements of this section and Appendix F are designed to control transfers of low-level radioactive waste intended for disposal at a land disposal facility (as defined in Part 61 of this chapter), establish a manifest tracking system, and supplement existing requirements concerning transfers and recordkeeping for those wastes.

(b) Each shipment of radioactive waste intended for disposal at a licensed land disposal facility must be accommanied by a shipment manifest as specified in Section I of Appendix F.

(c) Each shipment manifest must include a certification by the waste generator as specified in Section II of Appendix F.

(d) Each person involved in the transfer for disposal and disposal of waste, including the waste generator, waste collector, waste processor, and

disposal facility operator, shall comply with the requirements specified in Section III of Appendix F.

### § 20.1007 Compliance with environmental and health protection regulations

Nothing in this subpart relieves the licensee from complying with other applicable Federal, State, and local regulations governing any other toxic or hazardous properties of materials that may be disposed of under this subpart.

#### SUBPART L--RECORDS

### § 20.1101 General provisions.

(a) Each licensee shall use the units: curie, rad, rem, including multiples and subdivisions, and shall clearly indicate the units of all quantities on records required by this part.

(b) The licensee shall make a clear distinction among the quantities entered on the records required by this part (e.g., total effective dose equivalent, shallow-dose equivalent, eye dose equivalent, deep-dose equivalent, committed effective dose equivalent).

### § 20.1102 Records of radiation protection programs.

(a) Each licensee shall maintain records of the radiation protection program, including:

(1) The provisions of the program; and

(2) Audits and other reviews of program content and implementation.

(b) The licensee shall retain the records required by paragraph (a)(1) of this section until the Commission terminates each pertinent license requiring the record. The licensee shall retain the records required by paragraph (a)(2) of this section for 3 years after the record is made.

267

#### § 20.1103 Records of surveys.

(a) Each licensee shall maintain records showing the results of surveys and calibrations required by §§ 20.501 and 20.906(b) The licensee shall retain these records for 3 years after the record is made.

(b) The licensee shall retain each of the following records until the Commission terminates each pertinent license requiring the record:

(1) Records of the results of surveys to determine the dose from external sources and used, in the absence of or in combination with individual monitoring data, in the assessment of individual dose equivalents; and

(2) Records of the results of measurements and calculations used to determine individual intakes of radioactive material and used in the assessment of internal dose; and

(3) Records showing the results of air sampling, surveys, and bioassays required pursuant to § 20.703(a)(3)(i) and (ii); and

(4) Records of the results of measurements and calculations used to evaluate the release of radioactive effluents to the environment.

# § 20.1104 Determination of prior occupational dose.

(a) For each individual who may enter the licensee's restricted or controlled area and is likely to receive, in a year, an occupational dose requiring monitoring pursuant to § 20.502, the licensee shall--

(1) Determine the occupational radiation dose received during the current year; and

(2) Attempt to obtain the records of lifetime cumulative occupational radiation dose.

(b) Prior to permitting an individual to participate in a planned special exposure, the licensee shall determine--

(1) The internal and external doses from all previous planned special exposures; and

(2) All doses in excess of the limits (including doses received during accidents and emergencies) received during the lifetime of the individual.

(c) In complying with the requirements of paragraph (a) of this section, a licensee may--

(1) Accept, as a record of the occupational dose that the individual received during the current year, a written signed statement from the individual, or from the individual's most recent employer for work involving radiation exposure, that discloses the nature and the amount of any occupational dose that the individual may have received during the current year;

(2) Accept, as the record of lifetime cumulative radiation dose, an up-to-date NRC Form 4, or equivalent, signed by the individual and countersigned by an appropriate official of the most recent employer for work involving radiation exposure, or the individual's current employer (if the individual is not employed by the licensee); and

(3) Obtain reports of the individual's dose equivalent(s) from the most recent employer for work involving radiation exposure, or the individual's current employer (if the individual is not a loyed by the licensee) by telephone, telegram, electronic media, or letter. The licensee shall request a written verification of the dose data if the c thenticity of the transmitted report cannot be established.

(d) The licensee shall record the exposure history, as required by paragraph (a) of this section, on NRC Form 4, or other clear and legible record, of all the information required on that form.<sup>4</sup> The form or record must show each period in which the individual received occupational exposure to radiation or radioactive material and must be signed by the individual who received the exposure. For each period for which the licensee obtains reports, the licensee shall use the dose shown in the report in preparing NRC Form 4. For any period in which the licensee does not obtain a report, the licensee shall place a notation on NRC Form 4 indicating the periods of time for which data are not available.

<sup>4</sup> Licensees are not required to reevaluate the separate external dose equivalents and internal committed dose equivalents or intakes of radionuclides assessed under the regulations in this part in effect before January 1, 1993. Further, occupational exposure histories obtained and recorded on NRC Form 4 before January 1, 1993, would not have included effective dose equivalent, but may be used in the absence of specific information on the intake of radionuclides by the individual.

267 p10

(e) If the licensee is unable to obtain a complete record of an individual's current and previously accumulated occupational dose, the licensee shall assume--

(1) In establishing administrative controls under § 20.201(f) for the current year, that the allowable dose limit for the individual is reduced by 1.25 rems (12.5 mSv) for each quarter for which records were unavailable and the individual was engaged in activities that could have resulted in occupational radiation exposure; and

(2) That the individual is not available for planned special exposures.

(f) The licensee shall retain the records on NRC Form 4 or equivalent until the Commission terminates each pertinent license requiring this record. The licensee shall retain records used in preparing NRC Form 4 for 3 years after the record is made.

# § 20.1105 Records of planned special exposures.

(a) For each use of the provisions of § 20.206 for planned special exposures, the licensee shall maintain records that describe--

(1) The exceptional circumstances requiring the use of a planned special exposure; and

(2) The name of the management official who authorized the planned special exposure and a copy of the signed authorization, and

(3) What actions were necessary; and

(4) Why. the actions were necessary; and

(5) How doses were maintained ALARA, and

(6) What individual and collective doses were expected to result, and the doses actually received in the planned special exposure.

(b) The licensee shall retain the records until the Commission terminates each pertinent license requiring these records.

### § 20.1106 Records of individual monitoring results.

(a) <u>Recordkeeping requirement</u>. Each licensee shall maintain records of doses received by all individuals for whom monitoring was required pursuant to

267 p20

§ 20.502, and records of doses received during planned special exposures, accidents, and emergency conditions. These records<sup>5</sup> must include, when applicable--

(1) The deep-dose equivalent to the whole body, eye use equivalent, shallow-dose equivalent to the skin, and shallow-dose equivalent to the extremities, and

(2) The estimated intake or body burden of radionuclides (see § 20.202), and

(3) The committed effective dose equivalent assigned to the intake or body burden of radionuclides, and

(4) The specific information used to calculate the committed effective dose equivalent pursuant to § 20.204(c); and

(5) The total effective dose equivalent when required by § 20.202; and

(6) The total of the deep-dose equivalent and the committed dose to the organ receiving the highest total dose.

(b) <u>Recordkeeping frequency</u>. The licensee shall make entries of the records specified in paragraph (a) of this section at least annually.

(c) <u>Recordkeeping format</u>. The licensee shall maintain the records specified in paragraph (a) of this section on NRC Form 5, in accordance with the instructions for NRC Form 5, or in clear and legible records containing all the information required by NRC Form 5.

(d) <u>Privacy protection</u>. The records required under this section should be protected from public disclosure because of their personal privacy nature. These records are protected by most State privacy laws and, when transferred to the NRC, are protected by the Privacy Act of 1974, Pub.L. 93-579, 5 U.S.C. 552a, and the Commission's regulations in 10 CFR Part 9.

(e) The licensee shall maintain the records of dise to an embryo/fetus with the records of dose to the declared pregnant woman. The declaration of pregnancy shall also be kept on file, but may be maintained separately from the dose records.

5 Assessments of dose equivalent and records made using units in effect before January 1, 1993, need not be changed.

(f) The licensee shall retain each required form or record until the Commission terminates each pertinent license requiring the record.

### § 20.2107 Records of dose to individual members of the public.

 (a) Each licensee shall maintain records sufficient 0 demonstrate compliance with the dose limit for individual members of the public (see § 20.301).

(b) The licensee shall retain the records required by paragraph (a) of this section until the Commission terminates each pertinent license requiring the record.

### § 20.1108 Records of waste disposal.

(a) Each licensee shall maintain records of the disposal of licensed materials made under §§ 20.1002, 20.1003, 20.1004, 20.1005, Part 61, and disposal by buria: in soil, including burials authorized before January 28, 1981.

(b) The licensee shall retain the records required by paragraph (a) of this section until the Commission terminates each pertinent license requiring the record.

### § 20.1109 Records of testing entry control devices for very highradiation areas.

(a) Each licensee shall maintain records of tests made under

<sup>6</sup> A previous § 20.304 permitted burial of small quantities of licensed materials in soil before January 28, 1981, without specific Commission authorization.

§ 20.603(a)(9) on entry control devices for very high radiation areas. These records must include the date, time, and results of each such test of function.

(b) The licensee shall retain the records required by paragraph (a) of this section for 3 years after the record is made.

### § 20.1110 Form of records.

Each record required by this part must be legible throughout the specified retention period. The record may be the original or a reproduced copy or a microform provided that the copy or microform is authenticated by authorized personnel and that the microform is capable of producing a clear copy throughout the required retention period. The record may also be stored in electronic media with the capability for producing legible, accurate, and complete records during the required retention period. Records, such as letters, drawings, and specifications, must include all pertinent information, such as stamps, initials, and signatures. The licensee shall maintain adequate safeguards against tampering with and loss of records.

#### SUBPART M--REPORTS

# § 20.1201 Reports of theft or loss of licensed material.

(a) <u>Telephone</u> reports.

(1) Each licensee shall report by telephone as follows:

(i) Immediately after its occurrence becomes known to the licensee, any lost, stolen, or missing licensed material in an aggregate quantity equal to or greater than 1,000 times the quantity specified in Appendix C under such circumstances that it appears to the licensee that an exposure could result to persons in unrestricted areas; or

(ii) Within 30 days after the occurrence of any lost, stolen, or missing licensed material becomes known to the licensee, all licensed material in a quantity greater than 10 times the quantity specified in Appendix C that is still missing at this time.

[7590-01]

(2) Reports must be made as follows:

(i) Licensees having an installed Emergency Notification System shall make the reports to the NRC Operations Center in accordance with § 50.72 of this chapter, and

(ii) All other licensees shall make reports to the NRC Operations Center.

(b) Written reports.

(1) Each licensee required to make a report under paragraph (a) of this section shall, within 30 days after making the telephone report, make a written report setting forth the following information:

(i) A description of the licensed material involved, including kind, quantity, and chemical and physical form; and

(ii) A description of the circumstances under which the loss or theft occurred; and

(iii) A statement of disposition, or probable disposition, of the licensed material involved; and

(iv) Exposures of individuals to radiation, circumstances under which the exposures occurred, and the possible effective dose equivalent to persons in unrestricted areas; and

(v) Actions that have been taken, or will be taken, to recover the material; and

(vi) Procedures or measures that have been, or will be, adopted to ensure against a recurrence of the loss or theft of licensed material.

(2) Reports must be made as follows:

(i) For holders of an operating license for a nuclear power plant, the events included in paragraph (b) of this section must be reported in accordance with the procedures described in § 50.73(b), (c), (d), (e), and (g) of this chapter and must include the information required in paragraph (b)(1) of this section, and

(ii) All other licensees shall make reports to the Administrator of the appropriate NRC Regional Office listed in Appendix D.

(c) A duplicate report is not required under (b) if the licensee is also required to submit a report pursuant to \$ 30.55(c), 40.64(c), 50.72, 50.73, 70.52, 73.27(b), 73.67(e)(3)(vi), 73.67(g)(3)(iii), 73.71, or 150.19(c) of this chapter.
(d) Subsequent to filing the written report, the licensee shall also report any additional substantive information on the loss or theft within 30 days after the licensee learns of such information.

(e) The licensee shall prepare any report filed with the Commission pursuant to this section so that names of individuals who may have received exposure to radiation are stated in a separate and detachable part of the report.

## § 20.1202 Notification of incidents.

(a) <u>Immediate notification</u>. Notwithstanding any other requirements for notification, each licensee shall immediately report any event involving byproduct, source, or special nuclear material possessed by the licensee that may have caused or threatent to cause any of the following conditions--

(1) An individual to receive--

(i) A total effective dose equivalent of 25 rems (0.25 Sv) or more; or

(ii) An eye dose equivalent of 75 rems (0.75 Sv) or more; or

(iii) A shallow-dose equivalent to the skin or extremities of 250 rads (2.5 Gy) or more; or

(2) The release of radioactive material, inside or outside o. a restricted area, so that, had an individual been present for 24 hours, the individual could have received an intake five times the occupational annual limit on intake (the provisions of this paragraph do not apply to locations where personnel are not normally stationed during routine operations, such as hot-cells or process enclosures), or

(3) A loss of 1 working week or more of the operation of rny facilities affected; or

(4) Damage to property in excess of \$200,000.

(b) <u>Twenty-four hour notification</u>. Each licensee shall, within 24 hours of discovery of the event, report any event involving loss of control of licensed material possessed by the licensee that may have caused, or threatens to cause, any of the following conditions:

(1) An individual to receive, in a period of 24 hours--

(i) A total effective dose equivalent exceeding 5 rems (0.05 Sv), or

(ii) An eye dose equivalent exceeding 15 rems (0.15 Sv); or

(iii) A shallow-dose equivalent to the skin or extremities exceeding 50 rems (0.5 Sv); or

(2) The release of radioactive material, inside or outside of a restricted area, so that, had an individual been present for 24 hours, the individual could have received an intake in excess of one occupational annual limit on intake (the provisions of this paragraph do not apply to locations where personnel are not normally stationed during routine operations, such as hot-cells or process enclosures), or

(3) A loss of 1 day or more of the operation of any facilities affected; or

(4) Damage to property in excess of \$2,000.

(c) The licensee shall prepare any report filed with the Commission pursuant to this section so that names of individuals who have received exposure to radiation or radioactive material are stated in a separate and detachable art of the report.

(d) Reports made by licensees in response to the requirements of this section must be made as follows:

(1) Licensees having an installed Emergency Notification System shall make the reports required by paragraphs (a) and (b) of this section to the NRC Operations Center in accordance with § 50.72; and

(2) All other licensees shall make the reports required by paragraphs (a) and (b) of this section by telephone to the NRC Operations Center and by telegram, mailgram, or facsimile to the Administrator of the appropriate NRC Regional Office listed in Appendix D.

(e) The provisions of this section do not include doses that result from planned special exposures, that are within the limits for planned special exposures, and that are reported under § 20.1204.

# § 20.1203 <u>Reports of exposures, radiation levels, and concentrations of</u> radioactive material exceeding the limits.

(a) <u>Reportable events</u>. In addition to the notification required by
 \$ 20.1202, each licensee shall submit a written report within 30 days after learning of any of the following occurrences:

(1) Any incident for which notification is required by § 20.1202; or

[7590-01]

(2) Doses in excess of any of the following:

(i) The occupational dose limits for adults in § 20.201; or

(ii) The occupational dose limits for a minor in § 20.207; or

(iii) The limits for an embryo/fetus of a declared pregnant woman in
§ 20.208; or

(iv) The limits for an individual member of the public in § 20.301; or

(v) Any applicable limit in the license; or

(3) Levels of radiation or concentrations of radioactive material in--

(i) A restricted area in excess of any applicable limit in the license, or

(ii) An unrestricted area in excess of 10 times any applicable limit set forth in this part or in the license (whether or not involving exposure of any individual in excess of the limits in § 20.301), or

(4) For licensees subject to the provisions of EPA's generally applicable environmental radiation standards, levels of radiation or releases of radioactive material in excess of those standards, or of license conditions related to those standards.

(b) Contents of reports.

(1) Each report required by paragraph (a) of this section must describe the extent of exposure of individuals to radiation and radioactive material, including, as appropriate:

(i) Estimates of each individual's dose; and

(ii) The levels of radiation and concentrations of radioactive material involved; and

(iii) The cause of the elevated exposures, dose rates, or concentrations; and

(iv) Corrective steps taken or planned to ensure against a recurrence, including the schedule for achieving conformance with applicable limits, generally applicable environmental standards, and associated license conditions.

(2) Each report filed pursuant to paragraph (a) of this section must include for each individual<sup>7</sup> exposed: the name, Social Security account number, and date of birth. The report must be prepared so that this information is stated in a separate and detachable part of the report.

<sup>7</sup> With respect to the limit for the embryo/fetus (§ 20.208), the identifiers should be those of the declared pregnant woman.

(c) For holders of an operating license for a nuclear power plant, the occurrences included in paragraph (a) of this section must be reported in accordance with the procedures described in § 50.73(b), (c), (d), (e), and (g) of this chapter and must also include the information required by paragraph (b) of this section. Occurrences reported in accordance with § 50.73 of this chapter need not be reported by a duplicity sport under paragraph (a) of this section.

(d) All licensees, other than those holding an operating license for a nuclear power plant, who make reports under paragraph (a) of this section shall submit the report in writing to the U.S. Nuclear Regulatory Commission, Document Control Desk, Washington, DC 20555, with a copy to the appropriate NRC Regional Office listed in Appendix D.

# § 20.1204 Reports of planned special exposures.

The licensee shall submit a written report to the Administrator of the appropriate NRC Regional Office listed in Appendix D within 30 days following any planned special exposure conducted in accordance with § 20.206, informing the Commission that a planned special exposure was conducted and indicating the date the planned special exposure occurred and the information required by § 20.1105.

§ 20.1205 [Reserved].

# § 20.1206 Reports of individual monitoring.

(a) This section applies to each person licensed by the Commission to--

(1) Operate a nuclear reactor designed to produce electrical or heat energy pursuant to § 50.21(b) or § 50.22 of this chapter or a testing facility as defined in § 50.2 of this chapter; or

(2) Possess or use byproduct material for purposes of radiography pursuant to Parts 30 and 34 of this chapter; or

(3) Possess or use at any one time, for purposes of fuel processing, fabricating, or reprocessing, special nuclear materia? in a quantity exceeding 5,000 grams of contained uranium-235, uranium-233, or plutonium, or any combination thereof pursuant to Part 70 of this chapter; or

(4) Possess high-level radioactive waste at a geologic repository operations area pursuant to Part 60 of this chapter; or

(5) Possess spent fuel in an independent spent fuel storage installation (ISFSI) pursuant to Part 72 of this chapter; or

(6) Receive radioactive waste from other persons for disposal under Part 61 of this chapter; or

(7) Possess or use at any time, for processing or manufacturing for distribution pursuant to Parts 30, 32, 33 or 35 of thi. Thapter, byproduct material in quantities exceeding any one of the following quantities:

No second se	Quantity of F	adionuclide <sup>a</sup> in curies
Cesium-137 Cobalt-60 Gold-198		1
lodine-131 Iridium-192		100 1 10
Promethium-147 Technetium-99m		1,000 10 1,000

<sup>a</sup>The Commission may require as a license condition, or by rule, regulation, or order pursuant to § 20.1302, reports from licensees who are licensed to use radionuclides not on this list, in quantities sufficient to cause comparable radiation levels.

(b) Each licensee in a category listed in paragraph (a) of this section shall submit an annual report of the results of individual monitoring carried out by the licensee for each individual for whom monitoring was required by § 20.502 during that year. The licensee may include additional data for individuals for whom monitoring was provided but not required. The licensee shall use Form NRC 5 or electronic media containing all the information required by Form NRC 5.

(c) The licensee shall file the report required by § 20.1206(b), covering the preceding year, on or before April 30 of each year. The licensee shall

171

submit the report to the Director, Office of Nuclear Regulatory Research, Nuclear Regulatory Commission, Washington, DC 20555.

# SUBPART N--EXEMPTIONS AND ADDITIONAL REQUIREMENTS

# § 20.1301 Applications for exemptions.

The Commission may, upon application by a licensee or upon its own initiative, grant an exemption from the requirements of the regulations in this part if it determines the exemption is authorized by law and would not result in undue hazard to life or property.

# § 20.1302 Additional requirements.

The Commission may, by rule, regulation, or order, impose requirements on a licensee, in addition to those established in the regulations in this part, as it deems appropriate or necessary to protect health or to minimize danger to life or property.

## SUBPART O--ENFORCEMENT

## § 20.1401 Violations.

(a) The Commission may obtain an injunction or other court order to prevent a violation of the provisions of--

(1) The Atomic Energy Act of 1954, as amended;

(2) Title II of the Energy Reorganization Act of 1974, as amended; or

(3) A regulation or order issued pursuant to those Acts.

(b) The Commission may obtain a court order for the payment of a civil penalty imposed under Section 234 of the Atomic Energy Act:

(1) For violations of --

(i) Sections 53, 57, 62, 63, 81, 82, 101, 103, 104, 107, or 109 of the Atomic Energy Act of 1954, as amended;

(ii) Section 206 of the Energy Reorganization Act;

(iii) Any rule, regulation, or order issued pursuant to the sections specified in paragraph (b)(1)(i) of this section.

(iv) Any term, condition, or limitation of any license issued under the sections specified in paragraph (b)(1)(i) of this section.

(2) For any violation for which a license may be revoked under Section 186 of the Atomic Energy Act of 1954, as amended.

(c) Any. person who willfully violates a provision of the Atomic Energy Act or regulation or order issued under the requirements of that Act may be guilty of a crime and, upon conviction, be punished by fine or imprisonment or both, as provided by law.

## APPENDIX A

# PROTECTION FACTORS FOR RESPIRATORS<sup>8</sup>

	Description <sup>b</sup>		Protect	ion Factor	s	Tested & Certified Equipment	
			Modes <sup>C</sup>	Particu- lates only	Particu- lates, gases, & vapors	National Institute for Occupational Safety and Health / Mine Safety and Health Administration tests for permissibility	
1.	AIR	-PURIFYING RESPIRATORS				na na sana ana ang kananan kanan na sanan kanan ka	
	Face Face ful	epiece, half~mask <sup>9</sup> epiece, full epiece, half~mask 1, or hood	NP NP PP	10 50 1000		30 CFR Part 11, Subpart K.	
11.	ATM	DSPHERE-SUPPLYING PIRATORS					
	1.	Air-line respirator					
		Facepiece, half-mask Facepiece, half-mask Facepiece, full Facepiece, full Facepiece, full Hood Suit	CF D D D D CF CF		1000 5. 2000 5 2000 h i	30 CFR Part 11, Subpart J. J	
	2.	Self-contained breathing apparatus (SCBA)					
		Facepiece, full Facepiece, full Facepiece, full Facepiece, full	D PD RD RP		10,000 <sup>k</sup> 5,000 <sup>1</sup>	30 CFR Part 11, Subpart H.	
III.	COMBINATION RESPIRATORS						
	Any combination of air- purifying and atmosphere- supplying respirators			Protection for type a of operati listed abo	factor nd mode on as ve	30 CFR Part 11, §11.63(b).	

## FOOTNOTES

- a. For use in the selection of respiratory protective devices to be used only where the contaminants have been identified and the concentrations (or possible concentrations) are known.
- b. Only for shaven faces and where nothing interferes with the seal of tight-fitting facepieces against the skin. (Hoods and suits are excepted.)
- c. The mode symbols are defined as follows:
  - CF = continuous flow
  - D = demand
  - NP = negative pressure (i.e., negative phase during inhalation)
  - PD = pressure demand (i.e., always positive pressure)
  - PP = positive pressure
  - RD = demand, recirculating (closed circuit)
  - RP = pressure demand, recirculating (closed circuit)
- d.1. The protection factor is a measure of the degree of protection afforded by a respirator, defined as the ratio of the concentration of airborne radioactive material outside the respiratory protective equipment to that inside the equipment (usually inside the facepiece) under conditions of use. It is applied to the ambient airborne concentration to estimate the concentrations inhaled by the wearer according to the following formula:

Concentration inhaled = Ambient airborne concentration Protection factor

- 2. The protection factors apply:
  - (a) Only for individuals trained in using respirators and wearing properly fitted respirators that are used and maintained under supervision in a well-planned respiratory protective program.
  - (b) For air-purifying respirators only when high efficiency particulate filters (above 99.97% removal efficiency by thermally generated 0.3 µm dioctyl phthalate (DOP) test or equivalent) are used in atmospheres not deficient in oxygen and not containing radioactive gas or vapor respiratory hazards.

- (c) No adjustment is to be made for the use of sorbents against radioactive material in the form of gases or vapors.
- (d) For atmosphere-supplying respirators only when supplied with adequate respirable air. Respirable air shall be provided of the quality and quantity required in accordance with NIOSH/MSHA certification (described in 30 CFR Part 11). Oxygen and air shall not be used in the same apparatus.
- e. Excluding radioactive contaminants that present an absorption or submersion hazard. For tritium oxide, approximately one-third of the intake occurs by absorption through the skin so that an overall protection factor of less than 2 is appropriate when atmospheresupplying respirators are used to protect against tritium oxide. If the protection factor for a device is 5 the effective protection factor for tritium is about 1.4; for devices with protection factors of 10 the effective factor for tritium oxide is about 1.7, and for devices with protection factors of 100 or more the effective factor for tritium oxide is about 1.9. Air-purifying respirators are not suitable for protection against tritium oxide. See also footnote i concerning supplied-air suits.
- f. Canisters and cartridges shall not be used beyond service-life limitations.
- g. Under-chin type only. This type of respirator is not satisfactory for use where it might be possible (e.g., if an accident or emergency were to occur) for the ambient airborne concentrations to reach instantaneous values greater than 10 times the pertinent values in Table 1, Column 3 of Appendix B of this part. This type of respirator is not suitable for protection against plutonium or other high-toxicity materials. The mask is to be tested for fit prior to use, each time it is donned.

h.1. Equipment shall be operated in a manner that ensures that proper air flow-rates are maintained. A protection factor of no more than 1000 may be utilized for tested-and-certified supplied-air hoods when a minimum air flow of 6 cubic feet (0.17 cubic meters) per minute is maintained and calibrated airline pressure gauges or flow measuring devices are used. A protection factor of up to 2000 may be used for tested and certified hoods only when the air flow is maintained at the manufacturer's recommended maximum rate for the equipment, this rate is greater than 6 cubic feet (0.17 cubic meters) per minute, and calibrated airline pressure gauges or flow measuring devices are used.

2. The design of the supplied-air hood or helmet (with a minimum flow of 6 cfm (0.17 m<sup>3</sup> per minute) of air) may determine its overall efficiency and the protection it provides. For example, some hoods aspirate contaminated air into the breathing zone when the wearer works with hands-over-head. This aspiration may be overcome if a short cape-like extension to the hood is worn under a coat or overalls. Other limitations specified by the approval agency shall be considered before using a hood in certain types of atmospheres (see footnote i).

- i. Appropriate protection factors shall be determined, taking into account the design of the suit and its permeability to the contaminant under conditions of use. There shall be a s... by rescue person equipped with a respirator or other apparatus appropriate for the potential hazards and communications equipment whenever supplied-air suits are used.
- j. No approval schedules are currently available for this equipment. Equipment is to be evaluated by testing or on the basis of reliable test information.

- k. This type of respirator may provide greater protection and be used as an emergency device in unknown concentrations for protection against inhalation hazards. External radiation hazards and other limitations to permitted exposure, such as skin absorption, must be taken into account in such circumstances.
- Quantitative fit testing shall be performed on each individual and no more than 0.02% leakage is allowed with this type of apparatus. Perceptible outward leakage of gas from this or any positive pressure self-contained breathing apparatus is unacceptable because service life will be reduced substantially. Special training in the use of this type of apparatus shall be provided to the wearer.

Note 1: Protection factors for respirators as may be approved by the U.S. Bureau of Mines/National Institute for Occupational Safety and Health (NIOSH), according to applicable approvals for respirators for type and mode of use to protect against airborne radionuclides, may be used to the extent that they do not exceed the protection factors listed in this table. The protection factors listed in this table may not be appropriate to circumstances where chemical or other respiratory hazards exist in addition to radioactive hazards. The selection and use of respirators for such circumstances should take into account applicable approvals of the U.S. Bureau of Mines/NIOSH.

Note 2: Radioactive contaminants for which the concentration values in Table 1, Column 3 of Appendix B of this part are pased on internal dose due to inhalation may, in addition, present external exposure hizards at higher concentrations. Under these circumstances, limitations on occupancy may have to be governed by external dose limits.

ED to

Inst

other

### APPENDIX B

# ANNUAL LIMITS ON INTAKE (ALIS) AND DERIVED AIR CONCENTRATIONS (DACs) OF RADIONUCLIDES FOR OCCUPATIONAL EXPOSURE; EFFLUENT CONCENTRATIONS; CONCENTRATIONS FOR RELEASE TO SEWERAGE

## Introduction

For each radionuclide Table 1 indicates the chemical form which is to be used for selecting the appropriate ALL or DAC value. The ALIs and DACs for inhalation are given for an aerosol with an activity median aerodynamic diameter (AMAD) of 1 µm and for three classes (D,W,Y) of radioactive material, which refer to their retention (approximately days, weeks or years) in the pulmonary region of the lung. This classification applies to a range of clearance halftimes for 0 of less than 10 days, for W from 10 to 100 days, and for Y greater than 100 days. Table 2 provides concentration limits for airborne and liquid effluents released to the general environment. Table 3 provides concentration limits for discharges to sanitary sewer systems.

### Notation

The values in Tables 1, 2, and 3 are presented in the computer "E" notation. In this notation a value of 6E-02 represents a value of 6 x  $10^{-2}$  or 0.06, 6E+2 represents 6 x  $10^{-2}$  or 600, and 6E+0 represents 6 x  $10^{-2}$  or 6.

### Table 1 "Occupational"

Note that the columns in Table 1 of this appendix captioned "Oral Ingestion ALI," "Inhalation ALI," and "DAC," are applicable to occupational exposure to radioactive material.

The ALIs in this appendix are the annual intakes of given radionuclide by "Reference Man" which would result in either (1) a committed effective dose equivalent of 5 rems (stochastic ALI) or (2) a committed dose equivalent of 50 rems to an organ or tissue (non-stochastic ALI). The stochastic ALIs were derived to result in a risk, due to irradiation of organs and tissues, comparable to the risk associated with deep dose equivalent to the whole body of 5 rems. The derivation includes multiplying the committed dose equivalent to an organ or tissue by a weighting factor,  $w_{\rm T}$ . This weighting factor is the

moved

from 181

proportion of the risk of stochastic effects resulting from irradiation of the organ or tissue, T, to the total risk of stochastic effects when the whole body is irradiated uniformly. The values of  $w_T$  are listed under the definition of weighting factor in § 20.3. The non-stochastic ALIs were derived to avoid non-stochastic effects, such as prompt damage to tissue or reduction in organ function.

A value of  $w_T = 0.06$  is applicable to each of the five organs or tissues *Moved* in the "remainder" category receiving the highest dose equivalents, and the dose from equivalents of all other remaining tissues may be disregarded. The following *p* 181 parts of the GI tract -- stomach, small intestine, upper large intestine, and lower large intestine -- are to be treated as four separate organs.

Note that the dose equivalents for extremities (hands and forearms, feet and lower legs), skin, and lens of the eye are not considered in computing the committed effective dose equivalent, but are subject to limits that must be met separately.

When an ALI is defined by the stochastic dose limit, this value alone, is given. When an ALI is determined by the non-stochastic dose limit to an organ, the organ or tissue to which the limit applies is shown, and the ALI for the stochastic limit is shown in parentheses. (Abbreviated organ or tissue designations are used:

LLI wall = lower large intestine wall; St. wall = stomach wall; Blad wall = bladder wall; and Bone surf = bone surface.)

The use of the ALIs listed first, the more limiting of the stochastic and non-stochastic ALIs, will ensure that non-stochastic effects are avoided and that the risk of stochastic effects is limited to an acceptably low value. If, in a particular situation involving a radionuclide for which the nonstochastic ALI is limiting, use of that non-stochastic ALI is considered unduly

[7590-01]

conservative, the licensee may use the stochastic ALI to determine the committed effective dose equivalent. However, the licensee shall also ensure that the 50-rem dose equivalent limit for any organ or tissue is not exceeded by the sum of the external deep dose equivalent plus the internal committed dose to that organ (not the effective dose). For the case where there is no external dose contribution, this would be demonstrated if the sum of the fractions of the nonstochastic ALIs (ALI<sub>ns</sub>) that contribute to the committed dose equivalent to the organ receiving the highest dose does not exceed unity (i.e.,  $\Sigma$  (intake (in  $\mu$ Ci) of each radionuclide/ALI<sub>ins</sub>)  $\leq$  1.0). If there is an external deep dose equivalent contribution of H<sub>d</sub> then this sum must be less than 1 - (H<sub>d</sub>/50) instead of being  $\leq$  1.0.

climitrut

Note that the dose equivalents for extremities (hand and forearms, feet and lower legs), skin, and lens of the are are not considered in computing the committed effective dose equivalent, but are subject to limits that must be met separately.

The derived air concentration (DAC) values are derived limits intended to control chronic occupational exposures. The relationship between the DAC and the ALI is given by:

DAC = ALI(in  $\mu$ Ci)/(2000 hours per working year x 60 minutes/hour x 2 x 10<sup>4</sup> ml per minute) = [ALI/2.4 x 10<sup>8</sup>]  $\mu$ Ci/ml,

where 2 x  $10^4$  ml per minute is the volume of air breathed per minute at work by "Reference Man" under working conditions of "light work."

The DAC values relate to one of two modes of exposure: either external submersion or the internal committed dose equivalents resulting from inhalation of radioactive materials. Derived air concentrations based upon submersion are for immersion in a semi-infinite cloud of uniform concentration and apply to each radionuclide separately.

The ALI and DAC values relate to exposure to the single reconculide named, but also include contributions from the in-growth of any daughter radionuclide produced in the body by the decay of the parent. However, intakes that include both the parent and daughter radionuclides should be treated by the general method appropriate for mixtures. The value of ALI and DAC do not apply directly when the individual both ingests and inhales a radionuclide, when the individual is exposed to a mixture of radionuclides by either inhalation or ingestion or both, or when the individual is exposed to both internal and external irradiation ( see § 20.202). When an individual is exposed to radioactive materials which fall under several of the translocation classifications (i.e., Class D, Class W, or Class Y) of the same radionuclide, the exposure may be evaluated as if it were a mixture of different radionuclides.

It should be noted that the classification of a compound as Class D, W, or Y is based on the chemical form of the compound and does not take into account the radiological half-life of different radioisotopes. For this reason, values are given for Class D, W, and Y compounds, even for very short-lived radionuclides.

## Table 2

The columns in Table 2 of this appendix captioned "Effluents," "Air," and "Water," are applicable to the assessment and control of dose to the public, particularly in the implementation of the provisions of § 20.302. The concentration values given in Columns 1 and 2 of Table 2 are equivalent to the radionuclide concentrations which, if inhaled or ingested continuously over the course of a year, would produce a total effective dose equivalent of 0.05 rem (50 millirem or 0.5 millisieverts)

Consideration of non-stochastic limits has not been included in deriving the air and water effluent concentration limits because non-stochastic effects are presumed not to occur at the dose levels established for individual members of the public. For radionuclides, where the non-stochastic limit was governing in deriving the occupational DAC, the stochastic ALI was used in deriving the corresponding airborne effluent limit in Table 2. For this reason, the DAC and airborne effluent limits are not clways proportional as they were the previous Appendix B. The air concentration values listed in Table 2, Column 1, were derived by one of two methods. For those radionuclides for which the stochastic limit is governing, the occupational stochastic inhalation ALI was divided by 2.4 x  $10^8$ , relating the inhalation ALI to the DAC, as explained above, and then divided by a factor of 300. The factor of 300 includes the following components: a factor of 50 to relate the 5-rem annual occupational dose limit to the 0.1-rem limit for members of the public, a factor of 3 to adjust for the difference in exposure time and the inhalation rate for a worker and that for members of the public., and a factor of 2 to adjust the occupational values (derived for adults) so that they are applicable to other age groups.

For those radionuclides for which submersion (external dose) is limiting, the occupational DAC in Table 1, Column 3, was divided by 219. The factor of 219 is composed of a factor of 50, as described above, and a factor of 4.38 relating occupational exposure for 2,000 hours per year to full-time exposure (8,760 hours per year). Note that an additional factor of 2 for age considerations is not warranted in the submersion case.

The water concentrations were derived by taking the most restrictive occupational stochastic oral ingestion ALI and dividing by 7.3 x  $10^7$ . The factor of 7.3 x  $10^7$  (ml) includes the following components: the factors of 50 and 2 described above and a factor of 7.3 x  $10^5$  (ml) which is the annual water intake of "Reference Man."

Note 2 of this appendix provides groupings of radionuclides which are applicable to unknown mixtures of radionuclides. These groupings (including occupational inhalation ALIs and DACs, air and water effluent concentrations and sewerage) require demonstrating that the most limiting radionuclides in successive classes are absent. The limit for the unknown mixture is defined when the presence of the one of the listed radionuclides cannot be definitely excluded as being present either from knowledge of the radionuclide composition of the source or from actual measurements.

## Table 3 "Sewer Disposal"

The monthly average concentrations for release to sanitary sewers are applicable to the provisions in § 20.1003. The concentration values were derived by taking the most restrictive occupational stochastic oral ingestion ALI and dividing by  $7.3 \times 10^6$  (ml). The factor of  $7.3 \times 10^6$  (ml) is composed of a factor of  $7.3 \times 10^6$  (ml), the annual water intake by "Reference Man," and a factor of 10, such that the concentrations, if the sewage released by the licensee were the only source of water ingested by a reference man during a year, would result in a committed effective dose equivalent of 0.5 rem.

.

	Atomic			Atomic	
Name	Symbol	Number	Name	Symbol	Number
Actinium	Ac	89	Mancuny	Ha	00
Aluminum	Al	13	Molyhdonum	ng	00
Americium	Am	65	Noodumium	MO	42
Antimony	Sh	51	Neodymium	NQ	60
Argon	Ar	10	Neptunium	ND	93
Arsenic	Ac	10	Nickei	NI	28
Astatine	A+	30	NIODIUM	ND	41
Rarium	P.	60	Usmium	Os	76
Barkalium	Da	20	Palladium	Pd	46
Repullium	DK	97	Phosphorus	Ρ	15
Riemuth	be	4	Platinum	Pt	78
Promino	BI	83	Plutonium	Pu	94
Codmine	Br	35	Polonium	Po	84
Calaium	Cd	48	Potassium	K	19
Calcium	Ca	20	Praseodymium	Pr	59
Californium	Cf	98	Promethium	Pm	61
Carbon	C	6	Protactinium	Pa	91
Cerium	Ce	58	Radium	Ra	88
Cesium	Cs	55	Radon	Rn	86
Chlorine	C1	17	Rhenium	Re	75
Chromium	Cr	24	Rhodium	Rh	45
Cobalt	Co	27	Rubidium	Rh	37
Copper	Cu	29	Ruthenium	Ru	10
Curium	Cm	96	Samarium	Sm	62
Dysprosium	DV	66	Scandium	Se	02
Einsteinium	Es	99	Selectum	So	24
Erbium	Er	68	Silicon	Se ci	34
Europium	Eu	63	Silver	51	14
Fermium	Fm	100	Sodium	Mg	4/
Fluorine	F	9	Steptium	Na	11
Francium	Fr	87	Sulfur	sr	38
Gadolinium	Gd	64	Taptulum	5	16
Gallium	Ga	21	Tantalum	la	73
Germanium	Ge	20	Tellun	IC	43
Gold	Au	20	Tellurium	le	52
Hafnium	LIF.	19	lerbium	Tb	65
Holmium	Ho	12	Inallium	11	81
Hydrogen	ho	0/	Thorium	Th	90
Indium	n	1	Thulium	Tm	69
Indino	Ťu	49	Tin	Sn	50
Inidium		53	Titanium	Ti	22
Inco	Ir	77	Tungsten	W	74
Knupton	re	26	Uranium	U	92
Krypton	Kr	36	Vanadium	V	23
Lanthanum	La	57	Xenon	Xe	54
Lead	Pb	82	Ytterbium	Yb	70
Lutetium	Lu	71	Yttrium	Y	39
Magnesium	Mg	12	Zinc	Zn	30
Manganese	Mn	25	Zirconium	7 .	40
Mendelevium	Md	101		**	40

# LIST OF ELEMENTS

Radionuclide	Class	Table 1 Occupational Values			Table 2 Effluent Concentrations		Table 3 Release to Sewers
		Col. 1- Oral Ingestion ALI (µCi)	Col. 2- Inhalation ALI (µCi)	Col. 3_ DAC (µCi/ml)	Col. 1- Air (µCi/ml)	Col. 2- Water (uCi/ml)	Monthly Average
Hydrogen-3	Water, DAC includes skin absorption	8E+4	8E+4	2E-5	1E-7	1E-3	1E-2
Beryllium-7	W, all compounds except those given for Y	4E+4	2E+4	9E-6	3E-8	6E-4	6E-3
	Y, oxides, halides, and nitrates		2E+4	8E- 6	3E-8	-	-
Berj.lium-10	W, see <sup>7</sup> Be	1E+3 LLI wall	2E+2	6E-8	2E-10	-	-
	Y, see <sup>7</sup> Be	(1E+3)	- 1E+1	- 6E-9	- 25-11	2E-5	2E-4 -
Carbon-11 <sup>2</sup>	Monoxide Dioxide Compounds	- - 4E+5	1E+6 6E+5 4E+5	5E-4 3E-4 2E-4	2E-6 9E-7 6E-7	-	-
Carbon-14	Monoxide Dioxide	-	2E+6 2E+5	7E-4 9E-5	2E-6 3E-7	- -	ot-2 -
	Radionuclide Hydrogen-3 Beryllium-7 Ber;.lium-10 Carbon-11 <sup>2</sup> Carbon-14	Radionuclide       Class         Hydrogen-3       Water, DAC includes skin absorption         Beryllium-7       W, all compounds except those given for Y         Y, oxides, halides, and nitrates         Ber3.lium-10       W, see <sup>7</sup> Be         Carbon-11 <sup>2</sup> Monoxide Dioxide Compounds         Carbon-14       Monoxide Dioxide Compounds	Radionuclide       Class       Col. 1- Oral Ingestion ALI (µCi)         Hydrogen-3       Water, DAC includes skin absorption       8E+4         Beryllium-7       W, all compounds except those given for Y       4E+4         Bery.llium-7       W, see <sup>7</sup> Be       1E+3 LLI wall         y, see <sup>7</sup> Be       -         Carbon-11 <sup>2</sup> Monoxide Dioxide Compounds       -         Carbon-14       Monoxide Dioxide Compounds       -	Table 1 Occupational ValRadionuclideClass $\hline Col. 1^- \\ Oral \\ Inhalation \\ Ingestion \\ ALI \\ (\mu Ci) \\ (\mu $	Table 1 Occupational ValuesRadionuclideClassCol. 1- Oral Inhalation Ingestion ALI ( $\mu$ Ci)Col. 2- Inhalation DAC ( $\mu$ Ci)Col. 3_ DAC ( $\mu$ Ci)Hydrogen-3Water, DAC includes skin absorption8E+48E+42E-5Beryllium-7W, all compounds except those given for Y4E+42E+49E-6Y, oxides, halides, and nitrates-2E+48E-6Berg.1ium-10W, see <sup>7</sup> Be1E+3 UII wail (1E+3)2E+26E-9Carbon-11 <sup>2</sup> Monoxide Dixide-1E+65E-4 SE-4 Compounds-Carbon-14Monoxide Dixide-2E+67E-4 SE-5Carbon-14Monoxide Dixide-2E+67E-4 SE-5	Table 1 Occupational ValuesTable 1 Occupational ValuesTable 1 Occupational ValuesRadionuclideClassCol. 2- Oral Inhalation Ingestion ALI $(\mu Ci)$ Col. 3_ DACCol. 1-RadionuclideClassClassEt4Restion ALI $(\mu Ci)$ DACAir Air AirHydrogen-3Water, DAC includes skin absorptionBE+4Rest 2E+52E-5IE-7Beryllium-7W, all compounds except those given for Y4E+42E+49E-63E-8V, oxides, halides, and nitrates-2E+48E-63E-8Bery.lium-10W, see <sup>7</sup> BeIE+3 LII wall (IE+3)2E+26E-92E-10V, see <sup>7</sup> BeIE+16E-92E-11Carbon-11 <sup>2</sup> Monoxide Dioxide Compounds-1E+65E-42E-6Dioxide Dioxide Compounds-2E+67E-42E-6Dioxide Dioxide Compounds-2E+67E-42E-6Bioxide Compounds-2E+67E-42E-6	Table 1 Occupational ValuesTable 2 Effluent ConcentrationsRadionuclideClassTable 1 Occupational ValuesTable 2 Effluent ConcentrationsColl 1- Oral Ingestion ALI ( $\mu$ Ci)Coll 2- DACColl 2- DACColl 2- DACHydrogen-3Water, DAC includes skin absorptionBE+4EE+42E-5IE-7IE-3Beryllium-7W, all compounds except those given for Y4E+42E+49E-63E-86E-4Y, oxides, halides, and nitrates-2E+4BE-63E-8Bery.lium-10W, see <sup>7</sup> BeIE+3 (IE+3)2E+26E-82E-10-Y, see <sup>7</sup> BeIE+3 (IE+3)2E+26E-82E-10-Carbon-11 <sup>2</sup> Monoxide Dioxide-IE+65E-42E-6-Garbon-14Monoxide Dioxide-IE+65E-42E-6-Dioxide Compounds-2E+67E-42E-6-Zetys2E+59E-53E-7Zetys2E+67E-42E-6Ber2E+67E-42E-6Ber2E+67E-42E-6Carbon-11 <sup>2</sup> Monoxide 

## APPENDIX D

## UNITED STATES NUCLEAR REGULATORY COMMISSION REGIONAL OFFICES

	Address	Telephone (24 hours)
Region I: Connecticut, Deleware, District of Columbia, Maine, Maryland, Massachusetts, New Hampshire, New Jersey, New York, Pennsylvania, Rhode Island, and Vermont.	USNRC, Region I 475 Allendale Road King of Prussia, PA 19406	(215) 337-5000, (FTS) 346-5000.
Region II: Alabama, Florida, Georgia Kentucky, Mississippi, North Carolina, Puerto Rico, South Carolina, Tennessee, Virginia, Virgin Islands, and West Virginia.	USNRC, Region II 101 Marietta Street, NW Suite 2900 Atlanta, GA 30323	(404) 331-4503, (FTS) 841-4503.
Region III: Illinois, Indiana, Iowa, Michigan, Minnesota, Missouri, Ohio, and Wisconsin.	USNRC, Region III 799 Roosevelt Road Glen Ellyn, IL 60137	(708) 790-5500, (FTS) 388-5500.
Region IV: Arkansas, Colorado, Idaho, Kansas, Louisiana, Montana, Nebraska, New Mexico, North Dakota, Oklahoma, South Dakota, Texas, Utah, and Wyoming.	USNRC, Region IV 611 Ryan Plaza Drive Suite 1000 Arlington, TX 76011	(817) 860-8100 (FTS) 728-8100.
Region IV: Field Office	USNRC, Region IV Uranium Recovery Field Office 730 Simms Street, Suite 100a Golden, CO 80401 Mail: P.O. Box 25325 Denver, CO 80225	(303) 236-2805, (FTS) 776-2805.
Region V: Alaska, Arizona, California, Hawaii, Neveda, Oregon, Washington, and U.S. territories and possessions in the Pacific.	USNRC, Region V 1450 Maria Lane Suite 210 Walnut Creek, CA 94596	(415) 943-3700, (FTS) 463-3700.

#### APPENDIX E [RESERVED]

#### APPENDIX F

## REQUIREMENTS FOR LOW LEVEL-WASTE TRANSFER FOR DISPOSAL AT LAND DISPOSAL FACILITIES AND MANIFESTS

#### I. MANIFEST

The shipment manifest shall contain the name, address, and telephone number of the person generating the waste. The manifest shall also include the name, address, and telephone number or the name and EPA hazardous waste identification number of the person transporting the waste to the land disposal facility. The manifest must also indicate as completely as practicable: a physical description of the waste, the volume, radionuclide identity and quantity, the total radioactivity, and the principal chemical form. The solidification agent must be specified. Waste containing more than 0.1% chelating agents by weight must be identified and the weight percentage of the chelating agent estimated. Wastes classified as Class A, Class B, or Class C in § 61.55 of this chapter must be clearly identified as such in the manifest. The total quantity of the radionuclides H-3, C-14, Tc-99, and I-129 must be shown. The manifest required by this paragraph may be shipping papers used to meet Department of Transportation or Environmental Protection Agency regulations or requirements of the receiver, provided all the required information is included. Copies of manifests required by this section may be legible carbon copies or legible photocopies.

## II. CERTIFICATION

The waste generator shall include in the shipment manifest a certification that the transported materials are properly classified, described, packaged, marked, and labeled and are in proper condition for transportation according to the applicable regulations of the Department of Transportation and the Commission. An authorized representative of the waste generator shall sign and date the manifest.

#### III. CONTROL AND TRACKING

A. Any generating licensee who transfers radioactive waste to a land disposal facility or a licensed waste collector shall comply with the requirements in paragraphs A.1 through 8 of this section. Any generating licensee who transfers waste to a licensed waste processor who treats or repackages waste shall comply with the requirements of paragraphs A.4 through 8 of this section. A licensee shall:

Prepare all wastes so that the waste is classified according to § 61.55 of this chapter and meets the waste characteristics requirements in § 61.56 of this chapter.,

2. Label each package of waste to identify whether it is Class A waste. Class B waste, or Class C waste, in accordance with § 61.55 of this chapter;

 Conduct a quality control program to ensure compliance with §§ 61.55 and 61.56 of this chapter; the program must include management evaluation of audits;

 Prepare shipping manifests to meet the requirements of section I and II of this appendix;

5. Forward a copy of the manifest to the intended recipient, at the time of shipment, or deliver to a collector at the time the waste is collected, obtaining acknowledgment of receipt in the form of a signed copy of the manifest or equivalent documentation from the collector;

6. Include one copy of the manifest with the shipment;

 Retain a copy of the manifest and documentation of acknowledgment of receipt as the record of transfer of licensed material as required by Parts 30, 40, and 70 of this chapter; and

8. For any shipments or any part of a shipment for which acknowledgment of receipt has not been received within the times set forth in this section, conduct an investigation in accordance with paragraph E of this appendix.

B. Any waste collector licensee who handles only prepackaged waste shall:  Acknowledge receipt of the waste from the generator within 1 week of receipt by returning a signed copy of the manifest or equivalent documentation.,

2. Prepare a new manifest to reflect consolidated shipments; the new manifest shall serve as a listing or index for the detailed generator manifests. Copies of the generator manifests shall be a part of the new manifest. The waste collector may prepare a new manifest without attaching the generator manifests, provided the new manifest contains for each package the information specified in section <u>I</u> of this appendix. The collector licensee shall certify that nothing has been done to the waste that would invalidate the generator's certification;

267

p10

267

p10

3. Forward a copy of the new manifest to the land disposal facility operator at the time of shipment;

4. Include the new manifest with the shipment to the disposal site.,

5. Retain a copy of the manifest and documentation of acknowledgment of receipt as the record of transfer of licensed material as required by Parts 30, 40, and 70 of this chapter, and retain information from generator manifest until disposition is authorized by the Commission; and

6. For any shipments or any part of a shipment for which acknowledgment of receipt is not received within the times set forth in this section, conduct an investigation in accordance with paragraph E of this section.

C. Any licensed waste processor who treats or repackages wastes shall:

1. Acknowledge receipt of the waste from the generator within 1 week of receipt by returning a signed copy of the manifest or ec "valent documentation,

2. Prepare a new manifest that meets the requirements of sections I and II of this appendix. Preparation of the new manifest reflects that the processor is responsible for the waste;

Prepare all wastes so that the waste is classified according to
 § 61.55 of this chapter and meets the waste characteristics requirements in
 § 61.56 of this chapter;

 Label each package of waste to identify whether it is Class A waste, Class B waste, or Class C waste, in accordance with §§ 61.55 and 61.57 of this chapter; 5. Conduct a quality control program to ensure compliance with §§ 61.55 and 61.56 of this chapter. The program shall include management evaluation of audits,

6. Forward a copy of the new manifest to the disposal site operator or waste collector at the time of shipment, or deliver to a collector at the time the waste is collected obtaining acknowledgment of receipt in the form of a signed copy of the manifest or equivalent documentation by the collector.,

7. Include the new manifest with the shipment;

8. Retain copies of original manifests and new manifests and documentation of acknowledgment of receipt as the record of transfer of licensed material required by Parts 30, 40, and 70 of this chapter; and

9. For any shipment or part of a shipment for which acknowledgment is not received within the times set forth in this section, conduct an investigation in accordance with paragraph E of this section.

D. The land disposal facility operator shall:

1. Acknowledge receipt of the waste within 1 week of receipt by returning a signed copy of the manifest or equivalent documentation to the shipper. The shipper to be notified is the licensee who last possessed the waste and transferred the waste to the operator. The returned copy of the manifest or equivalent documentation shall indicate any discrepancies between materials listed on the manifest and materials received.

2. Maintain copies of all completed manifests or equivalent documentation until the Commission authorizes their disposition., and

3. Notify the shipper (i.e., the generator, the collector, or processor) and the Administrator of the nearest Commission Regional Office listed in Appendix D to this part when any shipment or part of a shipment has not arrived within 60 days after the advance manifest was received.

E. Any shipment or part of a shipment for which acknowledgment is not received within the times set forth in this section must:

1. Be investigated by the shipper if the shipper has not received notification or receipt within 20 days after transfer; and

2. Be traced and reported. The investigation shall include tracing the shipment and filing a report with the nearest Commission Regional Office listed

in Appendix D to this part. Each licensee who conducts a trace investigation shall file a written report with the appropriate NRC Regional Office within 2 weeks of completion of the investigation.

## CONFORMING AMENDMENTS

The following amendments to other parts of Chapter I of Title 10 generally update citations to 10 CFR Part 20 that are found in these other parts of the NRC regulations. Two amendments are particularly important as they go beyond updating cross-reference citations. The amendment to 10 CFR Part 2 Appendix C updates and modifies the examples of the severity levels associated with violations of 10 CFR Part 20. Because Appendix C relates to administrative policy of the Commission and because the listed violations are used as examples of different severity levels and are not all-inclusive, the Commission is issuing these Part 2 amendments in final form without public comment.

#### PART 2- RULES OF PRACTICE

The authority citation for Part 2 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

 Supplement IV -- Severity Categories of Appendix C to 10 CFR Part 2 is amended to read as follows:

> Appendix C--General Statement of Policy and Procedures for NRC Enforcement Actions

E

### Health Physics 10 CFR Part 2015

A. Severity I -- Violations involving for example:

1. Single radiation exposure of a worker in excess of 25 rems total effective dose equivalent, 75 rems to the lens of the eye, or 250 rads to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue.

2. Single radiation exposure of the embryo/fetus of a declared pregnant woman in excess of 2.5 rems total effective dose equivalent.,

3. Single radiation exposure of a minor in excess of 2.5 rems total effective dose equivalent, 7.5 rems to the lens of the eye. or 25 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Annual exposure of a member of the public in excess of 2.5 rems total effective dose equivalent;

5. Release of radioactive material to an unrestricted area at concentrations in excess of 50 times the limits for members of the public in Appendix B, Table 2 of 10 CFR Part 20; or

 Disposal of licensed material in quantities or concentrations in excess of 10 times the limits of 10 CFR 20.1003.

B. Severity II -- Violations involving for example:

1. Single radiation exposure of a worker in excess of 10 rems total effective dose equivalent, 30 rems to the lens of the eye, or 100 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue.,

Personnel overexposures and associated violations, incurred during a life-saving effort, will be treated on a case-by-case basis.

 Single radiation exposure of the embryo/fetus of a declared pregnant woman in excess of 1.0 rem total effective dose equivalent;

3. Single radiation exposure of a minor in excess of 1 rem total effective dose equivalent; 3.0 rems to the lens of the eye, or 10 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue.,

 Annual exposure of a member of the public in excess of 0.5 rem total effective dose equivalent;

5. Release of radioactive material to an unrestricted area at concentrations in excess of 10 times the limits for members of the public in Appendix B, Table 2 of 10 CFR Part 20;

6. Disposal of licensed material in quantities or concentrations in excess of five times the limits of 10 CFR 20.1003., or

 Failure to make an immediate notification as required by 10 CFR 20.1202(a)(1) or (a)(2).

C. Severity III -- Violations involving for example:

1. Single radiation exposure of a worker in excess of 5 rems total effective dose equivalent, 15 rems to the lens of the eye, or 50 rems to the skin of the whole body or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Single radiation exposure of the embryo/fetus of a declared pregnant woman in excess of 0.5 rem total effective dose equivalent;

3. Single radiation exposure of a minor in excess of 0.5 rem total effective dose equivalent; 1.5 rems to the lens of the eye, or 5 rems to the skin of the whole body, or to the feet, ankles, hands or forearms, or to any other organ or tissue;

 Worker exposure above regulatory limits when such exposure reflects a programmatic (rather than an isolated) weakness in the radiation control program;

5. Annual exposure of a member of the public in excess of 0.1 rem total effective dose equivalent (except when operation up to 0.5 rem a year has been approved by the Commission under § 20.301(c));

6. Release of radioactive material to an unrestricted area at concentrations in excess of two times the limits for members of the public in Appendix B to 10 CFR Part 20 (except when operation up to 0.5 rem a year has been approved by the Commission under § 20.301(c)),

7. Failure to make a 24-hour notification required by 10 CFR 20.1202(b) or an immediate notification required by 10 CFR 20.1201(a)(1)(i);

8. Substantial potential for exposures or releases in excess of the applicable limits in 10 CFR Part 20 whether or not such exposure or release occurs (e.g., operation of a radiation facility with a nonfunctioning interlock system or entry into high radiation areas, such as under reactor vessels or in the vicinity of exposed radiographic sources, without having performed an adequate survey),

9. Improper disposal of licensed material not covered in Severity Levels I or II;

10. Release for unrestricted use of contaminated or radioactive material or equipment that poses a realistic potential for exposure of the public to levels or doses exceeding the annual dose limits for member of the public, or that reflects a programmatic (rather than an isolated) weakness in the radiation control program;

Conduct of licensee activities by a technically unqualified person;
 or

12. Significant failure to control licensed material.

D. Severity IV -- Violations involving for example:

1. Exposures in excess of the limits of 10 CFR 20.201, 20.207 or 20.208 not constituting Severity Level I, II or III violations;

2. Release of radioactive material to an unrestricted area at concentrations in excess of the limits for members of the public in Appendix B to 10 CFR Part 20 (except when operation up to 0.5 rem a year has been approved by the Commission under § 20.301(c)),

3. A radiation dose rate in an unrestricted or controlled area in excess of 0.002 rem in any 1 hour (2 millirem/hour) or 50 millirems in a year:

 Failure to maintain and implement radiation programs to keep radiation exposures as low as is reasonably achievable;

5. Doses to a member of the public in excess of any EPA generally applicable environmental radiation standards, such as 40 CFR Part 190.

 Failure to make the 30-day notification required by 10 CFR 20.1201(a)(1)(ii) or 20.1203(a).

7. Failure to make a timely written report as required by 10 CFR Z0.1201(b), 20.1204, or 20.1206; or

8. Any other matter that has more than a minor safety, health, or environmental significance.

E. Severity V -- Violations that are of a minor safety, health, or environmental significance. PART 19 - NOTICES, INSTRUCTIONS, AND REPORTS TO WORKERS; INSPECTIONS

5. The authority citation for Part 19 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

6. Section 19.3 is amended by revising paragraph (e) to read as follows:

§ 19.3 Definitions.

(e) "Restricted area" means an area, access to which is limited by the licensee for the purpose of protecting individuals against undue risks from exposure to radiation and radioactive materials. Restricted area does not include areas used as residential quarters, but separate rooms in a residential building may be set apart as a restricted area.

7. In § 19.13, paragraph (d) is amended by changing the reference to "§20.405 and § 20.408" to read "§§20.1202, 20.1203, 20.1204 or 20.1206" and by revising paragraphs (b), (c), and (e) to read as follows:

## § 19.13 Notifications and reports to individuals.

\* \* \* \* \*

(b) Each licensee shall advise each worker annually of the worker's dose as shown in records maintained by the licensee pursuant to § 20.1106.

(c) At the request of a worker formerly engaged in licensed activities controlled by the licensee, each licensee shall furnish to the worker a report of the worker's exposure to radiation or radioactive material for each year the worker was required to be monitored under § 20.502. Such report shall be furnished within 30 days from the time the request is made, or within 30 days after the exposure of the individual has been determined by the licensee, whichever is later. This report shall cover the period of time that the worker's activities involved exposure to radiation from radioactive materials licensed by the Commission and shall include the dates and locations of licensed activities in which the worker participated during this period.

\*

(e) At the request of a worker who is terminating employment with the licensee that involved exposure to radiation or radioactive materials, during the current year, each licensee shall provide at termination to each such worker, or to the worker's designee, a written report regarding the radiation dose received by that worker from operations of the licensee during the current year or fraction thereof. If the most recent individual monitoring results are not available at that time, a written estimate of the dose shall be provided together with a clear indication that this is an estimate.

\*

\*

## PART 30 - RULES OF GENERAL APPLICABILITY TO DOMESTIC LICENSING OF BYPRODUCT MATERIAL

 The authority citation for Part 30 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

#### § 30.51 [Amended].

\*

\*

9. In § 30.51(c)(4), the reference to "§ 20.401(c)" is changed to read "§ 20.1108."

## PART 31 - GENERAL DOMESTIC LICENSES FOR BYPRODUCT MATERIAL

10. The authority citation for Part 31 continues to read in part as follows: Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

## § 31.5 [Amended].

In § 31.5(c)(10), the reference to "§§ 20.402 and 20.403" is changed to read "§§ 20.1201 and 20.1202."

## § 31.7 [Amended.].

12. In § 31.7(b), the reference to "§§ 20.402 and 20.403" is changed to read "§§ 20.1201 and 20.1202."

### § 31.10 [Amended].

13. In § 31.10(b)(1) the reference to "§ 20.301" is changed to read "§ 20.1001."

14. In § 31.10(b)(3) the reference to "§§ 20.301, 20.402, and 20.403" is changed to read "§§ 20.1001, 20.1201, and 20.1202."

#### § 31.11 Amended]

15. In § 31.11(c)(5), the reference to "§ 20.301" is changed to read "§ 20.1001."

16. In § 31.11(f), the reference to "§§ 20.301, 20.402, and 20.403" is changed to read "§§ 20.1001, 20.1201, and 20.1202."

# PART 32 - SPECIFIC DOMESTIC LICENSES TO MANUFACTURE OR TRANSFER CERTAIN ITEMS CONTAINING BYPRODUCT MATERIAL

17. The authority citation for Part 32 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

18. Section 32.51 is amended by revising paragraphs (a)(2)(ii) and (c) to read as follow:

§ 32.51 Byproduct material contained in devices for use under § 31.5;

requirements for license to manufacture or initially transfer.

(a) \* \* \*

(2) \* \* \*

(ii) Under ordinary conditions of handling, storage, and use of the device, the byproduct material contained in the device will not be released or inadvertently removed from the device, and it is unlikely that any person will receive in any period of 1 year a dose in excess of 10 percent of the annual limits specified in § 20.201(a) of this chapter., and

\* \* \* \* \*

(c) In the event the applicant desires that the general licensee under § 31.5 of this chapter, or under equivalent regulations of an Agreement State, be authorized to install the device, collect the sample to be analyzed by a specific licensee for leakage of radioactive material, service the device, test the on-off mechanism and indicator, or remove the device from installation, the applicant shall include in the application written instructions to be followed by the general licensee, estimated calendar quarter doses associated with such activity or activities, and the bases for these estimates. The submitted information must demonstrate that performance of this activity or activities by an individual untrained in radiological protection, in addition to other handling, storage, and use of devices under the general license, is unlikely to cause that individual to receive in a year a dose in excess of 10 percent of the annual limits specified in § 20.201(a) of this chapter.

§ 32.61 [Amended].

19. In § 32.61(d), the reference to "§ 20.203(a)" is changed to read
"§ 20.901(a)."

### § 32.71 [Amended].

20. In § 32.71(c)(2), the reference to "§ 20.203(a)(1)" is changed to read "§ 20.901(a)."

313

21. In § 32.71(e), the reference to "§ 20.301" is changed to read "§ 20.1001."

## PART 34 - LICENSES FOR RADIOGRAPHY AND RADIATION SAFETY REQUIREMENTS FOR RADIOGRAPHIC OPERATIONS

22. The authority citation for Part 34 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

## § 34:29 Amended

1

23. In § 34.29(a), the reference to "§ 20.203(c)(2)(ii), (2)(iii), or (4)" is changed to read "§ 20.601(a)(2), (a)(3), or (b)."

## § 34.41 [Amended].

24. In § 34.41(a), the reference to "§ 20.203(c)(2)" is changed to read "§ 20.601(a)(1), (a)(2), or (a)(3)."

## § 34.42 [Amended].

25. In § 34.42, the reference to "§ 20.204(c)" is changed to read "§ 20.903" and the reference to "§ 20.203(b) and (c)(1)" is changed to read "§ 20.902(a) and (b)."

## PART 35 - MEDICAL USE OF BYPRODUCT MATERIAL

26. The authority citation for Part 35 continues to read as follows:

267 p1/

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5941).

## § 35.92(a).

27. Change reference to "§ 20.301" to "§ 20.1001."

### § 35.315(a)(8)

28. Change reference to "§ 20.401(c)(1)" to "§ 20.1106(a)."

#### \$ 35.415.

29. Change reference to "§ 20.105(b)" to "§ 20.301(a)."

## § 35.630(a)(1).

30. Change reference to "National Bureau of Standards" to "National Institute of Standards and Technology."

## § 35.630(a)(2).

31. Change reference to "National Bureau of Standards" to "National Institute of Standards and Technology."

267

PIS

## § 35.641(a)(2)(i).

32. Change reference to "§ 20.101" to "§ 20.201."

## § 35.641(a)(2)(ii).

33. Change reference to "§ 20.105(b)" to "§ 20.301."

## § 35.641(b)(2).

34. Change reference to "§ 20.501" to "§ 20.1301."

## § 35.643 (a).

35. Change reference to "§ 20.105(b)" to "§ 20.301."

## § 35.643(a)(1).

36. Change reference to "§ 20.105(b)" to "§ 20.301."

### § 35.630(a)(2).

37. Change reference to "§ 20.105(a)" to "§ 20.301(c)."
38. Change reference to "§ 20.105(b)" to "§ 20.301(a)."

i,

PART 39 - LICENSES AND RADIATION SAFETY REQUIREMENTS FOR WELL-LOGGING

267011

e.,

39. The authority citation for Part 39 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

40. In § 39.15(a)(5)(iii)(B), the reference to "§ 20.203" is changed to read ""§ 20.901(a)."

41. In § 39.31(a)(1), the reference to "§ 20.203" is changed to read "§ 20.901(a).

42. In § 39.31(a)(2), the reference to "§ 20.203" is changed to read "§ 20.901(a)."

43. In § 39.77(b), the reference to "§§ 20.402, 20.403, and 20.405" is changed to read "§§ 20.1201 and 20.1203."

PART 40 - DOMESTIC LICENSING OF SOURCE MATERIAL

44. The authority tation for Part 40 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948 as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C. 5841).

45. Section § 40.34 is amended by revising paragraph (a)(2) to read as follows:

1 . . . . . . .

### § 40.34 Special requirements for issuance of specific licenses.

(a) \* \* \*

(2) The applicant submits sufficient information relating to the design, manufacture, prototype testing, quality control procedures, labeling or marking, proposed uses, and potential hazards of the industrial product or device to provide reasonable assurance that possession, use, or transfer of the depleted uranium in the product or device is not likely to cause any individual to receive in any period of 1 year a radiation dose in excess of 10 percent of the annual limits specified in § 20.201(a) of this chapter; and

\*

\* \* \* \*

### § 40.61 [Amended].

46. In § 40.61(c)(4), the reference to "§ 20.401(c)" is changed to read "§ 20.1108."

### Appendix A to Part 40.

47. In the Introduction to Appendix A, the reference to "§ 20.1(c)" is changed to read "§ 20.3."

## PART 50 - DOMESTIC LICENSING OF PRODUCTION AND UTILIZATION FACILITIES

48. The authority citation for Part 50 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

49. Section 50.34 is amended by revising paragraph (f)(2)(viii) to read as follows:

(f) \* \* \*

(2) \* \* \*

(viii) Provide a capability to promptly obtain and analyze samples from the reactor coolar system and containment that may contain TID-14844 source term radioactive materials without radiation exposures to any individual exceeding 5 rems to the whole body or 50 rems to the extremities. Materials to be analyzed and quantified include certain radionuclides that are indicators of the degree of core damage (e.g., noble gases, iodines and cesiums, and nonvolatile isotopes), hydrogen in the containment atmosphere, dissolved gases, chloride, and boron concentrations. (II.B.3)

50. In § 50.36a(a), the reference to "§ 20.106" is changed to read "§ 20.301" and paragraph (b) is revised to read as follows:

## § 50.36a Technical specifications on effluents from nuclear power reactors.

\*

(b) In establishing and implementing the operating procedures described in paragraph (a) of this section, the licensee shall be guided by the following considerations: Experience with the design, construction, and operation of nuclear power reactors indicates that compliance with the technical specifications described in this section will keep average annual releases of radioactive material in effluents and their resultant committed effective dose equivalents at small percentages of the values specified in § 20.301 of this chapter and in the operating license. At the same time, the licensee is permitted the flexibility of operation, compatible with considerations of health and safety, to assure that the public is provided a dependable source of power even under unusual operating conditions which may temporarily result in releases higher than such small percentages, but still within the dose values specified in § 20.301 of this chapter and in the operating license. It is expected that in using this operational flexibility under unusual operating conditions, the licensee will exert its best efforts to keep levels of radioactive material in effluents as low as is reasonably achievable. The guides set out in Appendix I provide numerical guidance on limiting conditions for operation for light-watercooled nuclear power reactors to meet the requirement that radioactive materials in effluents released to unrestricted areas be kept as low as is reasonably achievable.

51. In § 50.72 in paragraph (a), Footnote 1, the reference to "§ 20.205, § 20.403" is changed to read "§ 20.906, § 20.1202," and paragraphs (x)(2)(iv) (A) and (B) are revised to read as follows:

§ 50.72 Immediate notification requirements for operating nuclear power

	reactors.				
*	*	*	*	*	
(b)	* * *				
(2)	* * *				

(iv) (A) Any airborne radioactive release that results in concentrations in unrestricted areas that exceed 20 times the applicable concentration specified in Appendix B, Table 2, Column 1, of Part 20 of this chapter, when averaged over a time period of 1 hour.

(B) Any liquid effluent release that exceeds 20 times the applicable concentration specified in Appendix B, Table 2, Column 2, of Part 20 of this chapter at the point of entry into the receiving waters (i.e., unrestricted area) for all radionuclides except tritium and dissolved noble gases, when averaged over a time period of 1 hour. (Immediate notifications made under this paragraph also satisfy the requirements of paragraphs (a)(2) and (b)(2) of § 20.1202 of this chapter).

52. Section 50.73 is amended by revising paragraphs (a)(2)(viii)(A) and (B) and (ix) to read as follows:

§ 50.73 Licensee event report system.

(a) \* \* \* (2) \* \* \*

(viii)(A) Any airborne radioactivity release that exceeded 20 times the applicable concentrations specified in Appendix B, Table 2, Column 1, of Part

20 of this chapter in unrestricted areas, when averaged over a time period of 1 hour.

(B) Any liquid effluent release that exceeded 20 times the applicable concentrations specified in Appendix B, Table 2, Column 2, of Part 20 of this chapter at the point of entry into the receiving water (i.e., unrestricted area) for all radionuclides except tritium and dissolved noble gases, when averaged over a time period of 1 hour.

(ix) Reports submitted to the Commission in accordance with paragraph (a)(2)(viii) of this section also meet the effluent release reporting requirements of § 20.1203(a)(3) of this chapter.

## PART 61 - LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE

ŵ

53. The authority citation for Part 61 continues to read in part as follows:

<u>Authority</u>: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201); Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

§ 61.52 [Amended].

54. In § 61.52(a)(6), the reference to "§ 20.105" is changed to read "§§ 20.301 and 20.302."

### PART 70 - DOMESTIC LICENSING OF SPECIAL NUCLEAR MATERIAL

55. The authority citation for Part 70 continues to read in part as follows:

Authority: Sec. 161, 68 Stat. 948, as amended (42 U.S.C. 2201), Sec. 201, 88 Stat. 1242, as amended (42 U.S.C.5841).

§ 70.51 Amended].

¢.

]

56. In § 70.51(b)(6), the reference to "§ 20.401(c)" is changed to read "§ 20.1108."

Dated at Rockville, Maryland, this \_\_\_\_\_ day of \_\_\_\_\_ 1990. For the Nuclear Regulatory Commission.

> Samuel J. Chilk, Secretary of the Commission.

ENCLOSURE C FINAL REGULATORY ANALYSIS

1

1997 - 1944 - 1987 - 1987 -

Enclosure C

ſ

Sec. 1

- 2013

脉

PNL-6712

**Technical Evaluation Report** 

۴.

ľ

## Regulatory Analysis for the Revision of 10 CFR Part 20

S. E. Merwin	J. J. Tawil
J. A. MacLellan	R. J. Traub
M. F. Mullen	M. G. Woodruff
K. L. Swinth	

November 1988

Prepared for Division of Regulatory Applications Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission under Contract DE-AC06-76RLO 1830 NRC FIN B2918

Pacific Northwest Laboratory Operated for the U.S. Department of Energy by Battelle Memorial Institute



٢

10

## TECHNICAL EVALUATION REPORT

REGULATORY ANALYSIS FOR THE REVISION OF 10 CFR PART 20

S. E. Merwin J. A. MacLellan R. J. Traub M. G. Woodruff S. E. Merwin K. L. Swinth

J. J. Tawil

November 1988

Prepared for Division of Regulatory Applications Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission under Contract DE-AC06-76RL0 1830 NRC FIN B2918

Pacific Northwest Laboratory Richland, Washington

ABSTRACT

The revised 10 CFR Part 20, "Standards for Protection Against Radiation," would have substantial impacts on both U.S. Nuclear Regulatory Commission (NRC) and Agreement State licensees. It is estimated that the revision will cost licensees over \$170 million (present value); more than half of these costs are related to the new requirement to sum external and internal doses to demonstrate compliance with the annual whole-body effective dose. equivalent limit of 5 rem. Fuel fabrication and processing facilities will incur significant costs to reduce levels of airborne uranium to below the applicable derived air concentrations (DACs). The revision would also have substantial benefits, most notably by reducing doses to both workers and the unborn. Of the \$44 million (present value) in benefits identified in this analysis, over 80% are related to dose reductions. There were large uncertainties in these estimates, however, and the actual value of the benefit from the reduced doses could be much lower. Although the quantified benefits from the revision do not appear to outweigh the costs, many benefits identified in this analysis were not quantified and their consideration could favor revising 10 CFR Part 20 as planned.

1223

Winder

清朝

### EXECUTIVE SUMMARY

#### STATEMENT OF THE PROBLEM

The basic philosophy and scientific basis for the present 10 CFR Part 20 is over 30 years old. Newer concepts of radiation protection and updated biological data have rendered the present Part 20 outdated and inconsistent with current recommendations of national and international radiation protection organizations. Both the National Council on Radiation Protection and Measurements (NCRP) and the International Commission on Radiological Protection (ICRP) have published recommendations on limits for exposure to ionizing radiation that are in some ways significantly different from the limits promulgated in the present Part 20. In addition, the Environmental Protection Agency (EPA) has published in the Federal Register (1987) "Radiation Protection Guidance to Federal Agencies for Occupational Exposure," (referred to hereafter as Federal Guidance) signed by the President. This Federal Guidance incorporates the basic elements of both the ICRP and NCRP recommendations. In the past, it has been the Nuclear Regulatory Commission's (NRC's) policy to comply with Federal Guidance promulgated by the EPA.

The present Part 20 is in need of revision in six principal areas to become consistent with the Federal Guidance, ICRP recommendations, and/or NCRP recommendations:

- Use of the current 3 rem/quarter limit on whole-body exposures allows some (usually older) workers to receive doses greater than 5 rem/yr. The Federal Guidance includes an annual whole-body effective dose equivalent limit of 5 rem.
- 2. Under the current Part 20, internal doses are not included in the assessment of whole-body dose. It is possible for a worker to legally receive internal doses from some radionuclides that are equivalent to whole-body doses of 10 rem or more (using the newer ICRP method for risk assessment) in addition to annual external doses of 12 rem. Because the effective whole-body dose limit of 5 rem, as stated in the Federal Guidance, includes the sum of internal and external doses, under the current Part 20 a worker can legally receive an effective whole-body dose equivalent that is four or more times higher than the limit recommended in the Federal Guidance.
- 3. The current Part 20 limits on intake for some radionuclides do not reflect recent biological information or models of radionuclide retention and consequent risk. As a result, the current limits are too high for most alpha-emitting radionuclides and too low for most other radionuclides. The associated errors approach or exceed one order of magnitude in many cases.
- The current Part 20 dose limits for specific organs and parts of the body are not consistent with recent recommendations of the ICRP and NCRP.

- 5. The current Part 20 does not promulgate a dose limit for the embryo/ietus, even though such a limit is addressed in both an NRC Regulatory Guide and staff position paper. Recent studies suggest that the embryo/fetus is more biologically sensitive to radiationinduced damage from ionizing radiation, and the current dose limits for adults may not ensure adequate protection of the unborn. The Federal Guidance includes a dose limit of 0.5 rem to the embryo/ fetus.
- 6. For licensees not covered by the Uranium Fuel Cycle Standard (40 CFR 190) or the Clean Air Act Standard (40 CFR 61), there is no explicitly stated annual dose limit for individual members of the public in the current Part 20. Although the Federal Guidance does not include such a limit, the NCRP recommends an annual dose limit of 100 mrem for continuing long term exposures.

### OBJECTIVES

The principal objective of revising Part 20 is to make the NRC's "Standards for Protection Against Radiation" consistent with the recommendations of national (NCRP and EPA) and international (ICRP) organizations responsible for providing radiation protection standards. The ICRP and NCRP recommendations serve as the primary scientific basis for federal standards for protection against radiation. Also, as a matter of policy, the former Atomic Energy Commission (AEC) and the NRC have considered past Federal Radiation Council (FRC) and EPA federal guidance as binding and have implemented the guidance in their regulations.

The revision of Part 20 was also prepared to satisfy other objectives of the NRC. The revision would resolve several petitions for rulemaking and rulemaking proceedings that are still pending. These include limitation of dose to the embryo/fetus (NPRM, 1-3-75), deletion of the 5(N-18) doseaveraging formula (NPRM, 2-20-79) and petitions to lower dose limits (PRM-20-6 and PRM-20-6A). The revision would also correct some errors and inconsistencies in the current Part 20.

### ALTERNATIVES

This regulatory analysis focused on the consequences of the planned revision of Part 20. This alternative and others available to the NRC are described below.

### Alternative 1: Revise Part 20 as Planned

Adoption of this alternative would result in a comprehensive revision of 10 CFR Part 20. This would be the final step in a process initiated by the NRC almost 10 years ago to incorporate the recommendations of the ICRP into the rule. The revision would also incorporate many of the recent recommendations of the NCRP and would be consistent with the recent Federal Guidance.

### Alternative 2: No Action

Under this alternative, the NRC would terminate the existing rulemaking, publish a notice to this effect in the Federal Register, and initiate individual rulemaking actions as the situation demands. This would leave the principles underlying the current Part 20 in place and the rule would be inconsistent with ICRP and NCRP recommendations as well as the Federal' Guidance.

### Alternative 3: Modify the Revision and Proceed with the Rulemaking

Under this alterative, the NRC would modify the current revision to reduce the costs identified in this regulatory analysis. Several provisions of the revision could be modified to lessen the impact on licensees; however, some of these modifications would compromise the primary benefits of the revision.

### Alternative 4: Incorporate Only Those Provisions Necessary for Compliance with the Federal Guidance

Adoption of this alternative would require an extensize modification of the revision. The modified revision would be consistent with the Federal Guidance but would not contain some of the changes present in the currently planned revision.

#### CONSEQUENCES

### Alternative 1: Revise Part 20 as Planned

This alternative was the basis for the analysis provided in this report. The present value of the costs of the revision of Part 20 was estimated to be \$170,000,000, based on 1989 dollars, a discount rate of 10%, and a 30-year period. This estimate includes all costs that will be incurred in response to the promulgation of Part 20. If only those costs thought to be necessary for compliance are included in the cost estimate, this value would be reduced by about \$50 million.

Approximately 70% of the estimated to the revision are related to the revised provisions for monitoring, e aluating and recording internal doses. Table S.1 provides a breakdown of the costs by section of the rule.

The greatest impact of the revision is expected to be sustained by fuel fabrication facilities, where both the new Derived Air Concentration (DAC) values for uranium and the requirement to control intakes based on DACs and Annual Limits on Intakes (ALIs) will necessitate engineering modifications to reduce airborne uranium levels. It is estimated that 44% of the costs of the revision of Part 20 will be incurred by fuel fabrication facilities. Commercial nuclear power plants, medical facilities, academic/research institutions, and manufacturing and distribution facilities will incur an estimated 36%, 13%, 3%, and 2% of the costs of the revision, respectively. Other facilities would incur relatively insignificant costs. The costs to

12(a)
AL.
nal Dose 96 <sup>(a)</sup>
etus 5.2 <sup>(a)</sup>
dual Monitoring 1.4
oring 20 <sup>(a)</sup>
1ic 3.9
ring 11 <sup>(a)</sup>
10
10
edures 1.1

170

## TABLE S.1. Costs Incurred by Section of the Final Rule

(a) These costs are directly related to specific recommendations in the Federal Guidance. A total of \$144 million, or 85% of the quantified costs of the revision are directly related to those recommendations.

Viii

the NRC were estimated to be \$2,900,000, or about 2% of the total cost of ine revision. Tables 5.2 and 5.3 provide a summary of the quantified costs by facility type.

The benefits of revising Part 20 include both quantifiable and unquantifiable components. One of the principal benefits of the revision is that it would make NRC regulations consistent with ICRP and NCRP recommendations and the recent Federal Radiation Protection Guidance. This benefit is significant but cannot be measured in dollars. Other benefits of revising Part 20 are more readily quantified. These quantified benefits included an annual savings of \$880,000 in operating costs and reduced annual doses of 1200 rem and 300 rem to adult workers and the unborn, respectively. The present value of the quantified benefits was estimated to be \$44,000,000, over 80% of which was related to dose savings. The present value of the estimated dose savings was highly dependent on several assumptions, including the value of both avoiding a rem to an adult worker and avoiding a rem to a fetus (\$1000 and \$10,000, respectively).

One section of the revision was identified to have benefits that greatly exceeded the associated costs. It was estimated that Section 20.208, "Dose to an Embryo/Fetus." will result in costs of \$5,200,000 (present value) to licensees and reduced doses of 300 rem/yr to the unborn. In view of the perceived risk associated with dose to the embryo/fetus, this section of the rule is anticipated to provide a significant net benefit. The anticipated net benefit does not include consideration of the ramifications of the limits with regard to civil rights of female workers.

There were many uncertainties in the cost estimates provided in this report. A sensitivity analysis of the assumptions used to determine the net benefit of the revision suggested that the present value of the net benefit ranges from -\$40 million to -\$170 million. Based on this quantitative analysis, it appears that the revision of Part 20 does not have a favorable benefit/cost ratio. However, many of the benefits of the revision were not quantified in this analysis, and consideration of these benefits could justify the revision of Part 20 as planned.

#### Alternative 2: No Action

The primary consequence of this alternative would be that the NRC would not implement the Federal Radiation Protection Guidance. Licensees would be free to voluntarily implement some or all of the recommendations; however, they would not be required to do so by 10 CFR Part 20. Development costs already incurred by the NRC should not be considered as a consequence of this alternative.

## Alternative 3: Modify the Revision and Proceed with the Rulemaking

The primary consequences of this alternative would be twofold. First, the modifications would reduce the cost impact on licensees. Second, the modifications would lessen the consistency of the revision with the Federal Guidance. Three potential modifications to the rule are presented.

Contain Catogory	Cost, \$ Initial	Millions ( Annual Cost	1989 Dollars) Present Value of Costs	Total	Licensee Costs, ¥
Facility Lategory	33	5.3	75	44.7	(44.7) <sup>(a</sup>
Commercial Power Reactors	30	4.0	61	36.4	(81.1)
Medical	7.4	1.8	22	13.1	(94.2)
Academic/Research Institutions	1.8	0.34	4.5	2.7	(96.9)
Manufacturing and Distribution	0.53	0.24	2.6	1.6	(98.5)
Other Measuring Systems	0.31	0.043	0.65	0.4	(98.9)
Research and Test Reactors	0.65	0.004	0.62	0.4	(99.3)
Industrial Radiography	0.29	0.022	0.45	0.3	(99.6)
Well Logging	0.012	0.044	0.39	0.2	(99.8)
All Others	0.081	0.049	0.50		(100.0)
Total Licensee Costs	74.1	11.8	167.7	100%	
NRC Costs	1.4	0.21	2.9		
Total Costs	,5	12.0	170.6		

# TABLE S.2. Implementation Costs of the Part 20 Revision by Facility Type

(a) Numbers in parentheses refer to cumulative percentages.

×

--

.

.

Facility Category	Number	Present Value of Costs, \$ Millions	Present Value of Cost Per Facility, \$ Thousands	Annualized Cost Per Facility \$ Thousands/yr(a)
Fuel Fabrication and Processing	14	75	5,400	570
Commercial Power Reactors	109	61	560	59
Medical	6,506	22	3.4	0.38
Academic/Research Institutions	1,556	1.5	2.9	0.31
Manufacturing and Distribution	965	2.6	2.7	0.29
Other Measuring Systems	5,060	0.65	0.13	0.014
Research and Test Reactors	80	0.62	7.8	0.82
Industrial Radiography	851	0.45	0.53	0.056
h/ell Logging	454	0.39	0.86	0.091
All Others	1,752	0.50	0.29	0.030
All Facilities	17,347	168	9.68	1.03

INRLE 3.3. Cost Per Facility and Annualized Cost to Implement the Part 20 Revision

(a) Annualized cost factor (capital recovery factor) for JO years at a discount rate of 10% = 0.106.

P

×i

Reintroduce the provision for long-lived radionuclides. It has previously been reported that the provision present in the proposed rule would eliminate the need for facility modifications at fuel fabrication facilities. In short, this provision allowed some licensees to calculate effective dose equivalent based on annual effective dose equivalent rather than committed effective dose equivalent for intakes of radionuclides having long effective half-lives.

However, it appears that this provision would not eliminate the need for facility modifications even if they were required without the provision. That is because the provision specifically stated that a licensee must operate the facility such that "any individual is unlikely to have an intake from occupational exposure in one year in excess of the ALI value." In effect, the provision would allow licensees to calculate doses differently but not operate the facilities differently.

Another important factor is that it is likely that major facility modifications will not be required in all cases. Currently, most fuel fabrication facilities operate with airborne uranium concentrations at 10% to 25% of current Maximum Permissible Concentration (MPC) values; under the revision, licensees would be operating at about 50% to 125% of the DAC values. Apparently, licensees feel that this would be unacceptable. Based on the analysis in this report, however, very few workers would exceed the annual whole-body effective dose equivalent limit at these levels. For those who would, an increased use of respirators or appropriate job rotations would eliminate overexposures. I dividuals approaching the dose limits would be identified well in advance through air monitoring.

The reason that licensees would likely be unwilling to operate at a air concentrations near 1 DAC is that under the present system of d mitation, MPCs are often considered as limits and it is likely that the DACs will be incorrectly considered as limits as well. However, ALIs are the principal concern, and if particle size studies, studies on worker stay times in high-concentration areas, or solubility studies would show that workers can work safely under existing conditions, then proper use of the ALARA principle mandates that these studies be done. It is concluded that the provision is inconsistent with the Federal Guidance and would not result in major cost reductions in any case. Costs could be more effectively reduced through promotion of the concept that DACs need not be considered limits in all cases.

Remove the requirement to provide individual dose records to individuals. This modification would reduce the costs of the revision by an estimated \$9.1 million (present value). Because no significant quantifiable benefits were identified from this requirement, this modification would be cost-beneficial; however, it would be inconsistent with the Federal Guidance.

Revise Section 20.1106, "Records of Individual Monitoring Results." Approximately 12% of the costs of the revision will be related to this requirement. Licensees will be required to revise recordkeeping procedures to allow the proper assessment and recording of internal doses. Because few workers receive significant internal doses, these costs appear to be excessive. Some of these costs could be alleviated by relaxing the requirements for documenting internal exposures on NRC Form 5. However, this would be inconsistent with the recordkeeping requirements in the Federal Guidance.

### Alternative 4: Incorporate Only Those Provisions Necessary for Compliance with the Federal Guidance

This alternative would result in a complete overhaul of the current Part 20 revision. The revision would be rewritten to include only those provisions necessary for compliance with the Federal Guidance. The consequences would essentially be the same as the consequences of Alternative 1 because most of the costs and benefits of the planned revision are associated with provisions necessary for compliance with the Federal Guidance. Considering the costs of the revision that were unquantified, however, especially the costs associated with the revised limits for disposal to sewers, significant cost savings could be realized through this alternative.

### DECISION RATIONALE

Of the four alternatives, only Alternative 2 (no action) appears to be nonviable. Adoption of this alternative would allow the NRC's "Standards for Protection Against Radiation" (10 CFR Part 20) to remain inconsistent with the Federal Radiation Protection Guidance. This would disregard NRC policy and would contradict previous Commission comments to the EPA supporting development of the Federal Guidance.

Alternative 1 (revise Part 20 as planned) would incorporate both the Federal Radiation Protection Guidance and the ICRP and NCRP recommendations on limiting exposure to ionizing radiation. The cost of this alternative to · licensees and regulatory agencies is estimated to be approximately \$170 million (present value). However, adoption of the alternative would benefit society by reducing occupational exposures to both adult workers and the unborn being carried by pregnant workers. It was estimated that this alternative would result in reduced doses to workers of 1200 rem/yr and reduced doses to embryos/fetuses of 300 rem/yr. The present value of the benefits of the revision depends on the dollar values assigned to both adult and fetal doses and whether health effects were discounted; based on the assumptions used in this report. the present value was estimated to be \$44 million. Although the quantified benefits from this alternative do not appear to outweigh the costs, many benefits identified in this analysis were not quantified and their consideration could favor the adoption of this alternative. Also, note that relatively large uncertainties may be associated with the estimates provided in this analysis.

Alternative 3 (modify the revision and proceed with the rulemaking) is viable, provided that the modifications do not compromise the benefits from the revision. Each potential modification identified in the previous section would result in cost savings but would lessen the consistency of the revision with the Federal Guidance. Whether the individual modifications should be adopted depends primarily on subjective considerations of these competing factors. Mar Caller

Alternative 4 (incorporate only those provisions necessary for compliance with the Federal Guidance) may be desirable depending on whether the associated delay would be acceptable in view of the reduced costs. Of the costs identified in the analysis of Alternative 1, only a small fraction could be avoided by adoption of this alternative. However, this alternative would avoid many of the potential costs that were not quantified. Essentially all of the primary benefits associated with Alternative 1 would also be realized by adoption of Alternative 4.

### IMPLEMENTATION

### Alternative 1: Revise Part 20 as Planned

Considering that NRC planning has been based on adoption of this alternative. Alternative 1, if implemented, would probably result in publication of the final rule by December 31, 1988. The NRC would also allow a 5-year implementation period retroactive to the date that the proposed rule was published (January 9, 1986). This would effectively allow licensees until January 9, 1991, to implement the provisions of the revision. Granting a longer period of implementation, e.g., 5 years from the date the final rule is published, would allow licensees to defer the costs of implementation. A longer period of implementation would also provide the NRC with more time to develop and publish regulatory guides in support of the revised rule.

### Alternative 3: Modify the Revision and Proceed with the Rulemaking

If implemented, the preferred approach would be to identify the potential modifications and perform a separate cost-benefit analysis for each one. Favorable modifications would be introduced into the revision and unfavorable modifications would be dropped from consideration. Because most of the modifications would result in both reduced costs and reduced consistency with the Federal Guidance, the decisions to include each modification would require a comparison of the cost savings to the perceived importance of consistency with the Federal Guidance. Upon final modification of the rule, the rulemaking would proceed and the final rule would be published. An appropriate implementation period would be granted to licensees, and the NRC would develop and/or revise the regulatory guides required to support the revised rule.

### Alternative 4: Incorporate Only Those Provisions Necessary for Compliance with the Federal Guidance

If implemented, this alternative would result in a major interruption to the current schedule for revising Part 20. The preferred approach would be to carefully examine the current revision and eliminate or rewrite the provisions that are not necessary for compliance with the Federal Guidance. Those provisions identified as cost-beneficial tut not necessary for compliance with the Federal Guidance should remain in the revision. Once the appropriate changes are made, the revision could be published in the Federal Register. During this period, the NRC could develop and/or revise the regulatory guides required to support the revised rule.

### ACKNOWLEDGMENTS

The authors would like to thank the following: Jerry B. Martin, Matthew P. Moeller, and Michael J. Scott for their technical contributions; Jim Weber for editing the report; and Claudia H. Burk, Lisa M. Stoetzel, and Marianna Cross for typing the report. The authors would also like to acknowledge the contributions by Darrell R. Fisher and John P. Corley (retired), who provided thorough technical reviews of the report and made many valuable suggestions.

1 milt

## CONTENTS

ABSTI	RACT .			. 111
EXEC	UTIVE	SUMMARY		· · · ·
ACKN	OWLEDG	MENTS .		xv
1.0	INTRO	DUCTION	(	. 1.1
	1.1	BACKGRO	DUND	1.1
	1.2	HISTORY	OF PART 20 REVISIONS	1.1
	1.3	PURPOSE	E OF REVISION	1.2
	1.4	TECHNIC	CAL APPROACH	1.3
	1.5	CONTENT	TS OF THIS REPORT	1.4
2.0	COMP	ARISON C	OF PRESENT PART 20 WITH PART 20 REVISION	2.1
	2.1	SUMMARY	Y OF SIGNIFICANT CHANGES REGARDING INTERNAL DOSE	2.1
		2.1.1	Section 20.202: Compliance with Requirements for Summation of Internal and External Exposures .	2.1
		2.1.2	Section 20.204: Determination of Internal Exposure	2.1
		2.1.3	Section 20.502: Conditions Requiring Individual Monitoring of External and Internal Occupational Dose	2.2
	2.2	SUMMAR DOSE L	Y OF SIGNIFICANT CHANGES REGARDING OCCUPATIONAL IMITS	2.2
		2.2.1	Section 20.201: Occupational Dose Limits for Adults	2.2
		2.2.2	Section 20.206: Planned Special Exposures	2.2
		2.2.3	Section 20.208: Dose to an Embryo/Fetus	2.3
	2.3	SUMMAR DOSE T	Y OF SIGNIFICANT CHANGES REGARDING LIMITATION OF	2.3

		2.3.1 Section 20.301: Dose Limits for Individual Members of the Public	2.3
		2.3.2 Section 20.302: Compliance with Dose Limits for Individual Members of the Public	2.3
	2.4	SUMMARY OF SIGNIFICANT CHAMGES REGARDING RECORDKEEPING AND REPORTING REQUIREMENTS	2.3
		2.4.1 Section 20.1106: Records of Individual Monitoring Results	2.4
		2.4.2 Section 20.1107: Records of Dose to Individual Members of the Public	2.4
		2.4.3 Section 20.1206: Reports of Personnel Monitoring	2.4
	2.5	SUMMARY OF OTHER CHANGES	2.4
		2.5.1 Changes Regarding Control of Internal Exposure in Restricted Areas	2.4
		2.5.2 Changes Regarding Precautionary Procedures	2.4
		2.5.3 Changes Regarding Waste Disposal	2.5
		2.5.4 Changes Regarding Recordkeeping Requirements	2.5
		2.5.5 Changes Regarding Reporting Requirements	2.5
.0	CONS	EQUENCES OF REVISED DOSE EVALUATION REQUIREMENTS	3.1
	3.1	EXTERNAL DOSE EVALUATIONS	3.1
	3.2	INTERNAL DOSE EVALUATIONS	3.3
		3.2.1 Bioassay	3.4
		3.2.2 Air Monitoring	3.4
	3.3	DOSE EVALUATION COSTS	3.8
	3.4	DISTRIBUTION OF OCCUPATIONAL DOSES	3.13
		3.4.1 Fuel Fabrication Facilities	3.13
		3.4.2 Other Facilities	3.14
	3.5	RECORDKEEPING	3.15
	3.6	SUMMARY OF CONSEQUENCES	3.18

	a select as	A PROPERTY AND A PROP	4.1
1.0	CONSE	QUENCES OF REVISED LIMITS	4.1
	4.1	OCCUPATIONAL DOSE LIMITS FOR ADULTS	A 1
		4.1.1 Whole-Body Dose Equivalent Limits	4.1
		4.1.2 Other Limits and Their Consequences	4.0
	4.2	PLANNED SPECIAL EXPOSURES	4.9
		4.2.1 Use of the Planned Special Exposure Provision	4.9
		4.2.2 Effect on Distribution of Doses	4.10
		4.2.3 Cost Implications	4.10
	4.3	DOSE TO AN EMBRYD/FETUS	4.10
		4.3.1 Effect on Distribution of Doses	4.11
		4.3.2 Cost Implications	4.12
		4.3.3 Other Considerations	4.12
		DOSE LIMIT FOR INDIVIDUAL MEMBERS OF THE PUBLIC	4.13
	4.4	4.4.1 Effect on Distribution of Doses	4.13
		4.4.2 Cost 1mplications	4.14
	4.5	SUMMARY OF CONSEQUENCES	4.14
5.0	CON	SEQUENCES OF OTHER CHANGES	5.1
0.0	5.1	CONTROL OF INTERNAL EXPOSURE IN RESTRICTED AREAS	5.1
	5.2	PRECAUTIONARY PROCEDURES	5.3
	0.6	5.2.1 Labeling Requirements	5.3
		E 2 2 Posting Requirements	5.4
		5.2.2 Posting Reguling Requirements	5.4
		5.2.3 Package Handling Requirements fifthe	5.4
	5.3	3 WASTE DISPOSAL	5.5
	5.4	4 RECORDKEEPING REQUIREMENTS	5.0
	5.	5 REPORTING REQUIREMENTS	5.0
		5.5.1 Incident Reports	5.8

1,1996

- Sector

· Siles.

-41588

. Sim

÷

		5.5.2 Reports of Personnel Monitoring	5.8
		5.5.3 Reports to Individuals	5.9
		OTUED CHANGES	5.11
	5.0	E 6 1 Units of Dediction Dose	5.11
		5.6.1 Units of Radiation Dose	F 11
		5.6.2 Radiation Protection Programs	5.11
		5.6.3 Control of Access to High Radiation Areas	5.12
	5.7	SUMMARY OF CONSEQUENCES	5.12
6.0	OTHE	R CONSEQUENCES OF THE REVISION	6.1
	6.1	PERSONNEL TRAINING	6.1
	6.2	PROCEDURE REVISIONS	6.3
	6.3	NRC COSTS	6.6
		6.3.1 Development	6.7
		6.3.2 Implementation	6.7
		6.3.3 (peration	6.8
	6.4	EFFECTS ON SMALL BUSINESSES	6.9
	6.5	SUMMARY OF CONSEQUENCES	6.12
7.0	QUAL	ITATIVE BENEFITS FROM REVISING PART 20	7.1
	7.1	POTENTIAL BENEFITS FROM CONSISTENCY WITH NATIONAL AND INTERNATIONAL RECOMMENDATIONS	7.1
	7.2	POTENTIAL BENEFITS FROM REVISED MONITORING REQUIREMENTS	7.5
	7.3	POTENTIAL BENEFITS FROM REVISED RECORDKEEPING/REPORTING REQUIREMENTS	7.6
	7.4	POTENTIAL BENEFITS FROM ADDITIONS/REVISIONS TO CURRENT DOSE LIMITS	7.8
	7.5	POTENTIAL BENEFITS FROM SUPPLIED GUIDANCE FOR COMPLYING WITH THE REVISION	7.9
	7.6	POTENTIAL BENEFITS FROM DOSE LIMITS FOR PREGNANT WOMEN AND EMBRYOS/FETUSES	7.10
	7.7	POTENTIAL BENEFITS FROM REVISED LABELING REQUIREMENTS	7.10

198.1

8.0	SUMM	ARY OF COSTS AND BENEFITS 8.1
	8.1	COMPENDIUM OF NRC AND AGREEMENT STATE LICENSEES 8.1
	8.2	SUMMARY OF COSTS 8.4
		8.2.1 Summary by Facility Type 8.6
		8.2.2 Summary by Part 20 Section 8.6
		8.2.3 Discussion of Uncertainties 8.6
	8.3	SUMMARY OF BENEFITS 8.9
		8.3.1 Summary by Part 20 Section 8.9
		8.3.2 Discussion of Uncertainties
	8.4	SENSITIVITY ANALYSIS 8.10
	8.5	CONCLUSIONS 8.1
9.0	REFE	RENCES

## TABLES

5.1	Costs Incurred by Section of the Final Rule	viii
5.2	Implementation Costs of the Part 20 Revision by Facility Type	, ×
5.3	Cost Per Facility and Annualized Cost to Implement the Part 20 Revision	xi
3.1	Occupational Dose Limits Specified in the Present and Revised Part 20	3.1
3.2	Annual Limits on Intake of Common Radionuclides Compared to Detection Capabilities for Urinalysis and In Vivo Counting	3.
3.3	Comparison of Part 20 Current and Revised Air Concentration Limits for Certain Radionuclides	1
3.4	Estimated Costs of Increased Recordkeeping Requirements Resulting from the Revised Dose Evaluation Requirements	3.17
3.5	Summary of Consequences of the Revised Dose Evaluation Requirements	3.18
4.1	Doses Received by Licensee Employees in 1984	4.2
4.2	Number of Licensee Employees Reported to Have Received Doses Approaching or Exceeding 5 rem in 1979	4.3
4.3	Adjusted Number of Employees Who Received Doses >5 rem in 1979 and 1984	4.5
4.4	Number of Women Reported to Have Received Doses >0.5 rem in 1980	4.12
4.5	Summary of Consequences of the Revised Limits	4.15
5.1	Estimated Collective Dose Equivalent Savings During an Outage Through Reduced Respiratory Usage for Selected Tasks	5.2
5.2	Current and Revised Average Concentration Limits for Releases into Sanitary Sewerage	5.6
5.3	Cost of Providing Reports to Individuals by Facility Category	5.10
5.4	Summary of Marginal Cost Increases from Other Changes in the Revised Part 20	5.13
6.1	Fraction of Licensees That May Be Classified as Small Businesses	6.10

6.2	Per-facility Costs from the 10 CFR Part 20 Revision	6.10
6.3	Estimated Total Cost to Small Businesses from the 10 CFR Part 20 Revision	6.11
6.4	Summary of Marginal Cost Increases from Other Consequences of the 10 CFR Part 20 Revision	6.12
7.1	Comparison of the Revision of Part 20 with National and International Recommendations	7.3
8.1	Categories of Licensees	8.3
8.2	Number of NRC and Agreement State Licensees in 1985	8.4
8.3	Employee Data for NRC and Agreement State Licensees in 1985	8.5
8.4	Quantified Costs Incurred from the Revision of Part 20 by Facility Category	8.7
8.5	Quantified Costs Incurred from the Revision of Part 20 by Section of the Rule	8.8
8.6	Benefits of the Revision of Part 20 by Section of the Rule	8.10
8.7	Sensitivity of Net Benefits to Variations in Basic Assumptions .	8.12
8.8	Estimated Net Impact (Present Value) of the Revision of Part 20	8.13

- Martin

10.367.0

### 1.0 INTRODUCTION

The U.S. Nuclear Regulatory Commission (NRC) plans to promulgate a complete revision of Title 10, Part 20, <u>Code of Federal Regulations</u>, "Standards for Protection Against Radiation." This report presents a cost-benefit analysis of the revision which can be used by regulatory authorities to assess the impact of the revision on society. Groups affected by the revision will be NRC licensees, Agreement State licensees, NRC and Agreement State agencies, the general public, and businesses directly or indirectly related to the nuclear industry. The following four sections provide a historical perspective of Part 20 and describe the methodology used in this report to assess the impact of the revision.

### 1.1 BACKGROUND

<u>Code of Federal Regulations Title 10</u>, Part 20 provides standards for protection of workers and the public against radiation hazards arising out of activities under licenses issued by the NRC. In addition, these standards are used as the basis for the regulations promulgated by the 28 Agreement States.

The current standards are based primarily on the recommendations of the International Commission on Radiological Protection (ICRP) published in 1959 (ICRP 1959). Since that time, 10 CFR Part 20 has undergone numerous minor revisions, but most of the principles underlying the protection standards remain as they were three decades ago. In 1977, the ICRP published a new set of recommendations (ICRP 1977). These recommendations were based on a revised concept of total risk limitation and provided methods for normalizing risks from doses received by different parts of the body. This concept was adopted by both the National Council on Radiation Protection Agency (EPA) (FR 1987; EPA 1983a; EPA 1983b). If promulgated, the revision of Part 20 would incorporate the ICRP system of dose limitation.

### 1.2 HISTORY OF PART 20 REVISIONS

The NRC's standards for protection against radiation are basically unchanged from those first developed by the Atomic Energy Commission (AEC) approximately 30 years ago. These standa is were based on ICRP recommendations (ICRP 1959) and reflected the scientific understanding of the effects of radiation at that time. Since then, Part 20 has undergone numerous minor revisions, but the basic principles underlying the system of dose limitation have not changed.

Soon after the publication of the ICRP recommendations in ICRP Publication 26 (ICRP 1977), the NRC in tiated a relemaking activity to incorporate these recommendations into 10 "FR Part 20. On March 20, 1980, an "Advance Notice of Proposed Rulemakino" was published in the <u>Federal Register</u>. In 1982, the NRC prepared a Regulatory Analysis for the most recent version of the planned revision of Part 20. Subsequently, the revision underwent several iterations until a proposed revision was published in the <u>Federal Register</u> on January 9, 1986 (FR 1986). This was a corrected version of the proposed revision first published on December 20, 1985. Announcement of a 120-day comment period coincided with publication of the proposed revision.

Upon publication of the proposed revision, a draft backfit analysis to comply with Section 50.109 of 10 CFR 50 was requested by the Commission. This analysis was developed and published for public comment on August 29, 1986. Coincident with publication of the backfit analysis, the public comment period on the proposed revision was extended to October 31, 1986. This provided a 250-day comment period on the proposed revision, including a 60-day joint comment period on the proposed revision and the backfit analysis. A total of 813 comments were received during the 250-day comment period.

Based primarily on analysis of the comments on the proposed rule, the NRC performed further iterations of the revision until December 4, 1987, when the final rule language was approved by an NRC Steering Committee. Several changes to the rule were made after that date. The analysis provided in this report is based on the May 19, 1988, wording of the rule. Although further changes to the wording might occur before the Commission decides whether to promulgate the revised rule, any such changes should not affect the conclusions of this report.

### 1.3 PURPOSE OF REVISION

The primary purpose of revising Part 20 is to incorporate the basic principles of ICRP Publication 26, which were endorsed by the EPA in its Federal Radiation Protection Guidance (FR 1987). Under the current Part 20, the quarterly whole-body dose limit applies only to external doses. Internal doses are subject to separate limits based on the dose to critical organs. Under the revised Part 20, external doses and internal doses would be summed to demonstrate compliance with the annual whole-body dose limit. This system of dose limitation is consistent with current ICRP and NCRP recommendations on protection of both workers and the public from ionizing radiation (ICRP 1977; NCRP 10<sup>-7</sup>), and is also consistent with the Radiation Protection Guidance to Federal Agencies for Occupational Exposure, signed by the President (FR 1987).

The revised Part 20 would also incorporate other recommendations of the ICRP, NCRP, and EPA. Under the current Part 20, workers are allowed to receive an external whole-body dose of 3 rem per quarter, provided that their average annual dose after the age of 18 does not exceed 5 rem. Considering that internal doses are not considered in this limit, the potential exists for some workers under the current rule to receive doses significantly more than 12 rem per year. Under the revised Part 20, workers would be limited to 5 rem per year, except under special circumstances when a limit of 10 rem could be applied.

The revised Part 20 would include, for the first time, a dose limit to the embryo/fetus of pregnant workers. The ICRP, NCRP, and EPA all recommended a limit to the embryo/fetus that is separate and lower than the wholebody dose limit to adult workers. This provision in the revised Part 20 would introduce a limit that is already being applied to most radiation workers in the United States.

The revised Part 20 would correct outdated individual radionuclide intake limits in the current rule. Because the fundamental principles of the current rule are 30 years old, some of the intake limits are no longer consistent with current scientific knowledge and understanding of the behavior of radionuclides in the body. Inconsistencies associated with application of the limits would be corrected by the revision, as well.

In general, the revision of Part 20 represents a complete overhaul of the current rule. The revision would reflect current scientific knowledge of radiation protection principles and would provide a sound technical basis for the dose limits.

### 1.4 TECHNICAL APPROACH

This report documents a regulatory analysis of the impacts of the revision of Part 20. The methods and format of the analysis are consistent with both the regulatory analysis guidelines of the NRC (USNRC 1984) and accepted methods for performing value-impact assessments (Heaberlin et al. 1983). Although this analysis does not provide a backfit analysis consistent with 10 CFR 50.109, the information provided in this report is sufficient for the development of such an analysis.

The analysis provided in this report is a cost-benefit analysis of the revision of Part 20. Only the marginal costs and benefits are identified; that is, only those costs and benefits that represent additions or reductions compared to those currently being incurred under the present Part 20 are considered. The analysis is based on many sources of information, including a previous regulatory analysis of an earlier version of the revision (USNRC 1982), public comments on the regulatory analysis, public comments on the proposed rule and backfit analysis, and numerous publications relevant to specific provisions of the revision.

The current plan for implementing the revision is to allow a 5-year implementation period beginning the day the proposed rule was published (January 9, 1986). Because the final revision is not expected to be published until the end of 1988, licensees will have approximately 2 years after publication of the rule to fully implement its provisions. In this report, all quantified costs and benefits are normalized to January 1, 1989. For cost estimates obtained from cost information provided in the literature, a 5% per year increase was applied to account for inflation.

Because publication of the proposed rule in 1986 prompted many licensee, to begin "gearing up" for publication of the final rule, implementation cos's will be distributed over the 5-year implementation period. In many cases, licensees will defer implementation of the rule as long as possible. For this analysis, present value calculations are based on the assumption that 5%, 5%, 10%, 40%, and 40% of the implementation costs were or will be incurred in 1986, 1987, 1988, 1989, and 1990, respectively.

To be consistent with accepted guidance on performing value-impact assessments (Heaberlin et al. 1983), the resources spent to quantify the costs and benefits identified in this report were commensurate with their expected magnitude. As a result, many of the identified impacts have been labeled "negligible." This is not intended to imply that the impacts are nonexistent or unimportant to individual licensees; rather, impacts labeled "negligible" are assumed to have no significant impact on the results of this analysis.

#### 1.5 CONTENTS OF THIS REPORT

The Executive Summary summarizes the analysis in a format consistent with the regulatory analysis guidelines of the NRC (USNRC 1984). Section 2.0 compares the current Part 20 with the Part 20 revision to briefly familiarize the reader with the important changes introduced in the revision. Section 3.0 presents the consequences of the revised dose evaluation requirements, including both external and internal dose evaluation. Section 4.0 presents the consequences of the revised dose limits, and Section 5.0 presents the consequences of other changes to Part 20. In Section 6.0, consequences of the revision that are not related to specific provisions are described, including the costs associated with personnel training and procedure revisions, the costs that will be incurred by NRC to implement and enforce the revision, and the impact of the revision on small businesses. Section 7.0 discusses the important benefits from revising Part 20. Although most of these benefits cannot be quantified, they are presented in sufficient detail in this report so that the appropriate authorities can evaluate their significance with respect to the associated costs of the revision. Finally, Section 8.0 summarizes the costs and benefits associated with the revision.

As discussed in Section 1.2, the revision of Part 20 has undergone numerous iterations. In this report, the following definitions apply. The proposed rule is defined as the rule published for public comment on January 9, 1986 (FR 1986). It may also be referred to as "the proposed revision" or "the proposed Part 20." The <u>current rule</u> is defined as the rule present in the <u>Code of Federal Regulations</u> as of January 1, 1988. The current rule may also be referred to as "the present rule", "the current (or present) Part 20", or "the current (or present) regulations." The <u>revised rule</u> is defined as the current version of the planned revision, as of May 19, 1988. The revised rule may also be referred to as "the revised Part 20", "the revision of Part 20", or simply "the revision."

### 2.0 COMPARISON OF PRESENT PART 20 WITH PART 20 REVISION

The revision of Part 20 will involve changes to most sections of the present rule. Also, the revision will result in either the addition or deletion of certain sections. Some of the changes are major and are expected to impact licensees significantly. Other changes, however, are relatively, minor and are expected to have little or no impact. Sections 2.1 through 2.4 below describe changes to the rule that are likely to have significant impacts. Section 2.5, however, describes changes that are expected to have less significant impacts. Changes expected to have little or no impact are not discussed.

### 2.1 SUMMARY OF SIGNIFICANT CHANGES REGARDING INTERNAL DOSE ASSESSMENT

The primary purpose for revising Part 20 is to incorporate the current ICRP system of dose limitation (ICRP 1977) as adopted by the EPA (FR 1987). Basic to this system is the concept that both internal and external doses should be summed to determine relative risk. As a result, the revised Part 20 requires assessment of internal doses and summation of internal and external and external doses to determine the total effective dose equivalents, a measure of total health risk, to individuals. The present Part 20 does not require summation of internal and external doses.

An important consideration for this analysis is that under the revised Part 20, the summation requirement might result in increased whole-body dose equivalents being recorded compared to the external whole-body doses currently being recorded even though the actual dose (risk) to the worker remains the same. Because this increase in recorded dose would be a result of changing terminology rather than an actual increase in risk, no impact regarding health effects is associated with this change.

COLUMNS.

### 2.1.1 Section 20.202: Compliance with Requirements for Summation of Internal and External Exposures

The present Part 20 does not require summation of external and internal doses. In the revised rule, Section 20.202 specifies when and how a licensee must consider intakes of radionuclides in determination of total effective dose equivalent.

### 2.1.2 Section 20.204: Determination of Internal Exposure

This section defines acceptable procedures for determining internal exposures based on air sampling or bioassay measurements. The revised section is more detailed than the analogous section in the present Part 20 because the revised Part 20 requires summation of internal and external doses.

### 2.1.3 Section 20.502: Conditions Requiring Individual Monitoring of External and Internal Occupational Dose

The present Part 20 requires monitoring at 25% of the quarterly limits. However, Section 20.502 of the revised rule requires individual monitoring for workers who are likely to receive doses in excess of 10% of the occupational dose limits. This section mandates, in accordance with , Section 20.202, that suitable measurements be performed to allow summation of internal and external doses. Summation is required when internal doses exceed 10% of the applicable annual limits.

### 2.2 SUMMARY OF SIGNIFICANT CHANGES REGARDING OCCUPATIONAL DOSE LIMITS

This section describes the significant changes in occupational dose limits in the revised Part 20.

### 2.2.1 Section 20.201: Occupational Dose Limits for Adults

The present Part 20 specifies a 1.25 rem/quarter limit to the whole body, head and trunk, active blood-forming organs, lens of eyes, or gonads. If the licensee has a record of a worker's exposure history, the worker may receive up to 3 rem/quarter, provided the worker's lifetime dose does not exceed 5(N-18) rem, where N is the worker's age in years. The revised occupational dose limits for adults specify a total effective dose equivalent per year of 5 rem and a limit of 50 rem (sum of deep and committed dose equivalents) for organs or tissues other than the lens of the eye.

The present Part 20 specifies limits of 1.25 rem/quarter to the lens of the eye, 7.5 rem/quarter to the skin, and 18.75 rem/quarter to each of the extremities. The revised annual limits are 15 rem to the lens of the eye, 50 rem to the skin, and 50 rem to each of the extremities. The definition of "extremities" in the revision is slightly different than the definition in the present Part 20.

The present Part 20 also specifies limits for exposure of individuals to concentrations of radioactive materials in air. The limit can be calculated from air concentrations listed by radionuclide in Appendix B. The revision, on the other hand, does not specify such limits because the total effective dose equivalent limit includes intakes of radionuclides. However, the revision does specify a 10 mg/week limit for intake of soluble uranium because of its chemical toxicity. Limits on intake of other radioa live materials are implied by the 5 rem/yr limit on total effective dose equivalent for radiation workers.

### 2.2.2 Section 20.206: Planned Special Exposures

This section in the revised rule allows licensees to authorize adult workers to receive doses in excess of the occupational dose limits provided that certain conditions are met. Although there is no analogous provision in the present Part 20, planned special exposures would be similar to the 5(N-18) rule in that workers would be allowed to exceed the annual occupational dose limits under certain conditions. If the provision is used, individual workers could receive double the annual occupational dose limits for a maximum of 5 years during the worker's lifetime.

#### 2.2.3 Section 20.208: Dose to an Embryo/Fetus

A limit to the embryo/fetus is not addressed in the present Part 20. Under this section in the revised rule, however, the dose to an embryo/fetus is limited to 0.5 rem. The new rule effectively limits the total effective dose equivalent of a declared pregnant woman to 0.5 rem during the pregnancy. Most licensees currently practice this limit because it is addressed in both a NRC Regulatory Guide (USNRC 1987a) and a staff position paper.

### 2.3 SUMMARY OF SIGNIFICANT CHANGES REGARDING LIMITATION OF DOSE TO THE PUBLIC

Although the present rule only implies limits on dose to the public, the revised Part 20 would contain explicit limitations on such dose. This section describes the new requirements.

### 2.3.1 Section 20.301: Dose Limits for Individual Members of the Public

In the present Part 20, a whole-body limit of 0.5 rem/yr for each member of the public is implied but is not explicitly stated. This revised section, however, limits the total effective dose equivalent to individual members of the public to 0.1 rem/yr. The limit includes doses from all operations by a licensee, excluding disposal of radioactive material into sanitary sewerage. The revision allows licensees to apply for authorization to operate up to an annual effective dose equivalent of 0.5 rem/yr for individual members of the public.

1000000

196.1

Hotel

CONTRACTOR OF

### 2.3.2 <u>Section 20.302: Compliance with Dose Limits for Individual Members</u> of the Public

Although the present Part 20 does not include this provision, this section in the revised rule requires licensees to make measurements, as appropriate, to demonstrate compliance with the limits specified in Section 20.301.

### 2.4 SUMMARY OF SIGNIFICANT CHANGES REGARDING RECORDKEEPING AND REPORTING REQUIREMENTS

Because the revision to Part 20 includes a revised system of dose limitation, the information collected by licensees to assess doses will be more detailed than the information presently collected. As a result, recording and reporting of this information is expected to be more extensive than under the present rule.

### 2.4.1 Section 20.1106: Records of Individual Monitoring Results

This section specifies the information that will be required on NRC Form 5. Although the present Part 20 also requires individual monitoring results to be recorded on NRC Form 5, more information will be required under the revised rule to demonstrate compliance with occupational dose limits.

## 2.4.2 Section 20.1107: Records of Dose to Individual Members of the Public

Although not required in the present Part 20, the revised rule requires that records shall be maintained to demonstrate compliance with dose limits for individual members of the public.

### 2.4.3 Section 20.1206: Reports of Personnel Monitoring

The present Part 20 does not require that individual monitoring reports be submitted for each individual for whom monitoring was required, but does require certain licensees to submit an annual statistical summary report. Under Section 20.1206 of the revised rule, however, the same licensees are required to submit an annual report of the results of individual monitoring for each individual for whom monitoring was required. In addition, the revised 10 CFR Part 19 requires that all licensees submit, at least annually, reports to those individuals describing the doces that they received. Currently, licensees are required to submit reports to individuals only upon request.

### 2.5 SUMMARY OF OTHER CHANGES

### 2.5.1 Changes Regarding Control of Internal Exposure in Restricted Areas

Subpart H of the revision contains requirements pertaining to respiratory protection controls to restrict internal exposure in restricted areas Most of the changes that appear in the revision are either editorial or are insignificant with respect to cost. One exception is Section 20.702, which states that the licensee shall limit intakes so that the total effective dose equivalent is maintained ALARA. The present Part 20 discourages deliberate internal exposures that are a significant fraction of the dose limits and does not require that the total effective dose equivalent be maintained as low as reasonably achievable (ALARA).

### 2.5.2 Changes Regarding Precautionary Procedures

Most of the changes regarding precautionary procedures are expected to have insignificant impacts on licensees with the exception of labeling requirements. In the revised Appendix C, quantities of most radioactive materials that will require labeling are higher than the values provided in Appendix C of the present Part 20.
## 2.5.3 Changes Regarding Waste Disposal

Most of the changes regarding waste disposal requirements are expected to have insignificant impacts on licensees. One exception is that the amounts of specific radionuclides that can be released into sanitary sewerage have been changed. In most cases, the revised limits are 10-100 times more restrictive than the present limits.

# 2.5.4 Changes Regarding Recordkeeping Requirements

Subpart L of the revision contains sections that describe recordkeeping requirements. The requirements include records of radiation protection programs, records of surveys, records of prior occupational dose, records of planned special exposures, records of doses to both individual workers and individual members of the public, records of waste disposal, and records of testing entry control devices. Except for the sections that require records of doses to individual workers and individual members of the public (discussed in Section 2.4), the changes to the present rule are expected to have only minor impacts on licensees. In several cases, the sections in the revision contain more detail than the relevant sections in the present rule. The impacts of these revised requirements are expected to be minor because most licensees currently keep records that fulfill the requirements of the revision. In two other cases (records of radiation protection programs and records of testing entry control devices), the requirements are new but are not expected to have a significant cost impact on licensees.

## 2.5.5 Changes Regarding Reporting Requirements

Subpart M of the revision describes reporting requirements. Except for Section 20.1206, which was discussed in Section 2.4 of this report, the reporting requirements pertain to unusual situations such as thefts and overexposures. In these cashs, the requirements in the revision are similar to the requirements in the present Part 20 except for minor changes that either involve editorial changes or revised definitions of reportable events. Because reporting of incidents is very infrequent, these changes are expected to have minimal cost impacts.

and the second

# 3.0 CONSEQUENCES OF REVISED DOSE EVALUATION REQUIREMENTS

The present and proposed dose limits specified in 10 CFR Part 20 are summarized in Table 3.1. In addition to the changes to the limits, the fraction of the annual limit at which monitoring is required was effectively lowered from 25% to 10% for both internal and external doses. Also, the revised annual total dose equivalent limit is based on a weighted fraction concept designed to normalize the risk of adverse health effects from a dose delivered nonuniformly to the body to the risk from a uniform whole body dose. The total effective dose equivalent includes the sum of both external doses and weighted internal doses when both individually exceed 10% of the annual limit. The anticipated impacts of these changes are evaluated in the following sections.

> TABLE 3.1. Occupational Dose Limits Specified in the Present and Revised Part 20

Present Exposure Category	Present Dose Limit, rem	Revised Exposure Category	Annual Dose Limit, rem
Whole Body (head and trunk, active blood forming organs, or gonads)	3/quarter 5/yr (avg.)	Total effective dose equivalent	5
Lens of eyes	5/yr	Lens of eyes	15
Hands, forearms, feet and ankles	75/yr	Extremities	50
Skin of the whole body	30/yr	Skin	50
Internal Dose	520 MPC-h /quarter		
Committed Dose Equivalents Whole Body Most Organs Thyroid	5/yr 15/yr 30/yr	Committed dose equivalent (organs)	50

# 3.1 EXTERNAL DOSE EVALUATIONS

The present Part 20 limits external whole-body doses to 3 rem/quarter and to a lifetime average that is less than 5 rem/yr after the age of 18. Doses to individual organs are not considered in calculations for compliance with this limit. The revised annual limit of 5 rem includes the summation of external and internal doses if the latter exceed 10% of the annual effective dose equivalent limit. The annual limit for extremities was lowered from 75 to 50 rem and the limit for the skin was raised from 30 to 50 rem.

or the later

One potential cost associated with reduced external dose limits is the cost associated with improvements in personnel dosimeters. The most restrictive exposure limit regarding the sensitivity of personnel dosimeters is the annual whole body limit of 5 rem/yr. Assuming a 1-month exchange period, the minimum sensitivity for the dosimeter must be about 40 mrem (0.1 x 5000 mrem/12 months per year). Under both the current Part 20 and the revised Part 20, NRC licensees are required to obtain personnel dosimetry from a processor holding accreditation from the National Voluntary Laboratory Accreditation Program (NVLAP). Although NVLAP does not have a provision for lower limit of detection, dosimeters used currently can detect a 40 mrem dose. Therefore, no costs are anticipated regarding improvements in dosimetry methods.

Another potential cost is the cost associated with increased monitoring of workers. Because external doses must be monitored for workers likely in receive 10% of the revised dose limits in contrast to 25% of the current limits, it is possible that increased monitoring will be required at some facilities. However, essentially all facilities current'y monitor whole-body doses to workers who receive significantly less than 500 mrem/yr. The primary reason for monitoring relatively low doses is not to meet applicable state and federal regulations, but to both detect unsafe working conditions and protect the facility from litigation by documenting individual doses (Kelsey, Lane and Mettler 1984). Therefore, no increased monitoring of whole-body doses is anticipated from the revised monitoring requirements. However, extremity doses are not currently monitored at 10% of the revised extremity limits in all cases.

At nuclear power plants, extremity dose monitoring is not normally performed on a routine basis because extremity doses received during routine operations are generally well below 25% of the current limits (Huggins and Watson 1984).- However, considering that the revised monitoring requirement is 10% of the revised extremity limits, and that the revised extremity limits are 50 rem/yr as opposed to 18.75 rem/quarter, some workers who are not currently monitored will be monitored under the revised requirements. Based on a study of hand doses received during 4 months of routine operations at a typical nuclear power plant (Huggins and Watson 1984), it is estimated that additional extremity monitoring will be required for an average of 60 personmonths during routine operations per nuclear power plant per year. An additional 120 person-months/yr during outages is also anticipated.

seal.

· Sta

ichia.

Selleri.

increases in extremity monitoring are expected for some other facilities, as well. Procedures such as implantation of radiotherapy needles at hospitals, elution and activity measurements at nuclear pharmacies, and glove box operations at fuel fabrication facilities could result in doses to workers that may require increased extremity monitoring under the revised Part 20 (Thind 1987; Harty, Reece and MacLellan 1986). It is estimated that for 10% of medical facilities, average additional extremity monitoring of 12 person-months will be required. The remaining 90% of medical facilities are anticipated to incur no increased extremity monitoring costs. Additional extremity monitoring of 24 person-months per nuclear pharmacy and 60 person-months per fuel fabrication facility are also estimated. For other facilities, increases in extremity monitoring are expected to be relatively insignificant.

Based on the estimates provided above and the data presented in Section 8.1, increased extremity monitoring equivalent to 33,600 person-months/yr is anticipated in order to comply with the revised requirements. Assuming a monthly exchange period (Huggins and Watson 1984) and two dosimeters per person (one on a finger of each hand), 67,200 additional dosimeters per year will be processed. Based on an estimated cost of \$5 per dosimeter, which includes both the cost of evaluation and the cost of recording the evaluated dose information, the impact on NRC and Agreement State licensees is estimated to be \$340,000 per year. The present value of these costs is included in the summaries presented in Sections 3.6 and 8.2.

## 3.2 INTERNAL DOSE EVALUATIONS

Section 20.103 of the current Part 20 promulgates limits for exposure to concentrations of radioactive materials in air in restricted areas. Internal dose is limited by compliance with maximum permissible concentrations (MPC) of radionuclides in air. These MPC values were established using empirically derived biokinetic models and the calculated dose to critical organs per unit of radioactivity inhaled. The critical organs were defined as those receiving the greatest dose and were specific for each radionuclide.

Compliance with the regulations is currently demonstrated by comparison of the exposure limit with the actual 13-week average air concentration and by appropriate bioassay techniques. Doses do not need to be calculated for time-weighted exposures less than 2 MPC-h/d or 10 MPC-h/wk. The required sensitivity level for monitoring potential internal exposures is, therefore, 25% of the average weekly intake that would occur during exposure to an air concentration of one MPC.

The revised Part 20 requires licensees to monitor intakes of radioactive material for adults likely to receive, in 1 year, an intake in excess of 10% of the applicable Annual Limits on Intake (ALIs). Intakes through wounds or oral ingestion will be significant for only a small percentage of the worker population and will almost always be the result of accidents (unplanned exposures). The only mode of entry for which monitoring will be routinely affected by the changes in the revision is intake by inhalation.

In accordance with the revised Part 20, compliance with the limit on intake by inhalation may be demonstrated in any of three ways: 1) limiting the sum of the quotients of intakes divided by ALIs to unity, 2) limiting the DAC-hours of exposure to 2000, or 3) based on calculations from bioassay data, limiting the committed effective dose equivalents to all organs or tissues to the annual limit. Determination of internal exposure, therefore, requires either bioassay measurements to determine the uptake of radionuclides, or air monitoring to determine air concentrations to which workers may be exposed.

## 3.2.1 Bioassay

Bioassay techniques may be separated into two categories, in vitro and in vivo. In vitro techniques estimate uptakes or intakes through analysis of excreta from potentially exposed individuals and application of biokinetic models co represent the behavior of radionuclides in the body. In vivo techniques estimate uptakes or intakes by direct measurement of photons emitted from radioactive material present in the body. The measurement data are then extrapolated to uptakes using biokinetic models. Because intakes as small as 10% of an ALI must be included in the total effective dose equivalent under the revised Part 20 requirements, the bioassay technique used to assess internal uptakes must have a minimum detectable activity (MDA) at least as small as the uptake assumed to follow an intake of 10% of an ALI.

Using published data (Lessard et al. 1987) for excretion and lung retention fractions, the MDA necessary for detecting an intake of one-tenth of an ALI 30 days following the intake was calculated for a number of commonly-used radionuclides (Table 3.2). Thirty days was assumed to be the minimum measurement interval for a routine program, even though longer intervals are typical for many radionuclides. The calculated MDAs were then compared to acceptable MDAs (AMDAs) listed in draft ANSI Standard N13.30 <u>Performance Criteria for Radiobioassay</u> (see note "b" to Table 3.2). The <u>AMDAs</u> were established as the level expected to be obtainable by a competent counting laboratory and are not necessarily "acceptable" to meet health physics needs in all situations.

Table 3.2 indicates that for wost radionuclides, an intake of 10% of an ALI is detectable 30 days following the intake by either in vitro or in vivo techniques. However, detection at this level may not be possible at current levels of sensitivity for a few radionuclides (e.g., Sr and Pu isotopes). If the nuclide-specific MDAs are within the present capabilities of a bio-assay laboratory, implementation of the p posed regulations will incur no additional costs due to sensitivity requirements. If the required MDA is below the present sensitivity of the bioassay system, costs will be incurred either to improve the system sensitivity, increase the frequency of bioassay measurements, or implement an alternate monitoring system for internal exposure (air monitoring).

#### 3.2.2 Air Monitoring

Airborne radioactive material can be measured by air sampling and analysis. A volume of air is drawn through a detection device that measures the radioactivity directly, or through a collection device such as a filter or impactor, which removes the radioactive material from the air for counting.

Two general types of air samples can be collected: breathing zone and general area. Breathing zone samplers provide a relatively representative sample of the concentration of radioactive material in air breathed by an individual, but suffer from limitations regarding sampling rate. Although

				MDA(a) R Detect A 10 30 Days Po	equired To % ALI Intake st Exposure	Required MDA <amda(b)< th=""></amda(b)<>	
NU	Nuclide Class ALI, #Ci		Urine	Urine In Vivo		In Vivo	
	Зн	Vapor	8E4	2.5E4	_(c)	No	
	14 <sub>C</sub>	D	2E5	1.4E6		No	
	32 <sub>p</sub>	DW	9E2 1E2	4.2E1 1.9E1	:	No No	:
	35 <sub>S</sub>	DW	2E4 2E3	1.2E3 1.8E2	:	No No	:
	45 <sub>Ca</sub>	W	8E2	3.0E4		No	
	<sup>51</sup> Cr	DWY	5E4 2E4 2E4	2.8E6 5.1E5 1.4E5	9.7E4 1.4E5	NO NO NO	No No
	54 <sub>Mn</sub>	D	9E2 8E2	2.0E2 2.0E1	<1E-3 7.6E3	No No	Yes No
	<sup>59</sup> Fe	D	3E2 5E2	2.3E3 1.0E3	- 3.2E3	No No	No
	60 <sub>Co</sub>	WY	2E2 3E1	:	2.0E3 4.3E2	:	No No
	65 <sub>Zn</sub>	Y	3E2	5.3E3	4.0E3	No	No
	86 <sub>Rb</sub>	D	8E2	9.183		No	
	90 <sub>Sr</sub>	D	2E1	9.0E-1	•	No	: :
	99 <sub>Mo</sub>	DY	3E3 1E3	1.1E6 1.8E4	1.5E4	No No	No
	99m <sub>Tc</sub>	D	265	1.199			
	125 <sub>1</sub>	D	6E1	2.6E3	1.1E3	No	No
	129 <sub>1</sub>	D	9E0	5.5E2	2.2E2	No	No .
	131 <sub>1</sub>	D	5E1	2.7E-1	2.4E1	No	No
	137 <sub>Cs</sub>	D	2E2		4.8E1		No

. .

A ALARY

Sale.

10,691-1

in state

1. 200

TABLE 3.2. Annual Limits on Intake of Common Radionuclides Compared to Detection Capabilities for Urinalysis and In Vivo Counting

## TABLE 3.2 (contd)

Nuclide	Class	ALI, µCi	MDA(a) J Detect A 10 30 Days Po Urine	Required To D% ALI Intake DST Exposure In Vivo	Required	MDA <amda(b)< th=""></amda(b)<>
144 <sub>Ce</sub>	W Y	3E1 1E1	-(c)	2.9E2 6 7E1	:	No
235 <sub>U</sub>	D W Y	1E0 8E-1 4E-2	1.0E1 5.0E0 2.5E-1	<1E-3 8.2E0 5.8E-1	NC No No	Yes No No
Nat U(by 234Th)	D ₩ Y	1E0 8E-1 5E-2	1.0E1 5.0E0 2.5E-1	1.5E1 8.2E0 7.2E-1	No No	No No Yes
238 <sub>Pu</sub>	WY	6E-3 2E-2	8.0E-6 1.0E-6	6.1E-2 2.9E-1	Yes Yes	Yes Yes
241 <sub>Am</sub>	W	5E-3	5.0E0	5.1E-2	No	No

(a) Units are nCi for in vivo bioassay and nCi/L for all urinalyses except natural uranium urinalyses (µg/L).

(b) AMDAs specified in draft ANSI Standard N13.30, <u>Performance Criteria</u> for Radiobioassay. Draft ANSI Standard N13.30 is available from the Executive Secretary, Health Physics Society, 8000 Westpark Drive, Suite 400, McLean, VA 22102.

(c) Not applicable.

area air samplers can collect larger air samples, general area air concentrations may differ appreciably from the concentration in the breathing zone of the individual. Uptakes estimated from air samples have an estimated total uncertainty factor of 5 or more for breathing zone samplers and 20 or more for area air samplers (Booth, Bronson and Groth 1985).

Table 3.3 summarizes the revised changes in air concentration limits for a number of common radionuclides. For those radionuclides not detectable with current bioassay technology, increased sensitivity is most easily obtainable through air monitoring. The costs of implementing the revised limits will, therefore, involve procuring and maintaining additional air monitoring equipment, increasing labor costs for collecting and analyzing air samples, and establishing an appropriate perscanel monitoring records system. These costs are detailed in Sections 3.3, 3.4, and 3.5.

3830

TABLE 3.3. Comparison of Part 20 Current and Revised Air Concentration Limits for Certain Radionuclides

ŧ.

The stranged

() 例 ()

Total.

		P1C	MPC(air).	
Nuclide	Class	HC1/mL	MCi/mL(a)	DAC/MPC(air)
3 <sub>H</sub>	Vapor	2E-5	5E-6(S)	4
14 <sub>C</sub>	D	ic-6	4E-6(S)	0.3
32 <sub>p</sub>	D W	4E-7 2E-7	7E-8(S) 8E-8(1)	6 3
35 <sub>S</sub>	DW	7E-6 9E-7	3E-7(S) 3E-7(1)	20 3
45 <sub>Ca</sub>	w	4E-7	1E-7(1)	4
<sup>51</sup> Cr	D W Y	2E-5 1E-5 8E-6	1E-5(S) 1E-5(S) 2E-6(1)	2 1 4
54 <sub>Mn</sub>	D	4E-7 3E-7	4E-7(S) 4E-8(1)	1 8
59 <sub>Fe</sub>	DW	1E-7 2E-7	1E-7(S) 5E-8(I)	1 4
60 <sub>Co</sub>	WY	7E-8 1E-8	3E-7(S) 9E-9(I)	0.2
65 <sub>Zn</sub>	Y	1E-7	6E-8(1)	2
86 <sub>Rb</sub>	D	3E-7	3E-7(S)	1
90 <sub>Sr</sub>	D Y	8E-9 2E-9	1E-9(S) 5E-9(I)	8
99 <sub>MO</sub>	D Y	1E-6 6E-7	7E-7(S) 2E-7(1)	1 3
99m <sub>Tc</sub>	D	6E-5	4E-5(S)	2
125 <sub>1</sub>	D	3E-8	5E-9(S)	6
129 <sub>1</sub>	D	4E-9	2E-9(5)	2
131 <sub>I</sub>	D	2E-8	>E-9(S)	2
137 <sub>CS</sub>	D	6E-8	6E-8(S)	1

TABLE 3.3 (contd)

Nuclide	Class	DAC, #Ci/mL	MPC(air), #Ci/mL(a)	DAC/MPC(air)
<sup>144</sup> Ce	W Y	1E-8 6E-9	1E-8(S) 1E-9(I)	1 6
235 <sub>U</sub>	D W Y	6E-10 3E-10 2E-11	5E-10(S) 5E-10(S) 1E-10(1)	1 0.6 0.2
Natural Uranium	D W Y	5E-10 3E-10 2E-11	1E-10(S) 1E-10(S) 1E-10(I)	5 3 0.2
238 <sub>Pu</sub>	WY	3E-12 7E-12	2E-12(S) 4E-11(I)	2 0.2
241 <sub>Am</sub>	W	2E-12	6E-12(S)	0.3

(a) S = soluble, I = insoluble

## 3.3 POSE EVALUATION COSTS

The principal components of cost-benefit evaluations (value-impact assessments) of proposed regulatory actions are the attributes that are used to characterize the consequences of the proposed action. There are twelve attributes normally used for NRC value impact assessments (Heaberlin et al. 1983). These attributes may be categorized as factors affecting public health, accidental occupational exposure, routine occupational exposure, offsite property, onsite property, regulatory efficiency, improvements in knowledge industry implementation, industry operations, NRC development, NRC implementation, and NRC operations.

Changes in dose evaluation requirements will not affect actual doses received by the public, nor the frequency of accidents which could affect occupational exposure, nor property damage. The effect of the revised dose evaluation rec irements on routine occupational exposures will be discussed in Section 3.4. Changes in dose evaluation requirements may affect regulatory efficiency through changes in reporting requirements. This is discussed in Section 7.3. The changes will not improve knowledge of accident probabilities or consequences. The NRC development costs are sunken costs and are not considered in the regulatory analysis. The NRC implementation and operations costs will not be specific for dose evaluation requirements and are discussed in Section 6.3. Discussions in this section will therefore be limited to the effect of the revised dose evaluation requirements on industry implementation and operations costs. In this section, costs are summarized by facility type according to the facility categories listed in Table 8.1 of this report. For each affected facility type, the industry implementation and operations costs expected to be incurred from the revised dose evaluation requirements are estimated. Costs associated with recordkeeping are discussed separately in Section 3.5. Cost estimates are not provided for five facility categories (well logging, industrial radiography, other measuring systems, research and test reactors, and other facilities) because the associated costs are anticipated to be either zero or insignificant. Costs for these five facility types will be minimal because significant internal doses are rare and it is not anticipated that changes to existing procedures will be required from the revised dose evaluation requirements.

Medical Facilities. Potential internal exposures to workers in a hospital are primarily limited to the nuclear medicine department. By far the most common radionuclide used in nuclear medicine is TC. In one study it was estimated that 80% of the patient procedures involve TC (Wiatrowski et al. 1984).

The radiopharmacesticals of <sup>99</sup> Tc are stable in solution and are nonvolatile. If there are 30 mCi per patient and 8 patients per day per room of 10 X 15 X 8 feet with one air change per hour, 18% of the total dose could be volatilized without exceeding 10% of the DAC value. With no air changes (total recirculation), 0.8% of the total dose could be volatilized without exceeding 10% of the DAC value. The intakes of these radiopharmaceuticals by workers, therefore, are expected to be less than the 10% threshold for monitoring.

Radioioding compounds comprise most of the remaining radionuclides that are potentially inhaled at nuclear modicine departments. For all of these compounds, the DAC values in the revision are at least two times higher than the current MPC values. Therefore, wonitoring will not be required at lower levels of intake than are currently required even though monitoring will be required if intakes are likely to exceed 10% of the applicable ALIs. It is anticipated, however, that some of these licensees will respond to the revised Part 20 by performing additional monitoring for 1 year to demonstrate compliance. This monitoring will consist of a self-monitoring program by researchers who handle iodine (USNRC 1982). Associated cost estimates for one-third of the approximately 4800 nuclear medicine departments (i.e., 1600 departments) are \$400 each for instrument calibration services and \$550 each for equipment purchases, for a total one-time cost of \$1,500,000 (USNRC 1982). The remaining two-thirds of the facilities are not anticipated to incur costs because they currently do not have quantitative internal exposure monitoring programs and are not expected to require one based on the revised regulations.

A similar response is anticipated from the research hospitals possessing a broad license. There are approximately 200 broad-scope research hospitals with 52,000 potentially exposed employees, two-thirds of whom are monitored for internal exposures (USNRC 1982). Assuming a 1-year program of quarterly internal monitoring to demonstrate that intakes are less than 10% of the applicable ALIs, that most researchers handle H-3 or P-32 which can be evaluated using urinalysis, and that 20% of the researchers would be sampled at a cost of \$18 per urinalysis (USNRC 1982), the one-time cost would be 52,000 X 0.67 X 0.2 X \$18 X 4, or \$500,000. One whole-body count is also expected to be needed for 10% of the researchers who received urinalysis in order to evaluate internal deposition of gamma-emitters. The associated cost estimate, assuming \$200 per whole-body count, is \$140,000 (that is, 52,000 X 0.57 X 0.2 X 0.1 X \$200).

The costs to private practitioners of nuclear medicine will be similar to those of nonresearch hospitals. Of the approximately 1000 private practice licensees, one-third were assumed to have quantitative internal exposure monitoring programs (USNRC 1982). If these 330 facilities undergo a selfmonitoring program for 1 year at a cost of \$1000 each, the total cost estimate is \$330,000.

The current level of monitoring in nuclear pharmacies is sufficient to satisfy the provisions of the revised regulation. The only additional efforts would be the conversion of organ burden to dose, which falls under the category of recordkeeping (see Section 3.5).

Manufacturing and Distribution Facilities. It was assumed that the facilities involved in the manufacture and distribution of large sources currently have limited routine monitoring programs for internal exposures. Although exposures are probably less than 10% of the limits, monitoring would have to be performed for a year to demonstrate this to the inspectors (USNRC 1982). It is estimated that 100 workers in the industry would have to be monitored at a cost estimated to be \$200 per whole-body count. With an initial count and four quarterly counts the cost for the demonstration program would be about \$100,000.

Facilities involved in the manufacture and distribution of small sources utilize extensive routine bioassay and air monitoring at numerous locations. The programs are considerably more extensive than those expected to be required under the provisions of the revised Part 20 and should not need to be upgraded in response to the revision.

Academic/Research Institutions. There are currently about 1500 academic and research institutions employing 692,000 workers, 14% of whom are monitored for ionizing radiation (Table 8.3). The radionuclides commonly used at these institutions are H, 14C, 52P, 5Cl, 45Ca, 56Fe, 6CO, 125I, 131I, and 137Cs, among others. The DAC values in the revision for these radionuclides are an average of 2.4 times higher than the current MPC values. Considering that the revised requirement for monitoring intakes is effectively lower by a factor of 2.5 than the current requirement, no significant changes in longterm monitoring practices are expected to be necessary at these facilities.

1.3.2.84

- mil/

Even though long-term changes are not anticipated, it is expected that some of these licensees will incur initial costs required to demonstrate compliance to inspectors (USNRC 1982). Although these costs may not be necessary based on strict interpretation of the revision, they are included in this analysis because they are based on anticipated actual responses by licensees regardless of the necessity of their responses. Based on similar assumptions used to estimate these costs for broad-scope medical facilities, the total initial costs for academic/research institutions are estimated to be \$940,000 for urinalysis (i.e., 692,000 X 0.14 X 0.67 X 0.2 X \$18 X 4) and \$250,000 for whole-body counts (i.e., 692,000 X 0.14 X 0.67 X 0.2 X 0.1 X \$200). Increases in long-term operation costs are estimated to be negligible.

Fuel Fabrication and Processing Facilities. As will be discussed in Section 3.4.1, the effective dose equivalent to workers at fuel fabrication facilities from radioactive material deposited in the body may currently equal or, at least at one facility, exceed the effective dose equivalent from external radiation. The internal exposures at these facilities are thought to be chronic exposures to Class Y materials.

If the proposed DACs are implemented, if chronic exposures continue to exist at fuel fabrication plants, and, if the margin between the DACs and actual plant air concentrations will be equal to the present margin between MPCs and plant conditions, then it may be assumed that the chronic lung burdens of the average worker will decrease by a factor of 5, which is the difference between the revised DAC and current MPC for insoluble (Class Y) uranium. Since the lung burdens at one facility are near the detection limits of currently available lung counting equipment, a reduction of the lung burdens by a factor of 5 would result in lung burdens that are below the detection limits of currently available lung counting equipment (Booth, Bronson and Groth 1985; Palmer et al. 1987; Robinson et al. 1986). Booth, Bronson and Groth (1985) concluded that adequate estimates of intake can be accomplished with currently available air monitoring equipment, much of which should be already in place. They also concluded that procedural changes would be required for changing the method of compliance from one based on lung counting to one based on air monitoring, and estimated these costs to be about \$24,000 per facility (in 1989 dollars). Based on recent information on operating fuel fabrication facilities, these costs are anticipated to be A1incurred by five facilities in this category for a total of \$120,000. though other facilities in this category may incur related costs, they are assumed to be relatively insignificant for this analysis. Annual operation costs of \$43,000 per facility (Booth, Bronson and Groth 1985) for a total of \$215,000 are also anticipated for additional air sampling equipment.

The primary costs in these facilities will involve facility modifications to reduce existing airborne levels to below the applicable DACs for uranium. Booth, Bronson and Groth (1985) stated that fuel fabrication facilities may have areas where airborne activities are routinely 25% to 50% of current MPCs. More recently, it appears that existing fuel fabrication facilities operate with airborne levels that range from 10% to 25% of the MPCs. Considering that the revised DAC value for Class Y uranium is one-fifth of the current MPC value, facility modifications might be necessary to provide adequate safety margins.

Two courses of action are being considered by these facilities: ventilation changes and extensive use of glove boxes. Gne facility has estimated these costs to be \$11.5 million for the ventilation change option and \$42 million for the glove box option; the ventilation change option will also require an annual maintenance cost of \$2 million. Nilson and Malody (1982) did not quantify the costs but did estimate that it would be millions of dollars. Although apparently not yet evaluated, it might be possible to collect air samples to determine the actual particle size distribution at these facilities; the observed particle size distributions may allow operation at current activity levels. There is evidence that such studies would show that some engineering modifications may not be necessary (West, Scatt and Schultz 1979). Licensees may choose not to exercise such an option even if it is offered due to the difficulties in explaining to the public why air activity concentrations above those published in Part 20 do not pose a hazard to the workers. In consideration of these factors and assuming that five facilities will be most affected, the total cost estimate is \$30,000,000 initially (\$6 million per facility) and \$5,000,000 in annual operation costs (\$1 million per facility). These estimates are based on the assumption that particle size studies will render some modifications unnecessary.

It may be asked whether reductions " air concentration levels would be required at fuel fabrication plants 1. - mmissible to base dose equivalent limits for long-lived radionuclides on the annual effective dose equivalent rather than the committed effective dose equivalent. Even with the exemption described in Section 20.205 of the proposed revision, it appears that the facilities would be required to maintain the average air concentration less than the DAC [see 20.205(b)(2)]. Since the external contribution to the committed effective dose equivalent is small, on the average, and even the maximum external dose equivalents do not exceed approximately 1 rem (which would decrease the allowable air concentration by ~20%), it seems that fuel fabrication plants will be required to reduce their uranium air concentrations irrespective of whether the dose equivalent is based on annual effective dose equivalent or committed effective dose equivalent. The major question seems to be how much below the DAC the licensee would be required to maintain air concentrations. If the average air concentration is 1 DAC and a worker were to inhale 1 ALI (the apparent maximum in either case), the committed effective dose equivalent received is 5 rem, which does not allow for external doses to be accrued. If external doses were received by the worker, then air concentrations would need to be reduced below the DAC by corresponding amounts. In the case of fuel fabrication plants, it appears that 99% of the workers would not exceed the revised total effective dose equivalent limit if they inhaled 0.8 ALI. For those few who would, procedures such as increased local ventilation or the use of respirators would serve as sufficient countermeasures.

Commercial Power Reactors. Revised internal exposure provisions are not expected to alter internal monitoring programs already in existence at these plants. These programs greatly exceed the requirements of both the current and revised Part 20. Those utilities that use air monitoring as the surrogate for internal dose will continue to do so, and the calculated doses are so low that it is unlikely that any utility will refine their methods by measuring particle size, solubility fraction, etc. The few facilities that use bioassay data for compliance will probably not alter their procedures. Less than 0.03% of the individuals counted at nuclear power plants between 1978 and 1903 had measured body burdens in excess of 10% of the relevant ALIS (Booth, Bronson and Groth 1985) and it is unlikely that any utility will discontinue its existing monitoring program. The revised regulation does not require breathing zone monitoring for intake monitoring and allows general area sampling. The significant costs associated with the revised dose evaluation requirements are related to recordkeeping (see Section 3.5).

# 3.4 DISTRIBUTION OF OCCUPATIONAL DOSES

The revised dose evaluation requirements will result in internal and external doses being summed to demonstrate compliance with the dose limits. An important distinction between the current and revised rule is that the latter requires control of internal exposures as long as the controls are consistent with maintaining the total effective dose equivalent ALARA. In contrast, the current rule strongly discourages intakes of radioactive material by mandating both quarterly and weekly control measures. As a result, intakes of radioactive material are currently rare at most facilities. For those facilities where internal doses are significant compared to the revised annual dose limit, e.g., fuel fabrication facilities, added control measures should result in decreased internal doses.

#### 3.4.1 Fuel Fabrication Facilities

In 1984, there were 9379 workers at 11 uranium fuel fabrication plants, 5947 of whom received measurable external radiation doses. In 1984, the average measurable dose was about 140 mrem; in 1983 and 1982, the average measurable radiation doses were 160 mrem and 140 mrem, respectively (Brooks 1986). These numbers agree well with the measurable average of 170 mrem for 1980 that was reported by Kumazawa, Nelson and Richardson (1984), and the average annual dose equivalent reported by Booth, Bronson and Groth (1985) at a single facility. Brooks (1986) reported that for fuel fabrication plants, 99% of the external dose equivalents were less than 1.5 rem in 1982 and 1983, and that 99% were less than 0.95 rem in 1984.

Regarding internal doses, the workers at the facility studied by Booth, Bronson and Groth (1985) were found to have measurable lung burdens ranging from 60 to 200  $\mu$ g <sup>235</sup>U. The average of all workers (including those with burdens less than measurable) was found to be 47  $\mu$ g. The average body burden of 47  $\mu$ g of <sup>235</sup>U will result in an annual effective dose equivalent of 460 mrem and an estimated annual lung dose equivalent of 3900 mrem, based on the assumption that the body burdens resulted from chronic exposures to Class Y aerosols. The maximum observed lung burdens (200  $\mu$ g <sup>235</sup>U) result in annual lung dose equivalents of approximately 17 rem/yr and annual effective dose equivalents of about 2 rem/yr. Thus, for at least one facility, the average annual effective dose equivalent from internal radiation exceeded that from external radiation. Because these are values for only one facility, extrapolation to the entire industry is tenuous.

Lessard et al. (1987) have published tables that indicate the lung burden at various times after inhalation of uranium. These tables can be combined with a continuous inhalation model in order to compute the cumulative lung burden following chronic inhalation of uranium. Calculations indicate that during chronic inhalation, the lung burden guickly builds and essentially levels off at about 15 years, the lung burden increases at a rate of only about 1% per year at 15 years.

Kumazawa, Nelson and Richardson (1984) found that the median age of a nuclear fuel cycle worker was 35 years. For purposes of this analysis it was assumed that the average worker began work at 25 years of age, which results in a total work time of about 10 years. The average lung burden of 47  $\mu g'^{236}$ U found by Booth, Bronson and Groth (1985) will be achieved in 10 years if the chronic inhalation rate is about 102  $\mu g(^{236}$ U) per year. This annual intake will result in a committed dose equivalent to the lung of about 5.5 rem and a committed effective dose equivalent of 0.65 rem. This indicates that the average worker at fuel fabrication facilities receives committed effective dose equivalents well below 5 rem/yr.

Considering external doses, it appears that the average worker receives about 0.8 rem combined external and committed effective dose equivalent; the internal component is about four times the external component. The only instance in which fuel fabrication facilities appear not to be already in compliance with the revised requirements is for those individuals who receive the maximal doses, and even then it appears to be a problem only if the individuals who have the largest lung burdens have been exposed for less than 5 years. For a chronic exposure over 5 years, 99% of the workers would have a total of external radiation dose equivalent and committed effective dose equivalent that is <5 rem. For those 1% who currently receive doses that would be unacceptable under the revised Part 20, steps would need to be taken to reduce intakes, e.g., through the use of respirators.

Although the air concentrations in certain areas of fuel fabrication facilities exceed the revised DACs, it appears that few individuals would exceed the revised dose limits. Thus, it appears that one or more of the following may be true: 1) workers do not spend all of their time in the high air concentration areas of the facility, 2) the particle size distribution of the dust precludes lung deposition of much of the material, or 3) some of the material is more soluble than anticipated and is removed from the lung with a biological half-life less than 500 days.

10103488

- charle

Million .

-193

1.250

.

Calculations based on the data presented above indicate that a reduction in airborne levels in all areas of a facility by a factor of 5 would result in a decreased effective dose equivalent of 520 mrem (650 mrem X 0.8) for an average worker. Based on five facilities and assuming that 414 workers receive measurable doses at each of these facilities (see Section 8.1), the annual reduction in effective dose equivalent would be 1080 rem. However, in Section 3.3 as well as in this section it was determined that airborne levels in some areas of these facilities need not be reduced by a factor of five. It is assumed that an average reduction of 2.5 will be observed, for a total effective dose equivalent savings of 540 rem/yr.

#### 3.4.2 Other Facilities

Significant internal doses are rare at facilities other than fuel fabrication facilities. For nuclear power plant workers, average annual internal doses are negligible and maximum annual internal doses are a small fraction of the revised annual total dose equivalent limit (Booth, Bronson and Groth 1985). As a result, it is not expected that summation of internal and external doses will result in added control measures to ensure that workers do not exceed the revised total effective dose equivalent limit or corresponding administrative limits. This assumption is thought to be valid for all facilities (except fuel fabrication facilities) because currently most of the significant intakes of radioactive material result from accidental, not planned, exposures.

Under the current Part 20, most licensees attempt to avoid intakes of radionuclides by workers. The current rule emphasizes the limitation of internal exposures and many licensees feel that compliance is ascured if intakes are avoided altogether. Unfortunately, the avoidance of intakes of radionuclides often leads to increased external doses that exceed the internal doses avoided. This practice may or may not be consistent with the ALARA principle, depending on both the magnitude of the increase in total collective dose equivalent and the costs saved by avoiding the assessment and recording of intakes of radioactive material (Merwin, Brown and Martin 1987).

It is likely that the revised dose evaluation requirements will affect the distribution of doses to some workers who currently receive negligible internal doses. Because the DACs for most radionuclides are higher than the corresponding MPCs, it is anticipated that intakes of radionuclides will generally be more common under the revised Part 20 than under the current Part 20. This increase will result from both decreased controls on airborne activity and more consistent use of the ALARA philosophy regarding tradeoffs between internal and external doses.

In summary, the revised dose evaluation requirements may result in increased internal doses for some workers at facilities other than fuel fabrication facilities. However, these doses would be justified by a concomitant decrease in external doses and/or a cost savings, e.g., a reduction in the use of respirators. In other words, the fact that the revised requirements will likely result in improved implementation of the ALARA principle virtually assures a positive impact. Otherwise, licensees could not justify allowing increased internal doses. This potential impact is discussed further in Section 5.1.

#### 3.5 RECORDKEEPING

The revised dose evaluation requirements, in combination with the revised recordkeeping requirements, will result in substantial costs to licensees who have extensive internal dosimetry programs. These costs are estimated in this section. Recordkeeping costs not associated with the revised dose evaluation requirements are estimated in Section 5.4.

The current Part 20 requires that individual dose records be kept for all personnel for whom monitoring is required. The records are required to be kept on NRC Form 5 (or equivalent) and at least four separate entries per year are required because the current rule promulgates quarterly dose limits. Records of surveys and individual monitoring are also required to be maintained.

The revision of Part 20 also requires that individual dose records be kept for all personnel for whom monitoring was required. However, because individual doses will include internal doses when intakes greater than 10% of applicable ALIs occurred, more information will be required to evaluate doses. Thus, NRC Form 5 will require more information under the revised rule. For facilities with comprehensive internal dosimetry programs, such as nuclear power plants and fuel fabrication facilities, extensive modifications to existing dose evaluation procedures will be required. For facilities where internal doses are normally insignificant, increased effort to complete NRC Form 5 should not be required (USNRC 1982).

An extensive study on the dosimetry and recordkeeping implications of the revision of Part 20 was perfected by the Atomic Industrial Forum (AIF) (Booth, Bronson and Groth 1985). Additional information on recordkeeping costs anticipated from the revised dose evaluation requirements were available from public comments on the proposed revision and a report prepared for the NRC (SEA 1986).

Costs are anticipated to be incurred by some licensees to develop revised recordkeeping procedures and to implement and operate revised recordkeeping programs. Development costs may include development and approval of revised procedures. Implementation costs may include modification of computer programs and data bases and acquisition of additional computer-related equipment. Operation costs may include equipment replacement and maintenance, labor, and data review (Booth, Bronson and Groth 1985). These costs are expected to be incurred at some facilities even though there exists an extremely low frequency of significant internal exposures at the facilities. Some of the costs, therefore, will not be necessary for compliance with the revised Part 20; rather, the costs will be incurred voluntarily as options for demonstrating compliance.

An extensive study of these costs for nuclear power plants and fuel fabrication facilities was performed (Booth, Bronson and Groth 1985). Upon review of this study and other applicable information, the cost estimates from this study were adopted for this analysis. Table 3.4 lists the appropriate cost estimates in 1989 dollars.

Assuming that 109 nuclear power plants will be operating in the beginning of 1989 (ANS 1988), the total development and implementation costs incurred by nuclear power plant facilities is estimated to be \$2,700,000 and \$9,900,000, respectively. The annual operation costs are estimated to be \$2,300,000. It was assumed for this analysis that the number of nuclear power plants operating after January 1, 1989, would remain constant at 109.

TABLE 3.4.	Estimated	Costs of	Increased	Recordkeeping	Requirements
and a second second second	Resulting	From the	Revised D	ose Evaluation	Requirements

Cost Category	Cost per Facility Nuclear Power Plants	(1989 dollars) Fuel Fabrication
Development (initial cost)	25,000	13,900
Implementation (initial cost)	91,000	67,200
Operation (annual cost)	21,000	9,700
Present Value <sup>(a)</sup>	290,000	157,000

 (a) Calculated using methods described in Section 8.2 of this report.

Based on five fuel fabrication facilities that will incur most of the costs of the revision, the total costs for development and implementation are estimated to be \$70,000 and \$336,000, respectively. Total annual operation costs for these facilities are estimated to be \$49,000.

Although significant internal doses are relatively rare at other facilities, they do exist and health physicists at these facilities will be required to evaluate and record the appropriate committed and total dose equivalents. Because these instances will be rare, it is assumed that internal doses will be recorded manually at these facilities when appropriate. It is estimated that at academic and research facilities and at medical facilities, responsible personnel will spend an average of 8 person-hours developing and implementing revised recordkeeping procedures that pertain to dose evaluation. Average operation costs of 4 person-hours/yr are anticipated for manual evaluation and recording of information on NRC Form 5. Based on an hourly rate of \$25 per health physicist in 1989 dollars (SEA 1986), the development costs per facility are expected to be \$200. Annual operation costs are expected to be \$100 per facility. Based on the number of academic/research and medical facilities estimated in Section 8.1, academic/ research facilities are anticipated to incur \$310,000 in development costs and \$160,000 in annual operation costs. Medical facilities are anticipated to incur development costs of \$1,300,000 and annual operation costs of \$650,000. These estimates include both NRC and Agreement State licensees.

Most other licensees have limited or nonexistent internal dosimetry programs. Internal doses at these facilities will rarely exceed levels that must be included in the evaluation of total effective dose equivalent. Thus, no significant recordkeeping costs related to the revised dose evaluation requirements are anticipated for these facilities.

STATES AND

Because licensees will be required to monitor both external doses and intakes of radioactive material likely to exceed 10% of the applicable annual limits under the revised monitoring requirements, licensees that currently do not record doses over 10% of the current limits will have to record external doses under the revised requirements and might have to record internal doses depending on the radionuclide. Because the applicable DACs are higher for most radionuclides than the current MPCs, and considering that most licensees currently record all measured external and internal doses, the impact on recordkeeping with regard to an increased number of workers for whom records will be kept is considered to be negligible for this analysis.

## 3.6 SUMMARY OF CONSEQUENCES

This section summarizes the consequences of the revised dose evaluation requirements described in Sections 3.1-3.5. In Table 3.5, the costs are summarized by facility category in terms of both initial and annual costs. Initial costs include development and implementation costs; annual costs consist of operational costs. Table 3.5 also summarizes the annual benefits (reduced doses) from the revised dose evaluation requirements. The impacts from the revised dose evaluation requirements are anticipated to be negligible for facility categories not listed in Table 3.5.

## TABLE 3.5. Summary of Consequences of the Revised Dose Evaluation Requirements

Facility Category	Initial Cost, \$	Annual Cost, \$	Value of Annual Dose Savings, \$(a)	Value of Costs and Benefits, \$(b)
Medical	3,800,000	730,000	negligible	- 9,700,000
Manufacturing and Distribution	100,000	54,000	negligible	- 560,000
Academic/Research Institutions	1,500,000	160,000	negligible	- 2,700,000
Fuel Fabrication and Processing	31,000,000	5,300,000	540,000	- 59,000,000
Commercial Power Reactors	13,000,000	2,500,000	negligible	- 33,000,000
Total (c)	49,000,000	8,700,000	540,000	-115,000,000

(a) Based on a value of \$1000/person-rem.

(b) Calculated using methods described in Sections 8.2 and 8.3 of this report. The present values are based on 1989 dollars, a discount rate of 10% (applied to all costs and benefits), and a 30-year period. A negative sign indicates a negative impact.

(c) Throughout this report there may be minor variations in summed values because of rounding. It is probable that some of the costs that will be incurred by licensees in response to the promulgation of 10 CFR Part 20 will not be necessary for compliance with the revision. For example, because few workers at nuclear power plants currently receive doses greater than 10% of the applicable DACs, summation of external and internal doses will be required in only a small number of cases. It follows that some of the estimated costs of increased recordkeeping requirements resulting from the revised dose evaluation requirements (see Table 3.4) could be avoided if optimal compliance with the revision is achieved. An estimate of the costs of the revision based on optimal compliance is included in the sensitivity analysis provided in Section 8.4.

## 4.0 CONSEQUENCES OF REVISED LIMITS

The revision of Part 20 contains changes to most of the current dose limits. These include occupational whole-body dose limits and separate limits for the eyes, extremities, skin, and individual organs. In addition, the revised Part 20 contains dose limits for the embryo/fetus and for individual members of the public; neither of these limits exists in the current Part 20. The consequences of the revised limits are presented below.

## 4.1 OCCUPATIONAL DOSE LIMITS FOR ADULTS

The present occupational dose limits for adults are 1.25 rem/quarter to the whole-body, head and trunk, active blood-forming organs, lens of eyes, or gonads; 18.75 rem/quarter for hands and forearms or feet and ankles; and 7.5 rem/quarter for skin of the whole body. A licensee may, however, remmit a worker to receive whole-body doses up to 3 rem/quarter, provided that the worker's average dose to the whole body after the age of eighteen does not exceed 5 rem/yr. Inhalations of radioactive material per quarter are limited to the amount that would be inhaled by a worker present for 520 hours at the air concentrations listed in Appendix B, Table 1.

The revised occupational dose limits are a total effective dose equivalent of 5 rem/yr, a dose to any organ or tissue other than the lens of the eye of 50 rem/yr, an eye dose equivalent of 15 rem/yr, and a shallow dose equivalent of 50 rem/yr to each of the extremities and to the skin. Planned special exposures exceeding the annual limits are allowed, providing there is justification (see Section 4.2). A comparison of the current and revised limits was presented in Table 3.1.

#### 4.1.1 Whole-Body Dose Equivalent Limits

As discussed in Section 3.0, there are significant differences in the dosimetric principles that form the basis for the two sets of limits. Under the present Part 20, whole-body dose limits are based only on external doses. Internal doses are subject to separate controls on intake. Under the revised limits, however, external and internal doses are summed to determine the total effective dose equivalent. Weighting factors are applied to internal doses depending on the critical organ(s) for the radionuclides of concern to equate the risk to that from a whole-body exposure.

In this section, the costs and benefits associated with revising the whole-body dose limits are presented. Although it is difficult to compare the two dose limits because they are based on different methodologies for evaluating doses, very few workers currently receive both internal and external doses that approach the separate limits under the current Part 20. (It is assumed in this section that workers who were reported to receive whole-body doses approaching or exceeding 5 rem did not receive significant internal doses. The few cases where this assumption does not apply were discussed in Section 3.4).

### Number of Workers Who Receive High Doses

Certain licensees are currently required under Part 20.407 to submit an annual statistical summary report of whole-body doses received by individuals for whom monitoring was required. The reports list the number of individuals who received doses in any of eighteen dose ranges. The data in these reports are periodically compiled and analyzed by the NRC. To assess the impact of the revision, summary information on the relatively high doses received by individuals at NRC-licensed facilities was reviewed. Table 4.1 lists selected data for the year 1984 for the seven categories of licensees required to submit annual summary reports (Brooks 1986).

	No. of	No. of Employees	Average Dose Equivalent per Employee	No. of Employees Receiving Doses 4 rem or Greater	
Facility Type	Licensees	Monitored	Monitored, rem	<u>4-5 rem</u>	>5 rem
Industrial Radiography	361	8,458	0.30	24	13
Manufacturing and Distribution	38	5,009	0.13	1	0
High-Level Waste Repository	0	0		0	0
Low-Level Waste Disposal	2	925	0.08	0	0
Independent Fuel Storage	1	32	0.41	0	0
Fuel Fabrication and Processing	14	9,488	0.09	0	0
Commercial Power Reactors	88	170,928	0.32	380	<u>11</u>
Total	504	194,840	0.30	405	24

215

TABLE 4.1. Doses Received by Licensee Employees in 1984

Similar data for licensee categories other than the seven presented in Table 4.1 were not available for the year 1984 because licensees in other categories were not required to submit annual reports of occupational exposures. However, data for other categories of licensees were compiled for exposures occurring in 1979 (Brooks, McDonald and Richardson 1982). Selected data from that report are presented in Table 4.2.

Facility Category	No. of Employees Monitored	No. of El Receiving <u>4 rem or</u> <u>4-5 rem</u>	mployees Doses of <u>Greater</u> >5 rem
Academic	24,639	1	1
Medical	64,057	21	29
Marketing	11,037	31	21
Industrial Radiography	11,969	34	24
Research and Development	18,663	4	1
Other Byproduct Material (*)	27,335	9	22
Uranium Milling/Production	3,508	0	0
Fuel Fabrication and Processing	9,946	0	0
Other Special Nuclear Material	7,562	0	0
Research and Test Reactors .	3,003	0	0
Power Reactors	106,445	477	130
Total	288,164	577	228

## TABLE 4.2. Number of Licensee Employees Reported to Have Received Doses Approaching or Exceeding 5 rem in 1979

(a) Includes well logging.

Although in some cases the facility categories are not consistent between Tables 4.1 to 4.2, analysis of the data from which the tables were derived provides an indication of trends in dose distributions. Of the 577 individuals reported to have received occupational doses between 4 and 5 rem in 1979, only 35 (6%) were not employed by licensees categorized as facilities belonging to one of the seven types listed in Table 4.1. Of these 35 individuals, 30 were employed at either a medical facility or a well logging facility. The remaining 5 individuals were employed at other types of facilities. Similarly, of the 228 individuals reported to have received occupational doses greater than 5 rem in 1979, 53 (23%) were not employed by licensees categorized as facilities belonging to one of the types listed in Table 4.1. Of these 53 individuals, 46 were employed at either a medical facility or a well logging facility. The remaining 7 individuals were employed at other types of facilities.

The number of workers reported by licenses to have received doses in the various dose ranges does not reflect the actual distribution of doses received because of multiple reporting of transient (\*emporary) workers who worked at more than one facility during a year. Transient workers are often employed at nuclear power facilities for relatively brief periods, principally during plant outages or during special maintenance activities (Lawrence et al. 1984). Because a facility is required to report only those doses received at that facility, a summation of these reports results in transient workers being counted two or more times in relatively low dose ranges; such a practice does not reflect the doses actually received by these individuals during the year. However, their actual annual doses can be determined by compiling termination reports. When a transient worker terminates employment at a facility, a termination report indicating the dose received by that worker during employment at that facility is reported to the NRC as required under the present Part 20. Although termination reports are required to be submitted by all seven categories of licensees listed in Table 4.1, more than 95% of termination reports are filed by commercial nuclear power facilities (Brooks 1986).

Analysis of the termination reports submitted by nuclear power facilities indicates that from 1977 through 1984, multiple reporting of transient workers accounted for an underestimate of the number of workers who were reported to have received annual doses greater than 5 rem. The actual total was from 50 to 80 workers per year higher (Brooks 1986). Although this number applies only to transient workers employed at nuclear power plants, it was assumed that relatively few transient workers at other facilities received doses >5 rem during this period because more than 95% of transient worker reports were filed by nuclear power plants. In addition, the nature of the work performed during outages at nuclear power plants suggests that doses >5 rem/yr are most likely to be incurred by transient workers employed at nuclear power plants rather than transient workers employed elsewhere.

Using these adjusted dose estimates for transient workers at commercial power reactors, the actual number of workers who received annual doses -5 rem was derived for the years 1979 and 1984. The results are presented in Table 4.3. For 1984, the data for well logging, medical, and other facilities were extrapolated from the 1979 data assuming a reduction in numbers consistent with the reduction observed for industrial radiography facilities. Extrapolation was necessary because 1979 was the last year that data were compiled for facilities other than those required to submit annual summary reports.

The data in Table 4.3 suggest that the number of licensee workers who received annual doses >5 rem dropped significantly from 1979 to 1984. In 1979, 180 nuclear power plant workers received doses >5 rem, 130 (72%) of whom were reported by facilities on the annual statistical summary report form. The remaining 28% were transient workers reported by two or more facilities during the year. In 1984, 110 workers received >5 rem, none of whom were reported on the annual statistical summary report forms, i.e., all of whom were transient workers.

Exposure data for the years 1985 and 1986 suggest that doses received by workers in the nuclear power plant industry dropped dramatically during those two years. In 1985, collective doses received by nuclear power plant workers were 20% lower than the collective doses received in 1984 (Ryan 1986). Compared to the reduction in collective doses observed in previous years, the 20% reduction in collective doses in 1985 was unprecedented. Since 1973, the greatest single-year reduction in collective dose equivalent received by

Facility Type	No. of Employees Who Received >5 rem		
and the second se	1979	1984	
Industrial Radiography	24	13	
Medical	29	16	
Well Logging	17	9	
Commercial Power Reactors	180	110(a)	
All Others	7	5	
Total	257	153	

TABLE 4.3. Adjusted Number of Employees Who Received Doses >5 rem in 1979 and 1984

 (a) Estimate based on discussions with firms providing contract personnel (Brooks 1986).

nuclear power plant employees had been 4%. As in 1984, no worker at any one nuclear power facility received >5 rem in 1985.

In 1986, the total collective dose equivalent was lower than the 1985 level. Also in 1986, the average dose per worker reached the lowest level ever reported for the U.S. nuclear power industry (USNRC 1987b). For the third consecutive year, no worker received a dose >5 rem while at any one facility.

The recent dramatic decreases in collective dose equivalents, average individual dose equivalents, and number of workers who received >5 rem per year is not likely to be a short-lived phenomenon. Some of this change is attributable to an increased emphasis by the Institute of Nuclear Power Operations (INPO) on keeping doses low (Ryan 1985). INPO, an organization established in 1979 to promote the highest levels of safety and excellence in the nuclear industry, has recognized low personnel radiation exposures as one indicator of high plant performance (Pate 1986). Combined with NRC efforts to promote the ALARA concept, efforts by INPO to reduce individual and collectiv adiation doses throughout the nuclear industry have been successful.

Several other factors may have contributed to the recent reduction in doses to workers at both nuclear power plants and other NRC-licensed facilities. First, for most of this decade, NRC has been moving toward a revision of Part 20 that includes a 5-rem annual limit. Second, the concept of ALARA has been emphasized by essentially all influential organizations, including the NRC. Many facilities have responded by both reducing the collective doses incurred at the facilities and by reducing the number of workers to zero who are receiving an annual dose >5 rem. These two factors have contributed to the practice by all nuclear power plants to establish administrative limits that are <5 rem/yr. Finally, post-Three Mile Island modifications that involved high doses have now been completed (Ryan 1985).

## Direct Implications of Revised Occupational Limits

It is evident from the data presented in the preceding section that the number of licensee workers who receive annual doses >5 rem is small and, is continually decreasing. Based on the trends indicated by recent exposure data, it is estimated that the revised whole-body dose limit will reduce from 50 to zero the number of workers at Auclear power plants who will receive planned annual doses >5 rem. This estimate does not consider the use of the planned special exposure provision, which is discussed in Section 4.2; nor does it consider gradual implementation of the revision.

Because the workers affected by the revised limits will primarily be temporary workers, licensees will selectively hire temporary workers whose total annual doses from all employers are not approaching the dose limits. This is not expected to result in a change in the total collective dose equivalents received at the facilities, nor is it expected that significant costs will be incurred by licensees, because only 50 workers per year would be expected to receive annual doses >5 rem if the revised Part 20 were not promulgated. This is a very small percentage of the contractor work force, and licensees are not anticipated to experience a shortage of qualified workers in the foreseeable future (NESP 1980). The only significant cost associated with nuclear power plant facilities is that some workers will not be eligible for some jobs within the nuclear industry for portions of a year because their total annual doses from all employers will be near administrative or NRC dose limits. No attempt was made in this analysis to quantify these costs.

No impact from the revised limit is anticipated with regard to reducing annual doses below 5 rem to workers at facilities other than nuclear power plants. Table 4.3 indicated that in 1984, 43 non-power-plant workers were exposed to doses >5 rem. Of these, 16 were employed at medical facilities (hospitals) and it is likely that similar exposures in 1989 would not be planned, with or without promulgation of the revised Part 20. Twenty-two of the 43 workers were employed by either well logging or industrial radiography facilities. It is estimated that by 1989, only a few such workers per year would receive annual doses >5 rem if the revised Part 20 were not promulgated. This estimate considers the continuing overall reduction in relatively high doses throughout the nuclear industry as a result of emphasis on ALARA practices. For similar reasons, the five individual doses >5 rem in 1984 that were reported by other facilities would not be expected to occur in 1989.

# Indirect Implications of Revised Occupational Limits

An important effect of the revised occupational whole-body dose limit will be to reduce the administrative limits at some facilities. An administrative limit is a self-imposed limit set by a facility to provide a safety margin to ensure that the NRC limits are not exceeded. As stated previously in this section, nuclear power plants currently do not intentionally allow workers to receive whole-body doses greater than 5 rem/yr. To assure this, the administrative limits at the plants are set lower than 5 rem/yr. Annual administrative limits are common at many plants, as are quarterly and/or weekly administrative limits (either in addition to or instead of annual administrative limits) (Pelletier and Voilleque 1979). It is expected that a minor reduction in administrative limits will occur at some nuclear power plants in response to the revision of Part 20, resulting in hiring about 20 permanent workers for radiation areas who will lessen the burden on similar workers approaching the plant administrative limits. Although this will likely result in a decrease in efficiency for performance of some work, the anticipated slight increase in collective dose equivalent is considered negligible for this analysis.

It is likely that other facilities will reduce their administrative dose limits in response to the revised whole-body dose limits in Part 20. However, Table 4.1 indicated that few employees other than those at nuclear power plants receive doses >4 rem. Therefore, the impact of the revised whole-body dose limit with regard to lowered administrative limits is estimated to be negligible for facilities other than nuclear power plants.

Before the estimated impact of the revised whole-body dose limits can be quantified, two important factors must be considered. First, as stated previously, the number of workers who have received doses approaching or exceeding 5 rem/yr has decreased dramatically in recent years. It is likely that this trend will continue regardless of the limits specified in Part 20. However, because of the difficulty in estimating the rate of decrease, it is estimated that the work force in the nuclear industry will be permanently increased by 20 workers to ensure that administrative dose limits are not exceeded. This consideration partially accounts for the probability that some of the decrease in individual doses already observed may be directly attributable to licensee pre-planning in anticipation of the revised limits.

1000

The second important consideration in assessing the impact of the revised limits is that the implementation period for the revision is 5 years from the date the proposed revision was published (January 9, 1986). If licensees make use of this implementation period, costs could be deferred, resulting in a reduced present value of the costs of the revision. The method used in this report for calculating present values is described in Section 8.2.

#### Statement of Impact

The estimated impact of the revised whole-body occupational dose limit on NRC licensees is expected to be limited to the costs associated with hiring additional permanent workers at nuclear power plants. These workers will be hired in order to reduce annual doses to members of the existing workforce who currently receive doses approaching or exceeding 5 rem/yr. Based on data presented in this report and an annual cost of \$60,000/worker (SEA 1986), which includes all costs (labor and overhead) associated with the employment of the worker, the estimated annual impact of the revised whole-body dose limits is \$1,200,000 in 1989 dollars. A one-time marginal cost of \$280,000 is also estimated for hiring 20 workers at a cost of \$14,000/worker in 1989 dollars (Vallario et al. 1985). No other attributes are expected to be significantly affected by the revised whole-body dose limit.

# 4.1.2 Other Limits and Their Consequences

In addition to the revised whole-body dose limit, the revision of Part 20 includes revised dose limits for the eyes, extremities, skin, and, individual organs. The revised limits for the eyes and skin allow higher annual doses than are allowed by the present limits. The revised limit for the extremities is lower than the present limit; however, the revised rule clearly states that the limit is applicable to individual extremities. In contrast, the current rule does not specify that the limit is applicable to individual extremities. Finally, the dose limit to individual organs is essentially unchanged. However, the use of DACs and ALIs to limit the doses to individual organs will indirectly change the limit on intake of certain radionuclides. This consequence of the revised dose evaluation requirements was discussed in Section 3.4.

Dose limits for the eyes, skin and extremities are intended to prevent nonstochastic effects to those parts of the body (ICRP 1977, NCRP 1987). When the dose limit to an individual organ is not determined by the 5-rem annual limit for the whole-body, the limit to that organ (50 rem/yr) is also intended to prevent nonstochastic effects.

Because nonstochastic effects have a dose threshold, workers should not suffer from these effects, providing that their doses are kept below the limits. Under the current dose limits, no nonstochastic effects have been observed except when doses much greater than the limits were received from accidental overexposures. Assuming that the revised dose limits will not affect the probability of an accidental overexposure, no impact regarding occupational health is anticipated from the revised limits.

The revised limits could result in slight impacts on operations costs. For example, an increased annual eye dose limit could result in a decreased use of protective goggles (McGuire, Baker and Vandergrift 1983). Also, a decreased extremity limit could result in an increased use of shielding and gloves. Depending on the current interpretation of the dose limit to the skin, however, some individual licensees currently choose to adopt the limit to the skin for limiting dose to the extremities. Although the most recent guidance by NRC states that the dose limit to the skin of the whole body does not apply to the skin of the hand and forearm (USNRC 1983), licensees do not use consistent methods to determine doses to the skin. Therefore, an increased dose limit for skin cculd result in increased extremity doses.

Although there will likely be slight impacts on operations costs from the revised limits, the net impact is estimated to be negligible. Because the revised limits to the eyes and skin are less restrictive than the current limits, the costs saved are expected to approximately offset the costs incurred from the more restrictive extremity limits. In addition, previous case studies have indicated that, except for a few workers, eye, skin, and extremity doses have been maintained well below both the current and revised limits (USNRC 1982). Increased costs for extremity monitoring, which will result from both reduced limits and reduced doses that require monitoring, were discussed in Section 3.1.

## 4.2 PLANNED SPECIAL EXPOSURES

Under the planned special exposure provision in the revised Part 20, a licensee may authorize an adult worker under certain conditions to receive doses in excess of the prescribed annual limits. This provision may only be used in an exceptional situation, and the dose received by an individual from all planned special exposures in a year must not exceed one times the annual limit. In order to authorize a planned special exposure, a licensee must first ascertain the doses received from all previous planned special exposures in excess of the annual limits for each individual involved. The total lifetime dose from all planned special exposures and all doses in excess of the annual limits must not exceed 5 times the annual limits.

Although the present Part 20 does not include a planned special exposure provision, the present "5(N-18)" rule is somewhat similar because, under the rule, annual doses >5 rem/yr are permitted. An important difference, however, is that planned special exposures are limited to "exceptional" situations, whereas the "5(N-18)" rule was essentially unconditional. As a result, it is expected that fewer individuals will receive annual doses >5 rem under the revised Part 20 than under the present Part 20.

# 4.2.1 Use of the Planned Special Exposure Provision

Use of the planned special exposure provision for whole-body doses will be limited primarily to commercial nuclear power facilities. This is because, as demonstrated in Section 4.1.1, temporary workers at nuclear power plants are the only workers at NRC-licensed facilities who currently receive planned annual doses >5 rem. It is assumed that this would continue in the future even without implementation of the revised Part 20. As a result, facilities other than nuclear power plants would have no reason or justification for using the planned special exposure provision for whole-body doses. The few exceptions that might occur were assumed for this analysis to be negligible.

It was estimated that reduced administrative limits based on the revised Part 20 would result in the hiring of 20 additional permanent workers at all nuclear power plants combined (Section 4.1.1). In addition, it was estimated that annual individual doses >5 rem would be reduced from 50 to zero once the revised Part 20 is implemented. This reduction would be realized primarily through selective hiring of temporary workers. These estimates were made without consideration of the planned special exposure provision.

Instead of reducing individual doses through increased or selective hiring, it is possible that licensees will use the planned special exposure provision to authorize annual doses >5 rem in order to reduce expenses. However, this possibility is remote because of the conditions of the provision. The condition likely to limit the use of the planned special exposure provision is 20.206(a), which states that the provision can only be used when other alternatives are either "unavailable or impractical." Because the term "cost-effective" is not included in this condition, licensees will be likely to use the planned special exposure provision only when no other reasonable alternative exists.

It is estimated that an average of one nuclear power plant facility per year will use the planned special exposure provision. One reason for using the provision could be to avoid a safety hazard that may occur if repairs to a vital piece of equipment are delayed. Another reason could be to avoid substantial costs that would be incurred if vital work is delayed. In either case, the benefits would outweigh the costs. Otherwise, the licensee could not justify using the planned special exposure provision.

### 4.2.2 Effect on Distribution of Doses

Use of the planned special exposure provision should result in a decrease in collective dose equivalent because alternatives to use of the provision, such as performance of a job using more workers than are needed, would be less efficient. Therefore, use of the provision should partially offset the slight increase in collective dose equivalent anticipated from the revised whole-body dose equivalent limit. However, because the planned special exposure provision is anticipated to be used about once per year, the effect on doses to workers is assumed to be negligible.

### 4.2.3 Cost Implications

When the planned special exposure provision is used, a cost savings associated with its use is virtually assured because of the conditions that must be met before the rule can be implemented. Licensees will be unlikely to use the provision unless alternatives to its use are significantly less cost-effective. The associated cost savings could be significant if, for example, the down time at a nuclear power plant is reduced (the cost of replacement power averages \$400,000 - \$500,000 per day during a nuclear power plant outage). Because the planned special exposure provision is anticipated to be used infrequently, no attempt was made in this analysis to quantify the associated net cost savings. It is assumed, however, that these savings will partially offset the cost increases associated with the revised whole-body dose limit.

- inthe

LANSTE

计传输

## 4.3 DOSE TO AN EMBRYO/FETUS

The current Part 20 does not provide a limit on dose to the embryo/ fetus of a pregnant worker. Rather, pregnant workers are subject to the same occupational dose limits as all other adult workers. A 0.5-rem limit is practiced at many facilities because of a NRC Regulatory Guide (USNRC 1987a) and a staff position paper that address this subject. The revised Part 20 explicitly states that the dose to an embryo/fetus due to occupational exposure of a declared pregnant worker shall be limited to 0.5 rem during the entire pregnancy. The rule also states that efforts shall be made to avoid substantial variation above a uniform monthly rate that would satisfy the 0.5-rem limit. By comparison, the NCRP recommends a limit of 0.5 rem to the fetus during the entire pregnancy, and also recommends a limit of 0.05 rem/month once a pregnancy becomes known (NCRP 1987). Essentially, the revised Part 20 rule is consistent with the NCRP recommendations; however, the rule is worded less strongly.

## 4.3.1 Effect on Distribution of Doses

Because licensees are currently not required to submit detailed annual reports of doses received by individuals, it is not possible to precisely determine the number of female workers at licensee facilities who currently receive doses >0.5 rem/yr. However, based on a comprehensive review of occupational exposure to ionizing radiation in the United States in the year 1980 (Kumazawa, Nelson and Richardson 1984), it is possible to estimate the effect of the embryo/fetus limit on occupational exposures. In 1980, an estimated 80% of female workers monitored for exposure to ionizing radiation were between the ages of 18 and 40. The median age of these women was 27, 28, and 32 for workers in medicine, the nuclear fuel cycle, and industry, respectively. Approximately 10,000 women were reported to have received doses >0.5 rem in 1980. A distribution of these women by dose range and industry is presented in Table 4.4, where the data indicate that approximately 90% of the women who received doses >0.5 rem were employed in medicine.

There is evidence, however, that the number of pregnant women who receive annual doses >0.5 rem is small and is decreasing (USNRC 1982). Several factors could account for this trend. First, the NRC published a revised regulatory guide on prenatal radiation exposure in 1975 (USNRC 1975) which specifies that women assigned to work in a restricted area should be given specific instruction regarding prenatal exposure risks to the developing embryo and fetus. Women were instructed that they could request rearsignment to nonradiation work if they were pregnant or expected to be soon. Currently, most licensees either comply with or go beyond the recommendations in this regulatory guide. Two other factors that may contribute to this trend are the emphasis by ICRP (ICRP 1977) and NCRP (NCRP 1987) on limitation of dose to the unborn, and the trend toward reduced individual doses throughout the nuclear industry (see Section 4.1.1).

**TRAK** 

C. Martin

Similar.

Although few pregnant women currently receive doses >0.5 rem, the embryo/fetus dose limit in the revised Part 20 and the recent revision of Regulatory Guide 8.13 (USNRC 1987a) will result in a further reduction in the number of pregnant women who receive doses >0.5 rem. For the reasons discussed in the paragraph above, it is assumed in this report that essentially all pregnant women who currently receive doses >0.5 rem do so voluntarily.

It was estimated that about 7% of female radiation workers in the U.S. become pregnant in a given year (NCRP 1977). Based on the data in Table 4.4 and the recent trend toward decreased individual doses, and assuming that declared pregnant women will not receive doses >0.5 rem under the new limit, it is estimated that the doses to 200 pregnant women per year will be reduced below 0.5 rem due to the embryo/fetus limit. It is also estimated that an additional 1000 pregnant women per year will receive reduced doses even

		Number of Women Employed					
0	ose Range, rem	Medicine	Nuclear Fuel Cycle	Other Industries	Total		
	0.5-1.0	5,902	82	560	6,544		
	1.0-2.0	2,191	111	236	2,538		
	2.0-3.0	587	33	68	688		
	3.0-4.0	188	19	0	207		
	4.0-5.0	_ 100	0	_0	100		
	Total	8,968	245	864	10,077		

TABLE 4.4. Number of Women Reported to Have Received Doses >0.5 rem in 1980

though they would not have received doses >0.5 rem had the limit not been promulgated. This estimate is based on the assumption that licensees will be more cautious regarding the exposure of pregnant women to ionizing radiation once the limit is promulgated. The total dose reduction for the 1200 pregnant women is estimated to be 300 rem. Because these doses will likely be distributed to nonpregnant replacement workers in most cases, the net savings in collective dose is anticipated to be insignificant for workers at licensee facilities. However, it is estimated that 300 rem/yr to fetuses will be saved by the dose limit. It is assumed that 90% of this dose reduction will occur at medical facilities and 10% will occur at nuclear power plants.

## 4.3.2 Cost Implications

Because most licensees voluntarily keep doses to pregnant women below the limit specified in the Part 20 revision, they have experience in ensuring that declared pregnant workers do not receive doses >0.5 rem. Most licensees simply remove pregnant workers from work involving high radiation doses (USNRC 1982). The costs, therefore, are expected to be administrative costs required to either hire a replacement worker or reassign a present worker to the job vacated by the pregnant woman. The average estimated cost per pregnant worker who is removed from her work is \$500. Based on the estimates in Section 4.3.1, the total cost to licensees is expected to be \$600,000 per year, 90% and 10% of which will be incurred by medical facilities and nuclear power plants, respectively.

#### 4.3.3 Other Considerations

Because it will be difficult for licensees to determine in advance which women will become pregnant during their employment, it is possible that some licensees will selectively hire either men or older women for jobs that involve relatively high doses. This will limit the career opportunities for certain women. This phenomenon has already been observed at some facilities, especially nuclear power plants (USNRC 1982), and it is likely that promulgation of the limit will further affect the employment opportunities to some degree for a small number of women. This topic is discussed further in Section 7.6.

## 4.4 DOSE LIMIT FOR INDIVIDUAL MEMBERS OF THE PUBLIC

The current Part 20 does not explicitly state a dose limit for individual members of the public, but does state limits of 2 mrem in 1 hour and 100 mrem in 7 days, based on continuous presence of an individual in an unrestricted area. An annual limit of 500 mrem is implied, however, by wording in the current rule regarding license applications. A limit of 500 mrem/yr is also implied by the limit on releases of radioactive material to unrestricted areas. The present Part 20 also refers to 40 CFR Part ...) (EPA 1986a), which specifies much lower dose limits to the public from nuclear fuel cycle operations. Even though a 500 mrem/yr limit is not explicitly stated in the present Part 20, the NRC would not allow licensees to operate such that this annual dose is exceeded.

The Part 20 revision explicitly states an annual dose limit of 0.1 rem to individual members of the public from continuing operations by a licensee. A licensee may apply for authorization to operate temporarily up to an annual limit of 0.5 rem. In addition to the annual limit, the dose in any unrestricted area is limited to 2 mrem/h (there is no 7-day limit). An additional rule requires demonstration of compliance with the dose limits by either measurement or calculation using approved methods.

#### 4.4.1 Effect on Distribution of Doses

All licensees in the nuclear fuel cycle are constrained by 40 CFR Part 190 (EPA 1986a) to operate such that whole-body doses to individual members of the public are less than 25 mrem/yr; 40 CFR Part 61 (EPA 1986b) contains a similar limit for air emissions for most other NRC licensees. Nuclear power plants are subject to further release limits in 10 CFR Part 50, Appendix I. Because of the nature of operations of licensees not affected by 40 CFR Part 190 or 40 CFR Part 61, it is extremely unlikely that any licensee (NRC or Agreement State) currently exceeds the revised 100-mrem annual effective dose equivalent limit, except for some medical facilities (see below). For those licensees in the future who can justify exceeding this limit, the revised rule allows for application to operate up to 500 mrem/yr, which is the limit implied in the present Part 20.

It is possible that medical facilities where brachytherapy or radioimmunotherapy procedures are performed will be affected by the revised limits. Many hospitals performing brachytherapy have radiation levels outside a patient's room exceeding 2 mrem/h (Thomadsen et al. 1983). In some cases, these facilities must provide shielding or control patient admissions to ensure that dose rates to neighboring patients do not exceed the limits in Part 20. In many cases, the 100-mrem/7-day limit is the more restrictive limit because the wording of the rule requires limitation based on continuous presence of an individual in an unrestricted area (Thomadsen et al. 1983; Gitterman and Webster 1984). The revised limits could be both costly and beneficial to these facilities. The 100-mrem annual limit could be restrictive for patients who would otherwise remain at the facility for an extended period near a brachytherapy room. Conversely, the absence of the 100-mrem/7-day limit could allow facilities to control dose rates based on the less-restrictive 2-mrem/h limit. Based on these opposing factors and detailed descriptions of the ramifications of the current dose limits (Gitterman and Webster 1984), the net impact from the revised limits is anticipated to be insignificant for this analysis.

### 4.4.2 Cost Implications

Because licensee operations are not expected to be impacted by the revised limit, no operations costs are anticipated. However, the rule requiring demonstration of compliance may involve some added costs. The magnitude of these costs depends on the interpretation of the rule, which does not provide detailed procedures for how compliance must be demonstrated, nor does it provide details on which licensees are affected by the rule. Discussions with NRC personnel indicate that the intent of the rule is not to require increased monitoring of the environment, but to require licensees to maintain records showing compliance. Therefore, no significant costs other than recordkeeping costs are anticipated from promulgation of both the revised limits and the requirement to demonstrate compliance. The associated recordkeeping costs are discussed in Section 5.4 of this report.

## 4.5 SUMMARY OF CONSEQUENCES

In this section, the consequences of the revised limits are summarized. Table 4.5 summarizes these consequences by facility category (as defined in Section 8.1) and is a compendium of the consequences identified in Sections 4.1 to 4.4 in this report. The impacts from the revised limits are anticipated to be negligible for facility categories not listed in Table 4.5. The positive present value of \$10 million listed in the table indicates that the expected benefit from the dose savings to the embryo/fetus exceeds the costs that will be incurred by licensees to comply with the revised dose limits.

山谷

1 martin

15 merch

(注)))()))

TABLE 4.5. Summary of Consequences of the Revised Limits

Facility Category	Initial Cost, \$	Annual Cost, \$	Value of Annual Dose Savings, \$(a)	Net Present Value of Costs and Benefits, \$(b)
Medical Commercial Power Reactors	0	540,000	2,700,000	+19,000,000
	280,000	1,300,000	300,000	- 9,000,000
Total	280,000	1,800,000	3,000,000	+10,000,000

(a) Based on a value of \$10,000/fetus-rem.
(b) Calculated using methods described in Sections 8.2 and 8.3 of this report. The present values are based on 1989 dollars, a discount rate of 10% (applied to all costs and benefits), and a 30-year period. A positive sign indicates a positive impact and a negative sign indicates a negative impact.

# 5.0 CONSEQUENCES OF OTHER CHANGES

In this section, the consequences of other changes in the revision are discussed. These include changes regarding control of internal exposure in restricted areas, precautionary procedures, waste disposal, recordkeeping requirements not discussed in Section 3.5, reporting requirements, and others. These consequences are summarized in Section 5.7.

# 5.1 CONTROL OF INTERNAL EXPOSURE IN RESTRICTED AREAS

Under the present Part 20, in areas of airborne contamination, licensees are required to maintain intakes of radioactive material as low as is reasonably achievable without regard to external exposures. In many circumstances this mandates the use of respiratory protective equipment. Under the revised Part 20, internal exposures are considered a part of the worker's total radiation exposure and the licensee is required to keep the total effective dose equivalent ALARA. This permits the licensee to base ALARA decisions on both internal and external exposure rates, and could lead to a long-term net benefit in both collective dose and program costs.

Informal studies at Three Mile Island Unit 2 (TMI-2) indicated an overall dose savings of up to 50% (internal plus external dose) when the use of a respirator was discontinued (Cardarelli et al. 1986). The higher doses with a respirator were due to loss of productivity and consequent lengthened work times. It is well known that wearing respirators can lead to anxiety, which is associated with increased breathing resistance, increased body temperatures, decreased communication capabilities, limited vision, and general discomfort, all of which reduce work efficiency. A controlled experiment (Cardarelli et al. 1986) failed to show a statistically signific work time for a selected task with and without a respirator. It was observed that due to the discomfort of the respirator the workers operated faster, but this would probably not be true during longer, more complex tasks.

Information on accumulated doses is relily available for the various jobs involving respirator usage; however, information on present respirator usage, job frequency, and air concentrations by job is not readily available. Clearly, both cost savings and dose savings are likely, but the magnitude was difficult to assess accurately without additional information. If there were a 25% decrease in collective dose due to the greater efficiency from not using respirators in high-external, low-internal dose rate fields, this could result in an annual savings of over 600 person-rem during outages (Table 5.1). This is based on collective doses for selected jobs (Dionne and Baum 1985) and assumes a 25% decrease in collective doses for 10% of the outage tasks.

All And

The costs associated with the use of respiratory protection equipment (i.e., costs of canisters and additional personnel to process respirators and track stay times in airborne radioactivity areas) were estimated at over \$30,000 per outage week (Hendrixson, Wagner and Morris 1986). With a typical plant outage lasting 8 weeks (ANS 1988), the annual cost savings for the
Task	Typical Collective Dose Equivalent person-rem(b)	Annual Frequency(c)	No. of Reactors Affected	Saving in Collective Dose Equivalent, person-rem(d)
Assembly/Disassembly Fuel Shuffle	56	0.67	107	100
Snubber Inspection and Repair	75	0.6	107	120
Torus Inspection/Modification Repair	280	0.6	36	151
In-Service Inspection	68	0.6	107	109
Steam Generator(Test/Plug)	76	0.5	71	67
Decontamination	33	0.5	107	44
Reactor Coolant/Circulation Pump Seal Repair	14	0.47	107	
Total				603

TABLE 5.1. Estimated Collective Dose Equivalent Savings During an Outage Through Reduce Respiratory Usage for Selected Tasks(a)

ocar

(a) Data derived from Dionne and Baum (1985).

(b) Collective dose for typical plant based on  $\frac{1}{N_T} (\Sigma N_i \times D)$ 

where Ni is the number of plants of Type i and D is the average collective dose for type i.

(c) (Estimated frequency of task per outage) X (annual frequency of outage [0.67]).

(d) Assumes a 25% reduction in collective dose due to efficiency increases through reduced respiratory usage for 10% of the outage tasks.

5.2

utility industry would be \$1.75 million, counting 0.67 outages for reactor per year, 109 reactors (ANS 1988), and a 10% reduction in the overall use of respirators during outages. However, a decrease in the frequency of respirator use will likely result in increased costs for air monitoring, health physics support, and bioassay measurements (Merwin, Brown and Martin 1987). Therefore, the net annual cost savings are estimated to be one half of \$1,750,000 or \$880,000.

There are large uncertainties in these estimates, and benefits realized on other jobs or during routine operation were ignored. The impact on the non-power reactor segments of the industry were felt to be relatively insignificant and were not considered further.

#### 5.2 PRECAUTIONARY PROCEDURES

Precautionary procedures are set forth in Subpart J of the Part 20 revision. These include procedures for labeling containers of radioactive material, posting caution signs, and handling packages containing radioactive material.

#### 5.2.1 Labeling Requirements

The revised rules that address container labeling are essentially the same as the present rules. In both cases, containers of radioactive material must be labeled with caution signs, and the labels must provide information on the radioactive material present in the containers. Containers are exempt from the labeling requirement if they contain concentrations or quantities of radioactive material less than the values tabulated in Appendixes B and C.

The significant differences between the revised and present requirements are the exempt quantities provided in Appendix C. The present Appendix C lists the exempt quantities of 180 separate radionuclides, whereas the revised Appendix C lists the exempt quantities of 761 separate radionuclides. Of the 180 radionuclides listed in the revised Appendix C that are also listed in the present Appendix C, the exempt quantities in the revision are unchanged for 93 radionuclides, higher for 79 radionuclides, and lower for 8 radionuclides. Of the 87 exempt quantities that were changed, 76 were changed by a factor of ten and 11 were changed by a factor of 100.

The current exercit quantities of the radionuclides not listed in the present Appendix C are equal to one of two default values, depending on whether or not the radionuclide is an alpha emitter. Of the 581 radionuclides listed in the revised Appendix C that are not listed in the present Appendix C, 557 of the revised exempt quantities are equal to, or in most cases higher than the current default quantities. Only 24 revised quantities are lower than the relevant default values listed in the present Appendix C.

Because the Part 20 revision relaxes the requirement for labeling for most radionuclides, there could be a net cost savings from the revised labeling requirements. However, most of the radionuclides listed in the revised Appendix C are not commonly found in licensee facilities. For most Solution the revised labeling requirements is presented in Section 7.7.

#### 5.2.2 Posting Requirements

The revised posting requirements, which address the types of posting required for rooms containing radioactive material or having specified radiation dose rates, are essentially equivalent to the current requirements. The minor changes are not expected to result in significant cost impacts.

#### 5.2.3 Package Handling Requirements

These requirements address the receiving, monitoring, and spearing of packages containing radioactive material. The changes are relatively minor and no significant impacts from the revision are anticipated.

## 5.3 WASTE DISPOSAL

The revision of Part 20 explicitly permits onsite storage of radioactive wastes to allow the r dioactivity to decay. The expense of radioactive waste disposal and the lack of facilities has forced licensees to segregate, recycle, and compact wastes in an attempt to reduce waste volume and, thus, costs (Bunker 1985). More significant than cost is the possible exclusion from commercial low-level waste disposal facilitie: which the Low-Level Radioactive Waste Policy Act of 1990 (P.L. 96-573) portends. This has led to the building of onsite radwaste storage facilities as an alternative to immediate disposal. The NRC issued a generic letter (USNRC 1981) that provides guidance for temporary (5-yr) onsite radwaste storage at power reactors. Temporary onsite storage is regarded as a contingency that would become legalized by the new Part 20. This change will have little impact on the industry since facilities are already being built at medical research institutions (Masse 1984) and nuclear power plants (Rutland and Tuohy 1984; Kemper, Kohlerand and Scholz 1984). The NRC encourages medical licensees to modify their license to store wastes with half-lives up to 100 days for decay and disposal by conventional means. The value of this change is already being realized and the cost impact is already being borne by NRC licensees.

The only changes regarding waste disposal requirements that are expected to have a significant impact are the revised requirements for disposal by release into sanitary sewerage. Although there are some changes in the wording of the rule, they are relatively insignificant compared to the changes in the concentration limits listed in Appendix B. Under the current Part 20, releases to sewers must be in concentrations lower than the values listed in Appendix B, Table I, Column 2. The limiting concentrations in the revised Part 20 are listed in Appendix B, Table 3. A comparison of the current and revised average concentration limits for sewage disposal is presented in Table 5.2.

Table 5.2 indicates that for the radionuclides listed, the revised concentration limits are generally one to two orders of magnitude lower than the current values. For alpha emitters, the disparity is often greater than two orders of magnitude.

It is anticipated that some licensees will be required to take action to reduce the concentrations of radionuclides that are released to the sanitary sewerage system. Based on a review of published data on releases of radionuclides by licensees (Ti' ler and Norden 1986; Cock 1981) and on a review of numerous NRC inspection reports, it i, concluded that some medical, academic/ research, manufacturing and distribution, and nuclear laundry licensees could be affected by the revised release limits. Because of the large volumes of water released by most major medical and academic/research licensees, the impact on these licensees should be relatively small. It is likely, however, that some manufacturing and distribution and nuclear laundry licensees could be required to improve filtration systems, increase holdup times, and/or increase the amount of water released to sewers in order to comply with the revised limits. Nuclear power plants should not be affected because releases from these facilities are currently very low in accordance with the requirements in 10 CFR 50, Appendix I. Nilson and Malody (1982) suggested that some fuel fabrication facilities might be in acted by the revised limit for 20 U: however, the magnitude of this potential impact is uncertain, and may be smaller than originally anticipated. Because the available data are not sufficient for a complete evaluation of these potential impacts, the impact of the revised limits for releases into sanitary sewerage was not quantified in this analysis; however, the impact on some licensee operations could be significant.

#### 5.4 RECORDKEEPING REQUIREMENTS

Subpart L of the revision describes the records required to be kept by licensees. As discussed in Section 3.5 of this report, the requirements for recording individual monitoring results will be costly to licensees because more information must be evaluated and recorded than under the present Part 20. The requirement to maintain records of radiation protection programs is included in the discussion presented in Section 5.6. Changes regarding other recordseeping requirements will have relatively insignificant impacts, as described below: Same.

S. Mett

<u>Records of Surveys</u>. Although the revised requirements are more detailed than the present requirements, essentially all licensees currently keep records required by the revised Part 20. Therefore, no significant impact is anticipated. TABLE 5.2. Current and Revised Average Concentration Limits for Releases into Sanitary Sewerage

Radionuclide	Current Limit, <u>#Ci/mL</u>	Revised Limit, <u>wCi/mL</u>	Reduction Factor, Current/Revised
3 <sub>н</sub>	1E-1	1E-2	10 ,
14 <sub>C</sub>	2E-2	3E-4	70
32p	5E-4	9E-5	60
35 <sub>S</sub>	2E-3	18-3	2
45 <sub>Ca</sub>	3E-4	2E-4	2
51 <sub>Cr</sub>	5E-2	5E-3	10
54 <sub>Mr</sub>	4E-3	3E-4	10
59 <sub>Fe</sub>	2E-3	1E-4	20
60 <sub>Co</sub>	1E-3	3E-5	30
65 <sub>Zn</sub>	3E-3	5E-5	60
86 <sub>Rb</sub>	2E-3	7E-5	30
90 <sub>Sr</sub>	1E-5	4E-6	2
99 <sub>Mo</sub>	5E-3	1E-4	50
99m <sub>Tc</sub>	2E-1	1E-2	20
125 <sub>1</sub>	4E-5	2E-5	2
129 <sub>1</sub>	1E-5	32-6	3
131 <sub>I</sub>	6E-5	1E-5	6
137 <sub>Cs</sub>	4E-4	1E-5	40
144 <sub>Ce</sub>	3E-4	3E-5	10
235 <sub>U</sub>	8E-4	3E-6	300
238 <sub>Pu</sub>	1E-4	2E-6	50
241 <sub>Am</sub>	1E-4	3E-7	300

States.

112 A. (192)

<u>Ustermination of Prior Occupational Dose</u>. The primary change regarding requirements for determining prior occupational dose is that licensees must "attempt" to obtain lifetime records of dose before permitting individuals who will require monitoring to enter the licensee's restricted or controlled areas. Under the current Part 20, licensees must obtain record of prior doses only for the current calendar quarter, unless they plan to use the 5(N-18) dose-averaging provision. Only then are licensees required to determine lifetime cumulative doses.

There could be both cosis and benefits from the revised requirements. Because licensees will be required to determine lifetime cumulative doses for more workers, additional costs will be incurred. On the other hand, the revised requirements will result in better tracking of individual doses and licensees will expend less effort per worker to determine prior doses. The latter point is especially important for tracking transient worker doses, where under the current recordkeeping requirements time is often lost waiting for and verifying exposure histories (Hageman, Artz and Humphress 1982). Because the revised requirements will result in both costs and benefits that are difficult to quantify, the net impact was not estimated in this analysis.

Records of Planned Special Exposures. As discussed in Section 4.2, the planned special exposure provision will not be used frequently. Therefore, the associated recordkeeping costs are assumed to be insignificant for this analysis.

Records of Dose to Individual Members of the Public. Under this new requirement, licensees must maintain records sufficient to demonstrate compliance with the dose limit for individual members of the public. Although not explicitly required under the present Part 20, records currently being maintained by most licensees should be sufficient to demonstrate compliance. The only licensees that are likely to be affected by the new requirement are some medical facilities, academic/research institutions, and manufacturing and distribution facilities, where health physicists or radiation safety officers will maintain more detailed records of radioactive releases and inventories of radioactive materials. It is estimated that for facilities in these categories, a health physicist or other appropriate professional will devote an average of 2 hours per year to these tasks. Based on a cost of \$25 per hour for health physicists (SEA 1986) and 9027 licensees in these categories (see Section 8.1), the estimated annual cost is \$450,000.

<u>Records of Waste Disposal</u>. The changes that appear in the revised Part 20 are relatively minor and are expected to have an insignificant impact on licensees.

Records of Testing Entry Control Devices for Very High Radiation Areas. Although licensees are currently required to test entry control devices, they are not required to maintain test records. Although this requirement in the revision is new, most licensees voluntarily keep records to demonstrate compliance with the current testing requirements. Therefore, no significant impact is anticipated.

#### 5.5 REPORTING REQUIREMENTS

The NRC currently requires that licensees report the theft or loss of licensed material and incidents involving radiation exposures that exceed the annual limits. Certain licensees are also required to submit an annual summary report of radiation exposures received by workers (Brooks 1986). However, the revision contains three significant changes regarding reporting requirements. First, planned special exposures must be reported to the NRC within 15 days after the exposure occurs. A similar requirement is not included in the current Part 20 because planned special exposures are not currently allowed. Second, licensees that are currently required to submit annual exposure data will be required to submit separate reports for each individual for whom monitoring was required rather than one report containing summarized information. Third, all licensees will be required to submit reports, at least annually, to all individuals for whom monitoring was required, indicating the doses that they received in the workplace. Although the latter requirement will actually be included in the revised 10 CFR Part 19, Section 19.13, it is included in the analysis provided in this report.

# 5.5.1 Incident Reports

Some incidents that are not reportable under the current requirements will be reportable under the revised requirements. Such incidents, including planned special exposures, are expected to be rare. Also, the cost of reporting an incident to the NRC is relatively insignificant. As a result, no significant impact is anticipated from the revised incident reporting requirements.

#### 5.5.2 Reports of Personnel Monitoring

Under the current Part 20, certain licensees are required to submit an annual statistical summary report containing information on doses received by personnel. Under the revised requirements, individual reports rather than a summary report must be submitted. As a result, the affected licensees will annually submit one NRC Form 5 (or equivalent) for each individual for whom monitoring was required, rather than one report that summarizes all of the individual exposure information.

Because licensees are required to record individual dose information on NRC Form 5 (or equivalent) under a separate section of Part 20, the revised reporting requirement is not anticipated to result in significant costs regarding collection of information. The affected licensee will simply submit copies of forms they are required to maintain under the recordkeeping requirements. The amount of time required to assemble and submit these forms is anticipated to be equivalent to the amount of time expended under the current requirements to prepare a statistical summary report. Additional mailing costs will be incurred under the revised requirements, but these costs are assumed to be negligible for this analysis. Costs associated with updating computer programs and data bases so that licensees can process the reportable exposure information were included under the category of recordkeeping requirements (Section 3.5). The significant costs from the revised requirements for reporting personnel exposures to the NRC will be incurred by the NRC itself. The NRC personnel responsible for processing licensee exposure data estimate that the marginal annual cost increase for processing exposure data could range from zero to \$300,000 per year, depending on the format of the reports submitted by the licensees. If all licensees provide computer disks containing ' exposure information in NRC Form 5 format, no additional annual costs would be needed to process the data. However, the revised rule only requires that the reports contain all information required in NRC Form 5; the revised rule does not require that NRC Form 5 be used. Personnel at NRC estimate that approximately 30% of licensees will submit computer disks containing exposure information in NRC Form 5 format. Therefore, the estimated marginal cost increase is 70% of \$300,000, or \$210,000 per year. The NRC personnel also anticipate a one-time cost of \$20,000 to upgrade software for processing the new data.

#### 5.5.3 Reports to Individuals

Under the revised 10 CFR Part 19, all NRC licensees will be required to submit individual dose reports to all individuals for whom monitoring was required. As a minimum, licensees will be required to provide the reports at least annually. Three factors were considered in estimating the impact on licensees: 1) some licensees currently provide these reports voluntarily; 2) some licensees may choose to provide these reports more than once per year; and 3) under the current requirements, licensees must provide these reports to individuals who request them. Based on discussions with licensees and former licensee health physicists and radiation safety officers, it is assumed for this report that 1) 25% of licensees currently comply with the revised requirement, 2) licensees who choose to provide reports more frequently than once per year perceive a benefit from doing so (thus, the marginal cost can be estimated by assuming all licensees provide reports is insignificant.

Based on the data provided in Section 8.1 and the assumptions listed above, it is estimated that an additional 583,000 reports per year (one each for 75% of monitored employees, see Table 8.3) will be submitted to individual workers. This estimate is also based on the assumption that for convenience, licensees will provide reports to all individuals actually monitored, not only to those individuals for whom monitoring is required.

Licensees will have three primary options for providing reports to individuals. First, larger facilities may maintain a database of personnel exposures and simply print the information on a form that is provided to the individuals. Second, smaller facilities that record personnel exposure data by hand will probably prepare annual individual reports by hand. Third, licensees who use the services of a dosimetry processor may pay the processor to prepare the annual summary reports.

It is estimated that no matter which of these three options is used, the cost per individual report will range from \$1 to \$2. For this analysis,

an average value of \$1.50 per report is assumed. Based on 583,000 additional reports per year, the marginal annual cost would be \$870,000. In addition, licensees that choose to maintain a database will be required to modify existing software. It is estimated that 25% of all medical, academic/ research, and manufacturing and distribution licensees will select this option. This option will also be selected by an estimated 90% of all commercial power reactor and fuel fabrication facilities. A relatively insignificant fraction of other licensees will choose this option. A cost estimate for software modification is based on several factors: an estimated 1 person-week of effort by a computer programmer per licensee, as well as hourly costs of \$33 for programmers at nuclear power ; ants (SEA 1986) and fuel fabrication facilities and \$23 for programmers at other facilities. It is further assumed that 25% of the licensees currently provide reports voluntarily. Based on these factors, the estimated one-time cost for software modifications is \$1,700,000. Table 5.3 presents the estimated costs by facility category. The present values provided in the table were calculated using methods described in Section 8.2 of this report.

TABLE 5.3. Cost of Providing Reports to Individuals by Facility Category

Facility Category	Initial Cost, \$	Annual Cost, \$/yr	Present Value of Costs, \$
Medical	1,100,000	220,000	2,900,000
Well Logging	negligible	44,000	380,000
Industrial Radiography	negligible	22,000	190,000
Manufacturing and Distribution	170,000	140,000	1.400,000
Academic/Research Institutions	270,000	110,000	1,200,000
Other Measuring Systems	negligible	43,000	370,000
Fuel Fabrication and Processing	12,000	11,000	110,000
Research and Test Reactors	wegligible	4,000	34,000
Commercial Power Reactors	97,000	240,000	2,100,000
All Others	negligible	49,000	420,000
Totals	1,700,000	880,000	9,100,000

## 5.6 OTHER CHANGES

Miscellaneous changes in the revision that do not fall under one of the categories discussed previously in this report are discussed in this section. In general, these changes are expected to have minimal impacts on licensees. Changes in the revision that are editorial changes only are not discussed in this report.

#### 5.6.1 Units of Radiation Dose

Although the revision references the International System of Units (SI), it presents the conventional units found in the present Part 20. Thus, licensees will not be required to use the SI system for purposes of recordkeeping or reporting. In addition, the default quality factor (Q) for neutrons remains at 10 and it is unlikely that licensees will be impacted by the slight changes in the table that presents factors for converting neutron fluence to dose.

### 5.6.2 Radiation Protection Programs

The revised Part 20 states that licensees must develop and document a radiation protection program that is "commensurate with the scope and extent of licensed activities...." In addition, licensees are required to use, "to the extent practicable," procedures and controls to maintain doses ALARA. In comparison, the present Part 20 does not specifically require a radiation protection program to be maintained, but does require that doses be maintained ALARA. An important point here is that the revision does not require a documented ALARA program, a requirement that was present in the proposed Part 20.

Although the new requirement to develop a radiation protection program may appear to place a substantial burden on some licensees, the marginal impact will not be severe because most licensees currently main in a radiation protection program that is commensurate with the scope of their licensed activities. These programs are in place because of either license conditions, or regulatory guides, or both. For example, all nuclear power plants maintain both radiation protection programs and ALARA programs that greatly exceed the requirements in the revised Part 20. Also, most medical licensees maintain programs commensurate with Regulatory Guide 8.18 and a supporting publication (Brodsky 1982).

The only marginal costs anticipated from this requirement will be increased program documentation for a few licensees. For those licensees that have radiation protection manuals, the manuals should serve as acceptable documentation. (See Section 6.2 for cost estimates pertaining to revisions of these manuals.) For licensees that do not have radiation protection manuals, the limited scope of their activities should be sufficient to prevent the need for development of a program. Thus, for this analysis, it is assumed that the associated costs are negligible. However, the costs could be significant depending on the interpretation of the requirement by both licensees and the NRC. The NRC plans to publish several regulatory guides that will provide information on acceptable radiation protection programs for specific categories of licensees.

# 5.6.3 Control of Access to High Radiation Areas

The revised Part 20 provides more detailed requirements than the present Part 20 for controlling access to high and very high radiation, areas. Although the changes will affect some licensees, the associated costs are assumed to be negligible in this analysis.

#### 5.7 SUMMARY OF CONSEQUENCES

In this section, the estimated impacts from other changes in the revised Part 20 are summarized. Table 5.4 summarizes these impacts by licensee facility category. It is a compendium of the costs and benefits identified in Sections 5.1 through 5.6 of this report.

Facility Category	Initial Cost, \$	Annual Cost. \$	Value of Annual Dose Savings, \$(a)	Net Present Value of Costs and Benefits, \$(b)
Medical	1,100,000(c)	540,000(c)	negligible	-5,600,000(c)
Well Logging	negligible	44,000	negligible	-380,000
Industrial Radiography	negligible	22,000	negligible	-190,000
Manufacturing and Distribution	170,000(c)	190,000(c)	negligible	-1,800,000(c)
Academic/Research Institutions	270,000(c)	180,000(c)	negligible	-1,800,000(c)
Other Measuring Systems	negligible	43,000	negligible	-370,000
Fuel Fabrication and Processing	12,000(c)	11,000(c)	negligible	-110,000(c)
Research and Test Reactors	negligible	4,000	negligible	-34,000
Commercial Power Reactors	97,000	(640,000) (d)	610,000	+11,000,000
All Others	negligible(c)	50,000(c)	negligible	-430,000(0,
NRC	20,000	210,000	negligible	-1,800,000
Total	1,700,000(c)	650,000(c)	610,000	-2,000,000(c)

TABLE 5.4. Summary of Marginal Cost Increases from Other Changes in the Revised Part 20

(a) Based on a value of \$1000/person-rem.

(b) Calculated using methods described in Sections 8.2 and 8.3 of this report. The present values are based on 1989 dollars, a discount rate of 10% (applied to all costs and benefits), and a 30-year period. A positive sign indicates a positive impact and a negative sign indicates a negative impact.

WARE IN

(c) Not including potential costs from revised concentration limits for releases into sanitary sewerage.

(d) Number in parentheses indicates a cost savings.

5.13

# 6.0 OTHER CONSEQUENCES OF THE REVISION

Some of the costs associated with the revision of Part 20 cannot be attributed to specific sections of the rule. These costs include training of personnel and revision of procedures, and are discussed in Sections 6.1 and 6.2, respectively. Impacts on the NRC and small businesses are discussed in Sections 6.3 and 6.4, respectively. These consequences are summarized in Section 6.5.

Many of the costs identified in this section were calculated using published labor rates (including overhead costs) for nuclear power plant workers (SEA 1986). Because labor rates for non-power-reactor workers are generally lower, these costs were estimated by applying a factor of 0.7 to the rates for power reactor workers where applicable. All costs were adjusted to 1989 dollars.

#### 6.1 PERSCNNEL TRAINING

Some licensees will incur significant costs associated with training of personnel. Supervisory, health physics, and other professional personnel as well as general employees might require training depending on the facility. In the subsections below, cost estimates are provided for the ten categories of facilities defined in this report (see Section 8.1). In most cases, the costs per facility have a wide range within each category. Where appropriate, cost estimates are provided for facilities within a category.

An important consideration in developing these cost estimates was that personnel training is performed routinely at most facilities. Personnel are often required to attend periodic training sessions conducted or sponsored by the licensees. The content and leve' of training with regard to radiation protection depends on the work performed by the employee.

To eliminate nonmarginal costs from this analysis, a factor of 0.75 has been applied to some of the cost estimates to account for the increased training or retraining in response to the revised Part 20 that can be incorporated into routine training programs. In addition, intensive training immediately following promulgation of the revision will result in a decreased need for training in the short term. In some cases, the marginal cost factor should be higher or lower than 0.75. Where applicable, the appropriate factor is provided in the sections below.

Some of the costs estimated in this section (and other sections) are considered to be marginal even though it is likely that in many cases, the costs will be absorbed during the normal daily activities of the relevant personnel. For radiation protection personnel whose primary function is to ensure that operations by the facility are in compliance with radiation protection regulations, part of their responsibility is to be familiar with the regulations and familiarization with new regulations would not result in increased actual costs to the facilities. Medical. It is anticipated that nuclear physicians and nuclear medicine technologists would use their allotted time for continuing education to inform themselves of the content of the revised regulations and the ICRP Publication 26 recommendations (USNRC 1982). For this analysis, however, some of these costs are considered to be marginal and must be included in the assessment of the impact of the revision. It is estimated that 5% of the workers monitored for radiation exposure at thr e facilities will commit 2 hours each to studying the revised regulations and their implications. At a cost of \$14 per technician hour (SEA 1986) and applying the data in Tables 8.2 and 8.3, the cost estimate for these workers is 193,000 X 0.05 X 2 X \$14, or \$270,000. Applying the marginal cost factor of 0.75, the marginal costs are estimated to be \$200,000. Costs for professionals to study the regulations are included in the cost estimates provided in Section 6.2.

<u>Well Logging</u>. No significant retraining of workers or managers is anticipated for these facilities (USNRC 1982). All costs associated with studying the revision are accounted for in Section 6.2.

Industrial Radiography. In a survey of several of these facilities, none envisioned a need for retraining personnel (USNRC 1982). Thus, no significant costs are anticipated. All costs associated with studying the revision by responsible management are accounted for in Section 6.

Manufacturing and Distribution. It has been estimated that for facilities in this category, supervisors, health physicists, and other professionals will commit 250 person-hours, 200 person-hours, and 150 person-hours, respectively, to retraining (USNRC 1982). Other costs associated with retraining are assumed to be insignificant for this facility category. Assuming \$31 per professional hour (SEA 1986), the total cost is estimated to be \$19,000. The associated marginal cost estimate is \$19,000 X 0.75, or \$14,000.

<u>Academic/Research Institutions</u>. It is anticipated that users of radioactive material at these institutions will learn of the Part 20 revision and its implications through routine training and information programs. All of these costs are considered to be nonmarginal. Costs incurred by radiation safety officers to understand and implement the revised requirements are included in the cost estimates provided in Section 3.5.

Other Measuring Systems. No significant costs regarding personnel training will be required for these facilities. However, responsible personnel at most of the facilities will need to become familiar with the revision. An average of 2 person-hours for each facility is estimated. At a cost of \$31 per professional hour (SEA 1986) and 5060 facilities (see Section 8.1), the marginal cost estimate is \$310,000.

Fuel Fabrication and Processing. Booth, Bronson and Groth (1985) estimated that the revision would result in training costs of \$145,000 (in 1989 dollars) at a typical fuel fabrication facility. For this analysis, it is estimated that of the 14 facilities in this category (see Section 8.1), five will incur marginal costs of \$145,000 and nine will incur one-fourth of those costs, or \$36,000. The total cost estimate for fuel fabrication facilities is \$1,000,000.

Research and Test Reactors. For these facilities, retraining of personnel will be absorbed in the normal training cycles (USNRC 1982). However, health physics personnel will need to become familiar with the revision. It is estimated that 8 hours will be committed per facility to become familiar with the revision and its ramifications. Assuming a cost of \$31 per professional hour (SEA 1986) and 80 facilities (see Section 8.1), the total cost would be \$20,000.

<u>Commercial Power Reactors</u>. Published cost estimates for training and retraining of personnel at commercial power reactors in response to the revision of 10 CFR 20 are available (Booth, Bronson, and Groth 1985; USNRC 1986) and are given in the public comments on the proposed revision. It appears that training for a new employee under the revised Part 20 will be comparable to current training for new employees. However, most employees will need to be retrained to become familiar with the revised regulations and the relevant revisions to plant procedures. In addition, health physics and dosimetry personnel will need to be retrained in the areas of their work affected by the revision.

It has been estimated that at a typical nuclear power plant, total costs associated with training are \$175,000 (in 1989 dollars) (Booth, Bronson and Groth 1985). For this analysis, it is estimated that two-thirds of these costs are marginal. Further, a factor of 0.75 is applied to these costs to account for the fact that some nuclear power staticns have multiple units, i.e., the training costs per unit at these sites will be less than at a typical one-unit station. For 109 nuclear power plants expected to be operating in 1989 (ANS 1988), the marginal cost estimate is \$175,000 X 0.67 X 0.75 X 109, or \$9,500,000.

- 73962

Sala -

Section .

in anti-

<u>All Others</u>. In general, the licensees in this category do not have extensive radiation protection programs and general employee training will not be required. However, radiation protection personnel responsible for compliance with NRC regulations will probably read the revised Part 20 to become familiar with its provisions. It is estimated that an average of 2 hours per facility will be spent reading the revision and associated documents. Assuming a cost of \$31 per professional hour (SEA 1986) and 1752 facilities in this category (see Section 8.1), the total cost would be \$110,000. Applying the marginal cost factor of 0.75, the marginal cost estimate for this category of licensees is \$81,000.

## 6.2 PROCEDURE REVISIONS

Some licensees will incur significant costs to incorporate the revised requirements into existing procedures and related documents. The magnitude of the impact on individual licensees is related to the magnitude of the licensee's radiation protection program. The effected documents could include operating procedures, radiation protection manuals, and policy statements (training manual revisions were included in the estimates provided in Section 6.1). In this section, the costs associated with revising recordkeeping procedures are not included because they were included in the recordkeeping costs discussed in Section 3.5.

In the subsections below, cost estimates are provided for the ten categories of facilities defined for this analysis (see Section 8.1). In most cases, the costs per facility have a wide range within each category. Where appropriate, cost estimates are provided for separate classes of facilities within a category.

An important consideration in developing these cost estimates was that many procedures and related documents are routinely revised by licensees. Therefore, all of the costs associated with procedure revisions will not be marginal and should not be considered in this analysis. Several competing factors exist in separating the marginal costs from the nonmarginal costs: 1) considering the implementation period permitted by the NRC, some revisions can be incorporated without interrupting the usual schedule; 2) considering both the implementation period and the magnitude of the revised Part 20, some revisions cannot be incorporated without interrupting the usual schedule; 3) some revisions will be much more extensive than they would have been without a revised Part 20; and 4) extensive procedure revisions in response to the revised Part 20 will decrease the need for further revisions in the short term. In consideration of these factors, for this analysis the marginal costs of revising procedures are estimated to be 75% of the total costs, unless otherwise noted.

<u>Medical</u>. For these facilities, the cost of required procedure changes depends greatly on the type of facility. Of the 6506 facilities in this category (see Section 8.1), an estimated 4735 are classified as medical institutions (Hendrickson et al. 1987). The remaining facilities include private practice of nuclear medicine, in vitro labs, veterinary medicine, and others (see Table 8.1).

It is anticipated that a radiation protection manual at each medical institution will need to be revised. In addition to the procedure changes regarding recordkeeping that were estimated in Section 3.5, an estimated 8 person-hours will be committed to this task (USNRC 1982). At a cost of \$31 per professional hour (SEA 1986), and assuming that \$400 in printing and duplicating costs will be required per facility, the total cost estimate is \$3,100,000.

It is expected that the costs of required procedure changes at other medical facilities will be limited to recordkeeping changes associated with the recording of infrequent internal exposures. (These costs were estimated in Section 3.5.) The marginal cost estimate for procedure revisions for all medical facilities is \$3,100,000 X 0.75, or \$2,300,000.

Well Logging. It is estimated that only five of the licensees in this category have radiation protection programs extensive enough to warrant procedure revisions in radiation protection manuals (USNRC 1982). It is estimated that for each of the e firms, 80 person-hours of management time will be devoted to this work, and an additional \$800 per firm will be spent to produce revised manuals. At a first \$31 per hour for management personnel (SEA 1986), the total cost first five firms is estimated to be \$16,000. The other firms in the industry are not expected to incur significant costs. Therefore, the marginal cost of procedure revisions is estimated to be \$16,000 X 0.75, or \$12,000.

Industrial Radiography. Most of the facilities in this industry have an in-house document that serves as a radiation protection manual (USNRC 1982). Because internal exposures are not a problem in this segment of the nuclear industry, it is not anticipated that extensive revisions to the existing manuals would be required in response to the revised Part 20. It is estimated that at a typical facility, one professional will spend 8 hours revising the radiation protection manual. Also, \$200 will be spent to prepare and print the manual. At an hourly cost of \$31 (SEA 1986) and assuming 851 industrial radiography facilities currently in operation (see Section 8.1), the total cost to these facilities would be \$380,000. The marginal costs are estimated to be \$380,000 x 0.75, or \$290,000.

Manufacturing and Distribution. Of the facilities in these categories, required procedure revisions would be insignificant except for some source manufacturing firms and most nuclear pharmacies. It is anticipated that ten source manufacturing firms will require extensive procedure revisions, at a cost of 16 person-weeks of professional time per facility. At a cost of \$31 per professional hour (SEA 1986) and \$500 in printing and duplicating costs per facility, the total cost would be \$200,000.

For nuclear pharmacies, it is expected that minor revisions of radiation safety manuals will be necessary, requiring 8 hours of professional time per facility. Printing costs of about \$300 per facility will also be required. At a cost of \$31 per professional hour and with 221 facilities in this category (Hendrickson et al. 1987), the total cost for these facilities would be \$120,000.

Costs of required procedure changes at other facilities in this category are expected to be relatively insignificant. Therefore, the marginal cost estimate is (\$200,000 + \$120,000) X 0.75, or \$240,000.

Academic/Research Institutions. Most of the facilities in this category have a radiation protection manual that provides guidance to users of radioactive material. In general, these manuals will not require revisions based on the revised Part 20.

As discussed in Section 3.5, radiation safety officers or other personnel at these facilities will be required to revise recordkeeping procedures to allow calculation of internal doses should significant intakes occur. No additional costs related to procedure revisions have been identified for these licensees.

Other Measuring Systems. Few of these licensees have radiation protection manuals or procedures that would require revisions based on the revised Part 20 (USNRC 1982). For those that do, the costs associated with the revisions are assumed to be insignificant for this analysis. Fuel Fabrication and Processing. These facilities wiil require extensive revisions to air sampling and bioassay procedures in response to the revised dose evaluation requirements. Detailed cost estimates by Booth, Bronson, and Groth (1985) suggest that the associated costs at a typical facility are \$86,000 (in 1989 dollars). For this analysis, it is estimated that of the 14 facilities in this category, five will incur marginal costs of \$86,000 and nine will incur one-fourth of those costs, or \$22,000. The total cost, estimate for fuel fabrication facilities is \$630,000.

Research and Test Reactors. It has been suggested that revisions will be required to safety guides for approximately 70% of the facilities in this category (USNRC 1982). The associated costs per facility will be 2 personmonths of a professional, and \$5000 in preparation, printing and duplicating costs (USNRC 1982). At a cost of \$31 per professional hour (SEA 1986), the total cost based on 80 facilities in this category is \$840,000. The marginal cost estimate is \$840,000 X 0.75, or \$630,000.

Commercial Power Reactors. Commercial power reactors have extensive radiation protection programs that include detailed procedures for external dosimetry, internal dosimetry, recordkeeping, contamination control, etc. Detailed cost estimates for revising these procedures based on the revised Part 20 are available in the literature (Booth, Bronson and Groth 1985). Other than recordkeeping procedures, which were discussed in Section 3.5. licensees will be required to revise procedures related to external and internal dosimetry. Because of the magnitude of the radiation protection programs at these facilities, much of the time associated with these revisions will be spent studying the impact of the revised regulations and designing programs and procedures to comply with the regulations. It is estimated that the associated development costs at a typical nuclear power plant will be \$124,000 in 1989 dollars (Booth, Bronson and Groth 1985). For this analysis, a factor of 0.7 is applied to this estimate to account for the fact that multiple units at some sites share some of the procedure development activities, and as a result the costs of evaluating Part 20 and developing revised procedures will be smaller per facility than at a typical one-unit site. Assuming 109 operating units in 1989 (ANS 1988), the associated marginal cost estimate is \$124,000 X 109 X 0.75 X 0.7, or \$7,100,000.

<u>All Others</u>. In general, the licensees in this category do not have extensive radiation protection programs and do not have procedures that would require revisions based on the revised Part 20. Thus, the associated costs are assumed to be negligible for this analysis.

## 6.3 NRC COSTS

The NRC will incur costs related to the development, implementation, and operation of the revised rule. <u>Development</u> costs include the costs incurred by the NRC to prepare the revised rule for implementation. <u>Implementation</u> costs include the costs incurred to place the revised rule into operation. These costs include preparation and publication of the final rule. Costs associated with preparing and revising regulatory guides in support of the revised rule are also included in this category. <u>Operation</u> costs include those costs incurred by the NRC after the proposed action takes effect. They primarily consist of the costs of enforcing the requirements.

Because this regulatory analysis pertains to both NRC licensees and Agreement State licensees, costs incurred by state regulatory agencies must also be considered. These cost estimates are included in the applicable sections below.

# 6.3.1 Development

The NRC has incurred substantial costs required to research and develop numerous versions of the revision of Part 20 over the past 7 years. More costs are anticipated for further reviews and revisions. However, because this analysis includes only those costs and benefits that will be incurred after the decision is made whether or not to publish the rule, the costs incurred by NRC before that date are not considered. In this report, the assumed decision date is October 1, 1988. No significant NRC development costs after that date are anticipated.

#### 6.3.2 Implementation

The NRC has estimated that nine new regulatory guides will be necessary to support the revised Part 20. In addition, ten existing regulatory guides will require major revisions. It is estimated that 1.0 person-years per new guide and 0.5 person-year per revised guide will be committed. An additional 0.25 person-year is estimated to be required for minor revisions to a number of other existing regulatory guides. Based on a mean hourly rate of \$51 per hour (SEA 1986) and 28,500 total hours of effort, the NRC is expected to incur implementation costs of \$1,500,000 for developing and revising regulatory guides in support of the Part 20 revision.

Because only a few of the new and revised regulatory guides will be crucial to successful implementation of the revision by licensees, it is estimated that 30% of the costs estimated above will be incurred during the 2 years following publication of the revision (1989 and 1990). The remaining 70% of the costs are estimated to be distributed evenly over the following 5 years. In addition, only one-half of the latter are assumed to be marginal costs, because it is thought that the development and revision of similar regulatory guides would have occurred even without the revision of Part 20. As a result, the marginal costs associated with the development and revision of regulatory guides in response to the revision are estimated to be \$220,000 per year in 1989 and 1990 and \$100,000 per year from 1991 through 1995.

The NRC will also incur implementation costs related to the final preparation and publication of the rule. A total of 1 person-year of effort is anticipated to be required at a cost of \$100,000. Finally, 1 > NRC will be required to revise the various forms associated with Part 20, such as NRC Form 4 and NRC Form 5. The associated cost estimate for these revisions is \$25,000.

#### 6.3.3 Operation

Once the Part 20 revision is implemented, the NRC and Agreement States may incur costs related to additional inspection activities and additional training of inspectors. These cost estimates are provided below.

## Additional Inspection Activities

Currently, power reactors are inspected an average of two or three times per year. Most other licensees are inspected anywhere from one to five times per year, depending on past performance and the type of license.

No increase in inspection activities at power reactors is anticipated from promulgation of the revision. Although inspectors will use the revision as the basis for the inspections, the frequency of inspections and the person-hours per inspection are not expected to change.

For other types of licensees, some may require additional inspections to ensure compliance with the revision. However, in consideration of the limited resources that will be available for carrying out additional inspections, it is anticipated that the MRC will reschedule inspections so that the problem facilities receive additional attention and the facilities with satisfactory performance will receive less attention than usual. Problem facilities will be identified through response to a letter-writing campaign. After several years, it is anticipated that inspection scheduling will be unaffected by the presence of the revised Part 20.

In summary, only those costs associated with communication with nonreactor licensees and analyzing the responses will be significant. It is estimated that an average of 0.2 person-hour per facility will be committed, which includes preparation of letters, response by the licenses, and evaluation of the responses by the NRC. (Although costs incurred by licensees are not the subject of this section, they are included here for convenience. Exclusion of these costs from the cost estimates for facility categories does not significantly affect those cost estimates.) Because it is not known how individual Agreement States will schedule inspections in response to the revised Part 20, costs per Agreement State licensee are estimated to be similar. For a total of 17,238 non-reactor licensees (see Section 8.1) and a cost per person-hour of \$46 (SEA 1986), the associated cost estimate is \$160,000.

#### Inspector Training

Because the Part 20 revision is extensive, additional inspector training will be required following publication of the revision. In the five NRC regions, there are approximately 110 inspectors who will need retraining. Each inspector will need to receive an estimated 3 days of retraining, for a total of 2600 person-hours. In addition, an estimated 80 person-hours per region will be needed to revise training procedures and conduct the retraining. This amounts to an additional 400 person-hours for a total of 3000 person-hours. One benefit from the immediate retraining required for all inspectors is that routine training will be able to be curtailed for a few years thereafter. Currently, each inspector receives approximately 1 day of routine training per year. If each receives 3 days of training on the revised Part 20, future training in the short term will be less extensive than usual. This also applies to routine training instruction and training procedure revisions. It is estimated that the marginal inspector training costs associated with the revision are 50% of the 3000 person-hours estimated in the previous paragraph. The associated cost estimate, based on an hourly rate of \$46 (SEA 1986), is \$69,000. These costs are tripled to account for inspector training activities in Agreement States, for a total cost of \$210,000.

## 6.4 EFFECTS ON SMALL BUSINESSES

The Regulatory Flexibility Act of 1980 (Public Law 96-354, 5 USC 601-612) requires that, when a rulemaking action is likely to have a significant impact on a substantial number of small entities, the impact must be addressed specifically. The NRC's Regulatory Analysis Guidelines (USNRC 1984) specify that this analysis should be incorporated into the regulatory analysis for the action.

The NRC specifies that entities are to be considered small businesses for the purposes of the Regulatory Flexibility Act if they meet the following requirements (Hendrickson et al. 1987):

- for most licensees, annual billings of \$3.5 million or less
- for private practice physicians, annual billings of \$1 million or less
- for state or public educational institutions, an institution supported by a jurisdiction with a population of 50,000 or less
- for other educational institutions, an institution having 500 or fewer employees.

Henderickson et al. (1987) analyzed data on NRC and Agreement State licensees based on the results of licensee surveys (see Section 8.1). These surveys were used to estimate the fraction of the licensees in various categories that could be considered small entities. While the estimates in Hendrickson et al. may overstate the number of small businesses in some cases and understate it in others, they are sufficient to provide an indication of the impact on small businesses that is commensurate with the level of effort and detail expended on other portions of this analysis.

1000

Based on data in Hendrickson et al. (1987), estimates for the fraction of licensees in each facility category that could be classified as small businesses were obtained (see Table 6.1). These fractions are assumed to apply to the number of licensees tabulated in Section 8.1, although the absolute numbers differ from those in Hendrickson et al. (1987). The licensee

as small businesses	
Facility Category	Fraction
Medical	
Private Practice Physicians	934
Other Facilities	21%
Well Logging	63%
Industrial Radiography	40%
Manufacturing and Distribution	32%
Academic/Research Institutions	29%
Other Measuring Systems	37%
All Others	25%

TABLE 6.1. Fraction of Licensees That May Be Classified

categories of commercial power reactors, research and test reactors, and fuel fabrication and processing facilities are assumed to contain no small entities.

Table 6.2, which is based on Tables 8.2 and 8.4, presents the average impact of the Part 20 revision per facility for each facility category that includes small businesses.

The monetary impact of the Part 20 revision, averaged over all facilities, is not expected to be substantial for most groups of licensees. The most substantial impacts will be borne by medical facilities. However, the average impact does not account for the fact that not all medical facilities (and private practice physicians) will incur these costs; rather, the costs will often be incurred only by those facilities that already have the most

TABLE 6.2. Per-Facility Costs from the 10 CFR Part 20 Revision

Facility Category	Development and Implementation Costs, \$	Operation Costs, \$/yr	
Medical	1100	280	
Well Logging	26	97	
Industrial Radiography	340	26	
Manufacturing and Distribution	550	250	
Academic/Research Institutions	1200	220	
Other Measuring Systems	61	8	
All Others	46	28	

6.10

extensive radiation protection programs. These facilities are usually the larger businesses that are not classified as "small ent ties."

Table 6.3 indicates the magnitude of the total cost to small entities of the Part 20 revision, should all such entities incur the average cost given in Table 6.2.

In addition to the above mentionry impacts on small businesses, the Part 20 revision will have impacts that were not quantified. As discussed in Section 7.6, the embryo/fetus dose limit may significantly impact small businesses by causing the loss of a trained employee from certain jobs for a temporary period. A small firm usually does not have the ability to shift personnel around without a significant loss in productivity. Licensees may, therefore, be reluctant to employ women of childbearing age because of these costs.

Labeling requirements for products containing radioactive materials may also impact small businesses. As discussed in Sections 5.2 and 7.7, however, few changes with respect to container labeling are anticipated because most of the quantities of radioactive material that are exempt from labeling are either unchanged or less restrictive than under the current Part 20 quantities. Thus, costs incurred because of new labeling requirements are not expected to be substantial. As described in Section 5.2, a small positive net benefit is expected from the changes in labeling requirements. However, for those cases in which labeling requirements have become more stringent under the revised regulations, small businesses may bear a more significant impact than larger firms if the competitive positions of small businesses in both domestic and international markets are degraded when costs are passed along to customers.

Facility Category	Implementation Costs, \$	Operation Costs, \$/yr	
Medical			
Private Practice Physicians	1,000,000	260.000	
Other Facilities	1,300,000	320,000	
Kall Logging	8,000	28,000	
Industrial Radiography	120,000	9,000	
Manufacturing and Distribution	170,000	80,000	
Academic/Research Institutions	540,000	100,000	
Other Measuring Systems	110,000	16,000	
All Others	20,000	12,000	
Total	3,300,000	830,000	

## TABLE 5.3. Estimated Total Cost to Small Businesses from the 10 CFR Part 20 Revision

6.11

# 6.5 SUMMARY OF CONSEQUENCES

In this section, the consequences of the revised Part 20 that are not associated with specific sections of the rule are summarized. Table 6.4 summarizes these consequences by licensee facility category and is a compendium of the costs identified in Sections 6.1 through 6.3 of this report.

TABLE 6.4. Summary of Marginal Cost Increases from Other Consequences of the 10 CFR Part 20 Revision

Facility Category	Initial Cost, \$	Present Value of Costs, \$(a)
Medical	2,500,000	-2,300,000
Well Logging	12,000	-11,000
Industrial Radiography	290,000	-260,000
Manufacturing and Distribution	250,000	-240,000
Academic/Research Institutions	negligible	negligible
Other Measuring Systems	310,000	-280,000
Fuel Fabrication and Processing	1,700,000	-1,500,000
Research and Test Reactors	650,000	-590,000
Commercial Power Reactors	17,000,000	-15,000,000
All Others	81,000	-73,000
NRC	1,400,000	1,100,000
Total	24,000,000	-21,000,000

(a) Calculated using methods described in Section 8.2 of this report. The present values are based on 1989 dollars, a discount rate of 10%, and a 30-year period. A negative sign indicates a negative impact.

# 7.0 QUALITATIVE BENEFITS FROM REVISING PART 20

10 CFR Part 20 specifies the fundamental requirements for ensuring that workers and the public are protected from the hazards of ionizing radiation resulting from NRC-licensed activities. Consequently, Part 20 is the basic foundation for licensing, inspection, and enforcement activities relating to occupational and public health protection. Because the provisions of Part 20 are based primarily on recommendations by the ICRP and NCRP, recent changes in the basic radiation protection concepts promoted by these organizations have prompted the NRC to revise Part 20 to reflect these latest recommendations. The importance of revising Part 20 was reinforced when the EPA published its recent recommendations to Federal agencies, which essentially stressed that agencies incorporate the recent ICRP and NCRP recommendations in their conduct of programs for the protection of workers from ionizing radiation.

The previous four sections of this report presented a quantitative analysis, where possible, of the costs and benefits of the revision of Part 20. In most cases, quantitative estimates of the identified costs were possible. On the other hand, most of the potential benefits from revising Part 20 cannot be readily quantified. These potential benefits are discussed in this chapter.

Before describing these possible benefits, it is important to be clear about what is meant by a benefit. In social benefit/cost analysis, actions or projects are evaluated insofar as ineir effects increase or decrease the welfare of the individuals who are affected. Increases in welfare are associated with benefits, and decreases in welfare are associated with costs. A central theme in benefit/cost analysis is an individual's willingness to pay to receive benefits or accept compensation to bear costs. An effect is a benefit if one or more individuals can be identified who would be willing to pay for the effect from their own wealth.

# 7.1 POTENTIAL BENEFITS FROM CONSISTENCY WITH NATIONAL AND INTERNATIONAL RECOMMENDATIONS

The primary reason for completely revising Part 20 is that the current requirements are inconsistent with recent international and national recomme dations (FR 1986). In 1977, the ICRP published a recommended system of dost limitation based on risk (ICRP 1977). Since that time, this system has been recognized as a major improvement over the old system of dose limitation by m ny, but not all health physicists (Skrable et al. 1985; Healy 1982). Also, some other countries have incorporated the recent ICRP recommendations into heir radiation protection standards (Usui 1987). In 1987, the NCRP publis ed its recommendations, which are essentially consistent with, but in some cates more restrictive than, the ICRP recommendations (NCRP 1987). Also in 1987, the EPA published its recommendations on occupational radiation protection (which were signed by the President) in the Federal Register (FR 1987). As matter of policy, the NRC has considered past Federal Guidance as binding a d has implemented the guidance in its regulations. Unfortunately, the ICRP, NCRP, and EPA recommendations are not among themselves completely consistent. One reason for the inconsistencies is that the NCRP recommendations and the Federal Guidance (NCRP 1987; FR 1987) were published ten years after the relevant ICRP recommendations (ICRP 1977). Note that the proposed revision of Part 20 was published in 1986, before both the NCRP recommendations and the Federal Guidance were published. A comparison of the revision of Part 20 with these three sets of recommendations is presented in Table 7.1.

Table 7.1 indicates that, in general, the revision of Part 20 is either consistent with or more restrictive than the basic recommendations of ICRP 26 and 30, and is either consistent with or less restrictive than the basic recommendations of NCRP 91. The revision is also generally consistent with the Federal Radiation Protection Guidance. However, the revision provides detailed requirements in many cases where the Federal Guidance provides only general guidance.

Four potential benefits from consistency with national and international recommendations were identified and are discussed in this section:

- the knowledge that regulations are in agreement with currently accepted scientific concepts
- elimination of confusion caused by terminology that is based on inconsistent definitions and on inconsistent measurement concepts
- savings in tcaching resources
- increasing the usefulness and applicability of dosimetry data collected under the provisions of Part 20.

Knowledge that Regulations are in Agreement with Currently Accepted Scientific Concepts. Individuals may benefit from the knowledge that federal regulations on radiation protection are based on currently accepted practice. According to the willingness-to-pay principle, the social benefit from this knowledge is calculated conceptually as the total dollars that the affected individuals would be willing to pay to possess this knowledge.

While no attempt is made here to place a monetary value on this possible benefit, some useful insights can be obtained by inquiring into the type of individual who might benefit from it. For this purpose two groups of individuals are considered: a) workers at risk of receiving occupational doses, and b) members of the general public.

Workers at risk from occupational exposure to ionizing radiation might be comforted by knowing that current regulations reflect the latest scientific information on radiological risk. If the regulations are outmoded, a worker may fear that he is not being afforded adequate protection. The benefit to him from revising the current Part 20 is the reduction in his uncertainty regarding the radiological risk. This benefit will occur even if the revision leads to no substantive change in the current provisions relating to workers. TABLE 7.1. Comparison of the Revision of Part 20 with National and International Recommendations

	Consistency with Recommendations of		
Part 20 Section	ICRP(a)	NCRP(a)	Guidance
<ul> <li>20.4: Units of Radiation Dose</li> <li>20.102: Rar ation Protection Programs</li> <li>20.201: Occupational Dose Limits for Adults</li> <li>20.202: Compliance With Summation Requirements</li> <li>20.203: External Dose From Airborne Material</li> <li>20.204: Determination of Internal Exposure</li> <li>20.206: Planned Special Exposures</li> <li>20.207: Occupational Dose Limits for Minors</li> <li>20.208: Dose to an Embryo/Fetus</li> <li>20.301: Dose Limits for Members of the Public</li> <li>20.302: Compliance with Public Dose Limits</li> <li>20.502: Conditions Requiring Indiv. Monitoring</li> <li>20.601,2,3: Exposure Control in Restricted Areas</li> <li>20.702,3,4: Respiratory Protection</li> <li>20.901-6: Precautionary Procedures</li> <li>20.1102: Records of Radiation Protection Programs</li> <li>20.1103: Records of Surveys</li> <li>20.1104: Determination of Prior Occupational Dose</li> <li>20.1105: Records of Individual Monitoring Results</li> <li>20.1107: Pecords of Dose to the Public</li> <li>20.1108: Records of Waste Disposal</li> <li>20.1107: Pecords of Maste Disposal</li> <li>20.1108: Records of Waste Disposal</li> <li>20.1108: Records of Waste Disposal</li> <li>20.1201.2.3: Incident Reporting Requirements</li> </ul>	C(d) C(i(e) C(i(e)) I(c) I(e) I(e) I(e) I(e) I(e) INNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNNN	1((d) (b) (b) (b) (c) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	d) (d) (d) (d) (d) (d) (d) (d) (d) (d) (
20.1204: Reports of Planned Special Exposures 20.1206: Reports of Personnel Monitoring	N	N	G

Codes: C - consistent with recommendations (Part 20 may be more detailed) G - generally referred to in recommendations (Part 20 is more specific)

NAME IN

1.224

- I = inconsistent with recommendations
- N not addressed in recommendations
- (a) Consistency with ICRP 26 & 30 (ICRP 1977, ICRP 1979) and NCRP 91 (NCRP 1987). For those sections in Part 20 that are not applicable to the recommendations in these documents, they may be applicable to recommendations in other ICRP or NCRP documents.
- (b) Revised Part 20 is less restrictive.
- (c) The inconsistency is minor.
- (d) NRC Regulatory Guides are expected to provide detailed guidance consistent with the recommendations.
- (e) Revised Part 20 is more restrictive.

7.3

What can be said about the likely magnitude of this benefit? According to the willingness-to-pay principle, the benefit is equal to the total amount that workers would be willing to pay for the knowledge that, based on the latest scientific findings, the current provisions provide adequate worker protection. But workers who select risky occupations generally perceive their occupational risk to be lower than does the work force at large. Therefore, unless the reductions in uncertainty are relatively large, it seers unlikely that workers would be willing to pay very much to reduce this uncertainty.

There is an interesting sidelight to this effect. A number of studies indicate that occupational risks are reflected in wage levels, so that workers are compensated for the added risk (Low and McPheters 1983). This suggests that if perceptions of risk change, real wages should adjust accordingly. Furthermore, if the risk uncertainty is reduced equally on the upside and downside, then, because individuals tend to be risk-averse, the net benefit will be positive and real wages should adjust lower. It is important to realize, however, that this adjustment in real wages will occur only to the extent that workers are willing to give up real wages in exchange for less risk uncertainty.

Lastly, we inquire whether a regulatory review confers benefits on the general public. In more precise lerms, would a regulatory review that led to no changes in public exposures provide a perceived benefit to members of the general public? In response, it is important to note that the provisions of Part 20 deal primarily with occupational dose, which is probably of relatively little concern to the general public; the focus of the general public seems to be on accidental releases. The provisions of the revised Part 20 that do address public dose are not expected to significantly impact the public. Therefore, benefits of the revision to the general public are deemed to be insignificant.

出版

Elimination of Confusion Caused by Terminology that is Based on Inconsistent Definitions and on Inconsistent Measurement Concepts. Another possible benefit from updating present regulations is that the update may eliminate some confusion in terminology. For example, under a recent ICRP recommendation (ICRP 1977), total effective dose equivalent is defined as including both internal and external doses, with appropriate weighting factors applied. Under the current Part 20, however, external doses and internal doses are treated separately. In some situations it may not be clear which is the correct method for calculating whole-body dose and this could lead to faulty interpretations.

To demonstrate a real benefit from making definitions and concepts consistent with international practice, it is first necessary to show that the current situation does indeed lead to some confusion. At this time, no such situation has been identified.

Savings in Teaching Resources. The benefit in this category is that health physics students and professionals would not be required to understand two inconsistent systems of dose limitation. Currently, most health physics programs at universities teach both systems, but emphasize the newer system because it represents a fundamental improvement over the old system. While older texts are used to teach the principles in the current Part 20, recent textbooks on health physics have all but abandoned the system of dose limitation on which the current Part 20 is based (Cember 1983).

Increasing the Usefulness and Applicability of Dosimetry Data Collected under the Provisions of Part 20. Another possible benefit from bringing current law into agreement with current international standards is the potential for increasing the value of future databases. For example, in meeting reporting requirements to the NRC, licensees do not currently add the risk from internal exposures to the risk from external exposures. Thus, in situations where internal exposures are present, valid measures of risk are not reported and, hence, are not readily available for further research.

In many research endeavors, analysts find occasion to combine data from different sources in order to reach useful conclusions. In combining data, it is important to ensure that the databases are commensurable. This might not be the case if data generated under requirements of the current Part 20 are combined with data generated under ICRP or NCRP recommendations. However, because internal doses are currently rare at most NRC-licensed facilities, and because the revision is based on combining the risks from internal and external doses, it is unlikely that the proceeding the revision will result in a significant benefit in this area.

# 7.2 POTENTIAL BENEFITS FROM REVISED MONITORING REQUIREMENTS

Two possible benefits are identified in connection with the monitoring requirements imposed by the revision Part 20. These are benefits from 1) improving ALARA programs and 2) documenting individual doses that could later be used as supporting documents in litigation.

Improving ALARA Programs. Under the revised Part 20, monitoring will need to be upgraded to provide measurements of both internal and external individual doses that are likely to be in excess of 10 percent of the annual dose limits. By obtaining a more accurate record of the radiological environment within a facility, the facility operator may be able to discover relatively inexpensive (i.e., cost-effective) ways to reduce doses in accordance with ALARA. The extent to which such opportunities currently exist can only be conjectured.

Legal Defense by Documenting Individual Doses. The increased monitoring requirements will make it possible for a licensed firm to maintain better employee records of exposures. One function of an employee record is to reduce the uncertainty about whether a future adverse health effect is the result of past exposures. It may be argued, then that increased monitoring requirements reduce litigation (a societal benefit) by reducing uncertainties about exposures. For instance, if the records show exposures high enough to produce a relatively high probability of carsation (PC), then litigants will have an incentive to settle out of court, thus reducing litigation costs. Alternatively, if the records show that exposures were so low that the health effect was highly unlikely to have been caused by radiation exposures received at the facility, then the would-be plaintiff has little incentive to pursue a lawsuit. In either case, litigation costs are reduced, and this represents a social benefit. The social benefit is measured as the present value savings in litigation costs (attorneys' fees, court costs, etc.) plus the amount of the judgment that compensates for actual harm done to the plaintiff. For example, punitive damages should not be included since they are not a social cost; rather, they are simply a transfer of monetary assets from one party to another.

Further analysis indicates, however, that a real societal benefit is not likely to be found here. The reason for this is that any financially responsible factinty has a sufficient incentive to engage in monitoring for its own legal protection. Since the marginal benefits from increased monitoring will diminish at an increasing rate, there is an optimal level of monitoring which the firm will attempt to discover. The firm's optimizing (profit-maximizing) rule is to add monitoring protection until the marginal cost of the added protection just equals the expected marginal litigation costs. Since the facility is primarily concerned with the private costs rather than the social costs, it will include the cost of expected judgments in its calculation. This means that the marginal private costs will exceed the marginal social costs if the award is in excess of the actual harm sustained by the plaintiff, for example, because of punitive damages. In this case, the unregulated facility will tend to engage in more than the socially optimal level of monitoring, and monitoring requirements beyond this level will further exacerbate the resource misallocation, giving rist to even more excessive net societal costs. As discussed in Section 3.1, mos. licensees currently monitor occurrently doses less than 500 mrem/yr, which is the required level of monitoring under the revised Part 20. Therefore, the potential benefit from reduced litigation is likely to be small.

# 7.3 POTENTIAL BENEFITS FROM REVISED RECORDKEEPING/REPORTING REQUIREMENTS

One of the proposed reporting requirements will require some facilities to provide to the NRC individual exposure reports for all individuals for whom monitoring was required. In addition, all licensees will be required to notify individuals of the doses that they receive. Two possible benefits have been identified as arising from this requirement: 1) developing a database containing records of individual exposures that are readily converted to risk measures, and 2) providing the basis for a registry under which the accumulated doses of transient workers could be routinely monitored.

Developing a Database of Individual Exposure Records. Whether or not an actual benefit exists here depends on what the NRC will do with the individual exposure records it collects from licensees. For a benefit to exist, the records after collection must 1) be beneficial to workers receiving an annual accounting of their dose for the previous calendar year, 2) be processed by the NRC to produce information beneficial to society, 3) be beneficial to NRC licensees, 4) be made available to researchers outside the NRC who will produce information beneficial to society, or 5) be maintained by the NRC in a repository for possible future retrieval for a socially beneficial purpose. The first factor could give rise to a social benefit. Korkers receiving notification of their prior year's dose might behave in either of two ways. Workers receiving relatively large doses might make an extra effort to avoid unnecessary exposures. On the other hand, workers receiving doses well below the publicized limit might adopt a cavalier attitude toward exposures.

Regarding the second factor (NRC-produced information), a significant benefit is unlikely because the current reporting requirements are virtually the same as the revised requirements except that individual rather than summarized information will be sent to the NRC. Although the summarized information leads to overestimates of collective doses (Brooks 1986), statistical corrections could be applied.

A possible benefit was identified with respect to the third factor, the benefits accruing to licensees. A dose history is required if a licensee needs to ensure that a new worker wil! not violate the 5(N-18) formula. Currently, it may be costly to recover the dose history because many licensees currently maintain worker dose records by year rather than by worker. However ne revised Part 20 excludes the 5(N-18) formula, thereby significantly reducing the newd for a worker dose history more than 1 year in the past. If a worker changes jobs in the middle of a calendar year, he may need \* produce his dose record for the new employer so chat his dose for the remainder of the year can be planned. Since the revised Part 20 will result in better tracking of individual dose histories on NRC Form 4, providing the current year dose for transient workers will have very little cost. The benefit is expected to be small, though, because currently most employers experience little difficulty in obtaining a worker's current-year dase record from the worker's previous omployer.

With regard to the fourth factor--making individual dose records available to others outside the NRC--the National Cancer Is titute (NCI) has expressed interest in obtaining the individual dose records from the NRC to perform epidemiological studies. The records would be used to establish dose-response relationships from low levels of ionizing radiation. At least 10 years of data will be needed before statistically significant results can be expected from these studies. After the results of these studies are available, they may provide the basis for significant adjustments to dose limits. Of course, not all of the social benefits from more accurate dose limits can be attributed to the existence of the individual dose database; much of the benefit must be assigned to the epidemiological studies themselves.

S TERS

COLUMN THE

There is a potential benefit from maintaining a historical database in a repository for future retrieval. Currently, little is known about the effect of dose incidence on an individual's accumulated lifetime risk. For example, we might learn that 2 successive years of 5 rem doses pose much greater risk than if a year of low exposure is interposed. Armed with such knowledge, one could then retrieve and review the lifetime histories of current workers in order to modulate their lifetime risk (Newcombe 1980).

Providing the Basis for a Registry to Monitor Transient Workers. As discussed in Section 4.1.1, transient workers at nuclear power plants represent the majority of workers who currently receive doses greater than 5 rem/yr. Because workers will no longer be permitted to receive annual doses greater than 5 rem (except when the planned special exposure provision is used), some transient workers may falsify their accumulated dose to licensees so that they will be more likely to be hired.

Under the revised Part 20, both the revised requirements for determination of prior dose and the individual dose reporting requirements will provide for improved tracking of transient worker doses. This will reduce the likelihood that a worker will illegally receive doses greater than the limits. In terms of collective dose savings, however, no benefit is anticipated.

#### 7.4 POTENTIAL BENEFITS FROM ADDITIONS/REVISIONS TO CURRENT DOGE LIMITS

The proposed revisions to Part 20 contain several adjustments to current dose limits. Some of the adjustments are more restrictive, and in other cases, the limits are less restrictive (see Section 4.0). Where limits have become stricter, nearly all tasks that are currently undertaken would continue to be undertaken under the stricter dose limits. In cases where the dose limits have been relaxed, one would expect that few new tasks would be added; rather, tasks would be performed by fewer workers.

Assuming that the risk from radiation dose is proportional to dose and has no threshold, no benefit in terms of risk would accrue from dose reallocation if the collective dose remains unchanged (Peterson 1984). However, from a social benefit/cost perspective, there may be social benefits and costs. These derive from the fact that spreading of risk is usually socially beneficial, as discussed below.

Benefits from Spreading of Risk. Studies that undertake to put a dollar figure on the value of a human life typically derive the value by taking the dollar amount that is expended to obtain a reduction in risk and multiplying this by the inverse of the risk reduction. For example, if individuals are willing to pay \$10 to reduce a fatality risk from 1.0E-4 to 9.0E-5, then the imputed value of life is derived simply as \$10/(1.GE-4 - 9.OE-5) = \$10/1.OE-5 = \$1.0E6, or one million dollars. The problem with this approach is that it assumes that an individual's willingness to pay to reduce a given increment of risk is independent of the level of risk. Were this independence valid, then we should expect this same individual to play a game of Russian roulette for a certain payment of \$1E6/6 = \$1.6E5. Yet, there is no reason why this individual should feel compelled to play for this amount. Russian roulette represents a much greater risk, and the individual might well require a larger amount, perhaps will in excess of a million dollars. Such behavior could not be termed inconsistent or irrational. It simply implies that riskaversion is not a linear function of the risk level (Weinstein, Shepard and Pliskin 1980; Linnerooth 1979).

If risk aversion increases with the level of risk, then there will be a social benefit from spreading the risk over a greater number of individuals. In order to quantify the benefit, the following information is needed:

1) the level of risk to an individual before risk-spreading, 2) the level of risk to an individual after risk-spreading, and 3) for a representative individual, the functional relationship between willingness-to-pay to reduce risk and the level of risk.

The benefit from risk spreading may need to be adjusted for another effect. If risk is spread, then collective dose may not remain unchanged. To accomplish a task a manager will likely delegate it to the persons who can execute it at lowest cost. If additional workers must then be added (because of dose limitations), then the added workers generally will be less efficient than the original crew. As a result, collective dose may increase (Pelletier and Voilleque 1979). For example, four persons may be able to complete a task with a collective dose of 20 rem, but when five persons are assigned, the collective dose might increase to 23 rem. Collective dose increases although individual doses decrease.

In evaluating the net benefit from the revised dose limits, it is necessary to adjust the benefits from risk-spreading for the "less-criticientworker" effect just described. Considering that the dose distributions to only a small number of workers will be affected by the revised limits and that the anticipated increase in collective dose is small (see Section 4.1.1), any potential net costs or benefits are likely to be small.

In a few instances, dose limits that have been revised do not affect current applications. For example, civilian activities involving Pu are essentially nonexistent at NRC-licensed facilities. In the future, an activity involving Pu might be discovered that is sufficiently valuable to warrant compliance with existing regulations. However, stricter provisions in the new Part 20 might prevent this activity from occurring. In those cases where the requirements have been relaxed, new activities may be viable immediately or some time in the future.

#### 7.5 POTENTIAL BENEFITS FROM SUPPLIED GUIDANCE FOR COMPLYING WITH THE REVISION

The NRC plans to provide some detailed guidance in the form of regulatory guides on how licensees must meet the new Part 20 provisions (see Section 6.3.2). To the extent that the guidance is prescriptive, this will usually make licensee compliance more (socially) costly. It is easy to see why this is so. Consider, for example, a dose limitation with which licensees must comply. Let us assume that there are several different ways of complying with this limitation. Each profit-maximizing licensee will attempt to select the least costly compliance method. Because different licensees operate unde different conditions, there is no reason why the least costly compliance method for one licensee will also be the least costly method for other licensees. Thus, a prescriptive approach to enforcement will generally impose higher social costs than necessary, even if the prescribed method is the method that most licensees deem to be the least costly. On the other hand, the absence of guidance often leads licensees to adopt methods that are unnecessarily costly to ensure compliance with possibly ambiguous regulations.

The benefit from the prescriptive approach is that it will generally make enforcement by the regulating agency easier (i.e., less costly). If compliance with the revision is ensured once well-defined procedures are followed, then the enforcement agency need only check that the licensees have carried out the prescribed steps. This can also improve the consistency with which inspectors enforce the requirements.

# 7.6 POTENTIAL BENEFITS FROM DOSE LIMITS FOR PREGNANT WOMEN AND EMBRYOS/FETUSES

To estimate the benefit from the dose limits for declared pregnant women, it is necessary to obtain an estimate of the number of pregnant women who currently receive doses in excess of the proposed limits. It is then simply a matter of computing the risk to each embryo/fetus to obtain a value for the benefit. This benefit was estimated in Section 4.3.

However, there may also be unintended costs. Licensees, particularly those with small operations, will be reluctant to employ women of childbearing age. Loss of a trained employee for a temporary period could impose a substantial burden on a small firm, which usually does not have the ability to shift personnel around without a significant loss in productivity. However, because of potential legal liability, licensees may already be engaging in defensive hiring practices, in which case the new Part 20 would provide few added costs or benefits. The problems associated with a separate limit for the embryc/fetus are well documented (Taylor 1985) and are not discussed further in this report.

# 7.7 POTENTIAL BENEFITS FROM REVISED LABELING REQUIREMENTS

The proposed revision of Part 20 includes requirements that affect products containing radioactive materials. The levels of activity for which labeling will be required are lower for some radioactive materials and higher for others than under the current Part 20 (see Section 5.2.1). The main purpose of the proposed labeling requirements is to inform consumers of products posing health risks that exceed some determined level. It is noted that the health risk level at which labeling is required will be approximately the same for all radioactive materials because the exempt quantities are determined from occorrect onal ALIs for each radionuclide.

Because most of the revised exempt quantities are either higher by one order of magnitude than the current exempt quantities or are unchanged, few changes with respect to container labeling are anticipated. For those cases where labeling of containers will no longer be required, there are obvious benefits from the reduced number of labels that must be purchased and affixed. An additional benefit is identified for firms that currently spend resources to ensure that containers do not contain enough radioactive material to require labeling. Because the revised exempt quantities are generally higher than the current values, fewer resources will be required to reduce amounts of radioactive material to exempt levels.

# 8.0 SUMMARY OF COSTS AND BENEFITS

Previous sections in this report included analyses of the costs and benefits of the revision of Part 20. Where possible, the costs and benefits were discussed with respect to the relevant sections of the rule and the types of licensees that would be affected. In addition, the costs and benefits that would be incurred once were delineated from those that would be incurred annually.

In the following sections, the costs and benefits are summarized so that meaningful evaluations of the impact of the revision of Part 20 can be made. In Section 8.1, the current number of NRC and Agreement State licensees are summarized by facility category. In Section 8.2, the costs of the revision are summarized by both facility type and Part 20 section. The uncertainties of the cost estimates are also discussed. In Section 8.3, the benefits of the revision are summarized by section of the revision, and the uncertainties and qualitative aspects of the benefits are discussed. In Section 8.4, a sensitivity analysis of the estimated net ben fit is presented. Finally, the conclusions of this analysis are presented in Section 8.5.

#### 8.1 COMPENDIUM OF NRC AND AGREEMENT STATE LICENSEES

In rear to assess the impact of the revision of Part 20 on licensees, it was necessary to group the various types of licensees into categories that reflect the type of activities the licensees are involved in. Because both NRC and the Agreement States classify licensees under many categories, it was necessary to condense the number of categories of licensees for this analysis. The data presented below were derived primarily from a compilation of data presented in four references (Hendrickson et al. 1987; CRCPD 1987; Brooks 1986; Brooks, McDonald and Richardson 1982).

The Conference of Radiation Control Program Directors (CRCPD) periodically publishes a report of the number of Agreement State licenses by state and facility type. The most recent report (CRCPD 1987) was used to determine the number of Agreement State licensees in 1985. To determine this number from the data supplied in the report, several translations were necessary. First, the number of licenses issued by all States was determined for the various license categories listed in the CRCPD report. Second, the number of licenses issued by non-Agreement States was subtracted from the total number of licenses issued for each category of license. Finally, the total number of licenses was determined by dividing the number of licenses by 1.286, which is estimated to be the ratio of total licenses to licensees (Hendrickson et al. 1987). Although this recent was derived from data on NRC licensees, the same ratio was assumed to apply to Agreement State licensees for this report.

1. CONTRACT

Fekst Marsell,

The number of NRC licensees in 1985 was determined from several sources. For facilities other than power reactors and test reactors, and fuel fabrication and processing facilities, 'he number of licensees in each facility category was determined from a 1983 survey by the NRC's Division of Rules and Rocords (DRR). The data from this survey was analyzed and published (Hendrickson et al. 1987). The number of power reactor and fuel fabrication facilities was determined from the annual radiation exposure summary reports submitted to NRC in 1984 (Brooks 1986). The number of research and test reactor licensees was determined from data submitted to NRC for the year 1979 (Brooks, McDonald and Richardson 1982). It was assumed in this report that these data for NRC licensees accurately represent the number of NRC licensees in 1985.

Because the contegories of licensees listed in the various references were numerous and not always consistent, it was necessary for this analysis to condense the number of licensees into expressive categories. Ten general categories of licensees were identified based on the primary activities at the facilities. These categories and the associated categories used by NRC and CRCPD to classify licensees are listed in Table 8.1.

In 1985, there were 27 Agreement States. All licensees in Agreement States except power reactors, research and test reactors, and fuel fabrication and processing facilities are licensed by the States. All licensees not licensed by Agreement States are licensed by the NRC. Table 8.2 lists the number of NRC and Agreement State licensees in 1985 for each facility category. For this report, these data are assumed to represent accurately the number of licensees existing in 1989.

The data in Table 8.2 are generally consistent with the ratio of Agreement States to non-Agreement States. One inconsistency, however, is the number of licensees listed in the "Manufacturing and Distribution" facility category. For this category of licensees, NRC licensees outnumber Agreement State licensees by a factor of two. One reason for this apparent inconsistency is that the CRCPD data for Agreement States (CRCPD 1987) are not classified according to the NRC system of licensee classification. It is likely that some of the licensees listed in the "All Others" category for Agreement States would be listed in the "Manufacturing and Distribution" category under the NRC classification system.

Some of the costs of the revision of Part 20 identified in this analysis are directly related to the number of radiation workers at a facility. In order to assess these costs, it was necessary to estimate the number of employees, the number of employees monitored, are the number of employees reported to have measurable doses for each facility category. For facilities other than power reactors, research and test reactors, and fuel fabrication and processing facilities, the number of employees was determined by multiplying the average number of employees per licensee (Hendrickson et al. 1987) by the number of licensees listed in Table 8.2. The total number of employees monitored and the total number of employees having measurable doses were determined from data reported to NRC (Brooks 1986; Brooks, McDonald and Richardson 1982) indicating average numbers of these employees per licensee. The averages were multiplied by the number of licensees in each category listed in Table 8.2. It was assumed in this report that the most recent available data indicating both the average number of employees monitored per

No. Sec.

Sterit's.

Linger .
## TABLE 8.1. Categories of Licensees

Facility Category	Types of Licensees Included
Medical	Medical institutions (broad and other), medical private practice, teletherapy, cardiac pacemakers, eye applicators, nuclear medicine vans, veterinary, in vitro labs, and other medical
Well Logging	Well logging
Industrial Radiography	One location radiography, multilocation radiography, in-plant radiography, and field radiography
Manufacturing and Distribution	Manufacturing and distribution (broad and other), medical distribution, nuclear parmacies, pacemaker manufar ing and distribution, cover source material, source ial shield's and scorce material general distri
Academic/Research Institutions	Academic institutions (broad and other), and research and development institutions (broad and other)
Other Measuring Systems	Fixed gauge, portable gauge, and other measurement systems
Fuel Fabrication and Processing	Fuel fabrication and processing, UFs conversion and production, uranium mills, and uranium solution mining
Research and Test Reactors	Research reactors, test reactors, and critical experiment facilities
Commercial Power Reactors	Light water reactors, gas cooled reactors
All Others	All licensees licensed under 'Other Special Nuclear Material' codes (except pacemaker manufacturing and distribution), nuclear laundry, leak test service, irradiators, byproduct power sources, waste disposal, waste services, civil defense, and others

licensee and the average number of employees having measurable doses per licensee are applicable to the year 1985. Table 8.3 lists employee data for NRC and Agreement State licensees.

8.3

NRC and	Agreement State Licensee Number of Licensees	s in 1985
NRC	Agreement State	Total
2,432	4,074	6,506
130	324	454 ,
348	503	851
637	328	965
769	787	1,556
	. 2,824	5,060
14	0	14
80	0	80
88	a) 0	88 88
478	1,274	1,752
7,212	10,114	17,326
	NRC and A NRC 2,432 130 348 637 769 -, 14 80 88 478 7,212	NRC and Agreement State Licensees         NRC       Agreement State         2,432       4,074         130       324         348       503         637       328         769       787          2,824         14       0         80       0         88 <sup>(a)</sup> 0         478       1,274         7,212       10,114

(a) 109 reactors are assumed for the analysis in this report based on recent estimates for the year 1989 (ANS 1988).

## 8.2 SUMMURY OF COSTS

3

This section summarizes the costs identified in this report. The costs are summarized by facility type and Part 20 section, followed by a discussion of the uncertainties inherent in these estimates.

In this report, present values are calculated based on 1989 dollars, a discount rate of 10%, and a 30-year period. Because the NRC will grant a 5-year implementation period retroactive to January 9, 1986, development and implementation costs identified in this report will be distributed over the years 1986 to 1990. For this report, it was assumed that 5%, 5%, 10%, 40% and 40% of the development and implementation costs identified were or will be incurred in 1986, 1937, 1988, 1989 and 1990, respectively. In effect,

TABLE 8.3. Employee Data for NRC and Agreement State Licensees in 1985

Facility Category	Average No. Employees per Licensee	Total No. Employees	Total No. Employees Monitored	Total No. Employees Having Measurable Dose
Medical	460	3,000,000	193,000	109,000
Well Logging	170	77,100	39,300	37,000
Industrial Radiography	554	471,000	19,900	12,900
Manufacturing and Distribution	422	407,000	127,000	49,000
Academic/Research Institutions	445	692,000	93,700	27,900
Other Measuring Systems	292	1,480,000	38,000	13,700
Fuel Fabrication and Processing	679 <sup>(à)</sup>	9,500 <sup>(a)</sup>	9,500	5,800
Research and Test Reactors	45(a)	3,600 <sup>(a)</sup>	3,500	1,000
Commercial Power Reactors	1,920 <sup>(a)</sup>	169,000 <sup>(a)</sup>	169,000 <sup>(b</sup>	) 95,000 <sup>(b)</sup>
All Others	294	515,000	43,800	14,200
All Facilities	394	6,824,200	736,800	365,500

(a) Inferred from the number of workers monitored, assuming 100% of the workers at the facilities were monitored.

(b) A 24% increase in these numbers is assumed for the year 1989 based on recent estimates for the number of reactors operating in that year.

al the

i shiriya

development and implementation costs were multiplied by 0.91(a) to determine their present value (Heaberlin et al. 1983). The one exception is MRC costs to develop regulatory guides because it is expected that some of the guides will be developed after 1990. Appropriate corrections to the present value calculations were made for those costs.

(a)  $(0.05/1.1^{-2}) + (0.05/1.1^{-1}) + (0.1/1.1^{0}) + (0.4/1.1^{1}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4/1.1^{2}) + (0.4$ 0.91.

Operation costs will be incurred annually once the revision is fully implemented. For this analysis, present values were calculated based on a 30-year period beginning in 1989. Present values of operation costs were calculated by multiplying the annual operation costs by 8.60 (Heaberlin et al. 1983). This multiplication factor includes consideration of partial realization of operation costs from 1986 to 1990 (see above paragraph).

## 8.2.1 Summary By Facility Type

In Section 8.1, ten categories of licensee facilities were defined for this analysis. In Table 8.4, the costs identified in this report are summarized by facility type. The costs incurred by NRC are also included in Table 8.4. All costs estimates are based on 1989 dollars.

## 8.2.2 Summary By Part 20 Section

In this section, the costs of the revision are summarized by section of the rule. In some cases, costs are attributable to two or more related sections. For example, costs related to increased extremity monitoring are attributable to both the reduced extremity limits and the reduced fraction of the dose limit that requires monitoring. In these cases, the costs were assumed to be evenly divided among the relevant sections.

Some costs are either not attributable to a specific section of the rule or are attributable to many sections. For example, costs related to personnel training cannot be readily associated with a specific section of the rule. In these cases, the costs are summarized by cost description.

Table 8.5 summarizes the costs of the revised rule (in 1989 dollars) by section of the rule. Sections not listed in the table were not identified to have significant associated costs.

## 8.2.3 Discussion of Uncertainties

Each of the individual estimates used to develop the overall estimates provided in Tables 8.4 and 8.5 have an associated uncertainty; therefore, the overall estimates have an associated uncertainty as well.

The single most important cost estimate for this analysis was the estimated cost of required modifications at fuel fabrication facilities. Costs of approximately \$10 million initially and \$2 million annually per facility have been estimated in the past. However, these estimates are thought to be maximum estimates, i.e., worst case scenarios. Consideration must be given to the possibility that particle size studies, use of respirators, etc., may be used in place of facility modifications. The \$75 million cost estimate for fuel fabrication facilities could be a factor of five too high or a factor of two too low, depending on the steps that these facilities will actually take in response to the revised dose evaluation requirements.

Other assumptions necessary for this analysis were uncertain as well. For example, it was assumed throughout this report that Agreement State licensees will be subject to the same requirements as NRC licensees,

# TABLE 8.4. Quantified Costs Incurred from the Revision of Part 20 by Facility Category

Facility Category	Implementation Costs, \$	Operation Costs, \$/yr	Present Value of Costs, \$(a)
Medical	7,400,000	1,800,000	-22,000,000
Well Logging	12,000	44,000	-390,000
Industrial Radiography	290,000	22,000	-450,000
Manufacturing and Distribution	530,000	240,000	-2,600,000
Academic/Research Institutions	1,800,000	340,000	-4,500,000
Other Measuring Systems	310,000	43,000	-650,000
Fuel Fabrication and Processing	33,000,000	5,300,000	-75,000,000
Research and Test Reactors	650,000	4,000	-620,000
Commercial Power Reactors	30,000,000	4,000,000	-61,000,000
All Others	81,030	49,000	-500,000
NRC	1,400,000	210,000	-2,900,000
Totals	75,000,000	12,000,000	-170,000,000

制制的

200 F. Baller

POD

(a) The present values are based on 1989 dollars, a discount rate of 10%, and a 30-year period. A negative sign indicates a negative impact.

including the time allowed for implementation of the requirements. It is likely that Agreement State licensees will lag behind NRC licensees because Agreement State agencies will need time to evaluate the revised Part 20 and develop appropriate state regulations. If in fact Agreement State licensees take an average of 4 years longer to implement the provisions of the revised t 20, the present value of the costs for Agreement State licensees would bout 30% lower than calculated in this report. <u>TABLE 8.5</u>. Quantified Costs Incurred from the Revision of Part 20 by Section of the Rule

Section	Description	of Costs, \$(a)
20.201	Occupational dose limits for adults	-12,000,000
20.202, 20.204	Summation of internal and external doses	-96,000,000
20.208	Dose limit for embryo/fetus	-5,200,000
20.502	Conditions requiring individual ponitoring	-1,400,000
20.1106	Records of individual monitoring results	-20,000,000
20.1107	Records of dose to the public	-3,900,000
20.1206	Reports of personnel monitoring	-11,000,000(b)
NA(C)	Personnel training	-10,000,000
NA	Procedure revisions	-10,000,000
ы.). Э	NRC inspections/training	-1,100,000
Total		-170,000,000

(a) A negative sign indicates a negative impact.

(b) 84% of the costs are associated with the revised Part 19 requirements

to provide reports to individuals of the doses they received.

(c) Not applicable.

Another assumption inherent in this analysis was that all identified costs must be considred as actual costs. Because many of the costs will not result in an increased use of resources, this assumption may overestimate the actual costs associated with the revision. For example, if a health physicist's hourly wage is \$25 per hour (including overhead) and he must spend 8 hours revising procedures in response to the revised Part 20, he will most likely not work overtime to accomplish this task. Rather, he will likely omit performing another task of less importance. Because the most important tasks the health physicist performs would likely be worth more than \$25 per hour and the least important tasks would be worth less than \$25 per hour, the associated cost of revising the procedures would be less than 8 X \$25. Because many of the costs identified in this report may fall under this category, the overall cost estimate may misrepresent the actual costs that will be incurred.

There was one major cost identified in this report that was not quantified: the cost associated with the revised concentration limits in Appendix B for releases into sewers. These costs could be significant with respect to the overall cost estimates provided in this report.

#### 8.3 SUMMARY OF BENEFITS

In this section, the benefits of the revision of Part 20 are summarized. For those cases where the benefits were quantified in terms of cost savings, the present values were calculated using the same methods and assumptions used to calculate the present values of the annual costs (see Section 8,2). For those cases where the benefits were quantified in terms of dose reductions, present values were calculated based on a value of \$1000 per personrem (Heaberlin et al. 1983) and \$10,000 per fetus-rem. The latter value was arbitrarily chosen based on the ratio of the embryo/fetus limit to the occupational dose limit for adults. Health effects were discounted in calculation of present values in consideration of the arguments that favor discounting of future radiation effects (Cohen 1983; Nieves et al. 1983). The dependence of the present value calculations on variations in these assumptions is discussed in Section 8.3.2.

#### 8.3.1 Summary by Part 20 Section

In Table 8.6, the benefits of the revision are summarized by section of the rule. Some of the benefits are not attributable to a specific section of the rule and are therefore summarized by description only.

#### 8.3.2 Discussion of Uncertainties

The estimated benefit of \$44 million provided in Table 8.6 did not include consideration of many benefits that could not be quantified. It is likely that the actual value of the benefits of the revision, if such a value could be calculated, would be substantially higher than \$44 million. The actual value depends primarily on the actual or perceived importance of consistency with ICRP/NCRP recommendations and the importance of complying with the Federal Guidance. Other benefits that were not quantified are thought to be of less importance.

On the other hand, the assumptions used to determine the present value of the benefits may have resulted in overestimates of the actual benefits. For example, the value assigned to a rem to the embryo/fetus (\$10,000) was based on the assumption that a rem to the embryo/fetus is ten times as detrimental as a rem to an adult. If this assumption is valid, then it would be hard to justify any dose to pregnant women that could be avoided by job rotation practices; current practices in industry suggest that this may not be the case. Also, the value of \$1000 assigned to a person-rem is probably too high based on the actual risk of harm from radiation as compared to the risk of harm from other hazards. However, some believe that health effects should not be discounted to determine present value; at \$1000 per person-rem and \$10,000 per fetus-rem, the annual dose reductions identified in this report would be valued at over \$4 million dollars per year for 30 years, as opposed to a present value of \$36 million as calculated in this report based on a 10%

· .....

15000

- Anther

Section	Description of Benefit	Annual Benefit	Present Value of Benefit, \$(a)
20.202	Decreased collective doses	540 rem	+4,600,000
20.208	Reduced doses to the unborn	300 rem	+26,000,000
20.502	Increased knowledge of work environment Reduced litigation costs	NQ(b) NQ	NQ
20.702	Decreased collective doses Reduced operating costs	610 rem \$880,000	+5,300,000 +7,600,000
20.905	Reduced operating costs	NQ	NQ
20.1206	More complete data base	NQ	NQ
NA(c)	Consistency with ICRP/NCRP Recommendations	NQ	NQ
NA	Consistency with Federal Guidance	NQ	NQ
Total			+44,000,000(d)

TABLE 8.6. Benefits of the Revision of Part 20 by Section of the Rule

(a) A positive sign indicates a positive impact.

(b) Not quantified.

(c) Not applicable.

(d) Does not include unquantified benefits.

In consideration of the many uncertainties associated with the benefits from the revised Part 20, it is likely that the estimated \$44 million (present value) is too low. The information provided in this report should be of help in determining the degree to which the quantified benefits underestimate the overall benefit from the revision. State:

256

## 8.4 SENSITIVITY ANALYSIS

Many of the estimates provided in this report were uncertain. In most cases, however, minor variations in the assumptions used to derive the estimates would not affect significantly the conclusions of this report. Probably the most important assumptions were those required to determine the present values of the costs and benefits. The following variations were used to determine the dependence of the calculated net benefits on the basic assumptions:

- O) No variation from the basic assumptions (10% discount rate, all identified costs and benefits discounted, discount over a 30-year period, \$1000/person-rem, \$10,000/fetus-rem);
- A) Health effects not discounted;
- B) Health effects evaluated at \$100/person-rem and \$1000/fetus-rem;
- C) 5% discount rate;
- D) Lower-bound estimates of marginal costs incurred by licensees based on optimal compliance, i.e., only those costs thought to be necessary for compliance were included in the evaluation.

Table 8.7 lists the calculated net benefits for each facility category based on the variations listed above.

#### 8.5 CONCLUSIONS

Perhaps the most useful method for presenting the costs and benefits identified in this report is to present the net impact per monitored worker by facility category (Table 8.8). This allows one to approximately determine the magnitude of the impacts versus the magnitude of a licensee's radiation protection program.

It is apparent that fuel fabrication facilities will incur by far the highest negative impact per employee monitored. The negative impact of \$71 million listed in Table 8.8 includes about \$76 million in costs and \$5 million in dose savings (540 rea/yr at \$1000 per person-rem, discourted at 10%). Of course, there are great uncertainties in these estimates, the greatest of which is the uncertainty whether massive engineering modifications will indeed be necessary at these facilities.

The overall impact on medical facilities was determined to be positive because the anticipated dose savings to the embryo/fetus was estimated to outweigh the costs, even though medical facilities will incur an estimated \$22 million in costs. However, this conclusion relies heavily on both the estimated dose savings and the cost equated to a rem to the embryo/fetus; there were great uncertainties in both of these estimates.

Considering the best estimates of the quantified costs and benefits and the associated uncertainties, it is unlikely that the benefits associated with the revision of 10 CFR Part 20 will outweigh the costs. However, some of the costs and many of the benefits of the revision were not quantified; the unquantified benefits might have a high enough value to result in a favorable benefit/cost ratio. Because the most important unquantified benefit appears to be that the revision will be consistent with international (ICRP) and national (NCRP and EPA) recommendations, careful evaluation of this benefit is most important in determining whether the revision of Part 20 is acceptable from a benefit/cost standpoint.

	Present Va	lue of Net	Benefit (\$	K) for Varia	ation(a)
Facility	0	A	B	C	D
Medical	+1,400(b)	+58,000	-20,000	+5,800	+5,400
Well Logging	-390	-390	-390	-650	-390
Industrial Radiography	-450	-450	-450	-590	-390
Manufacturing and Distribution	-2,600	-2,600	-2,600	-4,000	-2,500
Academic/Research Institutions	-4,500	-4,500	-4,500	-6,600	-3,300
Other Measuring Systems	-650	-650	-650	-920	- 580
Fuel Fabrication and Processing	-71,000	-59,000	-75,000	-100,000	-36,000
Research and Test Reactors	-620	-620	-620	-680	-470
Commercial Power Reactors	-46,000	-26,000	-53,000	-61,000	-34,000
All Others	-500	-500	-500	-800	-480
Total (c)	-125,000	-37,000	-158,000	-169.000	-73.000

TABLE 8.7. Sensitivity of Net Benefits to Variations in Basic Assumptions

(a) See text for explanation of variations.
(b) A positive sign indicates a positive impact and a negative sign indicates a negative impact.
(c) Does not include costs that will be incurred by the NRC.

State.

TABLE 8.8. Estimated Net Impact (Present Value) of the Revision of Part 20

0

C. P. S. P.

States .

The first

-

Facility Category	No. of Employees Monitored	Net Present Value of Impact, \$(a)	Net Impact per Employee Monitored, \$(a)
Medical	193,000	+1,400,000	+7.3
Well Logging	39,300	-390,000	-9.9
Industrial Radiography	19,900	~450,000	-23
Manuf <sup>*</sup> cturing and Distribution	127,000	-2,600,000	-20
Academic/Research Institutions	93,700	-4,500;000	-48
Other Measuring Systems	38,000	-650,000	-17
Fuel Fabrication and Processing	9,500	-71,000,000	-7,500
Research and Test Reactors	3,600	-620,000	-170
Commercial Power Reactors	209,000	-46,000,000	-220
All Others	43,800	-500,000	1
Total	776,800	-125,000,000(b)	-161

(a) A positive sign indicates a positive impact and a negative sign indicates a negative impact.
(b) Does not include costs that will be incurred by the NRC.



## 9.0 REFERENCES

- American Nuclear Society (ANS). 1988. "World List of Nuclear Power Plants." Nuclear News 31(2):63-82.
- Booth, L. F., F. L. Bronson, and D. W. Groth. 1985. Dosimetry and Recordkeeping Implications of the Proposed Revisions to 10 CFR 20. AIF/NESP-030, Atomic Industrial Forum, Bethesda, Maryland.
- Brodsky, A. 1982. Principles and Practices for Keeping Occupational Radiation Exposures at Medical Institutions As Low As Reasonably Achievable. NUREG-0267, Rev. 1, U.S. Nuclear Regulatory Commission, Washington, D.C.
- Prooks, B. G. 1986. Occupational Radiation Exposure at Commercial Nuclear Power Reactors And Other Facilities, 1984, Seventeenth Annual Report. NUREG-0713, Vol. 6, U.S. Nuclear Regulatory Commission, Sashington, D.C.
- Brooks, B., S. McDonald, and E. Richardson. 1982. Occupational Radiation Exposure, Twelfth Annual Report, 1979. NUREG-0714, Vol. 1, U.S. Nuclear Regulatory Commission, Washington, D.C.
- Bunker, A. S. 1985. 'Dry Active Waste Management at Nuclear Power Stations." Radiation Protection Management 2(3):23-32.
- Cardarelli, R., M. J. Slobodien, K. L. Mossman, A. Brodsky, and J. L. Telford. 1986. "The Effect of Respiratory Protection on Worker Efficiency - ALARA Considerations." Paper presented at the 31st Annual Meeting of the Health Physics Society, June 29-July 3, 1986, Pittsburgh Pennsylvania.
- Cember, H. 1983. Introduction to Health Physics. 2nd ed. Pergamon Press, New York.
- Cohen, B. L. 1983. "Discounting in Assessment of Future Radiation Effects." Kealth Physics 45:687-697.
- Conference of Radiation Control Program Directors (CRCPD). 1987. "Profile of State and Local Radiation Control Programs in the United States for Fiscal Year 1985." <u>Conference Publication 87-3</u>, Conference of Radiation Control Program Directors, Inc., Frankfurt, Kentucky.
- Cook, J. R. 1981. <u>A Survey of Radioactive Effluent Releases from Byproduct</u> <u>Material Facilities</u>. NUREG-0819, U.S. Nuclear Regulatory Commission, Washington, D.C.
- Dionne, B. J., and J. W. Baum. 1985. Occupational Dose Reduction and ALARA at Nuclear Power Plants: Study on High-Dose Jobs, Radwaste Handling, and ALARA Incentives. NUREG/CR-42.4, U.S. Nuclear Regulatory Commission, Washington, D.C.

Environmental Protection Agency (EPA). 1983a. <u>Analysis of Costs for</u> <u>Compliance with rederal Radiation Protection Guidance for Occupational</u> <u>Exposure. Volume I: Cost of Compliance with Proposed Radiation Protection</u> <u>Guidance for Workers. EPA/520/1-83-013-1, U.S. Environmental Protection</u> <u>Agency, Washington, D.C.</u>

112003-5-

All and

- STER

- Environmental Protection Agency (EPA). 1983b. <u>Analysis of Costs for (Compliance with Federal Radiation Protection Guidence for Occupational Exposure. Volume II: Case Study Analysis of the Impacts of Proposed Radiation Protection Guidance for Workers. EPA/520/1-83-013-2, U.S. Environmental Protection Agency. Washington, D.C.</u>
- Environmental Radiation Protection Standards for Nuclear Power Plant Operations, 40 C.F.R. Part 190, Section 190.1 (1986). (Cited in text as EPA 1986a).
- Federal Register (FR). 1987. "Radiation Protection Guidance to Federal Agencies for Occupational Exposure. Recommendations Approved by the President." Vol. 52, No. 17, January 27, 1987, U.S. Environmental Protection Agency, Washington, D.C.
- Federal Register (FR). 1986. "Part II Nuclear Regulatory Commission. 10 CFR Parts 19 et al. Standards for Protection Against Radiation; Proposed Rule; Extension of Comment Period and Republication." Vol. 51, No. 6, January 9, 1986, U.S. Nuclear Regulatory Commission, Washington, D.C.
- Gitterman, M., and E. W. Hebster. 1984. "Shielding Hospital Rooms for Brachytherapy Patiencs: Design, Regulatory and Cost/Benefit Factors." Health Physics 46:617-625.
- Hageman, J. P., J. M. Artz, and G. B. Humphress. 1982. Study of a <u>Recordkceping System for In-Processing of Transient Workers at Nuclear</u> <u>Power Plants</u>. AIF/NESP-025, Atomic Industrial Forum, Bethesda, Maryland.
- Harty, R., W. D. Reece, and J. A. MacLellan. 1986. Extremity Dosimetry at U.S. Department of Energy Facilities. PNL-5831, Pacific Northwest Laboratory, Richland, Washington.
- Heaberlin, S. W., J. B. Burnham, R. H. V. Gallucci, M. F. Mullen, R. J. Nesse, L. A. Nieves, J. J. Tawil, M. B. Triplett, S. A. Weakley, and A. R. Wusterbarth. 1983. <u>A Handbook for Value-Impact Assessment</u>. NUREG/CR-3568, U.S. Nuclear Regulatory Commission, Washington, D.C.
- Healy, J. W. 1982. "The ICRP Dose Limitation System Solution or Problem?" Health Physics 42:407-413.
- Hendrickson, P. L., M. J. Scott, M. F. Mullen, A. K. Nicholls, and S. A. Smith. 1987. <u>Impact of Proposed Financial Assurance Requirements</u> on Nuclear Materials Licensees. NUREG/CR-4958, U.S. Nuclear Regulatory Commission, Washington, D.C.

- Hendrixson, E. S., D. C. Wagner, and L. Morris. 1986. "Containment Airborne I-131 Control Strates: Radiation Protection Management 3(3):51-59.
- Huggins, T. R., and J. "tson, Jr. 1984. "Routine Hand Doses vs. Monitoring Regulations at a Nuclear Power Plant." Health Physics 46:775-781.
- International Commission on Radiological Protection (ICRP). 1959. <u>Recom-</u> mendations of the International Commission on Radiological Protection. ICRP Publication 2, Pergamon Press, Oxford.
- International Commission on Radiological Protection (ICRP). 1977. <u>Recom-</u> mendations of the International Commission on Radiological Protection. ICRP Publication 26, Pergamon Press, Oxford.
- Tational Commission on Radiological Protection (ICRP). 1979. Limits
   Intakes of Radionuclides by Workers. ICRP Publication 30, Pergamon
   S, Oxford.
- International Commission on Radiological Protection (ICRP). 1982. <u>General</u> <u>Principles of Monitoring for Radiation Protection of Workers</u>. ICRP Publication 35, Pergamon Press, Oxford.
- Kelsey, C. A., R. G. Lane, and F. A. Mettler. 1984. "Comments on 'Proposed Criteria for Issuing Personnel Dosimeters'." Health Physics 47:649-650.
- Kemper, I. S., E. J. Kohlerand, and R. J. Scholz. 1984. "One Utility's Low-Level Waste Management Approach to Being in a Region Without a Burial Ground." ANS Transactions 47:76.
- Kumazawa, S., D. R. Nelson, and A. C. B. Richardson. 1984. <u>Occupational</u> Exposure to locizing Radiation in the United States. A Comprehensive Review for the Year 1980 and a Summary of Trends for the Years 1960-1985. EPA 520/1-84-0(5, U.S. Environmental Protection Agency, Washington, D.C.

The street

- Alleran

No.

- Lawrence, M. F., J. Skolnik, S. C. Cohen, and D. J. Goldin. 1984. Characterization of the Temporary Radiation Work Force at U.S. Nuclear Power Plants. AIF/IESP-028, Atomic Industrial Forum, Bethesda, Maryland.
- Lessard, E. T., X. Yihua, K. W. Skrable, G. E. Chabot, C. S. French, T. R. Labone, J. R. Johnson, D. R. Fisher R. Belanger, and J. L. Lipsztein. 1987. <u>Interpretation of Bioassay Measurements</u>. NUREG/CR-4884, U.S. Nuclear Regulatory Commission, Washington, D.C.
- Linnerooth, J. 1979. "The Value of Kuman Life: A Review of the Model." Economic Inquiry 17(1):52-74.
- Low, S. A., and L. R. McPheters. 1983. "Wage Differentials and Risk of Death: An Empirical Analysis." <u>Economic Inquiry</u> 21(2):271-280.
- Masse, F. 1984. "Low-Level Waste Management Practices in Academic and Medical Institutions." ANS Transactions 47:75.

- McGuire, E. L., M. L. Baker, and J. F. Vandergrift. 1983. "Evaluation of Radiation Exposures to Personnel in Fluoroscopic X-ray Facilities." <u>Health Physics</u> 45:975-980.
- Merwin, S. E., R. C. Brown, and J. B. Martin. 1987. "Optimization and ALARA Considerations in the Use of Respiratory Protection in Radia... 'reas." Paper presented at the 32nd Annual Meeting of the Health Physics Society, July 5-10, 1987, Salt Lake City, Utah.
- National Council on Radiation Protection and Measurements (NCRP). 1987. <u>Recommendations on Limits for Exposure to Ionizing Radiation</u>. NCRP Report No. 91, National Council on Radiation Protection and Measurements, Bethesda, Maryland.
- National Council on Radiation Protection and Measurements (NCRP). 1977. <u>Review of the NCRP Radiation Dose Limit for Embryo and Fetus in</u> <u>Occupationally Exposed Women</u>. NCRP Report No. 53, National Council on Radiation Protection and Measurements, Bethesda, Maryland.
- National Emission Standards for Hazardous Air Pollutants, 40 C.F.R. Part 61, Section 61.102 (1986). (Cited in text as EPA 1986b).
- National Environmental Studies Project (NESP). 1980. <u>Study of the Effects</u> of <u>Reduced Occupational Radiation Exposure Limits on the Nuclear Power</u> Industry. AIF/NESP-017, Atomic Industrial Forum, Bethesda, Maryland.
- Newcombe, H. B. 1980. "Design and Future Uses of National Dose Registers for Regulatory Control and Epidemiology." Health Physics 39:783-796.
- Nieves L. A., J. W. Curry, L. J. Hood, and T. M. Tierney, Jr. 1983. <u>Estimating the Economic Cost of Radiation-Induced Health Effects</u>. PNL-4664, Pacific Northwest Laboratory, Richland, Washington.
- Nilson, R., and C. W. Malody. 1982. "Radiation Protection Standards in Nuclear Fuel Manufacturing." Paper presented at the Conference on Radiation Issues for the Nuclear Industry, October 3-6, 1982, New Orleans, Louisiana.

and the second

- 10 14

出现的问题。

Contra-

- Palmer, H. E., C. P. Brim, G. A. Rieksts, and M. C. Rhoads. 1987. <u>Hanford</u> <u>Whole Body Counting Manual</u>. PNL-6198, Pacific Northwest Laboratory, R<sup>2</sup> Jand, Washington.
- Pace, Z. T. 1986. "INPO's Impact in the USA." International Atomic Energy Agency Bulletin 28(3).
- Pelletier, C. A., and P. G. Voilleque. 1979. <u>Potential Benefits of Reducing</u> <u>Occupational Radiation Exposure</u>. AIF/NESP-010R, Atomic Industrial Forum, Bethesda, Maryland.
- Peterson, Jr., H. T. 1984. "Regulatory Implications of Radiation Dose-Effect Relationships." <u>Health Physics</u> 47:345-359.

- Robinson, A. V., D. R. Fisher, W. D. Reece, and J. A. MacLellan. 1986. <u>Performance Testing of Radiobioassay Laboratories: In-Vivo Measurements,</u> <u>Pilot Study Report</u>. PNL-5840, Pacific Northwest Laboratory, Richland, Washington.
- Rutland, L., and J. M. Tuohy. 1984. "Some Effects of NRC Guidance on Design and Operation of an On-Site Radwaste Storage Facility." <u>ANS Transactions</u> 46:147.
- Ryan, M. L. 1985. "Average Exposures for U.S. Nuclear Plant Workers Hit Record Low in 1984." Inside N.R.C. 7(14).
- Ryan. M. L. 1986. "INPO Says Utility Worker Radiation Doses Cut 20% in 1985." Nucleonics Week 27(13).
- Schlenker, R. A. 1986. "Comparison of Intake and Committed Dose Equivalent Permitted by Radiation Protection Systems Based on Annual Dose Equivalent and Committed Dose Equivalent for a Nuclide of Intermediate Effective Half-Life." Health Physics 51:207-213.
- Science and Figineering Associates, Inc. (SEA). 1986. <u>Generic Cost</u> <u>Estimates</u>. <u>Abstracts From Generic Studies for Use in Preparing Regulatory</u> <u>Impact Analyses</u>. NUREG/CR-4627, U.S Nuclear Regulatory Commission, Washington, D.C.
- Skrable, K. W., G. E. Chabot, E. L. Alexander, and C. S. French. 1985. "Committed vs Annual Dose Limits for Internal Radiation Protection and Control." <u>Health Physics</u> 48:123.
- Taylor, L. S. 1985. "The Prob es of Radiation Double Standards: Exposure of Potentially Pregnant Persons " Health Physics 49:1043-1052.

Part Barry

1500

-LANSAGERS-

- ANTA

ANNINGS-

Strengther

- Thind, K. S. 1987. "Extremity Dose: Its Definition, Standards and Regulatory Limits, Radiobiological Significance, Measurement and Practical Considerations." Health Physics 52:695-705.
- Thomadson, B., J. van de Geijn, D. Buchler, and B. Paliwal. 1983. "Fortification of Existing Rooms Used for Brachytherapy Patients." <u>Health Physics</u> 45:607-615.
- Tichler, J., and K. Norden. 1986. <u>Radioactive Materials Released from</u> <u>Nuclear Power Plants</u>. NUREG/CR-2907, U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC). 1975. "Instruction Concerning Prenatal Radiation Exposure." <u>Regulatory Guide 8.13, Revision 1</u>. U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC). 1981. "Storage of Low-Level Radioactive Wastes at Power Reactor Sites." <u>Generic Letter 81-38</u>, November 10, 1981, U.S. Nuclear Regulatory Commission, Washington, D.C.

- U.S. Nuclear Regulatory Commission (USNRC). 1982. Enclosure 2 Regulatory Analysis. U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC). 1983. "Dose Assignment for Workers in Non-Uniform Fadiation Fields." Office of Inspection and Enforcement. <u>I.E. Information Notice No. 83-59</u>, U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC). 1984. Regulatory Analysis Guidelines of the U.S. Nuclear Regulatory Commission. NUREG/BR-0058, U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC), 1986. The Identification and Estimation of the Cost of Required Procedural Changes at Nuclear Power Plants. U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC). 1987a. "Instruction Concerning Prenatal Radiation Exposure." <u>Regulatory Guide 8.13, Revision 2</u>. U.S. Nuclear Regulatory Commission, Washington, D.C.
- U.S. Nuclear Regulatory Commission (USNRC). 1987b. "Occupational Exposures at Power Reactors Reach All-Time Low in 1986." <u>Highlights of Weekly</u> <u>Information Report</u>, Week Ending July 31, 1987. U.S. Nuclear Regulatory Commission, Washington, D.C.
- Usui, N. 1987. "Japan to Tighten Radiation Exposure Rules to Meet ICRP Standards." Nucleonics Week 28(50), December 10, 1987.
- Vallario, R. W., P. M. Lewis, M. F. Mullen, and S. C. Vickroy. 1985. <u>Regulatory Analysis of Recommendations for NRC Policy on Shift Scheduling</u> <u>and Overtime</u>. PNL-5593, Pacific Northwest Laboratory, Richland, Washington.
- Weinstein, M. C., D. S. Shepard, and J. S. Pliskin. 1980. "The Economic Value of Changing Mortality Probabilities: A Decision-Theoretic Approach." <u>Quarterly Journal of Economics</u> 94(2):373-396.
- West, C. M., L. M. Scott, and N. B. Schultz. 1979. "Sixteen Years of Uranium Personnel Monitoring - in Retrospect." <u>Health Physics</u> 36:665-669.

A STATISTICS

Wiatrowski, W. A., E. P. Cooke, D. T. Kopp, and D. W. Jordan. 1984. "Radiocontamination in Medical Centers from Diagnostic Nuclear Medicine Procedures." <u>Health Physics</u> 47:297-298.

Section 1

with a faith

A STATE AND

Att.

the .

The residence of the second second

#### **DISTRIBUTION**

No. of No. of Copies Copies ONSITE OFFSITE 50 Pacific Northwest Laboratory 12 H. T. Peterson, Jr. Radiation Protection and Health R. C. Adams Effects Branch W. J. Bair Division of Regulatory L. G. Faust Applications D. R. Fisher Office of Nuclear Regulatory C. A. Geffen Research J. R. Johnson U.S. Nuclear Regulatory J. A. MacLellan Commission J. B. Martin Washington, DC 20555 S. E. Merwin (24) M. F. Mullen (5) A. J. DiPalo J. M. Selby Regulation Development Branch K. L. Soldat Division of Regulatory K. L. Swinth Applications J. J. Ta: 11 Office of Nuclear Regulatory R. J. Traub M. G. Woodruff Research U.S. Nuclear Regulatory Health Physics Department Commission Library Washington, DC 20555 Publishing Coordination Technical Report Files (5) B. J. Richter Regulation Development Branch Division of Regulatory Applications Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission

Washington, DC 20555

ENCLOSURE D

FINAL ENVIRONMENTAL ASSESSMENT

Enclosure D

## ENVIRONMENTAL ASSESSMENT FOR THE REVISION OF 10 CFR FART 20 STANDARDS FOR PROTECTION AGAINST RADIATION

#### I. Introduction

The Nuclear Regulatory Commission (NRC) is revising its regulations in Part 20 of Title 10 of the Code of Federal Regulations (10 CFR Part 20). Because the regulations contain basic standards for protection against radiation, the proposed revision would affect all categories of NRC licensees. The intent of the revision is to improve NRC standards for protection against radiation by reflecting developments in the principles that underlie radiation protection and advances in related sciences that have occurred since the original promulgation of 10 CFR Part 20 over thircy years ago. The expected result of promulgating and implementing the revision is an improved rule that provides better assurance of protection of individuals, establishes a clear health protection basis for dose limits, applies to all licensees in a consistent manner, and reflects current information on health risk, dosimetry, and radiation protection practices and experiences.

The impact of this change in NRC's regulations on the environment could potentially arise in three areas, i.e., dose limits for members of the public, dose limits for radiation workers, and effluents from licensed facilities. The potential impact in each of these areas is discussed below.

#### II. The Need for the Proposed Action

The existing 10 CFR Part 20, "Standards for Protection Against Radiation," was developed and published in the late 1950s by the NRC's predecessor, the Atomic Energy Commission (AEC). These standards were based on

1

knowledge of radiation protection theory and practice developed through the 1950s. Since the initial publication of 10 CFR Part 20, there have been a number of revisions designed to bring the regulations into accordance with recommendations of scientific organizations having expertise in radiation protection and biological effects of ionizing radiation. However, the basic approach to radiation protection has been retained throughout these revisions--an approach which derives limits by implicit judgments on safety.

The principal concerns with the present 10 CFR Part 20 regulations and the benefits resulting from the revisions intended to address these concerns are outlined in Table 1.

## III. Potential Environmental Impact

10 CFR Part 20 establishes requirements for the protection of individuals from ionizing radiation resulting from operations of NRC-licensed tacilities. This overall goal is accomplished by regulations addressing three major topics, i.e., doses to members of the general public resulting from NRC licensed operations, controls on occupational exposure of radiation workers, and limits on the radionuclide concentrations in air and liquids discharged to the environment. This section addresses the potential impact on the environment of the revisions to 10 CFR Part 20 by reviewing the current status and likely impact of implementing the revisions in each of these three major topics of the regulation. For each subject area, the current status and likely effect of the revised regulation is quantified for the NRC licensed facilities with the largest potential for environmental impact, i.e. Part 50 licensees. Nuclear power reactors, in particular, have the potential for the most significant direct impact on the environment via plant effluents, as well as having the most significant impact on total occupational exposures.

#### TABLE 1. PRINCIPAL CONCERN'S AND BENEFITS OF THE PROPOSED REVISION OF 10 CFR PART 20

#### CONCERN WITH PRESENT PART 20

- 1. Mary Values In Appendix B Do Not Reflect Current Knowledge.
- Present Macs can cause underestimates of doses by a factor of 6 for most alpha emitters and 60 for thorium.
- Soluble" and "insoluble" designations in Part 20 and many other bases were abandoned by health physicists many years ago.

w

#### PART 20 KEVISION

 Revises and expands Appendix B to reflect contemporary knowledge of dosimetry and biology.

#### BENEFITS OF REVISION

- Derived values will reflect ICRP risk based system and make use of contemporary knowledge.
- Air concentrations are based on a lung model which permits adjustment for the particle sizes of aerosols.
- Values are presented for various compounds.
- Coverage of radionuclides has been increased from 260 to 757.
- o Of the radionuclides where comparisons can be made, about 65% of the new values are less restrictive, about 8% are unchanged, and about 27% are more restrictive.
- Annual and Lifetime doses to individuals receiving highest exposures will be reduced.
- Risks to radiation workers receiving highest exposures will be more comparable to those in low risk industries.

2. <u>5(N-18) Dose-/veraging Formula</u> <u>Permits Workers to Receive up to</u> <u>12 Rems Per Year from External</u> <u>Sources.</u>

o Deletes 5(N-18) and adopts 5 rem (0.05 Sv) per year

 Provides "planned special exposures" for necessary and unavoidable activities.

#### CONCERN WITH PRESENT PART 23

#### PART 20 REVISION

- Potential. Total risk could be substantial (3-10%) from 50 years external exposures at 5 rems per year and additional from internal exposures.
- 3. Dose Limits for Internal and External Doses are Independent.
- Risks from doses at the limits to various organs are unequal.

5

Requires summations of internal and external doses.

 Adopts ICRP "effective dose equivalent" which adjusts doses to various organs to whole body dose equivalent based on risk.

#### BENEFITS OF REVISION

- Provides substantial flexibility for licensee to manage justifiable exposures beyond selected annual dose limits.
- Provides for readily monitored records.
- Effective dose equivalent limits from combined external and internal exposures are related to individual risk.
- Limits for various organ doses reflect comparable risks.
- Dose weighting factors based on Quantified risk of radiationinduced health effects are consistent with Commission policies on use of quantitative risk.
- Workers and public can understand risk base which is more rational than present dose limit selection.
- Adherence to stated limits of doses to workers subjected to both external and internal exposures will be ensured.
- Ensures adequate radiation protection program and ALARA efforts by all licensees.

4. <u>No Requirements for Formal</u> <u>Radiation Protection Program</u> <u>or for ALARA</u>.  Requires written radiation protection program with ALARA provisions.

#### CONCERN WITH PRESENT PART 20

- Uneven requirements among types of licensees brought about mostly through licensing actions other than Part 20.
- 5. <u>All Values Below the Limits</u> <u>are Treated as Equally</u> <u>Acceptable</u>.
- De facto limits are established by licensing actions.
- 6. <u>Dose Data on Specific Workers</u> <u>Usually are not Available to</u> <u>Staff Until Workers Terminate</u> <u>Employment</u>.
- Workers are not required to be informed of annual or accumulated doses without request.
- 7. <u>Presents No Clear Dose Limits</u> for Members of the Public.
- Requires use of effective dose equivalents.
- o Establishes 100 mrem/yr (1 mSv/yr) o effective dose equivalent (external and internal sources). Permits licensee to use 500 mrem/yr (5 mSv/yr) upon application to and approval by NRC.

#### BENEFITS OF REVISION

- Provides basis for more effective ALARA efforts with reliance on licensee's judgment.
- Provides requirements in Part 20 for enforcement actions.
- Actions are taken to reduce exposures before dose limits are exceeded.
- Encourages reduction of doses to workers and public prior to reaching limits.
- Reporting requirements result in greater awareness of current conditions.
- Greater awareness of workers' accumulated doses is likely to increase their participation in dose reduction efforts.
- Dose limits for public would include possible multiple sources and multiple exposure modes.

## TABLE 1. (Continued)

#### PARI 20 REVISION

- Requires management commitment and participation.
- Requires selection of investigation levels for doses to workers below dose limits.
- o Emphasizes ALARA.

#### 57

The specific changes resulting from the revisions to 10 CFR Part 20 are itemized in Table 2. The potential environmental impact of these changes will be assessed by considering the current status and the expected impact of the Part 20 revisions for each of the three major subjects, i.e. dose limits to members of the public, occupational exposures, and concentration limits in plant effluents.

1. Dose Limits for Members of the Public

10 CFR Part 20 is not the only regulation which establishes limits for exposure to members of the general public, i.e., those people outside of restricted areas. Two other parts of the Code of Federal Regulations also govern these doses: 10 CFR Part 50 Appendix I Section  $II^{[1]}$  which sets design objectives and operating criteria for effluents from nuclear power plants and 40 CFR Part 190 Subpart  $B^{[2]}$  which sets dose limits for the uranium fuel cycle. The Appendix I design objectives are:

10 CFR Part 50 Appendix I Section II 3 mrems total body from liquid effluents 10 mrems any organ from liquid effluents 5 mrems total body from gaseous effluents 15 mrems to the skin from gaseous effluents 15 mrems to any organ from particulates and radioiodines in gaseous effluents

The EPA dose limits are:

40 CFR Part 190 Subpart B 25 mrems total body all effluents and direct radiation from uranium fuel cycle sources 75 mrems thyroid all effluents and direct radiation from uranium fuel cycle sources 25 mrems any other organ all effluents and direct radiation from uranium fuel cycle sources TABLE 2. COMPARISON OF SALIENT ISSUES IN THE EXISTING 10 CFR 20 WITH THE REVISED VERSION

Issue	Existing Part 20		Revision	
OCCUPATIONAL Limits	External			
	Whole body, head and trunk, active blood- forming organs, lens of eye, or gonads	1.25 rems/qtr or 3 rems/qtr with lifetime occupa- tional exposure history and within 5(N-18) dose- averaging formula.	Whole body, head, trunk, arm above elbow, and leg above knee	5 rems/year (0.05 Sv/ year) - includes summation of (external) deep dose equivalent and (internal) committed* effective dose equivalent.
			Lens of eye	15 rems/year (0.15 Sv/year
	Hand and forearms; feet and ankles	18 3/4 rems/qtr (75 rems/yr)	Hand, elbow, arm below elbow, foot, knee, and leg below knee	50 rems/year (0.5 Sv/year)
	Skin of whole body	7 1/2 rems/qtr (30 rems/yr)	Skin (1 cm <sup>2</sup> )	50 rems/year (0.5 Sv/year)
	No summation of internal (organ) doses.		Weighted organ doses for all organs are summed.	
	No summation of external and internal doses.		Doses from external and internal sources are summed.	

\*Except for selected uranium and transuranic radionuclides for which the derived air concentrations (DACs) and annual limits of intake (ALIs) are hard to measure at levels found in the workplace. For these nuclides, the regulation may be based upon the effective dose equivalent received in the year rather than the committed effective dose equivalent.

Issue	Existing Part 20		Revision			
	Internal					
	Intake equivalent to 520 MPC-hours/qtr. (= 2080 MPC-hours/yr) Calculated to result in a 50-year committed dose of: Whole body Bone, thyroid,	1.25 rems (5 rems/yr) 7.5 rems	Annual limit o 2000 DAC-hours Calculated DAC are assigned w of risk to tha estimate of ri body exposure. (0.5 Sv/year) For body parts	f intaka /year. s are ba eighting t organ sk per u "Capp used to other f	e (ALI) equivale ased on the foll factors, based per unit of dos unit of dose for ing" dose limit avoid nonstoch than those list	ent to lowing: Organs d on the estimates se relative to the r uniform whole of 50 rems/year astic effects. ed above:
	and skin Other organs	(30 rems/yr) (15 rems/yr)	Tissue	MI	Inferred Dose Limit (rems/year)	Actual Dose Limit (rems/year)
			Gonads	0.25	20	20
			Breast	0.15	33	33
			Red hone			
			marrow	0.12	42	42
			Lung	0.12	42	42
			Thyroid	0.03	167	50
			Rono surfaces	0.03	167	50
			Each of 5 remaining	0.05	83	50
			organs with the largest dose			
Planned Special	5(N-13) dose averaging provided - with quarter limits	·ly	Planned specia to the annual Limits set at events in a ye from all even sion is elimin may include in	al expos limits l x ann ear and ts. 5(N nated. nternal	ures allowed in from routine ex ual limits/year 5 x annual limi -18) dose avera Planned special as well as exte	addition posures. from all ts/lifetime ging provi- exposures rnal doses.

00

Issue	Existing Part 20	Revision
Embryo/Fetus	Not addressed.	0.5 rems (5 mSv) during the entire pregnancy due to occupational exposure of the "declared" pregnant woman.
ALARA	Recommended	ALARA program required.
BASIS FOR LIMITS	Biological damage or health effects would not be statistically observable.	"Acceptable risk (10 <sup>-4</sup> per year for workers, 10 <sup>-5</sup> to 10 <sup>-5</sup> per year for members of the public) based on estimated radiation-induced fatal cancers and serious hereditary disorders. Upper limit of organ dose set to avoid nonstochastic (threshold) effects, such as cataracts.
INTERNAL DOSIMETR METHODOLOGY	IC	
Irradiation	Dose to the most irradiated organ, i.e., "critical organ", used to limit intake via "Maximum Permissible Concentrations" (MPC)	Dose to each organ is calculated, weighted by a factor representing the ratio of the risk dose to that organ to risk from 5 rems (0.05 Sv) of whole body irradiation, and then the products are summed. Values for ALIs and DACs have been calculated for each radionuclide.
	Doses from radionuclides deposited in non-critical organs are ignored.	Weighted doses to organs from radionuclides deposited anywhere in the body are summed.
Lung model	1959 ICRP-2 model used.	Improved 1966 model of ICRP Task Group on Lung Dynamics used.
Retention in	Aerosols ranked "Soluble" or "Insoluble"	Aerosols ranked by translation and elimination rates, i.e., D (days), W (weeks), and Y (years).

9

Issue	Existing Part 20	Revision
	No consideration given for aerosol aerodynamic properties	Assumes lum AMAD. Adjustments for other aerosol size distributions, and physical and chemical properties are possible.
Translocation	Based on 1959 biological data.	Based on 1978 biological data from ICRP-30.
PUBLIC		
Limit	Implied limit for individuals of 0.5 rems/ year to whole body, blood-forming organs, and gonads; 3 rems/year to bone and thyroid; and 1.5 rems/year to other organs. No summation of external and internal dose. No consideration of food pathways.	Explicit limit of 0.1 rem/year (1 mSv/year) for individuals from all sources. Includes summation of external and internal doses and food pathways. Licensee may apply for and with approval operate up to 500 millirems/year (5 mSv/yr). For members of the public, report required for exceeding 0.1-re (ImSv) level to members of public, unless licensee received prior approval for conducting operations at higher doses.
MONITORING	김 사람이 있는 것은 것을 생각해 주는 것을 가지 않는 것을 했다.	
Adu?t	Required at 25% of the basic quarterly limit (0.312 rems/quarter).	Required at 10% of the annual limit for deep dose equivalent (0.5 rem or 5 mSv).
		Required at 10% of the annual limit for eyes, skin, or extremities.
	Required for intakes greater than 25% of 520 MPC-hours in a quarter.	Required at 10% of the ALIs.
Minor	Required at 5% of the basic quarterly limit 0.0625 rems.	Required at 5% of the external annual limits for adults. Required at 5% of the ALIs for adults.

Issue	Existing Part 20	Revision		
SEWER DISPOSAL	Concentration limits equivalent to 5 rems/year by potential ingestion.	Concentration limits equivalent to 0.5 rem/year (5 mSv/year) by potential ingestion.		
RECORDS				
Determination of prior dose	Occupational exposure history required as condition for allowing 3 rems per quarter and use of 5(N-18) dose-averaging formula. Signed statements of dose during last quarter required upon employment.	Occupational exposure history (effective dose equivalent received during the current year and, when appropriate, all planned special exposures and over-exposures received during the lifetime of the individual) required for all individuals requiring provision of individual monitoring devices or services.		
Current exposure records	Form NRC-5 includes only external dose. Includes items for calculating status under 5(N-18).	Revised Form NRC-5 includes external dose, internal dose, summation, and dose received during planned special exposures and as overexposures.		
Effluent releases	Implied under survey requirement.	Explicitly required.		
Planned Special No provision. Exposures		Records required.		
REPORTS				
Criteria for	20 times the basic quarterly dose limits	5 times the annual dose limits		
immediate notifi- cation of incidents	Property damage \$200,000.	Same		
Overexposures Required if limits for short-term F of public radiation levels or annual effluent a releases to unrestricted areas are exceeded.		Required if any individual in an unrestricted area exceeds 0.1 rem (1 mSv) in one year.		

1

11

.

Issue	Existing Part 20	Revision
REPORTS (Continued)		
Planned special exposures	No provisions for planned special exposures.	Report required in all cases.
Individual monitoring reports	Annual statistical summary report required of 7 categories of licensees. Termination report required of same 7 categories of licensees.	Data for the same 7 categories of licensees on all individuals required to be monitored (or all all individuals actually monitored at licensee's option), in lieu of statistical summary and termination reports to NRC.
		For members of the public, report required for exceeding 0.1-rem (ImSv) level to members of public, unless licensee received prior approval for conducting operations at higher doses.
Reports to individuals	Required by 19.13(d) for any infor- mation reported to NRC. Applies only to overexposures and termination reports. Pursuant to 19.13, other reports on exposures are available to the individual on request.	Reports required to employee on annual basis. Doses reported will be effective dose equivalents. In addition, licensees would report to individuals any planned special exposures and estimates of both annual effective dose equivalent and 50-year committed effective dose equivalent to their employees.

## Current Status

The existing regulation contains an implied whole body dose limit of 0.5 rems/y to an individual in an unrestricted area in 10 CFR Part 20.105.

The revised 10 CFR Part 20, Subpart D 20.301, explicitly states dose limits for members of the public. The limit is stated as the sum of external deep dose equivalents and internal committed effective dose equivalent to any individual.

The values identified in 10 CFR Part 50 Appendix I are objectives for the design, installation, and operation of equipment for the control of effluents. On the basis of this regulation, NRC has required licensees to develop radiological effluent technical specifications<sup>[3,4]</sup> which govern the operation of commercial light water reactors in the United States. Thus the design objectives are incorporated in the licenses of these facilities as operational limits. Compliance with the annual limits incorporated in the technical specifications of operating power reactors is demonstrated by calculation of individual doses, based on effluent and other measurements.

The calculated population dose commitment for  $1984^{[5]}$  was 280 person-rem. The range and geometric mean of the average total body doses for all light water reactors in the United States were  $1.4 \times 10^{-6}$  mrems ( $1.4 \times 10^{-8}$ mSv) to  $6.2 \times 10^{-2}$  ( $6.2 \times 10^{-4}$ mSv) and  $4.5 \times 10^{-4}$  ( $4.5 \times 10^{-6}$ mSv) mrems respectively.

#### Impact From Part 20 Revisions

In contrast to the implied limit of the existing rule, the revised Part 20 includes an explicit limit of 100 mrem/yr (lmSv/yr) effective dose equivalent from external and internal exposures. Based on the statistics of currently incurred doses to the public from NRC licensed facilities, and the current trend of declining exposures, it is concluded that the public dose limit of the revised Part 20 will not have any impact on members of the public. Instead, it is likely that public exposures will continue to be limited by other regulations, in particular, Appendix I of 10 CFR Part 50.

#### 2. Occupational Doses

The major emphasis of the existing Pari 20 regulation is on occupational doses. The following paragraphs address the current status and the likely effect of the modifications to Part 20 on this segment of the environmental impact of this regulation.

## Current Status

Occupational dose is monitored by all licensees and reported to the NRC annually. Since 1973 these dose statistics have been compiled in annual reports. The latest report published covers calendar year 1985.<sup>[6]</sup> Table 3a lists worker doses from 1973 to 1987 for all light water reactors (boiling water reactors and pressurized water reactors). With a few exceptions, average worker dose has continued to decrease since 1977. Table 3b shows a breakdown of the average worker dose by reactor type. A declining trend is evident, from this data, for both reactor types.

Under the current regulations, an individual may receive up to 3 rem per quarter and 12 rems per year whole body dose provided an occupational exposure history (Form NRC-4) is on file with the licensee. Table 3 lists a summary of annual whole body exposures reported to the Commission from 1968 to 1934. The data show that since 1982 no individual has received greater than 12 rems/y whole body dose and a small fraction, 0.1%, of the total monitored workers received greater than 5 rems/y. Approximately 95% of all workers received less than 2 rems/y. These statistics indicate that the licensees have been successful in keeping worker dose below the

		and the second se		_
Year	Number of workers with measurable exposure	Total person_rem	Average Dose rems	
1973	14780	13963	0.94	
1974	18466	13722	0.74	
1975	25489	20879	0.82	
1976	35447	26433	0.75	
1977	38858	32511	0.84	
1978	42674	31804	0.74	
1979	60160	39981	0.66	
1980	74503	53796	0.72	
1981	76730	54142	0.71	
1982	79224	52190	0.66	
1983	80804	56471	0.70	
1984	92918	55214	0.59	
1985	92864	43042	ũ.46	
1986*	103179	42725	0.41	
1987*		40879		

TABLE 3a. OCCUPATIONAL EXPOSURE AT LWRs

\*Preliminary Data

5

	Average Worker Dose (mrems)		
Year	All LWRs	All BWRs	A11 PWRs
1973	0.94	0.85	1.0
1974	0.74	0.81	0.68
1975	0.82	0.86	0.76
1976	0.75	0.71	0.79
1977	0.14	0.89	0.65
1978	0.74	0.74	0.65
1979	0.66	0.73	0.56
1980	0.72	0.87	0.52
1981	0.71	2.73	0.61
1982	0.66	0.76	0.53
1983	0.70	0.82	0.56
1984	0.59	0.66	0.49
1985	0.46	0.54	0.41

TABLE 3b. OCCUPATIONAL EXPOSURE AT LWRs
current regulatory limits, and in most cases at least a factor of 2.5 below the new limit.

Impact of Part 20 Revisions on Occupational Doses

The revised 10 CFR Part 20.201 states a limit of 5 rems/y for adult occupational radiation dose. In the area of external deep dose equivalent the impact of this limit should be minimal if judged by the data in Table 4. This table shows that more than 95% of the radiation workers received doses of less than 2 rems/y for 1984. This leaves a 3 rems margin for most workers to address the new requirement that the external deep dose equivalent and the internal committed effective dose equivalent due to internal deposition of radionuclides, be additive. With the increased awareness of airborne contaminants and the improvement of the worker environment, it is reasonable to expect that this margin will not be exceeded except in very rare instances.

In the area of internal dose limits the current regulations limit exposure to contaminated air to 520 MPC-hours/qtr (hours at maximum permissible concentration per quarter) by an individual in a restricted area. This restriction on breathing contaminated air indirectly limits internal organ doses. The change in the revised Part 20 establishes a limit of 2000 DAC-hours/y (derived air concentration hours per year) or inhalation of one ALI (allowed annual limit of intake), and puts a capping dose limit of 50 rem/y to any organ to avoid non-stochastic effects. The major difference between the two approaches to limit exposure from ingested contaminants is in the calculation of MPCs and DACs. MPCs are calculated based on a single critical organ concept. DACs are calculated based on all organs receiving a dose both directly and/or indirectly from internally deposited radionuclides and weighting the estimate of risk to an organ relative to the risk for uniform whole body exposure.

#### TABLE 4. SUMMARY OF ANNUAL DOSE DISTRIBUTIONS FOR CERTAIN NRC LICENSEES

1968 - 1984

	Total Number of Monitored Persons		Percent of Individuals	Percent of Individuals	Number of	
Year	Compiled Number	(Adjusted* Number)	With Doses <2 rems	With Doses >5 rems	With Doses >12 rems	
1968	36,836		97.2%	0.5%	3	
1969	31,176		96.5%	0.5%	7	
1970	36,164		96.1%	0.6%	0	
1971	36.311		95.3%	0.7%	1	
1972	44,690		95.7%	0.5%	8	
1973	67,862		95.0%	0.5%	1	
1974	85,097		96.4%	0.3%	1	
1975	78,713		94.8%	0.5%	1	
1976	92,773		95.0%	0.4%	3	
1977	98,212	(93,438)	93.8%*	0.4%	1	
1978	105,893	(100,818)	94.6%*	0.2%*	3	
1979	131,027	(125,316)	95.2%*	0.2%*	1	
1980	159,177	(150,675)	94.6%*	0.3%*	0	
1981	157.874	(149,314)	94.6%*	0.2%*	1	
1982	162,456	(154,117)	94.9%*	0.1%	0	
1983	172,927	(164,239)**	94.6%*	0.1%	0	
1984	204,069	(194,840)**	95.91*	0.1%	0	

\*Based on the distribution of individual doses after adjusting for the multiple counting of transient reactor workers (see Section 5). \*\*The termination data for about 15% of the individuals terminating during 1983 or 1984 have not been entered into the REIR System.

The regulations provide limits for planned special exposures. Under the current regulations a worker could exceed a 5 rems/y dose if an occupational exposure history was on file, in which case the worker could receive up to 12 rems/y. The revised regulation allows for planned special exposures exceeding the annual limits by an increment equal to the annual dose limit during any one year. No more than five times the annual limit may be permitted during a workers lifetime. These new criteria may have an affect on some licensee s operation, but the data in Table 4 indicates that the impact is rot likely to be significant.

The revisions include an explicit requirement to include the "as low as reasonably achievable" (ALARA) concept in radiation protection programs. The ALARA concept is not new. Although not an explicit general regulatory requirement heretofore, the NRC's regulatory practice has included this basic concept in a number of regulatory programs (e.g., effluent technical specifications discussed previously'. As a result, most, if not all, licensees currently have ALARA programs whose functions generally cover those listed in Section 20.102.

# 3. Concentration and Effluent Limits

A significant change occurs in the summation of both external and internal dose for a member of the public and the restriction of that dose to 0.1 rems/yr. The summation of the external and internal doses has required new derived limits in air and water to be calculated based on the 0.1 rem allowed dose. The revised effluent concentration limits are based upon an annual effective dose equivalent of 50 millirem in each release pathway (air and water).

The MPC changes, nevertheless, will have little environmental impact. This is a result of the defacto limitation on doses for members of the public arising from the more restrictive requirements in 10 CFR Part 50 Appendix I and 40 CFR Part 190. In the areas of monitoring and record keeping, the revised regulations do not require anything different than what the licensees are presently required to do by license conditions or effluent technical specifications.

Liquid and gaseous effluents are an indicator of the facility's impact on the environment. Table 5 presents reported effluent release activities<sup>[6]</sup> for boiling water and pressurized water reactors for the years 1972 and 1984, and 1973 and 1984 respectively. There are significant differences between the two types of reactors but the important comparison is the historical trend for each type of facility. It is readily apparent that total radioactivity released in gaseous and liquid effluents have decreased during the span of years covered by the data. The decrease in noble gas releases was enhanced in boiling water reactors by the installation of augmented offgas systems in the late 1970s.

## IV. Alternatives

The proposed revision of 10 CFR Part 20 reflects new scientific knowledge, increased uses of radiation and sources of radiation exposure, and experience gained during nearly thirty years. The proposed revision would adopt, in part, the approach to radiation protection and the system of dose limitations recommended by the ICRP. This revision would correct the deficiencies identified in the existing Part 20, and would also include a number of other changes required to make the regulations consistent with up-to-date principles that underlie radiation protection.

"ternative 1: No Action

To continue with the existing regulations would be to ignore recent scientific advancements. This alternative would result in a failure to implement the international system of dose limitations recommended

	Gaseous Effluents							
	BWRs			PWRs				
Nuclide	Year	Range	Average	Year	Range	Average		
Noble Gas	1972 1984	1.8(4)-8.77(5)** 118-1.67(5)	4.24(5) 5.22(4)	1973 1984	32-1.1(4) 28.4-4(4)	2280 6300		
Iodine	1972 1984	0.15-5.89 1.86(-4)-2.92	1.69 0.22	1973 1984	.01-1.61 2.07(-4)-0.41	0.41 0.049		

# TABLE 5. RADIOACTIVE EFFLUENTS FROM LWRs (curies/year)

Liquid Effluents

	BWRs			PWRs		
Nuclide	Year	Range	Average	Year	Range	Average
3 <sub>H</sub>	1972 1984	10.4-120 1.1-125	40.4 28.3	1973 1984	154-3900 1.7-3660	1196 589
MFP/AP*	1972 1984	1.1-51.5	22	1973 1984	0.1-27.8 0.019-13.0	6.1 2.6

\*MFP/AP = Mixed Fission products and activation products

\*\*8.77(5) = 8.77 x 10<sup>5</sup>

by ICRP in Publications 26 and 30, the NCRP in Publication 91, and most importantly, in the 1987 Federal Guidance for occupational exposure. The inclusion of the (internal) committed dose equivalent estimates, which could be an important part of determining the risks, would continue to be omitted from the total dose equivalent. Furthermore, if the present 10 CFR Part 20 is not changed, the present lack of clarity and organization will continue to exist.

While it may be shown that present regulatory efforts have been effective in providing an adequate level of protection for occupational workers, the failure to reevaluate the regulatory limits in view of the new scientific information would leave the NRC open to criticism and make it difficult for NRC to respond to Congressional (including GAO) and public inquiries about the adequacy of present regulations for protection against radiation.

Alternative 2: Delay Action Awaiting Further Developments

Revision of Part 20 could be delayed pending revised Federal Guidance EPA Standards, or NCRP recommendations on public exposure limits.

Federal Guidance

The existing Part 20 is in accord with present Federal Guidance to agencies in the conduct of radiation protection activities published in 1960 and approved by President Eisenhower. This guidance applies both to exposures of workers and members of the public. EPA has issued revised occupational exposure guidance which is in line with ICRP recommendations but has not issued revised guidance for the public. It is not necessary that NRC await such guidance prior to publishing a revision of Part 20 because the NRC has independent statutory authority to set such limits. NRC, as a matter of policy, will generally implement any new Federal Guidance in the conduct of its regulatory programs.

## EPA Standards

EPA has specific authority under the Atomic Energy Act, as amended, to establish "generally applicable environmental standards" for radiation and radioactive materials covered under the Act. This authority is limited to application at points beyond the site boundaries of NRC-licensed facilities. EPA has used this authority for control of effluents, from facilities in the uranium fuel cycle, in standards for high-level waste repositories, and in proposed low-level waste disposal standards. Since EPA is not immediately developing a generic standard for population exposure, no delay in Part 20 for this reason is justifiable.

EPA has issued final standards to control air emissions of radioactive materials which would apply to some NRC licensees, with exception for those NRC-licensed facilities already covered by EPA standards promulgated under the Atomic Energy Act. However, these standards have been remanded to EPA for revision by a Federal court. The details of implementation of the proposed standards by NRC have not been finally agreed to by NRC and EPA. Publication of the revised Part 20 is not.likely.to be affected by EPA's final standard, as the EPA standards did not require any modifications to the revised 10 CFR Part 20.

### NCRP Recommendations

The National Council on Radiation Protection and Measurements (NCRP) has issued a major revision of its basic radiation protection criteria in NCRP Report 91. This report and its recommendations were considered in the preparation of the final Part 20 rule. The degree of protection recommended in NCRP Report No. 91 is not substantially different than that provided following the 1977 ICRP recommendations which form the basis of the current Part 20 revisions.

# Alternative 3: Partial Updates of Part 20

3a. Update Only the Concentration Limits for Radionuclides in Air and Water.

If concentration limits were the only part of the regulations considered in the proposed revision, then the other shortcomings would still remain. The ALIs and DACs presented in ICRP Publications 30 and 32, and proposed for incorporation into Appendix B of the draft revision of Part 20, include . sighting factors and consideration of committed effective dose equivalents that would not be appropriate for incorporation into the present Part 20. Development of replacements for these limits without the ICRP-recommended weighting factors, would involve extensive calculations. In addition, updating the concentration limits could constitute about 30% of the initial cost of the entire revision and about 50% of the annual cost. While updating the concentration limits would seem to be the most important of the proposed provisions, the other provisions are considered to be very necessary at this time and the benefits that would result from implementation 're believed to outweigh the costs. While -the concentration limit changes would have the greatest impact on the licensees who process uranium and thorium, the costs from the remaining limit revisions are more evenly spread over the entire nuclear industry and are less likely to impact heavily on a single type of licensee.

3b. Limit Revisions to those Necessary to Make 10 CFP Part 20 Consistent with the Major Recommendations of the Federal Guidance and ICRP-26 and ICRP-30.

This alternative involves making only those revisions that would bring Part 20 in line with the dose limits and the concept of "effective" whole-body dose recommended in ICRP Publication 26. It would include incorporation of the dosimetric models and the biological information from ICRP. These changes would leave the other weakness noted above. This partial update, never-the-less, would entail most of the cost associated with the full revisions.

#### V. Summary and Conclusions

This revision of 10 CFR Part 20 was guided by (1) Federal Guidance issued by the President; (2) the recommendations of world-recognized authorities on radiation protection; (3) the most current national and international data on health effects and radiation dosimetry; (4) the advice of licensees, labor uniors, health physics organizations, and other interested parties; and (5) public comments received on the proposed rule.

In drafting the Part 20 revision, references were consulted including the following:

International Commission on Radiological Protection publications:

- #? "Report of Committee II on Permissible Dose for Internal Radiation" (1959)
- #23 "Report of the Task Group on Reference Man" (1974)
- #26 "Recommendations of the International Commission on Radiological Protection" (1977)
- #27 "Problems Involved in Developing an Index of Harm"
   (1977)
- #30 "Limits for Intakes of Radionuclides by Workers" (1978)
- #32 "Limits for Inhalation of Radon Daughters by Workers"
  (1981)

International Commission on Radiation Units and Measurements reports:

#33 - "Radiation Quantities and Units" (1980)

National Council on Radiation Protection and Measurements reports:

#38 - "Protection Against Neutron Radiation" (1971)

#39 - "Besic Radiation Protection Criteria" (1971)

#91 - "Recommendations on Limits for Exposure to Ionizing Radiation" (1987)

National Academy of Sciences Committee on the Biological Effects of Ionizing Radiations reports:

"The Effects on Populations of Exposure to Low Levels of Ionizing Radiation" (1972, 1980)

International Atomic Energy Agency Document: \*Basic Safety Standards for Radiation Protection\* (1981 Rev.)

European Atomic Energy Communities directive:

Official Journal of the European Communities L246 (1980) "Basic Safety Standards for the Health Protection of the General Public and Workers Against the Dangers of Ionizing Radiation"

United Nations Scientific Committee on the Effects of Atomic Radiation reports:

"Sources and Effects of Ionizing Radiation" (1977) "Ionizing Radiation: Sources and Biological Effects" (1982)

Section 3 discussed potential areas of environmental impact of the revised regulation. The revision will have little impact on doses to members of the public because more restrictive limits are found in 40 CFR Part 190, 10 CFR Part 50 Appendix I, and in some licensees' effluent technical specifications.

For those changes that affect the occupationally exposed individual there is a potential positive impact on health protection. In most

instances current practice of the licensees fits within the proposed changes, so that no major impact is anticipated for the majority of radiation workers as a result of reduced upper limits, or the combination of external and internal dose equivalents. For the very small fraction of the exposed work force with accumulated doses approaching the occupational dose limits, the determination of the total effective dose equivalent by combining both internal and external deep-dose equivalents provides a positive impact by ensuring appropriate limitation of the exposed individual's health risk.

In converting from concentration limits in the present Part 20 to the annual limits of intake (ALIs) and derived air concentrations (DACs) in the revision, many of the DACs are different from the limits currently listed in Appendix B, 10 CFR Part 20. Where comparison is possible, about 65% of the DAC listings are less restrictive, about 26% are more restrictive (in particular for radionuclides resulting in alpha particle emission), and the rest remain essentially unchanged.

The changes arise from a number of considerations that include:

- Dose contribution to all organs and tissues from radioactive material-deposited in all organs and tissues, rather than dose to -... a critical organ from radioactive material deposited in that organ;
- (2) Use of a quality factor (QF) of 1 for low-energy beta particles, and a QF of 20 for alpha particles, rather than 1.7 and 10, respectively;
- (3) Updated biological models; and
- (4) Application of the risk-based weighting factors within the ICRP system of dose limitation.

The current level of radiation protection is affected much more by consideration of what is ALARA than by specific dose limits. It is considered that a strong ALARA requirement will ensure that the level of radiation protection will remain high even though the limits for certain radionuclides might be increased. Furthermore, the proposed revision of Part 20 calls for the summation of ext. Hal dose and internal committed effective dose equivalent, thereby providing an additional constraint on the amount of radioactive material that may be taken into the body.

While greater assurances would be provided for worker and public protection against radiation, the revision of Part 20 will not have a significant impact on the quality of the human environment. Changes in the limits for environmental releases will not usually be limiting as other existing standards already in practice will be governing.

#### VI. Environmental Assessment: Negative Declaration Finding

The Commission has determined under the National Environmental Policy Act of 1969, as amended, and the Commission's regulations in Subpart A of 10 CFR Part 51, not to prepare an environmental impact statement for this revision of 10 CFR Part 20 because the Commission has concluded, on the basis of an environmental assessment, that promulgations of this revision of 10 CFR Part 20 as a final rule would not be a major Federal action significantly affecting the quality of the human environment.

The revised 10 CFR Part 20 changes the level for protection of the general public from an implicit limit of 500 millirems (5mSv) per year to an explicit limit of 100 millirems (1mSv) per year. There are also numerous changes in airborne and water radionuclide concentration limits. These changes result from changes in the models and parameters used to estimate the radiation dose associate with intake of a radionuclide. Some of the concentration limits for the general

public in this revision are new; some limits are similar to the present limits, while other limits are changed upward or downward.

Despite the changes in the dose and concentration limits, the Commission believes that issuance of the final Part 20 rule will not have a major impact on the environment. The primary basis for this conclusion is that in addition to 10 CFR Part 20, there are other regulations that govern allowable doses to members of the public which remain unchanged by the revision to Part 20. These other regulations include Appendix I to 10 CFR Part 50, 10 CFR Parts 60 and 61 and the EPA's generally-applicable environmental standards in 40 CFR Parts 190, 191, 192 and (proposed 193), and the National Emission Standards for Hazardous Air Pollutants (NESHAPS) in 40 CFR Part 61 (radionuclides). These standards set limits or design objectives (Appendix I) for releases of radioactive material to the general environment which are generally more restrictive than the dose limits in Part 20. Consequently, since these more restrictive standards. remained essentially unchanged by the Part 20 revision, the level of public protection and the associated environmental impact are not changed appreciable from those associated with the current rule and the aforementioned regulations.

### VII. Availability of Documents

This environmental assessment may be examined and copied (for a fee) at the NRC Public Document Room. Single copies of this assessment may be obtained from the NRC project manager:

Harold T. Peterson, Jr. Radiation Protection and Health Effects Branch Office of Nuclear Regulatory Research U.S. Nuclear Regulatroy Commission Washington, D.C. 20555 Telephone: (301) 492-3640

#### VIII. References

- Title 10, <u>Code of Federal Regulations</u>, Part 50, Appendix I, "Numerical Guides for Design Objectives and Limiting Conditions for Operation to Meet the Criterion, 'As Low As Is Reasonably Achievable,' for Radioactive Material in Light-Water-Cooled Nuclear Power Reactor Effluents."
- Title 40, <u>Code of Federal Regulations</u>, Part 190, "Environmental Radiation Protection Standards for Nuclear Power Operations."
- \*Radiological Effluent Technical Specifications for Boiling Water Reactors,\* Rev. 3, Draft 7,\* intended for contractor guidance in reviewing RETS proposals for operating reactors, NUREG-0473, September 1982.
- \*Radiological Effluent Technical Specifications for Pressurized Water Reactors,\* Rev. 3, Draft 7,\* intended for contractor guidance in reviewing RETS proposals for operating reactors, NUREG-0472, September 1982.
- D. A. Baker, "Population Dose Commitments Due to Radioactive Releases from Nuclear Power Plant Sites in 1984," NUREG/CR-2850, Vol. 6 (PNL-4221) Vol. 6, January 1988.
- J. Tichler, K. Norden, J. Congemi, "Radioactive Materials Released from Nuclear Power Plants, Annual Report 1985," NUREG/CR-2907, Vol. 5, BNL-NUREG-51581, Vol. 5, January 1988.
- B. G. Brooks, "Occupational Radiation Exposure at Commercial Nuclear Power Reactors and other Facilities 1985," NRC Report NUREG-0713, Vol. 7, April 1988.

ENCLOSURE E FINAL BACKFIT ANALYSIS

Enclosure E

FINAL BACKFIT ANALYSIS FOR THE REVISION OF 10 CFR PART 20, "CTANDARDS FOR PROTECTION AGAINST RADIATION"

August 1990

Division of Regulatory Applications Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555

# FINAL BACKFIT ANALYSIS FOR THE REVISION OF 10 CFR PART 20

#### A. Introduction

The comments of the NRC Commissioners in Section XXXVIII of the proposed rule indicated that the proposed rule was being published for public comment prior to a Commission decision on whether or not a backfit analysis should be prepared. Subsequently, at the direction of the Commission, the NRC staff prepared a draft backfit analysis for the Part 20 revision. This draft backfit analysis was published for public comment in the August 29, 1986 Federal Register (51 FR 30870). The public comment period on this analysis ran concurrently with the extended comment period on the proposed rule, both ending on October 31, 1986.

The Backfit Analysis was prepared to satisfy the requirements of § 50.109 of 10 CFR Part 50 (commonly called "The Backfit Rule") that:

- a systematic and documented analysis be prepared for any new requirements placed upon existing nuclear power reactor licensees;
- (2) the analysis shows that the new requirement provides a substantial increase in the overall protection of public health and safety or in the common defense and security; and
- (3) that the direct and indirect costs of implementation are justified in terms of the increased protection.

The Draft Backfit Analysis for the Part 20 rule concluded that: (1) the proposed Part 20 rule would provide improvements in health protection; (2) it could not be shown unequivocally that the direct and indirect costs of implementation would be justified in view of this increased protection; and (3) the continuation of the Part 20 rulemaking might be justified on the basis of qualitative as well as quantitative factors. The Draft Backfit Analysis also contained a request for public comment on these tentative conclusions, particularly on whether qualitative factors should be considered in evaluating costs and benefits.

There were 17 comments that responded specifically to the questions asked in the draft Backfit Analysis. Three of the 13 industry comments found the draft Backfit Analysis to be adequate, but only two commenters believed that it fully satisfied the backfit requirements in § 50.109. Seven commenters (all electric utilities) thought that the draft Backfit Analysis was inadequate. Some felt that a cost-benefit analysis should have been done for each of the major changes. [Note: The Regulatory Analysis contains estimates of the costs and benefits of each major change but does not perform a numerical balancing.]

One utility stated that the draft Backfit Analysis complies with § 50.109 when non-quantitative factors such as "contemporary scientific and philosophic approaches" are considered. Other comments state that nonquantitative factors should not be controlling, but could be considered in the analysis. There was no support from these comments for suspending

the application of the provisions of § 50.109 to the Part 20 rulemaking. Two commenters stated that the pertinent wording of § 50.109(c) would provide for Commission approval based on qualitative considerations. One of these commenters stated that: "...qualitative factors are of sufficient importance to establish compliance of the Part 20 backfit with [§] 50.109."

# C. Final Backfit Analysis For The Part 20 Revision

The revision of 10 CFR Part 20 is not anticipated to require physical modification to nuclear power reactors. However, the definition of a "backfit" in § 50.109(a)(1) includes the modification of or addi ion to the procedures or organization required to design, construct or operate a nuclear power reactor facility. The Part 20 rule is applicable to all NRC licensees and, therefore, is broader in scope than the "Backfit Rule," which applies only to nuclear power reactors. However, the revision would result in the need for revisions in the operating procedures dealing with radiation protection at nuclear power reactor facilities licensed under 10 CFR Part 50 so the application of the backfit rule must be considered.

A separate Regulatory Analysis for the 10 CFR Part 20 revision describes the anticipated benefits and anticipated costs that could be associated with the implementation of the Part 20 revision, were it to be adopted. This Regulatory Analysis is the primary source of the estimates of the benefits and the impacts for nuclear power reactors described in this Final Backfit Analysis and is incorporated as part of it. The Regulatory Analysis also contains estimates of the costs and benefits from the Part 20 revision for classes of NRC-licensed activities other than power reactors.

As the provisions of § 50.109 for a backfit analysis apply only to nuclear power reactors, these other impacts are not discussed in this Backfit Analysis.

D. The Requirements of § 50.109(c)

Paragraph 50.109(c) of the backfit rule sets forth certain factors which, to the extent relevant information is available, are to be considered in a backfit analysis. These factors are:

1. Statement of Specific Objectives to be Achieved.

The revision of 10 CFR Part 20 is intended to:

- Update the quarter-century-old 10 CFR Part 20 to incorrecte advances in science and new concepts of radiation protection methodology and philosophy;
- Implement the 1987 Federal Radiation Guidance on occupational radiation protection;
- c. Implement the principal current dose-limiting recommendations of the International Commission on Radiological Protection (ICRP) and the National Council on Radiation Protection and Measurements (NCRP) for both workers and members of the general public;
- d. Update the limits on airborne radionuclide intakes, effluent releases and doses from inhaled or ingested radionuclides using recent metabolic models and dose factors; and
- e. Add requirements keeping radiation levels "as low as is reasonably achievable."

The principal benefits associated with the rule are primarily associated with changes to or new limits on doses to the embryo/ fetus and to workers and to members of the public.  General Description of the Actions to be Required of the Licensee or Applicant.

The principal new or additional actions that would be required of licensees by the proposed 10 CFR Part 20 revisions are to:

- a. Sum the dose from radiation external to the body and from radionuclides deposited in the body as appropriate based upon survey and monitoring requirements;
- Provide increased protection for the embryo/fetus when female workers declare themselves pregnant;
- Employ the latest ICRP limits on airborne radionuclide intakes, effluent releases and doses from inhaled or ingested radionuclides;
- d. Modify training guides, operating procedures, and manuals to incorporate the new concepts and requirements and provide retraining of employees on these concepts and their implementation; and
- e. Develop radiation protection programs.
- Change in the Risk to the Public from Accidental Off-Site Release of Radioactive Material.

There is no direct impact on risks associated with accidental releases of radioactive materials because the principal thrust of the Part 20 applies to normal operation.

4. Potential Impact on Radiological Exposure of Facility Employees.

The principal impact of the revision would be to assure significantly better and more up-to-date worker protection. The added protection results from the following:

- a. The limit for annual worker doses would be 5 rems total effective dose equivalent in a given year. This is the sum of the external deep dose equivalent and the internal committed effective dose equivalent. Currently, the limit for dose from external radiation workers is 3 rems per quarter (12 rems per year). An additional limit is 5 rems whole-body or higher organ doses from internal radionuclides inhaled or ingested. These dose limits are applied independently so that a worker could conceivably receive a total dose (external and internal) of 17 rems in a given year. However, the present Part 20 also provides that the worker's long-term average external dose does not exceed 5 rems per year (cumulative dose less than 5(N-18) where N = age).
- b. The occupational dose limit for extremilies would be reduced from 75 to 50 rems per year; however, the limits for the skin and the eye would be raised.
- c. A limit of 0.5 rem would be placed or the dose to the embryo/ fetus of a declared pregnant woman. There is currently no specific limit in the NRC regulations to protect the embryo/ fetus. The exposure of the embryo-fetus constitutes not only a cancer risk. but also involves the possibility of producing mental retardation.
- d. Allowable intakes of radionuclides would be based upon updated radiobiological, metabolic, and dosimetric data. For a number of radionuclides the intake limits would be lowered. However, for many radionuclides the allowable intake would be increased due to the new methods of dose evaluation. These increases are not expected to affect airborne concentration levels in current facilities which will probably continue to operate below the older lower limits.
- e. The total effective dose equivalent would be the limiting quantity rather than separately limiting external or internal doses. In certain circumstances, internal and external radiation doses would be added together rather than evaluating them separately as required by the present rule.
- f. Dose limits would be expressed as the sum of organ doses weighted by the comparative biological risk of cancer in the organ. This will provide a better characterization of the radiological risks to the body by taking into account radiations from radioactive material located in parts of the body other than the "critical organ."

 Installation and Continuing Costs, Including the Cost of Facility Downtime or the Cost of Construction Delays

There should be no costs associated with reactor downtime or construction delays. The Part 20 changes apply primarily to operational procedures and should cause no modifications in facility design or in shielding. The initial and annual costs associated with various provisions in the revision are discussed and analyzed in the Regulatory Analysis. The total estimated costs for nuclear power reactor licensees are \$30 million for initial procedure modification and new procedure implementation and \$4 million in additional costs per year thereafter.

 Potential Safety Impact of Changes in Plant or Operational Complexity, Including Relationships to Proposed and Existing Regulatory Requirements.

Any safety impacts and changes in plant complexity would be negligible, since the proposed rule should not entail changes in plant design. Some of the proposed changes could increase operational complexity. However, once the new procedures are fully implemented they are expected to become routine.

The impact of modifying operating procedures, manuals, and records would be minimized by a deferred implementation date that will provide a period during which licensees may develop the necessary new procedures, manuals, and records.

 The Estimated Resource Burden on the NRC and the Availability of These Resources.

Costs to the NRC would primarily be associated with the preparation of new regulatory guides for implementing the new procedures and revising existing regulatory guides, branch technical positions, and inspection procedures to reflect the Part 20 revisions. This effort entails the preparation of 10 new regulatory guides requiring 0.2 man-year per guide or 2.0 person-years total and approximately \$500K of technical support effort over a three-year period. Seven existing regulatory guides would require major revisions, resulting in an additional person-year of effort. This effort to develop the new regulatory guides necessary to implement the revised Part 20 was started in 1989.

It is estimated that approximately one staff-year would be required in the Office of Nuclear Reactor Regulation (NRR) to modify license conditions and technical specifications to comply with the proposed revision. The largest impact in NRC relevant to Part 50 licensees covered by § 50.109 would be in NRR and the Regional Offices to revise inspection procedures and to train inspectors on the new regulations and procedures. It is estimated that this would require about 2 person-years total (excluding materials licensees). These impacts would be spread over the implementation period. For this reason the Part 20 implementation should not have a major impact on NRC programs. Once the new procedures are in place, there should not be any significant resource expenditures above current levels.

1

 Potential Impact of Differences in Facility Type, Design, or Age on the Relevancy and Practicality of the Proposed Action.

Since the proposed revisions principally affect operating procedures rather than facility physical design, there should be no significant impact due to differences in facility type, design or age.

9. Are the Proposed Revisions Interim or Final and if Interim, What is the Justification for Imposing Them on an Interim Basis?

This is a final rule.

10. Paragraph 50.109(c) requires consideration of the priority and scheduling of the action under consideration in light of other regulatory activities. Implementation of the proposed revision of 10 CFR Part 20 should not significantly affect any other backfits or safetyrelated activities. In order to minimize the impact of the retraining and revisions of procedures, the proposed implementation period of the Part 20 revision extends to 1992. The changes required to implement the Part 20 revision would not conflict with and do not need to be further prioritized with respect to other artivities at nuclear power plants.

#### E. Providing a "Substantial" Increase in Public Health & Safety

For the reasons given below, the Commission has concluded that the revisions to Part 20 as applied to nuclear power reactors, provide a substantial increase in overall protection of public health and safety both for workers and for members of the general public. The Commission's conclusion rests on both quantitative and qualitative grounds. The quantitative grounds are discussed first.

- (1). There is a requirement (§ 20.101) that licensees have a radiation protection program that describes measures to keep radiation exposures "as low as is reasonably achievable" or "ALARA," to the extent practicable. This makes the practice of ALARA mandatory for worker protection (consistent with the ALARA requirements for effluents in Appendix 1 to Part 50) rather than hortatory as in the present § 20.1(c).
- (2). The new occupational dose limit for radiation workers (§ 20.201) is a "total effective dose equivalent" (TEDE) of 5 rems per year. This limit is placed upon the sum of the external dose (deep dose equivalent) and the committed effective dose resulting from internal retention of inhaled radionuclides. (Occupational radionuclide intakes from ingestion are not considered as there should not be intake of contaminated food or water on the job.)

The new annual limit supersedes dose limits that would have allowed up to 12 rems per year (3 rems per quarter) from external doses and an additional 5 rems internal dose for the whole-body or up to 30 rems for doses to individual organs. Under the existing Part 20, it would have been legally possible to receive 17 rems in 1 year from the combined 12-rem<sup>1</sup> external and 5-rem internal doses. This is limited in the revised Part 20 to a total (internal + external) effective dose equivalent of 5 rems per year.

In practice, industry's own efforts to ensure compliance with the dose limits and to keep radiation doses "as low as is reasonably achievable" results in actual worker exposures being much less than the legally allowable maximum. Industry also generally sets practical operating limits lower than NRC limits to ensure compliance with the latter. For nuclear power facilities, guidelines issued by the Institute for Nuclear Power Operations, which are generally a fraction of the NRC limits, provide additional incentives for power reactor licencees to keep exposures below the NRC limits. Most workers in nuclear power reactors receive doses that are only a fraction of the dose limits in 10 CFR Part 20. For example, the average dose to workers with measurable radiation exposure in nuclear power plants is about 0.4 rem, an order of magnitude below the 5-rem limit (1.25 rems per quarter), and about a factor of 25 below the 12-rem (3 rems per quarter) maximum dose limit.

- (3). The new limit for doses to members of the public is an explicit 100 millirems per year. This is reduced from an implicit 500 millirems (and corresponding organ dose limits) per year found in the current Part 20 which formed the bases for the concentration limits in the existing Appendix B. Table 2. This change represents a substantial reduction in the numerical dose limit; however, most coses to the public from nuclear power reactors are only a very small fraction of Part 20 limits. This is due in part to ALARA design guidelines (e.g., Appendix I to 10 CFR Part 50) and ALARA-based operating limits (e.g., EPA's Uranium Fuel Cycle Standards in 40 CFR Part 190), and the licensee's own efforts to keep below the limits in order to assure compliance and maintain to releases "as low as is reasonable achievable." As was the case for occupational protection, the finding of a "substantial improvement in public health and safety" is based upon what would be legally permissible under the revised Part 20 as compared with the limits in the old Part 20.
- (4). The revised Part 20 i ins an explicit limit on the dose to the embryo/fetus of pregi iemale workers in licensed facilities (§ 20.208). There is no corresponding limit in the present Part 20 specifically to protect the unborn. The explicit limitation of doses to a pregnant woman who formally declares her pregnancy to her employer (i.e., a "declared pregnant woman") represents the largest estimated health protection benefit from the revised Part 20. This benefit (protection of pregnant women at power reactors) is estimated to have an economic value of \$2.6 million (Regulatory Analysis, page xv).

In addition to the quantifiable costs and safety benefits given above, the Commission has identified four qualitative arguments for the Commission's conclusion that the revisions to Part 20 provide a substantial increase in overall protection of public health and safety. These qualitative arguments arguing

- (1). The need for updating the principles and technical basis underlying the regulation. The system of protection presently used in 10 CFR Part 20 is based on the system of radioactive dose limitation and technical data in use over a quarter of a century ago. There have been broad revisions in the concepts and approaches for protection against radiation and numerous updates to the technical parameters used for estimating radionuclide metabolism and dose. A number of those changes result in more restrictive concentration limits as changes in the technical knowledge indicates that previous limits should be modified in order to provide the same margin of safety.
- (2). Maintaining consistency with international radiation protection standards. The European Economic Community, the International Atomic Energy Agency, the International Labor Organization, the World Health Organization, and the OECD/Nuclear Energy Agency base their radiation protection codes on ICRP recommendations. If the U.S. does not implement the new standards, we would not be compatible with international organizations and a large segment of the nations with advanced nuclear technology. This could lead to inconsistent limits being applied in areas close to U.S. borders and confusion regarding the proper degree of safety protection that should be applied.

(3). Consistency of radiation protection requirements with current risk assessment methodologies. There is widespread and growing usage of the dose and risk-related assessment approach arising from the concepts in ICRP Publication No. 26 and the metabolic and dose parameters in ICRP-30 and its companion reports. This means that dose assessments as currently conducted in practice by the NRC staff are not consistent with the dose assessment models underlying the present Part 20. Agreement is desirable for the sake of internal consistency between the techniques used in risk and impact assessment and the methodology used to set standards against which the acceptability of these impacts is judged.

In addition, the new estimates of risk as documented in the 1988 report of the United Nations Scientific Committee on the Effects of Atomic Radiation (UNSCEAR, 1988) and the 1989 report of the National Research Council's Committee on the Biological Effects of Ionizing Radiation (BEIR V, 1989) cannot be readily adopted into the risk limitation procedures in the present Part 20 because they use the effective dose concept rather than the critical organ approach. (The revised Part 20 uses the effective dose approach for dose assessment.)

The reduction in the dose limits and the new embryo/fetus dose limit found in the revised Part 20 is in accord with the higher risks predicted by the BEIR V and UNSCEAR 88 reports. Continued use of the older values that no longer adequately reflect current scientific knowledge may not provide sufficient assurance that there is a consistent and defensible basis for health protection.

(4). The 1987 revised Federal radiation protection guidance on occupational protection. As provided for in Sec. 274(h) of the Atomic Energy Act, on January 20, 1987, the President approved revised guidance to Federal agencies on the formulation of standards for occupational radiation exposure. Members of the NRC staff participated in an interagency working group that developed the recommendations for the President. Although the Federal guidance does not have the force of a statutory requirement, the AEC and the NRC have, as a matter of policy, considered Federal guidance as binding and in the past have implemented prior guidance in their regulations. Not following the guidance would lead to inconsistent regulations among Federal agencies. This would be particularly confusing for the Federal agencies that are NRC licensees and would have to follow both the Federal guidance and the NRC regulations. Several DOD component agencies have already requested advice on implementing the 1987 Presidential guidance on occupational radiation exposure because of the differences between the guidance and the current Part 20. This confusion could have adverse impacts on safety.

#### F. Determination

For the reasons stated above, the Commission believes that the reductions in allowable dose limits that are embodied in the revised Part 20 constitute substantial increases in the protection of public health and safety. Although current practice, including the philosophy of keeping radiation exposures as low as is reasonably achievable (ALARA), generally has kept radiation exposures well below the existing limits, the reductions in the allowable dose limits ensure that such doses will also remain low in the future.

In addition to the quantifiable safety benefits accruing from dose reductions and other improvements in the revised Part 20, there are several qualitative factors (described above) which support issuing the Part 20 revision. One of the main qualitative factors is that it is necessary to revise the 30-year-old existing Part 20 to ensure that the NRC regulations reflect the current state of radiation protection science. Any future revisions in dose limits recommended by ICRP or NCRP would undoubtedly be based upon the 1977 ICRP and 1987 NCRP recommendations and, therefore, would be more easily incorporated into the framework of the revised Part 20 than in the framework of the current Part 20.

Based upon the conclusions that the revised Part 20 provides: (1) a substantial increase in public health and safety and (2) that, when the quantitative and qualitative safety benefits of the revision are considered, the costs of implementing the revised Part 20 are justified, the Commission finds that the requirements of the "Backfit Rule" (§ 50.109) are satisfied and that the Part 20 revision should be issued as final rule.

0

The Commission is adopting the final rule based on the conclusion of this analysis that the rule provides for a substantial increase in the overall protection of the public health and safety and that the direct and indirect costs of its implementation are justified in terms of the quantitative and

qualitative benefits associated with the rule. The Commission would note, however, that, even had the analysis not concluded that the revised Part 20 provides a substantial increase in the overall public health and safety, it could have gone forward with the rule because the changes made to Part 20 also amount to a redefinition of the level of adequate protection and the backfit rule's substantial increase and cost justification standards do not apply to a redefinition of adequate protection. ENCLOSURE F

Enclosure F



#### UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

The Honorable Bob Graham, Chairman Subcommittee on Nuclear Regulation Committee on Environment and Public Works United States Senate Washington, D.C. 20510

Dear Mr. Chairman:

Enclosed for the information of the Subcommittee are copies of a public announcement and a final rule revising the Commission's regulations for protection against radiation in 10 CFR Part 20. This rule implements the Federal radiation guidance issued by President Reagan in January 1987.

The rule will become effective 30 days after issuance in the <u>Federal Register</u>, but licensees will have until January 1, 1993 to come into compliance. The early implementation date will allow new licensees or license renewals to adopt the new regulation without having to adopt the current Part 20 for only a short time.

The rule has been modified from a proposed rule published for public comment in January 1986. Over 800 public comments were received and considered in preparing the final rule. The rule is consistent with the recommendations of both the National Council on Radiation Protection and Measurements and the International Commission on Radiological Protection.

Sincerely,

Eric S. Beckjord, Director Office of Nuclear Regulatory Research

cc: Sen. Alan K. Simpson



#### UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

The Honorable Morris K. Udall, Chairman Subcommittee on Energy and the Environment Committee on Interior and Insular Affairs United States House of Representatives Washington, D.C. 20515

Dear Mr. Chairman:

Enclosed for the information of the Subcommittee are copies of a public announcement and a final rule revising the Commission's regulations for protection against radiation in 10 CFR Part 20. This rule implements the Federal radiation guidance issued by President Reagan in January 1987.

The rule will become effective 30 days after issuance in the <u>Federal Register</u>, but licensees will have until January 1, 1993 to come into compliance. The early implementation date will allow new licensees or license renewals to adopt the new regulation without having to adopt the current Part 20 for only a short time.

The rule has been modified from a proposed rule published for public comment in January 1986. Over 800 public comments were received and considered in preparing the final rule. The rule is consistent with the recommendations of both the National Council on Radiation Protection and Measurements and the International Commission on Radiological Protection.

Sincerely,

Eric S. Beckjord, Director Office of Nuclear Regulatory Research

cc: Rep. James V. Hansen


## UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

The Honorable Philip R. Sharp, Chairman Subcommittee on Energy and Power Committee on Energy and Commerce United States House of Representatives Washington, D.C. 20515

Dear Mr. Chairman:

Enclosed for the information of the Subcommittee are copies of a public announcement and a final rule revising the Commission's regulations for protection against radiation in 10 CFR Part 20. Inis rule implements the Federal radiation guidance issued by President Reagan in January 1987.

The rule will become effective 30 days after issuance in the <u>Federal Register</u>, but licensees will have until January 1, 1993 to come into compliance. The early implementation date will allow new licensees or license renewals to adopt the new regulation without having to adopt the current Part 20 for only a short time.

The rule has been modified from a proposed rule published for public comment in January 1986. Over 800 public comments were received and considered in preparing the final rule. The rule is consistent with the recommendations of both the National Council on Radiation Protection and Measurements and the International Commission on Radiological Protection.

Sincerely,

Eric S. Beckjord, Director Office of Nuclear Regulatory Research

cc: Rep. Carlos J. Moorhead

ENCLOSURE G PRESS RELEASE

Enclosure G

2

## NUCLEAR REGULATORY COMMISSION AMENDS RADIATION PROTECTION REQUIREMENTS

The Nuclear Regulatory Commission is amending its regulations governing protection against radiation to provide for a substantial increase in the overall protection of the public health and safety.

The new requirements are based on those that were proposed for public comment in January 1986. They incorporate Federal guidance for radiation protection of workers in the nuclear industry issued by the President in 1987 and recommendations of the International Commission on Radiological Protection (ICRP) and the National Council on Radiation Protection and Measurements (NCRP).

Highlights of the new requirements are:

- Annual radiation exposures to individual members of the public from NRC-licensed activities are limited to an explicit 0.1 rem per year or 100 millirem compared with the previous implicit limit of 0.5 rem or 500 millirem;
- -- The use of quarterly and cumulative radiation dose limits for nuclear industry workers are eliminated and provisions are added to control the sum of internal and external occupational dose limits while maintaining the annual dose limit for radiation workers at a level equivalent to the present five rem per year;

- -- For the first time, a standard is established for protection of the embryo/fetus of women radiation workers which limits the exposure to 0.5 rem over the duration of the pregnancy, if the pregnancy has been made known to the employer;
- -- The occupational radiation dose for minors (individuals under the age of 18) is limited to 10 percent of the annual dose limits specified for adult workers;
- -- The annual average concentrations of radioactive material released in gaseous and liquid effluents at the boundary of an unrestricted area is revised to reflect up-to-date metabolic models and dose factors--some limits decrease, some increase and others remain the same; and
- All NRC licensees are required to implement programs to assure that all radiation doses are kept as low as is reasonably achievable (ALARA).

The revised requirements reflect the first complete revision of the NRC's radiation protection requirements since they were first promulgated in 1957 and are based on three basic radiation protection principles:

-- That there is a potential health risk proportional to the radiation dose received and that there is an incremental health risk associated with even very small doses, even radiation doses much smaller than received from naturally occurring radiation sources;

- -- That the severity of random health effects such as cancer and genetic effects is independent of dose and at low doses and dose rates there is considerable uncertainty about the risks of radiation exposure; and
- That non-random occurrences of radiation-induced health effects such as radiation-induced cataracts can be prevented by limiting exposures to doses that are below the known thresholds for induction of such effects.

The revised Part 20 and conforming amendments to Parts 19, 32, 34, 39, 50 and 70 of the NRC regulations will become effective 30 days after issuance in the <u>Federal Register</u>. Licensees will not be required to implement the new regulations until January 1, 1993. Each implementation may be beneficial to applicants for new licenses or renewal of existing NRC licenses so that they will not have to commit to and implement the existing 10 CFR Part 20 for only a short period of time before the revised Part 20 would replace it. The NRC staff is developing guidance documents that will provide more details on the methods of implementing the new rule.

ŧ

## DRAFT

TRANSMITTAL TO:	Document Control Desk, P1-24
ADVANCE COPY TO:	Public Document Room*
DATE: 12/13/90	
FROM: SECY, Operations Bra	nch of

Attached are copies of SECY papers and related documents. They are being forwarded for entry on the Daily Accession List and placement in the Public Document Room. No other distribution is requested or required.

1. Decy-90-387	11.
2	12.
3	13
4	14
5	15
6.	16
7	17
8	18
9.	19.
10.	20

X

X

X

JF02

11

\* PDR is advanced two copies of each SECY paper and one copy of each related document.